MARÍA FERREIRO SUBRIDO

CONFORMAL STRUCTURES AND SOLITONS IN PSEUDO-RIEMANNIAN GEOMETRY



UNIVERSIDADE DE SANTIAGO DE COMPOSTELA

MARÍA FERREIRO SUBRIDO

CONFORMAL STRUCTURES AND SOLITONS IN PSEUDO-RIEMANNIAN GEOMETRY

157Publicaciones
del2023Departamento
de Geometría y Topología

UNIVERSIDADE DE SANTIAGO DE COMPOSTELA

© Universidade de Santiago de Compostela, 2023



Esta obra atópase baixo unha licenza internacional Creative Commons BY-NC-ND 4.0. Calquera forma de reprodución, distribución, comunicación pública ou transformación desta obra non incluída na licenza Creative Commons BY-NC-ND 4.0 só pode ser realizada coa autorización expresa dos titulares, salvo excepción prevista pola lei. Pode acceder Vde. ao texto completo da licenza nesta ligazón: <u>https://creativecommons.org/licenses/by-ncnd/4.0/deed.gl</u>



Esta obra se encuentra bajo una licencia internacional Creative Commons BY-NC-ND 4.0. Cualquier forma de reproducción, distribución, comunicación pública o transformación de esta obra no incluida en la licencia Cretative Commons BY-NC-ND 4.0 solo puede ser realizada con la autorización expresa de los titulares, salvo excepción prevista por la ley. Puede Vd. acceder al texto completo de la licencia en este enlace: <u>https://creativecommons.org/licenses/by-ncnd/4.0/deed.es</u>



This work is licensed under a Creative Commons BY NC ND 4.0 international license. Any form of reproduction, distribution, public communication or transformation of this work not included under the Creative Commons BY-NC-ND 4.0 license can only be carried out with the express authorization of the proprietors, save where otherwise provided by the law. You can access the full text of the license at https://creativecommons.org/licenses/by-nc-nd/4.0/legalcode



Tese de Doutoramento

Conformal structures and solitons in pseudo-Riemannian geometry

María Ferreiro Subrido

ESCOLA DE DOUTORAMENTO INTERNACIONAL DA UNIVERSIDADE DE SANTIAGO DE COMPOSTELA PROGRAMA DE DOUTORAMENTO EN MATEMÁTICAS

> SANTIAGO DE COMPOSTELA ANO 2023

A presente tese foi dirixida por Eduardo García Río e Ramón Vázquez Lorenzo. Defendida na Universidade de Santiago de Compostela o 7 de xullo de 2023.

Os resultados presentados nesta memoria foron obtidos coa axuda do financiamento da Consellería de Cultura, Educación e Ordenación Universitaria da Xunta de Galicia, na modalidade de Grupo de Referencia Competitiva GRC2013-045, e coas axudas dos proxectos MTM2016-75897-P, ED431F2017/03 e ED431F 2020/04 (incluído cofinanciamento do FEDER) e da convocatoria de axudas predoutorais FPU19/00130.

"Sobrio no tengo la sabiduría"

- Sebastian Pena

Contents

Ag	radeo	cementos / Acknowledgements	K	
Ab	strac	t	I	
Introduction XI				
1	Preli 1.1 1.2 1.3 1.4 1.5 1.6	minaries Pseudo-Riemannian geometry Curvature functionals Affine and projective geometry Lie groups Homogeneous spaces A note on Gröbner bases	1 1 2 8 9 0	
Ι	Loc Kähl Para- Null-	cally conformally flat structures 34 er structures 37 Kähler structures 37 Kähler structures 34 Kähler structures 34	5 7 8 0	
2	Loca 2.1 2.2 2.3	Illy conformally flat four-dimensional structures 43 Locally conformally flat four-dimensional Kähler and para-Kähler structures 43 Locally symmetric Kähler surfaces 53 Locally conformally flat null-Kähler structures 64	3 3 8 0	
3	Four 3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8	-dimensional Kähler and para-Kähler Lie groups6Summary of results6Symplectic Lie algebras6Para-Kähler structures on $\mathfrak{r}_{4,0}$ 6Para-Kähler structures on $\mathfrak{r}_{4,-1}$ 7Para-Kähler structures on $\mathfrak{r}_{2}\mathfrak{r}_{2}$ 7Para-Kähler structures on $\mathfrak{r}_{3,0}$ 7Para-Kähler structures on $\mathfrak{r}_{3,0}$ 7Para-Kähler structures on $\mathfrak{r}_{3,-1}$ 7	3 37 92 37 89	

	3.9 Para-Kähler structures on \mathfrak{r}'_2	. 81
	3.10 Para-Kähler structures on $\mathfrak{r}_{4,-1,\beta}$. 83
	3.11 Para-Kähler structures on $\mathfrak{r}_{4,-1,-1}$.	. 85
	3.12 Para-Kähler structures on $\mathfrak{r}_{4,-\alpha,\alpha}$. 87
	3.13 Para-Kähler structures on \mathfrak{h}_4	. 89
	3.14 Para-Kähler structures on $\mathfrak{d}_{4,1}$. 90
	3.15 Para-Kähler structures on $\mathfrak{d}_{4,\frac{1}{2}}$. 92
	3.16 Para-Kähler structures on $\mathfrak{d}_{4,\lambda}^{3/2}$. 94
	3.17 Para-Kähler structures on $\mathfrak{d}_{4,2}$. 95
	3.18 Kähler Lie algebras	. 101
тт		105
11	Solitons associated to geometric nows	105
	The Ricci flow: Ricci solitons	. 107
	The Bach flow: Bach solitons	. 110
4	Ricci solitons on four-dimensional Lorentzian Lie groups	113
	4.1 Extensions of Lorentzian Lie groups	. 119
	4.2 Extensions of Riemannian Lie groups	. 139
	4.3 Extensions of degenerate Lie groups	. 141
	4.4 Left-invariant Ricci solitons on pp-wave Lie groups	. 149
5	Algebraic solitons on four-dimensional Riemannian Lie groups	153
	5.1 Algebraic I -solitons	. 153
	5.2 Algebraic solitons on $SL(2, \mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$. 162
	5.3 Algebraic solutions on $\mathbb{R}^6 \rtimes \mathbb{R}$. 168
	5.4 Algebraic solitons on $E(1, 1) \rtimes \mathbb{R}$ and $E(2) \rtimes \mathbb{R}$. 188
	5.5 Algebraic solitons on $H^{\circ} \rtimes \mathbb{R}$. 195
II	I Homogeneous manifolds and homogeneous structures	211
6	Homogeneous four-manifolds with half-harmonic Weyl curvature	215
U	6.1 Summary of results	215
	6.2 Half-barmonic Weyl curvature on homogeneous manifolds	213
	0.2 man-narmonie weyr curvature on nomogeneous mannolus	. 210
7		
•	Three-dimensional Riemannian homogeneous structures	229
	Three-dimensional Riemannian homogeneous structures7.1Homogeneous structures	229 . 229
	Three-dimensional Riemannian homogeneous structures7.1Homogeneous structures7.2Riemannian homogeneous structures in dimension three	229 . 229 . 230
	Three-dimensional Riemannian homogeneous structures7.1Homogeneous structures7.2Riemannian homogeneous structures in dimension three7.3Self-dual and anti-self-dual homogeneous structures	229 . 229 . 230 . 239
Co	Three-dimensional Riemannian homogeneous structures 7.1 Homogeneous structures 7.2 Riemannian homogeneous structures in dimension three 7.3 Self-dual and anti-self-dual homogeneous structures onclusions and open problems	 229 229 230 239 242

Agradecementos / Acknowledgements

Nestas liñas tratarei de expresar toda a gratitude que sinto cara aquelas persoas que me acompañaron, total ou parcialmente, neste camiño. Foron uns anos intensos en moitos sentidos, dos que me levo moitos grandes momentos e persoas estupendas.

En primeiro lugar, non podo máis que darlles miles de grazas aos meus directores de tese, Eduardo e Ramón, por todo o seu apoio e paciencia comigo. En particular, moitas grazas, Ramón, por animarte a dirixir a miña tese aínda que a anterior fose a última. Eduardo, grazas por todos os cafés, algún día tocará que empece a invitar eu. Estouvos moi agradecida aos dous por todas as horas de traballo que lle dedicastes a este proxecto, dentro e fóra do despacho. Non podía pedir uns directores mellores, nin no profesional nin no persoal.

Tamén me gustaría agradecerlle a Esteban a súa colaboración neste proxecto e a súa bondade de asinar o consentimento de reprodución dos traballos que temos en común nesta memoria polo módico prezo dunha caña.

Grazas, Elena, por todas as horas de docencia compartidas, os cotilleos e a túa confianza nestes anos.

I would also like to thank professor Maciej Dunajski for his kindness when I visited him at the University of Cambridge and for helping me live the "Cambridge experience" as much as possible. Gracias también a Pablito por haberme acogido tan bien en St. Regis y acompañarme en mi camino desde entonces, aunque yo sea muy emo y tú no sepas cecear. Thanks to all the other people I met in Cambridge who still walk with me today. You made my time there worthwhile. I am inordinately happy we met.

Non sei que tería sido destes anos sen todos os meus compañeiros de fatigas na facultade. Desde os primeiros que estivestes comigo na charcu, Rodri e David, ata todos os que fostes pasando por ela; Tomás, Diego e Juanma. Alégrome moito de que fósedes vós con quen me tocou vivir esta experiencia. Moitas grazas polos cafés de media mañá e media tarde e mediodía, polas noites de xogos, as cañas, as ceas, os congresos e os dramas compartidos.

Grazas, Olgui, por traer a alegría contigo nas túas visitas, aínda que non chegamos a cumprir (aínda) o de ir de congreso xuntas.

Graciñas, Suso, Espe, Fina, Oliva, Andrés, Carlos e demais persoal da cafetería, por proporcionarme combustible e recibirme sempre cun sorriso, incluso de boa mañá, nestes anos de tese e en todos os anteriores.

Moitas grazas tamén a todas as persoas alleas á facultade que estivestes aí para min. En especial, Vic, por todas as horas de "terapia" e cervexas compartidas e por animarme tanto nas últimas etapas desta tese.

Non me podo esquecer de agradecer a todas as miñas compañeiras da Asemblea de Inves-

tigadoras de Compostela o traballo enorme que hai detrás de todas as cousas que conseguimos xuntas en prol dunha USC máis digna. Sodes xente marabillosa.

Muchísimas gracias, Natalia, porque sin tu ayuda me habría puesto la vida mucho más difícil a mí misma (o no, no lo sabemos).

Por último, e ben lonxe de ser o menos importante, infinitas grazas, Bea e Sandro, por acollerme no voso ático de Lugho en primeiro e convertervos en familia desde entón; polas noites de sofá, polas risas, polas bágoas e polas bágoas de risa. Fostes o maior dos meus apoios estes case catro anos e, non importa onde rematemos cada un, para min sempre seredes casa.

Abstract

This thesis is divided into three distinct parts, each of which explores different aspects of mathematical structures. The first part focuses on the investigation of locally conformally flat structures on four-dimensional manifolds. More precisely, the study delves into the local conformal flatness of Kähler, para-Kähler, and null-Kähler manifolds, and provides a complete geometric description of four-dimensional para-Kähler Lie algebras.

Moving on to the second part, the research focuses on the analysis of solitons associated to both the Bach and Ricci flows. This part offers a complete classification of four-dimensional left-invariant Lorentzian Ricci solitons and Riemannian algebraic Bach and Ricci solitons in dimension four.

Finally, the third part covers the description of four-dimensional homogeneous Riemannian manifolds that have half-harmonic Weyl curvature tensor and those homogenous manifolds which admit more than one homogeneous structure in dimension free.

Introduction

Let (M, g) be a pseudo-Riemannian manifold and denote by [g] the conformal class of all the pseudo-Riemannian metrics which are conformal to g. A basic problem in conformal Riemannian geometry is the existence of distinguished structures – such as Einstein metrics, Kähler or para-Kähler structures, Ricci solitons, etc. – in the same conformal class of a given metric. In this Ph.D. thesis we examine this sort of problems and obtain some classification results under certain conditions.

The existence of flat metrics in the conformal class of a given metric is a very restrictive property. From a local perspective, this property is characterized by the vanishing of the Weyl curvature tensor in dimension four, which ensures the existence of locally defined functions $\sigma: \mathcal{U} \subset M \to \mathbb{R}$ such that $(\mathcal{U}, \overline{g} = e^{2\sigma}g)$ is flat. Granted that there exist bountiful examples of locally conformally flat metrics, the particular example of the *n*-dimensional sphere \mathbb{S}^n shows that local conformal flatness cannot always be extended to a global condition.

Our first objective has been to completely understand the local structure of four-dimensional Kähler and para-Kähler metrics that have at least one flat representative in their conformal classes. It was already known that in dimension greater than four the existence of such representatives is strictly restricted to flat manifolds, but the four-dimensional situation still remained an open problem. Patterson proved the existence of non-flat four-dimensional examples when the associated Ricci operators are two-step nilpotent in the case of split-signature Kähler metrics (see [119]). In Chapter 2 in this thesis we complete the description of these metrics by finding the existence of examples with complex Ricci operators.

Theorem 2.2 Any indecomposable locally conformally flat Kähler surface (M, g, J_+) is locally isometric to the cotangent bundle $(T^*\Sigma, g)$ of a Riemannian surface (Σ, g_{Σ}) of constant curvature with a metric given by

- (i) $g = g_{\nabla^{\Sigma}}$ if the Gaussian curvature is non-zero, or
- (ii) $g = \iota J_{\Sigma} \circ \iota \operatorname{Id} + g_{\nabla^{\Sigma}}$ if the Gaussian curvature vanishes,

where ∇^{Σ} is the Levi-Civita connection of (Σ, g_{Σ}) and J_{Σ} is the Kähler structure on Σ associated to the Riemannian volume form. Furthermore, the complex structure J_+ on $T^*\Sigma$ is determined by the symplectic form $\Omega_+ = -d\iota J_{\Sigma}$.

A somehow analogous result describes the locally conformally flat para-Kähler surfaces.

Theorem 2.1 Any indecomposable locally conformally flat para-Kähler surface (M, g, J_{-}) is locally isometric to the cotangent bundle $T^*\Sigma$ of a flat affine surface (Σ, D) with a para-complex

structure determined by $J|_{\ker \pi_*} = \mathrm{Id}$, where π denotes the canonical projection from the cotangent bundle, and the metric g is given by $g = \iota T \circ \iota \mathrm{Id} + g_D$ where

- (i) T is a parallel nilpotent (1, 1)-tensor field on (Σ, D) , or
- (ii) T is a parallel (1,1)-tensor field on (Σ, D) satisfying $T^2 = -\kappa^2$ Id.

Moreover, in both cases the para-Kähler two-form $\Omega_{-}(X,Y) = g(J_{-}X,Y)$ is the canonical symplectic form of the cotangent bundle.

Even though the local structures correspond to modified Riemannian extensions of affine surfaces in both situations, the affine connections on the base surfaces are essentially different. In the Kähler case, the extended connection is the Levi-Civita connection of an affine surface of constant Gaussian curvature, whereas in the para-Kähler situation the extended connection is flat, which makes it less restrictive than the complex situation. The Ricci operators associated to the metrics in Theorem 2.1-(*ii*) and Theorem 2.2-(*ii*) determine self-dual parallel complex structures on the manifolds, which induce anti-Kähler structures (see [21]) whose associated metrics have the same Levi-Civita connection than the original metrics.

Since the metrics in Theorems 2.1 and 2.2 are locally symmetric, we pondered the possibility of their realization as left-invariant metrics on Lie groups. Ovando had already determined all the – both Riemannian and split-signature – Kähler structures on four-dimensional Lie groups in her work [116]. Nevertheless, the complete description of the four-dimensional para-Kähler Lie groups still remained an open problem, despite the number of previous attempts to understand it [31, 32, 101]. Para-Kähler structures are determined by the symplectic Lie algebras that admit a decomposition $\mathfrak{g} = \mathfrak{L} \oplus \mathfrak{L}'$ as a direct sum of Lagrangian subalgebras, so the possibilities for the existence of such structures are more flexible than in the Kählerian case and, consequently, they are plentiful.

In Chapter 3 we give a complete description of the left-invariant para-Kähler metrics on fourdimensional Lie groups, which allows us to describe all the locally conformally flat para-Kähler metrics in terms of these structures (see Corollary 3.2). We give our description in terms of the curvature of the left-invariant metrics. In Theorem 3.1 we determine all the classes of symmetric para-Kähler Lie groups. The non-symmetric situation splits into the semi-symmetric case – that occurs only when the corresponding Ricci tensors vanish – and the non-semi-symmetric case – which corresponds to four families among which we found some of the 3-symmetric spaces – as stated in Theorems 3.6 and 3.7. As a consequence of our study, we obtained an alternative description of the hypersymplectic Lie groups previously obtained by Andrada [5].

The existence of Einstein metrics in the conformal class of a given metric implies the existence of an underlying C-space structure, i.e., the existence of a vector field X on the manifold such that

$$\operatorname{div} W + \iota_X W = 0.$$

When the vector field X is the gradient of a function, this equation is equivalent to the existence of a conformal metric \overline{g} for which $\overline{\operatorname{div} W} = 0$, since the Weyl curvature of an Einstein manifold is harmonic – i.e., divergence-free. On four-dimensional oriented manifolds, the decomposition of the Weyl curvature tensor into its self-dual and anti-self-dual components $W = W^+ + W^-$, which is conformally invariant, makes the condition $\operatorname{div} W = 0$ become $\operatorname{div} W^+ + \operatorname{div} W^- = 0$. In this way, a four-dimensional oriented manifold has half-harmonic Weyl curvature if either one of the two summands in the previous equation vanishes. It is important to emphasize that this condition is not conformally invariant, since

$$\overline{\operatorname{div} W}^+ = \operatorname{div} W^+ - \iota_{\nabla_{\sigma}} W^+$$

for a conformal metric $\overline{g} = e^{2\sigma}g$. Four-dimensional homogeneous Riemannian manifolds have harmonic Weyl curvature tensor if and only if they are symmetric [123]. In Chapter 6 we give the complete description of the four-dimensional homogeneous spaces that have half-harmonic Weyl curvature, as stated in the main result of the chapter.

Theorem 6.1 Let (M, g) be a four-dimensional locally homogeneous Riemannian manifold with half-harmonic Weyl curvature tensor. Then it is symmetric or locally homothetic to one of the following semi-direct extensions of the Heisenberg group.

(i) The left-invariant metric on $H^3 \rtimes \mathbb{R}$ determined by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = -e_1, \quad [e_2, e_4] = -e_2, \quad [e_3, e_4] = -2e_3,$$

(ii) or the left-invariant metric on $H^3 \rtimes \mathbb{R}$ determined by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = \frac{1}{2}e_1, \quad [e_2, e_4] = -e_2, \quad [e_3, e_4] = -\frac{1}{2}e_3,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis of $\mathfrak{h}_3 \rtimes \mathfrak{r}$.

It is important to emphasize that the Weyl curvature tensor of a locally symmetric manifold is parallel and so it is divergence-free. Besides, even though orientation does not play a relevant role in Theorem 6.1, it does when additional structures on the manifold are considered. A Kähler surface satisfies div $W^- = 0$ if and only if it is weakly Bochner-flat, in which case it is locally symmetric when its scalar curvature is constant (see [89]). Furthermore, since the self-dual Weyl curvature of an oriented four-dimensional Kähler metric takes the form

$$W^+ = \frac{\tau}{12} \operatorname{diag}[2, -1, -1],$$

then $\tau \operatorname{div}_1 W^+ + \iota_{\nabla \tau} W^+ = 0$ and so $\operatorname{div}_1 W^+ = 0$ if the scalar curvature is a non-zero constant. If the scalar curvature is not constant, then it determines a conformal metric whose Weyl curvature tensor is half-harmonic [56,87]. The only non-symmetric homogeneous Kähler metric is the 3-symmetric space, which is necessarily half-harmonic, and it corresponds to Theorem 6.1-(*ii*).

The Lie groups given in Theorem 6.1 correspond to homogeneous manifolds that admit selfdual homogeneous structures. Generalizing Cartan's characterization of symmetric spaces as those whose curvature tensors are parallel, Ambrose and Singer [3] considered homogeneous structures on a Riemannian manifold (M, g) as tensor fields of type (1, 2) for which the connection $\tilde{\nabla} = \nabla - T$ makes the metric g, its associated curvature tensor R and the tensor field Tparallel. In this way, a complete and simply connected Riemannian manifold is homogeneous if it admits a homogeneous structure. Sekigawa proved in [126] that any three-dimensional simply connected homogeneous Riemannian manifold is either symmetric or isometric to a Lie group endowed with a left-invariant metric $(G, \langle \cdot, \cdot \rangle)$. Lie groups are trivially homogeneous, just considering the action of G on itself determined by the left translations. Therefore, all of them have a natural homogeneous structure, called the *canonical homogeneous structure* of the Riemannian Lie group.

Given the fact that a homogeneous manifold may admit different presentations as a homogeneous space, it may as well admit more than one homogeneous structure. Therefore, we decided to try and determine all the three-dimensional Riemannian homogeneous spaces that admit more than one homogeneous structure, and saw that in the non-symmetric situation this occurs only when their isometry groups are four-dimensional.

Theorem 7.1 A non-symmetric simply connected three-dimensional Riemannian Lie group admits a homogeneous structure different from the canonical one if and only if it admits a naturally reductive homogeneous structure. Moreover, in such a case, it admits exactly a one-parameter family of homogeneous structures.

In Theorems 7.2 and 7.3 we also determine all the possible homogeneous structures on threedimensional Riemannian Lie groups.

Since the four-dimensional homogeneous manifolds that have half-harmonic Weyl curvature admit self-dual homogeneous structures, we started the study of such structures in Section 7.3. Even though there already are classification results under certain additional conditions [109], the complete description of four-dimensional self-dual homogeneous structures seems to be an arduous problem that we only sketch in the mentioned section.

The Weyl curvature tensor of any locally conformally flat manifold vanishes and so the functional

$$g\mapsto \int_M \|W_g\|^2 dvol_g$$

reaches a minimum on every such manifold. This functional is conformally invariant in the four-dimensional situation and its gradient is determined by the Bach tensor

$$\mathfrak{B} = \operatorname{div}_2 \operatorname{div}_4 W + \frac{1}{2} W[\rho].$$

Therefore, Bach-flat four-dimensional manifolds are a natural generalization of locally conformally flat manifolds. The fact that four-dimensional both conformally Einstein and (anti-)selfdual manifolds are also Bach-flat places more importance on the study of such manifolds. The Bach flow

$$\frac{\partial}{\partial t}g_t = \mathfrak{B}_{g_t} + \frac{1}{12}(\Delta \tau_{g_t})g_t$$

has been intensively studied recently with the intention of improving the behaviour of a given metric in terms of its Bach tensor and obtaining Bach-flat metrics at the limit.

The existence of solitons associated to this flow has been studied in the context of product homogeneous manifolds and under the additional hypothesis of the soliton being a gradient. In Chapter 5 we tackle the study of such solitons from the algebraic point of view. A Riemannian Lie group $(G, \langle \cdot, \cdot \rangle)$ is an algebraic Bach soliton if and only if

$$\mathfrak{D} = \widehat{\mathfrak{B}} - \mu \mathrm{Id}$$

is a derivation of the Lie algebra of G, in which case it gives rise to a Bach soliton or, equivalently, to a self-similar solution $g_t = \sigma(t)\psi_t^* \langle \cdot, \cdot \rangle$, where ψ_t is one-parameter group of automorphisms of G.

Since every Einstein metric is Bach-flat, one might expect Ricci solitons to correspond to Bach solitons. This is actually the situation at the algebraic level, although there are algebraic Bach solitons which are not Ricci solitons. The following result provides a complete description of four-dimensional Riemannian algebraic Bach solitons.

Theorem 5.8 A four-dimensional simply connected Riemannian Lie group is an algebraic Bach soliton if and only if it is Bach-flat, an algebraic Ricci soliton or homothetic to one of the following Lie groups.

(i) The product Lie group $SU(2) \times \mathbb{R}$ with the product left-invariant metric determined by

$$[e_1, e_2] = 4e_3, \quad [e_1, e_3] = -4e_2, \quad [e_2, e_3] = e_1.$$

(ii) The semi-direct extension $H^3 \rtimes \mathbb{R}$ with the left-invariant metric determined by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = ae_1, \quad [e_2, e_4] = \frac{1}{a}e_2, \quad [e_3, e_4] = \frac{a^2+1}{a}e_3$$

for
$$a \in (0, 1)$$
.

Here $\{e_1, e_2, e_3, e_4\}$ *denotes an orthonormal basis of the corresponding Lie algebra.*

The Bach soliton obtained from Assertion (*i*) is a gradient soliton on $\mathbb{S}^3 \times \mathbb{R}$, where the metric on \mathbb{S}^3 is not the round metric, but a Berger one. On the other hand, the family of metrics in Assertion (*ii*) does not give rise to gradient solitons.

The four-dimensional algebraic Ricci solitons were determined – up to isomorphisms – by Lauret in [98]. In Theorem 5.4 we give a description of such solitons up to homotheties, which makes the description shorter and more manageable. The techniques developed in Chapter 5 are based on the analysis of general algebraic T-solitons for a geometric flow $\partial_t g_t = T_{g_t}$ given by an isometrically invariant symmetric, divergence-free (0, 2)-tensor field. This general approach led to unexpected simplifications which enable us to consider more complicated geometric flows.

Homogeneous Ricci solitons are critical for some curvature quadratic functional

$$g \mapsto \int_M \left\{ \|\rho\|^2 + t\tau^2 \right\} dvol_g$$

with zero energy in dimensions three and four (see [23]) and Riemannian signature. These solitons are algebraic or gradient solitons. In the same way, homogeneous gradient or algebraic Lorentzian Ricci solitons are critical for some curvature quadratic functional with zero energy in dimension three. Nevertheless, and in sharp contrast with the Riemannian situation, the Lorentzian signature allows the existence of left-invariant Ricci solitons on Lie groups. Such solitons had already been classified in the three-dimensional situation [24], and we tackle the four-dimensional problem in Chapter 4 of this thesis. The situation is far more complicated than the three-dimensional one and we obtained the following result.

Theorem 4.2 *A non-symmetric four-dimensional Lorentzian Lie group which is not a pp-wave is a non-trivial left-invariant Ricci soliton if and only if it is homothetic to one of the following:*

(i) $G_{\alpha} = \mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_4] = \alpha e_1, \ [e_2, e_4] = \varepsilon \left(1 - \frac{\alpha^2}{2}\right)^{\frac{1}{2}} e_2 - e_3, \ [e_3, e_4] = e_2 + \varepsilon \left(1 - \frac{\alpha^2}{2}\right)^{\frac{1}{2}} e_3,$$

where the parameter satisfies $0 \le \alpha \le \sqrt{2}$ and $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis with timelike e_3 . If $\alpha = 0$, then $\varepsilon = 1$. If $0 < \alpha < \sqrt{2}$, then $\varepsilon^2 = 1$. In the latter case, $\alpha \ne \frac{2\sqrt{3}}{3}$ when $\varepsilon = -1$.

(ii) $G_{\alpha} = \mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

 $[u_1, u_4] = \alpha u_1, \quad [u_2, u_4] = -\alpha u_2 + u_3, \quad [u_3, u_4] = u_1, \qquad \alpha > 0,$

where $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis for which the non-zero inner products are $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$.

(iii) $G = E(1,1) \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_2, e_4] = -[e_1, e_2] = e_2, \quad [e_1, e_3] = [e_3, e_4] = \frac{1}{2}[e_1, e_4] = e_3,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis with timelike e_3 .

(iv) $G_{\alpha\beta} = E(1,1) \rtimes \mathbb{R}$ with Lie algebra given by

$$[u_1, u_2] = u_1, \quad [u_1, u_4] = -2\alpha(\alpha\beta + 1)u_1, \quad [u_2, u_3] = u_3,$$
$$[u_2, u_4] = \beta u_1, \quad [u_3, u_4] = \alpha u_3,$$

where $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis for which the non-zero inner products are $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$, and the parameters $\alpha > 0$ and $\alpha \beta \notin \{-2, -1, -\frac{1}{2}\}$.

The metrics in Assertions (*i*) and (*iii*) have complex Ricci curvatures, while those of the remaining two are real. Besides, the metrics in Assertions (*i*), (*ii*) and (*iv*) are critical for some curvature quadratic functionals, whereas the metrics in Theorem 4.2-(*iii*) never are.

The case of left-invariant Ricci solitons which are pp-wave and plane wave Lie groups are described in Theorems 4.9 and 4.11, respectively.

The outline of this thesis

In Chapter 1 the reader can find the preliminary concepts that will be necessary for the complete understanding of the contents of this thesis, which is divided in three distinguished parts.

Part I is devoted to the study of locally conformally flat structures. In particular, in Chapter 2 we give a complete description of four-dimensional locally conformally flat Kähler, para-Kähler and null-Kähler structures. In Chapter 3 we study the realization of the para-Kähler families

obtained in the previous chapter as left-invariant metrics on Lie groups and give a complete geometric description of the four-dimensional para-Kähler Lie algebras.

In Part II we study the solitons associated to two particular geometric flows: the Bach flow and the Ricci flow. In Chapter 4 we give a complete description of the four-dimensional leftinvariant Lorentzian Ricci solitons and Chapter 5 is devoted to the study of the algebraic situation in Riemannian signature. In this chapter we introduce a general technique to describe four-dimensional Riemannian algebraic solitons associated to the flow determined by a generic symmetric and divergence-free (0, 2)-tensor field and use it to describe all the algebraic Bach and Ricci solitons in dimension four.

Part III covers Chapters 6 and 7 of this thesis. In Chapter 6 we give a complete description of the four-dimensional homogeneous Riemannian manifolds that have half-harmonic Weyl curvature. In Chapter 7 we determine all the non-symmetric three-dimensional Riemannian homogeneous spaces that admit more than one homogeneous structure. Besides, motivated by the fact that the metrics obtained in Chapter 6 admit self-dual homogeneous structures, we began to study these sort of structures in dimension four and the reader can find a sketch of this problem in Section 7.3. This remains an open problem for which we have obtained only partial results so far.

Chapter 1 Preliminaries

In this chapter we will introduce the concepts and notations that will be necessary for the complete understanding of this thesis. We will omit most of the proofs and remit the reader to different bibliographic references for more details.

1.1 Pseudo-Riemannian geometry

1.1.1 Pseudo-Riemannian manifolds

An *n*-dimensional *pseudo-Riemannian manifold* is a pair (M, g) where *M* is a smooth manifold and *g* is a metric tensor, i.e., a symmetric and non-degenerate (0, 2)-tensor field on *M*. The signature of the metric *g* is the pair $(n - \nu, \nu)$ such that $n - \nu$ and ν are the number of negative and positive eigenvalues of its associated matrix, respectively. Let (M, g) be an *n*-dimensional pseudo-Riemannian manifold. It is said to be *Riemannian* if its signature is (0, n) and *Lorentzian* if its signature is (1, n - 1). Moreover, if *M* is even-dimensional and the signature of *g* is $(\frac{n}{2}, \frac{n}{2})$ then the manifold has neutral (or split) signature.

We will denote by TM and T^*M the tangent and cotangent bundles of the manifold M and by $\mathfrak{X}(M)$ the space of vector fields which are tangent to M. We will use capital letters to denote vector fields and small letters to denote tangent vectors at a given point. Given a non-zero vector $v \in T_pM$ tangent to M at a point $p \in M$, it is said to be *timelike* if g(v, v) < 0, spacelike if g(v, v) > 0 and null or lightlike if g(v, v) = 0.

For any pseudo-Riemannian manifold (M, g) there exists a unique adapted linear connection ∇ which is torsion-free and parallel, i.e., such that

$$\nabla_X Y - \nabla_Y X - [X, Y] = 0$$
 and $\nabla g = 0$.

This connection is known as the *Levi-Civita connection* of the pseudo-Riemannian manifold and it is characterized by the Koszul formula

$$2g (\nabla_X Y, Z) = Xg (Y, Z) + Yg (X, Z) - Zg (X, Y) - g (X, [Y, Z]) - g (Y, [X, Z]) + g (Z, [X, Y]),$$

where $X, Y, Z \in \mathfrak{X}(M)$ and $[\cdot, \cdot]$ denotes the Lie bracket. The Levi-Civita connection can also be described by means of the *Christoffel symbols*. Let (x^1, \ldots, x^n) be local coordinates on M. The Christoffel symbols of the *first kind* are given by

$$\Gamma_{ij\ell} = \frac{1}{2} \left(\frac{\partial g_{\ell j}}{\partial x^i} + \frac{\partial g_{\ell i}}{\partial x^j} - \frac{\partial g_{ij}}{\partial x^\ell} \right)$$

and so the Christoffel symbols of the second kind are

$$\Gamma_{ij}{}^k = g^{k\ell} \Gamma_{ij\ell},$$

being (g^{ij}) the inverse matrix of (g_{ij}) . Therefore the Levi-Civita connection can be written in coordinates as

$$abla_{\partial_{x^i}}\partial_{x^j} = \Gamma_{ij}{}^k\partial_{x^k}$$

where $\partial_{x^i} := \frac{\partial}{\partial x^i}$ denote the locally defined coordinate vector fields.

1.1.2 The curvature tensor

In terms of the Levi-Civita connection we can define the *curvature operator*, or curvature tensor of type (1,3) by the convention

$$R(X,Y) Z = \nabla_{[X,Y]} Z - [\nabla_X, \nabla_Y] Z.$$

Considering local coordinates (x^1, \ldots, x^n) on M, the components of the curvature operator are given by $R(\partial_{x^i}, \partial_{x^j}) \partial_{x^k} = R_{ijk}^{\ell} \partial_{x^{\ell}}$. We can obtain the *curvature tensor* of type (0, 4) by lowering indices in the previous expression

$$R(X, Y, Z, V) = g(R(X, Y)Z, V),$$

so that its components are given by $R_{ijk\ell} = g_{\ell r} R_{ijk}^r$. Moreover, the curvature tensor has the following algebraic properties

(i)
$$R(X, Y, Z, V) = -R(Y, X, Z, V) = -R(X, Y, V, Z),$$

(ii) $R(X, Y, Z, V) + R(Y, Z, X, V) + R(Z, X, Y, V) = 0,$ (1.1)
(iii) $R(X, Y, Z, V) = R(Z, V, X, Y),$

and the differential identity

$$(iv) \quad (\nabla_X R) (Y, Z, U, V) + (\nabla_Y R) (Z, X, U, V) + (\nabla_Z R) (X, Y, U, V) = 0$$

Identities (ii) and (iv) are known as the first and second *Bianchi identities*, respectively.

A (0, 4)-tensor field $A: \mathcal{V} \times \mathcal{V} \times \mathcal{V} \to \mathbb{R}$ on a vector space \mathcal{V} satisfying identities (1.1) is said to be an *algebraic curvature tensor*.

The *sectional curvature* of a given Riemannian manifold (M, g) is the real function K define on the Grassmannian of 2-planes by

$$K(\Pi) = \frac{R(X, Y, X, Y)}{g(X, X)g(Y, Y) - g(X, Y)^2},$$

where $\Pi = \operatorname{span}\{X, Y\}$ is a two-dimensional subspace of T_pM . In the pseudo-Riemannian case we must consider the restriction to the Grassmannian of non-degenerate planes, i.e., those planes such that

$$g(X, X) g(Y, Y) - g(X, Y)^2 \neq 0.$$

If $K(\Pi)$ is independent of the plane Π , then the curvature tensor is given by

$$R(X, Y, Z, V) = KR^0(X, Y, Z, V),$$

where R^0 is the standard algebraic curvature tensor given by

$$R^{0}(X, Y, Z, V) = g(X, Z)g(Y, V) - g(X, V)g(Y, Z).$$
(1.2)

In dimension greater than two, if M is connected then the second Bianchi identity implies that if K is pointwise constant, then it is necessarily a global constant. A pseudo-Riemannian manifold has constant sectional curvature K if and only if its curvature tensor can be written as

$$R(X,Y)Z = K \left\{ g(X,Z)Y - g(Y,Z)X \right\},\,$$

in which case the manifold is locally isometric to a pseudo-sphere \mathbb{S}_{ν}^{n} (when K > 0), to a pseudo-Euclidean space \mathbb{E}_{ν}^{n} (when K = 0) or to a pseudo-hyperbolic space \mathbb{H}_{ν}^{n} (when K < 0). We refer to O'Neil's book [112] for more details on this topic.

We will denote by ρ the *Ricci tensor*, which is defined as the second trace of the curvature operator,

$$\rho(X,Y) = \operatorname{tr}\left(Z \mapsto R\left(X,Z\right)Y\right)$$

and its associated (1, 1)-tensor field, known as the *Ricci operator*, is characterized by

$$g(\operatorname{Ric}(X), Y) = \rho(X, Y).$$

The *scalar curvature* is defined as the trace of the Ricci operator $\tau = tr(Ric)$. It follows from the curvature identities (1.1) that the Ricci tensor is symmetric and, equivalently, the Ricci operator is self-adjoint. The Ricci tensor and the scalar curvature can be expressed in coordinates as

$$\rho_{ij} = g^{r\ell} R_{irj\ell}, \quad \tau = g^{ij} \rho_{ij}$$

Any two-dimensional pseudo-Riemannian manifold satisfies $\rho = \frac{\tau}{2}g$. A pseudo-Riemannian manifold of dimension greater than two is said to be an *Einstein space* if its Ricci tensor is a constant multiple of the metric, $\rho = \lambda g$. Tracing on the previous expression one sees that

$$\rho = \frac{\tau}{n}g,\tag{1.3}$$

and if M is connected, then the second Bianchi identity leads to the constancy of τ . In dimension three, satisfying the Einstein condition (1.3) is equivalent to having constant sectional curvature, while in dimension four there exist Einstein metrics which are not of constant sectional curvature. The four-dimensional case then appears as the first non-trivial case for consideration.

1

1.1.3 The Weyl tensor

The Schouten tensor of an algebraic curvature tensor A on an n-dimensional inner product vector space $(\mathcal{V}, \langle \cdot, \cdot \rangle)$ is the symmetric (0, 2)-tensor field defined as

$$\mathfrak{S}_A = \frac{1}{n-2} \left(\rho_A - \frac{\tau_A}{2(n-1)} \langle \cdot, \cdot \rangle \right),$$

where ρ_A and τ_A are the Ricci tensor and the scalar curvature associated to the algebraic curvature tensor A.

Let D and B be two symmetric bilinear forms on a vector space \mathcal{V} . Their *Kulkarni-Nomizu* product $D \odot B$ is the (0, 4)-tensor field on \mathcal{V} defined as

$$\begin{split} (D\odot B)\left(x,y,z,v\right) &= D(x,z)B(y,v) + D(y,v)B(x,z) \\ &\quad - D(x,v)B(y,z) - D(y,z)B(x,v), \end{split}$$

for $x, y, z, v \in \mathcal{V}$. It is easy to check that $D \odot B$ is an algebraic curvature tensor on $(\mathcal{V}, \langle \cdot, \cdot \rangle)$. For example, the standard curvature tensor $R^0 = \frac{1}{2} \langle \cdot, \cdot \rangle \odot \langle \cdot, \cdot \rangle$.

The Weyl curvature tensor arises from the Kulkarni-Nomizu product of the Schouten tensor and the metric tensor as $W_A = A - \mathfrak{S}_A \odot \langle \cdot, \cdot \rangle$. Therefore, the Weyl curvature tensor of a pseudo-Riemannian manifold (M, g) is defined as

$$W = R - \mathfrak{S} \odot g,$$

which can be written at each point $p \in M$ as

$$W(x, y, z, v) = R(x, y, z, v) + \frac{\tau}{(n-1)(n-2)} \{g(x, z)g(y, v) - g(x, v)g(y, z)\} - \frac{1}{n-2} \{\rho(x, z)g(y, v) - \rho(x, v)g(x, z) + \rho(y, v)g(x, z) - \rho(y, z)g(x, v)\},\$$

for all $x, y, z, v \in T_p M$. An important property of the Weyl curvature tensor is that it is trace-free and in dimension three it vanishes identically.

Local conformal flatness

A pseudo-Riemannian manifold (M, g) is said to be *locally conformally flat* if for every point $p \in M$ there exists an open neighbourhood \mathcal{U} of p and a smooth function $\sigma : \mathcal{U} \to \mathbb{R}$ such that the metric $\overline{g} = e^{2\sigma}g$ is flat.

The vanishing of the Weyl curvature tensor characterizes the locally conformally flat spaces in dimension greater than three. In the three-dimensional case, local conformal flatness is characterized by the total symmetry of the covariant derivative of the Schouten tensor, which means that it must be such that $(\nabla_X \mathfrak{S})(Y, Z) = (\nabla_Y \mathfrak{S})(X, Z)$ (see [99]).

In local differential geometry, the most important invariant of a conformal structure is given by the conformal Weyl curvature tensor of type (0, 4), which satisfies $\overline{W} = e^{2\sigma}W$ for any two conformally related metrics $\overline{g} = e^{2\sigma}g$.

1.1.4 Curvature decomposition

Consider for now an *n*-dimensional real vector space \mathcal{V} with basis $\{e_1, \ldots, e_n\}$ and let $\langle \cdot, \cdot \rangle$ be an inner product on \mathcal{V} . A *bivector* on \mathcal{V} is an element of the form $\sum_{i,j=1}^n a_{ij}e_i \wedge e_j$, where $a_{ij} \in \mathbb{R}$. The set of all bivectors is known as the bivector space and it is usually denoted by $\Lambda^2 \mathcal{V}$. The bivector space has the following properties:

(i)
$$e_i \wedge e_j = -e_j \wedge e_i$$
 and $e_i \wedge e_i = 0$ for all $i, j \in \{1, \ldots, n\}$.

(*ii*) The set

$$\{e_1 \wedge e_2, \ldots, e_1 \wedge e_n, e_2 \wedge e_3, \ldots, e_2 \wedge e_n, \ldots, e_{n-1} \wedge e_n\}$$

is a basis of $\Lambda^2 \mathcal{V}$.

Therefore $\Lambda^2 \mathcal{V}$ is an $\frac{n(n-1)}{2}$ -dimensional vector space. We define the *wedge product* of two elements $x = x^i e_i$ and $y = y^j e_j$ in \mathcal{V} by

$$x \wedge y = x^i e_i \wedge y^j e_j = \sum_{i < j} \left(x^i y^j - x^j y^i \right) e_i \wedge e_j \in \Lambda^2 \mathcal{V}.$$

The inner product on \mathcal{V} naturally extends to an inner product $\langle\!\langle \cdot, \cdot \rangle\!\rangle$ on the bivector space as [95]

$$\langle\!\langle x \wedge y, z \wedge v \rangle\!\rangle = \langle x, z \rangle \langle y, v \rangle - \langle x, v \rangle \langle y, z \rangle \tag{1.4}$$

and if $\{e_i : i = 1, ..., n\}$ is a $\langle \cdot, \cdot \rangle$ -orthonormal basis of \mathcal{V} , then $\{e_i \land e_j : i < j\}$ is an orthonormal basis of $\Lambda^2 \mathcal{V}$ with respect to $\langle\!\langle \cdot, \cdot \rangle\!\rangle$.

Given an algebraic curvature tensor A on $(\mathcal{V}, \langle \cdot, \cdot \rangle)$, it induces a unique self-adjoint endomorphism $\mathcal{A} \colon \Lambda^2 \mathcal{V} \to \Lambda^2 \mathcal{V}$ determined by

$$\langle\!\langle \mathcal{A}(x \wedge y), z \wedge v \rangle\!\rangle = A(x, y, z, v) \text{ for all } x, y, z, v \in \mathcal{V}.$$

The converse is not necessarily true, since a given self-adjoint endomorphism of $\Lambda^2 \mathcal{V}$ may not satisfy the first Bianchi identity, but there exists a bijective correspondence between the set of algebraic curvature tensors on $(\mathcal{V}, \langle \cdot, \cdot \rangle)$ and the set of self-adjoint endomorphisms of $(\Lambda^2 \mathcal{V}, \langle\!\!\langle \cdot, \cdot \rangle\!\!\rangle)$ satisfying

$$\langle\!\langle \mathcal{A}(x \wedge y), z \wedge v \rangle\!\rangle + \langle\!\langle \mathcal{A}(y \wedge z), x \wedge v \rangle\!\rangle + \langle\!\langle \mathcal{A}(z \wedge x), y \wedge v \rangle\!\rangle = 0.$$

In particular, the standard curvature tensor R^0 corresponds to the endomorphism $\mathcal{R}^0 = \mathrm{Id}_{\Lambda^2}$.

The following result provides a decomposition of all the algebraic curvature tensors which, in turn, motivates the tensors introduced above.

Theorem 1.1 ([95]). Any algebraic curvature tensor A on an inner product vector space $(\mathcal{V}, \langle \cdot, \cdot \rangle)$ decomposes as

$$A = \mathfrak{U}_A + \mathfrak{Z}_A + W_A,$$

1

where

$$\mathfrak{U}_{A} = \frac{\tau_{A}}{2n(n-1)} \langle \cdot, \cdot \rangle \odot \langle \cdot, \cdot \rangle, \qquad \mathfrak{Z}_{A} = \frac{1}{n-2} \left(\rho_{A} - \frac{\tau_{A}}{n} \langle \cdot, \cdot \rangle \right) \odot \langle \cdot, \cdot \rangle$$
$$W_{A} = A - \mathfrak{U}_{A} - \mathfrak{Z}_{A} = A - \mathfrak{S}_{A} \odot \langle \cdot, \cdot \rangle,$$

being ρ_A , τ_A and \mathfrak{S}_A the Ricci tensor, the scalar curvature and the Schouten tensor of the algebraic curvature tensor A, respectively.

The components \mathfrak{U}_A , \mathfrak{Z}_A and W_A in Theorem 1.1 correspond to the following orthogonal components:

- \mathfrak{U}_A is the orthogonal projection on the space of algebraic curvature tensors of constant sectional curvature.
- The vanishing of \mathfrak{Z}_A corresponds to Einstein algebraic curvature tensors.
- In dimension greater than three, the vanishing of the component W_A represents locally conformally flat algebraic curvature tensors.

1.1.5 Self-duality and anti-self-duality

The curvature decomposition given in Theorem 1.1 can be expressed in a more simple manner in low-dimensional cases. In dimension two every algebraic curvature tensor takes the form $A = \frac{\tau_A}{2} \langle \cdot, \cdot \rangle \odot \langle \cdot, \cdot \rangle$, while in dimension three every algebraic curvature tensor is determined by its associated Schouten tensor as $A = \mathfrak{S}_A \odot \langle \cdot, \cdot \rangle$. The situation in dimension four is more complicated, but the properties of the Hodge-star operator allow us to refine the curvature decomposition given above.

Let $\{e_1, e_2, e_3, e_4\}$ be an orthonormal basis of a four-dimensional inner product space that we will denote by $(\mathcal{V}, \langle \cdot, \cdot \rangle)$ and $\{e^1, e^2, e^3, e^4\}$ be its dual basis. Consider the volume form on \mathcal{V} , $vol = e^1 \wedge e^2 \wedge e^3 \wedge e^4$. The *Hodge-star operator* $\star: \Lambda^2 \mathcal{V} \to \Lambda^2 \mathcal{V}$ is defined as

$$\alpha \wedge \star \beta = \langle\!\langle \alpha, \beta \rangle\!\rangle vol$$

for all $\alpha, \beta \in \Lambda^2 \mathcal{V}$.

The properties of the Hodge-star operator depend on the signature of the inner product $\langle \cdot, \cdot \rangle$ under consideration. In this way, in Lorentzian signature the Hodge-star operator \star defines a complex structure ($\star^2 = -\text{Id}_{\Lambda^2 \mathcal{V}}$), while both in Riemannian and neutral signatures it defines a para-complex structure ($\star^2 = \text{Id}_{\Lambda^2 \mathcal{V}}$). In this case, the Hodge-star operator induces a decomposition of the space of bivectors $\Lambda^2 \mathcal{V} = \Lambda^2_+ \mathcal{V} \oplus \Lambda^2_- \mathcal{V}$, where $\Lambda^2_+ \mathcal{V}$ and $\Lambda^2_- \mathcal{V}$ denote the spaces of self-dual and anti-self-dual bivectors, respectively,

$$\Lambda^2_+\mathcal{V} = \left\{ \alpha \in \Lambda^2 \mathcal{V} \colon \star \alpha = \alpha \right\} \quad \text{and} \quad \Lambda^2_-\mathcal{V} = \left\{ \alpha \in \Lambda^2 \mathcal{V} \colon \star \alpha = -\alpha \right\},$$

which are the eigenspaces associated to the ± 1 eigenvalues of the Hodge-star operator.

For the orthonormal basis considered above and denoting $\varepsilon_i = \langle e_i, e_i \rangle$, the self-dual and anti-self-dual subspaces are spanned by $\{E_1^{\pm}, E_2^{\pm}, E_3^{\pm}\}$, where

$$\begin{split} E_1^{\pm} &= \frac{1}{\sqrt{2}} \left(e^1 \wedge e^2 \pm \varepsilon_3 \varepsilon_4 e^3 \wedge e^4 \right), \\ E_2^{\pm} &= \frac{1}{\sqrt{2}} \left(e^1 \wedge e^3 \mp \varepsilon_2 \varepsilon_4 e^2 \wedge e^4 \right), \\ E_3^{\pm} &= \frac{1}{\sqrt{2}} \left(e^1 \wedge e^4 \pm \varepsilon_2 \varepsilon_3 e^2 \wedge e^3 \right). \end{split}$$

The metric induced on the bivector space by the inner product on \mathcal{V} , given by (1.4), is Riemannian when $\langle \cdot, \cdot \rangle$ is positive definite and has signature (+ + - - -) when $\langle \cdot, \cdot \rangle$ has neutral signature (2, 2). In the latter case, the signature of the restriction of the metric to the subspaces $\Lambda_{\pm}^2 \mathcal{V}$ is (+ - -) and $\{E_1^{\pm}, E_2^{\pm}, E_3^{\pm}\}$ is an orthonormal basis in which E_1^{\pm} is spacelike and E_2^{\pm} and E_3^{\pm} are lightlike.

Now consider an algebraic curvature tensor A on \mathcal{V} as an endomorphism of the bivector space. The decomposition $\Lambda^2 \mathcal{V} = \Lambda^2_+ \mathcal{V} \oplus \Lambda^2_- \mathcal{V}$ induces a decomposition of the Weyl curvature tensor as $\mathcal{W} = \mathcal{W}^+ \oplus \mathcal{W}^-$, where \mathcal{W}^{\pm} denote the restriction of the Weyl curvature tensor acting on the space of bivectors to the self-dual and anti-self-dual subspaces, respectively. In dimension four the curvature decomposition given in Theorem 1.1 can be refined as

$$\mathcal{A} \equiv \frac{\tau}{12} \mathrm{Id}_{\Lambda^2 \mathcal{V}} + \rho_0 + \mathcal{W}^+ + \mathcal{W}^- \colon \Lambda^2 \mathcal{V} \to \Lambda^2 \mathcal{V},$$

where ρ_0 denotes the trace-less Ricci tensor. The Weyl curvature tensor is said to be *self-dual* if $\mathcal{W}^- = 0$ and *anti-self-dual* if $\mathcal{W}^+ = 0$.

1.1.6 Differential operators

Let (M, g) be an *n*-dimensional pseudo-Riemannian manifold and let $f: M \to \mathbb{R}$ be a differentiable function. The gradient of f is the vector field $\nabla f = {}^{\sharp}df$, which is determined by the gradient operator $\nabla: C^{\infty}(M) \to \mathfrak{X}(M)$ as

$$g(\nabla f, X) = X(f)$$
 for all $X \in \mathfrak{X}(M)$.

The gradient can be expressed in a system of local coordinates (x^1, \ldots, x^n) on M as

$$\nabla f = \sum_{i,k=1}^{n} g^{ik} \frac{\partial f}{\partial x^k} \partial_{x^i}.$$

The Hessian operator of f is defined as the endomorphism $h_f: \mathfrak{X}(M) \to \mathfrak{X}(M)$ that is given by the second covariant derivative

$$\mathbf{h}_f(X) = \nabla_X \nabla f.$$

Now, the *Hessian tensor* is the symmetric tensor field of type (0, 2) defined as

$$\operatorname{Hes}_{f}(X,Y) = g(\operatorname{h}_{f}(X),Y) = g(\nabla_{X}\nabla f,Y) = XY(f) - (\nabla_{X}Y)f.$$

1

In terms of a system of local coordinates on M, the Hessian tensor can be expressed as

$$\operatorname{Hes}_{f}(\partial_{x^{i}},\partial_{x^{j}}) = \frac{\partial^{2}f}{\partial x^{i}\partial x^{j}} - \Gamma_{ij}^{k}\frac{\partial f}{\partial x^{k}}.$$

Note that, since the Levi-Civita connection is trace-free and adapted to the metric, the Hessian operator is self-adjoint and so the corresponding (0, 2)-tensor field is symmetric. The Hessian operator also allows us to define the *Laplacian of a function f* as

$$\Delta f = \operatorname{tr}(\mathbf{h}_f).$$

Considering the extension of the Levi-Civita connection as a derivation of tensor fields on the manifold, the Laplacian of a (0, k)-tensor field T is defined as

$$\Delta T(X_1,\ldots,X_k) = \operatorname{tr}\left(\nabla^2 T\right)(X_1,\ldots,X_k) = \sum_{i=1}^n \left(\nabla^2_{E_i E_i} T\right)(X_1,\ldots,X_k),$$

where $\{E_1, \ldots, E_n\}$ is a local orthonormal frame.

The *divergence* of a vector field X is defined as the function $\operatorname{div} X = \operatorname{tr}(\nabla X)$. Considering a local orthonormal frame $\{E_1, \ldots, E_n\}$, then

$$\operatorname{div} X = \sum_{i=1}^{n} \varepsilon_{i} g(\nabla_{E_{i}} X, E_{i}),$$

where $\varepsilon_i = g(E_i, E_i)$. In general, if T is (0, s)-tensor field, its *r*-divergence is defined as the (0, s - 1)-tensor field given by

$$(\operatorname{div}_{r}T)(X_{1},\ldots,X_{s-1}) = \sum_{i=1}^{n} \varepsilon_{i}(\nabla_{E_{i}}T)(X_{1},\ldots,X_{r-1},E_{i},X_{r+1},\ldots,X_{s-1}),$$

for all $X_1, \ldots, X_{s-1} \in \mathfrak{X}(M)$. Since the *r*-divergence of the tensor field *T* is given by the *r*-th trace of ∇T , its definition does not depend on the choice of the local frame.

1.2 Curvature functionals

Let (M, g) be an *n*-dimensional pseudo-Riemannian manifold and consider a local orthonormal frame $\{E_i\}$. A *curvature scalar invariant* is a polynomial that involves the components of the curvature tensor and its covariant derivatives that does not depend on the choice of the local frame.

- The space of curvature scalar invariants of order one of a pseudo-Riemannian manifold has dimension one and is generated by the scalar curvature τ .
- The space of curvature scalar invariants of order two of a pseudo-Riemannian manifold has dimension four and is generated by $\{\tau^2, \|\rho\|^2, \|R\|^2, \Delta\tau\}$.

We refer the reader to [17, 80] for more information.

1.2.1 The Hilbert-Einstein functional

Given a compact *n*-dimensional pseudo-Riemannian manifold (M, g), the associated *Hilbert*-*Einstein functional* is given by

$$S_{HE} \colon g \mapsto \int_M \tau \, dvol_g$$

If we consider variations of the form g[t] = g + th, where h is a symmetric (0, 2)-tensor field, we will say that the metric g is critical for this variation if

$$\frac{d}{dt}\Big|_{t=0} S_{HE}(g[t]) = 0.$$

This expression can be written as

$$\frac{d}{dt}\Big|_{t=0} S_{HE}(g[t]) = \int_{M} \langle \nabla S_{HE}, h \rangle dvol_g$$

for some symmetric (0, 2)-tensor field ∇S_{HE} , which is called the *gradient of the functional* S_{HE} . The gradient of the Hilbert-Einstein functional is given by

$$\nabla S_{HE} = -\rho + \frac{\tau}{n}g.$$

Given that the Hilbert-Einstein functional is not homothetically invariant, we need to consider its restriction to the space of metrics of constant volume on a manifold M. Bearing in mind that all the metrics of the variation g[t] have constant volume as long as the tensor field h is orthogonal to the metric g, the corresponding Euler-Lagrange equations are determined by

$$\nabla S_{HE} = \lambda g.$$

Tracing both sides of this identity, it follows that

$$\lambda = \frac{1}{n} \operatorname{tr} \left(\nabla S_{HE} \right) = \frac{n-2}{2n} \tau,$$

$$\rho - \frac{\tau}{n} g = 0. \tag{1.5}$$

and so

As a consequence, the metrics which are critical for the Hilbert-Einstein functional – when restricted to variations of constant volume – are precisely the Einstein metrics, which are those whose scalar curvature is "better" distributed across the manifold.

1.2.2 Quadratic functionals

If we now consider the space of curvature scalar invariants of order two, a curvature quadratic functional is given by

$$g \mapsto \int_M \left\{ a\tau^2 + b \|\rho\|^2 + c \|R\|^2 + d\Delta\tau \right\} dvol_g,$$

1

for some $a, b, c, d \in \mathbb{R}$. If we assume that M is a compact manifold without boundary, then

$$g \mapsto \int_M \{a\tau^2 + b\|\rho\|^2 + c\|R\|^2 + d\Delta\tau\} \, dvol_g$$

= $\int_M \{a\tau^2 + b\|\rho\|^2 + c\|R\|^2\} \, dvol_g,$

and so every quadratic functional can be expressed in terms of the three functionals

$$\mathcal{S} \colon g \mapsto \int_{M} \tau^{2} dvol_{g}, \quad \mathcal{T} \colon g \mapsto \int_{M} \|\rho\|^{2} dvol_{g}, \quad \mathcal{R} \colon g \mapsto \int_{M} \|R\|^{2} dvol_{g},$$

which are what we will refer to as *curvature quadratic functionals*. The gradients of these three functionals are (see [16])

$$\nabla \mathcal{S} = 2\nabla^2 \tau - 2\Delta\tau g - \tau \left(2\rho - \frac{1}{2}g\right),$$
$$\nabla \mathcal{T} = -\Delta\rho + \nabla^2 \tau - \frac{1}{2}\Delta\tau g - 2R[\rho] + \frac{1}{2}\|\rho\|^2 g,$$
$$\nabla \mathcal{R} = -4\Delta\rho + 2\nabla^2 \tau - 2\check{R} + \frac{1}{2}\|R\|^2 g - 4R[\rho] + 4\check{\rho}.$$

The curvature tensor R of a three-dimensional pseudo-Riemannian manifold is completely determined by its Ricci tensor, which means that the three curvature scalar invariants satisfy the identity

$$||R||^2 = 2||\rho||^2 - \frac{1}{2}\tau^2$$

Consequently, the functional determined by the L^2 -norm of the curvature tensor can be expressed as a linear combination of the other two as

$$\mathcal{R} = 2\mathcal{T} - \frac{1}{2}\mathcal{S}$$

in the three-dimensional case. In the four-dimensional setting, the Chern-Gauss-Bonnet Theorem gives the Euler characteristic of a compact pseudo-Riemannian manifold with no boundary in terms of the three curvature scalar invariants as

$$\chi(M) = \frac{1}{8\pi^2} \int_M \left\{ \|R\|^2 - 4\|\rho\|^2 + \tau^2 \right\} dvol_g.$$

Therefore, the four-dimensional curvature functional \mathcal{R} is determined by

$$\mathcal{R} = t\pi^2 \chi(M) + 4\mathcal{T} - \mathcal{S},$$

and, since the Euler characteristic of a manifold is a topological invariant, the critical points of \mathcal{R} correspond to the critical points of $4\left(\mathcal{T}-\frac{1}{4}\mathcal{S}\right)$. Therefore, the functionals \mathcal{R} and $\mathcal{T}-\frac{1}{4}\mathcal{S}$ are equivalent. This shows that it is enough to study the functionals

$$\mathcal{S} \colon g \mapsto \int_{M} \tau^{2} dvol_{g}, \quad \mathcal{F}_{t} \colon g \mapsto \int_{M} \left\{ \|\rho\|^{2} + t\tau^{2} \right\} dvol_{g},$$

for all $t \in \mathbb{R}$. The gradients of the functionals \mathcal{F}_t are given by

$$\nabla \mathcal{F}_t = -\Delta \rho + (1+2t)\nabla^2 \tau - \frac{1+4t}{2}\Delta \tau g - 2R[\rho] + \frac{1}{2}\|\rho\|^2 g + \frac{1}{2}t\tau^2 g - 2t\tau\rho.$$

Proceeding in the same way as we did for the Hilbert-Einstein functional, when we restrict our study to the space of metrics of constant volume, we obtain that the Euler-Lagrange equations corresponding to these functionals are given by

$$\nabla^2 \tau - \frac{1}{n} \Delta \tau g - \tau \left(\rho - \frac{\tau}{n} g\right) = 0,$$

$$\Delta \rho - (1+2t) \nabla^2 \tau + \frac{2}{n} t \Delta \tau g + 2 \left(R[\rho] - \frac{1}{n} \|\rho\|^2 g\right) + 2t\tau \left(\rho - \frac{\tau}{n} g\right) = 0.$$

Note that Einstein metrics are critical for the functionals S and \mathcal{F}_t for any value of t. Consequently, Einstein metrics are critical for all the curvature quadratic functionals in dimensions three and four, but this is no longer true in higher dimensions.

The L^2 -norm of the Weyl curvature tensor: the Bach tensor

In dimension four, the functional given by

$$\mathcal{W} \colon g \mapsto \mathcal{W}(g) = \int_M \|W_g\|^2 dvol_g,$$

where W_g denotes the Weyl curvature tensor associated to g, quantifies the deflection of a Riemannian metric g from being locally conformally flat. A remarkable property of this functional is that it is conformally invariant in dimension four. Indeed, if $\bar{g} = e^{2\sigma}g$, then

$$\begin{split} \|\overline{W}\|^2 dvol_{\bar{g}} &= \overline{W}_{ijk\ell} \overline{W}^{ijk\ell} dvol_{\bar{g}} \\ &= e^{2\sigma} W_{ijk\ell} e^{2\sigma} e^{-8\sigma} W^{ijk\ell} e^{4\sigma} dvol_g = \|W\|^2 dvol_g. \end{split}$$

Furthermore, it follows from the Chern-Gauss-Bonnet Theorem that

$$\mathcal{W}(g) = 32\pi^2 \chi(M) + 2\mathcal{F}_{-1/3},$$

and so this functional is equivalent to $\mathcal{F}_{-1/3}$. The conformal invariance of \mathcal{W} shows that fourdimensional conformally Einstein metrics are $\mathcal{F}_{-1/3}$ -critical. Such metrics are equivalently characterized by the vanishing of their Bach tensor (see [10]), which is defined as the (0, 2)tensor field given by

$$\mathfrak{B} = \operatorname{div}_2 \operatorname{div}_4 W + \frac{1}{2} W[\rho],$$

where $W[\rho]_{ij} = W_{ijk\ell}\rho^{k\ell}$.

In addition to conformally Einstein metrics, half conformally flat metrics are also Bach-flat in dimension four. Recall that the Hirzebruch signature formula allows us to express the Hirzebruch signature as

$$\tau[M] = \frac{1}{12\pi^2} \int_M \left\{ \|W^+\|^2 - \|W^-\|^2 \right\} dvol_g$$

and so the functional \mathcal{W} can be written as

$$\begin{split} \mathcal{W}(g) &= \int_M \|W\|^2 dvol_g = \int_M \left\{ \|W^+\|^2 + \|W^-\|^2 \right\} dvol_g \\ &= \pm 12\pi^2 \tau[M] + 2\int_M \|W^{\mp}\|^2 dvol_g, \end{split}$$

which shows that half conformally flat metrics are critical for the functional \mathcal{W} , and therefore Bach-flat. Besides, since the functional \mathcal{W} is equivalent to the one given by the L^2 -norm of the self-dual Weyl curvature tensor W^+ , if g is a Kähler metric (see Part I), whose self-dual conformal operator W^+ is diagonalizable and takes the form $W^+ = \frac{\tau}{12} \operatorname{diag}[2, -1, -1]$, then

$$\mathcal{W}(g) = -12\pi^2 \tau[M] + 2\int_M \|W^+\|^2 dvol_g = -12\pi^2 \tau[M] + \frac{1}{12}\int_M \tau^2 dvol_g$$

Consequently, the Calabi functional – which is the restriction of S to metrics in the same Kähler class – is equivalent to $\mathcal{F}_{-1/3}$ when restricted to such variations (see [29, 127]).

1.3 Affine and projective geometry

An *affine manifold* is a pair (M, D) where M is a smooth manifold and D is an affine torsion-free connection on M. The Ricci tensor associated to D is defined as

$${}^{D}\rho(X,Y) := \operatorname{tr}\left(Z \to {}^{D}R(X,Z)Y\right),$$

and, since it need not be symmetric in general, we introduce the symmetrization ${}^{D}\rho_{s}$ and skew-symmetrization ${}^{D}\rho_{sk}$ as

$${}^{D}\rho_{s}(X,Y) := \frac{1}{2} \left\{ {}^{D}\rho(X,Y) + {}^{D}\rho(Y,X) \right\},$$

$${}^{D}\rho_{sk}(X,Y) := \frac{1}{2} \left\{ {}^{D}\rho(X,Y) - {}^{D}\rho(Y,X) \right\}.$$
 (1.6)

An affine manifold is *flat* if its associated curvature tensor ^{D}R vanishes. In this case, there exist local coordinates where the Christoffel symbols are zero.

A projective structure on an affine manifold (M, D) is an equivalence class [D] of affine connections on TM sharing the same unparametrized geodesics. Two affine connections D and \widetilde{D} are projectively equivalent if and only if there exists a one-form $\omega = \omega_i dx^i$ on M such that

$$\widetilde{\Gamma}_{ij}{}^{k} = \Gamma_{ij}{}^{k} + \delta_{i}{}^{k}\omega_{j} + \delta_{j}{}^{k}\omega_{i},$$

where $\Gamma_{ij}{}^k$ and $\widetilde{\Gamma}_{ij}{}^k$ denote the Christoffel symbols of the affine connections D and \widetilde{D} , respectively, and $\delta_i{}^k$ denotes the Kronecker delta. According to this, an affine manifold (M, D) is said to be *projectively flat* if there exists a flat representative in the class [D], i.e., if there exists a one-form ω on M such that

$$\Gamma_{ij}{}^{k} = -\left(\delta_{i}{}^{k}\omega_{j} + \delta_{j}{}^{k}\omega_{i}\right). \tag{1.7}$$

Furthermore, if there exists a real-valued function f locally defined on M such that $\omega = df$, then (M, D) is said to be *locally strongly projectively flat*. Two-dimensional projectively flat affine manifolds are characterized as follows (see [110]).

Theorem 1.2. Let (Σ, D) be an affine surface. Then (Σ, D) is projectively flat if and only if ${}^{D}\rho$ and $D^{D}\rho$ are totally symmetric.

An affine manifold is *curvature recurrent* (respectively, *Ricci recurrent*) if $D^D R = \xi \otimes {}^D R$ (respectively, $D^D \rho = \xi \otimes {}^D \rho$) for some one-form ξ on M and (M, D) is *locally symmetric* if $D^D R = 0$. Since the curvature tensor of any affine surface is determined by the associated Ricci tensor as

$${}^{D}R(X,Y)Z = {}^{D}\rho(X,Z)Y - {}^{D}\rho(Y,Z)X,$$

curvature recurrence and Ricci recurrence are equivalent in the two-dimensional case.

1.3.1 Walker structures

Let (M, g) be a pseudo-Riemannian manifold and consider a distribution \mathcal{D} of the tangent space. \mathcal{D} is parallel if $\nabla_X Y \in \mathcal{D}$ for every smooth vector field X and every $Y \in \mathcal{D}$, and degenerate if $g|_{\mathcal{D}}$ vanishes.

It is well-known that the existence of a parallel distribution on a Riemannian manifold induces a local de Rham decomposition as a product. This property is also true in the pseudo-Riemannian setting as long as the parallel distribution is non-degenerate. The situation in which the distribution is degenerate was studied by Walker [132], who gave a canonical expression for this kind of metrics. This is why pseudo-Riemannian manifolds admitting a parallel degenerate distribution are called *Walker manifolds*.

Walker metrics give rise to many strictly pseudo-Riemannian situations, such as degenerate homogeneous pseudo-Riemannian structures, strictly conformally symmetric manifolds, conformally flat metrics with two-step nilpotent Ricci operator, Einstein hypersurfaces in manifolds with constant sectional curvature and nilpotent shape operator, para-Kähler structures and so on.

In the most general situation, the existence of adapted Walker coordinates is given by the following result (see [132]).

Theorem 1.3. Let (M, g) be an *n*-dimensional Walker manifold and let \mathcal{D} be an *r*-dimensional, parallel degenerate distribution. There exists a local system of adapted coordinates

$$(x^1,\ldots,x^{n-r},x^{n-r+1},\ldots,x^n)$$

on M with respect to which the metric tensor takes the form

$$(g_{ij}) = \begin{pmatrix} B & H & \mathrm{Id}_r \\ {}^t\!H & A & 0 \\ \mathrm{Id}_r & 0 & 0 \end{pmatrix},$$

where Id_r is the identity matrix of order r and A, B and H are matrices whose components are functions of the adapted coordinates satisfying the following conditions:

- 1. A is an $(n-2r) \times (n-2r)$ matrix and B is an $r \times r$ matrix.
- 2. *H* is an $r \times (n 2r)$ matrix and ^tH denotes its transpose matrix.
- *3. A* and *H* are independent of the last n r coordinates.

Moreover, the distribution \mathcal{D} is spanned by the last n - r coordinate vector fields.

The canonical form given in the previous theorem turns out to be simpler when the manifold has even dimension n = 2m and the Walker distribution is of maximal dimension. In this case, there exist local Walker coordinates

$$\left(x^1,\ldots,x^m,x_{1'},\ldots,x_{m'}\right)$$

in which the metric tensor takes the form

$$g = \begin{pmatrix} B & \mathrm{Id}_m \\ \mathrm{Id}_m & 0 \end{pmatrix},\tag{1.8}$$

where B is a matrix whose components are functions of $(x^1, \ldots, x^m, x_{1'}, \ldots, x_{m'})$. This kind of metrics will play an important role in the work developed in this thesis. Their non-zero Christoffel symbols and curvature operator are given by the following results (see [41]).

Lemma 1.4. Let (M, g, D) be a Walker manifold of dimension n = 2m, where dim D = m. Then its Christoffel symbols are given, up to the corresponding symmetries, by

$$\begin{split} \Gamma_{ij}^{k} &= -\frac{1}{2}\partial_{k'}g_{ij}, \\ \Gamma_{i'j}^{k'} &= \frac{1}{2}\partial_{i'}g_{jk}, \\ \Gamma_{ij}^{k'} &= \frac{1}{2}\left(-\partial_{k}g_{ij} + \partial_{j}g_{ik} + \partial_{i}g_{jk} + g_{ks}\partial_{s'}g_{ij}\right), \end{split}$$

where $1 \leq s \leq m$.

where $1 \leq s$

Lemma 1.5. Let (M, g) be a Walker manifold of dimension n = 2m with m-dimensional parallel degenerate distribution \mathcal{D} . The non-zero components of its curvature tensor of type (1,3) are given, up to the corresponding symmetries, by

$$\begin{split} R^{h}_{ijk} &= -\frac{1}{2} \left(\partial_{j} \partial_{h'} g_{ik} - \partial_{i} \partial_{h'} g_{jk} \right) - \frac{1}{4} \left(\partial_{s'} g_{jk} \partial_{h'} g_{is} - \partial_{s'} g_{ik} \partial_{h'} g_{js} \right), \\ R^{h'}_{ijk} &= -\frac{1}{2} \left(\partial_{i} \partial_{k} g_{jh} - \partial_{i} \partial_{h} g_{jk} + \partial_{j} \partial_{h} g_{ik} - \partial_{j} \partial_{k} g_{ih} \right) \\ &- \frac{1}{4} \left\{ \partial_{s'} g_{jk} \left(\partial_{h} g_{is} - \partial_{s} g_{ih} - \partial_{i} g_{sh} - g_{ht} \partial_{t'} g_{js} \right) \\ &- \partial_{s'} g_{ik} \left(\partial_{h} g_{js} - \partial_{s} g_{jh} - \partial_{j} g_{sh} - g_{ht} \partial_{t'} g_{js} \right) \\ &- \partial_{s'} g_{ih} \left(\partial_{s} g_{jk} - \partial_{k} g_{js} - \partial_{j} g_{ks} - g_{st} \partial_{t'} g_{jk} \right) \\ &+ \partial_{s'} g_{jh} \left(\partial_{s} g_{ik} - \partial_{k} g_{is} - \partial_{i} g_{ks} - g_{st} \partial_{t'} g_{ik} \right) \\ &+ 2\partial_{i} \left(g_{hs} \partial_{s'} g_{jk} \right) - 2\partial_{j} \left(g_{hs} \partial_{s'} g_{ik} \right) \right\}, \end{split}$$

$$\begin{split} R^{h}_{ij'k} &= -\frac{1}{2}\partial_{j'}\partial_{h'}g_{ik}, \\ R^{h'}_{ij'k} &= -\frac{1}{2}\left(\partial_{h}\partial_{j'}g_{ik} - \partial_{k}\partial_{j'}g_{ih}\right) \\ &- \frac{1}{4}\left(\partial_{s'}g_{ik}\partial_{j'}g_{sh} + \partial_{s'}g_{ih}\partial_{j'}g_{sk} - 2\partial_{j'}(g_{hs}\partial_{s'}g_{ik})\right), \\ R^{h'}_{ijk'} &= -\frac{1}{2}\left(\partial_{i}\partial_{k'}g_{jh} - \partial_{j}\partial_{k'}g_{ih}\right) - \frac{1}{4}\left(\partial_{k'}g_{js}\partial_{s'}g_{ih} - \partial_{k'}g_{is}\partial_{s'}g_{jh}\right), \\ R^{h'}_{ij'k'} &= \frac{1}{2}\partial_{j'}\partial_{k'}g_{ih}, \\ \leq m \text{ and } 1 \leq t \leq m. \end{split}$$

In general, the fact that \mathcal{D} is parallel implies that the curvature tensor of any Walker manifold is such that

$$R(\mathcal{D}, \mathcal{D}^{\perp}, \cdot, \cdot) = 0, \quad R(\mathcal{D}, \mathcal{D}, \cdot, \cdot) = 0, \text{ and } R(\mathcal{D}^{\perp}, \mathcal{D}^{\perp}, \mathcal{D}, \cdot) = 0$$

1.3.2 Metrics on the cotangent bundle

A particular class of Walker metrics are those known as Riemannian extensions. The interest of these metrics resides in the fact that they allow the translation of problems in affine geometry to problems in pseudo-Riemannian geometry and vice versa.

Let T^*M denote the cotangent bundle of an *n*-dimensional smooth manifold and consider the natural projection from the cotangent bundle onto the base manifold $\pi: T^*M \to M$. A point $\tilde{p} \in T^*M$ is of the form $\tilde{p} = (p, \omega)$, where $p := \pi(\tilde{p}) \in M$ and $\omega \in T_p^*M$. We will give some basic notions about the geometry of the cotangent bundle before introducing Riemannian extensions.

Let $\tilde{p} = (p, \omega)$ be a point in T^*M and consider local coordinates (x^1, \ldots, x^n) on a neighbourhood \mathcal{U} of the point $p \in M$. In this neighbourhood we can write

$$\omega = x_{i'} dx^i,$$

which allows us to define a system of local coordinates on $\widetilde{\mathcal{U}} := \pi^{-1}(\mathcal{U}) \subset T^*M$ as

$$(x^1,\ldots,x^n,x_{1'},\ldots,x_{n'})$$
.

In terms of these local coordinates, the canonical symplectic structure of the cotangent bundle is given by

$$\Omega := d\omega = dx_{i'} \wedge dx^i. \tag{1.9}$$

Let us consider a vector field X on M. Its evaluation map ιX is the differentiable map on the cotangent bundle given by

$$\iota X(p,\omega) = \omega \left(X_p \right).$$

If we write $X = X^i \partial_{x^i}$, where $X^i = dx^i(X)$, then $\iota X(x^i, x_{i'}) = x_{i'}X^i$.

Vectors fields on T^*M are determined by their action on evaluation maps of vector fields on M. In this way, two vector fields \widetilde{Y} and \widetilde{Z} on M are the same if and only if $\widetilde{Y}(\iota X) = \widetilde{Z}(\iota X)$ for every vector field X on M. Bearing this in mind, the complete lift of a vector field X on M is a vector field on T^*M characterized by the identity

$$X^{C}(\iota Z) = \iota \left[X, Z \right],$$

for every vector field Z on M. The tangent space to the cotangent bundle of a smooth manifold is generated by the complete lifts of all smooth vector fields on M. Complete lifts allow us to characterize tensor fields of type (0, s) in the sense that two tensor fields \tilde{S} and \tilde{T} are the same if and only if

$$\widetilde{T}\left(X_{1}^{C},\ldots,X_{s}^{C}\right)=\widetilde{S}\left(X_{1}^{C},\ldots,X_{s}^{C}\right)$$
for any vector fields X_1, \ldots, X_s on M. Knowing this, it is easy to see that the two-form (1.9) does not depend on the system of local coordinates in consideration and is characterized by the identity

$$\omega\left(X^C, Y^C\right) = \iota\left[X, Y\right].$$

Now, if T is a (1, 1)-tensor field on M, it is an endomorphism of the tangent bundle TM, so one can define a one-form ιT on the cotangent bundle characterized by the identity

$$\iota T\left(X^{C}\right) = \iota\left(TX\right),$$

which takes the form $\iota T = x_{k'}T_i^k dx^i$ with respect to the coordinates induced on T^*M .

Riemannian extensions

The construction of this kind of metrics defines a Walker metric on the cotangent bundle of an affine *n*-manifold (M, D), where D denotes a torsion-free connection on M. T^*M can be equipped with a pseudo-Riemannian metric of neutral signature (n, n) given by the identity

$$g_D\left(X^C, Y^C\right) = -\iota\left(D_XY + D_YX\right)$$

This metric is called a *Riemannian extension* (see [120]) and in terms of the system of local coordinates induced on T^*M takes the form

$$g_D = \begin{pmatrix} -2x_{k'}\Gamma_{ij}^k & \mathrm{Id}_n \\ \mathrm{Id}_n & 0 \end{pmatrix}, \quad i, j = 1, \dots, n, \quad i' = i + n,$$
(1.10)

where $\Gamma_{ij}{}^k$ are the Christoffel symbols of the affine connection D with respect to the coordinates (x^i) on M.

These metrics are particular cases of Walker metrics for which the parallel, degenerate distributions have maximal dimension and are given by $\mathcal{D} = \ker \pi_*$. As we have already mentioned, Riemannian extensions provide a link between affine geometry and Riemannian geometry so that some properties of the affine connection D can be studied through the properties of g_D . For instance, D is projectively flat if and only if g_D is locally conformally flat.

Deformed Riemannian extensions

These metrics are a slight generalization of Riemannian extensions involving an additional (0, 2)tensor field on M that we will call Φ . Deformed Riemannian extensions are again pseudo-Riemannian metrics of neutral signature (n, n) on the cotangent bundle of an affine manifold and are given by

$$g_{D,\Phi} \left(X^C, Y^C \right) = g_D + \pi^* \Phi$$

= $-\iota \left(D_X Y + D_Y X \right) + \pi^* \Phi.$

Deformed Riemannian extensions can be expressed in terms of the coordinates induced on the cotangent bundle as

$$g_{D,\Phi} = \begin{pmatrix} -2x_{k'}\Gamma_{ij}{}^k + \Phi_{ij} & \mathrm{Id}_n \\ \mathrm{Id}_n & 0 \end{pmatrix}, \quad i,j = 1,\dots,n, \quad i' = i+n.$$

A criterium to characterize deformed Riemannian extensions amongst Walker manifolds was shown in [1], where it is stated that if (M, g, D) is a Walker manifold then g is a deformed Riemannian extension of an affine connection if and only if the curvature operator associated to g is such that

$$R\left(\cdot,\mathcal{D}\right)\mathcal{D}=0.$$

Moreover, the scalar curvature of every deformed Riemannian extension vanishes and its Ricci operator is nilpotent, so a deformed Riemannian extension will be Einstein if and only if it is Ricci-flat.

Modified Riemannian extensions

Going further in the generalization of Riemannian extensions we find a new kind of metrics known as modified Riemannian extensions. These metrics involve two new elements S and T which are (1, 1)-tensor fields on the affine manifold M. Modified Riemannian extensions are split-signature metrics on the cotangent bundle of M as well and are defined by

$$g_{D,\Phi,T,S} = \iota T \circ \iota S + g_{D,\Phi} = \iota T \circ \iota S + g_D + \pi^* \Phi,$$

where \circ denotes the symmetric product of one-forms. This kind of metrics can be expressed in terms of the local coordinates induced on the cotangent bundle as

$$g_{D,\Phi,T,S} = \begin{pmatrix} \frac{1}{2} x_{r'} x_{s'} \left(T_i^r S_j^s + T_j^r S_i^s \right) - 2x_{k'} \Gamma_{ij}^k + \Phi_{ij} & \mathrm{Id}_n \\ \mathrm{Id}_n & 0 \end{pmatrix},$$

where i, j = 1, ..., n and i' = i + n. In the particular case where $T = c \operatorname{Id}$ and $S = \operatorname{Id}$, the metric is denoted by $g_{D,\Phi,c}$ and it is a Walker metric for which the tensor B_{ij} in (1.8) is a quadratic function of the fibre coordinates $(x_{i'})$. The Walker distribution is given by $\mathcal{D} = \ker \pi_*$ and its scalar curvature is a multiple (depending on the dimension of the manifold) of the parameter c.

Modified Riemannian extensions can be characterized amongst the Walker metrics in terms of the covariant derivative of their curvature by

$$\nabla_{\mathcal{D}} R\left(\cdot, \mathcal{D}\right) \mathcal{D} = 0$$

(see [1]). As a consequence, an even-dimensional Walker manifold admitting a parallel, degenerate distribution of maximal dimension is locally symmetric if and only if it is a suitable modified Riemannian extension. In addition to all this, modified Riemannian extensions turn out to be a source of examples of Einstein metrics. In fact, the modified Riemannian extension $g_{D,\Phi,c}$, with $c \neq 0$, is Einstein if and only if $\Phi = \frac{4}{c(n-1)}^D \rho_s$.

If (M, D) is flat, its cotangent bundle is not only Einstein but also para-Kähler with constant para-holomorphic sectional curvature, when equipped with a modified Riemannian extension [41]. We will see this in detail in Part I.

1.4 Lie groups

A *Lie group* is a smooth manifold endowed with a group structure such that the operation in the group $\sigma: G \times G \to G$ and the inversion are differentiable. A group homomorphism that is differentiable as a map between two smooth manifolds is called a *Lie group homomorphism*. A *real Lie algebra* is a vector space g endowed with a skew-symmetric bilinear operator

$$[\cdot, \cdot] \colon \mathfrak{g} \times \mathfrak{g} \to \mathfrak{g}$$

satisfying the Jacobi identity

$$[x, [y, z]] + [z, [x, y]] + [y, [z, x]] = 0$$

for all $x, y, z \in \mathfrak{g}$. A Lie algebra homomorphism is a linear map $\varphi : \mathfrak{g} \to \mathfrak{h}$ that preserves the Lie brackets, i.e., such that $\varphi[x, y]_{\mathfrak{g}} = [\varphi(x), \varphi(y)]_{\mathfrak{h}}$ for all $x, y \in \mathfrak{g}$. Given a Lie algebra \mathfrak{g} , a vector subspace \mathfrak{h} is a Lie subalgebra of \mathfrak{g} if \mathfrak{h} is closed for the Lie brackets, i.e., if $[x, y] \in \mathfrak{h}$ for all $x, y \in \mathfrak{h}$. An *ideal* \mathfrak{a} of \mathfrak{g} is a Lie subalgebra satisfying the stronger condition that $[x, a] \in \mathfrak{a}$ for all $x \in \mathfrak{g}$ and $a \in \mathfrak{a}$. A particular example of an ideal is the *derived subalgebra* of a given Lie algebra \mathfrak{g} , which is the subspace $\mathfrak{g}' = \operatorname{span} \{[x, y] : x, y \in \mathfrak{g}\}$ and is also denoted by $[\mathfrak{g}, \mathfrak{g}]$. A Lie algebra such that the sequence of subalgebras given by

$$\mathfrak{g} \geq [\mathfrak{g}, \mathfrak{g}] \geq [\mathfrak{g}, [\mathfrak{g}, \mathfrak{g}]] \geq [\mathfrak{g}, [\mathfrak{g}, [\mathfrak{g}, \mathfrak{g}]]] \geq \dots$$

terminates in the zero subalgebra is said to be *nilpotent*. The Lie algebras whose derived subalgebras are nilpotent are called *solvable*, and those which have no non-zero solvable ideals are said to be *semi-simple*.

Given a Lie group G, a vector field X on G is said to be *left-invariant* if for all $g, h \in G$,

$$(L_q)_{*h}X_h = X_{L_q(h)},$$

where $L_q: h \in G \rightarrow gh \in G$ denotes the left translations on G.

The Lie brackets induce a Lie algebra structure in the space of left-invariant vector fields on a Lie group G, which is called the *Lie algebra of* G and is denoted by $Lie(G) = \mathfrak{g}$. The Lie algebra of a Lie group is isomorphic to the tangent space at the identity element of G. Besides, any Lie algebra of finite dimension \mathfrak{g} is isomorphic to the Lie algebra of a unique connected and simply connected Lie group – up to isomorphisms. This allows us to identify each simply connected Lie group with its Lie algebra.

In the situation where the Lie group is also a pseudo-Riemannian manifold, the need to study metrics for which L_g is an isometry – known as *left-invariant metrics* – arises naturally. Giving a left-invariant metric on a Lie group G is equivalent to giving an inner product on its Lie algebra g. The invariance of the metric allows us to obtain general expressions for the Levi-Civita connection, curvature, etc. of a pseudo-Riemannian Lie group just by knowing how the brackets and the metric behave.

Semi-direct products

Given two Lie algebras $(\mathfrak{h}_1, [\cdot, \cdot]_1)$ and $(\mathfrak{h}_2, [\cdot, \cdot]_2)$, if we consider the Lie algebra $\mathfrak{g} = \mathfrak{h}_1 \oplus \mathfrak{h}_2$ given by the direct sum of \mathfrak{h}_1 and \mathfrak{h}_2 as vector spaces, we can take the Lie brackets $[v, v'] = [v, v']_1$, $[w, w'] = [w, w']_2$ and [v, w] = 0 for all $v, v' \in \mathfrak{h}_1$ and $w, w' \in \mathfrak{h}_2$. Then $(\mathfrak{g}, [\cdot, \cdot])$ is the direct sum Lie algebra.

Let $\mathfrak{g}, \mathfrak{h}_1$ and \mathfrak{h}_2 be a Lie algebra, an ideal of \mathfrak{g} and a Lie subalgebra of \mathfrak{g} , respectively, such that $\mathfrak{g} = \mathfrak{h}_1 \oplus \mathfrak{h}_2$. If $w \in \mathfrak{h}_2$, and it follows from the Jacobi identity that $\mathfrak{D} = \mathrm{ad}(w) = [w, \cdot]$ is a derivation of the Lie algebra \mathfrak{h}_1 , i.e.,

$$\mathfrak{D}[u,v] = [\mathfrak{D}u,v] + [u,\mathfrak{D}v]$$

for all $u, v \in \mathfrak{h}_1$. In this way, $\operatorname{ad}: \mathfrak{h}_2 \to \operatorname{Der}(\mathfrak{h}_1)$ provides a Lie algebra homomorphism between \mathfrak{h}_2 and the Lie algebra of the derivations of \mathfrak{h}_1 , where the Lie brackets are given by the commutator. In fact, this homomorphism determines all the Lie brackets between the elements of \mathfrak{h}_2 and \mathfrak{h}_1 , since $[w, v] = \operatorname{ad}(w)v$ for all $v \in \mathfrak{h}_1$ and $w \in \mathfrak{h}_2$.

The construction given above can be extended to an arbitrary Lie algebra homomorphism $\varphi \colon \mathfrak{h} \to \text{Der}(\mathfrak{g})$. It can be checked that there exists a unique Lie algebra structure in the vector space $\mathfrak{g} \oplus \mathfrak{h}$ such that

$$[v, v'] = [v, v']_{\mathfrak{g}}, \quad [w, w'] = [w, w']_{\mathfrak{h}}, \quad [w, v] = \varphi(w)v,$$

for all $v, v' \in \mathfrak{g}$ and $w, w' \in \mathfrak{h}$. With these brackets, $\mathfrak{g} \oplus \mathfrak{h}$ is known as the *semi-direct product* of \mathfrak{g} and \mathfrak{h} via φ and is denoted by either $\mathfrak{h} \ltimes_{\varphi} \mathfrak{g}$ or $\mathfrak{g} \rtimes_{\varphi} \mathfrak{h}$.

At a Lie group level, given the product of two Lie groups $G \times H$, its Lie algebra is given by $Lie(G \times H) = \mathfrak{g} \oplus \mathfrak{h}$ were \mathfrak{g} and \mathfrak{h} are ideals of their direct sum. Conversely, given a Lie group G and two subgroups H_1 and H_2 such that $\mathfrak{g} = \mathfrak{h}_1 \oplus \mathfrak{h}_2$, then $G = H_1 \times H_2$ as Lie groups.

In the same way as in the Lie algebra situation, this notion can be extended to semi-direct products of Lie groups. Given two Lie groups G and H, G acts on H by homomorphisms if there exists a differentiable action $\sigma \colon G \times H \to H$ such that $\sigma(g, \cdot) \in \operatorname{Aut}(H)$ for all $g \in G$ – equivalently, there exists a Lie group homomorphism $\overline{\sigma} \colon G \to \operatorname{Aut}(H)$. The multiplication in $G \times H$ given by

$$(g_1, h_1) \cdot (g_2, h_2) = (g_1 g_2, \sigma(g_2^{-1}, h_1) h_2), \text{ where } g_1, g_2 \in G, h_1, h_2 \in H,$$

induces a Lie group structure in $G \times H$, known as the *semi-direct product* of G and H via σ and denoted by either $G \ltimes_{\sigma} H$ or $H \rtimes_{\sigma} G$. When the action σ is trivial, the structure corresponds to a direct product of Lie groups.

The Lie algebra of a semi-direct product of Lie groups $H \rtimes_{\sigma} G$ is the semi-direct product of their Lie algebras via the derivative of $\overline{\sigma}(g, \cdot)$, $\mathfrak{h} \rtimes_{d\overline{\sigma}} \mathfrak{g}$. Conversely, given a semi-direct product of Lie algebras $\mathfrak{h} \rtimes_{\varphi} \mathfrak{g}$, let G and H be the connected and simply connected Lie groups with Lie algebras \mathfrak{g} and \mathfrak{h} , respectively. There is a unique action by homomorphisms Φ from G on Hsuch that $d\Phi = \varphi$ and $H \rtimes_{\varphi} G$ is the unique connected and simply connected Lie group with Lie algebra $\mathfrak{h} \rtimes_{\varphi} \mathfrak{g}$. This gives a correspondence between semi-direct products of Lie groups and semi-direct products of Lie algebras.

1.4.1 Three-dimensional Lie groups

In dimension three we can make use of the Euclidean cross product operation to simplify the computations related to the Lie brackets and, in fact, classify the unimodular Lie groups (see [104]).

Let $(\mathfrak{g}, \langle \cdot, \cdot \rangle)$ be a Lie algebra endowed with a non-degenerate inner product and consider $\times : \mathfrak{g} \times \mathfrak{g} \to \mathfrak{g}$ the only cross product such that $\langle e_i \times e_j, e_k \rangle = \det(e_i, e_j, e_k)$ for any orthonormal basis $\{e_1, e_2, e_3\}$ of \mathfrak{g} . We can define a linear map $L : \mathfrak{g} \to \mathfrak{g}$ such that

 $L(e_1 \times e_2) = [e_1, e_2], \quad L(e_2 \times e_3) = [e_2, e_3], \quad L(e_1 \times e_3) = [e_1, e_3]$

with respect to an orthonormal basis of the Lie algebra. Therefore, L is such that $L(u \times v) = [u, v]$ for all vectors $u, v \in \mathfrak{g}$ and so it provides us with all the Lie brackets of the Lie algebra. L is known as *structure operator* of \mathfrak{g} .

Unimodular Lie groups are characterized in terms of their adjoint automorphism as those connected Lie groups G such that $\operatorname{ad}(x): \mathfrak{g} \to \mathfrak{g}$ is trace-free for all vectors $x \in \mathfrak{g} = Lie(G)$ (see [104]). Consequently, a Lie algebra \mathfrak{g} such that $\operatorname{tr} \operatorname{ad}(x) = 0$ for all $x \in \mathfrak{g}$ is said to be a *unimodular Lie algebra*.

Both in the Riemannian and the Lorentzian situations, writing $L(e_i) = c_i{}^j e_j$ and tracing $ad(e_i)$ in terms of the $c_i{}^j$'s one can see that G is unimodular if and only if L is self-adjoint. Therefore, it is possible to study unimodular Lie algebras through the study of these self-adjoint operators.

Riemannian unimodular Lie groups

Considering a three-dimensional Riemannian Lie algebra g, one has that

$$e_1 \times e_2 = e_3, \quad e_3 \times e_1 = e_2, \quad e_2 \times e_3 = e_1,$$

and so

$$L(e_1) = [e_2, e_3], \quad L(e_2) = [e_3, e_1], \quad L(e_3) = [e_1, e_2].$$

If \mathfrak{g} is a unimodular Lie algebra, then L is diagonalizable with respect to an orthonormal basis, i.e., if λ_1 , λ_2 and λ_3 are the eigenvalues of L, there exists an orthonormal basis of eigenvectors with respect to which

$$[e_2, e_3] = \lambda_1 e_1, \quad [e_3, e_1] = \lambda_2 e_2, \quad [e_1, e_2] = \lambda_3 e_3.$$

Unimodular Lie groups can therefore be described in terms of the signs of the eigenvalues of L as (see [104])

Associated Lie group
SU(2)
$\widetilde{SL}(2,\mathbb{R})$
$\widetilde{E}(2)$
E(1,1)
H^3
\mathbb{R}^{3}

The descriptions of the Lie groups that appear in the table are the following.

- SU(2) is the group of unitary 2×2 matrices of determinant 1. Its Lie algebra $\mathfrak{su}(2)$ is the algebra of skew-Hermitian 2×2 matrices with no trace.
- *SL*(2, ℝ) is the universal covering of the group of the 2 × 2 real matrices of determinant
 1. Its Lie algebra sl(2, ℝ) is the algebra of 2 × 2 real matrices with no trace.
- *E*(2) is the universal covering of the group of rigid motions of the Euclidean plane. Its Lie algebra e(2), known as the *Euclidean algebra*, consists of the semi-direct product r² × r determined by an endomorphisms of r² with imaginary eigenvalues.
- E(1,1) is the group of rigid motions of the Minkowskian plane. Its Lie algebra e(1,1), known as the *Poincaré algebra*, is the semi-direct product r²×r given by an endomorphism of r² with real and not-equal eigenvalues.
- H^3 is the Heisenberg group of order three, which consists of the 3×3 matrices of the form

$$\left(\begin{array}{ccc}1&a&b\\0&1&c\\0&0&1\end{array}\right),\quad\text{for}\quad a,b,c\in\mathbb{R}.$$

Its Lie algebra h^3 is the algebra of upper triangular matrices with zero diagonal.

• \mathbb{R}^3 is the Abelian group, whose Lie algebra is the Abelian algebra \mathfrak{r}^3 .

Remark 1.6. With the intention of making the reading of this thesis easier, and since Chapter 4 is the only chapter in which we work with Lorentzian signature, the description of Lorentzian unimodular Lie groups will be given there. The reader can find it in Sections 4.1, 4.2 and 4.3.

Riemannian non-unimodular Lie groups

If a three-dimensional Riemannian Lie group is not unimodular, then there exists a basis of its Lie algebra with respect to which the Lie brackets take the form (see [104])

$$[e_1, e_2] = \alpha e_2 + \beta e_3, \qquad [e_1, e_3] = \gamma e_2 + \delta e_3,$$

with $[e_2, e_3] = 0$, and so that the matrix associated to $ad(e_1)$

$$A = \left(\begin{array}{cc} \alpha & \gamma \\ \beta & \delta \end{array}\right)$$

has trace $\alpha + \delta = 2$ and $\alpha \gamma + \beta \delta = 0$. If we exclude the case where A = Id - in which case the group corresponds to a unimodular one –, the determinant $D = \alpha \delta - \beta \gamma$ provides a complete isomorphism invariant for this Lie algebra.

1

1.4.2 Four-dimensional Lie groups

Any real Lie algebra can be written as the semi-direct product of its radical – its maximal solvable ideal – and a semi-simple subalgebra known as the *Levi factor* (see [90]). Therefore, the classification of four-dimensional Lie algebras can be reduced to the study of low-dimensional semi-simple Lie algebras, solvable Lie algebras and semi-direct products of semi-simple and solvable Lie algebras. Semi-simple Lie algebras further decompose into direct sums of simple subalgebras – non Abelian Lie algebras without any non-zero proper ideals – which are orthogonal with respect to the Killing-Cartan form. Solvable Lie algebras have been classified in low dimensions – up to six – and we refer the reader to [6] for a description of four-dimensional solvable Lie algebras.

Simply connected four-dimension Lie groups are either isomorphic to the direct extensions $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$ or solvable Lie groups, which correspond to the semi-direct extensions of the Poincaré, Euclidean, Abelian and Heisenberg groups. Since solvable Lie groups play a special role in the study of left-invariant symplectic structures, we will outline some results on the classification of their corresponding Lie algebras.

Solvable Lie groups

Consider a four-dimensional Lie algebra \mathfrak{g} and an ideal \mathfrak{v} of codimension one in \mathfrak{g} . Taking $e_0 \in \mathfrak{g} \setminus \mathfrak{v}$, we can write $\mathfrak{g} = \mathbb{R} \ltimes_{\varphi} \mathfrak{v}$, where $\varphi \colon \mathbb{R}e_0 \to \text{Der}\mathfrak{v}$ is a linear map such that $\varphi(e_0) = \operatorname{ad}(e_0)$.

The following result, which can be found in [6, Proposition 1.3] (see [55] for a different proof), proves that any four-dimensional solvable real Lie algebra is a semi-direct product of \mathbb{R} and a three-dimensional unimodular Lie algebra. Therefore, the classification of four-dimensional solvable Lie algebras is reduced to the study of the derivations of three-dimensional unimodular Lie algebras.

Proposition 1.7. Let g be a four-dimensional solvable real Lie algebra. Then

$$\mathfrak{g}\cong\mathbb{R}\ltimes_{\varphi}\mathfrak{v},$$

where v is isomorphic to either r^3 , \mathfrak{h}_3 , $\mathfrak{e}(1,1)$ or $\mathfrak{e}(2)$.

Proof. Consider the Lie algebra homomorphism given by

$$\begin{array}{rcl} \chi \colon \mathfrak{g} & \to & \mathbb{R} \\ x & \mapsto & \chi(x) = \operatorname{tr}(\operatorname{ad}(x)). \end{array}$$

Its kernel $\mathfrak{u} = \ker \chi$, known as the *unimodular kernel*, is an ideal of the Lie algebra \mathfrak{g} containing the derived subalgebra $\mathfrak{g}' = [\mathfrak{g}, \mathfrak{g}]$.

If \mathfrak{g} is not unimodular, then its unimodular kernel $\mathfrak{u} = \ker \chi$ is three-dimensional, and therefore it is isomorphic to either \mathbb{R}^3 , \mathfrak{h}_3 , $\mathfrak{e}(1,1)$ or $\mathfrak{e}(2)$ and the result follows by taking $\mathfrak{v} = \mathfrak{u}$.

If \mathfrak{g} is unimodular, its derived subalgebra \mathfrak{g}' is nilpotent and its dimension is lower than or equal to three. Therefore, \mathfrak{g}' is isomorphic to either $\{0\}$, \mathbb{R} , \mathbb{R}^2 , \mathbb{R}^3 or \mathfrak{h}_3 .

- If $\mathfrak{g}' \cong \mathbb{R}^3$ or $\mathfrak{g}' \cong \mathfrak{h}_3$, the result immediately follows by taking $\mathfrak{v} = \mathfrak{g}'$.
- If $\mathfrak{g} \cong \{0\}$, \mathfrak{g} is Abelian and so $\mathfrak{v} = \mathbb{R}^3$ is an ideal of \mathfrak{g} .
- If g' ≅ ℝ, take g' = ℝe₃ and there exist e₁, e₂ ∈ g such that [e₁, e₂] = e₃ and, since g is unimodular, the set {e₁, e₂, e₃} is linearly independent, and so it generates an ideal which is isomorphic to h₃.
- If $\mathfrak{g}' \cong \mathbb{R}^2$, there are two different possibilities:

Case (i): There exists $x \notin \mathfrak{g}'$ such that $\operatorname{ad}(x)|_{\mathfrak{g}'}$ is non-singular.

Case (ii): $\operatorname{ad}(x)|_{\mathfrak{g}'}$ is singular for all $x \in \mathfrak{g}$.

Making use of the real Jordan normal form of the corresponding complex transformation in both cases we obtain that

$$\chi(x) = \operatorname{tr}(\operatorname{ad}(x)) = \lambda_1 + \lambda_2$$
 with $\lambda_i \in \mathbb{C}, i = 1, 2.$

In Case (i), there is a basis of \mathfrak{g}' such that the action of x is given – up to a non-zero multiple – by

(i.a)
$$\operatorname{ad}(x)|_{\mathfrak{g}'} = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$$
 or (i.b) $\operatorname{ad}(x)|_{\mathfrak{g}'} = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}$

where Case (i.b) corresponds to the complex eigenvalues $\pm i$. Thus, $\mathbb{R}x \oplus \mathfrak{g}'$ is an ideal of \mathfrak{g} that is isomorphic to $\mathfrak{e}(1,1)$ – in Case (i.a) – and $\mathfrak{e}(2)$ – in Case (i.b).

In Case (ii), since one of λ_1 and λ_2 vanishes, the unimodular condition implies that both of them have to. Therefore, for any fixed $x \notin \mathfrak{g}'$ there is a basis of \mathfrak{g}' with respect to which the action of $\operatorname{ad}(x)|_{\mathfrak{g}'}$ takes one of the following forms.

(ii.a)
$$\operatorname{ad}(x)|_{\mathfrak{g}'} = 0$$
 or (ii.b) $\operatorname{ad}(x)|_{\mathfrak{g}'} = \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}$.

As a consequence, $\mathbb{R}x \oplus \mathfrak{g}'$ is an ideal of \mathfrak{g} that is isomorphic to \mathbb{R}^3 in Case (ii.a) and to \mathfrak{h}_3 in Case (ii.b). This completes the proof.

Remark 1.8 (The Poincaré Lie algebra $\mathfrak{e}(1,1)$). The Poincaré Lie group can be equivalently described as

(i) the semi-direct product $\mathbb{R}^2 \rtimes_{\psi_1} \mathbb{R}e_1$ with

$$\psi_1 = \operatorname{ad}(e_1) = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix},$$

in which case its non-zero Lie brackets are given by

$$[e_1, e_2] = e_2, \quad [e_1, e_3] = -e_3,$$

(*ii*) or the semi-direct product $\mathbb{R}^2 \rtimes_{\psi_2} \mathbb{R}e_3$ with

$$\psi_2 = \operatorname{ad}(e_3) = \begin{pmatrix} 0 & -1 \\ -1 & 0 \end{pmatrix},$$

in which case its non-zero Lie brackets are given by

$$[e_3, e_1] = -e_2, \quad [e_3, e_2] = -e_1.$$

Let us consider the description of the Poincaré group given by (i). The derivations of this Lie algebra are

$$\operatorname{Der}(\mathfrak{e}(1,1)) = \left\{ \varphi = \left(\begin{array}{cc} 0 & 0 & 0 \\ c & a & 0 \\ d & 0 & b \end{array} \right) : a, b, c, d \in \mathbb{R} \right\},\$$

and so every semi-direct extension is of the form $E(1,1) \rtimes_{\varphi} \mathbb{R}e_0$, where φ belongs to $\text{Der}(\mathfrak{e}(1,1))$.

If we now rescale the vector e_0 and take $\hat{e}_0 = e_0 - ae_1 + ce_2 - de_3$, and consider the basis of $E(1,1) \rtimes_{\varphi} \mathbb{R}\hat{e}_0$ given by $\{e_1, e_2, e_3, \hat{e}_0\}$, then

$$[\hat{e}_0, e_1] = 0, \quad [\hat{e}_0, e_2] = 0, \quad [\hat{e}_0, e_3] = (a+b)e_3,$$

so there are two different possibilities depending on whether or not a + b = 0.

If a + b = 0, then $E(1, 1) \rtimes_{\varphi} \mathbb{R}\hat{e}_0$ is unimodular and its Lie brackets are

$$[\hat{e}_0, e_i] = 0, \quad [e_1, e_2] = e_2, \quad [e_1, e_3] = -e_3.$$

This corresponds to the product Lie group $E(1,1) \times \mathbb{R}\hat{e}_0$.

If $a + b \neq 0$, then $E(1, 1) \rtimes_{\varphi} \mathbb{R}e_0$ is non-unimodular and its non-zero Lie brackets are

$$[e_1, e_2] = e_2, \quad [e_1, e_3] = -e_3, \quad [e_3, \hat{e}_0] = \lambda e_3, \quad (\lambda \neq 0).$$

If we now consider the Lie group homomorphism Φ given by

$$\Phi(e_1) = -\frac{1}{\lambda}\hat{e}_0, \quad \Phi(e_2) = e_3, \quad \Phi(e_3) = e_1 - \frac{1}{\lambda}\hat{e}_0, \quad \Phi(\hat{e}_0) = \lambda e_2$$

and take the basis $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3, \mathbf{v}_4\}, \mathbf{v}_i = \Phi(e_i)$, of $E(1, 1) \rtimes_{\varphi} \mathbb{R}\hat{e}_0$, then the Lie brackets are

$$[\mathbf{v}_1,\mathbf{v}_2]=\mathbf{v}_2, \qquad [\mathbf{v}_3,\mathbf{v}_4]=\mathbf{v}_4$$

and this Lie algebra corresponds to $\mathfrak{aff}(\mathbb{R}) \times \mathfrak{aff}(\mathbb{R})$, where $\mathfrak{aff}(\mathbb{R})$ denotes the real affine Lie algebra.

The discussion in the previous remark can be summarized as follows.

Lemma 1.9. Let $G = E(1,1) \rtimes \mathbb{R}$ be a semi-direct extension of the Poincaré Lie group. Then

(i) G is unimodular if and only if it is isomorphic to the product $E(1,1) \times \mathbb{R}$, and

(ii) G is non-unimodular if and only if it is isomorphic to $\mathfrak{aff}(\mathbb{R}) \times \mathfrak{aff}(\mathbb{R})$.

Remark 1.10 (The Euclidean Lie algebra $\mathfrak{e}(2)$). The Euclidean Lie group can be described as the semi-direct product $\mathbb{R}^2 \rtimes_{\psi} \mathbb{R}$ given by

$$\psi = \operatorname{ad}(e_1) = \left(\begin{array}{cc} 0 & -1\\ 1 & 0 \end{array}\right)$$

so its non-zero Lie brackets are given by

$$[e_1, e_2] = e_3, \qquad [e_1, e_3] = -e_2.$$

The derivations of this Lie algebra are

$$\operatorname{Der}(\mathfrak{e}(2)) = \left\{ \varphi = \left(\begin{array}{cc} 0 & 0 & 0 \\ c & a & -b \\ d & b & a \end{array} \right) : a, b, c, d \in \mathbb{R} \right\}$$

and so every semi-direct extension is of the form $\widetilde{E}(2) \rtimes_{\varphi} \mathbb{R}e_0$, where $\varphi \in \text{Der}(\mathfrak{e}(2))$. We can rescale e_0 by $\hat{e}_0 = e_0 - be_1 + de_2 - ce_3$ and consider the basis of $\widetilde{E}(2) \times \mathbb{R}\hat{e}_0$ given by $\{e_1, e_2, e_3, \hat{e}_0\}$ so that

$$[\hat{e}_0, e_1] = 0, \quad [\hat{e}_0, e_2] = ae_2, \quad [\hat{e}_0, e_3] = ae_3.$$

This gives rise to two different possibilities depending on whether or not a = 0.

If a = 0, then $\widetilde{E}(2) \rtimes_{\varphi} \mathbb{R}\hat{e}_0$ is unimodular and its Lie brackets are

$$[\hat{e}_0, e_i] = 0, \quad [e_1, e_2] = e_3, \quad [e_1, e_3] = -e_2$$

This corresponds to the product Lie group $\widetilde{E}(2) \times \mathbb{R}\hat{e}_0$.

If $a \neq 0$, then the semi-direct product $\tilde{E}(2) \rtimes_{\varphi} \mathbb{R}\hat{e}_0$ is non-unimodular and its Lie algebra is isomorphic to

$$[\hat{e}_0, e_2] = e_2, \quad [\hat{e}_0, e_3] = e_3, \quad [e_1, e_2] = e_3, \quad [e_1, e_3] = -e_2,$$

which corresponds to the complex affine Lie algebra $\mathfrak{aff}(\mathbb{C})$.

The discussion above can be summarized as follows.

Lemma 1.11. Let $G = \widetilde{E}(2) \rtimes \mathbb{R}$ be a semi-direct extension of the Euclidean Lie group. Then

- (i) G is unimodular if and only if it is isomorphic to the product $\widetilde{E}(2) \times \mathbb{R}$, and
- (ii) G is non-unimodular if and only if it is isomorphic to $\mathfrak{aff}(\mathbb{C})$.

The left-invariant metrics on each four-dimensional solvable Riemannian Lie group can be described in terms of an orthonormal basis as follows.

Left-invariant Riemannian metrics on $H^3 \rtimes \mathbb{R}$

Let $\mathfrak{g} = \mathfrak{h}^3 \rtimes \mathbb{R}$ be the semi-direct extension of the Heisenberg algebra \mathfrak{h}^3 . Let $\langle \cdot, \cdot \rangle$ be an inner product in \mathfrak{g} and let $\langle \cdot, \cdot \rangle_3$ be its restriction to \mathfrak{h}^3 . It follows from Milnor's work [104] that there exists an orthonormal basis $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3\}$ of \mathfrak{h}^3 such that

$$[\mathbf{v}_1, \mathbf{v}_2] = \gamma \mathbf{v}_3, \quad [\mathbf{v}_1, \mathbf{v}_3] = 0, \quad [\mathbf{v}_2, \mathbf{v}_3] = 0, \quad \gamma \neq 0.$$
 (1.11)

The algebra of derivations of \mathfrak{h}^3 with respect to a rotated basis that we will also denote by $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3\}$ is given by

$$\operatorname{Der}(\mathfrak{h}^3) = \left\{ \left(\begin{array}{ccc} \tilde{a} & \tilde{c} & 0\\ -\tilde{c} & \tilde{d} & 0\\ \tilde{h} & \tilde{f} & \tilde{a} + \tilde{d} \end{array} \right) : \tilde{a}, \tilde{c}, \tilde{d}, \tilde{h}, \tilde{f} \in \mathbb{R} \right\}.$$

Let $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3, \mathbf{v}_4\}$ be a basis of \mathfrak{g} with $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3\}$ given by Equation (1.11), and $\mathfrak{g} = \mathfrak{h}^3 \oplus \mathbb{R}\mathbf{v}_4$. Since $\mathbb{R}\mathbf{v}_4$ need not be orthogonal to \mathfrak{h}^3 , we set $\tilde{k}_i = \langle \mathbf{v}_i, \mathbf{v}_4 \rangle$, for i = 1, 2, 3, and consider $\hat{e}_4 = \mathbf{v}_4 - \sum_i \tilde{k}_i \mathbf{v}_i$. If we now normalize it we get an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ of \mathfrak{g} such that

$$[e_1, e_2] = \gamma e_3, \qquad [e_4, e_1] = \frac{1}{R} \{ \tilde{a} e_1 - \tilde{c} e_2 + (\tilde{h} + \tilde{k}_2 \gamma) e_3 \}, [e_4, e_3] = \frac{1}{R} (\tilde{a} + \tilde{d}) e_3, \quad [e_4, e_2] = \frac{1}{R} \{ \tilde{c} e_1 + \tilde{d} e_2 + (\tilde{f} - \tilde{k}_1 \gamma) e_3 \},$$
(1.12)

where R > 0. In order to simplify the expressions we define

$$a = -\frac{\tilde{a}}{R}, \quad c = -\frac{\tilde{c}}{R}, \quad d = -\frac{\tilde{d}}{R}, \quad h = -\frac{\tilde{h}}{R}, \quad f = -\frac{\tilde{f}}{R},$$
$$k_1 = -\frac{\tilde{k}_1}{R}, \quad k_2 = -\frac{\tilde{k}_2}{R}$$

and use the notation $F = f - k_1 \gamma$ and $H = h + k_2 \gamma$. Now the Lie brackets in Equation (1.12) become

$$[e_1, e_2] = \gamma e_3, \qquad [e_1, e_4] = ae_1 - ce_2 + He_3, [e_3, e_4] = (a+d)e_3, \qquad [e_2, e_4] = ce_1 + de_2 + Fe_3.$$
 (1.13)

Left-invariant Riemannian metrics on $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$

Let $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathbb{R}$ be the semi-direct extension of the three-dimensional Lie algebra \mathfrak{g}_3 , being \mathfrak{g}_3 either $\mathfrak{e}(1,1)$ or $\mathfrak{e}(2)$. Let $\langle \cdot, \cdot \rangle$ be an inner product in \mathfrak{g} and let $\langle \cdot, \cdot \rangle_3$ denote its restriction to \mathfrak{g}_3 . According to Milnor's work [104], there exists an orthonormal basis $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3\}$ of \mathfrak{g}_3 such that

$$[\mathbf{v}_2, \mathbf{v}_3] = \lambda_1 \mathbf{v}_1, \quad [\mathbf{v}_3, \mathbf{v}_1] = \lambda_2 \mathbf{v}_2, \quad [\mathbf{v}_1, \mathbf{v}_2] = 0, \tag{1.14}$$

for $\lambda_1, \lambda_2 \in \mathbb{R}$ and $\lambda_1 \lambda_2 \neq 0$. The associated Lie groups correspond to E(2) whenever $\lambda_1 \lambda_2 > 0$ and E(1, 1) whenever $\lambda_1 \lambda_2 < 0$. Moreover, the algebra of derivations of \mathfrak{g}_3 is given by

$$\operatorname{Der}(\mathfrak{g}_3) = \left\{ \left(\begin{array}{ccc} \tilde{b} & \tilde{a} & \tilde{c} \\ -\frac{\lambda_2}{\lambda_1} \tilde{a} & \tilde{b} & \tilde{d} \\ 0 & 0 & 0 \end{array} \right) : \tilde{a}, \tilde{b}, \tilde{c}, \tilde{d} \in \mathbb{R} \right\}.$$

Let $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3, \mathbf{v}_4\}$ be a basis of \mathfrak{g} for which $\operatorname{ad}(\mathbf{v}_4)$ is determined by a derivation as above. After a normalization, like in the previous section, we get an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ for which the non-zero Lie brackets are given by

$$[e_{2}, e_{3}] = \lambda_{1}e_{1}, \qquad [e_{3}, e_{1}] = \lambda_{2}e_{2}, [e_{4}, e_{1}] = \frac{1}{R} \{\tilde{b}e_{1} - \lambda_{2}(\frac{\tilde{a}}{\lambda_{1}} + \tilde{k}_{3})e_{2}\}, [e_{4}, e_{2}] = \frac{1}{R} \{(\tilde{a} + \tilde{k}_{3}\lambda_{1})e_{1} + \tilde{b}e_{2}\}, [e_{4}, e_{3}] = \frac{1}{R} \{(\tilde{c} - \tilde{k}_{2}\lambda_{1})e_{1} + (\tilde{d} + \tilde{k}_{1}\lambda_{2})e_{2}\},$$
(1.15)

where R > 0. In order to simplify the notation, we define

$$a = -\frac{\tilde{a}}{R}, b = -\frac{\tilde{b}}{R}, c = -\frac{\tilde{c}}{R}, d = -\frac{\tilde{d}}{R}, k_1 = -\frac{\tilde{k}_1}{R}, k_2 = -\frac{\tilde{k}_2}{R}, k_3 = -\frac{\tilde{k}_3}{R}$$

and set $A = \frac{a}{\lambda_1} + k_3$, $C = c - k_2 \lambda_1$ and $D = d + k_1 \lambda_2$. Now the Lie brackets given in Equation (1.15) become

$$[e_1, e_3] = -\lambda_2 e_2, \qquad [e_2, e_3] = \lambda_1 e_1,$$

$$[e_1, e_4] = b e_1 - A \lambda_2 e_2, \qquad [e_2, e_4] = A \lambda_1 e_1 + b e_2,$$

$$[e_3, e_4] = C e_1 + D e_2.$$
(1.16)

Left-invariant Riemannian metrics on $\mathbb{R}^3 \rtimes \mathbb{R}$

Let $\mathfrak{g} = \mathfrak{r}^3 \rtimes \mathbb{R}$ be a semi-direct extension of the Abelian Lie algebra \mathfrak{r}^3 . Let $\langle \cdot, \cdot \rangle$ be a inner product on \mathfrak{g} and $\langle \cdot, \cdot \rangle_3$ be its restriction to \mathfrak{r}^3 . The algebra of all the derivations of \mathfrak{r}^3 is $\mathfrak{gl}(3,\mathbb{R})$ and there exists a $\langle \cdot, \cdot \rangle_3$ -orthonormal basis $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3\}$ of \mathfrak{r}^3 where a derivation decomposes as a sum of a diagonal matrix and a skew-symmetric matrix. Therefore, the algebra of derivations of \mathfrak{r}^3 is given by

$$\operatorname{Der}\left(\mathfrak{r}^{3}\right) = \left\{ \left(\begin{array}{ccc} \tilde{a} & -\tilde{b} & -\tilde{c} \\ \tilde{b} & \tilde{f} & -\tilde{h} \\ \tilde{c} & \tilde{h} & \tilde{p} \end{array} \right) : \tilde{a}, \tilde{b}, \tilde{c}, \tilde{f}, \tilde{h}, \tilde{p} \in \mathbb{R} \right\}.$$

Now, the corresponding semi-direct product $\mathfrak{g} = \mathfrak{r}^3 \rtimes \mathbb{R}$ is given by

$$\begin{split} [\mathbf{v}_1, \mathbf{v}_2] &= 0, \qquad [\mathbf{v}_4, \mathbf{v}_1] = \tilde{a}\mathbf{v}_1 + \tilde{b}\mathbf{v}_2 + \tilde{c}\mathbf{v}_3, \\ [\mathbf{v}_1, \mathbf{v}_3] &= 0, \qquad [\mathbf{v}_4, \mathbf{v}_2] = -\tilde{b}\mathbf{v}_1 + \tilde{f}\mathbf{v}_2 + \tilde{h}\mathbf{v}_3, \\ [\mathbf{v}_2, \mathbf{v}_3] &= 0, \qquad [\mathbf{v}_4, \mathbf{v}_3] = -\tilde{c}\mathbf{v}_1 - \tilde{h}\mathbf{v}_2 + \tilde{p}\mathbf{v}_4, \end{split}$$

with respect to some basis $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3, \mathbf{v}_4\}$ so that $\mathfrak{g} = \operatorname{span}\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3\} \oplus \mathbb{R}\mathbf{v}_4$. Since $\mathbb{R}\mathbf{v}_4$ need not be orthogonal to \mathfrak{r}^3 , we can consider $k_i = \langle \mathbf{v}_i, \mathbf{v}_4 \rangle$, for all i = 1, 2, 3, and define $\hat{e}_4 = \mathbf{v}_4 - \sum_i k_i \mathbf{v}_i$. If we normalize it, we obtain an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ of \mathfrak{g} so that

$$[e_4, e_1] = \frac{1}{R} (\tilde{a}e_1 + \tilde{b}e_2 + \tilde{c}e_3), \qquad [e_4, e_2] = \frac{1}{R} (-\tilde{b}e_1 + \tilde{f}e_2 + \tilde{h}e_3),$$

$$[e_4, e_3] = \frac{1}{R} (-\tilde{c}e_1 - \tilde{h}e_2 + \tilde{p}e_3), \quad R > 0.$$
(1.17)

In order to simplify the notation, we define

$$a = \frac{\tilde{a}}{R}, \quad b = \frac{\tilde{b}}{R}, \quad c = \frac{\tilde{c}}{R}, \quad f = \frac{\tilde{f}}{R}, \quad h = \frac{\tilde{h}}{R}, \quad p = \frac{\tilde{p}}{R}.$$

Now the Lie brackets given in Equation (1.17) become

$$[e_1, e_4] = ae_1 + be_2 + ce_3, \qquad [e_2, e_4] = -be_1 + fe_2 + he_3, [e_3, e_4] = -ce_1 - he_2 + pe_3.$$
 (1.18)

Non-solvable Lie groups

Four-dimensional non-solvable Lie groups are isomorphic to either one of the direct products $SU(2) \times \mathbb{R}$ and $\widetilde{SL}(2, \mathbb{R}) \times \mathbb{R}$ and the left-invariant metrics on them can be described in terms of an orthonormal basis as follows.

Left-invariant Riemannian metrics on $SU(2) \times \mathbb{R}$ and $SL(2, \mathbb{R}) \times \mathbb{R}$

Let $\mathfrak{g} = \mathfrak{g}_3 \times \mathbb{R}$ be a direct extension of the unimodular Lie algebra $\mathfrak{g}_3 = \mathfrak{sl}(2, \mathbb{R})$ or $\mathfrak{g}_3 = \mathfrak{su}(2)$. Let $\langle \cdot, \cdot \rangle$ be an inner product on \mathfrak{g} and let $\langle \cdot, \cdot \rangle_3$ denote its restriction to \mathfrak{g}_3 . Following Milnor's work [104], there exists an orthonormal basis { $\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3$ } of \mathfrak{g}_3 such that

$$[\mathbf{v}_2, \mathbf{v}_3] = \lambda_1 \mathbf{v}_1, \quad [\mathbf{v}_3, \mathbf{v}_1] = \lambda_2 \mathbf{v}_2, \quad [\mathbf{v}_1, \mathbf{v}_2] = \lambda_3 \mathbf{v}_3, \tag{1.19}$$

where $\lambda_1, \lambda_2, \lambda_3 \in \mathbb{R}$ and $\lambda_1 \lambda_2 \lambda_3 \neq 0$. Moreover, the associated Lie group corresponds to SU(2) if λ_1, λ_2 and λ_3 have the same sign, and to $SL(2, \mathbb{R})$ otherwise.

Let $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3, \mathbf{v}_4\}$ be a basis of \mathfrak{g} such that $\{\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3\}$ are given by Equation (1.19) and $\mathfrak{g} = \mathfrak{g}_3 \oplus \mathbb{R} \mathbf{v}_4$. Since $\mathbb{R} \mathbf{v}_4$ need not be orthogonal to \mathfrak{g}_3 , we consider $\tilde{k}_i = \langle \mathbf{v}_i, \mathbf{v}_4 \rangle$, for i = 1, 2, 3. Let $\hat{e}_4 = \mathbf{v}_4 - \sum_i \tilde{k}_i \mathbf{v}_i$ and normalize it to get an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \oplus \mathbb{R}$ such that

$$[e_{1}, e_{2}] = \lambda_{3}e_{3}, \qquad [e_{2}, e_{3}] = \lambda_{1}e_{1},$$

$$[e_{3}, e_{1}] = \lambda_{2}e_{2}, \qquad [e_{1}, e_{4}] = \frac{1}{R}(\tilde{k}_{3}\lambda_{2}e_{2} - \tilde{k}_{2}\lambda_{3}e_{3}), \qquad (1.20)$$

$$[e_{2}, e_{4}] = \frac{1}{R}(\tilde{k}_{1}\lambda_{3}e_{3} - \tilde{k}_{3}\lambda_{1}e_{1}), \quad [e_{3}, e_{4}] = \frac{1}{R}(\tilde{k}_{2}\lambda_{1}e_{1} - \tilde{k}_{1}\lambda_{2}e_{2}),$$

where R > 0. In order to simplify the expressions, we define $k_i = \frac{k_i}{R}$, so the Lie brackets now take the form

$$[e_1, e_2] = \lambda_3 e_3, \qquad [e_1, e_3] = -\lambda_2 e_2, \qquad [e_2, e_3] = \lambda_1 e_1,$$

$$[e_1, e_4] = k_3 \lambda_2 e_2 - k_2 \lambda_3 e_3, \qquad [e_2, e_4] = k_1 \lambda_3 e_3 - k_3 \lambda_1 e_1, \qquad (1.21)$$

$$[e_3, e_4] = k_2 \lambda_1 e_1 - k_1 \lambda_2 e_2.$$

1.5 Homogeneous spaces

Roughly speaking, in pseudo-Riemannian geometry, homogeneity means that the geometry of a manifold is the same at each of its points. What this means is that for any two points in the manifold, there exists an isometry sending one to the other. At the same time, in affine geometry, homogeneity means that for any two points there exists an affine transformation sending one point to the other. It is important to be aware of the fact that a pseudo-Riemannian manifold may be affine homogeneous for its Levi-Civita connection but not necessarily homogeneous (see [92]).

Riemannian homogeneous spaces

A connected Riemannian manifold (M, g) is said to be *homogeneous* if its isometry group acts transitively on M, i.e, if for any two points $p, q \in M$ there exists an isometry φ of (M, g) such that $\varphi(p) = q$. In this situation, the connected component of the identity of the isometry group also acts transitively on M. This definition of homogeneity is equivalent to the existence of a connected Lie group G and a smooth map

$$\begin{array}{rccc} G \times M & \longrightarrow & M \\ (q,p) & \longmapsto & L_q(p) = q \, p \end{array}$$

such that

(i) L_q is an isometry of (M, g).

(ii)
$$L_{q_1}L_{q_2} = L_{q_1q_2}$$
.

(iii) For any $p_1, p_2 \in M$, there exists an element $q_1 \in G$ such that $L_{q_1}(p_1) = p_2$.

If G acts effectively on M, i.e., if L_q is the identity transformation of M if and only if q is the identity element $e \in G$, we can always replace G by the quotient group G/K, where K is the kernel of the map $q \in G \mapsto L_q \in \text{Isom}(M)$. Therefore, if G is a connected Lie group that acts on (M, g) as a transitive and effective group of isometries, then G can be identified with a Lie subgroup of the isometry group of (M, g).

Let $p \in M$ and $H = \{q \in G : q p = p\}$ be the isotropy group of p. Then M is diffeomorphic to the quotient G/H and we have the canonical projection

$$\pi \colon G \longrightarrow G/H.$$

This gives a fibre bundle over M with structure group H, where the subgroup H is closed, but not necessarily connected. A Riemannian metric $\langle \cdot, \cdot \rangle$ on G/H is said to be G-invariant if the action

$$t_q \colon sH \in G/H \longmapsto t_q(sH) = q \, sH$$

is an isometry for all $q \in G$. In this case $(G/H, \langle \cdot, \cdot \rangle)$ is called a *Riemannian homogeneous space*. (M, g) is *locally homogeneous* if for each two points $p, q \in M$ there exist neighbourhoods \mathcal{U} of p and \mathcal{V} of q and a local isometry $\varphi : \mathcal{U} \to \mathcal{V}$ such that $\varphi(p) = q$.

Simply connected homogeneous Riemannian manifolds of dimension two are symmetric. Three-dimensional simply connected homogeneous Riemannian manifolds are either symmetric spaces or Lie groups endowed with left-invariant Riemannian metrics (see [126], [103] for a modern presentation and [30] for an extension to the three-dimensional Lorentzian case). Bérard-Bergery showed in [15] that the same result holds true in the four-dimensional Riemannian situation.

Theorem 1.12. Let (M, g) be a four-dimensional simply connected Riemannian homogeneous manifold. Then, it is either symmetric or isometric to a Lie group endowed with a left-invariant metric.

Let (M,g) be a connected *n*-dimensional Riemannian manifold. What is more, consider M = G/H, where G is a subgroup of the group of isometries of M acting transitively and effectively on M, and H is the isotropy group of a point $p \in M$. If we denote by g and h the Lie algebras of G and H, respectively, then M = G/H is said to be *reductive* if there exists a vector subspace m of g such that

$$g = \mathfrak{h} \oplus \mathfrak{m},$$

and \mathfrak{m} is the $\mathrm{ad}(H)$ -invariant subspace of \mathfrak{g} .

1.6 A note on Gröbner bases

Gröbner bases were introduced by Bruno Buchberger around the year 1960 and they have proven themselves to be extraordinarily useful in many different mathematical contexts. The aim of this section is to provide the reader with some basic knowledge on these algebraic objects, since we will be making use of them in different sections throughout this thesis as a tool to solve large system of polynomial equations.

1.6.1 Monomial order and ideals

Given a monomial $x^{\alpha} = x_1^{\alpha_1} \cdots x_n^{\alpha_n}$, the exponents $\alpha = (\alpha_1, \ldots, \alpha_n)$ are elements of $\mathbb{Z}_{\geq 0}^n$, which establishes a one-to-one correspondence between the monomials in the polynomial ring $\mathbb{R}[x_1, \ldots, x_n]$ and $\mathbb{Z}_{\geq 0}^n$. A *monomial order* on $\mathbb{R}[x_1, \ldots, x_n]$ is a relation > on $\mathbb{Z}_{\geq 0}^n$ or, equivalently, on the set of monomials x^{α} , satisfying the following properties.

- 1. > is a total order on $\mathbb{Z}_{>0}^n$.
- 2. If $\alpha > \beta$ and $\gamma \in \mathbb{Z}_{>0}^n$, then $\alpha + \gamma > \beta + \gamma$.
- 3. > is a well-order on $\mathbb{Z}_{>0}^n$.

There are many different monomial orders, but we will be most interested in the following three:

- Lexicographical order: $\alpha >_{lex} \beta$ if the leftmost non-zero entry in the vector $\alpha \beta \in \mathbb{Z}^n$ is positive.
- Graded lexicographical order: $\alpha >_{grlex} \beta$ if $|\alpha| > |\beta|$ or $|\alpha| = |\beta|$ with $\alpha >_{lex} \beta$.
- Graded reverse lexicographical order: $\alpha >_{grevlex} \beta$ if $|\alpha| > |\beta|$ or $|\alpha| = |\beta|$ and the rightmost non-zero entry of $\alpha \beta \in \mathbb{Z}^n$ is negative.

The lexicographical order corresponds to the alphabetical order and a variable dominates any monomial involving only smaller variables, regardless of its total degree. If we want to take into account the total degrees of the monomials so that the monomials of higher degree are the greatest, we can use the graded lexicographical order.

Let $\mathfrak{P} = \sum_{\alpha} a_{\alpha} x^{\alpha}$ be a non-zero polynomial in $\mathbb{R}[x_1, \ldots, x_n]$ and let > be a monomial order. The *multidegree* of \mathfrak{P} is the maximum (with respect to the monomial order >) $\alpha \in \mathbb{Z}_{\geq 0}^n$ so that $a_{\alpha} \neq 0$. The corresponding monomial is called the *leading term* $LT(\mathfrak{P}) = a_{\alpha}x^{\alpha}$. A *monomial ideal* in $\mathbb{R}[x_1, \ldots, x_n]$ is a polynomial ideal that can be generated by monomials. A polynomial \mathfrak{P} belongs to a monomial ideal \mathcal{I} if and only if all of its terms are elements of \mathcal{I} . We denote by $LT(\mathcal{I})$ the set of leading terms of the non-zero elements of \mathcal{I} , i.e.,

$$LT(\mathcal{I}) = \{ cx^{\alpha} \colon \exists \mathfrak{P} \in \mathcal{I} \setminus \{ 0 \} \text{ s.t. } LT(\mathfrak{P}) = cx^{\alpha} \},\$$

and $\langle LT(\mathcal{I}) \rangle$ denotes the ideal generated by the elements of $LT(\mathcal{I})$. Notice that if $\mathfrak{P}_i \in \mathcal{I}$, for $i = 1, \ldots, k$, then $LT(\mathfrak{P}_i) \in LT(\mathcal{I}) \subset \langle LT(\mathcal{I}) \rangle$ and so

$$\langle LT(\mathfrak{P}_1),\ldots,LT(\mathfrak{P}_k)\rangle \subset \langle LT(\mathcal{I})\rangle.$$

However, if $\mathcal{I} = \langle \mathfrak{P}_1, \dots, \mathfrak{P}_k \rangle$, $\langle LT(\mathcal{I}) \rangle$ might be strictly larger than $\langle LT(\mathfrak{P}_1), \dots, LT(\mathfrak{P}_k) \rangle$. Consider, for instance, the ideal $\mathcal{I} = \langle \mathfrak{P}_1, \mathfrak{P}_2 \rangle$ where

$$\mathfrak{P}_1 = y^3 - 2xy, \qquad \mathfrak{P}_2 = xy^2 - 2x^2 + y,$$

and fix the graded lexicographical order for monomials. The polynomial

$$y\mathfrak{P}_2 - x\mathfrak{P}_1 = y^2$$

belongs to \mathcal{I} and $y^2 = LT(x^2) \in \langle LT(\mathcal{I}) \rangle$, but

$$y^2 \notin \langle LT(\mathfrak{P}_1), LT(\mathfrak{P}_2) \rangle = \langle y^3, xy^2 \rangle.$$

Theorem 1.13 (Hilbert Basis Theorem). Every ideal $\mathcal{I} \subset \mathbb{R}[x_1, \ldots, x_n]$ has a finite generating set.

The analogue result for monomial ideals is called Dickson's Lemma. The Hilbert Basis Theorem guarantees that any non-zero ideal $\mathcal{I} \subset \mathbb{R}[x_1, \ldots, x_n]$ admits a Gröbner basis.

1

Definition 1.14. A finite subset $\mathcal{G} = \{g_1, \dots, g_\nu\}$ of an ideal \mathcal{I} with a fixed monomial order such that

$$\langle LT(g_1), \ldots, LT(g_{\nu}) \rangle = \langle LT(\mathcal{I}) \rangle$$

is said to be a *Gröbner basis* (or a *Gröbner-Shirshov basis*) with respect to the given monomial order.

Gröbner bases provide us with a powerful tool to find quite simple algorithmic solutions to various algebraic problems. For example:

The Ideal Membership Problem

The remainder of the division algorithm applied to a polynomial \mathfrak{P} divided by the elements of a Gröbner basis \mathcal{G} of an ideal \mathcal{I} is zero if and only if \mathfrak{P} belongs to \mathcal{I} , and this property does not necessarily hold if \mathcal{G} is not a Gröbner basis.

Solving large systems of polynomial equations

Let $\{\mathfrak{P}_i\}$ be a set of polynomials in $\mathbb{R}[x_1, \ldots, x_n]$ and consider the system of polynomial equations given by $\{\mathfrak{P}_i = 0\}$. If the system in consideration is simple, it is an elementary problem to find all the common roots, but if the number of unknowns, equations and their degree increase, then finding all the solutions might become quite an unmanageable problem. What can one do to make the task easier?

If two sets of polynomials generate the same ideal, the corresponding zero sets must be identical. The theory of Gröbner bases provides a well-known strategy to solve rather large polynomial systems obtaining "better" polynomials that belong to the ideal generated by the initial polynomial system.

We refer to [53] for more information regarding the theory of Gröbner bases.

1.6.2 Buchberger's algorithm

Buchberger's algorithm is the oldest algorithm ever introduced for the computation of Gröbner bases. It was devised by Bruno Buchberger at the same time that he introduced Gröbner bases. A crude version of this algorithm to find a Gröbner basis of an ideal of polynomials \mathcal{I} proceeds as follows:

Input: A set of polynomials P that generate \mathcal{I} . **Output:** A Gröbner basis \mathcal{G} for \mathcal{I} .

- (1) Set $\mathcal{G} = P$.
- (2) If $p_i, p_j \in P$ and we denote by f_i, f_j the coefficients of their leading terms with respect to a given monomial ordering, respectively, set $a_{ij} = \text{lcm}\{f_i, f_j\}$.
- (3) Define $g_{ij} = \frac{a_{ij}}{f_i} p_i \frac{a_{ij}}{f_j} p_j$. Note that the leading terms here will cancel. The polynomials p_i and p_j are called a *critical pair*.

- (4) Reduce g_{ij} as much as possible using the multivariate division algorithm with respect to \mathcal{G} . If the result is non-zero, then add g_{ij} to \mathcal{G} .
- (5) Repeat (2)–(4) until all possible pairs have been considered, including those involving the new polynomials added to \mathcal{G} in step (4).

There are many ways in which the algorithm above can be improved and, in fact, has been improved (see, for instance, [64, 65]).

Part I

Locally conformally flat structures

In this part we will devote ourselves to the study of Kähler, para-Kähler and null-Kähler structures. In Chapter 2 we will study the geometry of locally conformally flat four-dimensional structures and in Chapter 3 we will give the classification of four-dimensional para-Kähler Lie groups. Before we start, we will briefly outline some basic notions and notation on these structures.

Kähler structures

If a real 2n-dimensional manifold M admits a globally defined (1, 1)-tensor field J such that

$$J^2 = -\mathrm{Id},$$

then (M, J) is said to be an *almost complex manifold* and J is an *almost complex structure* on M. The almost complex structure is *integrable* if it corresponds to the underlying structure of a complex manifold – this is, a smooth manifold admitting a holomorphic atlas. Newlander and Nirenberg showed in [108] that the integrability of an almost complex structure on a manifold M is equivalent to the vanishing of its associated Nijenhuis tensor

$$N_J(X,Y) = [JX,JY] - J[JX,Y] - J[X,JY] + J^2[X,Y].$$
 (I.1)

A pseudo-Riemannian metric g on (M, J) is an *almost Hermitian metric* if the almost complex structure J is an isometry in each tangent space, i.e.,

$$g(JX, JY) = g(X, Y)$$
 for all $X, Y \in \mathfrak{X}(M)$.

Now the triple (M, g, J) is called an *almost Hermitian manifold*. In particular, if J is integrable, then (M, g, J) is a *Hermitian manifold*.

There always exists a non-degenerate two-form associated to any almost Hermitian structure (g, J) which is called the *Kähler form* and given by

$$\Omega(X,Y) = g(JX,Y).$$

This two-form is covariantly constant and so it is closed. The covariant derivative of an almost complex structure, its Nijenhuis tensor and its associated Kähler two-form are related by

$$2g((\nabla_X J)Y, Z) + 3d\Omega(X, Y, Z) - 3d\Omega(X, JY, JZ) - g(JX, N_J(Y, Z)) = 0.$$

An (almost) Hermitian manifold (M, g, J) whose associated Kähler two-form is closed is said to be an (*almost*) *Kähler* manifold. In other words, Kähler manifolds are characterized by the parallelizability of their complex structure, $\nabla J = 0$, and their curvature tensor satisfies

$$R(X, Y, Z, W) = R(JX, JY, Z, W).$$

As a consequence, any Kähler manifold of constant sectional curvature is necessarily flat. The restriction of the sectional curvature to non-degenerate planes Π which are invariant under the action of the complex structure J is called the *holomorphic sectional curvature* and given by

$$H(\Pi) = \frac{R(X, JX, X, JX)}{g(X, X)^2}$$

Notice that the holomorphic sectional curvature determines the curvature tensor of Kähler manifolds. What is more, a Kähler manifold has constant holomorphic sectional curvature c if and only if its curvature tensor decomposes as

$$R = \frac{c}{4} \left(R^0 + R^J \right)$$

where R^0 is the standard algebraic curvature tensor given in (1.2) and

$$R^{J}(X,Y)Z = g(JX,Z)JY - g(JY,Z)JX + 2g(JX,Y)JZ.$$

Such a Kähler manifold is locally isometric either to the complex space \mathbb{C}_{ν}^{2n} if c = 0, or to the complex projective space \mathbb{CP}_{ν}^{2n} if c > 0, or to the complex hyperbolic space \mathbb{CH}_{ν}^{2n} if c < 0 (see [13]).

An almost Hermitian manifold (M, g, J) is said to be *locally conformally Kähler* (resp. *locally conformally symplectic*) if there exists a local conformal deformation $\bar{g} = e^{2\sigma}g$ so that (M, \bar{g}, J) is a Kähler (resp. symplectic) manifold. Some characterizations for these kinds of manifolds have been given in [59, 131] as follows.

- (M, g, J) is locally conformally symplectic if and only if its associated two-form satisfies $d\Omega = \theta \wedge \Omega$ for some other one-form θ such that $d\theta = 0$.
- (M, g, J) is locally conformally Kähler if and only if it is locally conformally symplectic and J is integrable.

Assume that (M, g, J) is a four-dimensional Kähler manifold oriented so that the associated Kähler form is self-dual, i.e., $\Omega \in \Lambda^2_+$. Then its self-dual Weyl curvature operator satisfies

$$\mathcal{W}^+ = \frac{\tau}{12} \text{diag}[2, -1, -1].$$

Therefore, the self-dual part of the Weyl curvature tensor of any locally conformally Kähler metric has two equal eigenvalues and a distinguished one. Derdziński proved in [56] the following converse to this statement.

Theorem 1.15. Let (M, g) be an oriented four-dimensional Riemannian Einstein manifold such that W^+ has at most two different eigenvalues at each point. Then the metric $\overline{g} = (24 \|W^+\|^2)^{\frac{1}{3}}g$ is Kähler on the open set where $W^+ \neq 0$.

Para-Kähler structures

A (1, 1)-tensor field J on a 2n-dimensional manifold M is said to be an *almost product structure* if $J^2 = \text{Id}$. In this case, the pair (M, J) is called an *almost product manifold*. If the eigenspaces associated to the eigenvalues ± 1 of J have the same rank, the almost product manifold (M, J) is said to be an *almost para-complex manifold* and J is an *almost para-complex structure* on M.

A manifold endowed with an almost para-complex structure J and a metric tensor g for which J is an anti-isometry at each point, i.e., such that

$$g(JX, JY) = -g(X, Y),$$

is called an *almost para-Hermitian manifold*. Any almost para-Hermitian manifold has an associated almost symplectic form, i.e., a non-degenerate two-form, which is given by

$$\Omega(X,Y) = g(JX,Y),$$

for any vector fields X, Y on M. The pair (M, Ω) where Ω is an almost symplectic form is called an *almost symplectic manifold*. If $L \subset M$ is an *n*-dimensional submanifold of a 2*n*-dimensional symplectic manifold such that $\Omega|_L = 0$, then L is a *Lagrangian submanifold* of M. An almost symplectic manifold is an almost para-Hermitian manifold if its tangent bundle decomposes as a Whitney sum of Lagrangian subbundles. Indeed, in that case, $TM = L_1 \oplus L_2$ and the (1, 1)tensor field defined as $J = \pi_{L_1} - \pi_{L_2}$, where π_{L_i} are the projections from TM on L_i , determines an almost para-complex structure on M and the metric tensor is determined by the para-complex structure and the two-form Ω as $g(X, Y) = \Omega(JX, Y)$.

A para-Kähler manifold is a symplectic manifold that is diffeomorphic to a product of two Lagrangian submanifolds. For such manifolds, there is a relation between Ω and the covariant derivative and integrability of J given by the equation

$$2g((\nabla_X J)Y, Z) + 3d\Omega(X, Y, Z) + 3d\Omega(X, JY, JZ) + g(JX, N_J(Y, Z)) = 0,$$

where N_J is de Nijenhuis tensor associated to the almost para-complex structure J given as in (I.1). This relation allows us to characterize para-Kähler manifolds by the parallelism of J. Therefore, a triple (M, g, J) is a *para-Kähler manifold* if and only if

$$J^2 = \mathrm{Id}, \quad g(JX, JY) = -g(X, Y), \quad \mathrm{and} \quad \nabla J = 0.$$

In this case, the curvature tensor satisfies that

$$R(X, Y, Z, W) = -R(JX, JY, Z, W).$$

The restriction of the sectional curvature to non-degenerate planes Π that are invariant under the action of the para-complex structure J is called the *para-holomorphic sectional curvature* and given by

$$H(\Pi) = -\frac{R(X, JX, X, JX)}{g(X, X)^2}.$$

As it happened in the Kähler case, the para-holomorphic sectional curvature determines the curvature tensor of a para-Kähler manifold and it is constant if and only if it can be written as

$$R = \frac{c}{4}(R^0 - R^J),$$

where R^0 is the standard algebraic curvature tensor given in (1.2) and

$$R^{J}(X,Y)Z = g(JX,Z)JY - g(JY,Z)JX + 2g(JX,Y)JZ.$$

A para-Kähler manifold of constant para-holomorphic sectional curvature is locally isometric (or maybe anti-isometric) to \mathbb{R}^{2n} if c = 0, or to the para-complex projective space $\mathbb{P}^m(\mathbb{B})$ if $c \neq 0$ (see [74]), and in the latter case, it is isometric to the cotangent bundle of a flat affine manifold equipped with a suitable Riemannian extension [41].

The Bochner curvature tensor was introduced by S. Bochner in [20]. It is formally defined as an analogue to the Weyl curvature tensor, so that the curvature of a Bochner-flat manifold is completely determined by its Ricci tensor. The *Bochner curvature tensor* of a 2*n*-dimensional para-Kähler manifold is defined as

$$B(X,Y)Z = R(X,Y)Z + \frac{\tau}{(2n+2)(2n+4)}R^0(X,Y)Z - \frac{1}{2(n+2)}R^1(X,Y)Z$$

for all vector fields X, Y, Z on the manifold, where R^0 is the standard algebraic curvature tensor given in (1.2) and

$$R^{1}(X,Y)Z = g(X,Z)\operatorname{Ric}(Y) - g(Y,Z)\operatorname{Ric}(X) + g(X,JZ)\operatorname{Ric}(JY) - g(Y,JZ)\operatorname{Ric}(JX) + 2g(X,JY)\operatorname{Ric}(JZ) + \rho(X,Z)Y - \rho(Y,Z)X + \rho(X,JZ)JY - \rho(Y,JZ)JX + 2\rho(X,JY)JZ.$$

A para-Kähler manifold is said to be *Bochner-flat* if its Bochner tensor vanishes identically. Even though the condition of being Bochner-flat is somehow analogous to that of being locally conformally flat, it is more restrictive. Besides, the anti-self-dual curvature tensor of an oriented para-Kähler manifold is completely determined by its Bochner tensor, so $W^- = 0$ if and only if the manifold is Bochner-flat [27].

Null-Kähler structures

A (1, 1)-tensor field J on a 4n-dimensional manifold M is said to be an *almost tangent structure* if $J^2 = 0$. If, in addition, rank(J) = 2n, J is said to be a *null structure* on M.

A metric tensor g on M is said to be *null-Hermitian* if g(JX, Y) = g(X, JY) for all vector fields X, Y on M, where J is a null structure on M. This implies that g(X, JX) = 0. The signature of a null-Hermitian metric on M is neutral (2n, 2n) and to each null-Hermitian metric there is an associated two-form defined by $\Omega(X, Y) = g(JX, Y)$. The kernel of a null structure is an integrable distribution of the tangent bundle and this integrability condition is equivalent to the vanishing of the associated Nijenhuis tensor given by (I.1).

A null-Hermitian manifold (M, g, J) is said to be *null-Kähler* if $\nabla J = 0$, where ∇ denotes de Levi-Civita connection of g. The fundamental two-form given by $\Omega(X, Y) = g(JX, Y)$ is covariantly-constant and so it is closed. Besides, it satisfies $\Omega^{\wedge n} = \Omega \wedge \cdots \wedge \Omega \neq 0$ and $\Omega^{\wedge (n+1)} = 0$, in contrast with the Kähler and para-Kähler conditions, where $\Omega^{\wedge 2n} \neq 0$.

We refer to Dunajski's works [60,61] for more information about null-Kähler structures.

Notation

In what follows, (M,g,J_{ε}) with

$$J_{\varepsilon}^2 = -\varepsilon \operatorname{Id}, \quad g(J_{\varepsilon}X, Y) + g(X, J_{\varepsilon}Y) = 0, \quad \nabla J_{\varepsilon} = 0,$$

will denote a Kähler, a para-Kähler or a null-Kähler manifold for $\varepsilon = 1$, $\varepsilon = -1$, or $\varepsilon = 0$, respectively.

Chapter 2 Locally conformally flat four-dimensional structures

The contents of this chapter regarding locally conformally flat Kähler and para-Kähler surfaces are contained in the work [68].

2.1 Locally conformally flat four-dimensional Kähler and para-Kähler structures

Let (M, g, J_{ε}) be a (para-)Kähler manifold, where $J_{\varepsilon}^2 = -\varepsilon \operatorname{Id}$ is the (para-)complex structure satisfying $g(J_{\varepsilon}X, J_{\varepsilon}Y) = \varepsilon g(X, Y)$ for $\varepsilon = \pm 1$. Since the (para-)complex structure is parallel, the curvature identity $R(X, Y) \cdot J_{\varepsilon} = J_{\varepsilon} \cdot R(X, Y)$, which strictly restricts the curvature tensor, holds. A consequence of this is that the Ricci tensor of a 2*n*-dimensional locally conformally flat (para-)Kähler manifold satisfies

$$(2n-4) \rho(X,Y) = -\frac{\tau}{2n-1}g(X,Y)$$

and the manifold is flat if n is greater than or equal to three. In addition, the Ricci tensor of a locally conformally flat (para-)Kähler manifold is parallel an so (M, g) is locally symmetric in dimension four [111, 129]. Tanno proved in [129] that locally conformally flat positive definite four-dimensional Kähler manifolds are either flat or a product of two surfaces of constant opposite curvature. This result does not cover all the possibilities in the pseudo-Riemannian case of split signature, as Patterson pointed out in [119]. Motivated by this and some recent interest in locally conformally flat (para-)Kähler surfaces [2,4,82], in this chapter we will give the complete classification of locally conformally flat (para-)Kähler surfaces the classification in the para-Kähler setting is given by Theorem 2.1, which is the main result in this chapter.

Theorem 2.1. Any indecomposable locally conformally flat para-Kähler surface (M, g, J_{-}) is locally isometric to the cotangent bundle $T^*\Sigma$ of a flat affine surface (Σ, D) with a para-complex structure determined by $J|_{\ker \pi_*} = \operatorname{Id}$, where π denotes the canonical projection from the cotangent bundle, and the metric g is given by $g = \iota T \circ \iota \operatorname{Id} + g_D$ where

- (i) T is a parallel nilpotent (1, 1)-tensor field on (Σ, D) , or
- (ii) T is a parallel (1,1)-tensor field on (Σ, D) satisfying $T^2 = -\kappa^2 \operatorname{Id}$.

Moreover, in both cases the para-Kähler two-form $\Omega_{-}(X,Y) = g(J_{-}X,Y)$ is the canonical symplectic form of the cotangent bundle.

Recall that a pseudo-Riemannian manifold is indecomposable if it does not admit a nondegenerate subspace that is invariant under the action of its holonomy group. Besides, the holonomy group may act indecomposably without acting irreducibly.

Para-Kähler surfaces in Case (*i*) in Theorem 2.1 had already been reported by Patterson in [119], while Case (*ii*) seems to be missing in previous works.

Considering the Ricci operator in Case (*ii*), $\frac{1}{\kappa}$ Ric defines a self-adjoint complex structure that is parallel, and so it is a Riemannian complex structure. Since the Ricci operator and the para-complex structure J_{-} commute with each other, one has that $J_{+} = \frac{1}{\kappa}$ Ric J_{-} is a complex structure so that (g, J_{+}) is a locally conformally flat Kähler structure. Therefore, the result that completes the classification in the Kähler setting is given by Theorem 2.2.

Theorem 2.2. Any indecomposable locally conformally flat Kähler surface (M, g, J_+) is locally isometric to the cotangent bundle $(T^*\Sigma, g)$ of a Riemannian surface (Σ, g_{Σ}) of constant curvature with a metric given by

- (i) $g = g_{\nabla^{\Sigma}}$ if the Gaussian curvature is non-zero, or
- (ii) $g = \iota J_{\Sigma} \circ \iota \operatorname{Id} + g_{\nabla^{\Sigma}}$ if the Gaussian curvature vanishes,

where ∇^{Σ} is the Levi-Civita connection of (Σ, g_{Σ}) and J_{Σ} is the Kähler structure on Σ associated to the Riemannian volume form. Furthermore, the complex structure J_+ on $T^*\Sigma$ is determined by the symplectic form $\Omega_+ = -d\iota J_{\Sigma}$.

Remark 2.3. In Chapter 3 we will see that it is possible to give a description of the structures in Theorems 2.1 and 2.2 in terms of left-invariant metrics on Lie groups.

The metric tensors in *(ii)* in both Theorem 2.1 and Theorem 2.2 are the same, but their associated symplectic structures and underlying geometries are different. The metrics in Case *(i)* in both theorems above correspond to different curvature models that will be described in Section 2.1.1.

The Kähler metrics in Theorem 2.2-(*i*) are locally isometric (up to reversing the metric) to those studied by Guilfoyle and Klingenberg on the space of oriented affine lines in \mathbb{R}^3 by means of the minitwistor correspondence in [82]. Analogously, the metrics in Theorem 2.1-(*i*) are locally isometric (up to reversing the metric) to the space of oriented spacelike or timelike lines in \mathbb{R}^3_1 (see [2,4]). The metrics corresponding to Assertion (*ii*) in Theorem 2.1 and Theorem 2.2 are locally isometric (up to reversing the metric) to the non-Einstein para-Kähler and Kähler metrics in the space of spacelike and timelike oriented geodesics of the de Sitter space constructed by Anciaux in [4]. Besides, the non-locally conformally flat Kähler-Einstein metrics on the de Sitter space constructed in [4] correspond to those in Remark 2.13.

2.1.1 Curvature models

In this section we will work at a purely algebraic level to describe the curvature models found in our description of locally conformally flat Kähler and para-Kähler surfaces. Bearing this in mind, let $(\mathcal{V}, \langle \cdot, \cdot \rangle, J_{\pm})$ be a (para-)Hermitian inner product space. Recall that an algebraic curvature tensor is a multilinear map

$$A: \mathcal{V} \times \mathcal{V} \times \mathcal{V} \times \mathcal{V} \to \mathbb{R}$$

satisfying the identities (1.1). If the corresponding Weyl curvature tensor vanishes, then the algebraic curvature tensor A is determined by the associated Ricci tensor and a straightforward calculation shows that the Ricci operator of a locally conformally flat (para-)Kähler surface is either diagonalizable or (following the discussion in [112]) its Jordan normal form corresponds to one of the following.

1. The Ricci operator has two 2×2 Jordan blocks. At each point there is a basis $\{u_1, v_1, u_2, v_2\}$ of the tangent space so that the Ricci operator and the non-zero inner products are given by

$$\operatorname{Ric} = \begin{pmatrix} 0 & 0 & 0 & 0 \\ 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 \end{pmatrix}, \qquad \langle u_i, v_i \rangle = \varepsilon_i, \quad \varepsilon_i^2 = 1 \quad (i = 1, 2).$$

Besides, if the associated Weyl curvature vanishes, there are two different possibilities as $\varepsilon_1 \varepsilon_2 = \pm 1$ (up to reversing the metric).

1.a. If $\varepsilon_1 \varepsilon_2 = 1$, there is a unique (up to sign) Ricci-commuting Hermitian structure $(\langle \cdot, \cdot \rangle, J_+)$ given by

$$J_+u_1 = -u_2, \quad J_+v_1 = -v_2$$

and there are no Ricci-commuting para-Hermitian structures.

1.b. If $\varepsilon_1 \varepsilon_2 = -1$, there is a unique (up to sign) Ricci-commuting para-Hermitian structure $(\langle \cdot, \cdot \rangle, J_-)$ given by

$$J_{-}u_1 = v_2, \quad J_{-}v_1 = u_2$$

and there are no Ricci-commuting Hermitian structures.

2. The Ricci operator is complex diagonalizable with imaginary eigenvalues $\pm i\kappa$. At each point there is a basis $\{u_1, v_1, u_2, v_2\}$ of the tangent space so that the Ricci operator and the non-zero inner products are given by

$$\operatorname{Ric} = \begin{pmatrix} 0 & \kappa & 0 & 0 \\ -\kappa & 0 & 0 & 0 \\ 0 & 0 & 0 & \kappa \\ 0 & 0 & -\kappa & 0 \end{pmatrix}, \qquad \langle u_i, u_i \rangle = 1 = -\langle v_i, v_i \rangle, \quad (i = 1, 2).$$

One may assume that $\kappa > 0$ so that there are a unique (up to reversing the metric) Riccicommuting para-Hermitian structure $(\langle \cdot, \cdot \rangle, J_{-})$ given by

$$J_{-}u_1 = v_2, \quad J_{-}v_1 = -u_2$$

and a unique (up to reversing the metric) Ricci-commuting Hermitian structure $(\langle \cdot, \cdot \rangle, J_+)$ such that

$$J_+u_1 = u_2, \quad J_+v_1 = -v_2.$$

Regarding the discussion above, we introduce the following locally conformally flat algebraic curvature models $(\mathcal{V}, \langle \cdot, \cdot \rangle, A)$ given by $A = \frac{1}{2} \langle \cdot, \cdot \rangle \odot \rho_A$, where ρ_A denotes the Ricci tensors corresponding to the Ricci operators above and \odot is the Kulkarni–Nomizu's product.

 $(\mathfrak{M}^+): (\mathcal{V}, \langle \cdot, \cdot \rangle, A)$ given by

$$A_{1413} = A_{3231} = \frac{1}{2}$$

with respect to a basis $\{u_1, u_2, u_3, u_4\}$ for which the non-zero inner products are

$$\langle u_1, u_2 \rangle = 1 = \langle u_3, u_4 \rangle.$$

 $(\mathfrak{M}^{-}): (\mathcal{V}, \langle \cdot, \cdot \rangle, A)$ given by

$$A_{1413} = A_{3231} = -\frac{1}{2}$$

with respect to a basis $\{u_1, u_2, u_3, u_4\}$ for which the non-zero inner products are

$$\langle u_1, u_2 \rangle = 1 = -\langle u_3, u_4 \rangle$$

 $(\mathfrak{N}_{\kappa}):$ $(\mathcal{V},\langle\cdot,\cdot
angle,A)$ given by

$$A_{1413} = A_{1442} = A_{3224} = A_{3231} = \frac{\kappa}{2}$$

with respect to a basis $\{u_1, u_2, u_3, u_4\}$ where u_1 and u_3 are spacelike vectors and u_2 and u_4 are timelike vectors. Notice that, even though the curvature models (\mathfrak{N}_{κ}) are not isometric, they are all homothetic to the curvature model (\mathfrak{N}_1) .

We will see that the curvature tensors of the locally conformally flat para-Kähler manifolds in Assertions (*i*) and (*ii*) in Theorem 2.1 are modelled on (\mathfrak{M}^-) and (\mathfrak{N}_k) , respectively. The curvature tensors of the locally conformally flat Kähler manifolds in Assertions (*i*) and (*ii*) in Theorem 2.2 are modelled on (\mathfrak{M}^+) and (\mathfrak{N}_k) , respectively.

2.1.2 Self-dual Walker manifolds

Let (M, g, \mathcal{D}) be a four-dimensional Walker manifold, i.e., a pseudo-Riemannian manifold of split signature (M, g) admitting a parallel, degenerate plane field \mathcal{D} of maximal dimension. As we have already mentioned in Section 1.3.1, there exist local coordinates $(x^1, x^2, x_{1'}, x_{2'})$ so that the Walker distribution is $\mathcal{D} = \text{span} \{\partial_{x_{1'}}, \partial_{x_{2'}}\}$ and the metric takes the form

$$g = dx^{i} \otimes dx_{i'} + dx_{i'} \otimes dx^{i} + g_{ij} \left(x^{1}, x^{2}, x_{1'}, x_{2'} \right) dx^{i} \otimes dx^{j}.$$
(2.1)

The existence of a two-dimensional degenerate distribution \mathcal{D} on a split-signature fourdimensional manifold (M, g) naturally induces an orientation. Let $\{u, v\}$ be a basis of \mathcal{D}_p for $p \in M$, and denote by u^* and v^* their corresponding dual forms. The Hodge-star operator satisfies $\star (u^* \wedge v^*) = \pm (u^* \wedge v^*)$ and so, any four-dimensional Walker manifold is naturally oriented by the self-duality of $u^* \wedge v^*$. Considering local coordinates as in (2.1), the Walker orientation determined by

$$\star (dx_{1'} \wedge dx_{2'}) = dx_{1'} \wedge dx_{2'}$$

corresponds to the volume element $vol_g = dx^1 \wedge dx^2 \wedge dx_{1'} \wedge dx_{2'}$.

Self-dual Walker manifolds have been described by Calviño-Louzao, García-Río, Gilkey and Vázquez-Lorenzo as follows.

Theorem 2.4. ([41], Theorem 7.1) A four-dimensional Walker manifold is self-dual if and only if it is locally isometric to the cotangent bundle $T^*\Sigma$ of an affine surface (Σ, D) with metric

$$g = \iota X \left(\iota \operatorname{Id} \circ \iota \operatorname{Id}\right) + \iota T \circ \iota \operatorname{Id} + g_D + \pi^* \Phi$$
(2.2)

where X is a vector field on Σ and T and Φ are a (1, 1)-tensor field and a symmetric (0, 2)-tensor field on Σ , respectively.

A special case of this result describes the local structure of para-complex space forms as follows.

Theorem 2.5. ([41], Theorem 2.2) A para-Kähler surface of non-zero constant para-holomorphic sectional curvature *c* is locally isometric to the cotangent bundle of a flat affine surface equipped with the modified Riemannian extension $g = c \iota \operatorname{Id} \circ \iota \operatorname{Id} + g_D$.

Consider $\theta_{(p,\omega)} = \pi^* \omega_p = x_{\ell'} dx^{\ell}$ the tautological one-form of $T^* \Sigma$ and let

$$\Omega = d\theta = dx_{\ell'} \wedge dx^{\ell}$$

be the canonical symplectic form of $T^*\Sigma$. Given the modified Riemannian extension

$$g = c \iota \operatorname{Id} \circ \iota \operatorname{Id} + g_D,$$

one naturally has a para-complex structure J_{-} determined by $\Omega(X,Y) = g(J_{-}X,Y)$, whose components are

$$J_{-}\partial_{x_{i'}} = \partial_{x_{i'}}, \quad J_{-}\partial_{x^i} = -\partial_{x^i} + cx_{i'}x_{j'}\partial_{x_{i'}},$$

and the Walker distribution $\mathcal{D} = \ker \pi_*$ corresponds to the eigenspace $\mathcal{D}_+ = \ker (J_- - \mathrm{Id})$.

2.1.3 Locally symmetric self-dual Walker surfaces

Given that locally conformally flat Kähler surfaces with non-diagonalizable Ricci operator and locally conformally flat para-Kähler surfaces are locally symmetric, we study first which selfdual Walker surfaces in Theorem 2.4 are locally symmetric.

Let ${}^{D}\rho$ denote the Ricci tensor of (Σ, D) and decompose it as ${}^{D}\rho = {}^{D}\rho_{s} + {}^{D}\rho_{sk}$, where ${}^{D}\rho_{s}$ and ${}^{D}\rho_{sk}$ denote the symmetric and the skew-symmetric parts of ${}^{D}\rho$, respectively.

Lemma 2.6. Let (M, g) be a locally symmetric self-dual Walker manifold. Then the Riemannian extension g in Theorem 2.4 satisfies $g = \iota T \circ \iota \operatorname{Id} + g_D + \pi^* \Phi$ for some parallel (1, 1)-tensor field T on (Σ, D) .

Furthermore, if the scalar curvature is not zero, then (M, g) is locally isometric to a paracomplex space form as in Theorem 2.5.

Proof. The scalar curvature of any Riemannian extension in Theorem 2.4 is given by

$$\tau = 3 \operatorname{tr} T + 12\iota X.$$

Since the scalar curvature is constant, then X = 0 and tr $T = \kappa$ for some $\kappa \in \mathbb{R}$. Given the fact that any self-dual Walker manifold is locally isometric to a Riemannian extension given by Theorem 2.4, the covariant derivatives of the curvature operator are polynomials on the fibre coordinates $(x_{1'}, x_{2'})$.

$$\begin{aligned} (\nabla_{\partial_{x^1}} R)(\partial_{x^2}, \partial_{x^1}, \partial_{x^1}, \partial_{x^1}) &= -\frac{1}{8}\kappa \left(T_1{}^2\right)^2 x_{2'}^3 + \text{ other terms,} \\ (\nabla_{\partial_{x^2}} R)(\partial_{x^2}, \partial_{x^1}, \partial_{x^1}, \partial_{x^1}) &= \frac{1}{4}\kappa \left(T_2{}^1\right)^2 x_{1'}^3 + \text{ other terms,} \\ (\nabla_{\partial_{x^2}} R)(\partial_{x_{2'}}, \partial_{x^1}, \partial_{x^1}, \partial_{x^1}) &= -\frac{\kappa}{8} \left(\kappa - 2T_2{}^2\right) x_{1'} + \text{ other terms,} \end{aligned}$$

Assume in the first place that the scalar curvature $\tau = 3\kappa \neq 0$. If $\nabla R = 0$ then the previous expressions show that the tensor field T must be a scalar multiple of the identity, $T = c \operatorname{Id}$. Further calculations now show that

$$\begin{split} (\nabla_{\partial_{x^{1}}}R)(\partial_{x^{2}},\partial_{x^{1}},\partial_{x^{1}},\partial_{x^{1}}) &= c\{2^{D}\rho_{s}(\partial_{x^{1}},\partial_{x^{1}}) - \frac{1}{2}c\Phi_{11}\}x_{2'} + \text{ other terms,} \\ (\nabla_{\partial_{x^{2}}}R)(\partial_{x^{2}},\partial_{x^{1}},\partial_{x^{1}},\partial_{x^{1}}) &= \frac{3}{2}c\{2^{D}\rho_{s}(\partial_{x^{1}},\partial_{x^{2}}) - \frac{1}{2}c\Phi_{12}\}x_{2'} + \text{ other terms,} \\ (\nabla_{\partial_{x^{2}}}R)(\partial_{x^{2}},\partial_{x^{1}},\partial_{x^{1}},\partial_{x_{1'}}) &= \frac{1}{2}c^{2}\{2^{D}\rho_{s}(\partial_{x^{2}},\partial_{x^{2}}) - \frac{1}{2}c\Phi_{22}\}x_{1'}^{3} + \text{ other terms,} \end{split}$$

from where it follows that the symmetric (0, 2)-tensor field $\Phi = \frac{4}{c} D \rho_s$. In this situation, the Ricci operator $\operatorname{Ric} = \frac{3c}{2}$ Id and, according to [41, Theorem 2.1], the corresponding metric is Einstein. Furthermore, one has that for any unit vector field X, the Jacobi operators $R_X = R(\cdot, X)X$ have eigenvalues $\{0, c, \frac{1}{4}c, \frac{1}{4}c\}$ and the eigenspace associated to c is timelike. Consequently, (M, g) is locally a para-complex space form (see [77]), thus being locally isometric to a modified Riemannian extension given by Theorem 2.5.

Now assume that the scalar curvature $\tau = 0$. Let DT denote the covariant derivative of the (1, 1)-tensor field T with respect to the affine connection D. We set

$$DT = DT_{j;i}{}^k dx^i \otimes dx^j \otimes \partial_{x^k},$$

where $DT_{j;i}{}^{k} = \partial_{x^{i}}T_{j}{}^{k} + T_{j}{}^{\ell D}\Gamma_{i\ell}{}^{k}$. Since the scalar curvature vanishes, tr T = 0. Hence $T_{1}{}^{1} = -T_{2}{}^{2}$ and thus $DT_{1;1}{}^{1} = -DT_{2;1}{}^{2}$ and $DT_{1;2}{}^{1} = -DT_{2;2}{}^{2}$. A straightforward calculation

now shows that

$$(\nabla_{\partial_{x^{1}}} R)(\partial_{x_{2'}}, \partial_{x^{1}}, \partial_{x^{1}}, \partial_{x^{1}}) = \frac{1}{2} D T_{1;1}^{2}, (\nabla_{\partial_{x^{1}}} R)(\partial_{x_{2'}}, \partial_{x^{2}}, \partial_{x^{2}}, \partial_{x^{1}}) = \frac{1}{2} D T_{2;1}^{1}, (\nabla_{\partial_{x^{2}}} R)(\partial_{x_{2'}}, \partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{2}}) = \frac{1}{2} D T_{1;2}^{2}, (\nabla_{\partial_{x_{1'}}} R)(\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{2}}, \partial_{x^{1}}) = -\frac{1}{2} D T_{2;2}^{1}, (\nabla_{\partial_{x_{1'}}} R)(\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{1}}) = D T_{2;2}^{2} + \frac{1}{2} D T_{2;1}^{1}, (\nabla_{\partial_{x_{2'}}} R)(\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{2}}, \partial_{x^{2}}) = D T_{2;1}^{2} - \frac{1}{2} D T_{1;2}^{2},$$

from where it follows that the trace-free (1, 1)-tensor field T is D-parallel.

The existence of a parallel (1, 1)-tensor field on an affine surface (Σ, D) was considered in [40] showing that (besides the trivial case where T = 0) a parallel trace-free (1, 1)-tensor field corresponds to one of the following:

- (a) An affine para-Kähler structure (det $T = -k^2 < 0$), which in suitable adapted coordinates becomes $T = k(\partial_{x^1} \otimes dx^1 \partial_{x^2} \otimes dx^2)$.
- (b) An affine nilpotent Kähler structure ($T^2 = 0$), which in suitable adapted coordinates becomes $T = k \partial_{x^1} \otimes dx^2$.
- (c) An affine Kähler structure (det $T = k^2 > 0$), which in suitable adapted coordinates becomes $T = k(\partial_{x^2} \otimes dx^1 \partial_{x^1} \otimes dx^2)$.

Each of the three possibilities above gives rise to different geometric structures which are locally conformally flat. We study each case separately in what follows.

Locally symmetric self-dual Walker surfaces given by an affine para-Kähler structure

We will see that this case leads to product manifolds.

Lemma 2.7. Let $(T^*\Sigma, g)$ be a locally symmetric self-dual Walker manifold determined by an affine para-Kähler structure T on (Σ, D) . Then $(T^*\Sigma, g)$ is locally conformally flat and locally isometric to a product of two Lorentzian surfaces of constant opposite Gaussian curvature.

Proof. Let (Σ, D) be an affine surface and choose local coordinates (x^1, x^2) so that the parallel tensor field T is locally given by $T = k(\partial_{x^1} \otimes dx^1 - \partial_{x^2} \otimes dx^2)$. Then T is parallel if and only if the Christoffel symbols are such that

$${}^{D}\Gamma_{11}{}^{2} = {}^{D}\Gamma_{12}{}^{1} = {}^{D}\Gamma_{12}{}^{2} = {}^{D}\Gamma_{22}{}^{1} = 0.$$

We refer to [40] for more information on the matter. Let $(x^1, x^2, x_{1'}, x_{2'})$ be the induced coordinates on $T^*\Sigma$. Then the symmetric and skew-symmetric Ricci tensors of (Σ, D) are given by

$${}^{D}\rho_{s} = -(\partial_{x^{1}}{}^{D}\Gamma_{22}{}^{2} + \partial_{x^{2}}{}^{D}\Gamma_{11}{}^{1})dx^{1} \circ dx^{2} ,$$
$${}^{D}\rho_{sk} = \frac{1}{2}(\partial_{x^{1}}{}^{D}\Gamma_{22}{}^{2} - \partial_{x^{2}}{}^{D}\Gamma_{11}{}^{1})dx^{2} \wedge dx^{1}$$

A straightforward calculation now shows that the Ricci operator of $(T^*\Sigma, g)$, when expressed on the coordinate basis, satisfies

$$\operatorname{Ric} = \begin{pmatrix} k & 0 & 0 & 0 \\ 0 & -k & 0 & 0 \\ 0 & k\Phi_{12} + 2^D \rho_s(\partial_{x^1}, \partial_{x^2}) & k & 0 \\ -k\Phi_{12} + 2^D \rho_s(\partial_{x^1}, \partial_{x^2}) & 0 & 0 & -k \end{pmatrix},$$

from where it follows that the Ricci curvatures are $\pm k$. Moreover, a straightforward calculation shows that $(\operatorname{Ric} - k \operatorname{Id})(\operatorname{Ric} + k \operatorname{Id}) = 0$, so the Ricci operator is diagonalizable with respect to an orthonormal basis. If it is parallel, then the manifold is locally isometric to a product of two Lorentzian surfaces of constant opposite Gaussian curvature, thus being locally conformally flat.

Remark 2.8. Let (Σ, D, T) be an affine para-Kähler surface. Let (x^1, x^2) be local coordinates on Σ so that the tensor field T expresses locally as in (a) above. Then there exists a locally defined deformation tensor Φ so that $g = \iota T \circ \iota \operatorname{Id} + g_D + \pi^* \Phi$ is locally symmetric. A straightforward calculation shows that

$$(\nabla_{\partial_{x^2}} R)(\partial_{x^2}, \partial_{x^1}, \partial_{x^1}, \partial_{x_{1'}}) = -k^2 \{ k \Phi_{12} + 2^D \rho_{sk}(\partial_{x^1}, \partial_{x^2}) \} x_{1'}^2 x_{2'} + \text{other terms.}$$

Setting $\Phi_{12} = -\frac{2}{k} D \rho_{sk}(\partial_{x^1}, \partial_{x^2})$, one has

$$\begin{split} (\nabla_{\partial_{x^1}} R)(\partial_{x^2}, \partial_{x^1}, \partial_{x^1}, \partial_{x_{1'}}) &= \frac{k}{2} \left\{ k \partial_{x^2} \Phi_{11} - 2(D_{\partial_{x^1}}{}^D \rho)(\partial_{x^2}, \partial_{x^1}) \right\} x_{1'}^2 \\ &\quad + \text{ other terms }, \\ (\nabla_{\partial_{x^2}} R)(\partial_{x^2}, \partial_{x^1}, \partial_{x^1}, \partial_{x_{1'}}) &= -\frac{k}{2} \left\{ k \partial_{x^1} \Phi_{22} + 2(D_{\partial_{x^2}}{}^D \rho)(\partial_{x^1}, \partial_{x^2}) \right\} x_{1'}^2 \\ &\quad + \text{ other terms,} \end{split}$$

which determines Φ_{11} and Φ_{22} . A straightforward calculation now shows that the covariant derivative ∇R vanishes when choosing the deformation tensor Φ as above.

Locally symmetric self-dual Walker surfaces given by an affine nilpotent Kähler structure

The case given by an affine nilpotent Kähler structure (Σ, D, T) , where det T = 0 and $T \neq 0$, gives rise to a single curvature model.

Lemma 2.9. Let $(T^*\Sigma, g)$ be a locally symmetric self-dual Walker manifold determined by an affine nilpotent Kähler structure T on (Σ, D) . Then $(T^*\Sigma, g)$ is locally conformally flat modelled on (\mathfrak{M}^-) , and locally isometric to a modified Riemannian extension $g = \iota T \circ \iota \operatorname{Id} + g_D$ where (Σ, D) is a flat affine surface.

Proof. Let (x^1, x^2) be local coordinates on Σ in which the tensor field T is locally expressed as $T = k \partial_{x^1} \otimes dx^2$. Then it follows from [40] that the Christoffel symbols satisfy

$${}^{D}\Gamma_{11}{}^{1} = {}^{D}\Gamma_{11}{}^{2} = {}^{D}\Gamma_{12}{}^{2} = 0$$
 and ${}^{D}\Gamma_{12}{}^{1} = {}^{D}\Gamma_{22}{}^{2}$.

The symmetric and skew-symmetric components of the Ricci tensor of (Σ, D) are given by

$${}^{D}\rho_{s} = (\partial_{x^{1}}{}^{D}\Gamma_{22}{}^{1} - \partial_{x^{2}}{}^{D}\Gamma_{22}{}^{2})dx^{2} \otimes dx^{2}, \qquad {}^{D}\rho_{sk} = \partial_{x^{1}}{}^{D}\Gamma_{22}{}^{2}dx^{2} \wedge dx^{1}$$

Let $(x^1, x^2, x_{1'}, x_{2'})$ be the induced local coordinates on $T^*\Sigma$. A straightforward calculation shows that the component of the Weyl tensor

$$W(\partial_{x^2}, \partial_{x^1}, \partial_{x^1}, \partial_{x_{1'}}) = \frac{1}{k} \partial_{x_{1'}} (\nabla_{\partial_{x^1}} R) (\partial_{x^1}, \partial_{x^2}, \partial_{x^2}, \partial_{x^1})$$
$$= \frac{k}{4} \Phi_{11} - {}^D \rho_{sk} (\partial_{x^1}, \partial_{x^2}).$$

Assuming that $(T^*\Sigma, g)$ is locally symmetric, one has $\Phi_{11} = \frac{4}{k} \rho_{sk}(\partial_{x^1}, \partial_{x^2})$. Now, given the expression of Φ_{11} one gets that the only non-zero component of the Weyl tensor (up to the usual symmetries) is

$$W(\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{2}}) = \frac{1}{4k} x_{1'} \partial_{x_{1'}} (\nabla_{\partial_{x^{2}}} R) (\partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{2}}, \partial_{x_{1'}}) - \frac{1}{k^{2}} \partial_{x_{1'}} \partial_{x_{1'}} \partial_{x^{1}} (\nabla_{\partial_{x^{2}}} R) (\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{2}}, \partial_{x_{2'}}) + \frac{1}{4k^{2}} \partial_{x_{1'}} \partial_{x_{1'}} \partial_{x^{2}} (\nabla_{\partial_{x^{2}}} R) (\partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{2}}, \partial_{x_{1'}}) - \frac{1}{2k^{2}} {}^{D} \Gamma_{22} {}^{2} \partial_{x_{1'}} \partial_{x_{1'}} (\nabla_{\partial_{x^{2}}} R) (\partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{2}}, \partial_{x_{1'}}) .$$

This shows that if $(T^*\Sigma, g)$ is locally symmetric, then it is locally conformally flat. Moreover, since $\Phi_{11} = \frac{4}{k} \rho_{sk}(\partial_{x^1}, \partial_{x^2})$ it is easy to see that

$$(\nabla_{\partial_{x^1}} R)(\partial_{x^2}, \partial_{x^1}, \partial_{x^2}, \partial_{x^1}) = \frac{k}{2} \partial_{x^1} \Phi_{12} - (D_{\partial_{x^2}} {}^D \rho)(\partial_{x^1}, \partial_{x^2}) + \partial_{x^1} {}^D \rho_s(\partial_{x^2}, \partial_{x^2}) - \partial_{x^2} {}^D \rho_{sk}(\partial_{x^1}, \partial_{x^2}).$$

Therefore, $\Phi_{12} = \frac{2}{k} \left\{ \left({}^D \Gamma_{22} {}^2 \right)^2 - \partial_{x^2} {}^D \Gamma_{22} {}^2 - \partial_{x^1} {}^D \Gamma_{22} {}^1 \right\} + \phi_{12}(x^2)$ for some function $\phi_{12}(x^2)$. Using the expressions obtained for Φ_{11} and Φ_{12} one has

$$(\nabla_{\partial_{x^2}} R)(\partial_{x^2}, \partial_{x^1}, \partial_{x^2}, \partial_{x^1}) = \partial_{x^1} \left(\frac{k}{2} \Phi_{22} + 2\partial_{x^2} D \Gamma_{22}^{-1} - 2^D \Gamma_{22}^{-2D} \Gamma_{22}^{-1} \right) - \frac{k}{2} \phi_{12}'(x^2),$$

which gives $\Phi_{22} = \frac{4}{k} ({}^D\Gamma_{22}{}^{2D}\Gamma_{22}{}^1 - \partial_{x^2}{}^D\Gamma_{22}{}^1) + x^1 \phi'_{12}(x^2) + \phi_{22}(x^2)$ for some function $\phi_{22}(x^2)$.

A straightforward calculation shows that the metric $g = \iota T \circ \iota \operatorname{Id} + g_D + \pi^* \Phi$, where the tensor Φ is given by the expressions above, is locally symmetric (thus being locally conformally flat). Besides, the Ricci operator takes the form

$$\operatorname{Ric} = \begin{pmatrix} 0 & k & 0 & 0 \\ 0 & 0 & 0 & 0 \\ 0 & 2^{D} \rho_{sk}(\partial_{x^{2}}, \partial_{x^{1}}) & 0 & 0 \\ 2^{D} \rho_{sk}(\partial_{x^{1}}, \partial_{x^{2}}) & 2^{D} \rho_{s}(\partial_{x^{2}}, \partial_{x^{2}}) & k & 0 \end{pmatrix}$$

Hence, the Ricci operator is two-step nilpotent and a straightforward calculation shows that its Jordan normal form corresponds to that discussed in Section 2.1.1-(1.b) with $\varepsilon_1 \varepsilon_2 = -1$. Therefore, the curvature tensor is determined by the 0-model (\mathfrak{M}^-) .
Now let (Σ, D) be a flat affine surface and let T be an affine nilpotent Kähler structure. Setting $\Phi = 0$, a direct calculation using the expressions above shows that the Riemannian extension $g = \iota T \circ \iota \operatorname{Id} + g_D$ is locally symmetric, thus completing the proof.

Remark 2.10. Let (Σ, D, T) be an affine nilpotent Kähler surface. Let (x^1, x^2) be local coordinates on Σ so that the tensor field T expresses locally as $T = k \partial_{x^1} \otimes dx^2$. Then there exists a locally defined tensor field Φ given as in the proof of Lemma 2.9 so that the cotangent bundle $(T^*\Sigma, g = \iota T \circ \iota \operatorname{Id} + g_D + \pi^*\Phi)$ is locally symmetric.

Locally symmetric self-dual Walker surfaces given by an affine Kähler structure

We will see that this situation corresponds to the 0-model (\mathfrak{N}_k) .

Lemma 2.11. Let $(T^*\Sigma, g)$ be a locally symmetric self-dual Walker manifold determined by an affine Kähler structure T on (Σ, D) . Then $(T^*\Sigma, g)$ is locally conformally flat modelled on (\mathfrak{N}_k) and locally isometric to the Riemannian extension $g = \iota T \circ \iota \operatorname{Id} + g_D$, where (Σ, D) is a flat affine surface.

Proof. Let (x^1, x^2) be local coordinates on Σ so that the tensor field T locally takes the form $T = k \ (\partial_{x^2} \otimes dx^1 - \partial_{x^1} \otimes dx^2)$. Then it follows from the work in [40] that the Christoffel symbols are such that

$${}^{D}\Gamma_{11}{}^{1} = {}^{D}\Gamma_{12}{}^{2} = -{}^{D}\Gamma_{22}{}^{1}$$
 and ${}^{D}\Gamma_{12}{}^{1} = -{}^{D}\Gamma_{11}{}^{2} = {}^{D}\Gamma_{22}{}^{2}$.

Besides, the symmetric and skew-symmetric parts of the Ricci tensor are given by

$${}^{D}\rho_{s} = (\partial_{x^{1}}{}^{D}\Gamma_{22}{}^{1} - \partial_{x^{2}}{}^{D}\Gamma_{22}{}^{2})(dx^{1} \otimes dx^{1} + dx^{2} \otimes dx^{2}),$$

$${}^{D}\rho_{sk} = (\partial_{x^{2}}{}^{D}\Gamma_{22}{}^{1} + \partial_{x^{1}}{}^{D}\Gamma_{22}{}^{2})(dx^{2} \otimes dx^{1} - dx^{1} \otimes dx^{2}).$$

Now, since

$$(\nabla_{\partial_{x^1}} R)(\partial_{x^2}, \partial_{x^1}, \partial_{x^1}, \partial_{x_{1'}}) = \frac{k^2}{8} \left\{ k(\Phi_{11} + \Phi_{22}) - 4^D \rho_{sk}(\partial_{x^2}, \partial_{x^1}) \right\} x_{1'}^3 + \text{other terms}$$

assuming that $(T^*\Sigma, g)$ is locally symmetric we set $\Phi_{11} = -\Phi_{22} + \frac{4}{k} \rho_{sk}(\partial_{x^2}, \partial_{x^1})$. A long but straightforward calculation now shows that the only non-zero component (up to the usual symmetries) of the Weyl tensor is

$$W(\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{2}}) = \frac{1}{k} x_{1'} \partial_{x_{1'}} \partial_{x_{2'}} (\nabla_{\partial_{x^{1}}} R) (\partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{2}}, \partial_{x_{1'}}) + \frac{1}{k} x_{2'} \partial_{x_{2'}} \partial_{x_{2'}} (\nabla_{\partial_{x^{2}}} R) (\partial_{x^{2}}, \partial_{x^{1}}, \partial_{x^{1}}, \partial_{x_{1'}}) - \frac{1}{3k^{2}} \partial_{x^{2}} \partial_{x_{1'}} \partial_{x_{2'}} (\nabla_{\partial_{x^{1}}} R) (\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{1}}, \partial_{x_{2'}}) + \frac{2}{3k^{2}} D \Gamma_{22}^{2} \partial_{x_{1'}} \partial_{x_{2'}} (\nabla_{\partial_{x^{1}}} R) (\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{1}}, \partial_{x_{2'}}) - \frac{1}{k^{2}} \partial_{x^{1}} \partial_{x_{1'}} \partial_{x_{2'}} (\nabla_{\partial_{x^{2}}} R) (\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{1}}, \partial_{x_{2'}}) - \frac{2}{k^{2}} D \Gamma_{22}^{1} \partial_{x_{1'}} \partial_{x_{2'}} (\nabla_{\partial_{x^{2}}} R) (\partial_{x^{1}}, \partial_{x^{2}}, \partial_{x^{1}}, \partial_{x_{2'}}) .$$

Therefore, the local symmetry of $(T^*\Sigma, g)$ implies local conformal flatness. Moreover, the Ricci operator of $(T^*\Sigma, g)$ is given by

$$\operatorname{Ric} = \begin{pmatrix} 0 & -k & 0 & 0 \\ k & 0 & 0 & 0 \\ 2^{D} \rho_{s}(\partial_{x^{1}}, \partial_{x^{1}}) & 2^{D} \rho_{sk}(\partial_{x^{2}}, \partial_{x^{1}}) & 0 & k \\ 2^{D} \rho_{sk}(\partial_{x^{1}}, \partial_{x^{2}}) & 2^{D} \rho_{s}(\partial_{x^{1}}, \partial_{x^{1}}) & -k & 0 \end{pmatrix}$$

Hence, the Ricci curvatures are $\pm ik$ and $\operatorname{Ric}^2 = -k^2 \operatorname{Id}$, which shows that the Ricci operator is complex diagonalizable. Since the curvature tensor is completely determined by the Ricci operator, it corresponds to the 0-model (\mathfrak{N}_k) .

To conclude, let (Σ, D) be a flat affine surface and T be an affine Kähler structure. A direct calculation shows that the Riemannian extension $g = \iota T \circ \iota \operatorname{Id} + g_D$ is locally symmetric, thus completing the proof.

Remark 2.12. Let (Σ, D, T) be an affine Kähler surface. Then there exists a suitable locally defined deformation tensor Φ so that $(T^*\Sigma, g = \iota T \circ \iota \operatorname{Id} + g_D + \pi^* \Phi)$ is locally symmetric. A straightforward calculation as in the proof of Lemma 2.11 gives that

$$\Phi_{11} = -\Phi_{22} + \frac{4}{k} {}^D \rho_{sk}(\partial_{x^2}, \partial_{x^1})$$

and the coefficients Φ_{12} and Φ_{22} satisfy

$$\partial_{x^{1}}\Phi_{22} - \partial_{x^{2}}\Phi_{12} = \frac{2}{k} \left\{ (D\partial_{x^{1}}{}^{D}\rho_{sk})(\partial_{x^{1}},\partial_{x^{2}}) + (D\partial_{x^{2}}{}^{D}\rho_{s})(\partial_{x^{1}},\partial_{x^{1}}) \right. \\ \left. + \partial_{x^{1}}{}^{D}\rho_{sk}(\partial_{x^{2}},\partial_{x^{1}}) \right\}, \\ \partial_{x^{2}}\Phi_{22} + \partial_{x^{1}}\Phi_{12} = \frac{2}{k} \left\{ (D\partial_{x^{2}}{}^{D}\rho_{sk})(\partial_{x^{2}},\partial_{x^{1}}) + (D\partial_{x^{1}}{}^{D}\rho_{s})(\partial_{x^{1}},\partial_{x^{1}}) \right. \\ \left. + \partial_{x^{2}}{}^{D}\rho_{sk}(\partial_{x^{2}},\partial_{x^{1}}) \right\}.$$

The system of equations above is just the inhomogeneous Cauchy-Riemann equation $\partial_{\overline{z}}\phi = f$ for $\phi = \Phi_{22} + i\Phi_{12}$, where the function $f = f_1 + if_2$ is given by the right hand side of the expressions above, which admit local solutions for any affine Kähler surface (Σ, D) (see [106]).

Remark 2.13. Let (M, g) be a locally symmetric Walker metric as in Lemma 2.11. Then the Ricci operator defines a complex structure $S = \frac{1}{k}$ Ric which is self-adjoint and parallel (hence, a Riemannian complex structure). Furthermore, the twin metric

$$\hat{g}(X,Y) = g(\frac{1}{k}\operatorname{Ric} X,Y) = \frac{1}{k}\rho(X,Y)$$

is locally symmetric and Einstein with scalar curvature $\hat{\tau} = 4k$, which is a special case of the main Theorem in [22].

Locally symmetric self-dual Walker surfaces with T = 0

In contrast with the previous cases, the class of locally symmetric Riemannian extensions of the form $g = g_D + \pi^* \Phi$ is much larger and the underlying structure is not necessarily locally conformally flat.

Example 2.14. Let (Σ, D) be the flat plane and let Φ be the symmetric (0, 2)-tensor field

$$\Phi = x^1 x^2 (dx^1 \otimes dx^2 + dx^2 \otimes dx^1).$$

Then $(T^*\Sigma, g_D + \pi^*\Phi)$ is a Ricci-flat locally symmetric manifold which is not locally conformally flat. The curvature tensor is determined by the only non-zero component $R(\partial_{x^1}, \partial_{x^2}, \partial_{x^1}, \partial_{x^2}) = 1$ (up to the usual symmetries).

It follows from the work in [1] that if $(T^*\Sigma, g = g_D + \pi^*\Phi)$ is locally symmetric, then so is (Σ, D) . Locally symmetric affine surfaces were described by Opozda in [113], where it is stated that the affine connection corresponds to one of the following:

- (i) The flat affine surface modelled on \mathbb{R}^2 with ${}^D\Gamma_{ij}{}^k = 0$.
- (ii) The Levi-Civita connection of the hyperbolic plane $\mathbb{H}^2 = \mathbb{R}^+ \times \mathbb{R}$ given by

$${}^{D}\Gamma_{11}{}^{1} = {}^{D}\Gamma_{12}{}^{2} = -{}^{D}\Gamma_{22}{}^{1} = -\frac{1}{x^{1}}.$$

(iii) The Levi-Civita connection of the Lorentzian hyperbolic plane

$${}^{D}\Gamma_{11}{}^{1} = {}^{D}\Gamma_{12}{}^{2} = {}^{D}\Gamma_{22}{}^{1} = -\frac{1}{x^{1}}.$$

- (iv) The Levi-Civita connection of the standard sphere.
- (v) One of the two non-metrizable affine connections modelled on \mathbb{R}^2 whose non-zero Christoffel symbols are ${}^D\Gamma_{11}{}^1 = 1$ and ${}^D\Gamma_{22}{}^1 = \pm 1$.

The Ricci operator of $(T^*\Sigma, g = g_D + \pi^*\Phi)$ vanishes in Case (i) and it is two-step nilpotent otherwise. Besides, Ric has rank two in Cases (ii), (iii) and (iv), while it has rank one in Case (v).

In order to describe the locally symmetric Riemannian extensions which are locally conformally flat, we introduce the following algebraic curvature model:

 $(\mathfrak{P}): (V, \langle \cdot, \cdot \rangle, \mathcal{A})$ given by

$$\mathcal{A}_{1313} = \mathcal{A}_{1441} = \frac{1}{2}$$

with respect to a basis $\{u_1, u_2, u_3, u_4\}$ where the non-zero inner products are

$$\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = 1 = -\langle u_4, u_4 \rangle.$$

If $(T^*\Sigma, g = g_D + \pi^*\Phi)$ is locally conformally flat, then there exist coordinates on (Σ, D) so that one may assume $\Phi = 0$. This situation is summarized as follows:

Lemma 2.15. Let $(T^*\Sigma, g = g_D + \pi^*\Phi)$ be a locally symmetric and locally conformally flat Riemannian extension. Then it is either flat or locally isometric to a Riemannian extension $(T^*\Sigma, g_D)$ where (Σ, D) corresponds to one of the following affine connections.

- (i) The Levi-Civita connection of a Riemannian surface of constant curvature, in which case the curvature tensor of $(T^*\Sigma, g_D)$ is modelled on (\mathfrak{M}^+) .
- (ii) The Levi-Civita connection of a Lorentzian surface of constant curvature, in which case the curvature tensor of $(T^*\Sigma, g_D)$ is modelled on (\mathfrak{M}^-) .
- (iii) One of the two non-metrizable affine connections modelled on \mathbb{R}^2 whose Christoffel symbols are ${}^D\Gamma_{11}{}^1 = 1$ and ${}^D\Gamma_{22}{}^1 = \pm 1$. In this case, the curvature tensor of $(T^*\Sigma, g_D)$ is modelled on (\mathfrak{P}) .

2.1.4 Locally conformally flat para-Kähler surfaces

Let (M, g, J_{-}) be a para-Kähler surface. Since $J_{-}^{2} = \text{Id}$ and $J_{-}^{*}g = -g$, then $\mathcal{D}_{\pm} = \text{ker}(J_{-} \mp \text{Id})$ are totally degenerate. Moreover, the fact that $\nabla J_{-} = 0$ implies that \mathcal{D}_{\pm} are parallel, so (M, g, J_{-}) is a Walker manifold with respect to both distributions \mathcal{D}_{\pm} . We set the parallel distribution $\mathcal{D} = \mathcal{D}_{+}$ so that $J_{-}|_{\mathcal{D}} = \text{Id}$ and

$$g = dx^{i} \otimes dx_{i'} + dx_{i'} \otimes dx^{i} + g_{ij}(x^{1}, x^{2}, x_{1'}, x_{2'}) dx^{i} \otimes dx^{j}, \qquad (2.3)$$

with respect to the Walker coordinates $(x^1, x^2, x_{1'}, x_{2'})$. The locally defined almost para-Hermitian structures satisfying $J_{-}|_{\mathcal{D}} = \text{Id}$ are parametrized by a real-valued function $f(x^1, x^2, x_{1'}, x_{2'})$ and are given by

$$J_{-}^{f}\partial_{x^{1}} = -\partial_{x^{1}} + g_{11}\partial_{x_{1'}} + f\partial_{x_{2'}}, \qquad J_{-}^{f}\partial_{x_{1'}} = \partial_{x_{1'}}, J_{-}^{f}\partial_{x^{2}} = -\partial_{x^{2}} + (2g_{12} - f)\partial_{x_{1'}} + g_{22}\partial_{x_{2'}}, \quad J_{-}^{f}\partial_{x_{2'}} = \partial_{x_{2'}}.$$
(2.4)

Let (g, J_{-}^{f}) be an almost para-Hermitian structure determined by Equations (2.3)–(2.4). Then the para-Kähler form $\Omega_{f}(X, Y) = g(J_{-}^{f}X, Y)$ is given by

$$\Omega_f = (f - g_{12})dx^1 \wedge dx^2 + dx_{1'} \wedge dx^1 + dx_{2'} \wedge dx^2$$

and consequently

$$d\Omega_f = \partial_{x_{1'}} \left(f - g_{12} \right) dx_{1'} \wedge dx^1 \wedge dx^2 + \partial_{x_{2'}} \left(f - g_{12} \right) dx_{2'} \wedge dx^1 \wedge dx^2.$$

Therefore, $d\Omega_f = 0$ if and only if

$$f(x^1, x^2, x_{1'}, x_{2'}) = g_{12}(x^1, x^2, x_{1'}, x_{2'}) + h(x^1, x^2)$$

for some function $h(x^1, x^2)$ and the almost para-complex structure becomes

$$J_{-}^{h}\partial_{x^{1}} = -\partial_{x^{1}} + g_{11}\partial_{x_{1'}} + (g_{12} + h)\partial_{x_{2'}}, \quad J_{-}^{h}\partial_{x_{1'}} = \partial_{x_{1'}}, J_{-}^{h}\partial_{x^{2}} = -\partial_{x^{2}} + (g_{12} - h)\partial_{x_{1'}} + g_{22}\partial_{x_{2'}}, \quad J_{-}^{h}\partial_{x_{2'}} = \partial_{x_{2'}}.$$
(2.5)

Let (g, J^h_{-}) be an almost para-Hermitian structure determined by (2.3) and (2.5). The para-Kähler two-form is given by

$$\Omega_h = h \, dx^1 \wedge dx^2 + dx_{1'} \wedge dx^1 + dx_{2'} \wedge dx^2.$$

Notice that the para-Kähler orientation and the Walker orientation are opposite. Indeed, the para-Kähler two-form Ω_h is anti-self-dual for the para-Kähler orientation determined by the para-complex structure J_{-}^h , but it is self-dual for the Walker orientation.

Meaning to describe all the anti-self-dual para-Kähler surfaces of constant scalar curvature we consider the cotangent bundle $T^*\Sigma$ of an affine surface (Σ, D) with the metric $g = \iota T \circ \iota \operatorname{Id} + g_D + \pi^*\Phi$ as discussed in Section 2.1.3 and set the para-complex structure satisfying the condition $J_-|_{\ker\pi_*} = \operatorname{Id}$. The almost para-Hermitian structures (g, J_-^h) defined by (2.3) and (2.5) are not para-Kähler in general. In order to express the components of ∇J_-^h on $T^*\Sigma$ we use the notation $(\nabla_{\partial_{x^{\alpha}}} J_-^h)\partial_{x^{\beta}} = (\nabla J_-^h)_{\beta;\alpha}{}^{\gamma}\partial_{x^{\gamma}}$ and $(D\partial_{x^i}\Phi)(\partial_{x^j}, \partial_{x^k}) = D\Phi_{jk;i}$ to represent the covariant derivative of the symmetric (0, 2)-tensor field Φ on Σ . In this notation, the components of the covariant derivative of the para-complex structures J_-^h are given by the following result.

Lemma 2.16. Let $(T^*\Sigma, g)$ be a locally symmetric self-dual Walker structure on the cotangent bundle of an affine surface (Σ, D) . Let J^h_- be a (locally defined) almost para-complex structure determined by $J^h_-|_{\ker \pi_*} = \operatorname{Id}$ so that (g, J^h_-) is an almost para-Hermitian structure locally given by (2.5). Then the non-zero components of ∇J^h_- are determined by

$$\begin{split} 2\left(\nabla J_{-}^{h}\right)_{1;1}{}^{2'} &= \frac{1}{4}x_{1'}^{2}x_{2'}\left\{(T_{2}{}^{2})^{2} - (T_{1}{}^{1})^{2}\right\} - \frac{1}{2}x_{2'}\left\{8^{D}\rho(\partial_{x^{1}},\partial_{x^{1}}) - 4hT_{1}{}^{2}\right\} \\ &\quad + \frac{1}{2}x_{1'}\left\{8^{D}\rho(\partial_{x^{2}},\partial_{x^{1}}) + 5hT_{1}{}^{1} + hT_{2}{}^{2}\right. \\ &\quad + 2\left(T_{2}{}^{1}\Phi_{11} - T_{1}{}^{2}\Phi_{22} + (T_{2}{}^{2} - T_{1}{}^{1})\Phi_{12})\right\} \\ &\quad + 2\left\{\partial_{1}h - h\left({}^{D}\Gamma_{11}{}^{1} + {}^{D}\Gamma_{12}{}^{2}\right) + D\Phi_{11;2} - D\Phi_{12;1}\right\}, \\ 2\left(\nabla J_{-}^{h}\right)_{1;2}{}^{2'} &= \frac{1}{4}x_{1'}x_{2'}^{2}\left\{(T_{2}{}^{2})^{2} - (T_{1}{}^{1})^{2}\right\} + \frac{1}{2}x_{1'}\left\{8^{D}\rho(\partial_{x^{2}},\partial_{x^{2}}) + 4hT_{2}{}^{1}\right\} \\ &\quad - \frac{1}{2}x_{2'}\left\{8^{D}\rho(\partial_{x^{1}},\partial_{x^{2}}) - hT_{1}{}^{1} - 5hT_{2}{}^{2}\right. \\ &\quad - 2\left(T_{2}{}^{1}\Phi_{11} - T_{1}{}^{2}\Phi_{22} + (T_{2}{}^{2} - T_{1}{}^{1})\Phi_{12}\right)\right\} \\ &\quad + 2\left\{\partial_{2}h - h\left({}^{D}\Gamma_{22}{}^{2} + {}^{D}\Gamma_{12}{}^{1}\right) + D\Phi_{12;2} - D\Phi_{22;1}\right\}, \end{split}$$

where T is a trace-free parallel (1, 1)-tensor field on (Σ, D) and Φ is a symmetric (0, 2)-tensor field on Σ .

Theorem 2.1 follows immediately from the following result describing the local structure of anti-self-dual para-Kähler surfaces with constant scalar curvature.

Theorem 2.17. Let (M, g, J_{-}) be an anti-self-dual para-Kähler surface with constant scalar curvature. Then it is locally isometric to a Riemannian extension of the form $(T^*\Sigma, \tilde{g} = \iota T \circ \iota \operatorname{Id} + g_D)$ with para-complex structure determined by $J_{-}|_{\ker \pi_*} = \operatorname{Id}$, where T is a parallel (1, 1)-tensor field on a flat affine surface (Σ, D) satisfying one of the following conditions.

- (i) $T = c \operatorname{Id} and (M, g, J_{-})$ has constant para-holomorphic sectional curvature c.
- (*ii*) T = 0 and (M, g, J_{-}) is flat.
- (iii) $T^2 = k^2 \operatorname{Id} and (M, g, J_-)$ is locally isometric to a product of two Lorentzian surfaces of constant opposite curvature.
- (iv) $T^2 = 0$ and (M, g, J_-) is modelled on (\mathfrak{M}^-) .
- (v) $T^2 = -k^2 \operatorname{Id} and (M, g, J_-)$ is modelled on (\mathfrak{N}_k) .

In all the cases above the para-Kähler two-form is the canonical symplectic two-form of $T^*\Sigma$.

Proof. Let (M, g, J_{-}) be an anti-self-dual para-Kähler surface. Then there exists a Walker structure (M, g, \mathcal{D}) so that (M, g) is self-dual with respect to the Walker orientation and (M, g, J_{-}) is locally isometric to the cotangent bundle of an affine surface (Σ, D) with para-complex structure determined by $J_{-}|_{\ker \pi_{*}} = \operatorname{Id}$ and metric tensor $g = \iota X(\iota \operatorname{Id} \circ \iota \operatorname{Id}) + \iota T \circ \iota \operatorname{Id} + g_{D} + \pi^{*} \Phi$.

A para-Kähler surface is anti-self-dual if and only if its Bochner tensor vanishes (see [27]), and it is locally symmetric if and only if the scalar curvature is constant. Assertion (*i*) corresponds to the case when its scalar curvature is non-zero and Lemma 2.6 shows that the (1, 1)-tensor field T is parallel in this case.

Anti-self-dual para-Kähler surfaces whose scalar curvature is zero are locally conformally flat and locally symmetric. Therefore, the underlying structure is induced by an affine para-Kähler structure, an affine nilpotent Kähler structure, or an affine Kähler structure as discussed in Lemma 2.7, Lemma 2.9, and Lemma 2.11, respectively.

Let (Σ, D, T) be an affine surface equipped with a parallel trace-free (1, 1)-tensor field T. It follows from Lemma 2.16 that, if the almost para-complex structure J_{-}^{h} determined by $J_{-}^{h}|_{\ker \pi_{*}} = \mathrm{Id}$ is parallel, then it is uniquely determined. If (Σ, D, T) is an affine para-Kähler surface, then the coefficients of $x_{1'}$ and $x_{2'}$ in Lemma 2.16 show that $h = -\frac{2}{k}^{D} \rho_{s}(\partial_{x^{1}}, \partial_{x^{2}})$ for a deformation tensor field Φ given as in Remark 2.8. If (Σ, D, T) is an affine nilpotent Kähler surface, then Lemma 2.16 shows that $h = -\frac{2}{k}^{D} \rho(\partial_{x^{2}}, \partial_{x^{2}})$ and if (Σ, D, T) is an affine Kähler surface, then $h = \frac{2}{k}^{D} \rho(\partial_{x^{2}}, \partial_{x^{2}})$.

Moreover, a straightforward calculation shows that for any (Σ, D, T) there is an appropriate deformation tensor field Φ so that $(T^*\Sigma, \iota T \circ \iota \operatorname{Id} + g_D + \pi^*\Phi, J^h)$ is para-Kähler, where Φ is given as in Remark 2.8, Remark 2.10 and Remark 2.12. In all these cases, the cotangent bundle $(T^*\Sigma, \iota T \circ \iota \operatorname{Id} + g_D + \pi^*\Phi, J^h_-)$ is locally isometric to the Riemannian extension $\iota T \circ \iota \operatorname{Id} + g_D$ of a flat affine surface (Σ, D, T) that is affine para-Kähler, affine nilpotent Kähler or affine Kähler, and the two-form of the corresponding locally conformally flat para-Kähler manifold is the canonical symplectic form of $T^*\Sigma$, from where Assertions (*iii*), (*iv*) and (*v*) follow.

Finally, we consider the case T = 0 corresponding to Assertion (ii). Setting T = 0 in

Lemma 2.16, one has that the non-zero components of ∇J^h_{-} are given by

$$\left(\nabla J_{-}^{h}\right)_{1;1}{}^{2'} = 2x_{1'}\rho_{21}^{D} - 2x_{2'}\rho_{11}^{D} + \left\{\partial_{1}h - h\left({}^{D}\Gamma_{11}{}^{1} + {}^{D}\Gamma_{12}{}^{2}\right) \right. \\ \left. + D\Phi_{11;2} - D\Phi_{12;1}\right\},$$

$$\left(\nabla J_{-}^{h}\right)_{1;2}{}^{2'} = 2x_{1'}\rho_{22}^{D} - 2x_{2'}\rho_{12}^{D} + \left\{\partial_{2}h - h\left({}^{D}\Gamma_{22}{}^{2} + {}^{D}\Gamma_{12}{}^{1}\right) \right. \\ \left. + D\Phi_{12;2} - D\Phi_{22;1}\right\}.$$

It now follows from the coefficients of the terms of degree one above that if $\nabla J_{-}^{h} = 0$, the Ricci tensor ${}^{D}\rho$ vanishes and (Σ, D) is flat. Since the Ricci tensor of $g_{D} + \pi^{*}\Phi$ is determined by the symmetric part of ${}^{D}\rho$ one has that $(T^{*}\Sigma, \tilde{g} = g_{D} + \pi^{*}\Phi)$ is Ricci-flat. Therefore, if $(T^{*}\Sigma, \tilde{g} = g_{D} + \pi^{*}\Phi)$ is para-Kähler, then it must be Ricci-flat and thus flat – since it is locally conformally flat.

Observe that in all the cases above the para-complex structure J^h_- is uniquely determined because h = 0 if the base surface is flat and $T \neq 0$ (which follows in all cases from the expressions in Lemma 2.16). Moreover, the corresponding para-Kähler form is again the canonical symplectic two-form of the cotangent bundle.

Remark 2.18. The Ricci operator of any metric in Assertion (ν) of Theorem 2.17 satisfies $\operatorname{Ric}^2 = -k^2 \operatorname{Id}$ and, since the para-complex structure J_- commute with the Ricci operator, defining $J_+ = \frac{1}{k} \operatorname{Ric} J_-$ one has that (g, J_+) is a locally conformally flat indefinite Kähler structure.

2.2 Locally symmetric Kähler surfaces

Let (M, g, J_+) be a locally symmetric four-dimensional Kähler manifold. Then its Ricci operator is parallel. In the diagonalizable case the metric is Einstein or locally isometric to a product of two surfaces of constant curvature. The non-diagonalizability of the Ricci operator leads to a Walker structure and therefore to the situation in Section 2.1.3.

Lemma 2.19. A Kähler surface (M, g, J_+) with parallel and non-diagonalizable Ricci operator is a Walker manifold.

Proof. Since the Ricci operator Ric commutes with the complex structure J_+ , then the trace-less Ricci operator Ric⁰ = Ric $-\frac{\tau}{4}$ Id is either two-step nilpotent or complex diagonalizable.

If Ric^0 is two-step nilpotent, since ker Ric^0 is J_+ -invariant, then it must be two-dimensional and parallel thus determining a totally degenerate parallel distribution, which shows that (M, g) is a Walker manifold.

If Ric^0 is complex diagonalizable with eigenvalues $\pm ik$, then $\frac{1}{k}\operatorname{Ric}^0$ is a self-adjoint complex structure so that $(M, g, \frac{1}{k}\operatorname{Ric}^0)$ is complex Riemannian and the (1, 1)-tensor field $J_- = \frac{1}{k}\operatorname{Ric}^0 \cdot J_+$ determines a para-Kähler structure (g, J_-) so that (M, g, J_-) is a Walker manifold whose parallel distribution is J_{\pm} -invariant.

The proof of Theorem 2.2. Assertion (*ii*) in Theorem 2.2 follows immediately from Remark 2.18, which shows that any affine Kähler structure T satisfying $T^2 = -k^2 \operatorname{Id}$ on a flat affine surface (Σ, D) induces a Kähler structure (g, J_+) on $T^*\Sigma$ with

$$g = \iota T \circ \iota \operatorname{Id} + g_D$$

and $J_{+} = -\frac{1}{k} \operatorname{Ric} J_{-}$, where J_{-} is the para-Kähler structure determined by

$$J_{-}|_{\ker \pi_*} = \operatorname{Id}.$$

Moreover, let g_{Σ} be a flat Riemannian metric on Σ with Levi-Civita connection D. A straightforward calculation shows that the corresponding Kähler two-form is given by $\Omega_{+} = -d\iota J_{\Sigma}$, where J_{Σ} is the complex structure on (Σ, g_{Σ}) induced by the volume element of the flat metric g_{Σ} . Choosing local adapted coordinates (x^1, x^2) on Σ so that the metric tensor in given by $g_{\Sigma} = dx^1 \otimes dx^1 + dx^2 \otimes dx^2$, one has that the complex structure J_{+} on $T^*\Sigma$ is characterized by

$$J_+\partial_{x_{1'}} = \partial_{x_{2'}}, \qquad J_+\partial_{x_{2'}} = -\partial_{x_{1'}},$$

thus being a proper complex structure in the sense of [102] whose corresponding Kähler form is $\Omega_+ = dx^1 \wedge dx_{2'} - dx^2 \wedge dx_{1'}$.

We will construct a locally conformally flat Kähler surface showing the geometric realizability of the model (\mathfrak{M}^+) and thus proving Assertion (i) in Theorem 2.2. Let (Σ, g_{Σ}) be a Riemannian surface of non-zero constant Gaussian curvature and let D be its Levi-Civita connection. The Riemannian extension $(T^*\Sigma, g_D)$ is a locally symmetric four-dimensional manifold with curvature tensor modelled on (\mathfrak{M}^+) . Let ω_{Σ} be the Riemannian volume form of (Σ, g_{Σ}) and let J_{Σ} be the complex structure $g_{\Sigma}(J_{\Sigma}X, Y) = \omega_{\Sigma}(X, Y)$. Then $\Omega_+ = -d\iota J_{\Sigma}$ is a symplectic structure on $(T^*\Sigma, g = g_D)$ which induces a Kähler structure with complex structure $g_D(J_+\xi, \eta) = \Omega_+(\xi, \eta)$ for all vector fields ξ and η on $T^*\Sigma$.

Let (x^1, x^2) be a system of local coordinates on Σ so that the metric takes the form

$$g_{\Sigma} = \Psi(x^1, x^2)(dx^1 \otimes dx^1 + dx^2 \otimes dx^2)$$
 and $J_{\Sigma}\partial_{x^1} = \partial_{x^2}, J_{\Sigma}\partial_{x^2} = -\partial_{x^1}.$

Then the complex structure J_+ is determined, with respect to the induced system of local coordinates $(x^1, x^2, x_{1'}, x_{2'})$, by

$$J_+\partial_{x_{1'}} = \partial_{x_{2'}}, \qquad J_+\partial_{x_{2'}} = -\partial_{x_{1'}}$$

It corresponds to a proper Kähler structure whose corresponding Kähler form is given by

$$\Omega_+ = dx^1 \wedge dx_{2'} - dx^2 \wedge dx_{1'}$$

Remark 2.20. Let $\Sigma = \mathbb{H}_1^2$ be the Lorentzian hyperbolic plane and let $(T^*\Sigma, g_D)$ be the Riemannian extension of its Levi-Civita connection. Let J_{Σ} be the para-complex structure on (Σ, g_{Σ}) determined by the Lorentzian volume form and $\Omega_- = -d\iota J_{\Sigma}$. Then it determines a para-Kähler

structure (g_D, J_-) by $g_D(J_-\xi, \eta) = \Omega_-(\xi, \eta)$ which is locally conformally flat with curvature tensor modelled on (\mathfrak{M}^-) , and thus locally isometric to the one in Theorem 2.1-(*i*).

Furthermore, let (x^1, x^2) be a system of local coordinates on Σ so that the metric tensor $g_{\Sigma} = \frac{1}{(x^1)^2} (dx^1 \otimes dx^1 - dx^2 \otimes dx^2)$ and $J_{\Sigma} \partial_{x^1} = \partial_{x^2}$, $J_{\Sigma} \partial_{x^2} = \partial_{x^1}$. Let $(x^1, x^2, x_{1'}, x_{2'})$ be the induced coordinates on $T^*\Sigma$. It now follows that the para-complex structure J_- is given by

$$\begin{aligned} J_{-}\partial_{x_{1'}} &= -\partial_{x_{2'}} , \quad J_{-}\partial_{x^1} &= \partial_{x^2} - 2\frac{x_{2'}}{x^1}\partial_{x_{1'}} - 2\frac{x_{1'}}{x^1}\partial_{x_{2'}} , \\ J_{-}\partial_{x_{2'}} &= -\partial_{x_{1'}} , \quad J_{-}\partial_{x^2} &= \partial_{x^1} - 2\frac{x_{1'}}{x^1}\partial_{x_{1'}} - 2\frac{x_{2'}}{x^1}\partial_{x_{2'}} , \end{aligned}$$

with the corresponding Kähler two-form $\Omega_{-} = -d\iota J_{\Sigma} = dx^{1} \wedge dx_{2'} + dx^{2} \wedge dx_{1'}$.

2.3 Locally conformally flat null-Kähler structures

In this section we will focus on those (1, 1)-tensor fields that are two-step nilpotent, i.e., such that $J^2 = 0$. Such a tensor field is commonly said to be an *almost tangent structure* on M. The study of this kind of structures started with the works [51] and [52] around 1960 and they were further investigated during that decade in works such as [63] and [86]. If rank(J) = n, J is said to be a tangent structure or, as we will call it in what follows, a *null-Kähler structure* (see [60, 61]).

Lemma 2.21. A null-Kähler surface (M, g, J_0) is a Walker manifold.

Proof. $\mathcal{D} = \ker J = \operatorname{Im} J$ is a Walker distribution on (M, g, J_0) . Indeed, since $g(J_0X, Y) + g(X, J_0Y) = 0$, then

$$0 = g(J_0X, J_0Y) + g(X, J_0^2Y) = g(J_0X, JY),$$

so \mathcal{D} is degenerate. Besides, since J_0 is parallel with respect to the Levi-Civita connection of g,

$$\nabla_Z J_0 X = J_0 \nabla_Z X$$

for all vector field Z on M and all $X \in \mathcal{D}$, which shows that \mathcal{D} is also parallel.

As a consequence of Lemma 2.21, we can consider the induced Walker coordinates on M, with respect to which the metric is given by (2.1) and the Walker distribution is $\mathcal{D} = \text{span}\{\partial_{x_{1'}}, \partial_{x_{2'}}\}$. Since $\mathcal{D} = \text{Im}(J_0)$ we can write $J_0\partial_{x^1} = a\partial_{x_{1'}} + b\partial_{x_{2'}}$ and $J_0\partial_{x^2} = c\partial_{x_{1'}} + d\partial_{x_{2'}}$ for some real-valued smooth functions a, b, c and d on M. Imposing the condition for J_0 to be anti-self-adjoint we see that

$$0 = g(J_0\partial_{x^1}, \partial_{x^1}) + g(\partial_{x^1}, J_0\partial_{x^1}) = 2g(J_0\partial_{x^1}, \partial_{x^1}) = 2g(a\partial_{x_{1'}} + b\partial_{x_{2'}}, \partial_{x_1}) = 2a$$

and, analogously

$$2d = 0, \qquad c = -b.$$

Hence, in Walker coordinates, the null-Kähler structure takes the form

$$J_0 = \lambda(x^1, x^2, x_{1'}, x_{2'}) \left(dx_{2'} \otimes \frac{\partial}{\partial x^1} - dx_{1'} \otimes \frac{\partial}{\partial x^2} \right).$$

In this situation the associated two-form given by $\Omega(X, Y) = g(JX, Y)$ takes the form $\Omega = \lambda(x^1, x^2, x_{1'}, x_{2'})dx^1 \wedge dx^2$ and a standard calculation shows that it is closed if and only if $\lambda(x^1, x^2, x_{1'}, x_{2'}) = \lambda(x^1, x^2)$.

Theorem 2.22. Let (M, g, J) be a locally conformally flat null-Kähler structure. Then it is locally isometric to the cotangent bundle $T^*\Sigma$ of a strongly projectively flat affine surface (Σ, D) endowed with the Riemannian extension $g = g_D$.

Proof. It is known that the scalar curvature of a four-dimensional locally conformally flat Walker manifold $\tau = 3 \operatorname{tr}(T) + 12\iota X$ is necessarily zero, so the vector field X must vanish identically and the tensor field T must be trace-free. In this situation, the non-zero components of the covariant derivative ∇J_0 are given by

$$(\nabla_{\partial_{x^{1}}} J_{0}) \partial_{x^{1}} = T_{1}^{1} \lambda x_{1'} + T_{1}^{2} \lambda x_{2'} - \lambda (\Gamma_{11}^{1} + \Gamma_{12}^{2}) + \partial_{x^{1}} \lambda$$

$$= -(\nabla_{\partial_{x^{2}}} J_{0}) \partial_{x^{1}}$$

$$(\nabla_{\partial_{x^{1}}} J_{0}) \partial_{x^{2}} = T_{2}^{1} \lambda x_{1'} - T_{1}^{1} \lambda x_{2'} - \lambda (\Gamma_{12}^{1} + \Gamma_{22}^{2}) + \partial_{x^{2}} \lambda$$

$$= -(\nabla_{\partial_{x^{2}}} J_{0}) \partial_{x^{2}}$$

$$(2.6)$$

which are polynomials on the fibre coordinates. After analysing the conditions under which the expressions above vanish we easily deduce that the (1, 1)-tensor field T must be zero. This implies that the metric must be a deformed Riemannian extension $g = g_D + \pi^* \Phi$, which, in these coordinates, takes the form

$$g = \{-2\Gamma_{ij}^{\ k}y_k + \Phi_{ij}\}dx^i \otimes dx^j + dx^i \otimes dx_{i'} + dx_{i'} \otimes dx^i.$$

$$(2.7)$$

If we now consider $\bar{x}^i = x^i$ and $\bar{x}_{i'} = x_{i'} + \eta_i$ for some real-valued functions η_i on the base surface, and write $\partial_{\bar{x}^i} = a_i^{\ k} \partial_{x^k} + b_i^{\ k} \partial_{x_{k'}}$ and $\partial_{\bar{x}_{i'}} = c_i^{\ k} \partial_{x^k} + d_i^{\ k} \partial_{x_{k'}}$, we have

$$0 = d\bar{x}^{j}(\partial_{\bar{x}_{i'}}) = dx^{j}(c_{i}{}^{k}\partial_{x^{k}} + d_{i}{}^{k}\partial_{x_{k'}}) = c_{i}{}^{j},$$

$$\delta_{i}{}^{j} = d\bar{x}_{j'}(\partial_{\bar{x}_{i'}}) = dx_{j'}(d_{i}{}^{k}\partial_{x_{k'}}) = d_{i}{}^{j},$$

$$\delta_{i}{}^{j} = d\bar{x}^{j}(\partial_{\bar{x}^{i}}) = dx^{j}(a_{i}{}^{k}\partial_{x^{k}}) = a_{i}{}^{j},$$

$$0 = d\bar{x}_{j'}(\partial_{\bar{x}_{i}}) = (dx_{j'} + \partial_{x^{\ell}}\eta_{j}dx^{\ell})(a_{i}{}^{k}\partial_{x^{k}} + b_{i}{}^{k}\partial_{x_{k'}}) = \partial_{x^{i}}\eta_{j} + b_{i}{}^{j}.$$

so the coordinate vector fields are given by $\partial_{\bar{x}^i} = \partial_{x^i} - \partial_{x^i} \eta_k \partial_{y_k}$ and $\partial_{\bar{x}_{i'}} = \partial_{x_{i'}}$ for some real-valued functions η_i , i = 1, 2, on the base surface. It is easy to see that in these coordinates the metric takes the form

$$g = \{-2\Gamma_{ij}{}^k x_{k'} + 2\Gamma_{ij}{}^k \eta_k - \partial_{x^i} \eta_j - \partial_{x^j} \eta_i + \Phi_{ij}\} dx^i \otimes dx^j + dx^i \otimes dx_{i'} + dx_{i'} \otimes dx^i.$$

Afifi showed in [1] that it is possible to find two smooth functions η_1 and η_2 on Σ such that

$$\Phi_{ij} = -2\Gamma_{ij}{}^k\eta_k + \partial_{x^i}\eta_j + \partial_{x^j}\eta_i,$$

or, equivalently, a one-form η on Σ such that

$$\Phi(\partial_{x^i}, \partial_{x^j}) = (\nabla_{\partial_{x^i}} \eta) \partial_{x^j} + (\nabla_{\partial_{x^j}} \eta) \partial_{x^i}.$$
(2.8)

Therefore, the (0, 2)-tensor field Φ can be transformed away from Expression (2.7).

At this point, the only conditions we have left for J_0 to be parallel are

$$\lambda(\Gamma_{11}{}^{1} + \Gamma_{12}{}^{2}) + \partial_{x^{1}}\lambda = 0, \quad \lambda(\Gamma_{12}{}^{1} + \Gamma_{22}{}^{2}) + \partial_{x^{2}}\lambda = 0,$$
(2.9)

and the compatibility condition for this system of partial differential equations reduces to

$$\lambda \left\{ \partial_{x^2} (\Gamma_{12}{}^2 + \Gamma_{11}{}^1) - \partial_{x^1} (\Gamma_{12}{}^1 + \Gamma_{22}{}^2) \right\} = 0.$$
(2.10)

Besides, it is known [62, Sec. 34] that if a Riemannian extension of the form $g = g_D + \pi^* \Phi$ is locally conformally flat, then the base affine surface (Σ, D) must be projectively flat. This means that there exists a one-form ω on Σ such that

$$\Gamma_{ij}{}^{k} = -(\omega_i \delta_j{}^{k} + \omega_j \delta_i{}^{k}).$$

This implies that the compatibility condition (2.10) can be written as

$$-3\partial_{x^2}\omega(\partial_{x^1}) + 3\partial_{x^1}\omega(\partial_{x^2}) = 0, \qquad (2.11)$$

which means that the one-form ω must be closed and therefore there exists a real-valued function f locally defined on Σ such that $\omega = df$. Consequently, (Σ, D) must be locally strongly projectively flat. Under these conditions the metric is locally conformally flat and null-Kähler.

Conversely, if (Σ, D) is a strongly projectively flat affine surface, then there exist a one-form ω and a real-valued function f such that $\omega = df$ and the compatibility condition (2.11) automatically holds.

Chapter 3 Four-dimensional Kähler and para-Kähler Lie groups

In this chapter we will describe all the left-invariant para-Kähler structures on four-dimensional Lie groups and analyse their geometry, thus completing the analysis previously carried out in [31, 32, 101]. The geometry of the left-invariant Kähler structures obtained by Ovando in [116] will also be clarified. The results in this chapter are contained in the work [70].

3.1 Summary of results

A para-Kähler Lie algebra is a triple $(\mathfrak{g}, J, \langle \cdot, \cdot \rangle)$ so that

$$J^2 = \mathrm{Id}, \quad \langle Jx, Jy \rangle = -\langle x, y \rangle, \quad \nabla J = 0,$$

for all vectors $x, y \in \mathfrak{g}$. The associated two-form $\Omega(x, y) = \langle Jx, y \rangle$ is non-degenerate and closed, so (\mathfrak{g}, Ω) is a symplectic Lie algebra satisfying

$$\Omega(Jx, Jy) = -\Omega(x, y).$$

The eigenspaces $\mathfrak{L} = \ker (J - \mathrm{Id})$ and $\mathfrak{L}' = \ker (J + \mathrm{Id})$ are Lagrangian subalgebras and

$$\mathfrak{g}=\mathfrak{L}\oplus\mathfrak{L}'$$

is a Lagrangian decomposition of (\mathfrak{g}, Ω) .

For a fixed symplectic structure Ω on a Lie algebra \mathfrak{g} we will describe all the para-Kähler structures $(J, \langle \cdot, \cdot \rangle, \Omega)$, up to isometric automorphisms preserving the symplectic structure and modulo reversing the metric – in both cases the corresponding Lagrangian decomposition $\mathfrak{g} = \mathfrak{L} \oplus \mathfrak{L}'$ is preserved. The different Lagrangian decompositions, which are of much interest in the para-Kähler setting, will also be explicitly described in each case.

The geometry of four-dimensional para-Kähler Lie groups naturally splits into the symmetric and the non-symmetric cases. The latter splits into the semi-symmetric and the non-semi-symmetric situations. The symmetric case can be summarized as follows.

Theorem 3.1. Let $(G, \langle \cdot, \cdot \rangle, J)$ be a non-flat locally symmetric four-dimensional para-Kähler Lie group. Then, there are two distinct situations.

(i) If the Ricci operator is diagonalizable, then one of the following holds:

- *(i.a)* The Ricci operator vanishes and the anti-self-dual Weyl curvature operator is twostep nilpotent.
- (i.b) The para-holomorphic sectional curvature is a non-zero constant.
- *(i.c)* The metric is Einstein with non-zero scalar curvature, and the self-dual and anti-selfdual Weyl curvature operators are diagonalizable with the same eigenvalues.
- (i.d) The manifold is locally a product of two surfaces of constant Gaussian curvature. The self-dual and anti-self-dual Weyl curvature operators are diagonalizable with the same eigenvalues.
- (ii) If the Ricci operator is non-diagonalizable, then one of the following holds:
 - (ii.a) The Ricci operator has complex eigenvalues. The self-dual and anti-self-dual Weyl curvature operators are diagonalizable with the same eigenvalues.
 - *(ii.b)* The Ricci operator is two-step nilpotent and the anti-self-dual Weyl curvature operator either vanishes or is two-step nilpotent.

The structures in Assertion (*i.a*), which do not have a Kählerian counterpart, are realized on $\mathfrak{r}_{4,-1}$, $\mathfrak{r}_{4,-1,-1}$ and $\mathfrak{d}_{4,1}$, and they all correspond to symmetric Osserman manifolds – which are four-dimensional Einstein manifolds that are either self-dual or anti-self-dual – with non-diagonalizable Jacobi operators [77].

A para-Kähler manifold (M, g, J) is said to be *opposite para-Kähler* if there exists a para-Kähler structure (J', g) so that JJ' = J'J. In the four-dimensional setting, the corresponding para-Kähler forms Ω and Ω' induce opposite orientations and Q = JJ' defines a parallel product structure on the manifold. In this situation, (M, g) locally splits as a product of two oriented surfaces $M = N_1 \times N_2$ so that $J = J_1 \oplus J_2$ and $J' = J_1 \oplus -J_2$, where J_i is the para-complex structure induced by the volume form on N_i , for i = 1, 2. Conversely, a four-dimensional product of two oriented Lorentzian surfaces naturally inherits a para-Kähler and opposite para-Kähler structure. This is the case of the structures corresponding to Assertion (*i.d*) in Theorem 3.1.

As an immediate consequence of Theorem 3.1 and the results of Chapter 2 we conclude that locally conformally flat left-invariant para-Kähler structures can be modelled as follows.

Corollary 3.2. Let (M, g, J) be a locally conformally flat four-dimensional para-Kähler manifold. Then, it is either flat or locally isometric to one of the following para-Kähler Lie groups.

(i) The symplectic Lie algebra $(\mathfrak{r}'_2, \Omega)$ determined by $\Omega = e^{14} + e^{23}$, with the metric

$$\langle \cdot, \cdot \rangle = \frac{2}{\kappa} e^1 \circ e^2 - 2\kappa e^3 \circ e^4.$$

(ii) The symplectic Lie algebra $(\mathfrak{d}_{4,1},\Omega)$ determined by $\Omega = e^{12} - e^{34}$, with the metric

$$\langle \cdot, \cdot \rangle = 2(e^1 \circ e^3 + e^2 \circ e^4).$$

(iii) The symplectic Lie algebra $(\mathfrak{r}_2\mathfrak{r}_2,\Omega)$ determined by $\Omega = e^{12} + e^{34}$, with the metric

$$\langle \cdot, \cdot \rangle = \frac{1}{\kappa} (e^1 \circ e^1 - e^3 \circ e^3) - \kappa (e^2 \circ e^2 - e^4 \circ e^4)$$

Remark 3.3. Four-dimensional self-dual para-Kähler manifolds are either para-complex space forms – which are realizable as Lie groups and are the only ones that are strictly self-dual – or locally conformally flat, and so they are covered by the previous corollary (see [67]).

A pseudo-Riemannian manifold (M, g) is *semi-symmetric* if its curvature tensor satisfies $R(X, Y) \cdot R = 0$ for all vector fields X, Y on M, where R(X, Y) is acting on the curvature tensor R as a derivation. The special significance of the semi-symmetry condition lies in the fact that if a curvature tensor is semi-symmetric, then it is pointwise the curvature tensor of a symmetric space. However, the model symmetric space may change from point to point and there are examples of semi-symmetric manifolds which are not even locally homogeneous.

Another generalization of symmetric spaces is given by manifolds of *recurrent curvature*, which are those that admit a one-form ξ such that the covariant derivative of the curvature tensor satisfies $\nabla R = \xi \otimes R$. As we mentioned in Section 1.1.4, the curvature tensor induces a unique self-adjoint endomorphism \mathcal{R} of the space of two-forms defined by

$$\langle\!\langle \mathcal{R}(e^i \wedge e^j), e^k \wedge e^\ell \rangle\!\rangle = R(e_i, e_j, e_k, e_\ell).$$

A recurrent manifold is said to be *special* if there exist two two-forms $\alpha \wedge \beta$ and $\gamma \wedge \delta$ so that the curvature tensor acting on the space of two-forms is completely described by

$$\mathcal{R}(\alpha \wedge \beta) = \pm \gamma \wedge \delta.$$

Remark 3.4. Special recurrent manifolds are simply harmonic spaces, whose local structure was given in [125]. There it is shown that there exist local coordinates (x^1, x^2, x^3, x^4) with respect to which their metrics take the form

$$g = \Psi(x^1, x^2) dx^1 \circ dx^1 + 2dx^1 \circ dx^4 + 2dx^2 \circ dx^3,$$
(3.1)

where $\Psi(x^1, x^2)$ is an arbitrary function with non-constant $\partial_2 \partial_2 \Psi$ – otherwise, g would be locally symmetric. A standard calculation shows that special recurrent manifolds are semi-symmetric, since their curvature tensors correspond to those of the locally symmetric metrics given by $\Psi(x^1, x^2) = \pm (x^2)^2$. Moreover, given that

$$\begin{split} J_K &= \partial_{x^2} \otimes dx^1 + \partial_{x^3} \otimes dx^4 - \partial_{x^1} \otimes dx^2 - \partial_{x^4} \otimes dx^3 + \frac{1}{2} \Psi(\partial_{x^3} \otimes dx^1 + \partial_{x^4} \otimes dx^2), \\ J_{pK} &= \partial_{x^2} \otimes dx^2 + \partial_{x^4} \otimes dx^4 - \partial_{x^1} \otimes dx^1 - \partial_{x^3} \otimes dx^3 + \Psi \partial_{x^4} \otimes dx^1 \end{split}$$

are anti-commuting Kähler and para-Kähler structures with associated symplectic forms

$$\omega_{J_K} = \frac{1}{2} \Psi \, dx^1 \wedge dx^2 + dx^1 \wedge dx^3 - dx^2 \wedge dx^4 \quad \text{and} \quad \omega_{J_{pK}} = dx^1 \wedge dx^4 - dx^2 \wedge dx^3,$$

the metrics given by Equation (3.1) are (locally) hypersymplectic. Therefore, any special recurrent manifold admits a locally defined Kähler structure (g, J_K) and two locally defined para-Kähler structures (g, J_{pK}) and $(g, J_K J_{pK})$. Hypersymplectic structures on four-dimensional Lie algebras were classified by Andrada (see [5]), who showed that, besides the Abelian algebra \mathfrak{r}^4 , only \mathfrak{rh}_3 , $\mathfrak{r}_{4,-1,-1}$ or $\mathfrak{d}_{4,2}$ admit this kind of structures. The non-flat cases will be described in Section 3.11 and Section 3.17.1.

Remark 3.5. In addition to having a hypersymplectic structure, any special recurrent manifold is locally para-Kähler and opposite almost para-Kähler. Indeed, there exists an almost para-complex structure J'_{pK} , given by

$$J'_{nK} = \partial_{x^1} \otimes dx^1 + \partial_{x^2} \otimes dx^2 - \partial_{x^3} \otimes dx^3 - \partial_{x^4} \otimes dx^4 - \Psi \partial_{x^4} \otimes dx^1 \,,$$

that commutes with J_{pK} and (g, J'_{pK}) determines a locally defined almost para-Hermitian structure with closed fundamental form

$$\Omega_{J'_{nK}} = -dx^1 \wedge dx^4 - dx^2 \wedge dx^3.$$

Para-Kähler and opposite almost para-Kähler structures on non-flat Lie groups were considered in [45], where it is shown that their corresponding Ricci operators either vanish or are diagonalizable with two-dimensional kernels. In sharp contrast with the Kähler situation [7,44], there are many para-Kähler Lie groups which admit an opposite almost para-Kähler structure, even in the symmetric case. The Lie algebras $\mathfrak{rr}_{3,-1}$, $\mathfrak{r}_{4,-1,\beta}$ and $\mathfrak{r}_{4,-1,-1}$ admit Ricci-flat para-Kähler structures with associated opposite almost para-Kähler structures, and so do $\mathfrak{r}_{4,0}$, $\mathfrak{r}_{4,-1}$ and $\mathfrak{r}_2\mathfrak{r}_2$. The Lie algebra $\mathfrak{d}_{4,2}$ admits para-Kähler structures that are not semi-symmetric and have associated opposite almost para-Kähler structure.

The geometry of para-Kähler Lie groups is quite rigid in the non-symmetric case, where there are two essentially different possibilities depending on whether the curvature tensor is semi-symmetric or not. The semi-symmetric case can be summarized as follows.

Theorem 3.6. Let $(G, \langle \cdot, \cdot \rangle, J)$ be a four-dimensional para-Kähler Lie group that is not locally symmetric. Then, its curvature tensor is semi-symmetric if and only if its associated Ricci operator vanishes. Moreover, all these structures are recurrent and harmonic.

Although all the structures covered by Theorem 3.6 have analogous local descriptions as in Remark 3.4, they are not necessarily isometric, since their associated recurrence one-forms may have different causalities. Theorem 3.6 also holds true in the Kähler situation (cf. Theorem 3.16).

Finally, the class of non-symmetric para-Kähler Lie groups whose curvature tensors are not semi-symmetric is described as follows.

Theorem 3.7. Let $(G, \langle \cdot, \cdot \rangle, J)$ be a four-dimensional para-Kähler Lie group whose curvature tensor is not semi-symmetric. Then, one of the following statements holds:

(i) The Ricci operator has a single eigenvalue which is a double root of its minimal polynomial. The anti-self-dual Weyl curvature operator is three-step nilpotent.

- (ii) The Ricci operator is diagonalizable with two-dimensional kernel. Moreover,
 - (ii.a) the self-dual and anti-self-dual Weyl curvature operators have the same eigenvalues and W^- has a double root of its minimal polynomial, or
 - (ii.b) the self-dual and anti-self-dual Weyl curvature operators have opposite eigenvalues and both operators W^{\pm} are diagonalizable.

The only geometry in Theorem 3.7 with a Kähler counterpart corresponds to Assertion (*ii.b*), as shown in Theorem 3.16.

3.2 Symplectic Lie algebras

Four-dimensional symplectic Lie algebras are necessarily solvable (see [50]) and they were classified by Ovando in [115]. We follow the notation in Ovando's works to denote the different four-dimensional symplectic Lie algebras, which correspond to certain semi-direct extensions of the Abelian algebra \mathfrak{r}^3 , the Euclidean algebra $\mathfrak{e}(2)$, the Poincaré algebra $\mathfrak{e}(1,1)$ and the Heisenberg algebra \mathfrak{h}^3 admitting symplectic structures.

The non-zero Lie brackets on each Lie algebra are given in terms of a basis $\{e_1, e_2, e_3, e_4\}$ of the corresponding algebras and e^i will denote the dual of e_i . In order to simplify the notation, in what follows we will write e^{ij} to denote $e^i \wedge e^j$.

Solvable Lie algebra	Symplectic form
$\mathfrak{r}_2\mathfrak{r}_2$: $[e_1, e_2] = e_2, [e_3, e_4] = e_4$	$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{34}e^{34}$
	$\alpha_{12}\alpha_{34} \neq 0$
$\mathfrak{rh}_3 \colon [e_1, e_2] = e_3$	$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{23}e^{23} + \alpha_{24}e^{24}$
	$\alpha_{14}\alpha_{23} - \alpha_{13}\alpha_{24} \neq 0$
$\mathfrak{rr}_{3,0}$: $[e_1, e_2] = e_2$	$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{34}e^{34}$
	$\alpha_{12}\alpha_{34} \neq 0$
$\mathfrak{rr}'_{3,0}$: $[e_1, e_2] = -e_3, [e_1, e_3] = e_2$	$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{23}e^{23}$
	$\alpha_{14}\alpha_{23} \neq 0$
$\mathfrak{rr}_{3,-1}$: $[e_1,e_2] = e_2, [e_1,e_3] = -e_3$	$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{23}e^{23}$
	$\alpha_{14}\alpha_{23} \neq 0$
$[e_1, e_3] = e_3, [e_1, e_4] = e_4,$	$\omega = \alpha_{12}e^{12} + \alpha_{13}\left(e^{13} - e^{24}\right) + \alpha_{14}\left(e^{14} + e^{23}\right)$
$[e_2, e_3] = e_4, [e_2, e_4] = -e_3$	$\alpha_{13}^2 + \alpha_{14}^2 \neq 0.$
\mathfrak{n}_4 : $[e_1, e_4] = -e_2, [e_2, e_4] = -e_3$	$\omega = \alpha_{12}e^{12} + \alpha_{14}e^{14} + \alpha_{24}e^{24} + \alpha_{34}e^{34}$
	$\alpha_{12}\alpha_{34} \neq 0$
$\mathfrak{r}_{4,0}$: $[e_1, e_4] = -e_1, [e_3, e_4] = -e_2$	$\omega = \alpha_{14}e^{14} + \alpha_{23}e^{23} + \alpha_{24}e^{24} + \alpha_{34}e^{34}$
	$\alpha_{14}\alpha_{23} \neq 0$
$[e_1, e_4] = -e_1, [e_2, e_4] = e_2$	$\omega = \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{24}e^{24} + \alpha_{34}e^{34}$
$[e_3, e_4] = e_3 - e_2$	$\alpha_{13}\alpha_{24} \neq 0$

Solvable Lie algebra	Symplectic form
$[e_1, e_4] = -e_1, \ [e_2, e_4] = e_2,$	$\omega = \alpha_{12}e^{12} + \alpha_{14}e^{14} + \alpha_{24}e^{24} + \alpha_{34}e^{34}$
$[e_{4,-1,\beta}: [e_{3},e_{4}] = -\beta e_{3}, -1 < \beta < 0$	$\alpha_{12}\alpha_{34} \neq 0$
$[e_1, e_4] = -e_1, \ [e_2, e_4] = e_2,$	$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{24}e^{24} + \alpha_{34}e^{34}$
$[e_3, e_4] = e_3$	$\alpha_{13}\alpha_{24} - \alpha_{12}\alpha_{34} \neq 0$
$[e_1, e_4] = -e_1, [e_2, e_4] = \alpha e_2,$	$\omega = \alpha_{14}e^{14} + \alpha_{23}e^{23} + \alpha_{24}e^{24} + \alpha_{34}e^{34}$
$[e_{4,-\alpha,\alpha}]{}^{\epsilon}$ $[e_{3},e_{4}] = -\alpha e_{3}, \ 0 < \alpha < 1$	$\alpha_{14}\alpha_{23} \neq 0$
$[e_1, e_4] = -e_1, \ [e_2, e_4] = \delta e_3,$	$\omega = \alpha_{14}e^{14} + \alpha_{23}e^{23} + \alpha_{24}e^{24} + \alpha_{34}e^{34}$
$[e_{4,0,\delta}] [e_3, e_4] = -\delta e_2, \delta > 0$	$\alpha_{14}\alpha_{23} \neq 0$
$[e_1, e_2] = e_3, [e_1, e_4] = -\frac{1}{2}e_1$	$\omega = \alpha_{12} \left(e^{12} - e^{34} \right) + \alpha_{14} e^{14} + \alpha_{24} e^{24}$
$\mathfrak{h}_4\colon [e_3,e_4]=-e_3,$	$ a_{12} (0 \ 0 \) + a_{14} 0 \ + a_{24} 0 $
$[e_2, e_4] = -e_1 - \frac{1}{2}e_2$	
$[e_1, e_2] = e_3, [e_1, e_4] = -e_1,$	$\omega = \alpha_{12} \left(e^{12} - e^{34} \right) + \alpha_{14} e^{14} + \alpha_{24} e^{24}$
$[e_3, e_4] = -e_3$	$\alpha_{12} \neq 0$
$[e_1, e_2] = e_3, [e_1, e_4] = -\frac{1}{2}e_1$	$\omega = \alpha_{12} \left(e^{12} - e^{34} \right) + \alpha_{14} e^{14} + \alpha_{24} e^{24}$
$[e_4, \frac{1}{2}]$ $[e_2, e_4] = -\frac{1}{2}e_2, [e_3, e_4] = -e_3$	$\alpha_{12} \neq 0$
$[e_1, e_2] = e_3, \ [e_1, e_4] = -\lambda e_1,$	
$[e_2, e_4] = (\lambda - 1)e_2,$	$\omega = \alpha_{12} \left(e^{12} - e^{34} \right) + \alpha_{14} e^{14} + \alpha_{24} e^{24}$
$[e_{4,\lambda}: [e_3, e_4] = -e_3,$	$\alpha_{12} \neq 0$
$\lambda > \frac{1}{2}, \lambda \neq 1, 2$	
$[e_1, e_2] = e_3, [e_1, e_4] = -2e_1,$	$\omega = \alpha_{12} \left(e^{12} - e^{34} \right) + \alpha_{14} e^{14} + \alpha_{23} e^{23} + \alpha_{24} e^{24}$
$[e_2, e_4] = e_2, [e_3, e_4] = -e_3$	$\alpha_{12}^2 - \alpha_{14}\alpha_{23} \neq 0$
$[e_1, e_2] = e_3, [e_1, e_4] = e_2 - \frac{\delta}{2}e_1,$	$(1 - \alpha_{12})(e^{12} - \delta e^{34}) + \alpha_{14}e^{14} + \alpha_{24}e^{24}$
$\mathfrak{d}'_{4,\delta}$: $[e_2, e_4] = -e_1 - \frac{\delta}{2}e_2,$	$\omega = \alpha_{12} (c + cc + \alpha_{14}c + \alpha_{24}c)$ $\alpha_{12} \neq 0$
$[e_3, e_4] = -\delta e_3, \delta > 0$	

Table 3.1: Four-dimensional symplectic Lie algebras as given in [115]. We have highlighted those which do not admit a para-Kähler structure.

For each symplectic Lie algebra (\mathfrak{g}, ω) in Table 3.1 we will determine all the compatible para-Kähler structures. To do so, we start with an arbitrary (1, 1)-tensor field $J = (a_{ij})$ and determine all the conditions on its components that are necessary for it to satisfy

(1)
$$J^2 = \text{Id},$$

(2) $\omega(Jx, Jy) = -\omega(x, y)$, and

(3) the integrability condition given by

$$N_J(x, y) = [Jx, Jy] - J[Jx, y] - J[x, Jy] + [x, y] = 0,$$

so that the metric $\langle x, y \rangle = \omega(Jx, y)$ gives a para-Kähler metric. Note that the conditions (1)–(3) determine a system of polynomial equations on the parameters $\{a_{ij}\}$ which we will need to solve explicitly in order to describe all the possible para-Kähler structures in each case.

First of all, we point out that not all of the symplectic Lie algebras (\mathfrak{g}, ω) in Table 3.1 admit a para-Kähler structure. Indeed, even though the Lie algebras

$$\begin{aligned} \mathfrak{rr}_{3,0}': & [e_1, e_2] = -e_3, \ [e_1, e_3] = e_2, \\ \mathfrak{n}_4: & [e_1, e_4] = -e_2, \ [e_2, e_4] = -e_3, \\ \mathfrak{r}_{4,0,\delta}': & [e_1, e_4] = -e_1, \ [e_2, e_4] = \delta e_3, \ [e_3, e_4] = -\delta e_2, \\ \mathfrak{d}_{4,\delta}': & [e_1, e_2] = e_3, \ [e_1, e_4] = e_2 - \frac{\delta}{2} e_1, \ [e_2, e_4] = -e_1 - \frac{\delta}{2} e_2, \ [e_3, e_4] = -\delta e_3, \end{aligned}$$

where $\delta > 0$, admit symplectic structures, a straightforward calculation shows that none of them admits a para-Kähler structure.

Remark 3.8. According to [6], the four Lie algebras listed above correspond to the unimodular Lie group given by the trivial extension $\mathbb{R} \times \tilde{E}(2)$ (see Lemma 1.11-(*i*)), the semi-direct extensions $\mathbb{R}^3 \rtimes_{\varphi_i} \mathbb{R}e_4$ given by

$$\varphi_1 = \operatorname{ad}(e_4): \quad \varphi(e_1) = e_2, \quad \varphi(e_2) = e_3, \quad \varphi(e_3) = 0,$$

 $\varphi_2 = \operatorname{ad}(e_4): \quad \varphi(e_1) = e_1, \quad \varphi(e_2) = -\delta e_3, \quad \varphi(e_3) = \delta e_2,$

and the semi-direct extension $H^3 \rtimes_{\varphi} \mathbb{R}e_4$ given by

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = \frac{\delta}{2}e_1 - e_2, \quad \varphi(e_2) = e_1 + \frac{\delta}{2}e_2, \quad \varphi(e_3) = \delta e_3,$$

respectively.

Any other symplectic Lie algebra in Ovando's classification [115] does admit at least one para-Kähler structure. We will study all of them separately in the subsequent sections. For the sake of simplicity, we will omit the details of most of the calculations and outline the different para-Kähler structures and their curvatures. Our description of para-Kähler Lie algebras will be given up to symplectomorphical equivalence, i.e., up to Lie algebra automorphisms that preserve the symplectic structure.

3.3 Para-Kähler structures on $r_{4,0}$

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra $\mathfrak{r}_{4,0}$ determined by

$$[e_1, e_4] = -e_1, \quad [e_3, e_4] = -e_2,$$

which, as it was shown in [6], is equivalent to the Lie algebra of the semi-direct extension $\mathbb{R}e_4\ltimes_{\varphi}\mathbb{R}^3$ where

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = e_1, \quad \varphi(e_2) = 0, \quad \varphi(e_3) = -e_2.$$

The symplectic structures on $r_{4,0}$ are given by (see [115])

$$\omega = \alpha_{14}e^{14} + \alpha_{23}e^{23} + \alpha_{24}e^{24} + \alpha_{34}e^{34}, \qquad \alpha_{14}\alpha_{23} \neq 0.$$

Making use of the automorphisms of $r_{4,0}$, which are of the form

$$\Phi = \begin{pmatrix} z_{11} & 0 & 0 & z_{14} \\ 0 & z_{22} & z_{23} & z_{24} \\ 0 & 0 & z_{22} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix} \quad \text{with } z_{11}z_{22} \neq 0,$$

it is easy to see that any symplectic form on $r_{4,0}$ is symplectomorphically equivalent to

$$\Omega_{\varepsilon} = e^{14} + \varepsilon e^{23}, \quad \varepsilon^2 = 1.$$

Now, the automorphisms preserving the symplectic structure $(\mathfrak{r}_{4,0}, \Omega_{\varepsilon})$ are those determined by Φ above satisfying the conditions

$$\Phi_1: z_{11} = 1, \quad z_{24} = 0, \quad z_{34} = 0, \quad z_{22} = 1,$$

$$\Phi_2: z_{11} = 1, \quad z_{24} = 0, \quad z_{34} = 0, \quad z_{22} = -1.$$

Bearing this in mind, we determine the conditions that are necessary for the compatibility of the para-complex structure with the symplectic structures and the integrability of the para-complex structure (i.e., the integrability of the eigenspaces ker($J \mp Id$) corresponding to the eigenvalues ± 1 of J, which is equivalent to the vanishing of the Nijenhuis tensor).

The compatibility condition $\Omega_{\varepsilon}(J\cdot, J\cdot) = -\Omega_{\varepsilon}(\cdot, \cdot)$ determines the following system of polynomial equation on the components $\{a_{ij}\}$

$$\varepsilon a_{31} - a_{42} = 0, \quad -(\varepsilon a_{21} + a_{43}) = 0, \quad -\varepsilon (a_{22} + a_{33}) = 0,$$

 $-(a_{11} + a_{44}) = 0, \quad -(a_{12} + \varepsilon a_{34}) = 0, \quad -a_{13} + \varepsilon a_{24} = 0,$

and so the para-complex structure J must be such that

$$Je_1 = a_{11}e_1 + a_{21}e_2 + a_{31}e_3 + a_{41}e_4, \quad Je_3 = \varepsilon a_{24}e_1 + a_{23}e_2 - a_{22}e_3 - \varepsilon a_{21}e_4,$$

$$Je_2 = -\varepsilon a_{34}e_1 + a_{22}e_2 + a_{32}e_3 + \varepsilon a_{31}e_4, \quad Je_4 = a_{14}e_1 + a_{24}e_2 + a_{34}e_3 - a_{11}e_4.$$

Now, a straightforward calculation shows that the non-zero components of the Nijenhuis tensor are determined by

$$\begin{split} N_J(e_2, e_1) &= \varepsilon a_{34} a_{41} e_1 - (a_{32} a_{41} + \varepsilon a_{31} (a_{21} - a_{31})) e_2 - \varepsilon a_{31}^2 e_3 - \varepsilon a_{31} a_{41} e_4, \\ N_J(e_3, e_1) &= -\varepsilon a_{41} (a_{24} + a_{34}) e_1 + (2a_{22} a_{41} + \varepsilon a_{21} (a_{21} - a_{31})) e_2 \\ &+ (a_{32} a_{41} + \varepsilon a_{21} a_{31}) e_3 + \varepsilon a_{41} (a_{21} + a_{31}) e_4, \\ N_J(e_3, e_2) &= - (a_{24} a_{31} - (a_{21} - a_{31}) a_{34}) e_1 + \varepsilon (2a_{22} a_{31} - a_{21} a_{32}) e_2 + \varepsilon a_{31} a_{32} e_3 + a_{31}^2 e_4, \\ N_J(e_4, e_1) &= (1 - a_{11}^2 - a_{14} a_{41} + \varepsilon a_{31} a_{34}) e_1 - (a_{31} (a_{11} + a_{22}) + a_{34} a_{41}) e_2 - a_{31} a_{32} e_3 - \varepsilon a_{31}^2 e_4, \end{split}$$

$$\begin{split} N_J(e_4, e_2) &= -\varepsilon \left(a_{14}a_{31} - (2a_{11} + a_{32})a_{34} \right) e_1 \\ &- \left(a_{32}(a_{11} + a_{22}) - \varepsilon (a_{21} - a_{31})a_{34} \right) e_2 + \left(\varepsilon a_{31}a_{34} - a_{32}^2 \right) e_3 + \varepsilon (a_{34}a_{41} - a_{31}a_{32}) e_4, \\ N_J(e_4, e_3) &= \varepsilon \left(a_{14}a_{21} - a_{22}a_{34} - a_{11}(2a_{24} + a_{34}) \right) e_1 \\ &+ \left(1 + a_{22}(2a_{11} + a_{22}) + \varepsilon a_{21}(a_{34} - a_{24}) \right) e_2 + \left(a_{32}(a_{11} + a_{22}) - \varepsilon a_{24}a_{31} \right) e_3 \\ &+ \varepsilon \left(a_{31}(a_{11} + a_{22}) - a_{24}a_{41} \right) e_4, \end{split}$$

which again determine a system of polynomial equations on $\{a_{ij}\}\$ that needs to be solved in order to obtain the integrability conditions on J. It immediately follows from the expressions of $N_J(e_2, e_1)$ and $N_J(e_4, e_2)$ that both a_{31} and a_{32} must vanish. In this situation

$$N_J(e_3, e_2) = a_{21}a_{34}e_1$$

and so there are two different possibilities depending on whether $a_{34} = 0$ or not. A straightforward calculation shows that there can be no para-Kähler structures with $a_{34} \neq 0$. Therefore, we can assume that $a_{34} = 0$. Then the condition $J^2 = \text{Id}$ gives the following system of polynomial equations

$$a_{11}^2 + a_{14}a_{41} = 0,$$
 $a_{21}(a_{11} + a_{22}) + a_{24}a_{41} = 0,$
 $a_{22}^2 - 1 = 0,$ $a_{14}a_{21} + a_{24}(a_{22} - a_{11}) = 0,$

from where it follows that $a_{22} = \varepsilon_2$, with $\varepsilon_2^2 = 1$. At this point,

$$N_J(e_3, e_1) = -\varepsilon a_{24}a_{41}e_1 + (2\varepsilon_2 a_{41} + \varepsilon a_{21}^2)e_2 + \varepsilon a_{21}a_{41}e_4,$$

which implies that there are again two different possibilities depending on whether $a_{41} = 0$ or not. A straightforward calculation shows that the condition $a_{41} \neq 0$ is incompatible with the integrability of the para-complex structure. Consequently, we assume that $a_{41} = 0$. Now,

$$N_J(e_3, e_1) = \varepsilon a_{21}^2 e_2, \qquad N_J(e_4, e_1) = (1 - a_{11}^2) e_1,$$

so $a_{21} = 0$ and $a_{11} = \varepsilon_1$, for $\varepsilon_1^2 = 1$. At this point, the only non-zero component of the Nijenhuis tensor is

$$N_J(e_4, e_3) = -2\varepsilon_1\varepsilon a_{24}e_1 + 2(1+\varepsilon_1\varepsilon_2)e_2.$$

Therefore, $a_{24} = 0$ and $\varepsilon_2 = -\varepsilon_1$, and the condition $J^2 = \text{Id}$ automatically holds. What we have just obtained is the para-Kähler structures $(J, \langle \cdot, \cdot \rangle)$ on $\mathfrak{r}_{4,0}$ given by

$$\begin{cases} Je_1 = \varepsilon_1 e_1, Je_2 = -\varepsilon_1 e_2, Je_3 = a_{23}e_2 + \varepsilon_1 e_3, Je_4 = a_{14}e_1 - \varepsilon_1 e_4, \\ \langle \cdot, \cdot \rangle = 2(-\varepsilon_1 e^1 \circ e^4 - \varepsilon_1 \varepsilon e^2 \circ e^3) - \varepsilon a_{23}e_3 \circ e_3 - a_{14}e_4 \circ e_4. \end{cases}$$

It is easy to see that all these structures are symplectomorphically equivalent to

$$(J, \langle \cdot, \cdot \rangle) : \begin{cases} Je_1 = -e_1, \quad Je_2 = e_2, \quad Je_3 = -e_3, \quad Je_4 = e_4, \\ \langle \cdot, \cdot \rangle = 2(e^1 \circ e^4 - \varepsilon e^2 \circ e^3) \end{cases}$$

through the symplectomorphism determined by Φ_1 satisfying $z_{14} = -\frac{\varepsilon_1}{2}a_{14}$ and $z_{23} = \frac{\varepsilon_1}{2}a_{23}$. These structures correspond to the decomposition of $(\mathfrak{r}_{4,0}, \Omega_{\varepsilon})$ as a direct sum of Lagrangian subalgebras as

$$\mathfrak{r}_{4,0} = \mathfrak{L} \oplus \mathfrak{L}' = \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_1, e_3\},$$

where \mathcal{L} and \mathcal{L}' are the eigenspaces associated to the eigenvalues ± 1 of the para-complex structure J, respectively.

The structures $(\mathfrak{r}_{4,0}, J, \langle \cdot, \cdot \rangle)$ are Ricci-flat and a straightforward calculation shows that their curvature tensors satisfy $\nabla R = 2e^4 \otimes R$, so they are recurrent with recurrence one-form $\xi = 2e^4$. Besides, the corresponding curvature tensors acting on the space of two-forms are given by

$$\mathcal{R}(e^3 \wedge e^4) = -e^1 \wedge e^2 = \varepsilon R(e_3, e_4, e_3, e_4)e^1 \wedge e^2.$$

Therefore, the para-Kähler structures $(\mathfrak{r}_{4,0}, J, \langle \cdot, \cdot \rangle)$ are special recurrent and, consequently, simply harmonic, so they are locally modelled on (3.1) and their curvature tensors are semi-symmetric. These structures are thus covered by Theorem 3.6.

Direct calculations similar to the ones described above show that the structures $(J, \langle \cdot, \cdot \rangle)$ admit opposite almost para-Kähler structures compatible with the opposite symplectic forms $\Omega' = e^{14} - \varepsilon e^{23} - \mu e^{34}$, with $\mu \in \mathbb{R}$.

3.4 Para-Kähler structures on $r_{4,-1}$

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra $\mathfrak{r}_{4,-1}$ determined by

$$[e_1, e_4] = -e_1, \qquad [e_2, e_4] = e_2, \qquad [e_3, e_4] = e_3 - e_2,$$

which, according to [6], corresponds to the Lie algebra of the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} \mathbb{R}^3$ where

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = e_1, \quad \varphi(e_2) = -e_2, \quad \varphi(e_3) = e_3 - e_2.$$

Any symplectic form on $r_{4,-1}$ is of the form

$$\omega = \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{24}e^{24} + \alpha_{34}e^{34}, \qquad \alpha_{13}\alpha_{24} \neq 0,$$

and the automorphisms of this Lie algebra are given by

$$\Phi = \begin{pmatrix} z_{11} & 0 & 0 & z_{14} \\ 0 & z_{22} & z_{23} & z_{24} \\ 0 & 0 & z_{22} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix} \quad \text{with} \quad z_{11}z_{22} \neq 0.$$

Now, it is not difficult to check that any symplectic structure is symplectomorphically equivalent to $\Omega = e^{13} + e^{24}$. Moreover, the automorphisms preserving the symplectic structure $(\mathfrak{r}_{4,-1}, \Omega)$ are given by Φ above with $z_{11} = 1$, $z_{22} = 1$, $z_{34} = 0$ and $z_{23} = z_{14}$. Now, a straightforward

calculation as in the previous section shows that any para-Kähler structure on $r_{4,-1}$ is equivalent to

$$(J, \langle \cdot, \cdot \rangle) : \begin{cases} Je_1 = -e_1, & Je_2 = e_2, & Je_3 = -\kappa e_1 + e_3, & Je_4 = -e_4 \\ \langle \cdot, \cdot \rangle = 2(e^1 \circ e^3 - e^2 \circ e^4) + \kappa e^3 \circ e^3, & \kappa \in \mathbb{R}, \end{cases}$$

which correspond to the Lagrangian decomposition

$$\mathfrak{r}_{4,-1} = \mathfrak{L} \oplus \mathfrak{L}' = \operatorname{span}\{e_2, e_3 - \frac{\kappa}{2}e_1\} \oplus \operatorname{span}\{e_1, e_4\}.$$

The structures above are flat if $\kappa = 0$. Otherwise, they are Ricci-flat, locally symmetric – thus covered by Theorem 3.1 – and the underlying structures are locally modelled on (3.1) with

$$\Psi(x^1, x^2) = \pm (x^2)^2.$$

Furthermore, the structures above admit opposite almost para-Kähler structures which are compatible with the opposite symplectic form $\Omega' = e^{13} - e^{24} + \mu e^{34}$ with $\mu \in \mathbb{R}$.

3.5 Para-Kähler structures on r_2r_2

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra $\mathfrak{r}_2\mathfrak{r}_2$ determined by

$$[e_1, e_2] = e_2, \qquad [e_3, e_4] = e_4,$$

which, according to [6], corresponds to the Lie algebra of the non-unimodular semi-direct extension $E(1,1) \rtimes \mathbb{R}$ isomorphic to $\mathfrak{aff}(\mathbb{R}) \times \mathfrak{aff}(\mathbb{R})$ as in Lemma 1.9.

We make use of Lie algebra automorphisms to simplify the final expressions of the structures. The automorphisms of this Lie algebra are given by

$$\Phi_{1} = \begin{pmatrix} 0 & 0 & 1 & 0 \\ 0 & 0 & z_{23} & z_{24} \\ 1 & 0 & 0 & 0 \\ z_{41} & z_{42} & 0 & 0 \\ z_{24}z_{42} \neq 0 \end{pmatrix}, \qquad \Phi_{2} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ z_{21} & z_{22} & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & z_{43} & z_{44} \end{pmatrix}.$$

Any symplectic structure on r_2r_2 is given by

$$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{34}e^{34}, \qquad \alpha_{12}\alpha_{34} \neq 0,$$

and Ovando showed in [115] that these $(\mathfrak{r}_2\mathfrak{r}_2,\omega)$ are symplectomorphically equivalent to

$$\Omega_{\lambda} = e^{12} + e^{34} + \lambda e^{13} \quad \text{with} \quad \lambda \ge 0.$$

It is easy to check that the symplectic form Ω_{λ} is preserved by any automorphism Φ_2 with $z_{22} = z_{44} = 1$ (in the special case where $\lambda = 0$, the automorphisms Φ_1 with $z_{24} = z_{42} = 1$ also preserve

the symplectic form Ω_0). Considering the action of the symplectomorphisms Φ_2 that preserve Ω_{λ} we are allowed to restrict our study to the para-complex structures $J = (a_{ij})$ satisfying one of the following conditions.

Case 1. $a_{12} = 0$ and $a_{14} = 0$. Case 2. $a_{12} = 0$ and $a_{14} \neq 0$, in which case one may also assume $a_{13} = 0$. Case 3. $a_{12} \neq 0$ and $a_{14} = 0$, in which case one may also assume $a_{11} = 0$. Case 4. $a_{12} \neq 0$ and $a_{14} \neq 0$, in which case one may also assume $a_{11} = a_{13} = 0$.

In each case we determine the necessary conditions for the compatibility of the para-complex structure with the symplectic structures and the integrability of the latter. To do so, we distinguish the two possibilities corresponding to the symplectic structures Ω_{λ} (with $\lambda \neq 0$) and Ω_{0} , since they give rise to different geometries.

3.5.1 Para-Kähler structures on $(\mathfrak{r}_2\mathfrak{r}_2, \Omega_\lambda)$ with $\lambda > 0$

A straightforward calculation reveals that the Cases 2, 3 and 4 above are not compatible with the existence of para-Kähler structures. Therefore we focus on the case $a_{12} = a_{14} = 0$. Long but straightforward calculations show that the para-Kähler structures $(J, \langle \cdot, \cdot \rangle)$ which are compatible with Ω_{λ} are symplectomorphically equivalent to one of the following three families of para-Kähler structures

$$\begin{split} (J_{11}, \langle \cdot, \cdot \rangle_{11}) &: \begin{cases} J_{11}e_1 = -e_1, & J_{11}e_2 = e_2, & J_{11}e_3 = e_3, & J_{11}e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_{11} = 2(e^1 \circ e^2 + \lambda e^1 \circ e^3 - e^3 \circ e^4), \end{cases} \\ (J_{12}, \langle \cdot, \cdot \rangle_{12}) &: \begin{cases} J_{12}e_1 = -e_1 + 2\lambda e_2 - 2e_3, & J_{12}e_2 = e_2, \\ J_{12}e_3 = e_3, & J_{12}e_4 = 2e_2 - e_4, \\ \langle \cdot, \cdot \rangle_{12} = 2(e^1 \circ e^2 + \lambda e^1 \circ e^3 + 2e^1 \circ e^4 - e^3 \circ e^4), \end{cases} \\ (J_{13}, \langle \cdot, \cdot \rangle_{13}) &: \begin{cases} J_{13}e_1 = -e_1, & J_{13}e_2 = e_2 - 2e_4, \\ J_{13}e_3 = 2e_1 + e_3 + 2\lambda e_4, & J_{13}e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_{13} = 2(e^1 \circ e^2 + \lambda e^1 \circ e^3 - 2e^2 \circ e^3 - e^3 \circ e^4). \end{cases} \end{split}$$

The structures above correspond to the decompositions $\mathfrak{r}_2\mathfrak{r}_2 = \mathfrak{L}_{1i} \oplus \mathfrak{L}'_{1i}$ as direct sums of Lagrangian subalgebras given by

$$\begin{aligned} \mathbf{r}_{2}\mathbf{r}_{2} &= \mathfrak{L}_{11} \oplus \mathfrak{L}_{11}' = \operatorname{span}\{e_{2}, e_{3}\} \oplus \operatorname{span}\{e_{1}, e_{4}\} \\ &= \mathfrak{L}_{12} \oplus \mathfrak{L}_{12}' = \operatorname{span}\{e_{2}, e_{3}\} \oplus \operatorname{span}\{e_{4} - e_{2}, e_{1} + e_{3} - \lambda e_{2}\} \\ &= \mathfrak{L}_{13} \oplus \mathfrak{L}_{13}' = \operatorname{span}\{e_{4} - e_{2}, e_{1} + e_{3} + \lambda e_{2}\} \oplus \operatorname{span}\{e_{1}, e_{4}\}.\end{aligned}$$

Moreover, a straightforward calculation shows that all the structures above are non-flat and Ricci-flat with recurrent curvature, i.e., their curvature tensors satisfy

$$\nabla_{1i}R_{1i} = \xi_{1i} \otimes R_{1i},$$

and their recurrence one-forms ξ_{1i} are given by

$$\xi_{11} = 2(e^1 + e^3), \quad \xi_{12} = 2e^3, \quad \xi_{13} = 2e^1,$$

Furthermore, the corresponding curvature operators $\mathcal{R}_{1i} : \Lambda^2 \to \Lambda^2$ acting on the space of twoforms are given by

$$\mathcal{R}_{1i}(e^1 \wedge e^2) = -\lambda e^2 \wedge e^4 = R_{1i}(e_1, e_3, e_3, e_1)e^2 \wedge e^4, \text{ for } i = 1, 2, 3,$$

from where it follows that all the para-Kähler structures above are special recurrent and thus locally modelled on (3.1). Consequently, their curvature tensors are semi-symmetric and they are covered by Theorem 3.6.

Finally, the structures $(J_{11}, \langle \cdot, \cdot \rangle_{11})$ admit opposite almost para-Kähler structures compatible with the opposite symplectic forms $\Omega'_{11} = e^{12} - e^{34} + (\mu - \lambda)e^{13}$ with $\mu \in \mathbb{R}$, while the structures $(J_{12}, \langle \cdot, \cdot \rangle_{12})$ and $(J_{13}, \langle \cdot, \cdot \rangle_{13})$ do not admit opposite almost para-Kähler structures.

3.5.2 Para-Kähler structures on $(\mathfrak{r}_2\mathfrak{r}_2, \Omega_0)$

Proceeding as in the previous case, it is straightforward to check that no para-Kähler structures exist in Case 2 while the other three cases give rise to three essentially different geometries.

Para-Kähler structures of constant para-holomorphic sectional curvature

Assuming that $a_{12} \neq 0$ and $a_{14} \neq 0$ as in Case 4, any para-Kähler structure is equivalent to the structures $(J_{21}, \langle \cdot, \cdot \rangle_{21})$, which are given by

$$\begin{cases} J_{21}e_1 = e_3 - \frac{1}{\kappa}e_2, \ J_{21}e_2 = -\kappa(e_1 + e_3), \ J_{21}e_4 = e_4 - e_2 - \kappa(e_1 + e_3), \\ J_{21}e_3 = -e_3, \\ \langle \cdot, \cdot \rangle_{21} = \kappa(e^2 \circ e^2 + e^4 \circ e^4 + 2e^2 \circ e^4) - 2e^1 \circ e^4 + 2e^3 \circ e^4 - \frac{1}{\kappa}e^1 \circ e^1, \end{cases}$$

where $\kappa \neq 0$. These correspond to the Lagrangian decompositions

$$\mathfrak{r}_{2}\mathfrak{r}_{2} = \mathfrak{L}_{21} \oplus \mathfrak{L}_{21}' = \operatorname{span}\{e_{4} - e_{2}, e_{1} + e_{3} - \frac{1}{\kappa}e_{2}\} \oplus \operatorname{span}\{e_{3}, e_{2} + \kappa e_{1}\}.$$

Besides, one can easily check that their para-holomorphic sectional curvatures are constantly $H_{21} = \kappa$, so these structures correspond to Theorem 3.1-(*i.b*).

Flat para-Kähler structures

Assuming that $a_{12} = 0$ and $a_{14} = 0$ as in Case 1, if $a_{34} = 0$, then there are three inequivalent flat para-Kähler structures given by

$$(J_{22}, \langle \cdot, \cdot \rangle_{22}) : \begin{cases} J_{22}e_1 = -e_1, \ J_{22}e_2 = e_2 - 2e_4, \ J_{22}e_3 = 2e_1 + e_3, \\ J_{22}e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_{22} = 2(e^1 \circ e^2 - 2e^2 \circ e^3 - e^3 \circ e^4), \end{cases}$$

$$(J_{23}, \langle \cdot, \cdot \rangle_{23}) : \begin{cases} J_{23}e_1 = -e_1, \ J_{23}e_2 = e_2, \ J_{23}e_3 = \varepsilon e_3, \ J_{23}e_4 = -\varepsilon e_4, \\ \langle \cdot, \cdot \rangle_{23} = 2(e^1 \circ e^2 - \varepsilon e^3 \circ e^4), \quad \varepsilon = \pm 1, \end{cases}$$

$$(J_{24}, \langle \cdot, \cdot \rangle_{24}) : \begin{cases} J_{24}e_1 = -e_1 - 2e_3, \ J_{24}e_2 = e_2, \ J_{24}e_3 = e_3, \\ J_{24}e_4 = 2e_2 - e_4, \\ \langle \cdot, \cdot \rangle_{24} = 2(e^1 \circ e^2 + 2e^1 \circ e^4 - e^3 \circ e^4), \end{cases}$$

which correspond to the Lagrangian decompositions $\mathfrak{r}_2\mathfrak{r}_2=\mathfrak{L}_{2i}\oplus\mathfrak{L}'_{2i}$ given by

$$\begin{aligned} \mathbf{r}_{2}\mathbf{r}_{2} &= \mathfrak{L}_{22} \oplus \mathfrak{L}_{22}' = \operatorname{span}\{e_{1} + e_{3}, e_{4} - e_{2}\} \oplus \operatorname{span}\{e_{1}, e_{4}\} \\ &= \mathfrak{L}_{23} \oplus \mathfrak{L}_{23}' = \operatorname{span}\{e_{2}, e_{3}\} \oplus \operatorname{span}\{e_{1}, e_{4}\} \\ &= \mathfrak{L}_{23} \oplus \mathfrak{L}_{23}' = \operatorname{span}\{e_{2}, e_{4}\} \oplus \operatorname{span}\{e_{1}, e_{3}\} \\ &= \mathfrak{L}_{24} \oplus \mathfrak{L}_{24}' = \operatorname{span}\{e_{2}, e_{3}\} \oplus \operatorname{span}\{e_{1} + e_{3}, e_{4} - e_{2}\}. \end{aligned}$$

Para-Kähler and opposite para-Kähler structures

The remaining possibilities corresponding to Case 1 with $a_{34} \neq 0$ and Case 3 give rise to two families of para-Kähler structures determined by

$$(J_{25}, \langle \cdot, \cdot \rangle_{25}) : \begin{cases} J_{25}e_1 = -\frac{1}{\kappa_1}e_2, \ J_{25}e_2 = -\kappa_1e_1, \ J_{25}e_3 = -e_3, \ J_{25}e_4 = e_4, \\ \langle \cdot, \cdot \rangle_{25} = -\frac{1}{\kappa_1}e^1 \circ e^1 + \kappa_1e^2 \circ e^2 + 2e^3 \circ e^4, \end{cases}$$

$$(J_{26}, \langle \cdot, \cdot \rangle_{26}) : \begin{cases} J_{26}e_1 = -\frac{1}{\kappa_1}e_2, \ J_{26}e_2 = -\kappa_1e_1, \ J_{26}e_3 = -\frac{1}{\kappa_2}e_4, \\ J_{26}e_4 = -\kappa_2e_3, \\ \langle \cdot, \cdot \rangle_{26} = -\frac{1}{\kappa_1}e^1 \circ e^1 + \kappa_1e^2 \circ e^2 - \frac{1}{\kappa_2}e^3 \circ e^3 + \kappa_2e^4 \circ e^4, \end{cases}$$

where $\kappa_1 \kappa_2 \neq 0$. The corresponding Lagrangian decompositions are given by

$$\begin{aligned} \mathbf{r}_{2}\mathbf{r}_{2} &= \mathfrak{L}_{25} \oplus \mathfrak{L}_{25}' = \operatorname{span}\{e_{4}, e_{2} - \kappa_{1}e_{1}\} \oplus \operatorname{span}\{e_{3}, e_{2} + \kappa_{1}e_{1}\} \\ &= \mathfrak{L}_{26} \oplus \mathfrak{L}_{26}' = \operatorname{span}\{e_{4} - \kappa_{2}e_{3}, e_{2} - \kappa_{1}e_{1}\} \oplus \operatorname{span}\{e_{4} + \kappa_{2}e_{3}, e_{2} + \kappa_{1}e_{1}\}. \end{aligned}$$

Now, it is straightforward to see that both metrics are locally symmetric, so they are covered by Theorem 3.1-(*i.d*). Their Ricci operators are diagonalizable with respect to the basis $\{e_1, e_2, e_3, e_4\}$ with Ricci curvatures

$$\operatorname{Ric}_{25} = \operatorname{diag}[\kappa_1, \kappa_1, 0, 0]$$
 and $\operatorname{Ric}_{26} = \operatorname{diag}[\kappa_1, \kappa_1, \kappa_2, \kappa_2].$

Therefore, the underlying pseudo-Riemannian manifolds split locally as a product of two Lorentzian surfaces of constant Gaussian curvature, namely, $N_1(\kappa_1) \times \mathbb{L}^2$ or $N_1(\kappa_1) \times N_2(\kappa_2)$, where \mathbb{L}^2 denotes the Minkowskian plane. Thus they also admit opposite para-Kähler structures. Furthermore, the metrics $\langle \cdot, \cdot \rangle_{26}$ are Einstein if $\kappa_1 = \kappa_2$ and locally conformally flat if $\kappa_1 = -\kappa_2$, in which case they correspond to the metrics in Corollary 3.2-(*iii*).

3.6 Para-Kähler structures on \mathfrak{rh}_3

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra \mathfrak{rh}_3 determined by

$$[e_1, e_2] = e_3,$$

which corresponds to the Lie algebra of either $H^3 \times \mathbb{R}$ or the semi-direct extension $\mathbb{R}e_1 \ltimes_{\varphi} \mathbb{R}^3$ given by

$$\varphi = \operatorname{ad}(e_1): \quad \varphi(e_2) = e_3, \quad \varphi(e_3) = 0, \quad \varphi(e_4) = 0.$$

Symplectic forms on \mathfrak{rh}_3 are given by

$$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{23}e^{23} + \alpha_{24}e^{24},$$
$$\alpha_{14}\alpha_{23} - \alpha_{13}\alpha_{24} \neq 0,$$

as shown in [115]. All these symplectic structures $(\mathfrak{rh}_3, \omega)$ are symplectomorphically equivalent to $\Omega = e^{14} + e^{23}$ through an automorphism of the form

$$\Phi = \begin{pmatrix} z_{11} & z_{12} & 0 & 0 \\ z_{21} & z_{22} & 0 & 0 \\ z_{31} & z_{32} & z_{11}z_{22} - z_{12}z_{21} & z_{34} \\ z_{41} & z_{42} & 0 & z_{44} \end{pmatrix}, \quad \text{with} \quad (z_{11}z_{22} - z_{12}z_{21})z_{44} \neq 0.$$

The ones that preserve the symplectic structure $(\mathfrak{rh}_3, \Omega)$ determined by $z_{11}z_{22}^2 = 1$, $z_{12}z_{22} = -z_{34}$, $z_{21} = 0$, $z_{42} = z_{22}^3 z_{31} - z_{22} z_{34} z_{41}$ and $z_{44} = z_{22}^2$. A long but straightforward calculation

shows that any para-Kähler structure on \mathfrak{rh}_3 must be flat and equivalent to one of the following two para-Kähler structures

$$(J_1, \langle \cdot, \cdot \rangle_1) : \begin{cases} J_1 e_1 = e_1, & J_1 e_2 = -e_2, & J_1 e_3 = e_3, & J_1 e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_1 = 2(-e^1 \circ e^4 + e^2 \circ e^3), \\ (J_2, \langle \cdot, \cdot \rangle_2) : \begin{cases} J_2 e_1 = -e_2, & J_2 e_2 = -e_1, & J_2 e_3 = e_4, & J_2 e_4 = e_3, \\ \langle \cdot, \cdot \rangle_2 = 2(e^1 \circ e^3 + e^2 \circ e^4). \end{cases}$$

In each case, rh₃ decomposes as a direct sum of Lagrangian subalgebras as

$$\mathfrak{rh}_3 = \mathfrak{L}_1 \oplus \mathfrak{L}'_1 = \operatorname{span}\{e_1, e_3\} \oplus \operatorname{span}\{e_2, e_4\}$$
$$= \mathfrak{L}_2 \oplus \mathfrak{L}'_2 = \operatorname{span}\{e_3 + e_4, e_2 - e_1\} \oplus \operatorname{span}\{e_1 + e_2, e_4 - e_3\}.$$

3.7 Para-Kähler structures on $rr_{3,0}$

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra $\mathfrak{rr}_{3,0}$ determined by

$$[e_1, e_2] = e_2,$$

which corresponds to the Lie algebra of $E(1,1) \times \mathbb{R}$ or to the semi-direct extension $\mathbb{R}e_1 \ltimes_{\varphi} \mathbb{R}^3$ given by

$$\varphi = \operatorname{ad}(e_1): \quad \varphi(e_2) = e_2, \quad \varphi(e_3) = 0, \quad \varphi(e_4) = 0.$$

The symplectic structures on $\mathfrak{rr}_{3,0}$ are given by

$$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{34}e^{34},$$
$$\alpha_{12}\alpha_{34} \neq 0,$$

as shown in [115]. All these symplectic structures $(\mathfrak{rr}_{3,0},\omega)$ are symplectomorphically equivalent to $\Omega = e^{12} + e^{34}$ through an automorphism of the form

$$\Phi = \begin{pmatrix} 1 & 0 & 0 & 0 \\ z_{21} & z_{22} & 0 & 0 \\ z_{31} & 0 & z_{33} & z_{34} \\ z_{41} & 0 & z_{43} & z_{44} \end{pmatrix}, \quad \text{with} \quad (z_{33}z_{44} - z_{34}z_{43})z_{22} \neq 0.$$

Moreover, the automorphisms preserving the symplectic structure $(\mathfrak{rr}_{3,0},\Omega)$ are the ones determined by Φ above satisfying

$$\Phi_1 : z_{22} = 1, \quad z_{31} = 0, \quad z_{41} = 0, \quad z_{44}z_{33} = 1 + z_{34}z_{43} \quad \text{with } z_{33} \neq 0,$$

$$\Phi_2 : z_{22} = 1, \quad z_{31} = 0, \quad z_{41} = 0, \quad z_{33} = 0, \quad z_{43}z_{34} = -1.$$

Proceeding as in the previous cases, a straightforward calculation shows that any para-Kähler structure on $(\mathfrak{rr}_{3,0}, \Omega)$ is equivalent to one of the following two families

$$(J_1, \langle \cdot, \cdot \rangle_1) : \begin{cases} J_1 e_1 = \varepsilon \, e_1, & J_1 e_2 = -\varepsilon \, e_2, & J_1 e_3 = e_3, & J_1 e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_1 = -2(\varepsilon \, e^1 \circ e^2 + e^3 \circ e^4), & \varepsilon = \pm 1, \end{cases}$$

$$(J_2, \langle \cdot, \cdot \rangle_2) : \begin{cases} J_2 e_1 = \frac{1}{\kappa} e_2, & J_2 e_2 = \kappa e_1, & J_2 e_3 = e_3, & J_2 e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_2 = \frac{1}{\kappa} e^1 \circ e^1 - \kappa e^2 \circ e^2 - 2e^3 \circ e^4, & \kappa \neq 0. \end{cases}$$

These correspond to the Lagrangian decompositions

$$\begin{aligned} \mathfrak{rr}_{3,0} &= \mathfrak{L}_1 \oplus \mathfrak{L}_1' = \operatorname{span}\{e_1, e_3\} \oplus \operatorname{span}\{e_2, e_4\} & (\varepsilon = 1) \\ &= \mathfrak{L}_1 \oplus \mathfrak{L}_1' = \operatorname{span}\{e_2, e_3\} \oplus \operatorname{span}\{e_1, e_4\} & (\varepsilon = -1) \\ &= \mathfrak{L}_2 \oplus \mathfrak{L}_2' = \operatorname{span}\{e_2 + \kappa e_1, e_3\} \oplus \operatorname{span}\{e_2 - \kappa e_1, e_4\}. \end{aligned}$$

Both structures $(J_1, \langle \cdot, \cdot \rangle_1)$, with $\varepsilon = \pm 1$, are flat, while all the structures $(J_2, \langle \cdot, \cdot \rangle_2)$ are locally symmetric with diagonalizable Ricci operator

$$\operatorname{Ric}_2 = -\operatorname{diag}[\kappa, \kappa, 0, 0].$$

Thus, the underlying pseudo-Riemannian manifolds split locally as products of the form $N \times \mathbb{L}^2$, where N is a Lorentzian surface of constant Gaussian curvature $K_N = -\kappa$ and \mathbb{L}^2 . Therefore, $(J_2, \langle \cdot, \cdot \rangle_2)$ are para-Kähler and opposite para-Kähler structures, and correspond to the metrics in Theorem 3.1-(*i.d*)

3.8 Para-Kähler structures on $rr_{3,-1}$

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra $\mathfrak{rr}_{3,-1}$ determined by

$$[e_1, e_2] = e_2, \qquad [e_1, e_3] = -e_3.$$

which corresponds to a unimodular semi-direct extension of E(1, 1), given by the Lie algebra of $\mathbb{R} \times E(1, 1)$ as in Lemma 1.9, and can also be seen as the semi-direct extension $\mathbb{R}e_1 \ltimes_{\varphi} \mathbb{R}^3$ given by

$$\varphi = \operatorname{ad}(e_1): \quad \varphi(e_2) = e_2, \quad \varphi(e_3) = -e_3, \quad \varphi(e_4) = 0.$$

The symplectic structures on $rr_{3,-1}$ are given by

$$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{23}e^{23}, \qquad \alpha_{14}\alpha_{23} \neq 0,$$

as shown in [115]. All these symplectic structures $(\mathfrak{rr}_{3,-1},\omega)$ are symplectomorphically equivalent to $\Omega = e^{14} + e^{23}$ through an automorphism of the form

$$\Phi_{1} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ z_{21} & z_{22} & 0 & 0 \\ z_{31} & 0 & z_{33} & 0 \\ z_{41} & 0 & 0 & z_{44} \end{pmatrix} \quad \text{or} \quad \Phi_{2} = \begin{pmatrix} -1 & 0 & 0 & 0 \\ z_{21} & 0 & z_{23} & 0 \\ z_{31} & z_{32} & 0 & 0 \\ z_{41} & 0 & 0 & z_{44} \end{pmatrix}$$

with $z_{22}z_{33}z_{44} \neq 0$ in the first case and $z_{23}z_{32}z_{44} \neq 0$ in the second one. Moreover, the automorphisms preserving the symplectic structure $(\mathfrak{rr}_{3,-1}, \Omega)$ are given by

$$\Phi_1 : z_{21} = 0, \quad z_{31} = 0, \quad z_{33}z_{22} = 1, \quad z_{44} = 1,$$

$$\Phi_2 : z_{21} = 0, \quad z_{31} = 0, \quad z_{32}z_{23} = -1, \quad z_{44} = -1$$

Proceeding as in the previous sections we see that para-Kähler structure on $(\mathfrak{rr}_{3,-1}, \Omega)$ corresponds to one of the following possibilities.

Flat para-Kähler structures on $rr_{3,-1}$

Assuming that $a_{14} \neq 0$, the corresponding para-Kähler structures are all flat and equivalent to

$$(J_1, \langle \cdot, \cdot \rangle_1) : \begin{cases} J_1 e_1 = \frac{1}{\kappa} e_4, & J_1 e_2 = -e_2, & J_1 e_3 = e_3, & J_1 e_4 = \kappa e_1, \\ \langle \cdot, \cdot \rangle_1 = \frac{1}{\kappa} e^1 \circ e^1 + 2e^2 \circ e^3 - \kappa e^4 \circ e^4, & \kappa \neq 0. \end{cases}$$

These structures correspond to the decompositions

$$\mathfrak{rr}_{3,-1} = \mathfrak{L}_1 \oplus \mathfrak{L}'_1 = \operatorname{span}\{e_4 + \kappa e_1, e_3\} \oplus \operatorname{span}\{e_4 - \kappa e_1, e_2\}$$

of $(\mathfrak{rr}_{3,-1},\Omega)$ as direct sums of Lagrangian subalgebras.

Ricci-flat para-Kähler structures on $\mathfrak{rr}_{3,-1}$

If the component of the para-complex structure $a_{14} = 0$, any para-Kähler structure is equivalent to

$$(J_2, \langle \cdot, \cdot \rangle_2) : \begin{cases} J_2 e_1 = -e_1, & J_2 e_2 = -e_2, & J_2 e_3 = e_3 - \kappa e_2, & J_2 e_4 = e_4, \\ \langle \cdot, \cdot \rangle_2 = 2(e^1 \circ e^4 + e^2 \circ e^3) + \kappa e^3 \circ e^3, & \kappa = 0, \pm 1. \end{cases}$$

The symplectic Lie algebra $(\mathfrak{rr}_{3,-1}, \Omega)$ decomposes as a direct sum of Lagrangian subalgebras as $\mathfrak{rr}_{3,-1} = \mathfrak{L}_2 \oplus \mathfrak{L}'_2 = \operatorname{span}\{e_4, e_3 - \frac{\kappa}{2}e_2\} \oplus \operatorname{span}\{e_1, e_2\}.$

The para-Kähler structures $(J_2, \langle \cdot, \cdot \rangle_2)$ are flat if $\kappa = 0$ and Ricci-flat otherwise, so they are covered by Theorem 3.1-(*i.a*). Moreover, if $\kappa \neq 0$, then the metric has recurrent curvature with

recurrence one-form $\xi = 2e^1$ (i.e., $\nabla_2 R_2 = 2e^1 \otimes R_2$), and the corresponding curvature operator acting on the space of two-forms is given by

$$\mathcal{R}_2(e^1 \wedge e^3) = 2 \kappa e^2 \wedge e^4 = R_2(e_1, e_3, e_3, e_1)e^2 \wedge e^4.$$

Therefore, they are special recurrent and simply harmonic manifolds modelled on (3.1). Hence, their curvature tensor is semi-symmetric. Furthermore, the structures given by $(J_2, \langle \cdot, \cdot \rangle_2)$ with $\kappa \neq 0$ admit opposite almost para-Kähler structures compatible with the opposite symplectic form $\Omega'_2 = e^{14} - e^{23} + \mu e^{13}$ with $\mu \in \mathbb{R}$.

3.9 Para-Kähler structures on \mathfrak{r}'_2

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the symplectic Lie algebra \mathfrak{r}'_2 determined by

$$[e_1, e_3] = e_3, \quad [e_1, e_4] = e_4, \quad [e_2, e_3] = e_4, \quad [e_2, e_4] = -e_3$$

which, according to [6], corresponds to the non-unimodular semi-direct extension $\tilde{E}(2) \rtimes \mathbb{R}$ isomorphic to $\mathfrak{aff}(\mathbb{C}) \times \mathfrak{aff}(\mathbb{C})$ as in Lemma 1.11.

It was shown in [115] that the symplectic forms on r'_2 are given by

$$\omega = \alpha_{12}e^{12} + \alpha_{13}(e^{13} - e^{24}) + \alpha_{14}(e^{14} + e^{23}), \qquad \alpha_{13}^2 + \alpha_{14}^2 \neq 0$$

The automorphisms of \mathfrak{r}'_2 are of the form

$$\Phi_{1} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ z_{31} & z_{32} & z_{33} & z_{34} \\ -z_{32} & z_{31} & -z_{34} & z_{33} \end{pmatrix} \quad \text{or} \quad \Phi_{2} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & -1 & 0 & 0 \\ z_{31} & z_{32} & z_{33} & z_{34} \\ z_{32} & -z_{31} & z_{34} & -z_{33} \end{pmatrix},$$

with $z_{33}^2 + z_{34}^2 \neq 0$, from where it follows that any symplectic form is symplectomorphically equivalent to $\Omega_{\lambda} = \lambda e^{12} + e^{14} + e^{23}$, for some $\lambda \in \mathbb{R}$. We emphasize that even though there exist Kähler structures with associated symplectic form Ω_{λ} for $\lambda \neq 0$ (see [116]), a straightforward calculation shows that Ω_{λ} does not support any para-Kähler structure for $\lambda \neq 0$. On the other hand, there exist para-Kähler structures whose associated symplectic structure is $\Omega_0 = e^{14} + e^{23}$. In order to describe these structures, we make use of the automorphisms preserving the symplectic structure $(\mathfrak{r}'_2, \Omega_0)$, which are given by Φ_1 (resp., Φ_2) above with $z_{33} = 1$ and $z_{34} = 0$ (resp., $z_{33} = -1$ and $z_{34} = 0$). A long but straightforward calculation shows that any para-Kähler structure on $(\mathfrak{r}'_2, \Omega_0)$ corresponds to one of the different situations given by $a_{13} \neq 0$ or $a_{13} = 0$. While the Ricci operator is diagonalizable in the case $a_{13} = 0$, this does not happen when $a_{13} \neq 0$. If $a_{13} = 0$, there are three essentially different cases depending on whether the component of the para-complex structure $a_{14} = 0$ or $a_{14} \neq 0$.

Para-Kähler structures on r'_2 with complex Ricci operator

Assuming the component of the para-complex structure $a_{13} \neq 0$, then para-Kähler structures are equivalent to $(J_1, \langle \cdot, \cdot \rangle_1)$ given by

$$\begin{cases} J_1 e_1 = \widehat{\beta} e_3 + \widehat{\alpha} e_4, \ J_1 e_2 = -\widehat{\alpha} e_3 + \widehat{\beta} e_4, \ J_1 e_3 = \beta e_1 - \alpha e_2, \ J_1 e_4 = \alpha e_1 + \beta e_2, \\ \langle \cdot, \cdot \rangle_1 = \widehat{\alpha} (e^1 \circ e^1 - e^2 \circ e^2) + \alpha (e^3 \circ e^3 - e^4 \circ e^4) + 2(\widehat{\beta} e^1 \circ e^2 - \beta e^3 \circ e^4), \end{cases}$$

where $\widehat{\beta} = \frac{\beta}{\beta^2 + \alpha^2}$ and $\widehat{\alpha} = \frac{\alpha}{\beta^2 + \alpha^2}$, $\alpha, \beta \in \mathbb{R}$, and $\beta \neq 0$. In this situation, \mathfrak{r}'_2 decomposes into direct sums of Lagrangian subalgebras $\mathfrak{r}'_2 = \mathfrak{L}_1 \oplus \mathfrak{L}'_1$ where

$$\mathfrak{L}_{1} = \operatorname{span}\{e_{4} + \alpha e_{1} + \beta e_{2}, e_{3} + \beta e_{1} - \alpha e_{2}\}, \\
\mathfrak{L}_{1}' = \operatorname{span}\{e_{4} - \alpha e_{1} - \beta e_{2}, e_{3} - \beta e_{1} + \alpha e_{2}\}.$$

The Ricci operators associated to the metrics above are complex diagonalizable with eigenvalues $-2(\alpha \pm \beta \sqrt{-1})$ and the underlying pseudo-Riemannian structures are locally symmetric. Furthermore, their self-dual and anti-self-dual Weyl curvature operators are diagonalizable with the same eigenvalues

$$W^{\pm} = \operatorname{diag}\left[-\frac{4}{3}\alpha, \frac{2}{3}\alpha, \frac{2}{3}\alpha\right].$$

Therefore, they correspond to the metrics in Theorem 3.1-(*ii.a*) and the metrics are locally conformally flat if and only if $\alpha = 0$, as in Corollary 3.2-(*i*).

Remark 3.9. The Ricci tensor ρ_1 defines another para-Kähler structure $(\mathfrak{r}'_2, J_1, \rho_1)$ with the same Levi-Civita connection as $(\mathfrak{r}'_2, J_1, \langle \cdot, \cdot \rangle_1)$. Straightforward calculations show that $(\mathfrak{r}'_2, J_1, \rho_1)$ is an Einstein manifold of non-constant para-holomorphic sectional curvature so, after a normalization of the associated symplectic structure, it corresponds to one of the para-Kähler structures $(J_4, \langle \cdot, \cdot \rangle_4)$ described below.

Flat para-Kähler structures on \mathfrak{r}'_2

If $a_{13} = 0$ and $a_{14} = 0$, proceeding as in the previous sections, it is easy to see that any para-Kähler structure is equivalent to the flat structure given by

$$(J_2, \langle \cdot, \cdot \rangle_2) : \begin{cases} J_2 e_1 = e_1, \quad J_2 e_2 = e_2, \quad J_2 e_3 = -e_3, \quad J_2 e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_2 = -2(e^1 \circ e^4 + e^2 \circ e^3), \end{cases}$$

so that $\mathfrak{r}'_2 = \mathfrak{L}_2 \oplus \mathfrak{L}'_2 = \operatorname{span}\{e_1, e_2\} \oplus \operatorname{span}\{e_3, e_4\}.$

Einstein para-Kähler structures on \mathfrak{r}'_2 of non-zero curvature

Assume that $a_{13} = 0$ and $a_{14} \neq 0$. Then any para-Kähler structure is equivalent to another one in one of the following two families.

$$(J_3, \langle \cdot, \cdot \rangle_3) : \begin{cases} J_3 e_1 = -\frac{1}{\kappa} e_4, & J_3 e_2 = \frac{2}{\kappa} e_3 - e_2, & J_3 e_3 = e_3, & J_3 e_4 = -\kappa e_1, \\ \langle \cdot, \cdot \rangle_3 = -\frac{1}{\kappa} (e^1 \circ e^1 - 2e^2 \circ e^2) + 2e^2 \circ e^3 + \kappa e^4 \circ e^4, & \kappa \neq 0, \end{cases} \\ (J_4, \langle \cdot, \cdot \rangle_4) : \begin{cases} J_4 e_1 = -\frac{1}{\kappa} e_4, & J_4 e_2 = \frac{1}{\kappa} e_3, & J_4 e_3 = \kappa e_2, & J_4 e_4 = -\kappa e_1, \\ \langle \cdot, \cdot \rangle_4 = -\frac{1}{\kappa} (e^1 \circ e^1 - e^2 \circ e^2) - \kappa (e^3 \circ e^3 - e^4 \circ e^4), & \kappa \neq 0. \end{cases}$$

These give decompositions of the symplectic Lie algebra $(\mathfrak{r}'_2, \Omega_0)$ as direct sums of Lagrangian subalgebras as

$$\mathbf{\mathfrak{r}}_2' = \mathfrak{L}_3 \oplus \mathfrak{L}_3' = \operatorname{span}\{e_3, e_4 - \kappa e_1\} \oplus \operatorname{span}\{e_3 - \kappa e_2, e_4 + \kappa e_1\}$$
$$= \mathfrak{L}_4 \oplus \mathfrak{L}_4' = \operatorname{span}\{e_3 + \kappa e_2, e_4 - \kappa e_1\} \oplus \operatorname{span}\{e_4 + \kappa e_1, e_3 - \kappa e_2\}.$$

It is easy to see that the structures $(J_3, \langle \cdot, \cdot \rangle_3)$ have non-zero constant para-holomorphic sectional curvature $H_3 = \kappa$ – thus corresponding to those in Theorem 3.1-(*i.b*) – while the structures $(J_4, \langle \cdot, \cdot \rangle_4)$ are Einstein with $\operatorname{Ric}_4 = 2\kappa \operatorname{Id}(\neq 0)$ and their self-dual and anti-self-dual Weyl curvature operators are diagonalizable with the same eigenvalues – which implies that the paraholomorphic sectional curvature is not constant and they are covered by Theorem 3.1-(*i.c*).

A straightforward calculation shows that the structures $(J_4, \langle \cdot, \cdot \rangle_4)$ have a compatible anti-Kähler structure (see [21]) defined by a complex structure \mathfrak{J} given by $\mathfrak{J}e_1 = e_2$, $\mathfrak{J}e_3 = e_4$, which commutes with the para-complex structure J_4 . Hence, the structure $(\langle \cdot, \cdot \rangle^*, J_4)$ is also a locally symmetric para-Kähler structure, where $\langle X, Y \rangle^* = \langle \mathfrak{J}X, Y \rangle_4$ is the twin metric of $\langle \cdot, \cdot \rangle_4$. A straightforward calculation shows that the Ricci operator Ric^{*} has complex eigenvalues $\pm 2\kappa \sqrt{-1}$, so this metric corresponds to a metric in Section 3.9 (up to renormalization of the corresponding symplectic structure).

3.10 Para-Kähler structures on $r_{4,-1,\beta}$

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra $\mathfrak{r}_{4,-1,\beta}$ determined by

$$[e_1, e_4] = -e_1, \quad [e_2, e_4] = e_2, \quad [e_3, e_4] = -\beta e_3,$$

with $-1 < \beta < 0$. This Lie algebra corresponds to the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} \mathbb{R}^3$ given by

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = e_1, \quad \varphi(e_2) = -e_2, \quad \varphi(e_3) = \beta e_3.$$

Any symplectic form on this Lie algebra must take the form

$$\omega = \alpha_{12}e^{12} + \alpha_{14}e^{14} + \alpha_{24}e^{24} + \alpha_{34}e^{34}, \qquad \alpha_{12}\alpha_{34} \neq 0,$$

as shown in [115]. These symplectic forms are symplectomorphically equivalent to $\Omega = e^{12} + e^{34}$ through an automorphisms of $\mathfrak{r}_{4,-1,\beta}$, which are given by

$$\Phi = \begin{pmatrix} z_{11} & 0 & 0 & z_{14} \\ 0 & z_{22} & 0 & z_{24} \\ 0 & 0 & z_{33} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix} \quad \text{with} \quad z_{11}z_{22}z_{33} \neq 0.$$

The ones that preserve the symplectic structure $(\mathfrak{r}_{4,-1,\beta},\Omega)$ are those which satisfy $z_{14} = z_{24} = 0$, $z_{33} = 1$ and $z_{22}z_{11} = 1$. The para-Kähler structures on $(\mathfrak{r}_{4,-1,\beta},\Omega)$ split into two different classes depending on the value of the coefficient a_{43} of the para-complex structure as follows.

Ricci-flat para-Kähler structures on $\mathfrak{r}_{4,-1,\beta}$

Assuming that $a_{43} = 0$, the corresponding para-Kähler structures are equivalent to one of the following two families.

$$(J_1, \langle \cdot, \cdot \rangle_1) : \begin{cases} J_1 e_1 = -e_1 + \kappa e_2, & J_1 e_2 = e_2, & J_1 e_3 = -e_3, & J_1 e_4 = e_4, \\ \langle \cdot, \cdot \rangle_1 = \kappa e^1 \circ e^1 + 2(e^1 \circ e^2 + e^3 \circ e^4), & \kappa = 0, \pm 1, \end{cases}$$

$$(J_2, \langle \cdot, \cdot \rangle_2) : \begin{cases} J_2 e_1 = -e_1, & J_2 e_2 = -\kappa e_1 + e_2, & J_2 e_3 = e_3, & J_2 e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_2 = \kappa e^2 \circ e^2 + 2(e^1 \circ e^2 - e^3 \circ e^4), & \kappa = 0, \pm 1. \end{cases}$$

These two families of para-Kähler structures induce the Lagrangian decompositions

$$\begin{aligned} \mathbf{\mathfrak{r}}_{4,-1,\beta} &= \mathfrak{L}_1 \oplus \mathfrak{L}_1' = \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_3, e_2 - \frac{2}{\kappa}e_1\} \quad (\kappa \neq 0) \\ &= \mathfrak{L}_1 \oplus \mathfrak{L}_1' = \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_1, e_3\} \quad (\kappa = 0) \\ &= \mathfrak{L}_2 \oplus \mathfrak{L}_2' = \operatorname{span}\{e_3, e_2 - \frac{\kappa}{2}e_1\} \oplus \operatorname{span}\{e_1, e_4\}. \end{aligned}$$

The structures above are flat if $\kappa = 0$. Otherwise, they are Ricci-flat and recurrent with recurrence one-forms $\xi_1 = 2(\beta - 1)e^4$ and $\xi_2 = 2(\beta + 1)e^4$, respectively, and their curvature operators acting on the space of two-forms are given by

$$\mathcal{R}_1(e^1 \wedge e^4) = \kappa(\beta - 2)e^2 \wedge e^3 = R_1(e_1, e_4, e_1, e_4)e^2 \wedge e^3,$$

$$\mathcal{R}_2(e^2 \wedge e^4) = \kappa(\beta + 2)e^1 \wedge e^3 = R_2(e_2, e_4, e_4, e_2)e^1 \wedge e^3.$$

Therefore, the metrics above are simply harmonic and special recurrent, so they are modelled on (3.1). Hence, their curvature tensors are semi-symmetric and so they are covered by Theorem 3.6. Finally, both $(J_1, \langle \cdot, \cdot \rangle_1)$ and $(J_2, \langle \cdot, \cdot \rangle_2)$ admit opposite almost para-Kähler structures that are compatible with the opposite symplectic forms $\Omega'_1 = e^{12} - e^{34} + \mu e^{14}$ and $\Omega'_2 = e^{12} - e^{34} + \mu e^{24}$, respectively, where $\mu \in \mathbb{R}$.

Para-Kähler and opposite para-Kähler structures on $\mathfrak{r}_{4,-1,\beta}$

If $a_{43} \neq 0$, then any para-Kähler structure is equivalent to

$$(J_3, \langle \cdot, \cdot \rangle_3) : \begin{cases} J_3 e_1 = -e_1, \quad J_3 e_2 = e_2, \quad J_3 e_3 = \kappa e_4, \quad J_3 e_4 = \frac{1}{\kappa} e_3, \\ \langle \cdot, \cdot \rangle_3 = 2e^1 \circ e^2 + \kappa e^3 \circ e^3 - \frac{1}{\kappa} e^4 \circ e^4, \quad \kappa \neq 0, \end{cases}$$

so that $\mathfrak{r}_{4,-1,\beta} = \mathfrak{L}_3 \oplus \mathfrak{L}'_3 = \operatorname{span}\{e_2, e_4 + \frac{1}{\kappa}e_3\} \oplus \operatorname{span}\{e_1, e_4 - \frac{1}{\kappa}e_3\}.$

The metrics above are locally symmetric with diagonalizable Ricci operator

$$\operatorname{Ric}_3 = \operatorname{diag}[0, 0, \kappa\beta^2, \kappa\beta^2]$$

Therefore, the underlying manifolds are locally isometric to a product $\mathbb{L}^2 \times N$ of the Minkowskian plane and a Lorentzian surface of constant sectional curvature

$$K_N = \kappa \beta^2$$
.

Consequently, they are para-Kähler and opposite para-Kähler and they correspond to the metrics in Theorem 3.1-(*i.d*).

3.11 Para-Kähler structures on $r_{4,-1,-1}$

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra $\mathfrak{r}_{4,-1,-1}$ determined by

$$[e_1, e_4] = -e_1, \quad [e_2, e_4] = e_2, \quad [e_3, e_4] = e_3.$$

This Lie algebra corresponds to the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} \mathbb{R}^3$ given by

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = e_1, \quad \varphi(e_2) = -e_2, \quad \varphi(e_3) = -e_3$$

Any symplectic structure on $r_{4,-1,-1}$ is of the form

$$\omega = \alpha_{12}e^{12} + \alpha_{13}e^{13} + \alpha_{14}e^{14} + \alpha_{24}e^{24} + \alpha_{34}e^{34}, \quad \alpha_{13}\alpha_{24} - \alpha_{12}\alpha_{34} \neq 0,$$

and all of them are symplectomorphically equivalent to $\Omega = e^{12} + e^{34}$ through an automorphism of $\mathfrak{r}_{4,-1,-1}$, which are given by

$$\Phi = \begin{pmatrix} z_{11} & 0 & 0 & z_{14} \\ 0 & z_{22} & z_{23} & z_{24} \\ 0 & z_{32} & z_{33} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix} \quad \text{with} \quad z_{11}(z_{22}z_{33} - z_{23}z_{32}) \neq 0.$$

Moreover, the automorphisms preserving the symplectic structure $(\mathfrak{r}_{4,-1,-1},\Omega)$ are the ones above with $z_{23} = z_{24} = 0$, $z_{33} = 1$, $z_{32} = z_{14}z_{22}$ and $z_{22}z_{11} = 1$. The associated para-Kähler structures split into two cases depending on whether $a_{43} \neq 0$ or $a_{43} = 0$ as follows.

Para-Kähler and opposite para-Kähler structures on $r_{4,-1,-1}$

Straightforward calculations as in the previous sections show that any para-Kähler structure with $a_{43} \neq 0$ is equivalent to

$$(J_1, \langle \cdot, \cdot \rangle_1) : \begin{cases} J_1 e_1 = -e_1, \quad J_1 e_2 = e_2, \quad J_1 e_3 = \kappa e_4, \quad J_1 e_4 = \frac{1}{\kappa} e_3, \\ \langle \cdot, \cdot \rangle_1 = 2e^1 \circ e^2 + \kappa e^3 \circ e^3 - \frac{1}{\kappa} e^4 \circ e^4, \quad \kappa \neq 0, \end{cases}$$

so that $\mathfrak{r}_{4,-1,-1} = \mathfrak{L}_1 \oplus \mathfrak{L}'_1 = \operatorname{span}\{\frac{1}{\kappa}e_3 + e_4, e_2\} \oplus \operatorname{span}\{e_1, e_4 - \frac{1}{\kappa}e_3\}.$

It is not difficult to check that these structures are locally symmetric and their Ricci operators are diagonalizable

$$\operatorname{Ric}_1 = \operatorname{diag}[0, 0, \kappa, \kappa].$$

Therefore, the underlying pseudo-Riemannian structures split locally as a product $\mathbb{L}^2 \times N$ of a Lorentzian surface N of constant Gaussian curvature $K_N = \kappa$ and the Minkowskian plane. As a consequence, they also admit opposite para-Kähler structures. These structures correspond to those in Theorem 3.1-(*i.d*).

Ricci-flat para-Kähler structures on $r_{4,-1,-1}$

Assuming that $a_{43} = 0$, the para-Kähler structures on $(\mathfrak{r}_{4,-1,-1}, \Omega)$ are equivalent to one of the following families, which correspond to different geometric situations.

$$(J_2, \langle \cdot, \cdot \rangle_2) : \begin{cases} J_2 e_1 = -e_1, \quad J_2 e_2 = -\kappa e_1 + e_2, \quad J_2 e_3 = e_3, \quad J_2 e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_2 = \kappa e^2 \circ e^2 + 2(e^1 \circ e^2 - e^3 \circ e^4), \quad \kappa = 0, \pm 1, \end{cases}$$

so that $\mathfrak{r}_{4,-1,-1} = \mathfrak{L}_2 \oplus \mathfrak{L}'_2 = \operatorname{span}\{e_3, e_2 - \frac{\kappa}{2}e_1\} \oplus \operatorname{span}\{e_4, e_1\},$

$$(J_3, \langle \cdot, \cdot \rangle_3) : \begin{cases} J_3 e_1 = -e_1 + \kappa e_2, \quad J_3 e_2 = e_2, \quad J_3 e_3 = -e_3, \quad J_3 e_4 = e_4, \\ \langle \cdot, \cdot \rangle_3 = \kappa e^1 \circ e^1 + 2(e^1 \circ e^2 + e^3 \circ e^4), \quad \kappa = 0, \pm 1, \end{cases}$$

so that the Lie algebra decomposes as

$$\begin{aligned} \mathbf{\mathfrak{r}}_{4,-1,-1} &= \mathbf{\mathfrak{L}}_3 \oplus \mathbf{\mathfrak{L}}_3' = \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_3, e_2 - \frac{2}{\kappa}e_1\} \quad (\kappa \neq 0) \\ &= \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_3, e_1\} \qquad (\kappa = 0), \end{aligned}$$

or

$$(J_4, \langle \cdot, \cdot \rangle_4) : \begin{cases} J_4 e_1 = \kappa e_2 + e_4, \quad J_4 e_2 = e_3, \quad J_4 e_3 = e_2, \quad J_4 e_4 = e_1 - \kappa e_3, \\ \langle \cdot, \cdot \rangle_4 = \kappa (e^1 \circ e^1 + e^4 \circ e^4) + 2(e^1 \circ e^3 - e^2 \circ e^4), \quad \kappa \in \mathbb{R}, \end{cases}$$

so that $(r_{4,-1,-1}, \Omega)$ decomposes as direct sums of Lagrangian subalgebras

$$\mathfrak{r}_{4,-1,-1} = \mathfrak{L}_4 \oplus \mathfrak{L}'_4 = \operatorname{span}\{e_2 + e_3, e_1 + e_4 + \kappa e_2\} \oplus \operatorname{span}\{e_3 - e_2, e_4 - e_1 + \kappa e_2\}.$$

The structures $(J_2, \langle \cdot, \cdot \rangle_2)$ are flat if $\kappa = 0$. Otherwise, they are Ricci-flat and locally symmetric, thus locally modelled on (3.1) with $\Psi(x^1, x^2) = \pm (x^2)^2$. Moreover, these structures admit opposite almost para-Kähler structures compatible with the opposite symplectic two-forms $\Omega'_2 = e^{12} - e^{34} + \mu e^{24}$ with $\mu \in \mathbb{R}$. These structures correspond to Theorem 3.1-(*i.a*).

The structures $(J_3, \langle \cdot, \cdot \rangle_3)$ are flat if $\kappa = 0$ and Ricci-flat otherwise. They are special recurrent with recurrence one-form $\xi_3 = -4e^4$ and curvature operator determined by

$$\mathcal{R}_3(e^1 \wedge e^4) = -3\kappa e^2 \wedge e^3 = R_3(e_1, e_4, e_1, e_4)e^2 \wedge e^3$$

Therefore, these structures are simply harmonic and locally modelled on (3.1), and their curvature tensor is semi-symmetric. Furthermore, they admit opposite almost para-Kähler structures compatible with the opposite symplectic two-forms

$$\Omega'_3 = e^{12} - e^{34} + \mu e^{14}, \quad \mu \in \mathbb{R}.$$

These structures also have an associated one-parameter family of hypersymplectic structures $(J_3, J_\delta, \langle \cdot, \cdot \rangle_3)$ given by the Kähler structures

$$J_{\delta}e_1 = -\frac{\kappa\delta}{2}e_3 - \frac{1}{\delta}e_4, \quad J_{\delta}e_2 = -\delta e_3, \quad J_{\delta}e_3 = \frac{1}{\delta}e_2, \quad J_{\delta}e_4 = \delta e_1 - \frac{\kappa\delta}{2}e_2,$$

so that $J_{\delta}J_3 = -J_3J_{\delta}$ for any $\delta \neq 0$ (see Remark 3.4 and [5] for more information).

Finally, the structures $(J_4, \langle \cdot, \cdot \rangle_4)$ are flat if $\kappa = 0$. Otherwise, they are Ricci-flat and special recurrent with recurrence one-form $\xi_4 = -4e^4$ and curvature operators given by

$$\mathcal{R}_4(e^1 \wedge e^4) = -3\kappa e^2 \wedge e^3 = R_4(e_1, e_4, e_1, e_4)e^2 \wedge e^3.$$

Therefore, they are simply harmonic and locally modelled on (3.1), so their curvature tensors are semi-symmetric. Moreover, these structures admit opposite almost para-Kähler structures compatible with the opposite symplectic two-forms

$$\Omega'_4 = e^{13} + e^{24} + \mu e^{14}, \quad \mu \in \mathbb{R}.$$

The structures $(J_3, \langle \cdot, \cdot \rangle_3)$ and $(J_4, \langle \cdot, \cdot \rangle_4)$ are covered by Theorem 3.6.

3.12 Para-Kähler structures on $r_{4,-\alpha,\alpha}$

Let $\{e_1, e_2, e_3, e_4\}$ be a basis of the Lie algebra $\mathfrak{r}_{4,-\alpha,\alpha}$ determined by

$$[e_1, e_4] = -e_1, \quad [e_2, e_4] = \alpha e_2, \quad [e_3, e_4] = -\alpha e_3,$$

where $0 < \alpha < 1$. According to [6], this Lie algebra corresponds to the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} \mathbb{R}^3$ given by

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = e_1, \quad \varphi(e_2) = -\alpha e_2, \quad \varphi(e_3) = \alpha e_3.$$
The symplectic forms on $\mathfrak{r}_{4,-\alpha,\alpha}$ are given by

$$\omega = \alpha_{14}e^{14} + \alpha_{23}e^{23} + \alpha_{24}e^{24} + \alpha_{34}e^{34}, \quad \alpha_{14}\alpha_{23} \neq 0,$$

and all of them are symplectomorphically equivalent to $\Omega = e^{14} + e^{23}$ through a Lie algebra automorphism of the form

$$\Phi = \begin{pmatrix} z_{11} & 0 & 0 & z_{14} \\ 0 & z_{22} & 0 & z_{24} \\ 0 & 0 & z_{33} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix}, \quad \text{with} \quad z_{11}z_{22}z_{33} = 0.$$

Moreover, the automorphisms preserving the symplectic structure $(\mathfrak{r}_{4,-\alpha,\alpha},\Omega)$ are given by Φ above with $z_{11} = 1$, $z_{24} = z_{34} = 0$ and $z_{33}z_{22} = 1$. The different classes of para-Kähler structures on this Lie algebra arise from the cases $a_{41} \neq 0$ and $a_{41} = 0$, which we study separately in what follows.

Para-Kähler and opposite para-Kähler structures on $\mathfrak{r}_{4,-\alpha,\alpha}$

Straightforward calculations as in the previous sections show that any para-Kähler structure with $a_{41} \neq 0$ is equivalent to

$$(J_1, \langle \cdot, \cdot \rangle_1) : \begin{cases} J_1 e_1 = \kappa e_4, & J_1 e_2 = -e_2, & J_1 e_3 = e_3, & J_1 e_4 = \frac{1}{\kappa} e_1, \\ \langle \cdot, \cdot \rangle_1 = \kappa e^1 \circ e^1 - \frac{1}{\kappa} e^4 \circ e^4 + 2e^2 \circ e^3, & \kappa \neq 0, \end{cases}$$

which induces the decompositions

$$\mathfrak{r}_{4,-lpha,lpha} = \mathfrak{L}_1 \oplus \mathfrak{L}'_1 = \operatorname{span}\{e_3, e_4 + \frac{1}{\kappa}e_1\} \oplus \operatorname{span}\{e_2, e_4 - \frac{1}{\kappa}e_1\}$$

of $\mathfrak{r}_{4,-\alpha,\alpha}$ as direct sums of Lagrangian subalgebras. These structures are locally symmetric and their Ricci operators are diagonalizable

$$\operatorname{Ric}_1 = \operatorname{diag}[0, 0, \kappa, \kappa].$$

Therefore, the underlying manifolds split locally as products $\mathbb{L}^2 \times N$ of a Lorentzian surface N of constant Gaussian curvature $K_N = \kappa$ and the Minkowskian plane. Consequently, they also admit opposite para-Kähler structures and are covered by Theorem 3.1-(*i.d*).

Ricci-flat para-Kähler structures

Assuming that $a_{41} = 0$, the corresponding para-Kähler structures are equivalent to one of the following two families of para-Kähler structures.

$$(J_2, \langle \cdot, \cdot \rangle_2) : \begin{cases} J_2 e_1 = -e_1, \quad J_2 e_2 = -e_2 + \kappa e_3, \quad J_2 e_3 = e_3, \quad J_2 e_4 = e_4, \\ \langle \cdot, \cdot \rangle_2 = \kappa e^2 \circ e^2 + 2(e^1 \circ e^4 + e^2 \circ e^3), \quad \kappa = 0, \pm 1, \end{cases}$$

which induces the Lagrangian decompositions

$$\begin{aligned} \mathbf{\mathfrak{r}}_{4,-\alpha,\alpha} &= \mathfrak{L}_2 \oplus \mathfrak{L}_2' = \operatorname{span}\{e_3, e_4\} \oplus \operatorname{span}\{e_1, e_3 - \frac{2}{\kappa}e_2\} \quad (\kappa \neq 0) \\ &= \operatorname{span}\{e_3, e_4\} \oplus \operatorname{span}\{e_1, e_2\} \quad (\kappa = 0), \end{aligned}$$

or

$$(J_3, \langle \cdot, \cdot \rangle_3) : \begin{cases} J_3 e_1 = e_1, \quad J_3 e_2 = -e_2, \quad J_3 e_3 = -\kappa e_2 + e_3, \quad J_3 e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_3 = \kappa e^3 \circ e^3 + 2(e^2 \circ e^3 - e^1 \circ e^4), \quad \kappa = 0, \pm 1, \end{cases}$$

so that $\mathfrak{r}_{4,-\alpha,\alpha} = \mathfrak{L}_3 \oplus \mathfrak{L}'_3 = \operatorname{span}\{e_1, e_3 - \frac{\kappa}{2}e_2\} \oplus \operatorname{span}\{e_2, e_4\}.$

The structures $(J_2, \langle \cdot, \cdot \rangle_2)$ are flat if $\kappa = 0$, while structures $(J_3, \langle \cdot, \cdot \rangle_3)$ are flat if either $\kappa = 0$ or $\alpha = \frac{1}{2}$. Otherwise, all the structures above are Ricci-flat and special recurrent with recurrence one-forms given by $\xi_2 = 2(1 + \alpha)e^4$ and $\xi_3 = 2(1 - \alpha)e^4$, respectively, and curvature operators determined by

$$\mathcal{R}_2(e^2 \wedge e^4) = \alpha(1+2\alpha)\kappa e^1 \wedge e^3 = R_2(e_2, e_4, e_4, e_2)e^1 \wedge e^3,$$

$$\mathcal{R}_3(e^3 \wedge e^4) = \alpha(1-2\alpha)\kappa e^1 \wedge e^2 = R_3(e_3, e_4, e_3, e_4)e^1 \wedge e^2.$$

Therefore, they are simply harmonic and locally modelled on (3.1), so their curvature tensor is semi-symmetric. Furthermore, these structures admit opposite almost para-Kähler structures compatible with the opposite symplectic forms given by

$$\Omega_2'=e^{14}-e^{23}+\mu e^{24},\quad \Omega_3'=e^{14}-e^{23}+\mu e^{34}$$

where $\mu \in \mathbb{R}$. All these structures are covered by Theorem 3.6.

3.13 Para-Kähler structures on \mathfrak{h}_4

Let \mathfrak{h}_4 be the Lie algebra generated by $\{e_1, e_2, e_3, e_4\}$, so that

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = -\frac{1}{2}e_1, \quad [e_2, e_4] = -e_1 - \frac{1}{2}e_2, \quad [e_3, e_4] = -e_3.$$

According to [6], this Lie algebra corresponds to the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} H^3$ given by

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = \frac{1}{2}e_1, \quad \varphi(e_2) = e_1 + \frac{1}{2}e_2, \quad \varphi(e_3) = e_3.$$

The symplectic structures on h_4 , which are given by

$$\omega = \alpha_{12}(e^{12} - e^{34}) + \alpha_{14}e^{14} + \alpha_{24}e^{24}, \quad \alpha_{12} \neq 0,$$

are equivalent to $\Omega_{\varepsilon} = \varepsilon(e^{12} - e^{34}), \varepsilon = \pm 1$, through a Lie algebra automorphism

$$\Phi = \begin{pmatrix} z_{22} & z_{12} & 0 & z_{14} \\ 0 & z_{22} & 0 & z_{24} \\ 2z_{22}z_{24} & 2(z_{12} + 2z_{22})z_{24} - 2z_{14}z_{22} & z_{22}^2 & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix} \quad \text{with} \quad z_{22} \neq 0.$$

Moreover, the Lie algebra automorphisms preserving the symplectic structure $(\mathfrak{h}_4, \Omega_{\varepsilon})$ are given by Φ above with $z_{24} = z_{14} = 0$ and $z_{22} = \pm 1$. Now, it is easy to see that any para-Kähler structure on $(\mathfrak{h}_4, \Omega_{\varepsilon})$ is equivalent to

$$(J, \langle \cdot, \cdot \rangle) : \begin{cases} Je_1 = -\varepsilon e_1, \quad Je_2 = \varepsilon e_2, \quad Je_3 = \varepsilon e_3, \quad Je_4 = -\varepsilon e_4, \\ \langle \cdot, \cdot \rangle = 2(e^1 \circ e^2 + e^3 \circ e^4), \qquad \varepsilon = \pm 1, \end{cases}$$

which correspond to the Lagrangian decomposition

$$\mathfrak{h}_4 = \mathfrak{L} \oplus \mathfrak{L}' = \operatorname{span}\{e_2, e_3\} \oplus \operatorname{span}\{e_1, e_4\}.$$

A straightforward calculation shows that the structures $(J, \langle \cdot, \cdot \rangle)$ are Ricci-flat with recurrent curvature. The recurrence one-form is given by $\xi = e^4$ and their curvature operator is determined by

$$\mathcal{R}(e^2 \wedge e^4) = -e^1 \wedge e^3 = R(e_2, e_4, e_2, e_4)e^1 \wedge e^3.$$

As a consequence, $(\mathfrak{h}_4, J, \langle \cdot, \cdot \rangle)$ are simply harmonic manifolds with special recurrent curvature and locally modelled on (3.1) whose curvature tensor is semi-symmetric and so they are covered by Theorem 3.6. A straightforward calculation shows that the structures $(\mathfrak{h}_4, J, \langle \cdot, \cdot \rangle)$ do not admit any opposite almost para-Kähler structures.

Remark 3.10. These structures, together with the structures $(J_{13}, \langle \cdot, \cdot \rangle_{13})$ on $(\mathfrak{d}_{4,1}, \Omega_1)$ given in Section 3.14.1, $(J_{1i}, \langle \cdot, \cdot \rangle_{1i})$, i = 2, 3, on $(\mathfrak{d}_{4,2}, \Omega_1)$ given in Section 3.17.1, and $(J_{32}, \langle \cdot, \cdot \rangle_{32})$ on $(\mathfrak{d}_{4,2}, \Omega_3)$ given in Section 3.17.3, are the only Ricci-flat para-Kähler structures that do not admit any left-invariant opposite almost para-Kähler structures.

3.14 Para-Kähler structures on $\mathfrak{d}_{4,1}$

Let $\mathfrak{d}_{4,1}$ be the Lie algebra generated by $\{e_1, e_2, e_3, e_4\}$ so that

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = -e_1, \quad [e_3, e_4] = -e_3$$

According to [6], this Lie algebra corresponds to the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} H^3$ given by

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = e_1, \quad \varphi(e_2) = 0, \quad \varphi(e_3) = e_3.$$

The symplectic structures on $\mathfrak{d}_{4,1}$ are given by

$$\omega = \alpha_{12}(e^{12} - e^{34}) + \alpha_{14}e^{14} + \alpha_{24}e^{24}, \quad \alpha_{12} \neq 0.$$

All these two-forms are symplectomorphycally equivalent to either $\Omega_1 = e^{12} - e^{34}$ or $\Omega_2 = e^{12} - e^{34} + e^{24}$ through an automorphisms of $\mathfrak{d}_{4,1}$, which are given by

$$\Phi = \begin{pmatrix} z_{11} & 0 & 0 & z_{14} \\ 0 & z_{22} & 0 & 0 \\ z_{31} & -z_{14}z_{22} & z_{11}z_{22} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix}, \quad \text{with} \quad z_{11}z_{22} \neq 0.$$

Besides, the automorphisms Φ_i that preserve each symplectic structure $(\mathfrak{d}_{4,1}, \Omega_i)$, i = 1, 2, are determined by the conditions

 $\Phi_1: z_{31} = 0, \quad z_{22}z_{11} = 1$ and $\Phi_2: z_{31} = 0, \quad z_{22} = 1, \quad z_{11} = 1.$

We study the two symplectic structures separately.

3.14.1 Para-Kähler structures on $(\mathfrak{d}_{4,1}, \Omega_1)$

There are three distinct situations that give rise to different geometries.

Para-Kähler structures of non-zero constant para-holomorphic sectional curvature

Assume that $a_{43} \neq 0$. Then necessarily $a_{23} = 0$ and a straightforward calculation shows that any para-Kähler structure is equivalent to

$$(J_{11}, \langle \cdot, \cdot \rangle_{11}) : \begin{cases} J_{11}e_1 = -e_1, \quad J_{11}e_2 = e_2, \quad J_{11}e_3 = -\kappa e_4, \quad J_{11}e_4 = -\frac{1}{\kappa}e_3, \\ \langle \cdot, \cdot \rangle_{11} = 2e^1 \circ e^2 + \kappa e^3 \circ e^3 - \frac{1}{\kappa}e^4 \circ e^4, \quad \kappa \neq 0, \end{cases}$$

so that $\mathfrak{d}_{4,1}$ splits as direct sums of Lagrangian subalgebras as

$$\mathfrak{d}_{4,1} = \mathfrak{L}_{11} \oplus \mathfrak{L}'_{11} = \operatorname{span}\{e_2, e_4 - \frac{1}{\kappa}e_3\} \oplus \operatorname{span}\{e_1, e_4 + \frac{1}{\kappa}e_3\}.$$

A straightforward calculation shows that the structures $(J_{11}, \langle \cdot, \cdot \rangle_{11})$ have constant para-holomorphic sectional curvature $H_{11} = \kappa$, thus corresponding to those given by Theorem 3.1-(*i.b*).

Locally symmetric Ricci-flat para-Kähler structures

Set $a_{43} = 0$, $a_{23} = 0$. Then, any para-Kähler structure is equivalent to one of the following two families.

$$(J_{12}, \langle \cdot, \cdot \rangle_{12}) : \begin{cases} J_{12}e_1 = -e_1, \ J_{12}e_2 = -\kappa \ e_1 + e_2, \ J_{12}e_3 = e_3, \ J_{12}e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_{12} = \kappa \ e^2 \circ e^2 + 2(e^1 \circ e^2 + e^3 \circ e^4), \quad \kappa = 0, \pm 1, \end{cases}$$

$$(J_{13}, \langle \cdot, \cdot \rangle_{13}) : \begin{cases} J_{13}e_1 = e_1 + \kappa \ e_2, \ J_{13}e_2 = -e_2, \ J_{13}e_3 = e_3, \ J_{13}e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_{13} = \kappa \ e^1 \circ e^1 - 2(e^1 \circ e^2 - e^3 \circ e^4), \quad \kappa = 0, \pm 1. \end{cases}$$

These structures induce the Lagrangian decompositions

$$\mathfrak{d}_{4,1} = \mathfrak{L}_{12} \oplus \mathfrak{L}'_{12} = \operatorname{span}\{e_3, e_2 - \frac{\kappa}{2}e_1\} \oplus \operatorname{span}\{e_1, e_4\}$$
$$= \mathfrak{L}_{13} \oplus \mathfrak{L}'_{13} = \operatorname{span}\{e_3, e_2 + \frac{2}{\kappa}e_1\} \oplus \operatorname{span}\{e_2, e_4\} \quad (\kappa \neq 0)$$
$$= \mathfrak{L}_{13} \oplus \mathfrak{L}'_{13} = \operatorname{span}\{e_1, e_3\} \oplus \operatorname{span}\{e_2, e_4\} \quad (\kappa = 0).$$

The structures $(J_{12}, \langle \cdot, \cdot \rangle_{12})$ are all flat, while the structures $(J_{13}, \langle \cdot, \cdot \rangle_{13})$ are flat if and only if $\kappa = 0$. Otherwise, they are locally symmetric and Ricci-flat and so they correspond to those in Theorem 3.1-(*i*). As a consequence, their underlying pseudo-Riemannian structures are modelled on (3.1) with $\Psi(x^1, x^2) = \pm (x^2)^2$.

Finally, $(\mathfrak{d}_{4,1}, J_{13}, \langle \cdot, \cdot \rangle_{13})$ do not admit any invariant opposite almost para-Kähler structures (see Remark 3.10).

Locally symmetric para-Kähler structures with nilpotent Ricci operator

Setting $a_{43} = 0$ and $a_{23} \neq 0$, it is non difficult to see that any para-Kähler structure is equivalent to

$$(J_{14}, \langle \cdot, \cdot \rangle_{14}) : \begin{cases} J_{14}e_1 = \kappa e_2 - e_4, \ J_{14}e_2 = e_3, \ J_{14}e_3 = e_2, \ J_{14}e_4 = -e_1 + \kappa e_3, \\ \langle \cdot, \cdot \rangle_{14} = \kappa (e^1 \circ e^1 + e^4 \circ e^4) + 2(e^1 \circ e^3 + e^2 \circ e^4), \quad \kappa \in \mathbb{R}, \end{cases}$$

which correspond to the Lagrangian decompositions of $\mathfrak{d}_{4,1}$

$$\mathfrak{L}_{14} \oplus \mathfrak{L}'_{14} = \operatorname{span}\{e_2 + e_3, e_4 - e_1 - \kappa e_2\} \oplus \operatorname{span}\{e_3 - e_2, e_4 + e_1 - \kappa e_2\}.$$

The structures $(J_{14}, \langle \cdot, \cdot \rangle_{14})$ are locally symmetric and have two-step nilpotent Ricci operators. Furthermore, the metrics $\langle \cdot, \cdot \rangle_{14}$ are locally conformally flat if and only if $\kappa = 0$. In any other case, their anti-self-dual Weyl curvature operators are two-step nilpotent, and so this structures correspond to those in Theorem 3.1-(*ii.b*).

3.14.2 Para-Kähler structures on $(\mathfrak{d}_{4,1}, \Omega_2)$

Direct calculations show that all the para-Kähler structures are flat in this case and they are equivalent to the structures

$$(J_{21}, \langle \cdot, \cdot \rangle_{21}) : \begin{cases} J_{21}e_1 = -e_1, \ J_{21}e_2 = -\kappa e_1 + e_2, \ J_{21}e_3 = e_3, \ J_{21}e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_{21} = \kappa e^2 \circ e^2 + 2(e^1 \circ e^2 - e^2 \circ e^4 + e^3 \circ e^4), \quad \kappa \in \mathbb{R}, \end{cases}$$

which correspond to the Lagrangian decompositions

$$\mathfrak{d}_{4,1} = \mathfrak{L}_{21} \oplus \mathfrak{L}'_{21} = \operatorname{span}\{e_3, e_2 - \frac{\kappa}{2}e_1\} \oplus \operatorname{span}\{e_1, e_4\}.$$

3.15 Para-Kähler structures on $\mathfrak{d}_{4,\frac{1}{2}}$

Let $\mathfrak{d}_{4,\frac{1}{2}}$ be the Lie algebra generated by $\{e_1,e_2,e_3,e_4\}$ so that

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = -\frac{1}{2}e_1, \quad [e_2, e_4] = -\frac{1}{2}e_2, \quad [e_3, e_4] = -e_3,$$

whose automorphisms correspond to

$$\Phi = \begin{pmatrix} z_{11} & z_{12} & 0 & z_{14} \\ z_{21} & z_{22} & 0 & z_{24} \\ z_{31} & z_{32} & z_{33} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix}, \quad \text{with} \quad \begin{array}{l} z_{31} = 2(z_{11}z_{24} - z_{14}z_{21}), \\ z_{32} = 2(z_{12}z_{24} - z_{14}z_{22}), \\ z_{33} = z_{11}z_{22} - z_{12}z_{21} \neq 0. \end{array}$$

According to [6], this Lie algebra corresponds to the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} H^3$ given by

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = \frac{1}{2}e_1, \quad \varphi(e_2) = \frac{1}{2}e_2, \quad \varphi(e_3) = e_3.$$

All the symplectic forms on $\mathfrak{d}_{4,\frac{1}{2}}$ are given by

$$\omega = \alpha_{12}(e^{12} - e^{34}) + \alpha_{14}e^{14} + \alpha_{24}e^{24}, \quad \alpha_{12} \neq 0,$$

so they are symplectomorphically equivalent to $\Omega = e^{12} - e^{34}$ (see [115]). To give a description of the para-Kähler structures on this Lie algebra, we make use of the Lie algebra automorphisms that preserve Ω , i.e., those which satisfy

$$z_{11}z_{22} - z_{12}z_{21} = 1$$
, $z_{14}z_{21} - z_{11}z_{24} = 0$, $z_{14}z_{22} - z_{12}z_{24} = 0$.

Now, the description of para-Kähler structures depends on whether or not $a_{43} = 0$.

Para-Kähler structures of non-zero constant para-holomorphic sectional curvature

If $a_{43} \neq 0$, then any para-Kähler structures is equivalent to

$$(J_1, \langle \cdot, \cdot \rangle_1) : \begin{cases} J_1 e_1 = -e_1, \quad J_1 e_2 = e_2, \quad J_1 e_3 = -\kappa e_4, \quad J_1 e_4 = -\frac{1}{\kappa} e_3, \\ \langle \cdot, \cdot \rangle_1 = 2e^1 \circ e^2 + \kappa e^3 \circ e^3 - \frac{1}{\kappa} e^4 \circ e^4, \quad \kappa \neq 0, \end{cases}$$

which corresponds to the Lagrangian decompositions

$$\mathfrak{d}_{4,\frac{1}{2}} = \mathfrak{L}_1 \oplus \mathfrak{L}'_1 = \operatorname{span}\{e_2, e_4 - \frac{1}{\kappa}e_3\} \oplus \operatorname{span}\{e_1, e_4 + \frac{1}{\kappa}e_3\}.$$

The para-holomorphic sectional curvatures of these structures are constant $H_1 = \kappa$, so they correspond to those given in Theorem 3.1-(*i*.*b*)

Flat para-Kähler structures

If $a_{43} = 0$, then the corresponding para-Kähler structures are flat and equivalent to

$$(J_2, \langle \cdot, \cdot \rangle_2) : \begin{cases} J_2 e_1 = e_1, \quad J_2 e_2 = -e_2, \quad J_2 e_3 = e_3, \quad J_2 e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_2 = 2(-e^1 \circ e^2 + e^3 \circ e^4), \end{cases}$$

so that $\mathfrak{d}_{4,\frac{1}{2}} = \mathfrak{L}_2 \oplus \mathfrak{L}'_2 = \operatorname{span}\{e_3, e_1\} \oplus \operatorname{span}\{e_4, e_2\}.$

3.16 Para-Kähler structures on $\mathfrak{d}_{4,\lambda}$

Let $\mathfrak{d}_{4,\lambda}$ be the Lie algebra generated by $\{e_1, e_2, e_3, e_4\}$ so that

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = -\lambda e_1, \quad [e_2, e_4] = (\lambda - 1)e_2, \quad [e_3, e_4] = -e_3,$$

for $\lambda > \frac{1}{2}$, $\lambda \neq 1, 2$. This Lie algebra is equivalent to the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} H^3$ given by

 $\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = \lambda e_1, \quad \varphi(e_2) = (1 - \lambda)e_2, \quad \varphi(e_3) = e_3.$

The automorphisms of the Lie algebra are given by

$$\Phi = \begin{pmatrix} z_{11} & 0 & 0 & z_{14} \\ 0 & z_{22} & 0 & z_{24} \\ \frac{z_{11}z_{24}}{1-\lambda} & -\frac{z_{22}z_{14}}{\lambda} & z_{11}z_{22} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix} \quad \text{with} \quad z_{11}z_{22} \neq 0.$$

Thus, any symplectic form on $\mathfrak{d}_{4,\lambda}$, which are

$$\omega = \alpha_{12}(e^{12} - e^{34}) + \alpha_{14}e^{14} + \alpha_{24}e^{24}, \quad \alpha_{12} \neq 0,$$

is symplectomorphically equivalent to $\Omega = e^{12} - e^{34}$ (see [115]). Now, the Lie algebra automorphisms preserving Ω are given by Φ with $z_{24} = z_{14} = 0$ and $z_{22}z_{11} = 1$.

The description of the para-Kähler structures depends on whether the component a_{43} vanishes or not.

Para-Kähler structures of non-zero constant para-holomorphic sectional curvature

If $a_{43} \neq 0$, the para-Kähler structures on $(\mathfrak{d}_{4,\lambda}, \Omega)$ are equivalent to

$$(J_1, \langle \cdot, \cdot \rangle_1) : \begin{cases} J_1 e_1 = -e_1, \quad J_1 e_2 = e_2, \quad J_1 e_3 = -\kappa e_4, \quad J_1 e_4 = -\frac{1}{\kappa} e_3, \\ \langle \cdot, \cdot \rangle_1 = 2e^1 \circ e^2 + \kappa e^3 \circ e^3 - \frac{1}{\kappa} e^4 \circ e^4, \quad \kappa \neq 0. \end{cases}$$

These structures induce the Lagrangian decompositions

$$\mathfrak{d}_{4,\lambda} = \mathfrak{L}_1 \oplus \mathfrak{L}'_1 = \operatorname{span}\{e_2, e_4 - \frac{1}{\kappa}e_3\} \oplus \operatorname{span}\{e_1, e_4 + \frac{1}{\kappa}e_3\}.$$

A straightforward calculation shows that their para-holomorphic sectional curvatures are constant $H_1 = \kappa$, so these metrics correspond to those given in Theorem 3.1-(*i.b*).

Ricci-flat para-Kähler structures

If $a_{43} = 0$, then the resulting para-Kähler structures are Ricci-flat equivalent to one of the families

$$(J_{2}, \langle \cdot, \cdot \rangle_{2}) : \begin{cases} J_{2}e_{1} = -e_{1}, & J_{2}e_{2} = -\kappa e_{1} + e_{2}, & J_{2}e_{3} = e_{3}, & J_{2}e_{4} = -e_{4} \\ \langle \cdot, \cdot \rangle_{2} = 2(e^{1} \circ e^{2} + e^{3} \circ e^{4}) + \kappa e^{2} \circ e^{2}, & \kappa = 0, \pm 1, \end{cases}$$
$$(J_{3}, \langle \cdot, \cdot \rangle_{3}) : \begin{cases} J_{3}e_{1} = e_{1} + \kappa e_{2}, & J_{3}e_{2} = -e_{2}, & J_{3}e_{3} = e_{3}, & J_{3}e_{4} = -e_{4}, \\ \langle \cdot, \cdot \rangle_{3} = \kappa e^{1} \circ e^{1} - 2(e^{1} \circ e^{2} - e^{3} \circ e^{4}), & \kappa = 0, \pm 1. \end{cases}$$

These structures induce the Lagrangian decompositions

$$\begin{aligned} \mathfrak{d}_{4,\lambda} &= \mathfrak{L}_2 \oplus \mathfrak{L}_2' = \operatorname{span}\{e_3, e_2 - \frac{\kappa}{2}e_1\} \oplus \operatorname{span}\{e_1, e_4\} \\ &= \mathfrak{L}_3 \oplus \mathfrak{L}_3' = \operatorname{span}\{e_3, e_2 + \frac{2}{\kappa}e_1\} \oplus \operatorname{span}\{e_4, e_2\} \quad (\kappa \neq 0) \\ &= \mathfrak{L}_3 \oplus \mathfrak{L}_3' = \operatorname{span}\{e_3, e_1\} \oplus \operatorname{span}\{e_4, e_2\} \quad (\kappa = 0). \end{aligned}$$

Both structures above are flat if and only if $\kappa = 0$. Otherwise, they have recurrent curvature with recurrence one-forms $\xi_2 = 2\lambda e^4$ and $\xi_3 = 2(1 - \lambda)e^4$, respectively. Moreover, their curvature operators acting on the space of two-forms are given by

$$\mathcal{R}_2(e^2 \wedge e^4) = -\kappa(2\lambda^2 - 3\lambda + 1)e^1 \wedge e^3 = R_2(e_2, e_4, e_2, e_4)e^1 \wedge e^3,$$

$$\mathcal{R}_3(e^1 \wedge e^4) = \kappa\lambda(2\lambda - 1)e^2 \wedge e^3 = R_3(e_1, e_4, e_4, e_1)e^2 \wedge e^3,$$

respectively. Consequently, the corresponding manifolds are simply harmonic with special recurrent curvature tensor, thus locally isometric to a metric given by (3.1) and their curvature tensors are semi-symmetric. Furthermore, neither $(J_2, \langle \cdot, \cdot \rangle_2)$ nor $(J_3, \langle \cdot, \cdot \rangle_3)$ admit any opposite almost para-Kähler structure. These structures are thus covered by Theorem 3.6.

3.17 Para-Kähler structures on $\mathfrak{d}_{4,2}$

Let $\mathfrak{d}_{4,2}$ be the Lie algebra generated by $\{e_1, e_2, e_3, e_4\}$ with Lie brackets

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = -2e_1, \quad [e_2, e_4] = e_2, \quad [e_3, e_4] = -e_3,$$

which, according to [6], corresponds to the semi-direct extension $\mathbb{R}e_4 \ltimes_{\varphi} H^3$ given by

$$\varphi = \operatorname{ad}(e_4): \quad \varphi(e_1) = 2e_1, \quad \varphi(e_2) = -e_2, \quad \varphi(e_3) = e_3$$

The symplectic forms on $\mathfrak{d}_{4,2}$ are of the given by

$$\omega = \alpha_{12}(e^{12} - e^{34}) + \alpha_{14}e^{14} + \alpha_{23}e^{23} + \alpha_{24}e^{24}, \quad \alpha_{12}^2 - \alpha_{14}\alpha_{23} \neq 0.$$

Any such form is symplectomorphically equivalent to

$$\Omega_1 = e^{12} - e^{34}, \quad \Omega_2 = e^{14} + e^{23}, \quad \text{or} \quad \Omega_3 = e^{14} - e^{23},$$

through a Lie algebra automorphism given by (see [115])

$$\Phi = \begin{pmatrix} z_{11} & 0 & 0 & z_{14} \\ 0 & z_{22} & 0 & z_{24} \\ -z_{11}z_{24} & -\frac{z_{14}z_{22}}{2} & z_{11}z_{22} & z_{34} \\ 0 & 0 & 0 & 1 \end{pmatrix} \quad \text{with} \quad z_{11}z_{22} \neq 0.$$
(3.2)

The geometric situation is different for each Ω_i , so we consider the three possibilities separately.

3.17.1 Para-Kähler structures on $(\mathfrak{d}_{4,2}, \Omega_1)$

The Lie algebra automorphisms preserving the symplectic structure $(\mathfrak{d}_{4,2}, \Omega_1)$ with $\Omega_1 = e^{12} - e^{34}$ are the ones given by Φ above with $z_{14} = z_{24} = 0$ and $z_{22}z_{11} = 1$. Now, we consider the cases $a_{43} \neq 0$ and $a_{43} = 0$, which give rise to different geometries as follows.

Para-Kähler structures of non-zero constant para-holomorphic sectional curvature

If $a_{43} \neq 0$, then any para-Kähler structure is equivalent to

$$(J_{11}, \langle \cdot, \cdot \rangle_{11}) : \begin{cases} J_{11}e_1 = -e_1, \ J_{11}e_2 = e_2, \ J_{11}e_3 = -\kappa e_4, \ J_{11}e_4 = -\frac{1}{\kappa}e_3, \\ \langle \cdot, \cdot \rangle_{11} = 2e^1 \circ e^2 + \kappa e^3 \circ e^3 - \frac{1}{\kappa}e^4 \circ e^4, \quad \kappa \neq 0, \end{cases}$$

so that $\mathfrak{d}_{4,2} = \mathfrak{L}_{11} \oplus \mathfrak{L}'_{11} = \operatorname{span}\{e_2, e_4 - \frac{1}{\kappa}e_3\} \oplus \operatorname{span}\{e_1, e_4 + \frac{1}{\kappa}e_3\}.$

A straightforward calculation shows that all these structures have constant para-holomorphic sectional curvature $H_{11} = \kappa$, and so they correspond to those in Theorem 3.1-(*i.b*).

Ricci-flat para-Kähler structures

Assuming that $a_{43} = 0$, the corresponding para-Kähler structures on $(\mathfrak{d}_{4,2}, \Omega_1)$ are equivalent either to

$$(J_{12}, \langle \cdot, \cdot \rangle_{12}) : \begin{cases} J_{12}e_1 = -e_1 + \kappa e_2, \ J_{12}e_2 = e_2, \ J_{12}e_3 = -e_3, \ J_{12}e_4 = e_4, \\ \langle \cdot, \cdot \rangle_{12} = \kappa e^1 \circ e^1 + 2(e^1 \circ e^2 - e^3 \circ e^4), \quad \kappa = 0, \pm 1, \end{cases}$$

which induce the Lagrangian decompositions

$$\mathfrak{d}_{4,2} = \mathfrak{L}_{12} \oplus \mathfrak{L}'_{12} = \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_3, e_2 - \frac{2}{\kappa}e_1\} \qquad (\kappa \neq 0)$$
$$= \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_1, e_3\} \qquad (\kappa = 0),$$

or to

$$(J_{13}, \langle \cdot, \cdot \rangle_{13}) : \begin{cases} J_{13}e_1 = -e_1, \ J_{13}e_2 = -\kappa e_1 + e_2, \ J_{13}e_3 = e_3, \ J_{13}e_4 = -e_4 \\ \langle \cdot, \cdot \rangle_{13} = \kappa e^2 \circ e^2 + 2(e^1 \circ e^2 + e^3 \circ e^4), \quad \kappa = 0, \pm 1, \end{cases}$$

where $\mathfrak{d}_{4,2} = \mathfrak{L}_{13} \oplus \mathfrak{L}'_{13} = \operatorname{span}\{e_2 - \frac{\kappa}{2}e_1, e_3\} \oplus \operatorname{span}\{e_1, e_4\}.$

The structures $(J_{12}, \langle \cdot, \cdot \rangle_{12})$ and $(J_{13}, \langle \cdot, \cdot \rangle_{13})$ are flat if $\kappa = 0$. Otherwise, they are Ricciflat and recurrent, with recurrence one-forms $\xi_{12} = -2e^4$ and $\xi_{13} = 4e^4$, respectively. Their corresponding curvature operators are given by

$$\mathcal{R}_{12}(e^1 \wedge e^4) = 6\kappa e^2 \wedge e^3 = R_{12}(e_1, e_4, e_4, e_1)e^2 \wedge e^3,$$
$$\mathcal{R}_{13}(e^2 \wedge e^4) = -3\kappa e^1 \wedge e^3 = R_{13}(e_2, e_4, e_2, e_4)e^1 \wedge e^3.$$

Therefore, these two families are simply harmonic and special recurrent, locally modelled by (3.1). Consequently, their curvature tensors are semi-symmetric and so they are covered by Theorem 3.6. None of these structures admits any opposite almost para-Kähler structures (see Remark 3.10).

Besides, the structures $(J_{13}, \langle \cdot, \cdot \rangle_{13})$ have an associated one-parameter family of hypersymplectic structures $(J_{13}, J_{\delta}, \langle \cdot, \cdot \rangle_{13})$ which are given by the Kähler structures

$$J_{\delta}e_1 = \frac{1}{\delta}e_3, \quad J_{\delta}e_2 = \frac{\kappa}{2\delta}e_3 + \delta e_4, \quad J_{\delta}e_3 = -\delta e_1, \quad J_{\delta}e_4 = \frac{1}{\delta}(\frac{\kappa}{2}e_1 - e_2),$$

so that $J_{\delta}J_{13} = -J_{13}J_{\delta}$, for any $\delta \neq 0$ (see Remark 3.4 and [5] for information).

3.17.2 Para-Kähler structures on $(\mathfrak{d}_{4,2}, \Omega_2)$

The Lie algebra automorphisms that preserve the symplectic structure $(\mathfrak{d}_{4,2}, \Omega_2)$, where $\Omega_2 = e^{14} + e^{23}$, are the ones given by Φ in (3.2) with $z_{24} = z_{34} = 0$, $z_{11} = 1$ and $z_{22} = \pm 1$. Any para-Kähler structure on this Lie algebra satisfies $a_{21} = 0$ and different situations may occur depending on whether $a_{23} = 0$ or $a_{23} \neq 0$. We study each case separately.

Flat para-Kähler structures

Assuming that $a_{23} = 0$ and $a_{41} = 0$, straightforward calculations show that any para-Kähler structure is equivalent to the flat structure

$$(J_{21}, \langle \cdot, \cdot \rangle_{21}) : \begin{cases} J_{21}e_1 = e_1, \quad J_{21}e_2 = -e_2, \quad J_{21}e_3 = e_3, \quad J_{21}e_4 = -e_4, \\ \langle \cdot, \cdot \rangle_{21} = 2(e^2 \circ e^3 - e^1 \circ e^4), \end{cases}$$

so that $\mathfrak{d}_{4,2} = \mathfrak{L}_{21} \oplus \mathfrak{L}'_{21} = \operatorname{span}\{e_1, e_3\} \oplus \operatorname{span}\{e_2, e_4\}.$

Para-Kähler structures which are not semi-symmetric

Assuming that $a_{23} = 0$ and $a_{41} \neq 0$, straightforward calculations as in the previous sections show that the para-Kähler structures on this Lie algebra are equivalent to

$$(J_{22}, \langle \cdot, \cdot \rangle_{22}) : \begin{cases} J_{22}e_1 = e_1 + \kappa e_4, \ J_{22}e_2 = -e_2, \ J_{22}e_3 = e_3, \ J_{22}e_4 = -e_4 \\ \langle \cdot, \cdot \rangle_{22} = \kappa e^1 \circ e^1 + 2(e^2 \circ e^3 - e^1 \circ e^4), \quad \kappa \neq 0, \end{cases}$$

which induce the Lagrangian decomposition

$$\mathfrak{d}_{4,2} = \mathfrak{L}_{22} \oplus \mathfrak{L}'_{22} = \operatorname{span}\{e_3, e_4 + \frac{2}{\kappa}e_1\} \oplus \operatorname{span}\{e_2, e_4\}.$$

The Ricci operators of these structures are diagonalizable

$$\operatorname{Ric}_{22} = \operatorname{diag}[0, 0, 4\kappa, 4\kappa],$$

and their self-dual and anti-self-dual Weyl curvature operators have the same eigenvalues. Moreover, the anti-self-dual Weyl curvature operator W^- has a double root of its minimal polynomial. Besides, their curvature tensor is not semi-symmetric, so it does not correspond to the curvature of any symmetric space. These structures correspond to those given in Theorem 3.7-(*ii.a*).

Assuming that $a_{23} \neq 0$, the para-Kähler structures are equivalent to

$$(J_{23}, \langle \cdot, \cdot \rangle_{23}) : \begin{cases} J_{23}e_1 = \frac{\kappa}{2}e_4, \ J_{23}e_2 = -\frac{1}{\kappa}e_3, \ J_{23}e_3 = -\kappa e_2, \ J_{23}e_4 = \frac{2}{\kappa}e_1, \\ \langle \cdot, \cdot \rangle_{23} = \frac{\kappa}{2}e^1 \circ e^1 - \frac{1}{\kappa}e^2 \circ e^2 + \kappa e^3 \circ e^3 - \frac{2}{\kappa}e^4 \circ e^4, \ \kappa \neq 0, \end{cases}$$

so that

$$\mathfrak{d}_{4,2} = \mathfrak{L}_{23} \oplus \mathfrak{L}'_{23} = \operatorname{span}\{e_3 - \kappa e_2, e_4 + \frac{2}{\kappa}e_1\} \oplus \operatorname{span}\{e_3 + \kappa e_2, e_4 - \frac{2}{\kappa}e_1\}.$$

The Ricci operators of these structures are diagonalizable

$$\operatorname{Ric}_{23} = \operatorname{diag}[0, 0, 3\kappa, 3\kappa],$$

and their self-dual and anti-self-dual Weyl curvature operators are diagonalizable with opposite eigenvalues. Furthermore, their curvature tensor is not semi-symmetric. These structures correspond to those given in Theorem 3.7-(*ii.b*).

Remark 3.11. Let Q_{2i} be the almost product structures associated to the Ricci operators Ric_{2i} , for i = 2, 3, so that $Q_{2i} = -\operatorname{Id}$ on ker Ric_{2i} and $Q_{2i} = \operatorname{Id}$ on the orthogonal distribution corresponding to eigenspace of the non-zero Ricci curvature. These give opposite almost paracomplex structures defined by

$$J_{2i}' = J_{2i}Q_{2i}, \quad i = 2, 3$$

A straightforward calculation now shows that J'_{2i} are opposite almost para-Kähler structures commuting with J_{2i} that have associated symplectic forms

$$\Omega'_{2i}(x,y) = \langle J'_{2i}x, y \rangle_{2i} = \Omega_3, \quad i = 2, 3.$$

3.17.3 Para-Kähler structures on $(\mathfrak{d}_{4,2}, \Omega_3)$

The symplectic structure $(\mathfrak{d}_{4,2}, \Omega_3)$ given by $\Omega_3 = e^{14} - e^{23}$ is preserved by the automorphisms Φ in (3.2) with $z_{24} = z_{34} = 0$, $z_{11} = 1$ and $z_{22} = \pm 1$. Considering the action of the automorphisms which preserve the symplectic structure, the different possibilities arise from the following cases.

Case 1. $a_{21} = a_{23} = a_{41} = 0$, $a_{22} \neq 0$, in which case one may assume $a_{32} = 0$. Case 2. $a_{21} = a_{23} = a_{41} = a_{22} = 0$. Case 3. $a_{21} = a_{23} = 0$, $a_{41} \neq 0$, in which case one may assume $a_{11} = 0$. Case 4. $a_{21} = 0$, $a_{23} \neq 0$, in which case one may assume $a_{22} = 0$. Case 5. $a_{21} \neq 0$, in which case one may assume $a_{13} = 0$.

We will study all these situations separately.

Flat para-Kähler structures

Assuming that $a_{21} = a_{23} = a_{41} = 0$ and $a_{22} \neq 0$ as in Case 1, if $a_{31} = 0$ then the corresponding para-Kähler structures are equivalent to the flat para-Kähler structure given by

$$(J_{31}, \langle \cdot, \cdot \rangle_{31}) : \begin{cases} J_{31}e_1 = -e_1, \quad J_{31}e_2 = e_2, \quad J_{31}e_3 = -e_3, \quad J_{31}e_4 = e_4, \\ \langle \cdot, \cdot \rangle_{31} = 2(e^1 \circ e^4 + e^2 \circ e^3), \end{cases}$$

which induces the Lagrangian decomposition

$$\mathfrak{d}_{4,2} = \mathfrak{L}_{31} \oplus \mathfrak{L}'_{31} = \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_1, e_3\}.$$

The case where $a_{31} \neq 0$ will be considered below, since it gives rise to a different geometric situation.

Ricci-flat para-Kähler structures

Assuming that $a_{21} = a_{23} = a_{41} = a_{22} = 0$ as in Case 2, the corresponding para-Kähler structures are equivalent to the structures $(J_{32}, \langle \cdot, \cdot \rangle_{32})$ given by

$$\begin{cases} J_{32}e_1 = -e_3, \ J_{32}e_2 = -\kappa e_3 + e_4, \ J_{32}e_3 = -e_1, \ J_{32}e_4 = -\kappa e_1 + e_2, \\ \langle \cdot, \cdot \rangle_{32} = \kappa (e^2 \circ e^2 + e^4 \circ e^4) + 2(e^1 \circ e^2 + e^3 \circ e^4), \quad \kappa \in \mathbb{R}, \end{cases}$$

which correspond to the Lagrangian decomposition of $\mathfrak{d}_{4,2}$ as

$$\mathfrak{L}_{32} \oplus \mathfrak{L}'_{32} = \operatorname{span}\{e_3 - e_1, e_4 + e_2 - \kappa e_1\} \oplus \operatorname{span}\{e_3 + e_1, e_4 - e_2 + \kappa e_1\}.$$

These structures are flat if $\kappa = 0$. Otherwise, they are Ricci-flat and special recurrent with recurrence one-forms $\xi_{32} = 4e^4$ and curvature operators

$$\mathcal{R}_{32}(e^2 \wedge e^4) = -3\kappa e^1 \wedge e^3 = R_{32}(e_2, e_4, e_2, e_4)e^1 \wedge e^3.$$

Consequently, the underlying structures are locally modelled on (3.1) and thus their curvature tensor is semi-symmetric and these structures are covered by Theorem 3.6. Finally, these structures do not admit opposite almost para-Kähler structures (see Remark 3.10).

Para-Kähler structures with diagonalizable Ricci operator which are not semi-symmetric

Assuming that $a_{21} = a_{23} = 0$ and $a_{41} \neq 0$ as in Case 3, the para-Kähler structures are equivalent to

$$(J_{33}, \langle \cdot, \cdot \rangle_{33}) : \begin{cases} J_{33}e_1 = -e_1 + \kappa e_4, \ J_{33}e_2 = e_2, \ J_{33}e_3 = -e_3, \ J_{33}e_4 = e_4 \\ \langle \cdot, \cdot \rangle_{33} = \kappa e^1 \circ e^1 + 2(e^1 \circ e^4 + e^2 \circ e^3), \quad \kappa \neq 0, \end{cases}$$

so that

$$\mathfrak{d}_{4,2} = \mathfrak{L}_{33} \oplus \mathfrak{L}'_{33} = \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_3, e_4 - \frac{2}{\kappa}e_1\}.$$

Their Ricci operators are diagonalizable with eigenvalues $\{4\kappa, 4\kappa, 0, 0\}$ and their self-dual and anti-self-dual Weyl curvature operators have the same eigenvalues. Moreover, W^- has a double root of its minimal polynomial, and the curvature tensors are not semi-symmetric. These structures correspond to those in Theorem 3.7-(*ii.a*).

The assumption that $a_{21} = 0$ and $a_{23} \neq 0$ as in Case 4 leads to para-Kähler structures equivalent to

$$(J_{34}, \langle \cdot, \cdot \rangle_{34}) : \begin{cases} J_{34}e_1 = \frac{\kappa}{2}e_4, \ J_{34}e_2 = -\frac{1}{\kappa}e_3, \ J_{34}e_3 = -\kappa e_2, \ J_{34}e_4 = \frac{2}{\kappa}e_1, \\ \langle \cdot, \cdot \rangle_{34} = \frac{\kappa}{2}e^1 \circ e^1 + \frac{1}{\kappa}e^2 \circ e^2 - \kappa e^3 \circ e^3 - \frac{2}{\kappa}e^4 \circ e^4, \ \kappa \neq 0, \end{cases}$$

which correspond to the Lagrangian decompositions

$$\mathfrak{d}_{4,2} = \mathfrak{L}_{34} \oplus \mathfrak{L}'_{34} = \operatorname{span}\{e_4 + \frac{2}{\kappa}e_1, e_3 - \kappa e_2\} \oplus \operatorname{span}\{e_4 - \frac{2}{\kappa}e_1, e_3 + \kappa e_2\}.$$

Their Ricci operators are diagonalizable with eigenvalues $\{3\kappa, 3\kappa, 0, 0\}$ and their self-dual and the anti-self-dual Weyl curvature operators are diagonalizable with opposite eigenvalues. Moreover, the curvature tensor is not semi-symmetric and these structures correspond to those in Theorem 3.7-(*ii.b*).

Remark 3.12. Let Q_{3i} be the almost product structures associated to the Ricci operators Ric_{3i} , for i = 2, 3, so that $Q_{3i} = -\operatorname{Id}$ on ker Ric_{3i} and $Q_{3i} = \operatorname{Id}$ on the orthogonal distribution corresponding to the eigenspace of the non-zero Ricci curvature. These determine opposite almost para-complex structures defined by

$$J_{3i}' = J_{3i}Q_{3i}, \quad i = 2, 3.$$

A straightforward calculation now shows that J'_{3i} are opposite almost para-Kähler structures commuting with J_{3i} that have associated symplectic forms

$$\Omega'_{3i}(x,y) = \langle J'_{3i}x, y \rangle_{3i} = \Omega_2, \quad i = 3, 4.$$

Para-Kähler structures which are not semi-symmetric with non-diagonalizable Ricci operator

Assuming that $a_{21} = a_{23} = a_{41} = 0$ and $a_{22} \neq 0$ as in Case 1, if $a_{31} \neq 0$, the para-Kähler structures on this Lie algebra are equivalent to the structures $(J_{35}, \langle \cdot, \cdot \rangle_{35})$ given by

$$\begin{cases} J_{35}e_1 = -e_1 - 2e_3, \ J_{35}e_2 = 2e_4 - e_2 - \kappa e_3, \ J_{35}e_3 = e_3, \ J_{35}e_4 = e_4, \\ \langle \cdot, \cdot \rangle_{35} = \kappa e^2 \circ e^2 + 2(e^1 \circ e^4 + 2e^1 \circ e^2 - e^2 \circ e^3), \quad \kappa \in \mathbb{R}, \end{cases}$$

which induce the Lagrangian decompositions

$$\mathfrak{d}_{4,2} = \mathfrak{L}_{35} \oplus \mathfrak{L}'_{35} = \operatorname{span}\{e_3, e_4\} \oplus \operatorname{span}\{e_4 - e_2 + \frac{\kappa}{2}e_1, e_3 + e_1\}.$$

Their Ricci operators are two-step nilpotent and their curvature tensors are not semi-symmetric. Moreover, their anti-self-dual Weyl curvature operator is three-step nilpotent and these structures correspond to those in Theorem 3.7-(*i*).

Finally, if $a_{21} \neq 0$ as in Case 5, then the para-Kähler structures are equivalent to the structures $(J_{36}, \langle \cdot, \cdot \rangle_{36})$ given by

$$\begin{cases} J_{36}e_1 = -e_1 + \kappa(e_2 + e_4), \ J_{36}e_2 = e_2, \ J_{36}e_3 = \kappa(e_2 + e_4) - e_3, \ J_{36}e_4 = e_4, \\ \langle \cdot, \cdot \rangle_{36} = \kappa(e^1 \circ e^1 + e^3 \circ e^3 + 2e^1 \circ e^3) + 2(e^1 \circ e^4 + e^2 \circ e^3), \quad \kappa \neq 0, \end{cases}$$

so that

$$\mathfrak{d}_{4,2} = \mathfrak{L}_{36} \oplus \mathfrak{L}'_{36} = \operatorname{span}\{e_2, e_4\} \oplus \operatorname{span}\{e_2 - \frac{2}{\kappa}e_1 + e_4, e_3 - e_1\}.$$

The Ricci operators of these structures have a single eigenvalue $\frac{3}{2}\kappa \neq 0$ which is a double root of their minimal polynomials, and their anti-self-dual Weyl curvature operators are three-step nilpotent. Besides, their curvature tensors are not semi-symmetric and these structures correspond to those in Theorem 3.7-(*i*). Since their anti-self-dual Weyl curvature operators are three-step nilpotent the structures above do not admit any opposite almost para-Kähler structure commuting with the Ricci operator [45].

3.18 Kähler Lie algebras

Kähler Lie algebras were classified by Ovando in [116] and the geometry of the corresponding structures, similar though it may be to that of para-Kähler Lie algebras, is more rigid, allowing less possibilities. The symmetric case is essentially the same, but there are no left-invariant locally symmetric Ricci-flat Kählerian structures in contrast with Theorem 3.1-(1.a).

Theorem 3.13. Let $(G, \langle \cdot, \cdot \rangle, J)$ be a non-flat locally symmetric four-dimensional indefinite Kähler Lie group. Then, it corresponds to one of the following situations.

(1) Its Ricci operator is diagonalizable and one of the following holds:

- (1.a) Its holomorphic sectional curvature is a non-zero constant.
- (1.b) Its metric is Einstein with non-zero scalar curvature.
- (1.c) The underlying manifold is locally a product of two surfaces of constant Gaussian curvature.
- (2) Its Ricci operator is non-diagonalizable and one of the following holds:
 - (2.a) Its Ricci operator has complex eigenvalues.
 - (2.b) Its Ricci operator is two-step nilpotent.

Following Ovando's classification, the structures in Assertion (1.a) correspond to the Kähler structures on the Lie groups determined by $\mathfrak{d}_{4,\frac{1}{2}}$ and $\mathfrak{d}'_{4,\delta}$, where $\delta > 0$.

The Kähler structures in Assertion (1.b) correspond to the metrics

$$\langle \cdot, \cdot \rangle = a(e^1 \circ e^1 - e^2 \circ e^2 + e^3 \circ e^3 - e^4 \circ e^4)$$

on \mathfrak{r}'_2 with symplectic form $\Omega = a(e^{13} - e^{24})$.

The Kähler structures which admit an opposite Kähler structure as in Assertion (1.c) correspond to the Kähler structures on $\mathfrak{rr}_{3,0}$, $\mathfrak{r}'_{4,0,\delta}$ for $\delta > 0$, and the structures on $\mathfrak{r}_2\mathfrak{r}_2$ given by the metrics

$$\langle \cdot, \cdot \rangle = a(e^1 \circ e^1 + e^2 \circ e^2) + b(e^3 \circ e^3 + e^4 \circ e^4)$$

with the symplectic form $\Omega = ae^{12} + be^{34}$, for ab < 0.

The Kähler structures in Assertion (2.a) correspond to the metrics

$$\langle \cdot, \cdot \rangle = a(e^1 \circ e^1 - e^2 \circ e^2 + e^3 \circ e^3 - e^4 \circ e^4) + 2b(e^1 \circ e^2 + e^3 \circ e^4)$$

on \mathfrak{r}'_2 with the symplectic form $\Omega = a(e^{13} - e^{24}) + b(e^{14} + e^{23})$, for $b \neq 0$.

Assertion (2.b) corresponds to the Kähler structures on $\mathfrak{d}_{4,1}$.

Remark 3.14. The metrics corresponding to Assertions (1.b) and (2.a) are linked by anti-Kähler structures, so that they have the same Levi-Civita connection as in the para-Kähler case.

Moreover, amongst the structures above there are locally conformally flat Kähler Lie groups as stated in the following result.

Corollary 3.15. Let (M, g, J) be a locally conformally flat four-dimensional indefinite Kähler manifold. Then, it is flat or it is locally isometric to the Kähler Lie group determined by one of the following:

- (1) The Lie algebra $\mathfrak{r}_2\mathfrak{r}_2$ with the metrics $\langle \cdot, \cdot \rangle = a(e^1 \circ e^1 + e^2 \circ e^2 e^3 \circ e^3 e^4 \circ e^4)$ and the symplectic structure $\Omega = a(e^{12} e^{34})$.
- (2) The Lie algebra \mathfrak{r}'_2 with the metrics $\langle \cdot, \cdot \rangle = 2b(e^1 \circ e^2 + e^3 \circ e^4)$ and the symplectic structure $\Omega = b(e^{14} + e^{23})$.
- (3) The Lie algebra $\mathfrak{d}_{4,1}$ with the metrics $\langle \cdot, \cdot \rangle = 2a(e^2 \circ e^4 e^1 \circ e^3)$ and the symplectic structure $\Omega = a(e^{12} e^{34})$.

Finally, the non-symmetric case is significantly simpler than its para-Kähler counterpart since the existence of Kähler and opposite almost Kähler structures is much more rigid than the corresponding para-Kähler analogue (see [44, 72]).

Theorem 3.16. Let $(G, \langle \cdot, \cdot \rangle, \Omega)$ be a non-symmetric four-dimensional indefinite Kähler Lie group. Then, one of the following holds.

- (1) $(G, \langle \cdot, \cdot \rangle)$ is semi-symmetric if and only if its Ricci operator vanishes, in which case its curvature tensor is special recurrent and the metric is simply harmonic.
- (2) $(G, \langle \cdot, \cdot \rangle)$ is not semi-symmetric if and only if it corresponds to the 3-symmetric space determined by the Kähler metrics

$$\langle \cdot, \cdot \rangle = a \left(\frac{1}{2} e^1 \circ e^1 + 2e^4 \circ e^4 \right) + b(e^2 \circ e^2 + e^3 \circ e^3)$$

on $\mathfrak{d}_{4,2}$ with the symplectic form $\Omega = ae^{14} + be^{23}$, for ab < 0.

Assertion (1) corresponds to the metrics

$$\langle \cdot, \cdot \rangle = -c(e^1 \circ e^1 + e^2 \circ e^2) - 2a(e^1 \circ e^4 + e^2 \circ e^3) + 2b(e^1 \circ e^3 - e^2 \circ e^4)$$

on \mathfrak{r}'_2 with the symplectic form $\Omega = ce^{12} + a(e^{13} - e^{24}) + b(e^{14} + e^{23})$, where $c(a^2 + b^2) \neq 0$, the metrics

$$\langle \cdot, \cdot \rangle = -c(e^1 \circ e^1 + e^4 \circ e^4) + 2b(e^1 \circ e^2 - e^3 \circ e^4) - 2a(e^1 \circ e^3 + e^2 \circ e^4)$$

on $\mathfrak{r}_{4,-1,-1}$ with symplectic the form $\Omega = a(e^{12} + e^{34}) + b(e^{13} - e^{24}) + ce^{14}$, where $c(a^2 + b^2) \neq 0$, and the metrics

$$\langle \cdot, \cdot \rangle = b(e^2 \circ e^2 + e^4 \circ e^4) + 2a(e^1 \circ e^2 + e^3 \circ e^4)$$

on $\mathfrak{d}_{4,2}$ with the symplectic form $\Omega = be^{24} + a(e^{14} + e^{23})$, for $a \neq 0$.

Part II

Solitons associated to geometric flows

In this part, we will devote ourselves to the study of solitons associated to two particular geometric flows: the Ricci flow and the Bach flow. In Chapter 4 we will give a complete description of four-dimensional Lorentzian left-invariant Ricci solitons and in Chapter 5 we will introduce a general technique to approach the classification of algebraic solitons and give a complete classification of four-dimensional Riemannian both Ricci and Bach solitons. But before we start, we will briefly introduce the two geometric flows that are the subject of our study and their corresponding solitons.

The Ricci flow: Ricci solitons

The Ricci flow was introduced by Hamilton in [83] with the intention to solve the Poincaré conjecture, which declares that any three-dimensional closed and simply connected manifold is homeomorphic to \mathbb{S}^3 .

The Ricci flow is given by the evolution equation

$$\frac{\partial}{\partial t}g_t = -2\,\rho_{g_t},\tag{II.1}$$

where g_t is a one-parameter family of pseudo-Riemannian metrics on a manifold M. For any given differentiable metric g_0 on a closed manifold M, there exists a unique solution g_t , with $t \in [0, \varepsilon)$ for some $\varepsilon > 0$, to the Ricci flow equation such that $g_t|_{t=0} = g_0$.

The first examples of solutions to the Ricci flow are given by Einstein metrics, which provide solutions of the form

$$g_t = (1 - 2\mu t)g_0, \quad \text{where} \quad \begin{cases} t \in \left(-\infty, \frac{1}{2\mu}\right) & \text{if} \quad \mu > 0, \\ t \in \left(\frac{1}{2\mu}, \infty\right) & \text{if} \quad \mu < 0, \\ t \in (-\infty, \infty) & \text{if} \quad \mu = 0, \end{cases}$$

for an initial Einstein metric g_0 such that $\rho_{g_0} = \mu g_0$. In any of the three cases given by the different values of μ , g_0 remains invariant modulo homotheties. If we allow the initial metric to change not only by homotheties but also by diffeomorphisms, a solution g_t to the Ricci flow is said to be *self-similar* if there exists a positive function $\sigma(t)$ and a one-parameter group of diffeomorphisms $\psi_t \colon M \to M$ such that

$$g_t = \sigma(t)\psi_t^* g_0. \tag{II.2}$$

Remark 3.17. Assume that Equation (II.2) determines a solution to the Ricci flow and differentiate it to obtain

$$\frac{\partial}{\partial t}g_t = -2\rho_{g_t} = \frac{d\sigma}{dt}(t)\psi_t^*g_0 + \sigma(t)\psi_t^*\left(\mathcal{L}_X g_0\right),\tag{II.3}$$

where \mathcal{L} denotes the Lie derivative and X is the time-dependent vector field given by

$$X(\psi_t(p)) = \frac{d}{dt}(\psi_t(p))$$

for any $p \in M$. Now, since $\rho_{g_t} = \psi_t^* \rho g_0$, we can actually drop the pull-back in Equation (II.3) and so

$$-2\rho_{g_0} = \frac{d\sigma}{dt}(t)g_0 + \mathcal{L}_{\tilde{X}}g_0, \qquad (\text{II.4})$$

where $\tilde{X}(t) = \sigma(t)X(t)$. If we now set

$$\mu = -\frac{1}{2} \frac{d\sigma}{dt}\Big|_{t=0}$$
 and $X_0 = \tilde{X}(0),$

Equation (II.4) becomes

$$-2\rho_{g_0} = -2\mu g_0 + 2\mathcal{L}_{X_0} g_0$$

at t = 0. This proves that for any self-similar solution to the Ricci flow, there exists a vector field X on M such that

$$\mathcal{L}_X g + \rho = \mu g.$$

Conversely, let X be a complete vector field on a pseudo-Riemannian manifold (M, g) and denote by $\psi_t \colon M \to M$ the family of diffeomorphisms generated by X according to

$$\frac{\partial}{\partial t}\psi_t(p) = \frac{1}{1-2\mu t}X(\psi_t(p)) \text{ and } \psi(0) = \mathrm{Id}_M,$$

which is defined for all $t \in (-\infty, \frac{1}{2\mu})$ if $\mu > 0$ and for all $t \in (\frac{1}{2\mu}, \infty)$ if $\mu < 0$. If we now consider the one-parameter family of metrics

$$g_t = (1 - 2\mu t)\psi_t^* g,$$

then

$$\frac{\partial}{\partial t}g_t = -2\mu\psi_t^*g + (1-2\mu)\psi_t^*\mathcal{L}_{\frac{1}{1-2\mu t}X}g$$
$$= \psi_t^*\left(-2\mu g + \mathcal{L}_{X(\psi_t(p))}g\right).$$

Now, if $\mathcal{L}_X g + \rho = \mu g$, then

$$\frac{\partial}{\partial t}g_t = \psi_t^*(-2\rho) = -2\psi_t^*\rho = -2\rho(\psi_t^*g) = -2\rho(g_t),$$

and so g_t is a solution to the Ricci flow.

The remark above shows that there exists a correspondence between self-similar solutions to the Ricci flow and what are known as Ricci solitons.

Ricci solitons

A triple (M, g, X) where (M, g) is a pseudo-Riemannian manifold and X is a vector field on M satisfying the differential equation

$$\mathcal{L}_X g + \rho = \mu g, \quad \mu \in \mathbb{R}, \tag{II.5}$$

is called a *Ricci soliton*. These metrics not only generalize Einstein metrics but also are selfsimilar solutions to the Ricci flow and conversely, thus corresponding to geometric fixed points of the flow (modulo homotheties and diffeomorphisms). A Ricci soliton is said to be *expanding*, *steady*, or *shrinking* if the soliton constant $\mu < 0$, $\mu = 0$ or $\mu > 0$, respectively. Besides, if the soliton vector field X is the gradient of some potential function $f: M \to \mathbb{R}$, then the soliton is said to be a *gradient Ricci soliton*. In this case $\mathcal{L}_X g_0 = 2 \text{Hes}_{g_0}(f)$ and so

$$\operatorname{Hes}_{g_0}(f) + \rho_{g_0} = \mu g_0. \tag{II.6}$$

A Ricci soliton is said to be *trivial* if the corresponding pseudo-Riemannian metric is Einstein, in which case one may solve Equation (II.5) just by setting X = 0. It immediately follows from (II.5) that two Ricci soliton vector fields X_1 and X_2 on a given manifold (M, g) differ in a homothetic vector field $\xi = X_1 - X_2$. While the existence of homothetic vector fields is a very rigid condition in the positive definite case, Lorentzian manifolds may admit homothetic vector fields without being flat. Moreover, the Ricci soliton equation (II.5) is invariant by homotheties in the sense that (M, g, X) is a Ricci soliton with soliton constant μ if and only if $(M, \kappa g, \frac{1}{\kappa}X)$ is a Ricci soliton with soliton constant $\frac{\mu}{\kappa}$ for any $\kappa > 0$. Therefore, we develop our study modulo homotheties. We refer to [49] for more information.

The scalar curvature of a Ricci soliton satisfies

$$2\Delta \tau = \mu \tau - \|\rho\|^2 + \langle X, \nabla \tau \rangle$$

(see for example [47]). Therefore, homogeneous steady Ricci solitons are necessarily Ricci flat, and therefore flat. As a consequence of local homogeneity, a shrinking Ricci soliton is necessarily a gradient (see [105]).

A gradient Ricci soliton (M, g, f) is *rigid* if the manifold (M, g) splits isometrically as a product $N \times \mathbb{R}^k$, where the non-Euclidean factor (N, g_N) is Einstein and the potential function is given by the norm of the projection on the Euclidean factor, $f = \frac{\mu}{2} ||\pi_{\mathbb{R}^k}||^2$. Besides, the soliton constant is given by $\rho_N = \mu g_N$. It was shown in [121] that homogeneous gradient Ricci solitons are necessarily rigid, and Arroyo and Lafuente showed in [9] that any four-dimensional homogeneous expanding Ricci soliton is homothetic to an algebraic Ricci soliton. Hence, any four-dimensional homogeneous Ricci soliton is either Einstein, or an algebraic Ricci soliton or homothetic to a product $\mathbb{S}^2 \times \mathbb{R}^2$.

Algebraic Ricci solitons

A metric Lie group $(G, \langle \cdot, \cdot \rangle)$ is an *algebraic Ricci soliton* if its Ricci operator satisfies

$$\operatorname{Ric} = \mu \operatorname{Id} + \mathfrak{D}$$

for some derivation \mathfrak{D} of the corresponding Lie algebra (see [97]). Algebraic Ricci solitons are critical points of the scalar curvature for an appropriately restricted family of metrics [97], and they are critical for a quadratic curvature functional with zero energy in dimensions three and four [23]. On the other hand, Ricci solitons on Lie groups with left-invariant soliton vector field are not necessarily critical for any quadratic curvature functional, thus being of a different nature.

If a simply connected pseudo-Riemannian Lie group $(G, \langle \cdot, \cdot \rangle)$ is an algebraic Ricci soliton, then it is a Ricci soliton (see [133]). Indeed, let $\{\varphi_t : G \to G\}$ be the one-parameter group of automorphisms of G determined by

$$d(\varphi_t)_e = \operatorname{Exp}(\frac{t}{2}\mathfrak{D}),$$

where $\mathfrak{D} = \operatorname{Ric} - \mu \operatorname{Id}$ is the derivation of the Lie algebra determining the algebraic Ricci soliton. Define a vector field X on G as the infinitesimal generator of $\{\varphi_t\}$, i.e.,

$$X(p) = \left. \frac{d}{dt} \right|_{t=0} \varphi_t(p)$$

for any $p \in G$. Then

$$\begin{aligned} (\mathcal{L}_X \langle \cdot, \cdot \rangle)(e_i, e_j) &= \frac{d}{dt} (\varphi_t^* \langle \cdot, \cdot \rangle)(e_i, e_j) = \frac{1}{2} \left\{ \langle \mathfrak{D} e_i, e_j \rangle + \langle e_i, \mathfrak{D} e_j \rangle \right\} \\ &= \frac{1}{2} \left\{ \langle \operatorname{Ric}(e_i), e_j \rangle + \langle e_i, \operatorname{Ric}(e_j) \rangle \right\} - \mu \langle e_i, e_j \rangle \\ &= (\rho - \mu \langle \cdot, \cdot \rangle)(e_i, e_j), \end{aligned}$$

from where it follows that $\mathcal{L}_X\langle\cdot,\cdot\rangle - \rho = -\mu\langle\cdot,\cdot\rangle$. Replacing X by -X one gets the Ricci soliton equation (II.5).

The Bach flow: Bach solitons

The Bach tensor $\mathfrak{B} = \operatorname{div}_2 \operatorname{div}_4 W + \frac{1}{2}W[\rho]$ is conformally invariant, trace-free and divergence-free in dimension four.

The Bach flow is the differential equation

$$\frac{\partial}{\partial t}g_t = \mathfrak{B}_{g_t} + \frac{1}{12}(\Delta \tau_{g_t})g_t \tag{II.7}$$

where g_t is a one-parameter family of pseudo-Riemannian metrics on a manifold M. For any given differentiable metric g_0 on a closed manifold there exists a unique solution g_t with $t \in [0, \varepsilon)$ for some $\varepsilon > 0$, to the Bach flow equation such that $g_t|_{t=0} = g_0$ (see [11, 12]).

Given a homothetic transformation $\overline{g} = \lambda g$ of a pseudo-Riemannian metric on a manifold M, the corresponding Bach tensor rescales as $\overline{\mathfrak{B}} = \lambda^{-1}\mathfrak{B}$ and so the right-hand side in (II.7) is homogeneous of degree d = -1 under homotheties. Therefore, there is a one-to-one correspondence between self-similar solutions of the Bach flow and *Bach solitons*, i.e., (M, g, X, μ) such that

$$\mathcal{L}_X g + \left(\mathfrak{B} + \frac{1}{2}\Delta au g\right) = \mu g,$$

just like in the Ricci flow case (see [133]).

In the homogeneous case – or, in a more general situation, if the scalar curvature is constant –, Equation (II.7) becomes

$$\frac{\partial}{\partial t}g_t = \mathfrak{B}_{g_t},$$
 (II.8)

and the corresponding Bach solitons are determined by triples (M, g, X) so that

$$\mathcal{L}_X g + \mathfrak{B} = \mu g, \tag{II.9}$$

where X is a vector field on M and $\mu \in \mathbb{R}$. Following the terminology we introduced when we talked about Ricci solitons, a Bach soliton is said to be expanding, steady or shrinking if $\mu < 0$, $\mu = 0$ or $\mu > 0$, respectively. Moreover, the soliton is said to be a *gradient* Bach soliton if the vector field is the gradient of a potential function $f, X = \frac{1}{2}\nabla f$.

We now assume that $\dim M = 4$, as it is the situation that concerns our study of the Bach flow. Since the Bach tensor is trace-free, it is easy to see that

$$\mu = \frac{1}{2} \operatorname{div} X$$

just by tracing (II.9).

Let X be a vector field on a Riemannian manifold (M,g) and let Ψ be a symmetric (0,2)-tensor field on M. Then

$$\langle \mathcal{L}_X g, \Psi \rangle = 2 \operatorname{div}(\iota_X \Psi) + 2(\operatorname{div} \Psi)(X).$$

Applying the identity above to a Bach soliton $(\mathcal{L}_X g + \mathfrak{B} = \mu g)$ and setting $\Psi = \mathfrak{B}$, one has that

$$0 = \langle \mathcal{L}_X g + \mathfrak{B} - \mu g, \mathfrak{B} \rangle = \langle \mathcal{L}_X g, \mathfrak{B} \rangle + \|\mathfrak{B}\|^2 - \mu \langle g, \mathfrak{B} \rangle$$
$$= 2 \operatorname{div}(\iota_X \mathfrak{B}) + 2(\operatorname{div} \mathfrak{B})(X) + \|\mathfrak{B}\|^2 - \mu \operatorname{tr}_g \mathfrak{B}$$
$$= 2 \operatorname{div}(\iota_X \mathfrak{B}) + \|\mathfrak{B}\|^2,$$

from where it follows that compact Bach solitons are necessarily Bach-flat – even if the scalar curvature is not constant, just replacing the constant μ by the function $\hat{\mu} = \mu - \frac{1}{2}\Delta\tau$ and proceeding as above (see also [81]).

Just like gradient Ricci solitons, gradient Bach solitons are rigid in the homogeneous case. This follows immediately from the following general result obtained by Petersen and Wylie in [122], just by applying it to $\tilde{q} = \mu g - \mathfrak{B}$.

Theorem 3.18. Let (M, g) be a homogeneous manifold and let \tilde{q} be a symmetric (0, 2)-tensor field which is divergence-free and invariant by isometries. If there is a non-constant function f satisfying the equation $\text{Hes}_f = \tilde{q}$, then (M, g) splits as a product $N \times \mathbb{R}^k$ and f is a function on the Euclidean factor.

Griffin showed in [81] that homogeneous gradient Bach solitons either are Bach-flat or split as a product $N \times \mathbb{R}^k$, where the potential function depends only on the Euclidean factor and (N, g_N) is a homogeneous manifold. Moreover, homogeneous steady gradient Bach solitons are necessarily Bach-flat. The existence of gradient Bach solitons on those products were considered in [81], where it is shown that

 Non-Bach-flat homogeneous gradient shrinking Bach solitons reduce to the products S² × [∞] R² and H² × R².
 • The only non-Bach-flat homogeneous expanding Bach soliton on a product $N^3 \times \mathbb{R}$ corresponds to $\mathbb{S}^3 \times \mathbb{R}$, where the metric on \mathbb{S}^3 is not the round metric, but a Berger one.

The non-gradient case is more subtle. For instance, the existence of non-Bach-flat homogeneous steady Bach solitons is still an open question. We provide a partial answer in Chapter 5, where we show that steady algebraic Bach solitons are necessarily Bach-flat.

Let $(G, \langle \cdot, \cdot \rangle)$ be a Lie group equipped with a left-invariant Riemannian metric. We say that $(G, \langle \cdot, \cdot \rangle)$ is an *algebraic Bach soliton* if $\mathfrak{D} = \mathfrak{B} - \mu \mathrm{Id}$ is a derivation of the Lie algebra \mathfrak{g} of G for some $\mu \in \mathbb{R}$, where \mathfrak{B} is the (1, 1)-tensor field associated to the Bach tensor.

As in the Ricci flow case, if a simply connected Riemannian Lie group $(G, \langle \cdot, \cdot \rangle)$ is an algebraic Bach soliton, then it is a Bach soliton and thus a self-similar solution of the homogeneous Bach flow (see [133]). Indeed, let $\{\varphi_t : G \to G\}$ be the one-parameter family of automorphisms of G determined by

$$d(\varphi_t)_e = \operatorname{Exp}(\frac{t}{2}\mathfrak{D}),$$

where $\mathfrak{D} = \widehat{\mathfrak{B}} - \mu$ Id is the derivation of the Lie algebra determining the algebraic Bach soliton. Define a vector field X on G as the infinitesimal generator of $\{\varphi_t\}$, i.e., $X(p) = \frac{d}{dt}\Big|_{t=0} \varphi_t(p)$ for any $p \in G$. Then one has that

$$\begin{aligned} (\mathcal{L}_X \langle \cdot, \cdot \rangle)(e_i, e_j) &= \frac{d}{dt} (\varphi_t^* \langle \cdot, \cdot \rangle)(e_i, e_j) = \frac{1}{2} \left\{ \langle \mathfrak{D} e_i, e_j \rangle + \langle e_i, \mathfrak{D} e_j \rangle \right\} \\ &= \frac{1}{2} \left\{ \langle \widehat{\mathfrak{B}} e_i, e_j \rangle + \langle e_i, \widehat{\mathfrak{B}} e_j \rangle \right\} - \mu \langle e_i, e_j \rangle \\ &= (\mathfrak{B} - \mu \langle \cdot, \cdot \rangle)(e_i, e_j), \end{aligned}$$

from where it follows that $\mathcal{L}_X\langle\cdot,\cdot\rangle - \mathfrak{B} = -\mu\langle\cdot,\cdot\rangle$. Replacing X by -X one gets the homogeneous Bach soliton equation (II.9). Therefore, algebraic Bach solitons correspond to self-similar solutions to the homogeneous Bach flow (II.8) of the form $g_t = \sigma(t)\psi_t^*\langle\cdot,\cdot\rangle$, where $\sigma(t)$ is a positive real function and ψ_t is a one-parameter family of automorphisms of $(G, \langle\cdot, \cdot\rangle)$.

In Chapter 5 we determine all the algebraic Bach solitons, showing that they either are algebraic Ricci solitons or correspond to the above-mentioned gradient shrinking Bach soliton on $\mathbb{S}^3 \times \mathbb{R}$, or to a one-parameter family of left-invariant semi-direct extensions of the Heisenberg algebra (cf. Theorem 5.8). Due to the difficulty in manipulating the Bach tensor we approach the problem in an indirect way, considering the existence of general abstract algebraic *T*-solitons on each four-dimensional Lie group. The study of the existence of algebraic solitons associated to a generic (0, 2)-tensor field *T* provides us with some necessary conditions for the existence of such general *T*-solitons which turn out to be good enough to simplify the analysis of the algebraic Bach solitons.

Chapter 4 Ricci solitons on four-dimensional Lorentzian Lie groups

Non-trivial homogeneous Riemannian Ricci solitons are rigid or expanding. Moreover the expanding ones are necessarily algebraic in dimension four [9]. None of these four-dimensional Ricci solitons can be realized as a left-invariant vector field on a Lie group. Indeed it was shown in [58] that non-trivial left-invariant Ricci solitons do not exist on Riemannian unimodular Lie groups and there are no three-dimensional left-invariant Ricci solitons on Riemannian Lie groups. A straightforward calculation shows that no four-dimensional homogeneous Riemannian Ricci soliton may be determined by a left-invariant vector field unless they are trivial.

In sharp contrast, the Lorentzian signature does support such solitons (see [24]). In spite of this general discrepancy, there does exist a certain correspondence between Riemannian and Lorentzian algebraic Ricci solitons in the nilpotent case (see [135] for more information about this relation).

In this chapter we will give a complete classification of left-invariant Ricci solitons on fourdimensional Lorentzian Lie groups. After reviewing left-invariant Einstein metrics and plane waves, we will recall the situation in dimension three, which is much simpler than the fourdimensional one. The main result of this chapter, which is Theorem 4.2, provides a complete description, modulo homotheties, of non-trivial left-invariant Ricci solitons which are neither symmetric nor pp-waves. The symmetric case is treated in Remark 4.5 and the pp-wave Lie groups are considered in Section 4.4. The results in this chapter are contained in the work [69].

Einstein metrics on Lorentzian four-dimensional Lie groups

While four-dimensional homogeneous Einstein metrics are locally symmetric in the Riemannian setting [88], the Lorentzian signature allows other possibilities. Left-invariant Einstein metrics on four-dimensional Lorentzian Lie groups were studied in [36] and a different approach shows that left-invariant Einstein metrics split into three categories: symmetric spaces, plane waves and left-invariant metrics which do not correspond to any of these.

Indecomposable locally symmetric Lorentzian spaces either are irreducible (and therefore of constant sectional curvature), or they correspond to Cahen-Wallach symmetric spaces [28], which are a special class of plane waves. Four-dimensional products $N^3 \times \mathbb{R}$ are Einstein if and only if they are flat and so the only decomposable four-dimensional Einstein Lorentzian symmetric spaces of non-constant sectional curvature are products $M_1(c) \times M_2(c)$ of two surfaces with the same constant sectional curvature. The other possibilities are covered by the following result (see [114]). **Theorem 4.1.** Let $(G, \langle \cdot, \cdot \rangle)$ be a four-dimensional Lie group with a left-invariant Einstein Lorentzian metric which is neither locally symmetric nor a plane wave. Then, it is locally homothetic to the Lie group determined by one of the following:

(i) The Ricci-flat semi-direct product $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_4] = -2e_1, \quad [e_2, e_4] = e_2 + \sqrt{3}e_3, \quad [e_3, e_4] = -\sqrt{3}e_2 + e_3, \quad or$$

(ii) the semi-direct product $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[u_1, u_4] = -u_1 + \delta u_2, \quad [u_2, u_4] = 5u_2, \quad [u_3, u_4] = 2u_3, \quad \delta \neq 0, \quad or$$

(iii) the semi-direct product $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[u_1, u_4] = 4u_1, \quad [u_2, u_4] = -2u_2 + \delta u_3, \quad [u_3, u_4] = \delta u_1 + u_3, \quad \delta \neq 0,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis with e_3 timelike, and $\{u_1, u_2, u_3, u_4\}$ is a pseudoorthonormal basis with $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$.

The curvature operator $\mathcal{R} : \Lambda^2 \to \Lambda^2$ of the metrics corresponding to Assertion (*i*) has real and complex eigenvalues, and $\|\nabla R\|^2 \neq 0$. The metrics corresponding to Assertion (*ii*) have scalar curvature $\tau = -48$ and their Weyl curvature operator is two-step nilpotent. Moreover, they are locally isometric to the only non-reductive homogeneous space which is Einstein but not of constant sectional curvature [35, 66]. The metrics corresponding to Assertion (*iii*) have scalar curvature $\tau = -12$ and their Weyl curvature operator is three-step nilpotent.

Homogeneous pp-waves and plane waves

Let (M, g, \mathcal{U}) be a Brinkmann wave, i.e., a Lorentzian manifold admitting a parallel degenerate line field \mathcal{U} . (M, g, \mathcal{U}) is said to be a *pp-wave* if the parallel line field is locally generated by a parallel null vector field and (M, g) is transversally flat, i.e., its curvature tensor satisfies R(X, Y) = 0 for all $X, Y \in \mathcal{U}^{\perp}$. In such case there exist local coordinates (u, v, x^1, x^2) so that

$$g = du \circ dv + H(v, x^1, x^2) dv \circ dv + dx^1 \circ dx^1 + dx^2 \circ dx^2.$$

Leistner showed in [100] that a Brinkmann wave (M, g, U) is a pp-wave if and only if it is transversally flat and Ricci isotropic, i.e., g(Ric X, Ric X) = 0 for any vector field X on M.

A pp-wave is said to be a *plane wave* if the covariant derivative of the curvature tensor satisfies $\nabla_X R = 0$ for all $X \in \mathcal{U}^{\perp}$. In this case the local coordinates above can be specialized so that $H(v, x^1, x^2) = a_{ij}(v)x^ix^j$. The Ricci operator of any pp-wave is two-step nilpotent and the metric is Ricci-flat if $\Delta_x H = 0$, where $\Delta_x = \partial_{x^1x^1} + \partial_{x^2x^2}$ is the spacelike Laplacian. It was shown in [78] that locally homogeneous Ricci-flat pp-waves are plane waves in the four-dimensional case. Homogeneous steady Ricci solitons on pp-waves which are not plane waves are given in Section 4.4, where we show that the result in [78] does not extend to Ricci solitons.

Homogeneous plane waves of dimension four are described in terms of a 2 × 2 skewsymmetric matrix F and a 2 × 2 symmetric matrix A_0 so that the defining function $H(v, x^1, x^2)$ takes the form $H = \vec{x}^T A(v)\vec{x}$, where the matrix A(v) is given by (see [19])

$$A(v) = e^{vF} A_0 e^{-vF}$$
, or $A(v) = \frac{1}{(v+b)^2} e^{\log(v+b)F} A_0 e^{-\log(v+b)F}$.

Furthermore, the plane wave metric is Ricci-flat if and only if A_0 is trace-free.

The existence of Ricci solitons on plane waves was investigated in [25] where it is shown that any plane wave is a steady gradient Ricci soliton. Due to the existence of homothetic vector fields, one also has the existence of expanding and shrinking Ricci solitons on some special classes of plane waves. In any case, the soliton vector field needs not be left-invariant for a plane wave Lie group, and hence the existence of left-invariant Ricci solitons on plane wave Lie groups will be considered in Section 4.4.

Left-invariant Ricci solitons on three-dimensional Lorentzian Lie groups

Non-trivial three-dimensional left-invariant Ricci solitons are either non-symmetric pp-waves or locally isometric to a left-invariant metric on G = O(1, 2), the universal cover of $SL(2, \mathbb{R})$ or the non-unimodular semi-direct extension $\mathbb{R}^2 \rtimes \mathbb{R}$, respectively given by the Lorentzian Lie algebras

(i)
$$[u_1, u_2] = \lambda u_3$$
, $[u_1, u_3] = -\lambda u_1 \mp u_2$, $[u_2, u_3] = \lambda u_2$, $\lambda \neq 0$,
(ii) $[u_1, u_2] = u_1 + \lambda u_3$, $[u_1, u_3] = -\lambda u_1$, $[u_2, u_3] = \lambda u_2 + u_3$, $\lambda \neq 0$,
(iii) $[e_1, e_3] = e_1 - e_2$, $[e_2, e_3] = e_1 + e_2$

where $\{u_1, u_2, u_3\}$ is a pseudo-orthonormal basis with $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = 1$, and $\{e_1, e_2, e_3\}$ is an orthonormal basis with timelike e_1 .

The three-dimensional Lorentzian Lie groups corresponding to cases (i) and (ii) have a single Ricci curvature which is either a double or a triple root of the minimal polynomial of the Ricci operator (see [24]). Moreover, the Lie group corresponding to (iii), which was omitted in [24], has complex Ricci curvatures $-2 \pm 2i$.

There are two different possibilities for three-dimensional left-invariant pp-waves which are Ricci solitons: a locally conformally flat plane wave (which is locally isometric to a \mathcal{P}_c -space), or a pp-wave that is locally isometric to a \mathcal{N}_b -space. We recommend to consult [76] for a classification of homogeneous pp-waves in dimension three, definitions of \mathcal{P}_c and \mathcal{N}_b -spaces and more details.

Left-invariant Ricci solitons on four-dimensional Lorentzian Lie groups

As in the Einstein case, the four-dimensional situation is more complicated than the corresponding three-dimensional one. We consider the case of left-invariant Ricci solitons on pp-wave Lie groups separately in Section 4.4. The remaining possibilities are given the main result of this chapter as follows. **Theorem 4.2.** A non-symmetric four-dimensional Lorentzian Lie group which is not a pp-wave is a non-trivial left-invariant Ricci soliton if and only if it is homothetic to one of the following:

(i) $G_{\alpha} = \mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_4] = \alpha e_1, \ [e_2, e_4] = \varepsilon \left(1 - \frac{\alpha^2}{2}\right)^{\frac{1}{2}} e_2 - e_3, \ [e_3, e_4] = e_2 + \varepsilon \left(1 - \frac{\alpha^2}{2}\right)^{\frac{1}{2}} e_3,$$

where the parameter satisfies $0 \le \alpha \le \sqrt{2}$ and $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis with timelike e_3 . If $\alpha = 0$, then $\varepsilon = 1$. If $0 < \alpha < \sqrt{2}$, then $\varepsilon^2 = 1$. In the latter case, $\alpha \ne \frac{2}{\sqrt{3}}$ when $\varepsilon = -1$.

(ii) $G_{\alpha} = \mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[u_1, u_4] = \alpha u_1, \quad [u_2, u_4] = -\alpha u_2 + u_3, \quad [u_3, u_4] = u_1, \qquad \alpha > 0,$$

where $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis for which the non-zero inner products are $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$.

(iii) $G = E(1,1) \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_2, e_4] = -[e_1, e_2] = e_2, \quad [e_1, e_3] = [e_3, e_4] = \frac{1}{2}[e_1, e_4] = e_3,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis with timelike e_3 .

(iv) $G_{\alpha\beta} = E(1,1) \rtimes \mathbb{R}$ with Lie algebra given by

$$[u_1, u_2] = u_1,$$
 $[u_1, u_4] = -2\alpha(\alpha\beta + 1)u_1,$ $[u_2, u_3] = u_3,$
 $[u_2, u_4] = \beta u_1,$ $[u_3, u_4] = \alpha u_3,$

where $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis for which the non-zero inner products are $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$, and the parameters $\alpha > 0$ and $\alpha\beta \notin \{-2, -1, -\frac{1}{2}\}$.

Remark 4.3. The left-invariant Ricci solitons corresponding to G_{α} in Assertion (*i*) are steady and their left-invariant soliton vector field is defined by $X = X_1e_1 + e_4$ if the parameter $\alpha = 0$, and by $X = \frac{1}{2} \left(\alpha + \varepsilon \sqrt{4 - 2\alpha^2} \right) e_4$ otherwise. Moreover, the Ricci operator has eigenvalues

$$\xi_{1} = 0, \qquad \xi_{2} = -\alpha \left(\alpha + \varepsilon \left(4 - 2\alpha^{2} \right)^{\frac{1}{2}} \right),$$

$$\xi_{3} = \alpha^{2} - 2 - \varepsilon \alpha \left(1 - \frac{\alpha^{2}}{2} \right)^{\frac{1}{2}} + \left(\alpha^{2} - 4 - 2\varepsilon \alpha \left(4 - 2\alpha^{2} \right)^{\frac{1}{2}} \right)^{\frac{1}{2}},$$

$$\xi_{4} = \alpha^{2} - 2 - \varepsilon \alpha \left(1 - \frac{\alpha^{2}}{2} \right)^{\frac{1}{2}} - \left(\alpha^{2} - 4 - 2\varepsilon \alpha \left(4 - 2\alpha^{2} \right)^{\frac{1}{2}} \right)^{\frac{1}{2}}.$$

Therefore, the Ricci curvatures are $\{0, 0, -2 \pm 2i\}$ if $\alpha = 0$, $\{0, \lambda, \alpha \pm \beta i\}$ with $\lambda \alpha \beta \neq 0$ if $0 < \alpha < \sqrt{2}$, and $\{0, -2, \pm \sqrt{2}i\}$ if $\alpha = \sqrt{2}$.

The left-invariant Ricci solitons corresponding to G_{α} in Assertion (*ii*) are steady and their left-invariant soliton vector field is defined by $X = X_1u_1 - X_1\alpha u_3 - \frac{1}{2}\alpha u_4$. Moreover, their Ricci operator is three-step nilpotent.

The left-invariant Ricci solitons corresponding to Assertion (*iii*) are steady and their left-invariant soliton vector field is defined by $X = -\frac{1}{2}e_1 + \frac{3}{2}e_4$. Furthermore, their Ricci operator has eigenvalues $\{0, -2, -2 \pm \sqrt{6}i\}$.

The left-invariant Ricci solitons corresponding to $G_{\alpha\beta}$ in Assertion (*iv*) are expanding with $\mu = -(2(\alpha\beta + 1)^2 + 1)\alpha^2$ and their left-invariant soliton vector field is defined by $X = X_1u_1 + X_2u_2 + X_4u_4$, where

$$X_1 = \frac{1}{2(2\alpha\beta+1)}(\alpha\beta+2)(2(\alpha\beta+1)\alpha\beta-1),$$

$$X_2 = \frac{1}{2\alpha\beta+1}(\alpha\beta+2)(2(\alpha\beta+2)\alpha\beta+3)\alpha^2,$$

$$X_4 = \frac{1}{2\alpha\beta+1}(\alpha\beta+2)^2\alpha.$$

Moreover, their Ricci operator is diagonalizable with non-zero real eigenvalues

$$\xi_1 = \xi_2 = -(2\alpha\beta + 1)(\alpha\beta + 1)\alpha^2, \xi_3 = (2\alpha\beta + 1)\alpha^2, \qquad \xi_4 = -(2(\alpha\beta + 2)\alpha\beta + 3)\alpha^2.$$

Remark 4.4. Let $(G_1, \langle \cdot, \cdot \rangle_1)$ and $(G_2, \langle \cdot, \cdot \rangle_2)$ be two Lorentzian Lie groups with non-zero scalar curvatures. If $(G_1, \langle \cdot, \cdot \rangle_1)$ and $(G_2, \langle \cdot, \cdot \rangle_2)$ are homothetic, then

$$au_1^{-2} \|R_1\|^2 = au_2^{-2} \|R_2\|^2$$
 and $au_1^{-2} \|W_1\|^2 = au_2^{-2} \|W_2\|^2$,

where R_i and W_i denote the curvature tensor and the Weyl conformal curvature tensor for i = 1, 2, respectively. We use the quadratic scalar curvature invariants to show that left-invariant metrics in different assertions in Theorem 4.2 correspond to distinct homothetic classes. It also follows that different values of the parameter in Assertion (*i*) determine distinct homothetic classes. Metrics in Assertion (*iv*) with different $\alpha\beta$ correspond to distinct homothetic classes.

Remark 4.5. Locally symmetric Lorentzian spaces which are neither of constant sectional curvature nor a Cahen-Wallach symmetric space split as a product [28]. Besides, left-invariant symmetric Ricci solitons which are neither Einstein nor plane waves are locally isometric to $\mathbb{L}^2 \times N(c)$, where N(c) is a surface of constant curvature, and correspond to one of the following Lie groups:

- $G_{\alpha\beta}$ in Assertion (*iv*) of Theorem 4.2 for $\alpha\beta = -1$, as discussed in Section 4.1.4.
- The Lie group $H^3 \rtimes \mathbb{R}$ determined by the Lie algebra

$$[u_1, u_2] = \lambda_1 u_1, \qquad [u_1, u_4] = -\frac{\gamma_3 \lambda_1^2}{\gamma_4} u_1,$$
$$[u_2, u_4] = \gamma_3 \lambda_1^2 u_3, \qquad [u_3, u_4] = \gamma_4 \lambda_1 u_3,$$

where $\lambda_1 \gamma_4 \neq 0$ and $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis such that $\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle = 1$. This is a expanding Ricci soliton with $\mu = -\lambda_1^2$ and left-invariant soliton vector field $X = -\frac{\gamma_3 \lambda_1^2}{\gamma_4^2} u_2 + \frac{\gamma_3^2 \lambda_1^3}{2\gamma_4^3} u_3 - \frac{\lambda_1}{\gamma_4} u_4$, as discussed in Section 4.3.2.

Remark 4.6. The Bach tensor of a four-dimensional manifold is defined by

$$\mathfrak{B} = \operatorname{div}_2 \operatorname{div}_4 W + \frac{1}{2} W[\rho].$$

Four-dimensional Bach-flat metrics are conformally invariant and Bach-flatness is a necessary condition for being conformally Einstein. The left-invariant metrics in Theorem 4.2 are Bach-flat if and only if they correspond to Assertion (*iv*) when the parameters satisfy $\alpha\beta = -\frac{5}{4}$. In this case, the vector field $X = \frac{3}{2}u_1 - \frac{3\alpha}{2}u_4$ is locally a gradient and satisfies $\operatorname{div}_4 W + \frac{1}{2}W(\cdot, \cdot, \cdot, X) = 0$. A straightforward calculation shows that the Weyl operator acting on the space of two-forms has non-zero eigenvalues and thus the metric is weakly-generic. Therefore, it is conformally Einstein (see [94] for more information).

Remark 4.7. The metrics in Theorem 4.2-(*i*) are critical for the functional given by the L^2 -norm of the Ricci tensor or, equivalently \mathcal{F}_0 -critical with zero energy. These metrics also are steady algebraic Ricci solitons, which do not exist in the Riemannian situation.

The scalar curvature of the metrics in Theorem 4.2-(*ii*) is zero, so they are S-critical, but they are not critical for any other curvature quadratic functional and nor are they algebraic Ricci solitons.

The metrics in Theorem 4.2-(*iii*) are never critical for any curvature quadratic functional and they are not algebraic Ricci solitons either.

The metrics in Theorem 4.2-(*iv*) are \mathcal{F}_t -critical with zero energy for

$$t = -\frac{2\alpha^2\beta^2 + 4\alpha\beta + 3}{6\alpha^2\beta^2 + 8\alpha\beta + 4},$$

and they are never algebraic Ricci solitons.

Left-invariant metrics and Gröbner bases

Connected and simply connected four-dimensional Lie groups are either products $SU(2) \times \mathbb{R}$, $\widetilde{SL}(2, \mathbb{R}) \times \mathbb{R}$, or one of the solvable semi-direct extensions of three-dimensional unimodular Lie groups $\widetilde{E}(2) \rtimes \mathbb{R}$, $E(1,1) \rtimes \mathbb{R}$, $H^3 \rtimes \mathbb{R}$ or $\mathbb{R}^3 \rtimes \mathbb{R}$, where $\widetilde{E}(2)$, E(1,1), H^3 and \mathbb{R}^3 denote the Euclidean, the Poincaré, the Heisenberg and the Abelian three-dimensional Lie algebras, respectively. Since our purpose is to investigate left-invariant Ricci solitons, we work at the purely algebraic level, and therefore we restrict to the corresponding Lie algebras. Left-invariant Riemannian metrics are described, using the work of Milnor [104], in terms of the corresponding derivations on the three-dimensional unimodular Lie subalgebras. The Lorentzian situation is more subtle due to the fact that the restriction of the metric to the three-dimensional subalgebras $\mathfrak{su}(2)$, $\mathfrak{sl}(2,\mathbb{R})$, $\mathfrak{e}(2)$, $\mathfrak{e}(1,1)$, \mathfrak{h} or \mathfrak{r}^3 may be a positive definite, Lorentzian or degenerate inner product. We follow [33] and consider separately the three possibilities above.

Let $(G, \langle \cdot, \cdot \rangle)$ be a four-dimensional Lorentzian Lie group and let X be a left-invariant vector field on G. Then $(G, \langle \cdot, \cdot \rangle, X, \mu)$ is a left-invariant Ricci soliton if and only if the symmetric tensor field $\frac{1}{2}\mathfrak{P} = \mathcal{L}_X \langle \cdot, \cdot \rangle + \rho - \mu \langle \cdot, \cdot \rangle$ vanishes identically. It is now immediate, since the vector field X is left-invariant, that the condition $\mathfrak{P} = 0$ equals to a system of polynomial equations on the structure constants which needs to be solved in order to obtain a complete classification. The theory of Gröbner bases provides a well-known strategy to solve rather large polynomial systems obtaining "better" polynomials that belong to the ideal generated by the initial polynomial system. We make use of Gröbner bases to show non-existence results in some cases. We refer to Section 1.6 and [53] for more information on Gröbner bases.

4.1 Extensions of Lorentzian Lie groups

Let $(G, \langle \cdot, \cdot \rangle)$ be a four-dimensional Lorentzian Lie group of the form $G_3 \rtimes \mathbb{R}$ so that the restriction of the metric to the three-dimensional subalgebra \mathfrak{g}_3 is Lorentzian. Three-dimensional unimodular Lie algebras are completely described by a Milnor-type frame associated to the selfdual structure tensor L given by

$$L(X \times Y) = [X, Y],$$

where "×" denotes the vector-cross product $\langle X \times Y, Z \rangle = \det(X, Y, Z)$. The self-duality of L ensures the existence of an orthonormal basis $\{e_1, e_2, e_3\}$ of \mathfrak{g}_3 that diagonalizes the structure tensor in the positive definite case [104]. If the inner product is of Lorentzian signature, then L may have non-trivial Jordan normal form as follows (see, for example [112]).

- Ia. L is real diagonalizable. Therefore, there exists an orthonormal basis $\{e_1, e_2, e_3\}$, where we assume e_3 to be timelike, so that $L(e_i) = \lambda_i e_i$.
- **Ib.** L has complex eigenvalues. Consequently, there exists an orthonormal basis $\{e_1, e_2, e_3\}$, where we assume e_3 to be timelike, so that

$$L = \begin{pmatrix} \lambda & 0 & 0 \\ 0 & \alpha & \beta \\ 0 & -\beta & \alpha \end{pmatrix}, \quad \beta \neq 0$$

II. L has a double root of its minimal polynomial. Then there exists a pseudo-orthonormal basis $\{u_1, u_2, u_3\}$ so that

$$L = \begin{pmatrix} \lambda_1 & 0 & 0 \\ \varepsilon & \lambda_1 & 0 \\ 0 & 0 & \lambda_2 \end{pmatrix}, \quad \varepsilon = \pm 1, \quad \text{where} \quad \langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = 1.$$

III. L has a triple root of its minimal polynomial. Then there exists a pseudo-orthonormal basis $\{u_1, u_2, u_3\}$ so that

$$L = \begin{pmatrix} \lambda & 0 & 1 \\ 0 & \lambda & 0 \\ 0 & 1 & \lambda \end{pmatrix}, \text{ where } \langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = 1.$$

In what follows, we set $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ and L denotes the structure operator of the unimodular subalgebra \mathfrak{g}_3 . We follow the work of Rahmani [124] to describe Lorentzian left-invariant metrics on \mathfrak{g}_3 and analyse the existence of left-invariant Ricci solitons on each of the possibilities above. It follows that all the left-invariant metrics in Theorem 4.2 are realized as extensions of unimodular Lorentzian Lie groups.

4.1.1 The structure operator L is diagonalizable

There exists an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$, with timelike e_3 , where $\mathfrak{g}_3 = \operatorname{span}\{e_1, e_2, e_3\}$ and $\mathfrak{r} = \operatorname{span}\{e_4\}$, so that

$$[e_1, e_2] = -\lambda_3 e_3, \quad [e_1, e_3] = -\lambda_2 e_2, \quad [e_2, e_3] = \lambda_1 e_1, \quad [e_i, e_4] = \sum_{j=1}^3 \alpha_i{}^j e_j,$$

for certain $\alpha_i^{j} \in \mathbb{R}$ depending on the eigenvalues λ_i . The Jacobi identity leads to the following different possibilities.

Structure operator with non-zero eigenvalues on $\widetilde{SL}(2,\mathbb{R})\times\mathbb{R}$ or $SU(2)\times\mathbb{R}$

Assume $\lambda_1 \lambda_2 \lambda_3 \neq 0$. The left-invariant metrics on $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ or $SU(2) \times \mathbb{R}$ are described by the corresponding Lie algebra structure

$$[e_1, e_2] = -\lambda_3 e_3, \quad [e_1, e_3] = -\lambda_2 e_2, \qquad [e_1, e_4] = \gamma_1 \lambda_2 e_2 + \gamma_2 \lambda_3 e_3, \\ [e_2, e_3] = \lambda_1 e_1, \qquad [e_2, e_4] = -\gamma_1 \lambda_1 e_1 + \gamma_3 \lambda_3 e_3, \quad [e_3, e_4] = \gamma_2 \lambda_1 e_1 + \gamma_3 \lambda_2 e_2,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis. A straightforward calculation shows that a leftinvariant vector field $X = \sum_{\ell} X_{\ell} e_{\ell}$ is a Ricci soliton if and only if the tensor field $\frac{1}{2}\mathfrak{P} = \mathcal{L}_X \langle \cdot, \cdot \rangle + \rho - \mu \langle \cdot, \cdot \rangle$ vanishes identically. Equivalently $\{\mathfrak{P}_{ij} = 0\}$, where the polynomials \mathfrak{P}_{ij} are given by

$$\begin{split} \mathfrak{P}_{11} &= (\gamma_1^2 - \gamma_2^2 - 1)\lambda_1^2 - (\gamma_1^2 - 1)\lambda_2^2 + (\gamma_2^2 + 1)\lambda_3^2 - 2\lambda_2\lambda_3 - 2\mu, \\ \mathfrak{P}_{12} &= \gamma_2\gamma_3(\lambda_3^2 - \lambda_1\lambda_2) - 2(X_4\gamma_1 - X_3)(\lambda_1 - \lambda_2), \\ \mathfrak{P}_{13} &= -\gamma_1\gamma_3(\lambda_2^2 - \lambda_1\lambda_3) + 2(X_4\gamma_2 - X_2)(\lambda_1 - \lambda_3), \\ \mathfrak{P}_{14} &= \gamma_3(\lambda_2 - \lambda_3)^2 + 2(X_2\gamma_1 - X_3\gamma_2)\lambda_1, \\ \mathfrak{P}_{22} &= -(\gamma_1^2 - 1)\lambda_1^2 + (\gamma_1^2 - \gamma_3^2 - 1)\lambda_2^2 + (\gamma_3^2 + 1)\lambda_3^2 - 2\lambda_1\lambda_3 - 2\mu, \\ \mathfrak{P}_{23} &= \gamma_1\gamma_2(\lambda_1^2 - \lambda_2\lambda_3) + 2(X_4\gamma_3 + X_1)(\lambda_2 - \lambda_3), \\ \mathfrak{P}_{24} &= -\gamma_2(\lambda_1 - \lambda_3)^2 - 2(X_1\gamma_1 + X_3\gamma_3)\lambda_2, \\ \mathfrak{P}_{33} &= -(\gamma_2^2 + 1)\lambda_1^2 - (\gamma_3^2 + 1)\lambda_2^2 + (\gamma_2^2 + \gamma_3^2 + 1)\lambda_3^2 + 2\lambda_1\lambda_2 + 2\mu, \\ \mathfrak{P}_{34} &= \gamma_1(\lambda_1 - \lambda_2)^2 + 2(X_1\gamma_2 + X_2\gamma_3)\lambda_3, \\ \mathfrak{P}_{44} &= -\gamma_1^2(\lambda_1 - \lambda_2)^2 + \gamma_2^2(\lambda_1 - \lambda_3)^2 + \gamma_3^2(\lambda_2 - \lambda_3)^2 - 2\mu \,. \end{split}$$

Since $\lambda_1 \lambda_2 \lambda_3 \neq 0$, we may assume $\lambda_1 = 1$ in our computations, which means we will be working with the homothetic metric determined by $\hat{e}_i = \frac{1}{\lambda_1} e_i$. Now, let $\mathcal{I} \subset \mathbb{R}[\gamma_1, \gamma_2, \gamma_3, \lambda_2, \lambda_3, \mu, X_1, X_2, X_3, X_4]$ be the ideal generated by the polynomials \mathfrak{P}_{ij} . We compute a Gröbner basis \mathcal{G} of \mathcal{I} with respect to the graded reverse lexicographical order and we see that the polynomials

$$\mathbf{g}_1 = \mu^2$$
 and $\mathbf{g}_2 = 4\lambda_2\lambda_3 + 3(\lambda_2 + \lambda_3 + 1)\mu$

belong to \mathcal{G} . Since $\lambda_2 \lambda_3 \neq 0$, there are no left-invariant Ricci solitons in this case.

Structure operator with a zero eigenvalue on $\tilde{E}(2) \rtimes \mathbb{R}$ or $E(1,1) \rtimes \mathbb{R}$

We distinguish two possibilities depending on the causality of ker L. If ker L is spacelike then either $\lambda_1 = 0$ or $\lambda_2 = 0$, while if ker L is timelike then $\lambda_3 = 0$. Next we will show that left-invariant Ricci solitons exist only in the flat case.

Structure operator L with spacelike kernel

Without loss of generality, we can assume $\lambda_1 = 0$ and $\lambda_2 \lambda_3 \neq 0$. Left-invariant metrics in this case are given by

$$\begin{split} [e_1, e_2] &= -\lambda_3 e_3, \\ [e_1, e_3] &= -\lambda_2 e_2, \\ [e_2, e_4] &= \gamma_3 e_2 + \gamma_4 \lambda_3 e_3, \\ [e_3, e_4] &= \gamma_4 \lambda_2 e_2 + \gamma_3 e_3, \end{split}$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis. We focus on the following components of the tensor field \mathfrak{P} :

$$\begin{aligned} \mathfrak{P}_{11} &= (\lambda_3 - \lambda_2)^2 - \gamma_1^2 + \gamma_2^2 - 2\mu, \\ \mathfrak{P}_{14} &= \gamma_4 (\lambda_3 - \lambda_2)^2, \\ \mathfrak{P}_{22} &= -(\gamma_4^2 + 1)(\lambda_2^2 - \lambda_3^2) + \gamma_1^2 - 4(\gamma_3 - X_4)\gamma_3 - 2\mu, \\ \mathfrak{P}_{33} &= -(\gamma_4^2 + 1)(\lambda_2^2 - \lambda_3^2) + \gamma_2^2 + 4(\gamma_3 - X_4)\gamma_3 + 2\mu, \\ \mathfrak{P}_{44} &= \gamma_4^2 (\lambda_3 - \lambda_2)^2 - \gamma_1^2 + \gamma_2^2 - 4\gamma_3^2 - 2\mu. \end{aligned}$$

It is easy to check that

$$\mathfrak{P}_{11} + \gamma_4 \mathfrak{P}_{14} - \mathfrak{P}_{44} = (\lambda_2 - \lambda_3)^2 + 4\gamma_3^2.$$

Therefore, $\lambda_3 = \lambda_2$ and $\gamma_3 = 0$. Now, we have

$$\mathfrak{P}_{22} + \mathfrak{P}_{33} = \gamma_1^2 + \gamma_2^2,$$

which implies $\gamma_1 = \gamma_2 = 0$, so the metric is flat.

Structure operator L with timelike kernel

If $\lambda_3 = 0$ and $\lambda_1 \lambda_2 \neq 0$ then left-invariant metrics are described by

$$[e_1, e_3] = -\lambda_2 e_2, \qquad [e_1, e_4] = \gamma_1 e_1 + \gamma_2 \lambda_2 e_2, \quad [e_2, e_3] = \lambda_1 e_1,$$
$$[e_2, e_4] = -\gamma_2 \lambda_1 e_1 + \gamma_1 e_2, \quad [e_3, e_4] = \gamma_3 e_1 + \gamma_4 e_2,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis. In this situation, a straightforward calculation leads to the following components of the tensor field \mathfrak{P} :

$$\begin{aligned} \mathfrak{P}_{11} &= (\gamma_2^2 - 1)(\lambda_1^2 - \lambda_2^2) - \gamma_3^2 - 4(\gamma_1 - X_4)\gamma_1 - 2\mu, \\ \mathfrak{P}_{34} &= \gamma_2(\lambda_1 - \lambda_2)^2, \\ \mathfrak{P}_{33} &= -(\lambda_1 - \lambda_2)^2 - \gamma_3^2 - \gamma_4^2 + 2\mu, \\ \mathfrak{P}_{44} &= -\gamma_2^2(\lambda_1 - \lambda_2)^2 - 4\gamma_1^2 + \gamma_3^2 + \gamma_4^2 - 2\mu. \end{aligned}$$

It is easy to see that

$$\mathfrak{P}_{33} + \gamma_2 \mathfrak{P}_{34} + \mathfrak{P}_{44} = -(\lambda_1 - \lambda_2)^2 - 4\gamma_1^2,$$

so $\lambda_2 = \lambda_1$ and $\gamma_1 = 0$. Now,

$$\mathfrak{P}_{11} + \mathfrak{P}_{33} = -2\gamma_3^2 - \gamma_4^2,$$

which implies that $\gamma_3 = \gamma_4 = 0$ and the metric is flat as in the previous case.

Structure operator of rank one: metrics on $H^3 \rtimes \mathbb{R}$

We will consider the cases when the restriction of the metric to ker L is positive definite ($\lambda_3 \neq 0$) or Lorentzian ($\lambda_3 = 0$) separately, and make use of Gröbner bases to show non-existence of left-invariant Ricci solitons in both cases.

Structure operator L with positive definite kernel

Setting $\lambda_1 = \lambda_2 = 0$ and $\lambda_3 \neq 0$ left-invariant metrics are described by

$$[e_1, e_2] = -\lambda_3 e_3, \qquad [e_1, e_4] = \gamma_1 e_1 + \gamma_2 e_2 + \gamma_3 e_3,$$
$$[e_2, e_4] = \gamma_4 e_1 + \gamma_5 e_2 + \gamma_6 e_3, \qquad [e_3, e_4] = (\gamma_1 + \gamma_5) e_3,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis. Now, a vector $X \in \mathfrak{h} \rtimes \mathbb{R}$ determines a leftinvariant Ricci soliton if and only if the system of polynomial equations given by $\{\mathfrak{P}_{ij} = 0\}$ is satisfied, where

$$\begin{split} \mathfrak{P}_{11} &= \lambda_3^2 - 4\gamma_1^2 - \gamma_2^2 + \gamma_3^2 + \gamma_4^2 - 4\gamma_1\gamma_5 + 4X_4\gamma_1 - 2\mu, \\ \mathfrak{P}_{12} &= -\gamma_1\gamma_2 - 3\gamma_1\gamma_4 - 3\gamma_2\gamma_5 + \gamma_3\gamma_6 - \gamma_4\gamma_5 + 2X_4(\gamma_2 + \gamma_4), \\ \mathfrak{P}_{13} &= 2X_2\lambda_3 + 2\gamma_1\gamma_3 + 3\gamma_3\gamma_5 - \gamma_4\gamma_6 - 2X_4\gamma_3, \\ \mathfrak{P}_{14} &= \gamma_6\lambda_3 - 2X_1\gamma_1 - 2X_2\gamma_4, \\ \mathfrak{P}_{22} &= \lambda_3^2 + \gamma_2^2 - \gamma_4^2 - 4\gamma_5^2 + \gamma_6^2 - 4\gamma_1\gamma_5 + 4X_4\gamma_5 - 2\mu, \\ \mathfrak{P}_{23} &= -2X_1\lambda_3 + 3\gamma_1\gamma_6 - \gamma_2\gamma_3 + 2\gamma_5\gamma_6 - 2X_4\gamma_6, \\ \mathfrak{P}_{24} &= -\gamma_3\lambda_3 - 2X_1\gamma_2 - 2X_2\gamma_5, \\ \mathfrak{P}_{33} &= \lambda_3^2 + 4\gamma_1^2 + \gamma_3^2 + 4\gamma_5^2 + \gamma_6^2 + 8\gamma_1\gamma_5 - 4X_4(\gamma_1 + \gamma_5) + 2\mu, \\ \mathfrak{P}_{34} &= 2\{X_3(\gamma_1 + \gamma_5) + X_1\gamma_3 + X_2\gamma_6\}, \\ \mathfrak{P}_{44} &= -4\gamma_1^2 - \gamma_2^2 + \gamma_3^2 - \gamma_4^2 - 4\gamma_5^2 + \gamma_6^2 - 4\gamma_1\gamma_5 - 2\gamma_2\gamma_4 - 2\mu. \end{split}$$

Since $\lambda_3 \neq 0$, we can assume $\lambda_3 = 1$ and work with the homothetic metric determined by $\hat{e}_i = \frac{1}{\lambda_3} e_i$. Let $\mathcal{I}_1 \subset \mathbb{R}[\gamma_1, \gamma_2, \gamma_3, \gamma_4, \gamma_5, \gamma_6, \mu, X_1, X_2, X_3, X_4]$ be the ideal generated by the polynomials \mathfrak{P}_{ij} . We compute a Gröbner basis \mathcal{G}_1 of \mathcal{I}_1 with respect to the lexicographical order and obtain that the polynomials

$$\begin{aligned} \mathbf{g}_{11} &= X_3 (2617344 X_4^8 + 13139712 X_4^6 + 18557248 X_4^4 + 7213356 X_4^2 + 61803), \\ \mathbf{g}_{12} &= X_4 (83755008 X_4^{14} + 776429568 X_4^{12} + 2689679360 X_4^{10} + 4517104000 X_4^8 \\ &\quad + 4237066048 X_4^6 + 2362718304 X_4^4 + 591574590 X_4^2 + 5006043) \end{aligned}$$
belong to \mathcal{G}_1 . Thus, $X_3 = X_4 = 0$. We compute a second Gröbner basis \mathcal{G}_2 of the ideal generated by $\mathcal{G}_1 \cup \{X_3, X_4\} \subset \mathbb{R}[\gamma_1, \gamma_2, \gamma_3, \gamma_4, \gamma_5, \gamma_6, \mu, X_1, X_2, X_3, X_4]$ with respect to the lexicographical order, obtaining that the polynomial

$$\mathbf{g}_{21} = X_1^2 + X_2^2$$

belongs to \mathcal{G}_2 , which shows that X = 0 and Ricci solitons are Einstein metrics, which do not exist in this case.

Structure operator L with Lorentzian kernel

In this case, $\lambda_3 = 0$ and we can assume without loss of generality that $\lambda_1 = 0$ and $\lambda_2 \neq 0$. Therefore, left-invariant metrics are described by

$$\begin{split} & [e_1, e_3] = -\lambda_2 e_2, \quad [e_1, e_4] = \gamma_1 e_1 + \gamma_2 e_2 + \gamma_3 e_3, \\ & [e_2, e_4] = \gamma_4 e_2, \qquad [e_3, e_4] = \gamma_5 e_1 + \gamma_6 e_2 - (\gamma_1 - \gamma_4) e_3, \end{split}$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis. Proceeding as in the previous case, the polynomials \mathfrak{P}_{ij} are given by

$$\begin{aligned} \mathfrak{P}_{11} &= \lambda_2^2 - \gamma_2^2 + \gamma_3^2 - \gamma_5^2 - 4\gamma_1\gamma_4 + 4\gamma_1X_4 - 2\mu, \\ \mathfrak{P}_{12} &= -2X_3\lambda_2 + \gamma_1\gamma_2 - 3\gamma_2\gamma_4 - \gamma_5\gamma_6 + 2X_4\gamma_2, \\ \mathfrak{P}_{13} &= -2\gamma_1(\gamma_3 + \gamma_5) - \gamma_2\gamma_6 + 3\gamma_3\gamma_4 - \gamma_4\gamma_5 - 2X_4(\gamma_3 - \gamma_5), \\ \mathfrak{P}_{14} &= \gamma_6\lambda_2 - 2(X_1\gamma_1 + X_3\gamma_5), \\ \mathfrak{P}_{22} &= -\lambda_2^2 + \gamma_2^2 - 4\gamma_4^2 - \gamma_6^2 + 4X_4\gamma_4 - 2\mu, \\ \mathfrak{P}_{23} &= 2X_1\lambda_2 - \gamma_1\gamma_6 - \gamma_2\gamma_3 - 2\gamma_4\gamma_6 + 2X_4\gamma_6, \\ \mathfrak{P}_{24} &= -2(X_1\gamma_2 + X_2\gamma_4 + X_3\gamma_6), \\ \mathfrak{P}_{33} &= -\lambda_2^2 + \gamma_3^2 + 4\gamma_4^2 - \gamma_5^2 - \gamma_6^2 - 4\gamma_1\gamma_4 + 4X_4(\gamma_1 - \gamma_4) + 2\mu, \\ \mathfrak{P}_{34} &= \gamma_2\lambda_2 - 2X_3(\gamma_1 - \gamma_4) + 2X_1\gamma_3, \\ \mathfrak{P}_{44} &= -4\gamma_1^2 - \gamma_2^2 + \gamma_3^2 - 4\gamma_4^2 + \gamma_5^2 + \gamma_6^2 + 4\gamma_1\gamma_4 - 2\gamma_3\gamma_5 - 2\mu. \end{aligned}$$

Since $\lambda_2 \neq 0$, we assume $\lambda_2 = 1$ and work with the homothetic metric determined by $\hat{e}_i = \frac{1}{\lambda_2} e_i$. Let $\mathcal{I}_1 \subset \mathbb{R}[\gamma_1, \gamma_2, \gamma_3, \gamma_4, \gamma_5, \gamma_6, \mu, X_1, X_2, X_3, X_4]$ be the ideal generated by the polynomials \mathfrak{P}_{ij} . We compute a Gröbner basis \mathcal{G}_1 of \mathcal{I}_1 with respect to the lexicographical order and see that the polynomials

$$\begin{aligned} \mathbf{g}_{11} &= X_2 (2617344X_4^8 + 13139712X_4^6 + 18557248X_4^4 + 7213356X_4^2 + 61803), \\ \mathbf{g}_{12} &= X_4 (83755008X_4^{14} + 776429568X_4^{12} + 2689679360X_4^{10} + 4517104000X_4^8 \\ &\quad + 4237066048X_4^6 + 2362718304X_4^4 + 591574590X_4^2 + 5006043) \end{aligned}$$

belong to \mathcal{G}_1 . Therefore, $X_2 = X_4 = 0$.

We compute a second Gröbner basis \mathcal{G}_2 of the ideal generated by the polynomials $\mathcal{G}_1 \cup \{X_2, X_4\}$ in $\mathbb{R}[\gamma_1, \gamma_2, \gamma_3, \gamma_4, \gamma_5, \gamma_6, \mu, X_1, X_2, X_3, X_4]$ with respect to the lexicographical order and see that the polynomial

$$\mathbf{g}_{21} = \gamma_4^2 + 1$$

belongs to \mathcal{G}_2 , which shows that there are no left-invariant Ricci solitons in this case.

Structure operator with zero eigenvalues: metrics on $\mathbb{R}^3 \rtimes \mathbb{R}$

Since $\lambda_1 = \lambda_2 = \lambda_3 = 0$, any linear map $D : \mathfrak{r}^3 \to \mathfrak{r}^3$ is a derivation. In order to simplify the structure constants, we proceed as follows. Let

$$\Phi(x,y) = \langle Dx,y \rangle$$

be the bilinear form associated to $D(\cdot) = [\cdot, e_4]$, and let

$$\Phi_s = \frac{1}{2}(\Phi + {}^t\Phi) \quad \text{and} \quad \Phi_a = \frac{1}{2}(\Phi - {}^t\Phi)$$

be the symmetric and skew-symmetric parts of Φ , respectively. We denote the corresponding self-adjoint and anti-self-adjoint endomorphisms by D_{sad} and D_{asad} , which are defined by

$$\Phi_s(x,y) = \langle D_{sad}x, y \rangle$$
 and $\Phi_a(x,y) = \langle D_{asad}x, y \rangle$,

respectively. We analyse the different Jordan normal forms of D_{sad} separately.

The self-adjoint part of the derivation D_{sad} is diagonalizable

In this case, there exists an orthonormal basis $\{e_1, e_2, e_3\}$ of \mathfrak{r}^3 with timelike e_3 , so that

$$D_{sad} = \begin{pmatrix} \eta_1 & 0 & 0 \\ 0 & \eta_2 & 0 \\ 0 & 0 & \eta_3 \end{pmatrix}, \quad D_{asad} = \begin{pmatrix} 0 & \gamma_1 & \gamma_2 \\ -\gamma_1 & 0 & \gamma_3 \\ \gamma_2 & \gamma_3 & 0 \end{pmatrix}.$$

Therefore left-invariant metrics are described by

$$[e_1, e_4] = \eta_1 e_1 - \gamma_1 e_2 + \gamma_2 e_3, \quad [e_2, e_4] = \gamma_1 e_1 + \eta_2 e_2 + \gamma_3 e_3,$$
$$[e_3, e_4] = \gamma_2 e_1 + \gamma_3 e_2 + \eta_3 e_3,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis of $\mathfrak{r}^3 \rtimes \mathbb{R}$ with timelike e_3 . After a straightforward calculation, we obtain the following polynomials $\tilde{\mathfrak{P}}_{ij} = \frac{1}{2}\mathfrak{P}_{ij}$:

$$\begin{split} \tilde{\mathfrak{P}}_{11} &= -\eta_1(\eta_1 + \eta_2 + \eta_3 - 2X_4) - \mu, \\ \tilde{\mathfrak{P}}_{22} &= -\eta_2(\eta_1 + \eta_2 + \eta_3 - 2X_4) - \mu, \\ \tilde{\mathfrak{P}}_{33} &= \eta_3(\eta_1 + \eta_2 + \eta_3 - 2X_4) + \mu, \\ \tilde{\mathfrak{P}}_{44} &= -\eta_1^2 - \eta_2^2 - \eta_3^2 - \mu. \end{split}$$

Considering the combinations

$$\eta_2 \tilde{\mathfrak{P}}_{11} - \eta_1 \tilde{\mathfrak{P}}_{22} = (\eta_1 - \eta_2)\mu,$$

$$\eta_3 \tilde{\mathfrak{P}}_{11} + \eta_1 \tilde{\mathfrak{P}}_{33} = (\eta_1 - \eta_3)\mu,$$

together with the expression of $\tilde{\mathfrak{P}}_{44}$, we see that $\eta_1 = \eta_2 = \eta_3 = \kappa$. Now, a standard calculation shows that the corresponding left-invariant metric has constant sectional curvature $-\kappa^2$.

The self-adjoint part of the derivation D_{sad} has complex eigenvalues

If the self-dual part of the derivation, D_{sad} , has complex eigenvalues then there exists an orthonormal basis $\{e_1, e_2, e_3\}$ of \mathfrak{r}^3 with timelike e_3 , so that

$$D_{sad} = \begin{pmatrix} \eta & 0 & 0 \\ 0 & \delta & \nu \\ 0 & -\nu & \delta \end{pmatrix}, \quad D_{asad} = \begin{pmatrix} 0 & \gamma_1 & \gamma_2 \\ -\gamma_1 & 0 & \gamma_3 \\ \gamma_2 & \gamma_3 & 0 \end{pmatrix},$$

where $\nu \neq 0$. The corresponding left-invariant metrics are described by

$$[e_1, e_4] = \eta e_1 - \gamma_1 e_2 + \gamma_2 e_3, \qquad [e_2, e_4] = \gamma_1 e_1 + \delta e_2 + (\gamma_3 - \nu) e_3,$$
$$[e_3, e_4] = \gamma_2 e_1 + (\gamma_3 + \nu) e_2 + \delta e_3,$$

and a standard calculation shows that the polynomials $\tilde{\mathfrak{P}}_{ij} = \frac{1}{2}\mathfrak{P}_{ij}$ are given by

$$\begin{split} \tilde{\mathfrak{P}}_{11} &= -\eta^2 - 2(\delta - X_4)\eta - \mu, & \tilde{\mathfrak{P}}_{12} &= \gamma_1(\delta - \eta) - \gamma_2\nu, \\ \tilde{\mathfrak{P}}_{13} &= \gamma_2(\delta - \eta) + \gamma_1\nu, & \tilde{\mathfrak{P}}_{14} &= -X_1\eta - X_2\gamma_1 - X_3\gamma_2, \\ \tilde{\mathfrak{P}}_{22} &= -2\delta^2 - (\eta - 2X_4)\delta - 2\gamma_3\nu - \mu, & \tilde{\mathfrak{P}}_{23} &= -(2\delta + \eta - 2X_4)\nu, \\ \tilde{\mathfrak{P}}_{24} &= X_1\gamma_1 - X_2\delta - X_3(\nu + \gamma_3), & \tilde{\mathfrak{P}}_{33} &= 2\delta^2 + (\eta - 2X_4)\delta - 2\gamma_3\nu + \mu, \\ \tilde{\mathfrak{P}}_{34} &= X_1\gamma_2 - X_2(\nu - \gamma_3) + X_3\delta, & \tilde{\mathfrak{P}}_{44} &= -2\delta^2 - \eta^2 + 2\nu^2 - \mu \,. \end{split}$$

Since $\nu \neq 0$, we can rescale the basis to work with the homothetic metric determined by $\hat{e}_i = \frac{1}{\nu} e_i$ and assume that $\nu = 1$ for the rest of our calculations. If we now consider the linear combinations

$$\gamma_2 \tilde{\mathfrak{P}}_{12} - \gamma_1 \tilde{\mathfrak{P}}_{13} = -\gamma_1^2 - \gamma_2^2 \quad \text{and} \quad \tilde{\mathfrak{P}}_{22} + \tilde{\mathfrak{P}}_{33} = -4\gamma_3,$$

it follows that $\gamma_1 = \gamma_2 = \gamma_3 = 0$. Now, the combinations

$$\begin{split} \tilde{\mathfrak{P}}_{23} &= -2\delta - \eta + 2X_4, \\ \tilde{\mathfrak{P}}_{24} &- \delta \, \tilde{\mathfrak{P}}_{34} = -X_3(\delta^2 + 1) \\ \delta \, \tilde{\mathfrak{P}}_{24} &+ \tilde{\mathfrak{P}}_{34} = -X_2(\delta^2 + 1) \end{split}$$

lead to $X_2 = X_3 = 0$ and $X_4 = \delta + \frac{1}{2}\eta$. Consequently, the system of polynomial equations $\{\tilde{\mathfrak{P}}_{ij} = 0\}$ reduces to

$$\mathfrak{P}_{11} = \mathfrak{P}_{22} = -\mathfrak{P}_{33} = -\mu = 0,$$

 $\tilde{\mathfrak{P}}_{14} = -X_1\eta = 0,$
 $\tilde{\mathfrak{P}}_{44} = -2\delta^2 - \eta^2 - \mu + 2 = 0,$

which shows that $X_1\eta = 0$ and the left-invariant metric given by

$$[e_1, e_4] = \eta e_1, \quad [e_2, e_4] = \varepsilon \sqrt{1 - \frac{1}{2}\eta^2} e_2 - e_3, \quad [e_3, e_4] = e_2 + \varepsilon \sqrt{1 - \frac{1}{2}\eta^2} e_3,$$

with $-\sqrt{2} \le \eta \le \sqrt{2}$ and $\varepsilon^2 = 1$, is a left-invariant steady Ricci soliton which corresponds to Assertion (*i*) in Theorem 4.2. Moreover, the left-invariant soliton vector field is given by $X = X_1 e_1 + \varepsilon e_4$ if $\eta = 0$, and $X = \frac{1}{2} \left(\eta + \varepsilon \sqrt{4 - 2\eta^2} \right) e_4$ if $\eta \neq 0$, as stated in Remark 4.3.

 $X = X_1 e_1 + \varepsilon e_4$ if $\eta = 0$, and $X = \frac{1}{2} \left(\eta + \varepsilon \sqrt{4 - 2\eta^2} \right) e_4$ if $\eta \neq 0$, as stated in Remark 4.3. Note that $(e_1, e_2, e_3, e_4) \mapsto (e_1, e_2, -e_3, -e_4)$ defines an isometry interchanging (η, ε) and $(-\eta, -\varepsilon)$. Therefore, we can assume $0 \le \eta \le \sqrt{2}$. In particular, if $\eta = 0$, the same isometry interchanges $\varepsilon = 1$ and $\varepsilon = -1$. A straightforward calculation shows that the metrics described above are never symmetric and they are Einstein if and only if $\eta = -\frac{2\varepsilon}{\sqrt{3}}$, which corresponds to Assertion (*i*) in Theorem 4.1.

The self-adjoint part of the derivation D_{sad} has a double root

In this case, there exists a pseudo-orthonormal basis $\{u_1, u_2, u_3\}$ of \mathfrak{r}^3 such that $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = 1$ and

$$D_{sad} = \begin{pmatrix} \eta_1 & 0 & 0 \\ \varepsilon & \eta_1 & 0 \\ 0 & 0 & \eta_2 \end{pmatrix}, \quad D_{asad} = \begin{pmatrix} \gamma_1 & 0 & \gamma_2 \\ 0 & -\gamma_1 & \gamma_3 \\ -\gamma_3 & -\gamma_2 & 0 \end{pmatrix},$$

where $\varepsilon^2 = 1$. In this situation, the corresponding left-invariant metrics are described by

$$[u_1, u_4] = (\eta_1 + \gamma_1)u_1 + \varepsilon u_2 - \gamma_3 u_3, \quad [u_2, u_4] = (\eta_1 - \gamma_1)u_2 - \gamma_2 u_3,$$
$$[u_3, u_4] = \gamma_2 u_1 + \gamma_3 u_2 + \eta_2 u_3,$$

where $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis for which the non-zero inner products are $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$. We consider the following polynomials $\tilde{\mathfrak{P}}_{ij} = \frac{1}{2} \mathfrak{P}_{ij}$:

$$\begin{split} \tilde{\mathfrak{P}}_{11} &= -\varepsilon (2\eta_1 + \eta_2 + 2\gamma_1 - 2X_4), \qquad \tilde{\mathfrak{P}}_{12} = -\eta_1 (2\eta_1 + \eta_2) + 2X_4\eta_1 - \mu, \\ \tilde{\mathfrak{P}}_{13} &= -\gamma_3 (\eta_1 - \eta_2) - \varepsilon \gamma_2, \qquad \tilde{\mathfrak{P}}_{23} = -\gamma_2 (\eta_1 - \eta_2), \\ \tilde{\mathfrak{P}}_{33} &= -\eta_2 (2\eta_1 + \eta_2) + 2X_4\eta_2 - \mu, \qquad \tilde{\mathfrak{P}}_{44} = -2\eta_1^2 - \eta_2^2 - \mu \,. \end{split}$$

It is easy to check that

$$\begin{split} &\gamma_{2}\tilde{\mathfrak{P}}_{13} - \gamma_{3}\tilde{\mathfrak{P}}_{23} = -\varepsilon\gamma_{2}^{2}, \\ &\eta_{2}\tilde{\mathfrak{P}}_{12} - \eta_{1}\tilde{\mathfrak{P}}_{33} = (\eta_{1} - \eta_{2})\mu, \\ &\varepsilon\eta_{1}\tilde{\mathfrak{P}}_{11} - \tilde{\mathfrak{P}}_{33} + \tilde{\mathfrak{P}}_{44} = -\eta_{1}(4\eta_{1} - \eta_{2} + 2\gamma_{1}) + 2X_{4}(\eta_{1} - \eta_{2}), \end{split}$$

and it now follows from the expression of \mathfrak{P}_{44} above that $\gamma_2 = 0$, $\eta_2 = \eta_1$ and $\eta_1(3\eta_1 + 2\gamma_1) = 0$.

If $3\eta_1 + 2\gamma_1 = 0$, the resulting left-invariant metric is Einstein and it corresponds to Assertion (*ii*) in Theorem 4.1. Finally, if $\eta_1 = \gamma_2 = \eta_2 = 0$ and $\gamma_1 \neq 0$, then the left-invariant metric corresponds to

$$[u_1, u_4] = \gamma_1 u_1 + \varepsilon u_2 - \gamma_3 u_3, \quad [u_2, u_4] = -\gamma_1 u_2, \quad [u_3, u_4] = \gamma_3 u_2, \tag{4.1}$$

and u_2 is a recurrent null vector. Furthermore, a straightforward calculation shows that the curvature tensor is transversally flat (i.e., R(Y,Z) = 0 for all $Y, Z \in u_2^{\perp}$) and the Ricci operator is isotropic ($\rho_{11} = -2\varepsilon\gamma_1$ is the only non-zero component of the Ricci tensor). Therefore, the underlying structure is that of a pp-wave which is neither symmetric nor locally conformally flat.

The self-adjoint part of the derivation D_{sad} has a triple root

Let $\{u_1, u_2, u_3\}$ be a pseudo-orthonormal basis of \mathfrak{r}^3 with $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = 1$, so that

$$D_{sad} = \begin{pmatrix} \eta & 0 & 1 \\ 0 & \eta & 0 \\ 0 & 1 & \eta \end{pmatrix}, \quad D_{asad} = \begin{pmatrix} \gamma_1 & 0 & \gamma_2 \\ 0 & -\gamma_1 & \gamma_3 \\ -\gamma_3 & -\gamma_2 & 0 \end{pmatrix}.$$

The corresponding left-invariant metrics are given by

$$[u_1, u_4] = (\eta + \gamma_1)u_1 - \gamma_3 u_3, \qquad [u_2, u_4] = (\eta - \gamma_1)u_2 - (\gamma_2 - 1)u_3,$$
$$[u_3, u_4] = (\gamma_2 + 1)u_1 + \gamma_3 u_2 + \eta u_3,$$

where $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis of of $\mathfrak{r}^3 \rtimes \mathbb{R}$ for which the non-zero inner products are $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$. A straightforward calculation shows that the non-zero polynomials $\tilde{\mathfrak{P}}_{ij} = \frac{1}{2} \mathfrak{P}_{ij}$ are given by

$$\begin{split} \tilde{\mathfrak{P}}_{12} &= -3\eta^2 + 2X_4\eta + \gamma_3 - \mu, \\ \tilde{\mathfrak{P}}_{22} &= 2\gamma_2, \\ \tilde{\mathfrak{P}}_{23} &= -3\eta + \gamma_1 + 2X_4, \\ \tilde{\mathfrak{P}}_{24} &= -X_1(\eta + \gamma_1) - X_3(\gamma_2 + 1), \\ \tilde{\mathfrak{P}}_{34} &= -X_3\eta + X_2(\gamma_2 - 1) + X_1\gamma_3, \\ \tilde{\mathfrak{P}}_{44} &= -3\eta^2 - \mu. \end{split}$$

It follows from the combinations

$$\begin{split} \tilde{\mathfrak{P}}_{22} &= 2\gamma_2, \\ \tilde{\mathfrak{P}}_{12} - \tilde{\mathfrak{P}}_{33} &= 3\gamma_3, \\ \tilde{\mathfrak{P}}_{12} - \eta \, \tilde{\mathfrak{P}}_{23} - \tilde{\mathfrak{P}}_{44} &= \eta (3\eta - \gamma_1) + \gamma_3 \end{split}$$

that $\gamma_2 = \gamma_3 = 0$ and $\eta(3\eta - \gamma_1) = 0$.

Now, if $3\eta - \gamma_1 = 0$, the corresponding left-invariant metric is Einstein, and it corresponds to Assertion *(iii)* in Theorem 4.1 if $\gamma_1 = 3\eta \neq 0$. The case where $\eta = \gamma_1 = 0$ corresponds to a Ricci-flat plane wave.

If $\eta = 0$ and $\gamma_1 \neq 0$, then a straightforward calculation shows that the corresponding leftinvariant metrics, which are given by

$$[u_1, u_4] = \gamma_1 u_1, \quad [u_2, u_4] = -\gamma_1 u_2 + u_3, \quad [u_3, u_4] = u_1,$$

are neither Einstein nor symmetric. Moreover, the system of polynomial equations $\{\tilde{\mathfrak{P}}_{ij} = 0\}$ reduces to

$$\begin{split} \tilde{\mathfrak{P}}_{12} &= \tilde{\mathfrak{P}}_{33} = \tilde{\mathfrak{P}}_{44} = -\mu = 0, & \tilde{\mathfrak{P}}_{24} = -X_1 \gamma_1 - X_3 = 0, \\ \tilde{\mathfrak{P}}_{14} &= X_2 \gamma_1 = 0, & \tilde{\mathfrak{P}}_{34} = -X_2 = 0, \\ \tilde{\mathfrak{P}}_{23} &= \gamma_1 + 2X_4 = 0, \end{split}$$

and it defines a left-invariant steady Ricci soliton with left-invariant soliton vector field $X = X_1 u_1 - X_1 \gamma_1 u_3 - \frac{1}{2} \gamma_1 u_4$.

Finally, notice that $(u_1, u_2, u_3, u_4) \mapsto (-u_1, -u_2, u_3, -u_4)$ defines an isometry interchanging γ_1 and $-\gamma_1$. Therefore, we can restrict the parameter γ_1 to $\gamma_1 > 0$ without losing generality. Setting $\alpha = \gamma_1$, this case corresponds to Assertion *(ii)* in Theorem 4.2.

4.1.2 The structure operator *L* has complex eigenvalues

If the structure operator L is of type **Ib**, there exists an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ with timelike e_3 , where $\mathfrak{g}_3 = \operatorname{span}\{e_1, e_2, e_3\}$ and $\mathfrak{r} = \operatorname{span}\{e_4\}$, so that

$$[e_1, e_2] = -\beta e_2 - \alpha e_3, \ [e_1, e_3] = -\alpha e_2 + \beta e_3, \ [e_2, e_3] = \lambda e_1, \ [e_i, e_4] = \sum_{j=1}^3 \alpha_i^j e_j,$$

for certain $\alpha_i^j \in \mathbb{R}$ and $\beta \neq 0$. Next, we consider the cases where the real eigenvalue $\lambda = 0$ and $\lambda \neq 0$ separately.

Case of zero real eigenvalue: metrics on $E(1,1) \rtimes \mathbb{R}$

If $\lambda = 0$, then the corresponding metrics are given by

$$[e_1, e_2] = -\beta e_2 - \alpha e_3, \qquad [e_1, e_3] = -\alpha e_2 + \beta e_3, \qquad [e_1, e_4] = \gamma_1 e_2 + \gamma_2 e_3,$$
$$[e_2, e_4] = 2\gamma_3 \beta e_2 + (\gamma_3 - \gamma_4) \alpha e_3, \qquad [e_3, e_4] = (\gamma_3 - \gamma_4) \alpha e_2 + 2\gamma_4 \beta e_3,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis of $\mathfrak{e}(1, 1) \rtimes \mathfrak{r}$ with timelike e_3 . A straightforward calculation shows that the polynomials \mathfrak{P}_{ij} are given by

$$\begin{split} \mathfrak{P}_{11} &= -4\beta^2 - \gamma_1^2 + \gamma_2^2 - 2\mu, \\ \mathfrak{P}_{12} &= (\gamma_2(\gamma_3 - \gamma_4) - 2X_3)\alpha - 2(\gamma_1(2\gamma_3 + \gamma_4) + X_2)\beta + 2X_4\gamma_1, \\ \mathfrak{P}_{13} &= -(\gamma_1(\gamma_3 - \gamma_4) - 2X_2)\alpha + 2(\gamma_2(\gamma_3 + 2\gamma_4) - X_3)\beta - 2X_4\gamma_2 \\ \mathfrak{P}_{14} &= -4(\gamma_3 - \gamma_4)\beta^2, \\ \mathfrak{P}_{22} &= -8\gamma_3(\gamma_3 + \gamma_4)\beta^2 + 4(2X_4\gamma_3 + X_1)\beta + \gamma_1^2 - 2\mu, \\ \mathfrak{P}_{23} &= -4((\gamma_3 - \gamma_4)^2 + 1)\alpha\beta - \gamma_1\gamma_2, \\ \mathfrak{P}_{24} &= -(\gamma_2 + 2X_3(\gamma_3 - \gamma_4))\alpha + (\gamma_1 - 4X_2\gamma_3)\beta - 2X_1\gamma_1, \\ \mathfrak{P}_{33} &= 8(\gamma_3 + \gamma_4)\gamma_4\beta^2 - 4(2X_4\gamma_4 - X_1)\beta + \gamma_2^2 + 2\mu, \\ \mathfrak{P}_{34} &= (\gamma_1 + 2X_2(\gamma_3 - \gamma_4))\alpha + (\gamma_2 + 4X_3\gamma_4)\beta + 2X_1\gamma_2 \\ \mathfrak{P}_{44} &= -8(\gamma_3^2 + \gamma_4^2)\beta^2 - \gamma_1^2 + \gamma_2^2 - 2\mu \,. \end{split}$$

Since $\beta \neq 0$, we can assume that $\beta = 1$ and work with the homothetic metric determined by $\hat{e}_i = \frac{1}{\beta} e_i$. Using the expressions for \mathfrak{P}_{14} , \mathfrak{P}_{11} , \mathfrak{P}_{23} and \mathfrak{P}_{44} given above, together with the combination

$$\mathfrak{P}_{22} + \mathfrak{P}_{33} = \gamma_1^2 + \gamma_2^2 - 8(\gamma_3^2 - \gamma_4^2 - X_4(\gamma_3 - \gamma_4) - X_1),$$

we see that

$$\gamma_4 = \gamma_3, \quad \mu = -\frac{1}{2}(\gamma_1^2 - \gamma_2^2 + 4), \quad \alpha = -\frac{1}{4}\gamma_1\gamma_2, \quad \gamma_3 = \frac{\varepsilon_1}{2}, \quad X_1 = -\frac{1}{8}(\gamma_1^2 + \gamma_2^2),$$

where $\varepsilon_1^2 = 1$. Now, it is easy to check that

$$\varepsilon_{1}\mathfrak{P}_{12} - \mathfrak{P}_{24} - \frac{1}{4}\gamma_{1}\gamma_{2}\mathfrak{P}_{34} + \frac{1}{2}\gamma_{1}\mathfrak{P}_{33} = \frac{1}{16}\gamma_{1}(\gamma_{2}^{2} - 8)(\gamma_{2}^{2} + 2\gamma_{1}^{2} + 8),$$

$$\varepsilon_{1}\mathfrak{P}_{13} - \frac{1}{4}\gamma_{1}\gamma_{2}\mathfrak{P}_{24} + \mathfrak{P}_{34} - \frac{1}{2}\gamma_{2}\mathfrak{P}_{33} = -\frac{1}{16}\gamma_{2}(\gamma_{1}^{2} + 8)(2\gamma_{2}^{2} + \gamma_{1}^{2} - 8),$$

from where it follows that $\gamma_1 = 0$ and $\gamma_2 \in \{-2, 0, 2\}$. A standard calculation shows that the corresponding left-invariant metric, which is given by

$$[e_1, e_2] = -e_2, \qquad [e_1, e_3] = e_3, \qquad [e_1, e_4] = \gamma_2 e_3,$$
$$[e_2, e_4] = \varepsilon_1 e_2, \qquad [e_3, e_4] = \varepsilon_1 e_3,$$

is Einstein (and locally isometric to a product of two surfaces with the same constant curvature) if and only if $\gamma_2 = 0$. If $\gamma_2 \neq 0$, we take $\gamma_2 = 2\varepsilon_2$, with $\varepsilon_2^2 = 1$, and the system of polynomial equations $\{\mathfrak{P}_{ij} = 0\}$ is now given by

$$\begin{aligned} \mathfrak{P}_{12} &= -2X_2 = 0, \\ \mathfrak{P}_{13} &= -2(X_3 + 2\varepsilon_2 X_4) + 6\varepsilon_1 \varepsilon_2 = 0, \\ \mathfrak{P}_{22} &= 4\varepsilon_1 X_4 - 6 = 0, \end{aligned} \qquad \begin{aligned} \mathfrak{P}_{23} &= -2\varepsilon_1 X_2 = 0, \\ \mathfrak{P}_{33} &= -4\varepsilon_1 X_4 + 6 = 0, \\ \mathfrak{P}_{34} &= 2\varepsilon_1 X_3 = 0, \end{aligned}$$

which shows that $X_2 = X_3 = 0$, $X_4 = \frac{3\varepsilon_1}{2}$, and the left-invariant metric given by

$$[e_1, e_2] = -e_2,$$
 $[e_1, e_3] = e_3,$ $[e_1, e_4] = 2\varepsilon_2 e_3,$
 $[e_2, e_4] = \varepsilon_1 e_2,$ $[e_3, e_4] = \varepsilon_1 e_3,$

is a steady Ricci soliton with left-invariant soliton vector field $X = -\frac{1}{2}e_1 + \frac{3\varepsilon_1}{2}e_4$.

Notice that $(e_1, e_2, e_3, e_4) \mapsto (e_1, -e_2, -e_3, -e_4)$ is an isometry interchanging $\varepsilon_1 = 1$ and $\varepsilon_1 = -1$, and $(e_1, e_2, e_3, e_4) \mapsto (e_1, -e_2, -e_3, e_4)$ defines an isometry which interchanges $\varepsilon_2 = 1$ and $\varepsilon_2 = -1$. Therefore, we can set $\varepsilon_1 = \varepsilon_2 = 1$, which leads to Assertion *(iii)* in Theorem 4.2.

Case of non-zero real eigenvalue: metrics on $\widetilde{SL}(2,\mathbb{R})\times\mathbb{R}$

If $\lambda \neq 0$ then the corresponding left-invariant metrics are given by

$$[e_1, e_2] = -\beta e_2 - \alpha e_3, \qquad [e_1, e_4] = (\alpha^2 + \beta^2)(\gamma_1 e_2 + \gamma_2 e_3)$$
$$[e_1, e_3] = -\alpha e_2 + \beta e_3, \qquad [e_2, e_4] = -(\gamma_1 \alpha - \gamma_2 \beta)\lambda e_1 + \gamma_3 \beta e_2 + \gamma_3 \alpha e_3,$$
$$[e_2, e_3] = \lambda e_1, \qquad [e_3, e_4] = (\gamma_2 \alpha + \gamma_1 \beta)\lambda e_1 + \gamma_3 \alpha e_2 - \gamma_3 \beta e_3,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis of $\mathfrak{sl}(2, \mathbb{R}) \rtimes \mathfrak{r}$ with timelike e_3 . A straightforward calculation shows that the polynomials \mathfrak{P}_{ij} are given by

$$\begin{aligned} \mathfrak{P}_{11} &= -\left((\alpha^{2} + \beta^{2})^{2} - (\alpha^{2} - \beta^{2})\lambda^{2}\right)(\gamma_{1}^{2} - \gamma_{2}^{2}) - 4\alpha\beta\lambda^{2}\gamma_{1}\gamma_{2} - 4\beta^{2} - \lambda^{2} - 2\mu, \\ \mathfrak{P}_{12} &= \left(2X_{4}(\alpha^{2} + \beta^{2} - \alpha\lambda) - (\alpha^{2} + \beta^{2} + 2\alpha\lambda)\beta\gamma_{3}\right)\gamma_{1} \\ &+ \left(\left((\alpha^{2} + \beta^{2})\alpha - (\alpha^{2} - \beta^{2})\lambda\right)\gamma_{3} + 2X_{4}\beta\lambda\right)\gamma_{2} - 2(X_{3}(\alpha - \lambda) + X_{2}\beta), \\ \mathfrak{P}_{13} &= -\left(\left((\alpha^{2} + \beta^{2})\alpha - (\alpha^{2} - \beta^{2})\lambda\right)\gamma_{3} - 2X_{4}\beta\lambda\right)\gamma_{1} \end{aligned}$$

$$- (2X_4(\alpha^2 + \beta^2 - \alpha\lambda) + (\alpha^2 + \beta^2 + 2\alpha\lambda)\beta\gamma_3)\gamma_2 + 2(X_2(\alpha - \lambda) - X_3\beta),$$

$$\mathfrak{P}_{14} = 2(X_2\alpha - X_3\beta)\lambda\gamma_1 - 2(X_3\alpha + X_2\beta)\lambda\gamma_2 - 4\beta^2\gamma_3,$$

$$\mathfrak{P}_{22} = ((\alpha^2 + \beta^2)^2 - \alpha^2\lambda^2)\gamma_1^2 - \beta^2\lambda^2\gamma_2^2 + 2\alpha\beta\lambda^2\gamma_1\gamma_2 + 4X_4\beta\gamma_3 + 4X_1\beta - (2\alpha - \lambda)\lambda - 2\mu,$$

$$\mathfrak{P}_{23} = \alpha\beta(\lambda^2(\gamma_1^2 - \gamma_2^2) - 4\gamma_3^2) - ((\alpha^2 + \beta^2)^2 - (\alpha^2 - \beta^2)\lambda^2)\gamma_1\gamma_2 - 2(2\alpha - \lambda)\beta,$$

$$\mathfrak{P}_{24} = ((\alpha^2 + \beta^2)(\beta - 2X_1) - \beta\lambda^2)\gamma_1 - ((\alpha^2 + \beta^2)(\alpha - 2\lambda) + \alpha\lambda^2)\gamma_2 - 2(X_3\alpha + X_2\beta)\gamma_3,$$

$$\mathfrak{P}_{33} = -\beta^2\lambda^2\gamma_1^2 + ((\alpha^2 + \beta^2)^2 - \alpha^2\lambda^2)\gamma_2^2 - 2\alpha\beta\lambda^2\gamma_1\gamma_2 + 4X_4\beta\gamma_3 + 4X_1\beta + (2\alpha - \lambda)\lambda + 2\mu,$$

$$\mathfrak{P}_{34} = ((\alpha^2 + \beta^2)(\alpha - 2\lambda) + \alpha\lambda^2)\gamma_1 + ((\alpha^2 + \beta^2)(2X_1 + \beta) - \beta\lambda^2)\gamma_2 + (2X_2\alpha - 2X_3\beta)\gamma_3,$$

$$\mathfrak{P}_{44} = -(\alpha^2 + \beta^2 - (\alpha + \beta)\lambda)(\alpha^2 - \alpha\lambda + (\beta + \lambda)\beta)(\gamma_1^2 - \gamma_2^2) - 4\beta^2\gamma_3^2 - 4(\alpha^2 + \beta^2 - \alpha\lambda)\beta\lambda\gamma_1\gamma_2 - 2\mu.$$

In this case, we make use of Gröbner bases again but, since it is extraordinarily difficult to obtain such a basis using the above polynomials \mathfrak{P}_{ij} , we need to reduce the number of variables. Let us focus on the expressions of \mathfrak{P}_{11} , \mathfrak{P}_{22} and the linear combinations

$$\beta \mathfrak{P}_{12} - (\alpha - \lambda) \mathfrak{P}_{13}$$
 and $(\alpha - \lambda) \mathfrak{P}_{12} + \beta \mathfrak{P}_{13}$.

They allow us to clear μ , X_1 , X_2 and X_3 , respectively, as

$$\mu = -\frac{1}{2}((\alpha^{2} + \beta^{2})^{2} - (\alpha^{2} - \beta^{2})\lambda^{2})(\gamma_{1}^{2} - \gamma_{2}^{2}) - 2\alpha\beta\lambda^{2}\gamma_{1}\gamma_{2} - 2\beta^{2} - \frac{1}{2}\lambda^{2},$$

$$X_{1} = -\frac{1}{4\beta}((\alpha^{2} + \beta^{2})^{2} - \alpha^{2}\lambda^{2})\gamma_{1}^{2} + \frac{1}{4}\beta\lambda^{2}\gamma_{2}^{2} - \frac{1}{2}\alpha\lambda^{2}\gamma_{1}\gamma_{2} - X_{4}\gamma_{3}$$

$$-\frac{1}{4\beta}((\lambda - 2\alpha)\lambda - 2\mu),$$

$$X_{2} = \left(\frac{1}{2}\left(\alpha^{2} - \beta^{2} - \frac{4\alpha\beta^{2}\lambda}{(\alpha - \lambda)^{2} + \beta^{2}}\right)\gamma_{3} + X_{4}\beta\right)\gamma_{1} + \left(\frac{\alpha\beta(\alpha^{2} + \beta^{2} - \lambda^{2})}{(\alpha - \lambda)^{2} + \beta^{2}}\gamma_{3} + X_{4}\alpha\right)\gamma_{2},$$

$$X_{3} = -\left(\frac{\alpha\beta(\alpha^{2} + \beta^{2} - \lambda^{2})}{(\alpha - \lambda)^{2} + \beta^{2}}\gamma_{3} - X_{4}\alpha\right)\gamma_{1} + \left(\frac{1}{2}\left(\alpha^{2} - \beta^{2} - \frac{4\alpha\beta^{2}\lambda}{(\alpha - \lambda)^{2} + \beta^{2}}\right)\gamma_{3} - X_{4}\beta\right)\gamma_{2}.$$

Consequently, we can eliminate these variables from the polynomials \mathfrak{P}_{ij} and, as a consequence, X_4 is also eliminated. Let us denote by \mathfrak{Q}_{ij} the expressions obtained from the polynomials \mathfrak{P}_{ij} after substituting μ , X_1 , X_2 and X_3 . These expressions are not polynomials, since they contain fractional expressions with variables in the denominators, but we can easily avoid this inconvenience by considering \mathfrak{Q}'_{ij} , which are given by

$$\begin{split} \mathfrak{Q}_{14}' &= ((\alpha - \lambda)^2 + \beta^2) \mathfrak{Q}_{14}, \qquad \mathfrak{Q}_{23}' = \mathfrak{Q}_{23}, \\ \mathfrak{Q}_{24}' &= 2((\alpha - \lambda)^2 + \beta^2) \beta \mathfrak{Q}_{24}, \qquad \mathfrak{Q}_{33}' = \mathfrak{Q}_{33}, \\ \mathfrak{Q}_{34}' &= 2((\alpha - \lambda)^2 + \beta^2) \beta \mathfrak{Q}_{34}, \qquad \mathfrak{Q}_{44}' = \mathfrak{Q}_{44}, \end{split}$$

and the remaining ones are zero. This way, $\mathfrak{Q}'_{ij} \in \mathbb{R}[\gamma_1, \gamma_2, \gamma_3, \lambda, \alpha, \beta]$. Now, let $\mathcal{I} \subset \mathbb{R}[\gamma_1, \gamma_2, \gamma_3, \lambda, \alpha, \beta]$ be the ideal generated by the polynomials \mathfrak{Q}'_{ij} . We compute a Gröbner basis \mathcal{G} of \mathcal{I}

with respect to the lexicographical order and see that the polynomial

$$\mathbf{g} = (\alpha^2 + \beta^2)^2 \beta^2$$

belongs to \mathcal{G} . Since $\beta \neq 0$, there are no left-invariant Ricci solitons in this case.

4.1.3 The structure operator L has a double root of its minimal polynomial

If the structure operator is of type II then there exists a pseudo-orthonormal basis $\{u_1, u_2, u_3, u_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ with $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$, where $\mathfrak{g}_3 = \operatorname{span}\{u_1, u_2, u_3\}$ and $\mathfrak{r} = \operatorname{span}\{u_4\}$, so that

$$[u_1, u_2] = \lambda_2 u_3, \quad [u_1, u_3] = -\lambda_1 u_1 - \varepsilon u_2, \quad [u_2, u_3] = \lambda_1 u_2, \quad [u_i, u_4] = \sum_{\substack{j=1 \\ (i=1,2,3)}}^3 \alpha_i^j u_j,$$

for certain $\alpha_i^j \in \mathbb{R}$ and $\varepsilon^2 = 1$. Next, depending on the eigenvalues λ_i , we are led to the following different possibilities.

Case $\lambda_i = 0$: metrics on $H^3 \rtimes \mathbb{R}$

If $\lambda_1 = \lambda_2 = 0$ then the corresponding metrics are determined by

$$\begin{split} & [u_1, u_3] = -\varepsilon u_2, \qquad [u_1, u_4] = \gamma_1 u_1 + \gamma_2 u_2 + \gamma_3 u_3, \\ & [u_2, u_4] = \gamma_4 u_2, \qquad [u_3, u_4] = \gamma_5 u_1 + \gamma_6 u_2 - (\gamma_1 - \gamma_4) u_3 \end{split}$$

and the polynomials \mathfrak{P}_{ij} are

$$\begin{aligned} \mathfrak{P}_{12} &= -2\gamma_4^2 - 2\gamma_1\gamma_4 + \gamma_5\gamma_6 + 2X_4\gamma_1 + 2X_4\gamma_4 - 2\mu, \\ \mathfrak{P}_{14} &= -2X_1\gamma_2 - 2X_2\gamma_4 - 2X_3\gamma_6 - \varepsilon\gamma_5, \\ \mathfrak{P}_{22} &= \gamma_5^2, \\ \mathfrak{P}_{24} &= -2(X_1\gamma_1 + X_3\gamma_5), \\ \mathfrak{P}_{33} &= -2(2\gamma_4^2 - 2\gamma_1\gamma_4 + \gamma_5\gamma_6 + 2X_4\gamma_1 - 2X_4\gamma_4 + \mu), \\ \mathfrak{P}_{34} &= 2(X_3\gamma_1 - X_1\gamma_3 - X_3\gamma_4), \\ \mathfrak{P}_{44} &= -3\gamma_1^2 - 3\gamma_4^2 + 2\gamma_1\gamma_4 - 2\gamma_3\gamma_5 - 2\gamma_5\gamma_6 - 2\mu. \end{aligned}$$

Therefore γ_5 must vanish. Considering the linear combination

$$0 = 2(\gamma_1 - \gamma_4)\mathfrak{P}_{12} + (\gamma_1 + \gamma_4)\mathfrak{P}_{33} = 2(\gamma_4 - 3\gamma_1)\mu$$

we see that there are two different possibilities depending on whether $\mu = 0$ or $\gamma_4 = 3\gamma_1$. If $\mu = 0$ then

$$\mathfrak{P}_{44} = -2\gamma_1^2 - 2\gamma_4^2 - (\gamma_1 - \gamma_4)^2,$$

and if $\gamma_4 = 3\gamma_1$, it is easy to check that

$$\begin{aligned} \mathfrak{P}_{24} &= -2X_1\gamma_1, \qquad \gamma_1\mathfrak{P}_{34} - \gamma_3\mathfrak{P}_{24} = -4X_3\gamma_1^2, \\ 2\gamma_1^2\mathfrak{P}_{14} - (2\gamma_1\gamma_2 - \gamma_3\gamma_6)\mathfrak{P}_{24} - \gamma_1\gamma_6\mathfrak{P}_{34} = -12X_2\gamma_1^3, \\ \mathfrak{P}_{12} - \mathfrak{P}_{44} = 8X_4\gamma_1. \end{aligned}$$

In any case, $\gamma_1 = \gamma_4 = 0$, and the left-invariant metrics are given by

$$[u_1, u_3] = -\varepsilon u_2, \quad [u_1, u_4] = \gamma_2 u_2 + \gamma_3 u_3, \quad [u_3, u_4] = \gamma_6 u_2.$$
(4.2)

A straightforward calculation shows that u_2 is parallel and the curvature tensor satisfies R(Y, Z) = 0 and $\nabla_Y R = 0$ for all $Y, Z \in u_2^{\perp}$. Thus, the underlying structure is a plane wave.

Case $\lambda_1 = 0, \lambda_2 \neq 0$: metrics on $\tilde{E}(2) \rtimes \mathbb{R}$ or $E(1, 1) \rtimes \mathbb{R}$

In this case the left-invariant metrics are described by

$$\begin{split} & [u_1, u_2] = \lambda_2 u_3, & [u_2, u_4] = \gamma_3 u_2 + \gamma_4 \lambda_2 u_3, \\ & [u_1, u_3] = -\varepsilon u_2, & [u_3, u_4] = -\varepsilon \gamma_4 u_2 + \gamma_3 u_3, \\ & [u_1, u_4] = \gamma_1 u_2 + \gamma_2 u_3, \end{split}$$

and a straightforward calculation leads to the polynomials

$$\mathfrak{P}_{12} = \lambda_2^2 - \gamma_2 \gamma_4 \lambda_2 - 2(\gamma_3^2 - X_4 \gamma_3 + \mu),$$

$$\mathfrak{P}_{24} = -\gamma_4 \lambda_2^2,$$

$$\mathfrak{P}_{44} = 2(\gamma_4 \varepsilon - \gamma_2) \gamma_4 \lambda_2 - 3\gamma_3^2 - 2\mu,$$

$$\mathfrak{P}_{33} = 2\gamma_2 \gamma_4 \lambda_2 - \lambda_2^2 - 2(2\gamma_3^2 - 2X_4 \gamma_3 + \mu)$$

It now follows that

$$2\mathfrak{P}_{12} - \frac{2(\gamma_2 + \varepsilon \gamma_4)}{\lambda_2}\mathfrak{P}_{24} - \mathfrak{P}_{33} - \mathfrak{P}_{44} = 3(\lambda_2^2 + \gamma_3^2)$$

and, since $\lambda_2 \neq 0$, there are no left-invariant Ricci solitons in this case.

Case $\lambda_1 \neq 0$, $\lambda_2 = 0$: metrics on $E(1,1) \rtimes \mathbb{R}$

If $\lambda_1 \neq 0$ and $\lambda_2 = 0$ then

$$\begin{split} & [u_1, u_3] = -\lambda_1 u_1 - \varepsilon u_2, & [u_2, u_4] = -(2\varepsilon \gamma_2 \lambda_1 - \gamma_1) u_2, \\ & [u_1, u_4] = \gamma_1 u_1 + \gamma_2 u_2, & [u_3, u_4] = \gamma_3 u_1 + \gamma_4 u_2, \\ & [u_2, u_3] = \lambda_1 u_2 \,. \end{split}$$

Straightforward calculations show that the non-zero polynomials \mathfrak{P}_{ij} are given by

$$\begin{split} \mathfrak{P}_{11} &= -4\varepsilon\lambda_1 + \gamma_4^2 - 4\gamma_1\gamma_2 + 4X_4\gamma_2 - 4\varepsilon X_3, \\ \mathfrak{P}_{12} &= -4\gamma_2^2\lambda_1^2 + 4\varepsilon(2\gamma_1 - X_4)\gamma_2\lambda_1 - 4\gamma_1^2 + \gamma_3\gamma_4 + 4X_4\gamma_1 - 2\mu, \\ \mathfrak{P}_{13} &= (2\varepsilon\gamma_2\gamma_4 - 2X_2)\lambda_1 - 3\gamma_1\gamma_4 - \gamma_2\gamma_3 + 2X_4\gamma_4 + 2\varepsilon X_1, \\ \mathfrak{P}_{14} &= (4\varepsilon X_2\gamma_2 - \gamma_4)\lambda_1 - 2X_2\gamma_1 - 2X_1\gamma_2 - \varepsilon\gamma_3 - 2X_3\gamma_4, \\ \mathfrak{P}_{22} &= \gamma_3^2, \\ \mathfrak{P}_{23} &= 2(2\varepsilon\gamma_2\gamma_3 + X_1)\lambda_1 - 3\gamma_1\gamma_3 + 2X_4\gamma_3, \\ \mathfrak{P}_{24} &= \gamma_3\lambda_1 - 2X_1\gamma_1 - 2X_3\gamma_3, \\ \mathfrak{P}_{33} &= -2(\gamma_3\gamma_4 + \mu), \\ \mathfrak{P}_{44} &= -4\gamma_2^2\lambda_1^2 + 8\varepsilon\gamma_1\gamma_2\lambda_1 - 4\gamma_1^2 - 2\gamma_3\gamma_4 - 2\mu. \end{split}$$

We immediately see that γ_3 must be zero, so that

$$\mathfrak{P}_{23} = 2X_1\lambda_1$$
 and $\mathfrak{P}_{33} = -2\mu$,

and so $X_1 = \mu = 0$. Now,

$$\mathfrak{P}_{44} = -4(\varepsilon\gamma_2\lambda_1 - \gamma_1)^2,$$

which implies that $\gamma_1 = arepsilon \gamma_2 \lambda_1$ and now the polynomial

$$\mathfrak{P}_{13} = -(\varepsilon \gamma_2 \gamma_4 + 2X_2)\lambda_1 + 2X_4 \gamma_4,$$

from where we get $X_2 = -\frac{\varepsilon}{2}\gamma_2\gamma_4 + X_4\frac{\gamma_4}{\lambda_1}$. At this point, the left-invariant metric is given by

$$\begin{split} & [u_1, u_3] = -\lambda_1 u_1 - \varepsilon u_2, \quad [u_1, u_4] = \varepsilon \gamma_2 \lambda_1 u_1 + \gamma_2 u_2, \quad [u_2, u_3] = \lambda_1 u_2, \\ & [u_2, u_4] = -\varepsilon \gamma_2 \lambda_1 u_2, \qquad [u_3, u_4] = \gamma_4 u_2, \end{split}$$

and the system of polynomial equations $\{\mathfrak{P}_{ij}=0\}$ is now given by

$$\mathfrak{P}_{11} = -4\varepsilon(\gamma_2^2 + 1)\lambda_1 + \gamma_4^2 + 4X_4\gamma_2 - 4\varepsilon X_3 = 0$$

$$\mathfrak{P}_{14} = -\gamma_4\{(\gamma_2^2 + 1)\lambda_1 + 2(X_3 - \varepsilon X_4\gamma_2)\} = 0.$$

Let us set

$$v_1 = u_1, \qquad v_2 = \frac{1}{2}u_2, \qquad v_3 = \varepsilon\gamma_2 u_3 + u_4, \qquad v_4 = u_3.$$

A straightforward calculation shows that $[v_i, v_j] = 0$ for all $i, j \in \{1, 2, 3\}$ and $[v_4, v_i] \in$ span $\{v_1, v_2, v_3\}$. Therefore, every left-invariant metric above is isometric to some left-invariant metric on $\mathbb{R}^3 \rtimes \mathbb{R}$ as discussed in Section 4.1.1.

Case of non-zero eigenvalues: metrics on $\widetilde{SL}(2,\mathbb{R})\times\mathbb{R}$

In this case one has the metric expressed in terms of the Lie brackets

$$\begin{split} & [u_1, u_2] = \lambda_2 u_3, \quad [u_1, u_3] = -\lambda_1 u_1 - \varepsilon u_2, \quad [u_2, u_3] = \lambda_1 u_2, \\ & [u_1, u_4] = \lambda_1 \gamma_1 u_1 + \varepsilon \gamma_1 u_2 + \gamma_2 \lambda_2 u_3, \qquad [u_2, u_4] = -\gamma_1 \lambda_1 u_2 + \gamma_3 \lambda_2 u_3, \\ & [u_3, u_4] = -\gamma_3 \lambda_1 u_1 - (\gamma_2 \lambda_1 + \varepsilon \gamma_3) u_2, \end{split}$$

and a straightforward calculation shows that the polynomials \mathfrak{P}_{ij} are given by

$$\begin{split} \mathfrak{P}_{11} &= \gamma_2^2 (\lambda_1^2 - \lambda_2^2) - 2\varepsilon (2\gamma_1^2 - \gamma_2\gamma_3 + 2)\lambda_1 + 2\varepsilon\lambda_2 + \gamma_3^2 + 4\varepsilon X_4\gamma_1 - 4\varepsilon X_3, \\ \mathfrak{P}_{12} &= \gamma_2\gamma_3\lambda_1^2 - (\gamma_2\gamma_3 - 1)\lambda_2^2 - 2\lambda_1\lambda_2 + \varepsilon\gamma_3^2\lambda_1 - 2\mu, \\ \mathfrak{P}_{13} &= \gamma_1\gamma_2(\lambda_1^2 - \lambda_1\lambda_2) + 2(\varepsilon\gamma_1\gamma_3 - X_4\gamma_2 - X_2)\lambda_1 + (\varepsilon\gamma_1\gamma_3 + 2X_4\gamma_2 + 2X_2)\lambda_2 \\ &- 2\varepsilon\gamma_3X_4 + 2\varepsilon X_1, \\ \mathfrak{P}_{14} &= \gamma_2(\lambda_1 - \lambda_2)^2 + 2(X_2\gamma_1 + X_3\gamma_2 + \varepsilon\gamma_3)\lambda_1 - 2\varepsilon\gamma_3\lambda_2 - 2\varepsilon X_1\gamma_1 + 2\varepsilon X_3\gamma_3, \\ \mathfrak{P}_{22} &= \gamma_3^2(\lambda_1^2 - \lambda_2^2), \\ \mathfrak{P}_{23} &= -\gamma_1\gamma_3(\lambda_1^2 - \lambda_1\lambda_2) - 2(X_4\gamma_3 - X_1)(\lambda_1 - \lambda_2), \\ \mathfrak{P}_{24} &= -\gamma_3(\lambda_1^2 + \lambda_2^2) + 2\gamma_3\lambda_1\lambda_2 - 2(X_1\gamma_1 - X_3\gamma_3)\lambda_1, \\ \mathfrak{P}_{33} &= -2\gamma_2\gamma_3\lambda_1^2 + (2\gamma_2\gamma_3 - 1)\lambda_2^2 - 2\varepsilon\gamma_3^2\lambda_1 - 2\mu, \\ \mathfrak{P}_{34} &= -2(X_1\gamma_2 + X_2\gamma_3)\lambda_2, \\ \mathfrak{P}_{44} &= -2\gamma_2\gamma_3(\lambda_1 - \lambda_2)^2 - 2\varepsilon\gamma_3^2(\lambda_1 - \lambda_2) - 2\mu. \end{split}$$

Let $\mathcal{I} \subset \mathbb{R}[\gamma_1, \gamma_2, \gamma_3, \varepsilon, \lambda_1, \lambda_2, \mu, X_1, X_2, X_3, X_4]$ be the ideal generated by the polynomials \mathfrak{P}_{ij} . We compute a Gröbner basis \mathcal{G} of \mathcal{I} with respect to the graded reverse lexicographical order

and obtain that the polynomial $g = \lambda_2^3$ belongs to \mathcal{G} . Since $\lambda_2 \neq 0$, there are no left-invariant Ricci solitons in this case.

4.1.4 The structure operator L has a triple root of its minimal polynomial

If the structure operator L is of type III, there exists a pseudo-orthonormal basis $\{u_1, u_2, u_3, u_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ with $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$, where $\mathfrak{g}_3 = \operatorname{span}\{u_1, u_2, u_3\}$ and $\mathfrak{r} = \operatorname{span}\{u_4\}$, so that

$$[u_1, u_2] = u_1 + \lambda u_3, \quad [u_1, u_3] = -\lambda u_1, \quad [u_2, u_3] = \lambda u_2 + u_3, \quad [u_i, u_4] = \sum_{j=1}^3 \alpha_i^j u_j,$$

for certain $\alpha_i^j \in \mathbb{R}$. In what follows, we will consider the cases $\lambda = 0$ and $\lambda \neq 0$ separately.

Case $\lambda = 0$: metrics on $E(1, 1) \rtimes \mathbb{R}$

If $\lambda = 0$, then

$$\begin{split} & [u_1, u_2] = u_1, & [u_1, u_4] = \gamma_1 u_1, & [u_2, u_3] = u_3, \\ & [u_2, u_4] = \gamma_2 u_1 + \gamma_3 u_3, & [u_3, u_4] = \gamma_4 u_3, \end{split}$$

and a straightforward calculation shows that the non-zero polynomials \mathfrak{P}_{ij} are given by

$$\begin{aligned} \mathfrak{P}_{12} &= -\gamma_1^2 - \gamma_1 \gamma_4 + 2X_4 \gamma_1 + 2X_2 - 2\mu, \qquad \mathfrak{P}_{23} = -2\gamma_3 \gamma_4 + 2X_4 \gamma_3 + 2X_3, \\ \mathfrak{P}_{22} &= -\gamma_3^2 - 2\gamma_2 \gamma_4 + 4X_4 \gamma_2 - 4X_1 - 4, \qquad \mathfrak{P}_{34} = -2(X_2 \gamma_3 + X_3 \gamma_4), \\ \mathfrak{P}_{24} &= -(2X_1 + 1)\gamma_1 - 2X_2 \gamma_2 + 2\gamma_4, \qquad \mathfrak{P}_{44} = -\gamma_1^2 - 2\gamma_4^2 - 2\mu, \\ \mathfrak{P}_{33} &= -2(\gamma_4^2 + \gamma_1 \gamma_4 - 2X_4 \gamma_4 + 2X_2 + \mu). \end{aligned}$$

From the expressions of $\mathfrak{P}_{22}, \mathfrak{P}_{23}$ and \mathfrak{P}_{44} we obtain

$$X_1 = -\frac{1}{4}\gamma_3^2 - \frac{1}{2}(\gamma_4 - 2X_4)\gamma_2 - 1, \quad X_3 = (\gamma_4 - X_4)\gamma_3, \quad \mu = -\frac{1}{2}\gamma_1^2 - \gamma_4^2.$$

Therefore,

$$\mathfrak{P}_{12} = 2\gamma_4^2 - (\gamma_4 - 2X_4)\gamma_1 + 2X_2$$

and

$$X_2 = -\gamma_4^2 + \frac{1}{2}(\gamma_4 - 2X_4)\gamma_1.$$

Now,

$$\mathfrak{P}_{24} = \frac{1}{2}(\gamma_3^2 + 2)\gamma_1 + 2(\gamma_2\gamma_4 + 1)\gamma_4$$

so

$$\gamma_1 = -\frac{4(\gamma_2\gamma_4 + 1)\gamma_4}{\gamma_3^2 + 2}.$$

At this point, the system of polynomial equations $\{\mathfrak{P}_{ij} = 0\}$ reduces to

$$\mathfrak{P}_{33} = \frac{4}{(\gamma_3^2 + 2)^2} \left\{ (\gamma_3^2 + 2\gamma_2\gamma_4 + 4)^2 \gamma_4 + X_4(\gamma_3^2 + 2)(\gamma_3^2 - 4\gamma_2\gamma_4 - 2) \right\} \gamma_4 = 0,$$

$$\mathfrak{P}_{34} = \frac{2}{\gamma_3^2 + 2} \left\{ 2(\gamma_2\gamma_4 + 1)\gamma_4 + (\gamma_3^2 - 4\gamma_2\gamma_4 - 2)X_4 \right\} \gamma_3\gamma_4 = 0.$$

It is easy to check that

$$\frac{\gamma_3}{2}\mathfrak{P}_{33} - \mathfrak{P}_{34} = \frac{1}{2(\gamma_3^2 + 2)^2} \left\{ 3\gamma_3^4 + 12\gamma_3^2 + (\gamma_3^2 + 4\gamma_2\gamma_4 + 6)^2 + 12 \right\} \gamma_3\gamma_4^2,$$

which implies that $\gamma_3 \gamma_4 = 0$.

If $\gamma_4 = 0$ (which implies $\gamma_1 = 0$), the left-invariant metrics are given by

$$[u_1, u_2] = u_1, \qquad [u_2, u_3] = u_3, \qquad [u_2, u_4] = \gamma_2 u_1 + \gamma_3 u_3,$$
 (4.3)

and a standard calculation shows that u_1 is a recurrent null vector. Moreover, the only non-zero component of the Ricci tensor

$$\rho_{22} = -2 - \frac{1}{2}\gamma_3^2$$

shows that the Ricci operator is isotropic, R(Y,Z) = 0, and $\nabla_Y R = 0$ for all $Y, Z \in u_1^{\perp}$. Consequently, the underlying structure corresponds to a plane wave.

If $\gamma_4 \neq 0$, then $\gamma_3 = 0$ and

$$\mathfrak{P}_{33} = 4\left\{ (\gamma_2\gamma_4 + 2)^2 \gamma_4 - X_4(2\gamma_2\gamma_4 + 1) \right\} \gamma_4.$$

Notice that if $2\gamma_2\gamma_4 + 1 = 0$ then $\mathfrak{P}_{33} \neq 0$. Hence the left-invariant metric is given by

$$[u_1, u_2] = u_1, \qquad [u_1, u_4] = -2(\gamma_2 \gamma_4 + 1)\gamma_4 u_1, \qquad [u_2, u_3] = u_3,$$

$$[u_2, u_4] = \gamma_2 u_1, \qquad [u_3, u_4] = \gamma_4 u_3,$$

and it is an expanding left-invariant Ricci soliton with $\mu = -(2(\gamma_2\gamma_4 + 1)^2 + 1)\gamma_4^2$ and left-invariant soliton vector field $X = X_1u_1 + X_2u_2 + X_4u_4$, where

$$X_{1} = \frac{1}{2(2\gamma_{2}\gamma_{4}+1)} (\gamma_{2}\gamma_{4}+2) (2(\gamma_{2}\gamma_{4}+1)\gamma_{2}\gamma_{4}-1),$$

$$X_{2} = \frac{1}{2\gamma_{2}\gamma_{4}+1} (\gamma_{2}\gamma_{4}+2) (2(\gamma_{2}\gamma_{4}+2)\gamma_{2}\gamma_{4}+3)\gamma_{4}^{2},$$

$$X_{4} = \frac{1}{2\gamma_{2}\gamma_{4}+1} (\gamma_{2}\gamma_{4}+2)^{2} \gamma_{4}.$$

A straightforward calculation shows that the metric above is symmetric if and only if $(\gamma_2\gamma_4 + 1)(\gamma_2\gamma_4 + 2) = 0$. Moreover, it is Einstein if and only if $\gamma_2\gamma_4 + 2 = 0$, in which case the sectional curvature is constant. Otherwise, if $\gamma_2\gamma_4 + 1 = 0$, the metric is locally a product $\mathbb{L}^2 \times N(c)$, where \mathbb{L}^2 is the Minkowskian plane and N(c) is a surface of constant curvature c. Finally, note that $(u_1, u_2, u_3, u_4) \mapsto (u_1, u_2, u_3, -u_4)$ defines an isometry interchanging (γ_4, γ_2) and $(-\gamma_4, -\gamma_2)$, which allows us to restrict the parameter γ_4 to $\gamma_4 > 0$ without losing generality. Setting $\alpha = \gamma_4$ and $\beta = \gamma_2$, this case corresponds to Assertion (*iv*) in Theorem 4.2 and Remark 4.5.

Case $\lambda \neq 0$: metrics on $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$

If $\lambda \neq 0$, then

$$\begin{split} & [u_1, u_2] = u_1 + \lambda u_3, & [u_1, u_3] = -\lambda u_1, \\ & [u_1, u_4] = \gamma_1 \lambda u_1 + \gamma_2 \lambda^2 u_3, & [u_2, u_3] = \lambda u_2 + u_3 \\ & [u_3, u_4] = -\gamma_3 \lambda u_1 - \gamma_2 \lambda^2 u_2 - \gamma_2 \lambda u_3, \\ & [u_2, u_4] = \gamma_3 u_1 - (\gamma_1 - \gamma_2) \lambda u_2 - (\gamma_1 - \gamma_2 - \gamma_3 \lambda) u_3, \end{split}$$

and we see that

$$\mathfrak{P}_{12} = -(\gamma_2^2 - \gamma_1\gamma_2 + 1)\lambda^2 + 2X_4\gamma_2\lambda + 2X_2 - 2\mu, \qquad \mathfrak{P}_{13} = 3\gamma_2^2\lambda^3,$$

$$\mathfrak{P}_{33} = (2\gamma_2^2 - 2\gamma_1\gamma_2 - 1)\lambda^2 - 4X_4\gamma_2\lambda - 4X_2 - 2\mu, \qquad \mathfrak{P}_{44} = -3\gamma_2^2\lambda^2 - 2\mu.$$

One easily checks that

$$2\mathfrak{P}_{12} - rac{3}{\lambda}\mathfrak{P}_{13} + \mathfrak{P}_{33} - 3\mathfrak{P}_{44} = -3\lambda^2.$$

Since $\lambda \neq 0$, there are no left-invariant Ricci solitons in this case.

4.2 Extensions of Riemannian Lie groups

In this section we analyse the left-invariant Lorentzian metrics which are extensions of threedimensional unimodular Riemannian Lie groups. In particular, we show that any left-invariant Ricci soliton in this setting is trivial.

Lemma 4.8. A four-dimensional Lie group $G = G_3 \rtimes \mathbb{R}$ equipped with a left-invariant Lorentzian metric whose restriction to G_3 is Riemannian is a left-invariant Ricci soliton if and only if it is a space of non-negative constant sectional curvature.

Let $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ and let *L* be the structure operator of \mathfrak{g}_3 . Since *L* is self-adjoint and diagonalizable, there exists an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ of \mathfrak{g} , with timelike e_4 , where $\mathfrak{g}_3 = \operatorname{span}\{e_1, e_2, e_3\}$ and $\mathfrak{r} = \operatorname{span}\{e_4\}$, so that

$$[e_1, e_2] = \lambda_3 e_3, \quad [e_1, e_3] = -\lambda_2 e_2, \quad [e_2, e_3] = \lambda_1 e_1, \quad [e_i, e_4] = \sum_{\substack{j=1 \\ (i=1,2,3)}}^3 \alpha_i^j e_j,$$

for certain $\alpha_i^j \in \mathbb{R}$. Next, depending on the eigenvalues λ_i and imposing the Jacobi identity, we are led to the following different possibilities.

4.2.1 Case of non-zero eigenvalues: $\widetilde{SL}(2,\mathbb{R})\times\mathbb{R}$ and $SU(2)\times\mathbb{R}$

If $\lambda_1 \lambda_2 \lambda_3 \neq 0$, the left-invariant Lorentzian metrics are described by

$$[e_1, e_2] = \lambda_3 e_3, \quad [e_1, e_3] = -\lambda_2 e_2, \qquad [e_1, e_4] = \gamma_1 \lambda_2 e_2 + \gamma_2 \lambda_3 e_3,$$
$$[e_2, e_3] = \lambda_1 e_1, \quad [e_2, e_4] = -\gamma_1 \lambda_1 e_1 + \gamma_3 \lambda_3 e_3, \quad [e_3, e_4] = -\gamma_2 \lambda_1 e_1 - \gamma_3 \lambda_2 e_2,$$

and, proceeding as in Section 4.1.1, a straightforward calculation shows that there are no left-invariant Ricci solitons in this case.

4.2.2 Case of a single null eigenvalue: $\tilde{E}(2) \rtimes \mathbb{R}$ and $E(1,1) \rtimes \mathbb{R}$

Without loss of generality, we assume $\lambda_3 = 0$ and $\lambda_1 \lambda_2 \neq 0$. The Lorentzian left-invariant metrics on $\tilde{E}(2) \rtimes \mathbb{R}$ or $E(1,1) \rtimes \mathbb{R}$ are given by

$$[e_1, e_3] = -\lambda_2 e_2, \qquad [e_1, e_4] = \gamma_1 e_1 + \gamma_2 \lambda_2 e_2, \quad [e_2, e_3] = \lambda_1 e_1,$$
$$[e_2, e_4] = -\gamma_2 \lambda_1 e_1 + \gamma_1 e_2, \quad [e_3, e_4] = \gamma_3 e_1 + \gamma_4 e_2.$$

Proceeding as in Section 4.1.1, we see that the existence of left-invariant Ricci solitons leads to the conditions $\lambda_2 = \lambda_1$, $\gamma_1 = \gamma_3 = \gamma_4 = 0$, which determine flat metrics on $\tilde{E}(2) \rtimes \mathbb{R}$.

4.2.3 Structure operator of rank one: $H^3 \rtimes \mathbb{R}$

Set $\lambda_1 = \lambda_2 = 0$ and $\lambda_3 \neq 0$ so the left-invariant Lorentzian metrics can be expressed as

$$[e_1, e_2] = \lambda_3 e_3, \qquad [e_1, e_4] = \gamma_1 e_1 + \gamma_2 e_2 + \gamma_3 e_3,$$
$$[e_2, e_4] = \gamma_4 e_1 + \gamma_5 e_2 + \gamma_6 e_3, \qquad [e_3, e_4] = (\gamma_1 + \gamma_5) e_3.$$

A straightforward calculation as in Section 4.1.1 shows that there are no left-invariant Ricci solitons in this case.

4.2.4 Case of zero eigenvalues: $\mathbb{R}^3 \rtimes \mathbb{R}$

Proceeding as in Section 4.1.1, the left-invariant metrics are described by

$$[e_1, e_4] = \eta_1 e_1 - \gamma_1 e_2 - \gamma_2 e_3, \qquad [e_2, e_4] = \gamma_1 e_1 + \eta_2 e_2 - \gamma_3 e_3,$$
$$[e_3, e_4] = \gamma_2 e_1 + \gamma_3 e_2 + \eta_3 e_3.$$

Analogous calculations to those in Section 4.1.1 show that $\mathbb{R}^3 \rtimes \mathbb{R}$ is a left-invariant Ricci soliton if and only if $\eta_1 = \eta_2 = \eta_3 = \kappa$, in which case the sectional curvature is constantly κ^2 .

4.3 Extensions of degenerate Lie groups

In this section we will study left-invariant Lorentzian metrics which are extensions of threedimensional unimodular Lie groups with a degenerate metric. We will show that the underlying structure of any non-Einstein soliton is either a plane wave (obtained in Section 4.3.1 and Section 4.3.2) or a symmetric product $\mathbb{L}^2 \times N(c)$ (studied in Section 4.3.2). While the products of the form $\mathbb{L}^2 \times N(c)$, discussed in Section 4.3.2, are left-invariant Ricci solitons, the case of plane waves is more complicated and will be analysed in Section 4.4.

Let $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ be a four-dimensional Lie algebra with a Lorentzian inner product $\langle \cdot, \cdot \rangle$ which is degenerate when restricted to \mathfrak{g}_3 . Let $\mathfrak{g}'_3 = [\mathfrak{g}_3, \mathfrak{g}_3]$ be the derived subalgebra of \mathfrak{g}_3 . We consider the different cases given by dim $\mathfrak{g}'_3 \in \{0, 1, 2, 3\}$ separately.

4.3.1 dim $\mathfrak{g}'_3 = 0$: left-invariant metrics on $\mathbb{R}^3 \rtimes \mathbb{R}$

In this case the Lie algebra \mathfrak{g}_3 is Abelian. There exists a pseudo-orthonormal basis $\{u_1, u_2, u_3, u_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \operatorname{span}\{u_4\}$ with $\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle = 1$, so that

$$[u_1, u_4] = \gamma_1 u_1 + \gamma_2 u_2 + \gamma_3 u_3, \qquad [u_2, u_4] = \gamma_4 u_1 + \gamma_5 u_2 + \gamma_6 u_3,$$
$$[u_3, u_4] = \gamma_7 u_1 + \gamma_8 u_2 + \gamma_9 u_3,$$

for $\gamma_i \in \mathbb{R}$. A straightforward calculation leads to the polynomials

$$\mathfrak{P}_{11} = -\gamma_7^2 + 4X_4\gamma_1 - 2\mu, \qquad \mathfrak{P}_{13} = 2X_4\gamma_7,$$

$$\mathfrak{P}_{34} = \gamma_7^2 + \gamma_8^2 + 2X_4\gamma_9 - 2\mu, \qquad \mathfrak{P}_{23} = 2X_4\gamma_8.$$

It follows from the expressions of \mathfrak{P}_{13} and \mathfrak{P}_{23} , together with the combination

$$\mathfrak{P}_{11} - \mathfrak{P}_{34} = -2\gamma_7^2 - \gamma_8^2 + 2X_4(2\gamma_1 - \gamma_9),$$

that $\gamma_7 = \gamma_8 = 0$. Therefore, the left-invariant metric is given by

$$[u_1, u_4] = \gamma_1 u_1 + \gamma_2 u_2 + \gamma_3 u_3, \qquad [u_3, u_4] = \gamma_9 u_3,$$

$$[u_2, u_4] = \gamma_4 u_1 + \gamma_5 u_2 + \gamma_6 u_3.$$

(4.4)

It is not difficult to check that u_3 is a recurrent null vector such that R(Y, Z) = 0 and $\nabla_Y R = 0$ for all $Y, Z \in u_3^{\perp}$. Moreover, the only non-zero component of the Ricci tensor is

$$\rho_{44} = -\gamma_1^2 - \frac{1}{2}(\gamma_2 + \gamma_4)^2 - \gamma_5^2 + (\gamma_1 + \gamma_5)\gamma_9,$$

which shows that the Ricci operator is isotropic and the underlying structure is a plane wave.

4.3.2 dim $\mathfrak{g}'_3 = 1$: left-invariant metrics on $H^3 \rtimes \mathbb{R}$

Since the restriction of the metric to \mathfrak{g}_3 has signature (+, +, 0), then $\mathfrak{g}'_3 = \operatorname{span}\{v\}$ can be a null or a spacelike subspace. We analyse these two possibilities separately.

$\mathfrak{g}_3' = \operatorname{span}\{v\}$ is a null subspace

In this case, setting $u_3 = v$, there exists a pseudo-orthonormal basis $\{u_1, u_2, u_3, u_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ with $\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle = 1$, where $\mathfrak{g}_3 = \operatorname{span}\{u_1, u_2, u_3\}$ and $\mathfrak{r} = \operatorname{span}\{u_4\}$, so that

$$[u_1, u_2] = \lambda_1 u_3, \quad [u_1, u_3] = \lambda_2 u_3, \quad [u_2, u_3] = \lambda_3 u_3, \quad [u_i, u_4] = \sum_{j=1}^3 \alpha_i^j u_j,$$

for certain $\alpha_i^j \in \mathbb{R}$ and where at least one of λ_1 , λ_2 and λ_3 is non-zero. We are led to the following different possibilities depending on the values of the λ_i 's.

Case $\lambda_2 = \lambda_3 = 0$

If $\lambda_2 = \lambda_3 = 0$, then necessarily $\lambda_1 \neq 0$ and

$$[u_1, u_2] = \lambda_1 u_3, \qquad [u_1, u_4] = \gamma_1 u_1 + \gamma_2 u_2 + \gamma_3 u_3, [u_2, u_4] = \gamma_4 u_1 + \gamma_5 u_2 + \gamma_6 u_3, \qquad [u_3, u_4] = (\gamma_1 + \gamma_5) u_3.$$
(4.5)

A standard calculation shows that u_3 is a recurrent vector field and the curvature tensor satisfies R(Y, Z) = 0 and $\nabla_Y R = 0$ for all $Y, Z \in u_3^{\perp}$. The only non-zero component of the Ricci tensor is

$$\rho_{44} = \frac{1}{2} \left\{ \lambda_1^2 + 4\gamma_1 \gamma_5 - (\gamma_2 + \gamma_4)^2 \right\},\,$$

which shows that the Ricci operator is isotropic. Therefore, the underlying structure is a plane wave.

Case $\lambda_2 = 0$, $\lambda_3 \neq 0$

In this case one has

$$\begin{split} & [u_1, u_2] = \lambda_1 u_3, & [u_1, u_4] = \gamma_1 \lambda_3 u_1 + (\gamma_1 - \gamma_2) \lambda_1 u_3, & [u_2, u_3] = \lambda_3 u_3, \\ & [u_2, u_4] = \gamma_3 u_1 + \gamma_4 u_3, & [u_3, u_4] = \gamma_2 \lambda_3 u_3, \end{split}$$

and the non-zero polynomials \mathfrak{P}_{ij} are given by

$$\begin{split} \mathfrak{P}_{11} &= 4X_4 \gamma_1 \lambda_3 - 2\mu, \\ \mathfrak{P}_{12} &= 2X_4 \gamma_3, \\ \mathfrak{P}_{14} &= \lambda_1 \lambda_3 + 2(X_4(\gamma_1 - \gamma_2) + X_2)\lambda_1 - 2X_1 \gamma_1 \lambda_3 - 2X_2 \gamma_3, \\ \mathfrak{P}_{22} &= -\lambda_3^2 - 2\mu, \\ \mathfrak{P}_{24} &= -\gamma_1 \lambda_3^2 - 2X_1 \lambda_1 + 2X_3 \lambda_3 + 2X_4 \gamma_4, \\ \mathfrak{P}_{34} &= (2X_4 \gamma_2 - 2X_2)\lambda_3 - \lambda_3^2 - 2\mu, \\ \mathfrak{P}_{44} &= \lambda_1^2 - 2\gamma_1 (\gamma_1 - \gamma_2) \lambda_3^2 - 4X_1 (\gamma_1 - \gamma_2)\lambda_1 - 4X_3 \gamma_2 \lambda_3 - 4X_2 \gamma_4 - \gamma_3^2 \,. \end{split}$$

It is easy to check that

$$\mathfrak{P}_{11} - \mathfrak{P}_{22} = (\lambda_3 + 4X_4\gamma_1)\lambda_3$$

which implies that $X_4 \neq 0$ (since $\lambda_3 = 0$). Now, this combination, together with the expressions of \mathfrak{P}_{12} and \mathfrak{P}_{22} , lead to

$$\gamma_3 = 0, \quad \gamma_1 = -\frac{1}{4X_4}\lambda_3, \quad \mu = -\frac{1}{2}\lambda_3^2,$$

and a direct calculation shows that

$$2\lambda_1\mathfrak{P}_{14} - 2\gamma_2\lambda_3\mathfrak{P}_{24} + \frac{2}{\lambda_3}(\lambda_1^2 - \gamma_3\lambda_1 + \gamma_4\lambda_3)\mathfrak{P}_{34} - \lambda_3\mathfrak{P}_{44} = \frac{1}{8X_4^2}\lambda_3^5.$$

Since $\lambda_3 \neq 0$, there are no left-invariant Ricci solitons in this case.

Case $\lambda_2 \neq 0$

If $\lambda_2 \neq 0$, then the Lie algebra structure is given by

$$\begin{split} & [u_1, u_2] = \lambda_1 u_3, \quad [u_1, u_3] = \lambda_2 u_3, \quad [u_2, u_3] = \lambda_3 u_3, \quad [u_3, u_4] = \gamma_4 \lambda_2 u_3, \\ & [u_1, u_4] = -\gamma_1 \lambda_2 \lambda_3 u_1 + \gamma_1 \lambda_2^2 u_2 + \gamma_2 \lambda_2 u_3, \\ & [u_2, u_4] = -\gamma_3 \lambda_3 u_1 + \gamma_3 \lambda_2 u_2 + (\gamma_1 \lambda_1 \lambda_3 - (\gamma_3 - \gamma_4) \lambda_1 + \gamma_2 \lambda_3) u_3, \end{split}$$

and the non-zero polynomials \mathfrak{P}_{ij} are

$$\begin{split} \mathfrak{P}_{11} &= -\lambda_2^2 - 4X_4 \gamma_1 \lambda_2 \lambda_3 - 2\mu, \\ \mathfrak{P}_{12} &= 2X_4 \gamma_1 \lambda_2^2 - \lambda_2 \lambda_3 - 2X_4 \gamma_3 \lambda_3, \\ \mathfrak{P}_{14} &= \gamma_1 \lambda_2^2 \lambda_3 - \gamma_3 \lambda_2^2 + \lambda_1 \lambda_3 + 2X_2 (\lambda_1 + \gamma_3 \lambda_3) + 2(X_4 \gamma_2 + X_3 + X_1 \gamma_1 \lambda_3) \lambda_2, \\ \mathfrak{P}_{22} &= -\lambda_3^2 + 4X_4 \gamma_3 \lambda_2 - 2\mu, \\ \mathfrak{P}_{24} &= \gamma_1 \lambda_2 \lambda_3^2 - 2X_1 \gamma_1 \lambda_2^2 - \lambda_1 \lambda_2 + 2X_4 \gamma_1 \lambda_1 \lambda_3 - \gamma_3 \lambda_2 \lambda_3 \\ &\quad - 2(X_4 (\gamma_3 - \gamma_4) + X_1) \lambda_1 - 2X_2 \gamma_3 \lambda_2 + 2(X_4 \gamma_2 + X_3) \lambda_3, \\ \mathfrak{P}_{34} &= -\lambda_2^2 - \lambda_3^2 + 2(X_4 \gamma_4 - X_1) \lambda_2 - 2X_2 \lambda_3 - 2\mu, \\ \mathfrak{P}_{44} &= -\gamma_1^2 \lambda_2^4 - 2\gamma_1^2 \lambda_2^2 \lambda_3^2 + 2\gamma_1 (\gamma_3 - \gamma_4) \lambda_2^2 \lambda_3 + \lambda_1^2 - 2\gamma_3 (\gamma_3 - \gamma_4) \lambda_2^2 - \gamma_3^2 \lambda_3^2 \\ &\quad - 4X_2 \gamma_1 \lambda_1 \lambda_3 + 4X_2 (\gamma_3 - \gamma_4) \lambda_1 - 4(X_1 \gamma_2 + X_3 \gamma_4) \lambda_2 - 4X_2 \gamma_2 \lambda_3 \,. \end{split}$$

Since $\lambda_2 \neq 0$, if we consider the expressions

$$\mathfrak{P}_{11} - \mathfrak{P}_{22} = -\lambda_2^2 + \lambda_3^2 - 4X_4(\gamma_1\lambda_3 + \gamma_3)\lambda_2,$$
$$\mathfrak{P}_{12} = -\lambda_2\lambda_3 + 2X_4(\gamma_1\lambda_2^2 - \gamma_3\lambda_3),$$

then $X_4 \neq 0$. Now, from the expressions of $\mathfrak{P}_{12}, \mathfrak{P}_{11}$ and \mathfrak{P}_{22} we obtain

$$\gamma_1 = \frac{(\lambda_2 + 2X_4\gamma_3)\lambda_3}{2X_4\lambda_2^2}, \quad \mu = -\frac{\lambda_2^3 + 2(\lambda_2 + 2X_4\gamma_3)\lambda_3^2}{2\lambda_2}, \quad \gamma_3 = -\frac{\lambda_2}{4X_4},$$

and a direct calculation shows that

$$2(\gamma_4\lambda_2^2 - \lambda_1\lambda_3)\mathfrak{P}_{14} + 2(\lambda_1 + \gamma_4\lambda_3)\lambda_2\mathfrak{P}_{24} + (\lambda_2^2 + \lambda_3^2)\mathfrak{P}_{44}$$
$$-2\left(\lambda_1^2 + (\gamma_1\lambda_1 + \gamma_2)\left(\lambda_2^2 + \lambda_3^2\right) + \gamma_4\lambda_1\lambda_3\right)\mathfrak{P}_{34} = -\frac{(\lambda_2^2 + \lambda_3^2)^3}{8X_4^2}$$

Since $\lambda_2 \neq 0$, there are no left-invariant Ricci solitons in this case.

$\mathfrak{g}_3' = \operatorname{span}\{v\}$ is a spacelike subspace

Considering $u_1 = \frac{v}{\|v\|}$, there exists a pseudo-orthonormal basis $\{u_1, u_2, u_3, u_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ with $\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle = 1$, where $\mathfrak{g}_3 = \operatorname{span}\{u_1, u_2, u_3\}$ and $\mathfrak{r} = \operatorname{span}\{u_4\}$, so that

$$[u_1, u_2] = \lambda_1 u_1, \quad [u_1, u_3] = \lambda_2 u_1, \quad [u_2, u_3] = \lambda_3 u_1, \quad [u_i, u_4] = \sum_{j=1}^3 \alpha_i^j u_j,$$

for certain $\alpha_i^j \in \mathbb{R}$ and where at least one of λ_1, λ_2 and λ_3 is non-zero. Depending on the values of the λ_i 's, we are led to the following different possibilities.

Case $\lambda_1 = \lambda_2 = 0$

Since $\lambda_3 \neq 0$ one has the Lie algebra structure

$$[u_1, u_4] = \gamma_1 u_1, \qquad [u_2, u_3] = \lambda_3 u_1,$$
$$[u_2, u_4] = \gamma_2 u_1 + \gamma_3 u_2 + \gamma_4 u_3, \qquad [u_3, u_4] = \gamma_5 u_1 + \gamma_6 u_2 + (\gamma_1 - \gamma_3) u_3.$$

A direct calculation shows $\mathfrak{P}_{33} = -\lambda_3^2$. Therefore, there are no left-invariant Ricci solitons in this case.

Case $\lambda_1 = 0$, $\lambda_2 \neq 0$

In this case, the left-invariant metrics are described as

$$[u_1, u_3] = \lambda_2 u_1, \qquad [u_1, u_4] = \gamma_1 \lambda_2 u_1, \qquad [u_2, u_3] = \lambda_3 u_1,$$
$$[u_2, u_4] = (\gamma_1 - \gamma_2) \lambda_3 u_1 + \gamma_2 \lambda_2 u_2, \quad [u_3, u_4] = \gamma_3 u_1 + \gamma_4 u_2.$$

It now follows from $\mathfrak{P}_{33} = -2\lambda_2^2 - \lambda_3^2$ that there are no left-invariant Ricci solitons in this case.

Case $\lambda_1 \neq 0$

If $\lambda_1 \neq 0$, then the Lie algebra structure takes the form

$$[u_1, u_2] = \lambda_1 u_1, \quad [u_1, u_3] = \lambda_2 u_1, \quad [u_1, u_4] = \gamma_1 \lambda_1 u_1, \quad [u_2, u_3] = \lambda_3 u_1,$$
$$[u_2, u_4] = \lambda_1 \gamma_2 u_1 - \gamma_3 \lambda_1 \lambda_2 u_2 + \gamma_3 \lambda_1^2 u_3,$$
$$[u_3, u_4] = -(\gamma_3 \lambda_2 \lambda_3 - \gamma_2 \lambda_2 + (\gamma_1 - \gamma_4) \lambda_3) u_1 - \gamma_4 \lambda_2 u_2 + \gamma_4 \lambda_1 u_3,$$

and the polynomials \mathfrak{P}_{ij} are

$$\begin{aligned} \mathfrak{P}_{11} &= -\gamma_3^2 \lambda_2^2 \lambda_3^2 + 2\gamma_3 \lambda_1 \lambda_2^2 + 2\gamma_2 \gamma_3 \lambda_2^2 \lambda_3 - 2(\gamma_1 - \gamma_4) \gamma_3 \lambda_2 \lambda_3^2 - 2\lambda_1^2 - \gamma_2^2 \lambda_2^2 \\ &- (\gamma_1 - \gamma_4)^2 \lambda_3^2 - 2(2\gamma_1 + \gamma_4) \lambda_1 \lambda_2 + 2\gamma_2 \lambda_1 \lambda_3 + 2(\gamma_1 - \gamma_4) \gamma_2 \lambda_2 \lambda_3 \\ &+ 4(X_4 \gamma_1 + X_2) \lambda_1 + 4X_3 \lambda_2 - 2\mu, \end{aligned}$$
$$\begin{aligned} \mathfrak{P}_{12} &= -\gamma_3 \gamma_4 \lambda_2^2 \lambda_3 + \gamma_2 \gamma_4 \lambda_2^2 - 2\gamma_2 \lambda_1 \lambda_2 - (2\gamma_1 + \gamma_4) \lambda_1 \lambda_3 - (\gamma_1 - \gamma_4) \gamma_4 \lambda_2 \lambda_3 \\ &+ 2(X_4 \gamma_2 - X_1) \lambda_1 + 2X_3 \lambda_3, \end{aligned}$$

$$\mathfrak{P}_{13} = 2\gamma_3\lambda_2^2\lambda_3 - 2\gamma_2\lambda_2^2 + 2\lambda_1\lambda_3 + 2(\gamma_1 - \gamma_4 - X_4\gamma_3)\lambda_2\lambda_3 + (2X_4\gamma_2 - 2X_1)\lambda_2 - 2(X_4(\gamma_1 - \gamma_4) + X_2)\lambda_3,$$

$$\mathfrak{P}_{14} = \gamma_3^2 \lambda_1 \lambda_2^2 \lambda_3 + \gamma_3 \lambda_1^2 \lambda_3 - \gamma_2 \gamma_3 \lambda_1 \lambda_2^2 - (\gamma_1 + \gamma_4) \gamma_3 \lambda_1 \lambda_2 \lambda_3 + 2\gamma_2 \lambda_1^2 + 2\gamma_1 \gamma_2 \lambda_1 \lambda_2 \\ - 2\gamma_1 (\gamma_1 - \gamma_4) \lambda_1 \lambda_3 + 2X_3 \gamma_3 \lambda_2 \lambda_3 - 2(X_1 \gamma_1 + X_2 \gamma_2) \lambda_1 - 2X_3 \gamma_2 \lambda_2$$

$$\begin{aligned} &+ 2X_{3}(\gamma_{1} - \gamma_{4})\lambda_{3}, \\ \mathfrak{P}_{22} &= 2\gamma_{3}\lambda_{1}\lambda_{2}^{2} - 2\lambda_{1}^{2} - \gamma_{4}^{2}\lambda_{2}^{2} - 4X_{4}\gamma_{3}\lambda_{1}\lambda_{2} - 2\gamma_{2}\lambda_{1}\lambda_{3} - 2\mu, \\ \mathfrak{P}_{23} &= \gamma_{3}\lambda_{2}\lambda_{3}^{2} + \gamma_{4}\lambda_{2}^{2} + (\gamma_{1} - \gamma_{4})\lambda_{3}^{2} - 2\lambda_{1}\lambda_{2} - \gamma_{2}\lambda_{2}\lambda_{3} - 2X_{4}\gamma_{4}\lambda_{2}, \\ \mathfrak{P}_{24} &= -2\gamma_{3}\lambda_{1}^{2}\lambda_{2} + 2\gamma_{3}\gamma_{4}\lambda_{1}\lambda_{2}^{2} - \gamma_{2}\gamma_{3}\lambda_{1}\lambda_{2}\lambda_{3} - 2(\gamma_{1} - X_{4}\gamma_{3})\lambda_{1}^{2} + (\gamma_{2}^{2} - \gamma_{1}\gamma_{4} + 2X_{2}\gamma_{3})\lambda_{1}\lambda_{2} \\ &- (\gamma_{1} - \gamma_{4})\gamma_{2}\lambda_{1}\lambda_{3} + 2X_{3}\gamma_{4}\lambda_{2}, \end{aligned}$$

$$\mathfrak{P}_{33} &= -2\lambda_{2}^{2} - \lambda_{3}^{2}, \\ \mathfrak{P}_{34} &= \gamma_{3}^{2}\lambda_{2}^{2}\lambda_{3}^{2} - 2\gamma_{2}\gamma_{3}\lambda_{2}^{2}\lambda_{3} + 2(\gamma_{1} - \gamma_{4})\gamma_{3}\lambda_{2}\lambda_{3}^{2} + (\gamma_{2}^{2} + \gamma_{4}^{2})\lambda_{2}^{2} + (\gamma_{1} - \gamma_{4})^{2}\lambda_{3}^{2} \\ &- (2\gamma_{1} + \gamma_{4})\lambda_{1}\lambda_{2} - \gamma_{2}\lambda_{1}\lambda_{3} - 2(\gamma_{1} - \gamma_{4})\gamma_{2}\lambda_{2}\lambda_{3} + 2X_{4}\gamma_{4}\lambda_{1} - 2\mu, \end{aligned}$$

$$\mathfrak{P}_{44} &= -2\gamma_{3}^{2}\lambda_{1}^{2}\lambda_{2}^{2} + 2\gamma_{3}\lambda_{1}^{3} - (2\gamma_{1}^{2} + \gamma_{2}^{2} - 2\gamma_{1}\gamma_{4} + 4X_{2}\gamma_{3})\lambda_{1}^{2} - 4X_{3}\gamma_{4}\lambda_{1}. \end{aligned}$$

It follows from the expression of \mathfrak{P}_{33} that both λ_2 and λ_3 must be zero. Now,

$$\mathfrak{P}_{22} - \mathfrak{P}_{34} = -2(\lambda_1 + X_4\gamma_4)\lambda_1$$

implies that $X_4 \neq 0$ and $\gamma_4 \neq 0$. At this point, the expressions

$$\gamma_2 \mathfrak{P}_{11} - 2\gamma_1 \mathfrak{P}_{12} + 2 \mathfrak{P}_{14} - \gamma_2 \mathfrak{P}_{22} = 4\gamma_2 \lambda_1^2,$$

$$\gamma_1 \mathfrak{P}_{22} - \mathfrak{P}_{24} - \gamma_1 \mathfrak{P}_{34} = -2X_4 (\gamma_3 \lambda_1 + \gamma_1 \gamma_4) \lambda_1,$$

lead to $\gamma_2 = 0$, and $\gamma_1 = -\frac{\gamma_3\lambda_1}{\gamma_4}$. Finally, a standard calculation shows that the corresponding left-invariant metric, which given by

$$[u_1, u_2] = \lambda_1 u_1, \quad [u_1, u_4] = -\frac{\gamma_3 \lambda_1^2}{\gamma_4} u_1, \quad [u_2, u_4] = \gamma_3 \lambda_1^2 u_3, \quad [u_3, u_4] = \gamma_4 \lambda_1 u_3,$$

is symmetric and locally isometric to a product $\mathbb{L}^2 \times N(c)$, where N is a surface of constant curvature c. Furthermore, it is a expanding Ricci soliton with $\mu = -\lambda_1^2$ and left-invariant soliton vector field

$$X = -\frac{\gamma_3 \lambda_1^2}{\gamma_4^2} u_2 + \frac{\gamma_3^2 \lambda_1^3}{2\gamma_4^3} u_3 - \frac{\lambda_1}{\gamma_4} u_4.$$

4.3.3 dim $\mathfrak{g}'_3 = 2$: left-invariant metrics on $\tilde{E}(2) \rtimes \mathbb{R}$ and $E(1,1) \rtimes \mathbb{R}$

Let $\mathfrak{g}' = [\mathfrak{g}, \mathfrak{g}]$ be the derived subalgebra of \mathfrak{g} . We can assume that $\mathfrak{g}' = \mathfrak{g}_3$ without losing generality. Indeed, if dim $\mathfrak{g}' < 3$, then there exist two linearly independent vectors $x_1, x_2 \in \mathfrak{g}$ acting as derivations on \mathfrak{g} . Since \mathfrak{g} is Lorentzian, we can choose a non-null vector $y \in \operatorname{span}\{x_1, x_2\}$ so that $\mathfrak{g} = \mathfrak{h} \rtimes \operatorname{span}\{y\}$, where the restriction of the metric to the three-dimensional subalgebra \mathfrak{h} is non-degenerate. Thus, \mathfrak{g} corresponds to one of the cases already studied in Sections 4.1 and 4.2.

Let $\mathfrak{g}'_3 = \operatorname{span}\{w_1, w_2\}$ with $w_i = v_i + \xi_i u_3$, where v_i is a spacelike vector field and u_3 is null and orthogonal to v_1 and v_2 .

If $\{v_1, v_2\}$ are linearly independent, which means that \mathfrak{g}'_3 is a spacelike subspace, we choose an orthonormal basis $\{u_1, u_2\}$ for span $\{v_1, v_2\}$ that can be completed to a pseudo-orthonormal basis $\{u_1, u_2, u_3, u_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ such that $\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle = 1$, where $\mathfrak{g}_3 = \operatorname{span}\{u_1, u_2, u_3\}$ and $\mathfrak{r} = \operatorname{span}\{u_4\}$. In this situation, the left-invariant metrics are of the form

$$[u_1, u_2] = \gamma_1 u_1 + \gamma_2 u_2, \qquad [u_1, u_3] = \gamma_3 u_1 + \gamma_4 u_2,$$
$$[u_2, u_3] = \gamma_5 u_1 + \gamma_6 u_2, \qquad [u_i, u_4] = \sum_{j=1}^3 \alpha_i^j u_j,$$
$$_{(i=1,2,3)}$$

for certain $\gamma_i, \alpha_i^j \in \mathbb{R}$.

If $\{v_1, v_2\}$ are linearly dependent, i.e., the restriction of the metric to \mathfrak{g}'_3 is degenerate, then $\{u_1 = \frac{v_1}{\|v_1\|}, u_3\}$ is a basis of \mathfrak{g}'_3 that can be completed to a pseudo-orthonormal basis $\{u_1, u_2, u_3, u_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ such that $\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle = 1$, where $\mathfrak{g}_3 =$ $\operatorname{span}\{u_1, u_2, u_3\}$ and $\mathfrak{r} = \operatorname{span}\{u_4\}$. In this case, the left-invariant metrics are expressed as

$$[u_1, u_2] = \gamma_1 u_1 + \gamma_2 u_3, \qquad [u_1, u_3] = \gamma_3 u_1 + \gamma_4 u_3,$$
$$[u_2, u_3] = \gamma_5 u_1 + \gamma_6 u_3, \qquad [u_i, u_4] = \sum_{j=1}^3 \alpha_i^j u_j,$$

for certain $\gamma_i, \alpha_i^j \in \mathbb{R}$.

In both cases above, a straightforward calculation shows that the Jacobi identity is not satisfied since dim $g'_3 = 2$ and dim g' = 3. Therefore, there are no left-invariant Ricci solitons in this case.

4.3.4 dim $\mathfrak{g}'_3 = 3$: left-invariant metrics on $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$

In this case, $\mathfrak{g}'_3 = \mathfrak{g}_3$ and we consider the pseudo-orthonormal basis $\{u_1, u_2, u_3, u_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \operatorname{span}\{u_4\}$ with $\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle$ and $\operatorname{ad}_{u_3} : \mathfrak{g}_3 \to \mathfrak{g}_3$. Since $\mathfrak{g}'_3 = \mathfrak{g}_3$, ad_{u_3} must be of rank 2 and, apart from 0, it must have either two real eigenvalues or two conjugate complex eigenvalues. Moreover, writing

$$u_3 = [x_1, x_2], \qquad x_1, x_2 \in \mathfrak{g}_3,$$

we have

$$\mathrm{ad}_{u_3} = \mathrm{ad}_{x_1} \circ \mathrm{ad}_{x_2} - \mathrm{ad}_{x_2} \circ \mathrm{ad}_{x_1},$$

which implies $tr(ad_{u_3}) = 0$. Consequently, there are two different possibilities, none of them supporting left-invariant Ricci solitons.

ad_{u_3} has real eigenvalues $\{0, \lambda, -\lambda\}$, with $\lambda \neq 0$

Let v_1 and v_2 be unit eigenvectors, i.e., $[v_1, u_3] = \lambda v_1$ and $[v_2, u_3] = -\lambda v_2$. The Jacobi identity implies $[v_1, v_2] \in \text{span}\{u_3\}$. Thus, rescaling u_3 if necessary, we obtain a basis $\{v_1, v_2, v_3, v_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ with $\langle v_1, v_1 \rangle = \langle v_2, v_2 \rangle = \langle v_3, v_4 \rangle = 1$ and $\langle v_1, v_2 \rangle = \kappa \neq \pm 1$, where $\mathfrak{g}_3 = \text{span}\{v_1, v_2, v_3\}$ and $\mathfrak{r} = \text{span}\{v_4\}$. In this situation

$$\begin{bmatrix} v_1, v_2 \end{bmatrix} = v_3, \qquad \begin{bmatrix} v_1, v_3 \end{bmatrix} = \lambda v_1, \qquad \begin{bmatrix} v_1, v_4 \end{bmatrix} = \gamma_1 v_1 + \gamma_2 v_3, \\ \begin{bmatrix} v_2, v_3 \end{bmatrix} = -\lambda v_2, \qquad \begin{bmatrix} v_2, v_4 \end{bmatrix} = -\gamma_1 v_2 + \gamma_3 v_3, \qquad \begin{bmatrix} v_3, v_4 \end{bmatrix} = \gamma_3 \lambda v_1 + \gamma_2 \lambda v_2$$

We compute $\mathfrak{P}_{33} = \frac{4\lambda^2}{\kappa^2 - 1}$ and, since $\lambda \neq 0$, there are no left-invariant Ricci solitons in this case.

 ad_{u_3} has complex eigenvalues $\{0, i\beta, -i\beta\}$, with $\beta \neq 0$

Let v_1 and v_2 be unit vectors so that $[v_1, u_3] = \beta v_2$ and $[v_2, u_3] = -\beta v_1$. The Jacobi identity implies $[v_1, v_2] \in \text{span}\{u_3\}$. Thus, rescaling u_3 if necessary, we obtain a basis $\{v_1, v_2, v_3, v_4\}$ of $\mathfrak{g} = \mathfrak{g}_3 \rtimes \mathfrak{r}$ with $\langle v_1, v_1 \rangle = \langle v_2, v_2 \rangle = \langle v_3, v_4 \rangle = 1$ and $\langle v_1, v_2 \rangle = \kappa \neq \pm 1$, where $\mathfrak{g}_3 = \text{span}\{v_1, v_2, v_3\}$ and $\mathfrak{r} = \text{span}\{v_4\}$. In this case

$$\begin{bmatrix} v_1, v_2 \end{bmatrix} = v_3, \qquad \begin{bmatrix} v_1, v_3 \end{bmatrix} = \beta v_2, \qquad \begin{bmatrix} v_1, v_4 \end{bmatrix} = \gamma_1 v_2 + \gamma_2 v_3, \\ \begin{bmatrix} v_2, v_3 \end{bmatrix} = -\beta v_1, \qquad \begin{bmatrix} v_2, v_4 \end{bmatrix} = -\gamma_1 v_1 + \gamma_3 v_3, \qquad \begin{bmatrix} v_3, v_4 \end{bmatrix} = \gamma_2 \beta v_1 + \gamma_3 \beta v_2$$

We will consider the polynomials $\tilde{\mathfrak{P}}_{ij} = (\kappa^2 - 1)\mathfrak{P}_{ij}$ given by

$$\begin{split} \tilde{\mathfrak{P}}_{11} &= -(\kappa^2 - 1)\beta^2(\gamma_2 + \kappa\gamma_3)^2 - 4(2\kappa\beta - X_4(\kappa^2 - 1))\kappa\gamma_1 \\ &+ 2(2X_3\kappa^3 - \kappa^2 - 2X_3\kappa + 1)\beta - 2(\kappa^2 - 1)\mu, \\ \tilde{\mathfrak{P}}_{12} &= -(\kappa^2 - 1)\kappa\beta^2(\gamma_2^2 + \gamma_3^2) - (\kappa^4 - 1)\beta^2\gamma_2\gamma_3 - 8\kappa\beta\gamma_1 - 2(\kappa^2 - 1)\kappa\mu, \\ \tilde{\mathfrak{P}}_{22} &= -(\kappa^2 - 1)\beta^2(\kappa\gamma_2 + \gamma_3)^2 - 4(2\kappa\beta + X_4(\kappa^2 - 1))\kappa\gamma_1 \\ &- 2(2X_3\kappa^3 + \kappa^2 - 2\kappa X_3 - 1)\beta - 2(\kappa^2 - 1)\mu, \\ \tilde{\mathfrak{P}}_{33} &= 4\beta^2\kappa^2, \\ \tilde{\mathfrak{P}}_{44} &= 4\kappa^2\gamma_1^2 - 2(\kappa^2 - 1)\beta(\gamma_2^2 + \gamma_3^2) - 4(\kappa^2 - 1)(X_1\gamma_2 + X_2\gamma_3) - 1. \end{split}$$

Since $\beta \neq 0$, the expression of $\tilde{\mathfrak{P}}_{33}$ shows that $\kappa = 0$. It is now easy to check that

$$\gamma_3 \tilde{\mathfrak{P}}_{11} - \gamma_2 \tilde{\mathfrak{P}}_{12} - \gamma_3 \tilde{\mathfrak{P}}_{22} = -\beta^2 \gamma_3^3, \qquad \gamma_2 \tilde{\mathfrak{P}}_{11} + \gamma_3 \tilde{\mathfrak{P}}_{12} - \gamma_2 \tilde{\mathfrak{P}}_{22} = \beta^2 \gamma_2^3.$$

Therefore, $\gamma_2 = \gamma_3 = 0$ and so $\tilde{\mathfrak{P}}_{44} = -1$, which shows that there are no left-invariant Ricci solitons in this case.

4.4 Left-invariant Ricci solitons on pp-wave Lie groups

Based on the analysis carried out in the previous sections, left-invariant Ricci solitons on pp-wave Lie groups split naturally into two distinct possibilities depending on whether they are plane waves or not. The case of pp-wave Lie groups which are not plane waves can be summarized as follows

Theorem 4.9. A four-dimensional Lorentzian pp-wave Lie group that is not a plane wave is a non-trivial left-invariant Ricci soliton if and only it is homothetic to the Lie group $G = \mathbb{R}^3 \rtimes \mathbb{R}$ endowed with a left-invariant metric given by the two-parameter family of Lie algebras

$$[u_1, u_4] = \gamma_1 u_1 + \varepsilon u_2, \quad [u_2, u_4] = -\gamma_1 u_2,$$

where $\gamma_1 \neq 0$, $\varepsilon = \pm 1$, and $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis such that $\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1$.

Proof. The Lorentzian Lie groups above are extensions of unimodular Lorentzian Lie groups and have been discussed in Section 4.1.1. Since the sectional curvature is independent of the structure constant γ_3 , we set $\gamma_3 = 0$ in Equation (4.1) and work at the homothetic level (see [91,96]). Now, the non-zero polynomials $\tilde{\mathfrak{P}}_{ij} = \frac{1}{2}\mathfrak{P}_{ij}$ reduce to

$$\tilde{\mathfrak{P}}_{11} = 2\varepsilon(X_4 - \gamma_1), \qquad \tilde{\mathfrak{P}}_{12} = \tilde{\mathfrak{P}}_{33} = \tilde{\mathfrak{P}}_{44} = -\mu,$$
$$\tilde{\mathfrak{P}}_{14} = X_2\gamma_1 - \varepsilon X_1, \qquad \tilde{\mathfrak{P}}_{24} = -X_1\gamma_1,$$

so we have a left-invariant steady Ricci soliton with left-invariant soliton vector field $X = X_3 u_3 + \gamma_1 u_4$, for any $\gamma_1 \neq 0$.

Remark 4.10. Globke and Leistner proved in [78] that four-dimensional Ricci-flat homogeneous pp-waves are plane waves. The examples in Theorem 4.9 show that the result above cannot be extended to steady Ricci soliton pp-waves. Moreover, the pp-wave Lie groups in Theorem 4.9 are conformal C-spaces, but not conformally Einstein (see [26, 79] for more information).

Theorem 4.11. A four-dimensional Lorentzian plane wave Lie group is a non-trivial left-invariant Ricci soliton if and only it is homothetic to one of the following Lie groups.

(i) $G = H^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[u_1, u_3] = u_2, \quad [u_1, u_4] = \gamma_3 u_3,$$

where $\gamma_3 \neq 0$ and $\{u_1, u_2, u_3, u_4\}$ is pseudo-orthonormal such that

$$\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1.$$

(ii) $G = E(1,1) \rtimes \mathbb{R}$ with Lie algebra given by

 $[u_1, u_2] = u_1, \quad [u_2, u_3] = u_3, \quad [u_2, u_4] = \gamma_3 u_3,$

and $\{u_1, u_2, u_3, u_4\}$ is pseudo-orthonormal with

$$\langle u_1, u_2 \rangle = \langle u_3, u_3 \rangle = \langle u_4, u_4 \rangle = 1.$$

(iii) $G = \mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

 $[u_1, u_4] = \gamma_1 u_1 + \gamma_2 u_2, \quad [u_2, u_4] = \gamma_4 u_1 + \gamma_5 u_2, \quad [u_3, u_4] = u_3,$

where $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis with

$$\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle = 1$$

(iv) $G = H^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$\begin{split} & [u_1, u_2] = \lambda_1 u_3, & [u_1, u_4] = \gamma_1 u_1 + \gamma_2 u_2, \\ & [u_2, u_4] = \gamma_4 u_1 + \gamma_5 u_2, & [u_3, u_4] = (\gamma_1 + \gamma_5) u_3, \end{split}$$

where $\gamma_1 + \gamma_5 \neq 0$, and $\{u_1, u_2, u_3, u_4\}$ is a pseudo-orthonormal basis with

$$\langle u_1, u_1 \rangle = \langle u_2, u_2 \rangle = \langle u_3, u_4 \rangle = 1$$

Proof. The Lie groups in Assertions (*i*) and (*ii*) are Lorentzian extensions of unimodular Lorentzian Lie groups. Assertion (*i*) was considered in Section 4.1.3 and a straightforward calculation shows that the curvature tensor does not involve the structure constants ε and γ_2 , so we can take $\varepsilon = -1$ and $\gamma_2 = 0$ in Equation (4.2) (see [91,96]). Now, the only non-zero component of the Ricci tensor is

$$\rho_{11} = -\frac{1}{2}(\gamma_3^2 - \gamma_6^2)$$

and the non-zero polynomials \mathfrak{P}_{ij} are

$$\mathfrak{P}_{11} = -\frac{1}{2} \{ \gamma_3^2 - \gamma_6^2 - 4X_3 \}, \qquad \mathfrak{P}_{12} = \mathfrak{P}_{33} = \mathfrak{P}_{44} = -\mu,$$

$$\mathfrak{P}_{13} = X_4(\gamma_3 + \gamma_6) - X_1, \qquad \mathfrak{P}_{14} = -X_3\gamma_6, \qquad \mathfrak{P}_{34} = -X_1\gamma_3$$

Now the metric is a left-invariant steady Ricci soliton if and only if it is Ricci-flat ($\gamma_3^2 = \gamma_6^2$) or, otherwise, $\gamma_6 = 0$ and $\gamma_3 \neq 0$. In this latter case, the left-invariant soliton vector field is given by

$$X = X_2 u_2 + \frac{1}{4} \gamma_3^2 u_3.$$

Assertion (*ii*) was treated in Section 4.1.4 and since the curvature tensor does not depend on the structure constant γ_2 , we can eliminate it in Equation (4.3) remaining in the same homothetic class, according to the work [96] (see also [91]). The non-zero polynomials \mathfrak{P}_{ij} now reduce to

$$\mathfrak{P}_{12} = X_2 - \mu, \qquad \mathfrak{P}_{22} = -\frac{1}{2}\gamma_3^2 - 2(X_1 + 1), \qquad \mathfrak{P}_{23} = X_4\gamma_3 + X_3,$$

$$\mathfrak{P}_{33} = -2X_2 - \mu, \qquad \mathfrak{P}_{34} = -X_2\gamma_3, \qquad \qquad \mathfrak{P}_{44} = -\mu,$$

and we obtain a left-invariant steady Ricci soliton with left-invariant soliton vector field

$$X = -(\frac{1}{4}\gamma_3^2 + 1)u_1 - X_4\gamma_3 u_3 + X_4 u_4.$$

The plane wave Lie groups in Assertion (*iii*) are Lorentzian extensions of unimodular degenerate Lie groups and correspond to those studied in Section 4.3.1. First of all, observe that proceeding as in the previous cases the constants γ_3 and γ_6 can be removed from Equation (4.4) and we would still be working in the same homothetic class. A straightforward calculation shows that the Ricci tensor vanishes if and only if

$$\rho_{44} = -\gamma_1^2 - \frac{1}{2}(\gamma_2 + \gamma_4)^2 - \gamma_5^2 + (\gamma_1 + \gamma_5)\gamma_9 = 0.$$

and the non-zero polynomials $ilde{\mathfrak{P}}_{ij} = rac{1}{2} \mathfrak{P}_{ij}$ are given by

$$\begin{aligned} & \mathfrak{P}_{11} = 2X_4\gamma_1 - \mu, & \mathfrak{P}_{14} = -X_1\gamma_1 - X_2\gamma_4, & \mathfrak{P}_{12} = X_4(\gamma_2 + \gamma_4), \\ & \tilde{\mathfrak{P}}_{34} = X_4\gamma_9 - \mu, & \tilde{\mathfrak{P}}_{22} = 2X_4\gamma_5 - \mu, & \tilde{\mathfrak{P}}_{24} = -X_1\gamma_2 - X_2\gamma_5, \\ & \tilde{\mathfrak{P}}_{44} = \rho_{44} - 2X_3\gamma_9 \,. \end{aligned}$$

We consider the cases $\gamma_9 \neq 0$ and $\gamma_9 = 0$ separately. Assuming $\gamma_9 \neq 0$, we can take $\gamma_9 = 1$ without losing generality. If $\gamma_2 = -\gamma_4$ and $\gamma_1 = \gamma_5 = \frac{1}{2}$, then

$$X = \frac{1}{2}\rho_{44}u_3 + \mu u_4$$

defines a locally conformally flat expanding, steady or shrinking left-invariant Ricci soliton. Otherwise, if $\gamma_2 \neq -\gamma_4$, or $\gamma_1 \neq \frac{1}{2}$, or $\gamma_5 \neq \frac{1}{2}$, then

$$X = \frac{1}{2}\rho_{44}u_3$$

determines a steady left-invariant Ricci soliton. Finally, if $\gamma_9 = 0$, then G is a left-invariant Ricci soliton if and only if it is Ricci-flat.

The plane wave Lie groups in Assertion (*iv*) are Lorentzian extensions of unimodular degenerate Lie groups and correspond to those in Section 4.3.2. We proceed as in the previous case and eliminate the structure constants γ_3 and γ_6 , so that the Ricci tensor vanishes if and only if

$$\rho_{44} = \frac{1}{2} \left\{ \lambda_1^2 + 4\gamma_1 \gamma_5 - (\gamma_2 + \gamma_4)^2 \right\} = 0,$$

and the non-zero polynomials $\tilde{\mathfrak{P}}_{ij} = \frac{1}{2} \mathfrak{P}_{ij}$ are given by

$$\begin{split} \tilde{\mathfrak{P}}_{11} &= 2X_4\gamma_1 - \mu, \\ \tilde{\mathfrak{P}}_{14} &= -X_1\gamma_1 + X_2(\lambda_1 - \gamma_4), \\ \tilde{\mathfrak{P}}_{24} &= -X_1(\lambda_1 + \gamma_2) - X_2\gamma_5, \\ \tilde{\mathfrak{P}}_{44} &= \rho_{44} - 2X_3(\gamma_1 + \gamma_5). \end{split}$$

$$\begin{split} \tilde{\mathfrak{P}}_{12} &= X_4(\gamma_2 + \gamma_4), \\ \tilde{\mathfrak{P}}_{22} &= 2X_4\gamma_5 - \mu, \\ \tilde{\mathfrak{P}}_{34} &= X_4(\gamma_1 + \gamma_5) - \mu, \\ \tilde{\mathfrak{P}}_{44} &= \rho_{44} - 2X_3(\gamma_1 + \gamma_5). \end{split}$$

If $\gamma_1 + \gamma_5 = 0$, left-invariant Ricci solitons only exist in the Ricci-flat case. Assuming that $\gamma_1 + \gamma_5 \neq 0$, there are two distinct possibilities.

If $\gamma_2 + \gamma_4 = 0$ and $\gamma_1 - \gamma_5 = 0$, then

$$X = \frac{1}{2(\gamma_1 + \gamma_5)}\rho_{44}u_3 + \frac{1}{\gamma_1 + \gamma_5}\mu u_4$$

determines a locally conformally flat expanding, steady or shrinking left-invariant Ricci soliton. Otherwise, if $\gamma_2 + \gamma_4 \neq 0$ or $\gamma_1 - \gamma_5 \neq 0$, then

$$X = \frac{1}{2(\gamma_1 + \gamma_5)} \rho_{44} u_3$$

defines a steady left-invariant Ricci soliton.

Remark 4.12. The plane wave Lie groups in Theorem 4.11 have vanishing Cotton tensor, and thus they are conformally Einstein [26]. The plane wave Lie groups corresponding to Assertion *(iii)* and Assertion *(iv)* which admit expanding, steady and shrinking left-invariant Ricci solitons are locally conformally flat.

Chapter 5 Algebraic Ricci and Bach solitons on four-dimensional Riemannian Lie groups

In this chapter we will give a complete description of the four-dimensional Riemannian algebraic Bach and Ricci solitons. In an endeavour to simplify the calculations, we will first work with a generic (0, 2)-tensor field T and study the conditions it must satisfy in order to determine an algebraic soliton for its associated flow. Imposing these conditions on specific tensor fields significantly eases the problem of determining all the algebraic solitons associated to the corresponding geometric flows. The results in this chapter are partially contained in the work [71].

5.1 Algebraic *T*-solitons

Let T be a left-invariant symmetric (0, 2)-tensor field on a Lie group $(G, \langle \cdot, \cdot \rangle)$ that is endowed with a left-invariant Riemannian metric and denote by \widehat{T} its associated (1, 1)-tensor field. Throughout this chapter, we will assume that T is divergence-free and isometrically invariant. $(G, \langle \cdot, \cdot \rangle)$ is said to be an *algebraic T-soliton* if

$$\widehat{T} = \mu \operatorname{Id} + \mathfrak{D}$$

for some derivation \mathfrak{D} of the corresponding Lie algebra \mathfrak{g} and some $\mu \in \mathbb{R}$, and it will be expanding, steady or shrinking if $\mu < 0$, $\mu = 0$, or $\mu > 0$, respectively.

If $\{e_1, e_2, e_3, e_4\}$ is a basis of \mathfrak{g} , we will use the notation $T_{ij} = T(e_i, e_j)$ and, since T is symmetric, when we need to write it in matrix form we will omit the entries below the diagonal and, for the sake of simplicity, write '*' instead.

The endomorphism $\mathfrak{D} = \widehat{T} - \mu \operatorname{Id}$ is a derivation of the corresponding Lie algebra \mathfrak{g} if it satisfies the condition

$$\mathfrak{D}[e_i, e_j] - [\mathfrak{D}e_i, e_j] - [e_i, \mathfrak{D}e_j] = 0, \quad i, j = 1, \dots, 4,$$

which, when expressed with respect to the basis $\{e_1, e_2, e_3, e_4\}$, is equivalent to

$$\mathfrak{P}_{ijk} = \mathfrak{D}_{\ell}{}^{k}c_{ij}{}^{\ell} - \mathfrak{D}_{i}{}^{\ell}c_{\ell j}{}^{k} - \mathfrak{D}_{j}{}^{\ell}c_{i\ell}{}^{k} = 0,$$

where $\mathfrak{D}_s^r = \widehat{T}_s^r - \mu \delta_s^r$ and the structure constants c_{ij}^{ℓ} are given by $[e_i, e_j] = c_{ij}^{\ell} e_{\ell}$. Note that $T_{rs} = T(e_r, e_s) = \sum_{\alpha=1}^4 \widehat{T}_{\alpha}^r \langle e_{\alpha}, e_s \rangle$ and therefore $\widehat{T}_s^r = T_{rs}$ if the basis $\{e_1, e_2, e_3, e_4\}$ is orthonormal.

We say that an algebraic *T*-soliton is *trivial* if $T = \kappa \langle \cdot, \cdot \rangle$ for some real constant κ (or equivalently, \hat{T} and \mathfrak{D} are multiples of the identity). Note that trivial algebraic Ricci solitons correspond to Einstein spaces, while trivial algebraic Bach solitons are precisely the Bach-flat manifolds.

Remark 5.1. A Riemannian manifold (M, g) is a *T*-soliton if there is a vector field X on M so that

$$\mathcal{L}_X g + T = \lambda g$$

for some $\lambda \in \mathbb{R}$, and it is expanding, steady or shrinking if $\lambda < 0$, $\lambda = 0$, or $\lambda > 0$, respectively. If the vector field X is a gradient of a real-valued function, $X = \frac{1}{2}\nabla f$, then the equation above becomes

$$\operatorname{Hes}(f) + T = \lambda g.$$

In such a case we say that (M, g, T, f) is a gradient *T*-soliton and refer to f as the potential function of the *T*-soliton. Given a geometric evolution equation

$$\frac{\partial}{\partial t}g_t = T_{g_t}$$

associated to an isometrically invariant symmetric (0, 2)-tensor field T, a solution that evolves by scaling and diffeomorphisms is said to be a *self-similar solution*. These solutions are of the form

$$g_t = \sigma(t)\psi_t^*g,$$

where ψ_t is a one-parameter family of diffeomorphisms of M. Any self-similar solution gives rise to a T-soliton and the converse is true if the tensor field T is homogeneous of degree d, i.e., $\overline{T} = \kappa^d T$ for any homothetic transformation $\overline{g} = \kappa g$ (see [133]). Since the Ricci and Bach tensors are homothetically invariant, self-similar solutions are equivalent to solitons for both the Ricci and Bach flows.

If a simply connected Riemannian Lie group $(G, \langle \cdot, \cdot \rangle)$ is an algebraic *T*-soliton, then it is a *T*-soliton (see [133]). Indeed, let $\{\varphi_t : G \to G\}$ be the one-parameter group of automorphisms of *G* determined by

$$d(\varphi_t)_e = \operatorname{Exp}(\frac{t}{2}\mathfrak{D}),$$

where $\mathfrak{D} = \widehat{T} - \mu \operatorname{Id}$ is the derivation of the Lie algebra determining the algebraic *T*-soliton. Define a vector field *X* on *G* as the infinitesimal generator of $\{\varphi_t\}$, i.e., $X(p) = \frac{d}{dt}\Big|_{t=0} \varphi_t(p)$ for any $p \in G$. Then

$$\begin{aligned} (\mathcal{L}_X \langle \cdot, \cdot \rangle)(e_i, e_j) &= \frac{d}{dt} (\varphi_t^* \langle \cdot, \cdot \rangle)(e_i, e_j) = \frac{1}{2} \left\{ \langle \mathfrak{D} e_i, e_j \rangle + \langle e_i, \mathfrak{D} e_j \rangle \right\} \\ &= \frac{1}{2} \left\{ \langle \widehat{T} e_i, e_j \rangle + \langle e_i, \widehat{T} e_j \rangle \right\} - \mu \langle e_i, e_j \rangle \\ &= (T - \mu \langle \cdot, \cdot \rangle)(e_i, e_j), \end{aligned}$$

from where it follows that $\mathcal{L}_X \langle \cdot, \cdot \rangle - T = -\mu \langle \cdot, \cdot \rangle$. Replacing X by -X one gets the T-soliton equation above with $\lambda = \mu$.

Remark 5.2. If a Riemannian manifold (M, g) admits two distinct *T*-solitons, i.e., vector fields X_i so that $\mathcal{L}_{X_i}g+T = \lambda_i g$, i = 1, 2, then one has that $\xi = X_1 - X_2$ satisfies $\mathcal{L}_{\xi}g = (\lambda_1 - \lambda_2)g$ and so ξ is a homothetic vector field. What is more, if we assume that (M, g) is homogeneous, then either the manifold is flat or ξ is a Killing vector field. This shows that, if they exist, *T*-solitons are unique (up to Killing vector fields) in the homogeneous category.

Remark 5.3. Petersen and Wylie showed in [122] that if (M, g) is a homogeneous manifold and \widetilde{T} is a divergence-free, symmetric, and isometrically invariant tensor field of type (0, 2), then any non-constant function satisfying $\text{Hes}(f) = \widetilde{T}$ induces a splitting of the manifold as a product $N \times \mathbb{R}^k$, and f is a function on the Euclidean factor.

For a given T-flow, considering the tensor field $\tilde{T} = \lambda g - T$, it is a direct consequence of the previous result that any homogeneous gradient T-soliton (such that $\text{Hes}(f) + T = \lambda g$) splits as a product if the tensor field T is divergence-free. The result above is no longer true in the T-soliton is not a gradient, which is the case of the solitons constructed in this chapter.

5.1.1 Four-dimensional algebraic Ricci solitons

First of all note that the Ricci tensor is invariant by homotheties. Equivalently, the Ricci operator, $\operatorname{Ric}_i{}^k = \rho_{ij}g^{jk}$, is homogeneous of degree d = -1. Therefore, there is a one-to-one correspondence between self-similar solutions of the Ricci flow and Ricci solitons. Algebraic Ricci solitons correspond to self-similar solutions which are invariant up to homotheties and automorphisms of the group.

Four-dimensional homogeneous Einstein manifolds were described by Jensen, who showed in [88] that they are necessarily symmetric and, in the simply connected case, homothetic to a real space form (\mathbb{S}^4 , \mathbb{R}^4 or \mathbb{H}^4), to a complex space form ($\mathbb{C}P^2$ or $\mathbb{C}H^2$), or to a product $\mathbb{S}^2 \times \mathbb{S}^2$ or $\mathbb{H}^2 \times \mathbb{H}^2$.

Recall that, even though any four-dimensional homogeneous expanding Ricci soliton is homothetic to an algebraic Ricci soliton, algebraic Ricci solitons may be shrinking, in which case they are rigid gradient Ricci solitons (see Theorem 5.4-(v)). Four-dimensional algebraic Ricci solitons were described by Lauret in [98].

One of our aims in this chapter is to understand the Riemannian geometry of algebraic Ricci solitons. To do so, we work modulo homotheties also considering non-isomorphic homotheties in order to give a shorter description. Lauret's classification follows as a special situation of the analysis carried out in this chapter.

Theorem 5.4. A four-dimensional simply connected Riemannian Lie group is an algebraic Ricci soliton if and only if it is Einstein or either of the following conditions holds.

- (a) The Riemannian Lie group is not locally symmetric and it is homothetic to one of the following:
 - (i) The semi-direct extension $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_2, e_4] = e_2 + e_3, \quad [e_3, e_4] = -e_2 - e_3.$$

(ii) The semi-direct extension $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_4] = \frac{\sqrt{2}}{2}(e_2 + e_3), \quad [e_2, e_4] = -\frac{\sqrt{2}}{2}e_1 + e_2, \quad [e_3, e_4] = -\frac{\sqrt{2}}{2}e_1 - e_3.$$

(iii) The semi-direct extension $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_4] = e_1, \quad [e_2, e_4] = fe_2, \quad [e_3, e_4] = pe_3,$$

where

$$(f,p) \in \{(f,p) \in \mathbb{R}^2: -1 \le f \le p \le 1\} \setminus \{(-1,p) \in \mathbb{R}^2: -1 \le p < 0\}$$

- and $(f, p) \notin \{(0, 0), (0, 1), (1, 1)\}.$
- (iv) The semi-direct extension $H^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = ae_1, \quad [e_2, e_4] = de_2, \quad [e_3, e_4] = (a+d)e_3,$$

where $a \in \left[-\frac{\sqrt{3}}{2}, \frac{1}{2}\right)$. For a fixed a, the parameter d is given by the only positive solution of $4(a^2 + d^2 + ad) - 3 = 0$.

(b) The Riemannian Lie group is locally symmetric and it is homothetic to one of the following:

(v) The product Lie group $SU(2) \times \mathbb{R}$ with the product metric determined by

$$[e_1, e_2] = e_3, \quad [e_1, e_3] = -e_2, \quad [e_2, e_3] = e_1,$$

which is homothetic to $\mathbb{S}^3 \times \mathbb{R}$.

(vi) The semi-direct extension $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

 $[e_3, e_4] = e_3,$

which is homothetic to the product $\mathbb{H}^2 \times \mathbb{R}^2$.

(vii) The semi-direct extension $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

 $[e_1, e_4] = e_1, \quad [e_2, e_4] = e_2,$

which is homothetic to the product $\mathbb{H}^3 \times \mathbb{R}$.

In all the cases above $\{e_1, e_2, e_3, e_4\}$ denotes an orthonormal basis.

Remark 5.5. It follows from the result above that a simply connected four-dimensional homogeneous Ricci soliton is either a rigid gradient Ricci soliton or homothetic to a metric (i)-(iv) in Theorem 5.4.

Observe that the assumption of simply connectedness is relevant. Indeed, the metric in (*i*) above is isomorphically homothetic to the left-invariant product metric on the product $H^3 \times \mathbb{R}$. It is clear that $H^3 \times \mathbb{R}$ admits discrete subgroups Γ so that the quotient nilmanifold $H^3 \times \mathbb{R}/\Gamma$ is a compact (but not simply connected) nilmanifold. Although the left-invariant metric descends to the quotient (and thus $H^3 \times \mathbb{R}$ and the corresponding nilmanifold are locally isometric), the Ricci soliton cannot pass to the quotient, since compact expanding Ricci solitons are necessarily Einstein.

Remark 5.6. All the algebraic Ricci solitons in the previous theorem are also algebraic Bach solitons. Moreover, while the algebraic Ricci solitons are expanding in all the cases but (v), the algebraic Bach solitons may be shrinking, expanding or steady (the latter only in the Bach-flat situation). We analyse their structure case by case in what follows.

(i) A straightforward calculation shows that the Ricci and Bach operators in the orthonormal basis $\{e_k\}$ take the forms

$$\operatorname{Ric} = \begin{pmatrix} 0 & 0 & 0 & 0 \\ 0 & 0 & 2 & 0 \\ 0 & 2 & 0 & 0 \\ 0 & 0 & 0 & -2 \end{pmatrix}, \quad \widehat{\mathfrak{B}} = -\frac{8}{3} \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 4 & 0 \\ 0 & 4 & 1 & 0 \\ 0 & 0 & 0 & -3 \end{pmatrix}$$

In this situation,

 $\mathfrak{D} = \operatorname{Ric} + 6\operatorname{Id}$ and $\mathfrak{D} = \widehat{\mathfrak{B}} - \frac{88}{3}\operatorname{Id}$

are derivations of the Lie algebra, from where it follows that $(G, \langle \cdot, \cdot \rangle)$ is an expanding algebraic Ricci soliton ($\mu = -6$) and a shrinking algebraic Bach soliton ($\mu = \frac{88}{3}$).

The metric in Theorem 5.4-(*i*) is isometric to the product metric on $H^3 \times \mathbb{R}$, which is a nongradient Ricci soliton. Moreover, it follows from the work in [81] that the corresponding Bach soliton is not a gradient either.

(ii) The Ricci and Bach operators of the left-invariant metric in Theorem 5.4-(ii) are given by

$$\operatorname{Ric} = \frac{1}{\sqrt{2}} \begin{pmatrix} 0 & -1 & 1 & 0 \\ -1 & 0 & 0 & 0 \\ 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & -2\sqrt{2} \end{pmatrix}, \quad \widehat{\mathfrak{B}} = \frac{7}{6} \begin{pmatrix} -1 & \sqrt{2} & -\sqrt{2} & 0 \\ \sqrt{2} & -1 & 0 & 0 \\ -\sqrt{2} & 0 & -1 & 0 \\ 0 & 0 & 0 & 3 \end{pmatrix}.$$

Therefore,

$$\mathfrak{D} = \operatorname{Ric} + 3\operatorname{Id}$$
 and $\mathfrak{D} = \widehat{\mathfrak{B}} - \frac{35}{6}\operatorname{Id}$

are derivations of the Lie algebra, which shows that $(G, \langle \cdot, \cdot \rangle)$ is an expanding algebraic Ricci soliton and a shrinking algebraic Bach soliton. Although the Ricci operator has a zero eigenvalue with associated eigenvector given by

$$e_2 + e_3$$
,

the corresponding eigenspace is not parallel and thus the associated Ricci and Bach solitons are not gradient solitons.

(iii) The Ricci operators of the metrics in Theorem 5.4-(iii) are given by

$$\operatorname{Ric} = -\operatorname{diag}\left[f + p + 1, f(f + p + 1), p(f + p + 1), f^2 + p^2 + 1\right],$$

from where it follows that $\mathfrak{D} = \text{Ric} + (f^2 + p^2 + 1)$ Id are derivations of the Lie algebras determining expanding algebraic Ricci solitons.

Moreover, the Bach operators are given by $\widehat{\mathfrak{B}} = \frac{1}{6} \operatorname{diag}[\mathfrak{B}_{11}, \mathfrak{B}_{22}, \mathfrak{B}_{33}, 6\mu]$, where

$$\begin{split} \mathfrak{B}_{11} &= -\left(f^2 - 2(p+1)f + (p-1)^2\right)\left(f^2 + (p-1)f + (p-1)p - 1\right),\\ \mathfrak{B}_{22} &= -\left(f^2 - 2(p+1)f + (p-1)^2\right)\left(f^2 + (p+1)f - (p+1)p - 1\right),\\ \mathfrak{B}_{33} &= -\left(f^2 - 2(p+1)f + (p-1)^2\right)\left(f^2 - (p-1)f - (p+1)p + 1\right),\\ \mu &= \frac{1}{6}\left(f^2 - 2(p+1)f + (p-1)^2\right)\left(f^2 - (p+1)f + (p-1)p + 1\right). \end{split}$$

Thus, $\mathfrak{D} = \widehat{B} - \mu \text{Id}$ are derivations of the corresponding Lie algebras and hence they are algebraic Bach solitons, which may be expanding, steady or shrinking depending on the values of the parameters (f, p). They are steady algebraic Bach solitons if and only if $f = (\sqrt{p} - 1)^2$, with $\frac{1}{4} \le p < 1$, in which case the corresponding metrics are Bach-flat (see Case (*ii*) in Remark 5.7).

(*iv*) The Ricci and Bach operators of the left-invariant metrics in Theorem 5.4-(*iv*) are given by

Ric = diag
$$\left[2(d^2-1), -2d(a+d) - \frac{1}{2}, -2ad - 1, -\frac{3}{2}\right]$$
,

$$\widehat{\mathfrak{B}} = \frac{1}{24} \operatorname{diag}[\mathfrak{B}_{11}, -\mathfrak{B}_{11}, -(4ad-1)(8ad+1), (4ad-1)(8ad+1)],$$

$$\mathfrak{B} = -\frac{32(d^4}{d^2} - \frac{d^3}{d^3} - \frac{20ad}{d^3} - \frac{3}{2} \operatorname{In this situation}$$

where $\mathfrak{B}_{11} = -32(d^4 - d^2 - ad^3) - 20ad - 3$. In this situation,

$$\mathfrak{D} = \operatorname{Ric} + \frac{3}{2}\operatorname{Id}, \quad \mathfrak{D} = \mathfrak{B} - \frac{1}{24}(4ad - 1)(8ad + 1)\operatorname{Id}$$

are derivations of the Lie algebra, from where it follows that these metrics are expanding algebraic Ricci solitons and also algebraic Bach solitons, which may be expanding, steady or shrinking depending on the values of the parameters (a, d). They are steady algebraic Bach solitons if and only if

$$(4ad - 1)(8ad + 1) = 0,$$

in which case they are Bach-flat and correspond to Case (i) in Remark 5.7.

(v) The metric in Theorem 5.4-(v) is homothetic to the product metric on $\mathbb{S}^3 \times \mathbb{R}$. Therefore, it is locally conformally flat and thus Bach-flat, so it is a steady algebraic Bach soliton. Moreover, its associated Ricci operator is diagonal, Ric = $\frac{1}{2}$ diag[1, 1, 1, 0], and $\mathfrak{D} =$ Ric $-\frac{1}{2}$ Id, which shows that it is a shrinking Ricci soliton that is a gradient.

- (vi) The product $\mathbb{H}^2 \times \mathbb{R}^2$ is an expanding rigid gradient Ricci soliton which is also algebraic since the Ricci operator is diagonal Ric = diag[0, 0, -1, -1] and $\mathfrak{D} = \text{Ric}+\text{Id}$. Moreover, it is a shrinking algebraic Bach soliton determined by $\mathfrak{D} = \widehat{\mathfrak{B}} - \frac{1}{6}$ Id, where the Bach operator is given by $\widehat{\mathfrak{B}} = \frac{1}{6}$ diag[-1, -1, 1, 1]. The Bach soliton associated to the algebraic Bach soliton coincides with the gradient Bach soliton given by Ho in [85] up to some Killing vector field.
- (vii) The metric in Theorem 5.4-(vii) is homothetic to the product metric on $\mathbb{H}^3 \times \mathbb{R}$, thus being locally conformally flat. It is a steady algebraic Bach soliton and an expanding rigid gradient Ricci soliton with $\operatorname{Ric} = -2 \operatorname{diag}[1, 1, 0, 1]$ and $\mathfrak{D} = \operatorname{Ric} + 2\operatorname{Id}$.

The proof of Theorem 5.4 is obtained as the result of the general study of the algebraic Ricci solitons on four-dimensional Lie groups.

5.1.2 Four-dimensional algebraic Bach solitons

The Bach tensor \mathfrak{B} is conformally invariant in dimension four and thus it is also homothetically invariant. Besides, the Bach operator $\widehat{\mathfrak{B}}$ is homogeneous of degree d = -1 and thus there is a one-to-one correspondence between self-similar solutions to the Bach flow and Bach solitons.

Remark 5.7. Locally conformally flat manifolds are trivially Bach-flat. In the homogeneous situation Takagi showed that they are also symmetric [128], and thus they either correspond to real space forms, or are homothetic to a product $\mathbb{S}^3 \times \mathbb{R}$, $\mathbb{H}^3 \times \mathbb{R}$ or $\mathbb{S}^2 \times \mathbb{H}^2$.

In contrast with the Einstein and the locally conformally flat situations, there are non-symmetric Bach-flat homogeneous four-manifolds, which have already been described in [39] as Riemannian Lie groups that are homothetic to those given by

(i) The semi-direct product $H^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_2] = e_3, \qquad [e_1, e_4] = -\frac{1}{4}\sqrt{7 - 3\sqrt{5}} e_1,$$
$$[e_2, e_4] = \frac{1}{4}\sqrt{7 + 3\sqrt{5}} e_2, \qquad [e_3, e_4] = \frac{\sqrt{5}}{2\sqrt{2}} e_3.$$

(ii) The semi-direct product $\mathbb{R}^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_4] = e_1, \quad [e_2, e_4] = (\sqrt{p} - 1)^2 e_2, \quad [e_3, e_4] = p e_3, \quad \frac{1}{4} \le p < 1.$$

(*iii*) The semi-direct product $E(1,1) \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_3] = (2 - \sqrt{3}) e_2, \quad [e_1, e_4] = \sqrt{6 - 3\sqrt{3}} e_1,$$
$$[e_2, e_3] = e_1, \qquad [e_2, e_4] = \sqrt{6 - 3\sqrt{3}} e_2.$$

(iv) The semi-direct product $H^3 \rtimes \mathbb{R}$ with Lie algebra given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = e_1, \quad [e_2, e_4] = e_2, \quad [e_3, e_4] = 2e_3.$$
Note that, while the left-invariant metrics (i) and (ii) above are algebraic Ricci solitons, the metrics corresponding to (iii)-(iv) are not isometric to any Ricci soliton. Moreover, the left-invariant metric determined in (iv) is anti-self-dual, and its Weyl conformal tensor is half-harmonic (see Theorem 6.1 in this thesis).

Proceeding as in Remark 5.3, Griffin [81] showed that homogeneous gradient Bach solitons either are Bach-flat or split as a product $N \times \mathbb{R}^k$, where the potential function depends only on the Euclidean factor, and (N, g_N) is a homogeneous manifold. Moreover, homogeneous steady gradient Bach solitons are necessarily Bach-flat. Gradient Bach solitons on products $N \times \mathbb{R}$ which are not Bach-flat reduce to the case $\mathbb{S}^3 \times \mathbb{R}$, where the metric on \mathbb{S}^3 is not the round metric, but a Berger one [81].

Moving from the gradient to the algebraic situation, the next result shows that algebraic Bach solitons are steady if and only if they are Bach-flat and, moreover, all but two of them also are algebraic Ricci solitons.

Theorem 5.8. A four-dimensional simply connected Riemannian Lie group is an algebraic Bach soliton if and only if it is Bach-flat, an algebraic Ricci soliton or homothetic to one of the following Lie groups.

(i) The product Lie group $SU(2) \times \mathbb{R}$ with the product left-invariant metric determined by

$$[e_1, e_2] = 4e_3, \quad [e_1, e_3] = -4e_2, \quad [e_2, e_3] = e_1.$$

(ii) The semi-direct extension $H^3 \rtimes \mathbb{R}$ with the left-invariant metric determined by

 $[e_1, e_2] = e_3, \quad [e_1, e_4] = ae_1, \quad [e_2, e_4] = \frac{1}{a}e_2, \quad [e_3, e_4] = \frac{a^2+1}{a}e_3,$

for $a \in (0, 1)$.

Here $\{e_1, e_2, e_3, e_4\}$ *denotes an orthonormal basis of the corresponding Lie algebra.*

Remark 5.9. The left-invariant metric in Theorem 5.8-(*i*) is the product metric of the Berger sphere determined by the left-invariant metric on SU(2) given by

$$[e_1, e_2] = 4e_3, \quad [e_1, e_3] = -4e_2, \quad [e_2, e_3] = e_1,$$

and the real line. Moreover, it follows from [81, 84] that the induced Bach soliton is a gradient soliton. The Berger sphere above is the only non-symmetric homogeneous three-dimensional manifold which is critical for the curvature quadratic functional

$$\mathcal{F}_{-1/3}(g) = \int_M \left\{ \|\rho\|^2 - \frac{1}{3}\tau^2 \right\} dvol_g.$$

The associated Bach operator is diagonal of the form $\widehat{\mathfrak{B}} = \frac{1}{2} \operatorname{diag}[1, 1, 1, -3]$ and the algebraic Bach soliton is shrinking, with $\mu = \frac{1}{2}$.

The left-invariant metrics in (ii) have diagonal Bach operator of the form

$$\widehat{\mathfrak{B}} = \frac{1}{2a^2} \left[-3(a^4 - 1), 3(a^4 - 1), (a^2 - 1)^2, -(a^2 - 1)^2 \right]$$

and $\mathfrak{D} = \widehat{\mathfrak{B}} + \frac{(a^2-1)^2}{2a^2}$ Id is a derivation determining an expanding algebraic Bach soliton. Moreover, the corresponding Ricci operator is diagonal,

$$\operatorname{Ric} = -\operatorname{diag}\left[2a^2 + \frac{5}{2}, \frac{5a^2 + 4}{2a^2}, \frac{4a^4 + 7a^2 + 4}{2a^2}, \frac{2(a^4 + a^2 + 1)}{a^2}\right],$$

and therefore its eigenvalues are always non-zero. This shows that the metrics cannot split as products $N \times \mathbb{R}^k$ and thus the corresponding solitons are not gradient.

Remark 5.10. Gradient Bach solitons on products $N^2 \times \mathbb{R}^2$ were considered by Ho in [85], where the existence of non-trivial gradient Bach solitons on $\mathbb{S}^2 \times \mathbb{R}^2$ and $\mathbb{H}^2 \times \mathbb{R}^2$ was shown. While the soliton on $\mathbb{S}^2 \times \mathbb{R}^2$ cannot be algebraic, the gradient Bach soliton on $\mathbb{H}^2 \times \mathbb{R}^2$ corresponds to the one in Theorem 5.4-(*vi*).

Remark 5.11. The algebraic Ricci solitons in Theorem 5.4 are critical metrics for some curvature quadratic functional

$$\mathcal{F}_t: g \mapsto \mathcal{F}_t(g) = \int_M \{ \|\rho\|^2 + t\tau^2 \} dvol_g$$

with zero energy, i.e., $\|\rho\|^2 + t\tau^2 = 0$ (see [23]). This is no longer true for algebraic Bach solitons corresponding to Cases (*i*) and (*ii*) in Theorem 5.8, which are not critical for any curvature quadratic functional.

Remark 5.12. The two-loop renormalization group flow – RG2 flow for short – is a perturbation of the Ricci flow $\partial_t g_t = -2\rho_t$, which is mathematically described by

$$\partial_t g_t = -2RG[\Upsilon]_t,$$

where $RG[\Upsilon]$ is the symmetric (0, 2)-tensor field given by

$$RG[\Upsilon] = \rho + \frac{\Upsilon}{4}\check{R},$$

where Υ denotes the coupling constant, and \check{R} is the symmetric (0,2)-tensor field given by

$$\check{R}_{ij} = R_{iabc} R_j^{\ abc}.$$

Let G be a Lie group with left-invariant metric $\langle \cdot, \cdot \rangle$ and let $(\mathfrak{g}, \langle \cdot, \cdot \rangle)$ denote the corresponding Lie algebra. An RG2 *algebraic soliton* corresponds to a derivation of the Lie algebra \mathfrak{g} given by

$$\mathfrak{D} = \widehat{RG}[\Upsilon] - \mu \operatorname{Id},$$

where $\widehat{RG}[\Upsilon]$ is the (1, 1)-tensor field metrically equivalent to $RG[\Upsilon]$ and $\mu \in \mathbb{R}$. RG2 algebraic solitons give rise to RG2 solitons as in the Ricci flow and the Bach flow cases (see [97, 133]).

The metric in Theorem 5.8-(i) is an algebraic soliton for the RG2 flow. Indeed,

$$\mathfrak{D} = \widehat{RG} \left[\frac{-4}{7} \right] - \frac{13}{28} \operatorname{Id}$$

is a derivation of the Lie algebra, where $RG[\frac{-4}{7}] = \rho - \frac{1}{7}\check{R}$.

The metrics in Theorem 5.8-(ii) are algebraic RG2 solitons as well. Indeed,

$$\mathfrak{D} = \widehat{R}\widehat{G}[\Upsilon] - \mu \operatorname{Id}$$

is a derivation of the Lie algebra for $\mu = -\frac{20(a^4+a^2+1)}{4a^4+15a^2+4}$ and $\Upsilon = \frac{8a^2}{4a^4+15a^2+4}$.

The proof of Theorem 5.8 is obtained as the result of the general study of the algebraic Bach solitons on four-dimensional Lie groups.

Remark 5.13. Four-dimensional homogeneous expanding Ricci solitons are homothetic to algebraic solitons [9]. It is an open question whether the analogous situation holds true for homogeneous Bach solitons.

5.2 Algebraic solitons on $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$

The left-invariant metrics on the product Lie groups $SL(2, \mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$ have already been described in Section 1.4.2. We recall that in this case there exists an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ of the Lie algebras $\mathfrak{sl}(2, \mathbb{R}) \times \mathbb{R}$ and $\mathfrak{su}(2) \times \mathbb{R}$ such that

$$[e_{1}, e_{2}] = \lambda_{3}e_{3}, \qquad [e_{1}, e_{3}] = -\lambda_{2}e_{2}, \qquad [e_{2}, e_{3}] = \lambda_{1}e_{1},$$

$$[e_{1}, e_{4}] = k_{3}\lambda_{2}e_{2} - k_{2}\lambda_{3}e_{3}, \qquad [e_{2}, e_{4}] = k_{1}\lambda_{3}e_{3} - k_{3}\lambda_{1}e_{1},$$

$$[e_{3}, e_{4}] = k_{2}\lambda_{1}e_{1} - k_{1}\lambda_{2}e_{2},$$
(5.1)

where $\lambda_1 \lambda_2 \lambda_3 \neq 0$. The associated Lie group corresponds to $SU(2) \times \mathbb{R}$ if λ_1 , λ_2 and λ_3 have the same sign, and to $\widetilde{SL}(2, \mathbb{R}) \times \mathbb{R}$ otherwise.

Remark 5.14. $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ is never locally symmetric while $SU(2) \times \mathbb{R}$ is locally symmetric if and only if $\lambda_1 = \lambda_2 = \lambda_3$, in which case it is homothetic to $\mathbb{S}^3 \times \mathbb{R}$.

5.2.1 Algebraic *T*-solitons on $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$

The existence of algebraic T-solitons on $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$ is a very restrictive condition which essentially reduces to the fact that \widehat{T} decomposes according to the product and the restriction of \widehat{T} to the semi-simple Lie algebra is a multiple of the identity.

Theorem 5.15. $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$ are non-trivial algebraic *T*-solitons with soliton constant μ if and only if

 $\widehat{T} = \operatorname{diag}[\mu, \mu, \mu, T_{44}], \quad T_{44} \neq \mu,$

and the left-invariant metrics correspond to the product metric, i.e.,

$$k_1 = k_2 = k_3 = 0$$

Proof. First of all, note that the vanishing of the divergence of T leads to

$$(\lambda_{2} - \lambda_{3})T_{23} - k_{3}\lambda_{2}T_{24} + k_{2}\lambda_{3}T_{34} = 0,$$

$$(\lambda_{1} - \lambda_{3})T_{13} - k_{3}\lambda_{1}T_{14} + k_{1}\lambda_{3}T_{34} = 0,$$

$$(\lambda_{1} - \lambda_{2})T_{12} - k_{2}\lambda_{1}T_{14} + k_{1}\lambda_{2}T_{24} = 0,$$

$$k_{3}(\lambda_{1} - \lambda_{2})T_{12} - k_{2}(\lambda_{1} - \lambda_{3})T_{13} + k_{1}(\lambda_{2} - \lambda_{3})T_{23} = 0.$$
(5.2)

To determine the conditions for $\mathfrak{D} = \hat{T} - \mu \operatorname{Id}$ to be a derivation we will need to solve a system of twenty four (up to duplicities) polynomial equations on the soliton constant μ , the structure constants (5.1) and the components T_{ij} , given by the vanishing of the polynomials

$$\begin{split} \mathfrak{P}_{211} &= -(\lambda_1 + \lambda_3)T_{13} + k_3\lambda_1T_{14}, \\ \mathfrak{P}_{212} &= -(\lambda_2 + \lambda_3)T_{23} + k_3\lambda_2T_{24}, \\ \mathfrak{P}_{213} &= \lambda_3(T_{11} + T_{22} - T_{33} - k_1T_{14} - k_2T_{24} - \mu), \\ \mathfrak{P}_{214} &= -\lambda_3T_{34}, \\ \mathfrak{P}_{311} &= (\lambda_1 + \lambda_2)T_{12} - k_2\lambda_1T_{14}, \\ \mathfrak{P}_{312} &= -\lambda_2(T_{11} - T_{22} + T_{33} - k_1T_{14} - k_3T_{34} - \mu), \\ \mathfrak{P}_{313} &= (\lambda_2 + \lambda_3)T_{23} - k_2\lambda_3T_{34}, \\ \mathfrak{P}_{314} &= \lambda_2T_{24}, \\ \mathfrak{P}_{321} &= -\lambda_1(T_{11} - T_{22} - T_{33} + k_2T_{24} + k_3T_{34} + \mu), \\ \mathfrak{P}_{322} &= -(\lambda_1 + \lambda_2)T_{12} + k_1\lambda_2T_{24}, \\ \mathfrak{P}_{323} &= -(\lambda_1 + \lambda_3)T_{13} + k_1\lambda_3T_{34}, \\ \mathfrak{P}_{324} &= -\lambda_1T_{14}, \\ \mathfrak{P}_{411} &= -k_3(\lambda_1 + \lambda_2)T_{12} + k_2(\lambda_1 + \lambda_3)T_{13}, \\ \mathfrak{P}_{412} &= k_3\lambda_2(T_{11} - T_{22} + T_{44} - \mu) - k_1\lambda_2T_{13} + k_2\lambda_3T_{23} - \lambda_2T_{34}, \\ \mathfrak{P}_{413} &= -k_2\lambda_3(T_{11} - T_{33} + T_{44} - \mu) + k_1\lambda_3T_{12} - k_3\lambda_2T_{23} + \lambda_3T_{24}, \\ \mathfrak{P}_{422} &= k_3(\lambda_1 + \lambda_2)T_{12} - k_1(\lambda_2 + \lambda_3)T_{23}, \\ \mathfrak{P}_{423} &= k_1\lambda_3(T_{22} - T_{33} + T_{44} - \mu) - k_2\lambda_3T_{12} + k_3\lambda_1T_{13} - \lambda_3T_{14}, \\ \mathfrak{P}_{424} &= k_3\lambda_1T_{14} - k_1\lambda_3T_{34}, \\ \mathfrak{P}_{433} &= -k_2\lambda_1(T_{11} - T_{33} - T_{44} + \mu) - k_2\lambda_1T_{12} + k_3\lambda_1T_{13} - \lambda_1T_{24}, \\ \mathfrak{P}_{433} &= -k_2(\lambda_1 + \lambda_3)T_{13} + k_1(\lambda_2 + \lambda_3)T_{23}, \\ \mathfrak{P}_{434} &= -k_2\lambda_1(T_{14} - K_{13}X_{24}, \\ \mathfrak{P}_{434} &= -k_2\lambda_1(T_{14} + k_1\lambda_2T_{24}. \\ \end{array}$$

We start by considering the polynomials

$$\begin{aligned} \mathfrak{P}_{324} &= -\lambda_1 T_{14}, \quad \mathfrak{P}_{314} = \lambda_2 T_{24}, \quad \mathfrak{P}_{214} = -\lambda_3 T_{34}, \\ \mathfrak{P}_{321} &= -\lambda_1 (T_{11} - T_{22} - T_{33} + k_2 T_{24} + k_3 T_{34} + \mu), \\ \mathfrak{P}_{312} &= -\lambda_2 (T_{11} - T_{22} + T_{33} - k_1 T_{14} - k_3 T_{34} - \mu), \\ \mathfrak{P}_{213} &= -\lambda_3 (T_{11} + T_{22} - T_{33} - k_1 T_{14} - k_2 T_{24} - \mu), \end{aligned}$$

from where it follows that

$$T_{14} = T_{24} = T_{34} = 0, \quad T_{11} = T_{22} = T_{33} = \mu.$$
 (5.3)

Under the conditions above we obtain

$$\mathfrak{P}_{311} = (\lambda_1 + \lambda_2)T_{12}, \quad \mathfrak{P}_{211} = -(\lambda_1 + \lambda_3)T_{13}, \quad \mathfrak{P}_{212} = -(\lambda_2 + \lambda_3)T_{23},$$

and now the conditions necessary for the vanishing of the divergence of T given by Equation (5.2) lead to

$$(\lambda_1 - \lambda_2)T_{12} = 0, \quad (\lambda_1 - \lambda_3)T_{13} = 0, \quad (\lambda_2 - \lambda_3)T_{23} = 0.$$

Since $\lambda_1 \lambda_2 \lambda_3 \neq 0$, these equations lead to

$$T_{12} = T_{13} = T_{23} = 0. (5.4)$$

Equations (5.3) and (5.4) imply that $\hat{T} = \text{diag}[\mu, \mu, \mu, T_{44}]$. Moreover, T is divergence-free (see Equation (5.2)) and the system $\{\mathfrak{P}_{ijk} = 0\}$ reduces to

$$k_1(T_{44} - \mu) = 0, \quad k_2(T_{44} - \mu) = 0, \quad k_3(T_{44} - \mu) = 0,$$

which complete the proof.

5.2.2 Algebraic Ricci solitons on $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$

A straightforward calculation shows that the components ρ_{ij} of the Ricci tensor of both $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$ are determined by

$$\begin{aligned} &2\rho_{11} = (\lambda_1^2 - \lambda_3^2)k_2^2 + (\lambda_1^2 - \lambda_2^2)k_3^2 + (\lambda_1 + \lambda_2 - \lambda_3)(\lambda_1 - \lambda_2 + \lambda_3), \\ &2\rho_{12} = (\lambda_3^2 - \lambda_1\lambda_2)k_1k_2, \\ &2\rho_{13} = (\lambda_2^2 - \lambda_1\lambda_3)k_1k_3, \\ &2\rho_{14} = (\lambda_2 - \lambda_3)^2k_1, \\ &2\rho_{22} = (\lambda_2^2 - \lambda_3^2)k_1^2 - (\lambda_1^2 - \lambda_2^2)k_3^2 - (\lambda_1 - \lambda_2 - \lambda_3)(\lambda_1 + \lambda_2 - \lambda_3), \\ &2\rho_{23} = (\lambda_1^2 - \lambda_2\lambda_3)k_2k_3, \\ &2\rho_{24} = (\lambda_1 - \lambda_3)^2k_2, \\ &2\rho_{33} = (\lambda_3^2 - \lambda_2^2)k_1^2 - (\lambda_1^2 - \lambda_3^2)k_2^2 - (\lambda_1 - \lambda_2 - \lambda_3)(\lambda_1 - \lambda_2 + \lambda_3), \end{aligned}$$

$$2\rho_{34} = (\lambda_1 - \lambda_2)^2 k_3,$$

$$2\rho_{44} = -(\lambda_2 - \lambda_3)^2 k_1^2 - (\lambda_1 - \lambda_3)^2 k_2^2 - (\lambda_1 - \lambda_2)^2 k_3^2.$$

In the locally symmetric case, which can only occur on $SU(2) \times \mathbb{R}$, since

$$\lambda_1 = \lambda_2 = \lambda_3 = \lambda \neq 0$$

(see Remark 5.14) a direct calculation shows that the Ricci operator is given by

$$\operatorname{Ric} = \frac{1}{2}\lambda^2 \operatorname{diag}[1, 1, 1, 0].$$

Thus, by Theorem 5.15, we obtain a non-trivial algebraic Ricci soliton for $k_1 = k_2 = k_3 = 0$ and $\mu = \frac{1}{2}\lambda^2$. Note that we can take $\lambda = 1$ remaining in the same homothetic class and hence the left-invariant metric is isomorphically homothetic to the one given by

$$[e_1, e_2] = e_3, \quad [e_1, e_3] = -e_2, \quad [e_2, e_3] = e_1,$$

which is non-Einstein and corresponds to Assertion (v) in Theorem 5.4.

Next we analyse non-trivial algebraic Ricci solitons in the non-symmetric case. Note that λ_1 , λ_2 , λ_3 must be not equal. Also, according to Theorem 5.15, the possible non-zero components of the Ricci tensor must reduce to ρ_{ii} , satisfying

$$\rho_{11} = \rho_{22} = \rho_{33} = \mu,$$

and moreover $k_1 = k_2 = k_3 = 0$. Setting $k_1 = k_2 = k_3 = 0$ the non-zero components of the Ricci tensor are given by

$$\rho_{11} = \frac{1}{2}(\lambda_1 + \lambda_2 - \lambda_3)(\lambda_1 - \lambda_2 + \lambda_3),$$

$$\rho_{22} = -\frac{1}{2}(\lambda_1 - \lambda_2 - \lambda_3)(\lambda_1 + \lambda_2 - \lambda_3),$$

$$\rho_{33} = -\frac{1}{2}(\lambda_1 - \lambda_2 - \lambda_3)(\lambda_1 - \lambda_2 + \lambda_3).$$

Since λ_1 , λ_2 and λ_3 are not equal, $\rho_{11} = \rho_{22} = \rho_{33}$ never holds. Hence, there does not exist any non-symmetric algebraic Ricci soliton on $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ or $SU(2) \times \mathbb{R}$.

5.2.3 Algebraic Bach solitons on $\widetilde{SL}(2,\mathbb{R})\times\mathbb{R}$ and $SU(2)\times\mathbb{R}$

A long but straightforward calculation shows that the components \mathfrak{B}_{ij} of the Bach tensors of $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$ are determined by the following polynomials on the structure constants.

$$24\mathfrak{B}_{11} = -4(\lambda_2 - \lambda_3)^2(\lambda_2^2 + \lambda_3^2 + \lambda_2\lambda_3)k_1^4 -4(5\lambda_1^4 - 3\lambda_3^4 - \lambda_1\lambda_3(3\lambda_1^2 - \lambda_3^2))k_2^4 -4(5\lambda_1^4 - 3\lambda_2^4 - \lambda_1\lambda_2(3\lambda_1^2 - \lambda_2^2))k_3^4 +(8\lambda_3^4 + 4\lambda_3^3(\lambda_1 - \lambda_2) - 24\lambda_1^2\lambda_2^2 + 3\lambda_1^2\lambda_3^2 - \lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(6\lambda_1 + 2\lambda_2 - \lambda_3))k_1^2k_2^2$$

$$\begin{split} + & (8\lambda_{2}^{4} + 4\lambda_{2}^{3}(\lambda_{1} - \lambda_{3}) + 3\lambda_{1}^{2}\lambda_{2}^{2} - 24\lambda_{1}^{2}\lambda_{3}^{2} - \lambda_{2}^{2}\lambda_{3}^{2} + 2\lambda_{1}\lambda_{2}\lambda_{3}(6\lambda_{1} - \lambda_{2} + 2\lambda_{3}))k_{1}^{2}k_{3}^{2} \\ - & (40\lambda_{1}^{4} - 12\lambda_{1}^{3}(\lambda_{2} + \lambda_{3}) - \lambda_{1}^{2}\lambda_{2}^{2} - \lambda_{1}^{2}\lambda_{3}^{2} - 24\lambda_{2}^{2}\lambda_{3}^{2} + 2\lambda_{1}\lambda_{2}\lambda_{3}(\lambda_{1} + 2\lambda_{2} + 2\lambda_{3}))k_{2}^{2}k_{3}^{2} \\ + & (\lambda_{2} - \lambda_{3})^{2}(3\lambda_{1}^{2} + 8\lambda_{2}^{2} + 8\lambda_{3}^{2} + 4\lambda_{1}\lambda_{2} + 4\lambda_{1}\lambda_{3} + 8\lambda_{2}\lambda_{3})k_{1}^{2} \\ - & (\lambda_{1} - \lambda_{3})(40\lambda_{1}^{3} + 24\lambda_{3}^{3} - 4\lambda_{1}^{2}(3\lambda_{2} - 4\lambda_{3}) - \lambda_{2}^{2}(\lambda_{1} + 3\lambda_{3}) \\ & \quad + 4\lambda_{3}^{2}(4\lambda_{1} - 3\lambda_{2}) - 8\lambda_{1}\lambda_{2}\lambda_{3})k_{2}^{2} \\ - & (\lambda_{1} - \lambda_{2})(40\lambda_{1}^{3} + 24\lambda_{2}^{3} + 4\lambda_{1}^{2}(4\lambda_{2} - 3\lambda_{3}) + 4\lambda_{2}^{2}(4\lambda_{1} - 3\lambda_{3}) \\ & \quad - \lambda_{3}^{2}(\lambda_{1} + 3\lambda_{2}) - 8\lambda_{1}\lambda_{2}\lambda_{3})k_{3}^{2} \\ - & 4(5\lambda_{1}^{4} - 3\lambda_{2}^{4} - 3\lambda_{3}^{4} - 3\lambda_{1}^{3}(\lambda_{2} + \lambda_{3}) + \lambda_{2}^{3}(\lambda_{1} + 3\lambda_{3}) + \lambda_{3}^{3}(\lambda_{1} + 3\lambda_{2}) \\ & \quad + \lambda_{1}\lambda_{2}\lambda_{3}(\lambda_{1} - \lambda_{2} - \lambda_{3})), \end{split}$$

$$\begin{split} 12\mathfrak{B}_{12} &= -k_1k_2\left((8\lambda_3^4 - 8\lambda_1\lambda_3^3 - 4\lambda_2\lambda_3^3 - \lambda_2^2\lambda_3^2 + \lambda_1\lambda_2\lambda_3(4\lambda_2 + \lambda_3))k_1^2 \\ &\quad + (8\lambda_3^4 - 8\lambda_1^3\lambda_2 - 4\lambda_1\lambda_3^3 - \lambda_1^2\lambda_3^2 + \lambda_1\lambda_2\lambda_3(4\lambda_1 + \lambda_3))k_2^2 \\ &\quad + (\lambda_3^2 - \lambda_1\lambda_2)(8\lambda_1^2 + 8\lambda_2^2 + 5\lambda_1\lambda_2)k_3^2 \\ &\quad + 8\lambda_3^4 - 8\lambda_1^3\lambda_2 - 8\lambda_1\lambda_2^3 - 4\lambda_3^3(\lambda_1 + \lambda_2) - 5\lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 - \lambda_2^2\lambda_3^2 \\ &\quad + \lambda_1\lambda_2\lambda_3(10\lambda_1 + 10\lambda_2 + 3\lambda_3))\,, \end{split}$$

$$\begin{split} 12\mathfrak{B}_{13} &= -k_1k_3\left((8\lambda_2^4 - 4\lambda_2^3\lambda_3 - 8\lambda_1\lambda_3^3 - \lambda_2^2\lambda_3^2 + \lambda_1\lambda_2\lambda_3(\lambda_2 + 4\lambda_3))k_1^2 \\ &\quad + (\lambda_2^2 - \lambda_1\lambda_3)(8\lambda_1^2 + 8\lambda_3^2 + 5\lambda_1\lambda_3)k_2^2 \\ &\quad + (8\lambda_2^4 - 8\lambda_1^3\lambda_3 - 4\lambda_1\lambda_2^3 - \lambda_1^2\lambda_2^2 + \lambda_1\lambda_2\lambda_3(4\lambda_1 + \lambda_2))k_3^2 \\ &\quad + 8\lambda_2^4 - 8\lambda_1^3\lambda_3 - 4\lambda_2^3(\lambda_1 + \lambda_3) - 8\lambda_1\lambda_3^3 - \lambda_1^2\lambda_2^2 - 5\lambda_1^2\lambda_3^2 - \lambda_2^2\lambda_3^2 \\ &\quad + \lambda_1\lambda_2\lambda_3(10\lambda_1 + 3\lambda_2 + 10\lambda_3))\,, \end{split}$$

$$\begin{split} 12\mathfrak{B}_{14} &= -k_1 \left(8(\lambda_2-\lambda_3)^2(\lambda_2^2+\lambda_3^2+\lambda_2\lambda_3)k_1^2\right.\\ &\quad + (8\lambda_3^4-4\lambda_3^3(\lambda_1+\lambda_2)+8\lambda_1^2\lambda_2^2-\lambda_1^2\lambda_3^2-\lambda_2^2\lambda_3^2-2\lambda_1\lambda_2\lambda_3(2\lambda_1+2\lambda_2-\lambda_3))k_2^2\\ &\quad + (8\lambda_2^4-4\lambda_2^3(\lambda_1+\lambda_3)-\lambda_1^2\lambda_2^2+8\lambda_1^2\lambda_3^2-\lambda_2^2\lambda_3^2-2\lambda_1\lambda_2\lambda_3(2\lambda_1-\lambda_2+2\lambda_3))k_3^2\\ &\quad - (\lambda_2-\lambda_3)^2(\lambda_1^2-8\lambda_2^2-8\lambda_3^2+4\lambda_1\lambda_2+4\lambda_1\lambda_3-8\lambda_2\lambda_3)\right), \end{split}$$

$$\begin{split} 24\mathfrak{B}_{22} &= -4(5\lambda_2^4 - 3\lambda_3^4 - \lambda_2\lambda_3(3\lambda_2^2 - \lambda_3^2))k_1^4 \\ &\quad -4(\lambda_1 - \lambda_3)^2(\lambda_1^2 + \lambda_3^2 + \lambda_1\lambda_3)k_2^4 \\ &\quad +4(3\lambda_1^4 - 5\lambda_2^4 - \lambda_1\lambda_2(\lambda_1^2 - 3\lambda_2^2))k_3^4 \\ &\quad +(8\lambda_3^4 - 4\lambda_3^3(\lambda_1 - \lambda_2) - 24\lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 3\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(2\lambda_1 + 6\lambda_2 - \lambda_3))k_1^2k_2^2 \\ &\quad +(-40\lambda_2^4 + 12\lambda_2^3(\lambda_1 + \lambda_3) + \lambda_1^2\lambda_2^2 + 24\lambda_1^2\lambda_3^2 + \lambda_2^2\lambda_3^2 - 2\lambda_1\lambda_2\lambda_3(2\lambda_1 + \lambda_2 + 2\lambda_3))k_1^2k_3^2 \\ &\quad +(8\lambda_1^4 + 4\lambda_1^3(\lambda_2 - \lambda_3) + 3\lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 - 24\lambda_2^2\lambda_3^2 - 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 6\lambda_2 - 2\lambda_3))k_2^2k_3^2 \\ &\quad -(\lambda_2 - \lambda_3)(40\lambda_2^3 + 24\lambda_3^3 - \lambda_1^2(\lambda_2 + 3\lambda_3) - 4\lambda_2^2(3\lambda_1 - 4\lambda_3) \\ &\quad -4\lambda_3^2(3\lambda_1 - 4\lambda_2) - 8\lambda_1\lambda_2\lambda_3)k_1^2 \\ &\quad +(\lambda_1 - \lambda_3)^2(8\lambda_1^2 + 3\lambda_2^2 + 8\lambda_3^2 + 4\lambda_1\lambda_2 + 8\lambda_1\lambda_3 + 4\lambda_2\lambda_3)k_2^2 \\ &\quad +(\lambda_1 - \lambda_2)(24\lambda_1^3 + 40\lambda_2^3 + 4\lambda_1^2(4\lambda_2 - 3\lambda_3) + 4\lambda_2^2(4\lambda_1 - 3\lambda_3) \\ &\quad -\lambda_3^2(3\lambda_1 + \lambda_2) - 8\lambda_1\lambda_2\lambda_3)k_3^2 \\ &\quad +4(3\lambda_1^4 - 5\lambda_2^4 + 3\lambda_3^4 - \lambda_1^3(\lambda_2 + 3\lambda_3) + 3\lambda_2^3(\lambda_1 + \lambda_3) - \lambda_3^3(3\lambda_1 + \lambda_2) \\ &\quad +\lambda_1\lambda_2\lambda_3(\lambda_1 - \lambda_2 + \lambda_3)), \end{split}$$

$$\begin{split} 12\mathfrak{B}_{23} &= -k_2k_3\left((\lambda_1^2 - \lambda_2\lambda_3)(8\lambda_2^2 + 8\lambda_3^2 + 5\lambda_2\lambda_3)k_1^2\right. \\ &\quad + (8\lambda_1^4 - 4\lambda_1^3\lambda_3 - 8\lambda_2\lambda_3^3 - \lambda_1^2\lambda_2^2 + \lambda_1\lambda_2\lambda_3(\lambda_1 + 4\lambda_3))k_2^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_1^3(\lambda_2 + \lambda_3) - 8\lambda_2^3\lambda_3 - 8\lambda_2\lambda_3^3 - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 - 5\lambda_2^2\lambda_3^2 \\ &\quad + 8\lambda_1^4 - 4\lambda_3^3(\lambda_1 + \lambda_2) + 8\lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 - \lambda_2^2\lambda_3^2 \\ &\quad - 2\lambda_1\lambda_2\lambda_3(2\lambda_1 + 2\lambda_2 - \lambda_3))k_1^2 \\ &\quad + 8(\lambda_1 - \lambda_3)^2(\lambda_1^2 + \lambda_3^2 + \lambda_1\lambda_3)k_2^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_3^3(\lambda_1 + \lambda_2) + 8\lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 8\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 2\lambda_3))k_3^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_3^3(\lambda_2 + \lambda_3) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 8\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 2\lambda_3))k_3^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_3^3(\lambda_1 + \lambda_2) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 8\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 2\lambda_3))k_3^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_3^3(\lambda_1 + \lambda_3) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 8\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 2\lambda_3))k_2^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_1^3(\lambda_2 + \lambda_3) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 8\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 2\lambda_3))k_2^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_3^3(\lambda_1 + \lambda_2) - 24\lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 8\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 2\lambda_3))k_2^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_1^2(\lambda_2 + \lambda_3) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 8\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 2\lambda_3))k_2^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_1^2(\lambda_2 + \lambda_3) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 + 8\lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 2\lambda_3))k_2^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_1^2(\lambda_2 - \lambda_3) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 - \lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(2\lambda_1 - 2\lambda_2 + 3\lambda_3))k_1^2k_2^2 \\ &\quad + (8\lambda_1^4 - 4\lambda_1^2(\lambda_2 - \lambda_3) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 - \lambda_2^2\lambda_3^2 + 2\lambda_1\lambda_2\lambda_3(2\lambda_1 - 2\lambda_2 - 6\lambda_3))k_1^2k_3^2 \\ &\quad + (\lambda_1 - \lambda_2)^2(\lambda_1^2 + \lambda_2) - \lambda_1^2\lambda_2^2 - \lambda_1^2\lambda_3^2 - 2\lambda_2^2\lambda_3^2 - 2\lambda_1\lambda_2\lambda_3(\lambda_1 - 2\lambda_2 - 6\lambda_3))k_1^2k_3^2 \\ &\quad + (\lambda_2 - \lambda_3)(24\lambda_3^2 + 40\lambda_3^3 - 4\lambda_1^2(3\lambda_2 - 4\lambda_3) - 4\lambda_2^2(3\lambda_1 - 4\lambda_2) - 8\lambda_1\lambda_2\lambda_3)k_3^2 \\ &\quad + (\lambda_1 - \lambda_3)(24\lambda_3^2 + 40\lambda_3^3 - 4\lambda_1^2(3\lambda_2 - 4\lambda_3) - 4\lambda_2^2(3\lambda_1 - 4\lambda_2) - 8\lambda_1\lambda_2\lambda_3)k_3^2 \\ &\quad + (\lambda_1 - \lambda_2)^2(8\lambda_1^2 + 8\lambda_2^2 + 3\lambda_3^2 + 8\lambda_1\lambda_2 + 4\lambda_1\lambda_3 + 4\lambda_2\lambda_3)k_3^2 \\ &\quad + (\lambda_1 - \lambda_2)^2(8\lambda_1^2 + 8\lambda_2^2 + 3\lambda_3^2 + 8\lambda_1\lambda_2 + 4\lambda_1\lambda_3 + 4\lambda_2\lambda_3)$$

Recall that $\mathfrak{B}_{44} = -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33}$ since the Bach tensor is trace-free. If the space is locally symmetric we have $\lambda_1 = \lambda_2 = \lambda_3$ (see Remark 5.14) and a direct calculation shows it is locally conformally flat and thus Bach-flat.

In the non-symmetric case – where λ_1 , λ_2 , λ_3 are not equal –, we make use of Theorem 5.15 to analyse the existence of non-trivial algebraic Bach solitons. Thus, the only possible non-zero components of the Bach tensor must correspond to \mathfrak{B}_{ii} , with

$$\mathfrak{B}_{11}=\mathfrak{B}_{22}=\mathfrak{B}_{33}=\mu,$$

and moreover, $k_1 = k_2 = k_3 = 0$. In what follows we take an isomorphic homothety to assume

 $\lambda_1 = 1$. A direct computation shows that $\mathfrak{B}_{ij} = 0$ for $i \neq j$ and, moreover,

$$0 = (\lambda_2 - 1)(\mathfrak{B}_{22} - \mathfrak{B}_{33}) + (\lambda_2 - \lambda_3)(\mathfrak{B}_{11} - \mathfrak{B}_{22})$$

= $-\frac{1}{3}(\lambda_2 - \lambda_3)(\lambda_2 - 1)(\lambda_3 - 1)$
 $\times ((\lambda_2 + 1)^2 + (\lambda_3 + 1)^2 + (\lambda_2 + \lambda_3)^2 + 3(\lambda_2^2 + \lambda_3^2 + 1)).$

Since the last factor above does not vanish, then either $\lambda_2 = \lambda_3$, or $\lambda_2 = 1$, or $\lambda_3 = 1$.

If $\lambda_2 = \lambda_3$, then

$$\mathfrak{B}_{11} = -\frac{1}{6}(\lambda_2 - 1)(\lambda_2 - 5),$$

$$\mathfrak{B}_{22} = \mathfrak{B}_{33} = \frac{1}{6}(\lambda_2 - 1)(\lambda_2 - 3),$$

so the condition $\mathfrak{B}_{11} = \mathfrak{B}_{22}$ implies that $\lambda_2 = 4$. Thus,

$$\widehat{B} = \frac{1}{2} \operatorname{diag}[1, 1, 1, -3], \quad \mu = \frac{1}{2},$$

and the left-invariant metric is given by

$$[e_1, e_2] = 4e_3, \quad [e_1, e_3] = -4e_2, \quad [e_2, e_3] = e_1.$$

This space, which is not a Ricci soliton, corresponds to Assertion (i) in Theorem 5.8.

If $\lambda_2 = 1$, we proceed as in the previous case and obtain that $\lambda_3 = \frac{1}{4}$, so the left-invariant metric is given by

$$[e_1, e_2] = \frac{1}{4}e_3, \quad [e_1, e_3] = -e_2, \quad [e_2, e_3] = e_1.$$

Now, the transformation $(e_1, e_2, e_3, e_4) \mapsto 4(e_3, e_2, -e_1, e_4)$ shows that this case is isomorphically homothetic to the previous one.

Finally, if $\lambda_3 = 1$, proceeding exactly as in the previous cases we get $\lambda_2 = \frac{1}{4}$. Hence, the left-invariant metric is given by

$$[e_1, e_2] = e_3, \quad [e_1, e_3] = -\frac{1}{4}e_2, \quad [e_2, e_3] = e_1,$$

and the transformation $(e_1, e_2, e_3, e_4) \mapsto (e_1, -e_3, e_2, e_4)$ shows that this case is isomorphically isometric to the previous one.

5.3 Algebraic solitons on $\mathbb{R}^3 \rtimes \mathbb{R}$

The left-invariant metrics on semi-direct extensions of the Abelian Lie algebra have already been described in Section 1.4.2. In this case there exists an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ such that

$$[e_1, e_4] = ae_1 + be_2 + ce_3, \qquad [e_2, e_4] = -be_1 + fe_2 + he_3,$$

$$[e_3, e_4] = -ce_1 - he_2 + pe_3.$$
 (5.5)

Remark 5.16. $\mathbb{R}^3 \rtimes \mathbb{R}$ is locally symmetric if and only if it is Einstein (a = f = p) in which case it is homothetic to \mathbb{H}^4 . Otherwise, it is isomorphically isometric to (5.5) with a = f = c = h = 0, $p \neq 0$ (thus being homothetic to $\mathbb{H}^2 \times \mathbb{R}^2$), or $a = f \neq 0$, p = c = h = 0 (thus being homothetic to $\mathbb{H}^3 \times \mathbb{R}$).

Note that in the Abelian case any metric is flat and any tensor field T is an algebraic T-soliton. Hence, we exclude this case throughout this section and assume that at least one of the structure constants in (5.5) is non-zero.

Remark 5.17. The isomorphic isometry determined by

$$(e_1, e_2, e_3, e_4) \mapsto (e_2, e_1, e_3, e_4)$$

implies that $(a, f, p, b, c, h) \sim (f, a, p, -b, h, c)$. Analogously,

$$(e_1, e_2, e_3, e_4) \mapsto (e_3, e_2, e_1, e_4)$$

gives $(a, f, p, b, c, h) \sim (p, f, a, -h, -c, -b)$ and

$$(e_1, e_2, e_3, e_4) \mapsto (e_1, e_3, e_2, e_4)$$

shows that $(a, f, p, b, c, h) \sim (a, p, f, c, b, -h)$.

5.3.1 Algebraic *T*-solitons on $\mathbb{R}^3 \rtimes \mathbb{R}$

Recall that we are assuming that the left-invariant tensor field T is divergence-free, so the following conditions hold:

$$(2a + f + p)T_{14} + bT_{24} + cT_{34} = 0,$$

$$bT_{14} - (a + 2f + p)T_{24} - hT_{34} = 0,$$

$$cT_{14} + hT_{24} - (a + f + 2p)T_{34} = 0,$$

$$aT_{11} + fT_{22} + pT_{33} - (a + f + p)T_{44} = 0.$$

(5.6)

The conditions for $\mathfrak{D} = \hat{T} - \mu \operatorname{Id}$ to be a derivation are determined by a system of twenty one (up to duplicities) polynomial equations on the soliton constant μ , the structure constants in (5.5) and the components T_{ij} , given by $\{\mathfrak{P}_{ijk} = 0\}$, where

$$\begin{aligned} \mathfrak{P}_{211} &= bT_{14} + aT_{24}, \\ \mathfrak{P}_{212} &= -fT_{14} + bT_{24}, \\ \mathfrak{P}_{213} &= -hT_{14} + cT_{24}, \\ \mathfrak{P}_{311} &= cT_{14} + aT_{34}, \\ \mathfrak{P}_{312} &= hT_{14} + bT_{34}, \\ \mathfrak{P}_{313} &= -pT_{14} + cT_{34}, \end{aligned}$$

$$\begin{split} \mathfrak{P}_{321} &= cT_{24} - bT_{34}, \\ \mathfrak{P}_{322} &= hT_{24} + fT_{34}, \\ \mathfrak{P}_{323} &= -pT_{24} + hT_{34}, \\ \mathfrak{P}_{411} &= a(T_{44} - \mu) - 2bT_{12} - 2cT_{13}, \\ \mathfrak{P}_{412} &= b(T_{11} - T_{22} + T_{44} - \mu) - (a - f)T_{12} - hT_{13} - cT_{23}, \\ \mathfrak{P}_{413} &= c(T_{11} - T_{33} + T_{44} - \mu) + hT_{12} - (a - p)T_{13} - bT_{23}, \\ \mathfrak{P}_{414} &= -aT_{14} - bT_{24} - cT_{34}, \\ \mathfrak{P}_{421} &= b(T_{11} - T_{22} - T_{44} + \mu) + (a - f)T_{12} - hT_{13} - cT_{23}, \\ \mathfrak{P}_{422} &= f(T_{44} - \mu) + 2bT_{12} - 2hT_{23}, \\ \mathfrak{P}_{423} &= h(T_{22} - T_{33} + T_{44} - \mu) + cT_{12} + bT_{13} - (f - p)T_{23}, \\ \mathfrak{P}_{424} &= bT_{14} - fT_{24} - hT_{34}, \\ \mathfrak{P}_{431} &= c(T_{11} - T_{33} - T_{44} + \mu) + hT_{12} + (a - p)T_{13} - bT_{23}, \\ \mathfrak{P}_{432} &= h(T_{22} - T_{33} - T_{44} + \mu) + cT_{12} + bT_{13} + (f - p)T_{23}, \\ \mathfrak{P}_{433} &= p(T_{44} - \mu) + 2cT_{13} + 2hT_{23}, \\ \mathfrak{P}_{434} &= cT_{14} + hT_{24} - pT_{34}. \end{split}$$

In order to determine all the algebraic T-solitons, it is important to analyse the behaviour of the distinguished direction e_4 , since it must be an eigenspace of the endomorphism \hat{T} in most cases.

Lemma 5.18. Let $\mathbb{R}^3 \rtimes \mathbb{R}$ be an algebraic *T*-soliton with soliton constant μ . Then the following relations hold.

(a) With no further assumptions, either e_4 is an eigenvector of \hat{T} , i.e.,

$$T_{14} = T_{24} = T_{34} = 0,$$

or a, f and p are different with a f p = 0 and b ch = 0. In the latter case, the original metric is isomorphically isometric to a metric with a = 0 and b = c = 0.

(b) Assume that a, f and p are different. If e_4 is an eigenvector of \widehat{T} with associated eigenvalue $T_{44} \neq \mu$, then afp = 0. In addition, the original metric is isomorphically isometric to a metric with a = 0.

Proof. We prove each assertion separately in what follows.

Assertion (a)

Let us assume that e_4 is not an eigenvector of \hat{T} , i.e., at least one of the components T_{14} , T_{24} and T_{34} does not vanish. Firstly, we show that a, f and p are necessarily different. If two of them are equal, using Remark 5.17 one may assume f = a. If f = a, we consider

$$\mathfrak{P}_{211} = bT_{14} + aT_{24}, \quad \mathfrak{P}_{212} = -aT_{14} + bT_{24}.$$

Hence, either $T_{14} = T_{24} = 0$ or a = b = 0. If $T_{14} = T_{24} = 0$ then

$$\mathfrak{P}_{311} = aT_{34}, \quad \mathfrak{P}_{434} = -pT_{34}, \quad \mathfrak{P}_{312} = bT_{34}, \quad \mathfrak{P}_{313} = cT_{34}, \quad \mathfrak{P}_{323} = hT_{34},$$

so $T_{34} \neq 0$ implies that the Lie group is Abelian. If some of T_{14} and T_{24} does not vanish, then a = b = 0 and

$$\mathfrak{P}_{313} + \mathfrak{P}_{414} = -pT_{14}, \quad \mathfrak{P}_{323} + \mathfrak{P}_{424} = -pT_{24},$$

which together with

$$\mathfrak{P}_{311} = cT_{14}, \quad \mathfrak{P}_{321} = cT_{24}, \quad \mathfrak{P}_{312} = hT_{14}, \quad \mathfrak{P}_{322} = hT_{24},$$

imply that p = c = h = 0, so the Lie group turns out to be Abelian again. This shows that a, f and p have to be different.

Secondly, we show that bch = 0 and afp = 0. The linear combinations

$$\begin{aligned} \mathfrak{P}_{213} - \mathfrak{P}_{312} - \mathfrak{P}_{321} &= -2hT_{14}, \\ \mathfrak{P}_{213} + \mathfrak{P}_{312} + \mathfrak{P}_{321} &= -2cT_{24}, \\ \mathfrak{P}_{213} + \mathfrak{P}_{312} - \mathfrak{P}_{321} &= -2bT_{34}, \end{aligned}$$

clearly lead to bch = 0. If b = 0, then

$$\mathfrak{P}_{212} = -fT_{14}, \quad \mathfrak{P}_{211} = aT_{24}, \quad f\mathfrak{P}_{311} + c\mathfrak{P}_{212} = afT_{34}.$$

Consequently, afp = 0. The same conclusion is obtained if c = 0, since

$$\mathfrak{P}_{313} = -pT_{14}, \quad p\mathfrak{P}_{211} + b\mathfrak{P}_{313} = apT_{24}, \quad \mathfrak{P}_{311} = aT_{34},$$

or if h = 0, using

$$p\mathfrak{P}_{212} + b\mathfrak{P}_{323} = -fpT_{14}, \quad \mathfrak{P}_{323} = -pT_{24}, \quad \mathfrak{P}_{322} = fT_{34}$$

Thus, afp = 0 and bch = 0. Finally, by Remark 5.17 we can take a = 0 working, if necessary, with an isomorphically isometric metric. Now, if b = 0, we compute

$$\mathfrak{P}_{311} = cT_{14}, \quad \mathfrak{P}_{321} = cT_{24}, \quad \mathfrak{P}_{414} = -cT_{34},$$

so necessarily c = 0. If c = 0, then b = 0 since

$$\mathfrak{P}_{211} = bT_{14}, \quad \mathfrak{P}_{414} = -bT_{24}, \quad \mathfrak{P}_{321} = -bT_{34}.$$

Finally, if h = 0 we use

$$\mathfrak{P}_{211} = bT_{14}, \quad \mathfrak{P}_{212} = -fT_{14} + bT_{24}, \quad \mathfrak{P}_{312} = bT_{34},$$

to obtain that b = 0 and, therefore, necessarily c = 0. This proves Assertion (a).

Assertion (b)

Since

$$\mathfrak{P}_{411} + \mathfrak{P}_{422} + \mathfrak{P}_{433} = (a + f + p)(T_{44} - \mu),$$

and we are assuming $T_{44} \neq \mu$, it follows that a + f + p = 0. Therefore, we can take p = -a - f, and with this condition we obtain

$$\mathfrak{P}_{421} - \mathfrak{P}_{412} = 2\left((a-f)T_{12} - b(T_{44} - \mu)\right),$$

$$\mathfrak{P}_{431} - \mathfrak{P}_{413} = 2\left((2a+f)T_{13} - c(T_{44} - \mu)\right),$$

$$\mathfrak{P}_{432} - \mathfrak{P}_{423} = 2\left((a+2f)T_{23} - h(T_{44} - \mu)\right).$$

Since p = -a - f and a, f, p are different, we have $a - f \neq 0, 2a + f \neq 0$ and $a + 2f \neq 0$, so

$$T_{12} = \frac{b}{a-f}(T_{44} - \mu), \quad T_{13} = \frac{c}{2a+f}(T_{44} - \mu), \quad T_{23} = \frac{h}{a+2f}(T_{44} - \mu).$$

We set $T_{14} = T_{24} = T_{34} = 0$. We need to see that afp = 0, i.e., af(a + f) = 0. Now, we compute the polynomials

$$\begin{aligned} \mathfrak{P}_{411} &= \left(a - \frac{2b^2}{a-f} - \frac{2c^2}{2a+f}\right) (T_{44} - \mu), \\ \mathfrak{P}_{422} &= \left(f - \frac{2h^2}{a+2f} + \frac{2b^2}{a-f}\right) (T_{44} - \mu), \\ \mathfrak{P}_{433} &= -\left((a+f) - \frac{2c^2}{2a+f} - \frac{2h^2}{a+2f}\right) (T_{44} - \mu), \\ \mathfrak{P}_{421} &= b(T_{11} - T_{22}) - \frac{3ch(a+f)}{(2a+f)(a+2f)} (T_{44} - \mu), \\ \mathfrak{P}_{413} &= c(T_{11} - T_{33}) + \frac{3bhf}{(a-f)(a+2f)} (T_{44} - \mu), \\ \mathfrak{P}_{432} &= h(T_{22} - T_{33}) + \frac{3bca}{(a-f)(2a+f)} (T_{44} - \mu). \end{aligned}$$
(5.7)

If bch = 0, it follows immediately from the expressions above that af(a + f) = 0. For instance, assume that b = 0. Hence, $\mathfrak{P}_{411} = \mathfrak{P}_{422} = 0$ in Equation (5.7) imply that $ch \neq 0$ or, otherwise, a = 0 or f = 0. Now, if $ch \neq 0$, $\mathfrak{P}_{421} = 0$ in the same equation leads to a + f = 0. Thus, in any case, af(a + f) = 0. A similar argument works both for c = 0 and h = 0.

Next we analyse the case $bch \neq 0$. We assume that $af(a + f) \neq 0$ and argue for a contradiction. At this point, we consider Equation (5.6), which characterizes the vanishing of the divergence of T. A direct calculation shows that such equation reduces to

$$a(T_{11} - T_{33}) + f(T_{22} - T_{33}) = 0.$$
(5.8)

Since $ch \neq 0$, we can isolate $T_{11} - T_{33}$ and $T_{22} - T_{33}$ in $\mathfrak{P}_{413} = 0$ and $\mathfrak{P}_{432} = 0$ in Equation (5.7), respectively. Replacing them in Equation (5.8) we get

$$abf(T_{44} - \mu)\left((a + 2f)c^2 + (2a + f)h^2\right) = 0$$

and therefore $(a+2f)c^2 + (2a+f)h^2 = 0$, which leads to

$$c^2 = -\frac{(2a+f)h^2}{a+2f}$$

Finally, using this last expression in \mathfrak{P}_{433} in Equation (5.7) we obtain

$$\mathfrak{P}_{433} = -(a+f)(T_{44}-\mu),$$

which leads to a contradiction since this expression does not vanish.

Hence, we have shown that afp = 0. Note that, by Remark 5.17, we can take a = 0 working, if necessary, with an isomorphically isometric metric. This proves Assertion (b).

Theorem 5.19. If $\mathbb{R}^3 \rtimes \mathbb{R}$ is not Einstein, then it is a non-trivial algebraic *T*-soliton with soliton constant μ if and only if one of the following holds.

(i) The self-adjoint part of $ad(e_4)$ has two equal eigenvalues. In this case, the space is isomorphically isometric to a Lie group given by $a = f \neq p$, and

$$T = \begin{pmatrix} T_{11} & T_{12} & 0 & 0 \\ * & T_{22} & 0 & 0 \\ * & * & T_{33} & 0 \\ * & * & * & \mu \end{pmatrix} \neq \mu \langle \cdot, \cdot \rangle,$$

with

$$bT_{12} = 0, \quad b(T_{11} - T_{22}) = 0, \quad a(T_{11} + T_{22}) + pT_{33} - (2a + p)\mu = 0,$$

 $c(T_{11} - T_{33}) + hT_{12} = 0, \quad h(T_{22} - T_{33}) + cT_{12} = 0.$

(ii) The self-adjoint part of $ad(e_4)$ has three different eigenvalues and the tensor field \hat{T} is diagonal,

$$\widehat{T} = \text{diag}[T_{11}, T_{22}, T_{33}, \mu] \neq \mu \text{ Id},$$

with

$$b(T_{11} - T_{22}) = 0, \quad c(T_{11} - T_{33}) = 0, \quad h(T_{22} - T_{33}) = 0,$$

 $aT_{11} + fT_{22} + pT_{33} - (a + f + p)\mu = 0.$

(iii) The self-adjoint part of $ad(e_4)$ has three different eigenvalues and the tensor field \widehat{T} is not diagonal. In this case, the space is isomorphically isometric to a Lie group given by one of the following:

(iii.a) $a = 0, p = -f \neq 0, b = c, 2b^2 - f^2 + h^2 = 0, and$

$$T = \begin{pmatrix} T_{11} & -\frac{b}{f}(T_{44} - \mu) & \frac{b}{f}(T_{44} - \mu) & 0\\ * & T_{22} & \frac{h}{2f}(T_{44} - \mu) & 0\\ * & * & T_{22} & 0\\ * & * & * & T_{44} \end{pmatrix},$$

with

$$T_{44} \neq \mu$$
, $b(2f(T_{11} - T_{22}) - 3h(T_{44} - \mu)) = 0$.

(*iii.b*) $a = 0, p = -f \neq 0, b = c = 0, h = f$ and

$$T = \begin{pmatrix} T_{11} & 0 & 0 & 0 \\ * & T_{22} & \frac{1}{2}(T_{44} - \mu) & T_{24} \\ * & * & T_{22} & -T_{24} \\ * & * & * & T_{44} \end{pmatrix},$$

with
$$T_{24} \neq 0$$
.

Proof. We consider the self-adjoint part of $ad(e_4)$, diag[a, f, p], and analyse the cases where two of those structure constants are equal or all of them are different separately. Recall that the metric is Einstein if a = f = p (see Remark 5.16), and if two of the structure constants a, f, p are equal, we may assume $a = f \neq p$ just working, if necessary, with an isomorphically isometric metric (see Remark 5.17).

Case 1: $a = f \neq p$.

By Lemma 5.18-(*a*), we know that $T_{14} = T_{24} = T_{34} = 0$. Besides, considering the linear combination

$$\mathfrak{P}_{411} + \mathfrak{P}_{422} + \mathfrak{P}_{433} = (a + f + p)(T_{44} - \mu), \tag{5.9}$$

we must analyse the cases $T_{44} = \mu$ and a + f + p = 0 separately.

Case 1.1: $T_{44} = \mu$.

Since

$$\mathfrak{P}_{431} - \mathfrak{P}_{413} = 2(a-p)T_{13}, \quad \mathfrak{P}_{432} - \mathfrak{P}_{423} = 2(a-p)T_{23},$$

we immediately see that

$$T_{13} = T_{23} = 0.$$

Now, a direct calculation shows that the system of polynomial equations $\{\mathfrak{P}_{ijk} = 0\}$ is determined by

$$\mathfrak{P}_{422} = 2bT_{12} = 0,$$

 $\mathfrak{P}_{413} = c(T_{11} - T_{33}) + hT_{12} = 0,$
 $\mathfrak{P}_{423} = h(T_{22} - T_{33}) + cT_{12} = 0$

while the vanishing of the divergence of the tensor T given by Equation (5.6) reduces to

$$a(T_{11} + T_{22}) + pT_{33} - (2a + p)\mu = 0.$$

This proves Assertion (i).

Case 1.2: $T_{44} \neq \mu$ and a + f + p = 0. In this situation we have that f = a and p = -2a, with $a \neq 0$. A direct calculation shows that

$$\begin{aligned} \mathfrak{P}_{421} - \mathfrak{P}_{412} &= -2b(T_{44} - \mu), \\ \mathfrak{P}_{431} - \mathfrak{P}_{413} &= 6aT_{13} - 2c(T_{44} - \mu), \\ \mathfrak{P}_{432} - \mathfrak{P}_{423} &= 6aT_{23} - 2h(T_{44} - \mu), \end{aligned}$$

and therefore $T_{44} \neq \mu$ and $a \neq 0$ imply that

$$b = 0$$
, $T_{13} = \frac{c}{3a}(T_{44} - \mu)$, $T_{23} = \frac{h}{3a}(T_{44} - \mu)$.

Now, the polynomials $\mathfrak{P}_{421}, \mathfrak{P}_{411}$ and \mathfrak{P}_{433} reduce to

$$\begin{aligned} \mathfrak{P}_{421} &= -\frac{2}{3a}ch(T_{44} - \mu), \\ \mathfrak{P}_{411} &= -\frac{1}{3a}(3a^2 - 2c^2)(T_{44} - \mu), \\ \mathfrak{P}_{433} &= -\frac{2}{3a}(3a^2 - c^2 - h^2)(T_{44} - \mu), \end{aligned}$$

which imply

$$ch = 0$$
, $3a^2 - 2c^2 = 0$, $3a^2 - c^2 - h^2 = 0$

Since $a \neq 0$ these equations are incompatible, so we conclude that there are no algebraic T-solitons in this case.

Case 2: $a \neq f \neq p$, $a \neq p$.

If a, f and p are all different we distinguish two possibilities depending on whether T_{14} , T_{24} and T_{34} are all zero or not. When $T_{14} = T_{24} = T_{34} = 0$ we analyse the two cases given by Equation (5.9), i.e., $T_{44} = \mu$ and a + f + p = 0, separately.

Case 2.1: $T_{14} = T_{24} = T_{34} = 0$ and $T_{44} = \mu$. Since $T_{44} = \mu$, it is not difficult to check that

$$\begin{aligned} \mathfrak{P}_{421} - \mathfrak{P}_{412} &= 2(a-f)T_{12}, \\ \mathfrak{P}_{431} - \mathfrak{P}_{413} &= 2(a-p)T_{13}, \\ \mathfrak{P}_{432} - \mathfrak{P}_{423} &= 2(f-p)T_{23}, \end{aligned}$$

and so

$$T_{12} = T_{13} = T_{23} = 0$$

Applying these relations, a direct calculation shows that the tensor \widehat{T} is diagonal,

$$\widehat{T} = \operatorname{diag}[T_{11}, T_{22}, T_{33}, \mu],$$

and the vanishing of the divergence of T (see Equation (5.6)) is given by

$$aT_{11} + fT_{22} + pT_{33} - (a+f+p)\mu = 0,$$

and the system of polynomial equations $\{\mathfrak{P}_{ijk} = 0\}$ is now determined by

$$\mathfrak{P}_{412} = b(T_{11} - T_{22}) = 0,$$

$$\mathfrak{P}_{413} = c(T_{11} - T_{33}) = 0,$$

$$\mathfrak{P}_{423} = h(T_{22} - T_{33}) = 0.$$

This proves Assertion (ii).

Case 2.2: $T_{14} = T_{24} = T_{34} = 0$, $T_{44} \neq \mu$ and a + f + p = 0.

By Lemma 5.18-(b) we can take a = 0 working, if necessary, with an isomorphically isometric metric. Hence, $p = -f \neq 0$. Now, we have the following linear combinations

$$\begin{aligned} \mathfrak{P}_{421} - \mathfrak{P}_{412} &= -2 \left(f T_{12} + b (T_{44} - \mu) \right), \\ \mathfrak{P}_{431} - \mathfrak{P}_{413} &= 2 \left(f T_{13} - c (T_{44} - \mu) \right), \\ \mathfrak{P}_{432} - \mathfrak{P}_{423} &= 2 \left(2 f T_{23} - h (T_{44} - \mu) \right), \end{aligned}$$

from where we obtain

$$T_{12} = -\frac{b}{f}(T_{44} - \mu), \quad T_{13} = \frac{c}{f}(T_{44} - \mu), \quad T_{23} = \frac{h}{2f}(T_{44} - \mu).$$

Using these expressions, the vanishing of the divergence of the tensor T given by Equation (5.6) reduces to

$$f(T_{22} - T_{33}) = 0.$$

Besides,

$$\mathfrak{P}_{411} = \frac{2}{f}(b^2 - c^2)(T_{44} - \mu).$$

As a consequence

$$T_{22} = T_{33}, \quad c = \varepsilon b, \quad \text{with } \varepsilon^2 = 1$$

and a direct calculation shows that the system of polynomial equations $\{\mathfrak{P}_{ijk}=0\}$ is given by

$$\mathfrak{P}_{412} = \varepsilon \mathfrak{P}_{413} = \mathfrak{P}_{421} = \varepsilon \mathfrak{P}_{431} = \frac{1}{2f} b \left(2f(T_{11} - T_{22}) - 3\varepsilon h(T_{44} - \mu) \right) = 0$$
$$\mathfrak{P}_{422} = -\mathfrak{P}_{433} = -\frac{1}{f} (2b^2 - f^2 + h^2)(T_{44} - \mu) = 0.$$

The associated left-invariant metric is given by

$$[e_1, e_4] = be_2 + \varepsilon be_3,$$

$$[e_2, e_4] = -be_1 + fe_2 + he_3,$$

$$[e_3, e_4] = -\varepsilon be_1 - he_2 - fe_3,$$

and the isometry $e_3 \mapsto -e_3$ interchanges (ε, f, b, h) and $(-\varepsilon, f, b, -h)$, so we can take $\varepsilon = 1$ working with an isomorphically isometric metric if necessary. Thus Assertion *(iii.a)* is obtained.

Case 2.3: Some of T_{14} , T_{24} and T_{34} does not vanish.

In this case, according to Lemma 5.18-(a), we can assume a = b = c = 0 and work, if necessary, with an isomorphically isometric metric. In this situation

$$\mathfrak{P}_{212} = -fT_{14}, \quad \mathfrak{P}_{313} = -pT_{14}$$

and, since $f \neq p$, it follows that

$$T_{14} = 0. (5.10)$$

Now, we compute

$$\mathfrak{P}_{322} = hT_{24} + fT_{34}, \quad \mathfrak{P}_{323} = -pT_{24} + hT_{34},$$
$$\mathfrak{P}_{424} = -fT_{24} - hT_{34}, \quad \mathfrak{P}_{434} = hT_{24} - pT_{34}.$$

Since $T_{24} = T_{34} = 0$ is not possible, we have $f^2 = p^2 = h^2$. Moreover, $f \neq p$ and $fp \neq 0$ lead to

$$p = -f \neq 0, \quad h = \varepsilon f, \quad T_{34} = -\varepsilon T_{24} \neq 0,$$
(5.11)

where $\varepsilon^2 = 1$.

Using Equations (5.10) and (5.11) the conditions given in Equation (5.6) for the vanishing of the divergence of T reduce to

$$f(T_{22} - T_{33}) = 0,$$

and

$$\mathfrak{P}_{412} = f(T_{12} - \varepsilon T_{13}), \quad \mathfrak{P}_{421} = -f(T_{12} + \varepsilon T_{13}), \quad \mathfrak{P}_{433} = f(2\varepsilon T_{23} - T_{44} + \mu).$$

Therefore,

$$T_{22} = T_{33}, \quad T_{12} = T_{13} = 0, \quad T_{23} = \frac{\varepsilon}{2}(T_{44} - \mu),$$
 (5.12)

and we obtain an algebraic T-soliton with associated left-invariant metric given by

$$[e_2, e_4] = fe_2 + \varepsilon fe_3, \quad [e_3, e_4] = -\varepsilon fe_2 - fe_3.$$

Note that the isometry $e_3 \mapsto -e_3$ interchanges ε and $-\varepsilon$. Thus, working with an isomorphically isometric metric if necessary, we can take $\varepsilon = 1$. Now Equations (5.10), (5.11) and (5.12) lead to the proof of Assertion (*iii.b*).

In the previous result we determined the algebraic T-solitons in the non-Einstein case. Even though the Einstein case can usually be handled directly when dealing with particular tensor fields, we include this case in the following result for the sake of completeness.

Theorem 5.20. If $\mathbb{R}^3 \rtimes \mathbb{R}$ is Einstein, then it is a non-trivial algebraic *T*-soliton with soliton constant μ if and only if it is isomorphically isometric to a metric with a = f = p and

$$T = \begin{pmatrix} T_{11} & T_{12} & T_{13} & 0 \\ * & T_{22} & T_{23} & 0 \\ * & * & T_{33} & 0 \\ * & * & * & \mu \end{pmatrix} \neq \mu \langle \cdot, \cdot \rangle,$$

with

$$a(T_{11} + T_{22} + T_{33} - 3\mu) = 0, \qquad b(T_{11} - T_{22}) - hT_{13} - cT_{23} = 0,$$

$$bT_{12} + cT_{13} = 0, \qquad c(T_{11} - T_{33}) + hT_{12} - bT_{23} = 0,$$

$$cT_{13} + hT_{23} = 0, \qquad h(T_{22} - T_{33}) + cT_{12} + bT_{13} = 0.$$

Proof. Recall that $\mathbb{R}^3 \rtimes \mathbb{R}$ is Einstein if and only if a = f = p (see Remark 5.16) so, according to Lemma 5.18-(*a*), $T_{14} = T_{24} = T_{34} = 0$. Moreover, since

$$\mathfrak{P}_{411} + \mathfrak{P}_{422} + \mathfrak{P}_{433} = 3a(T_{44} - \mu),$$

then either $T_{44} = \mu$ or a = 0.

If $T_{44} = \mu$, the conditions for the divergence of T to vanish given by Equation (5.6) reduce to

$$a(T_{11} + T_{22} + T_{33} - 3\mu) = 0,$$

and a direct calculation shows that the system of polynomial equations $\{\mathfrak{P}_{ijk} = 0\}$ is determined by

$$\mathfrak{P}_{411} = -2(bT_{12} + cT_{13}) = 0,$$

$$\mathfrak{P}_{433} = 2(cT_{13} + hT_{23}) = 0,$$

$$\mathfrak{P}_{412} = b(T_{11} - T_{22}) - hT_{13} - cT_{23} = 0$$

$$\mathfrak{P}_{413} = c(T_{11} - T_{33}) + hT_{12} - bT_{23} = 0$$

$$\mathfrak{P}_{423} = h(T_{22} - T_{33}) + cT_{12} + bT_{13} = 0$$

Assume now that $T_{44} \neq \mu$ and a = 0. Since either b, c or h must be non-zero, Remark 5.17 implies that we can assume that $b \neq 0$ (working, if necessary, with an isomorphically isometric metric). Now, a direct calculation shows that

$$\mathfrak{P}_{412} - \mathfrak{P}_{421} = 2b(T_{44} - \mu),$$

which implies that this case cannot occur. This proves the result.

5.3.2 Algebraic Ricci solitons on $\mathbb{R}^3 \rtimes \mathbb{R}$

Since Einstein metrics are trivially Ricci solitons, we exclude them from our study and part from the conditions in Theorem 5.19. First of all, a direct calculation shows that the Ricci tensor is determined by

$$\rho_{11} = -(a + f + p)a, \qquad \rho_{23} = (f - p)h,$$

$$\rho_{12} = (a - f)b, \qquad \rho_{33} = -(a + f + p)p,$$

$$\rho_{13} = (a - p)c, \qquad \rho_{44} = -a^2 - f^2 - p^2,$$

$$\rho_{22} = -(a + f + p)f.$$

In the locally symmetric case we analyse the two non-Einstein cases given in Remark 5.16. If a = f = c = h = 0 and $p \neq 0$, clearly we have

$$\operatorname{Ric} = -p^2 \operatorname{diag}[0, 0, 1, 1],$$

while if $a = f \neq 0$ and p = c = h = 0, we get that

$$Ric = -2a^2 \operatorname{diag}[1, 1, 0, 1].$$

Theorem 5.19-(i) guarantees that both are non-trivial algebraic Ricci solitons, and it is immediate to see that their soliton constants are $\mu = -p^2$ and $\mu = -2a^2$, respectively. Setting p = 1 in the

former case and a = 1 in the latter, we remain in the same homothetic classes and the associated left-invariant metrics are given by

$$[e_1, e_4] = be_2, \quad [e_2, e_4] = -be_1, \quad [e_3, e_4] = e_3,$$
 (5.13)

and

$$[e_1, e_4] = e_1 + be_2, \quad [e_2, e_4] = -be_1 + e_2,$$

respectively. A straightforward calculation shows that the sectional curvatures of the metrics above are independent of the structure constant b. Therefore, these metrics are homothetic – although not isomorphically homothetic – to the corresponding metrics with b = 0 (see [96]), which are the metrics in Assertions (*vi*) and (*vii*) in Theorem 5.4.

In the non-symmetric case we analyse every possibility in Theorem 5.19. Note that with no further assumption, $\rho_{14} = \rho_{24} = \rho_{34} = 0$, so Theorem 5.19-(*iii.b*) cannot occur.

Case (i).

We set $a = f \neq p$. Since algebraic Ricci solitons must satisfy the conditions

$$\rho_{13} = (a-p)c = 0 \text{ and } \rho_{23} = (a-p)h = 0,$$

then c = h = 0. Now note that $\rho_{12} = 0$, so the algebraic Ricci solitons are determined by the vanishing of the polynomials

$$\mathfrak{Q}_1 = b(\rho_{11} - \rho_{22}), \quad \mathfrak{Q}_2 = \rho_{44} - \mu, \quad \mathfrak{Q}_3 = a(\rho_{11} + \rho_{22}) + p\rho_{33} - (2a+p)\mu.$$

A direct calculation leads to

$$\mathfrak{Q}_1 = 0$$
, $\mathfrak{Q}_2 = -\mu - 2a^2 - p^2$, $\mathfrak{Q}_3 = -(2a+p)(\mu + 2a^2 + p^2)$

Thus, the system $\{\mathfrak{Q}_i = 0\}$ is equivalent to $\mu = -2a^2 - p^2$, which provides an algebraic Ricci soliton with associated left-invariant metric given by

$$[e_1, e_4] = ae_1 + be_2, \quad [e_2, e_4] = -be_1 + ae_2, \quad [e_3, e_4] = pe_3.$$
 (5.14)

It is not difficult to see that the sectional curvatures of these metrics do not depend on the structure constant b, so the metrics above are homothetic – although not isomorphically homothetic – to the corresponding metrics with b = 0 (see [96]). Moreover, the space is locally symmetric if and only if a = 0 or p = 0. Hence, working in the same homothetic class, we can assume a = 1 and b = 0, and these metrics are covered by Theorem 5.4-(*iii*) (see the Case (*ii*) for the restrictions on the parameter).

Case (ii).

In this case, $a \neq f \neq p$ and $a \neq p$. The Ricci operator must be diagonal, which occurs if and only if

$$\rho_{12} = (a - f)b = 0, \quad \rho_{13} = (a - p)c = 0 \quad \text{and} \quad \rho_{23} = (f - p)h = 0,$$

i.e., b = c = h = 0. Thus, we obtain an algebraic Ricci soliton if the polynomials

$$\mathfrak{Q}_1 = \rho_{44} - \mu, \quad \mathfrak{Q}_2 = a\rho_{11} + f\rho_{22} + p\rho_{33} - (a+f+p)\mu_{53}$$

vanish. A direct calculation shows that

$$\mathfrak{Q}_1 = -\mu - a^2 - f^2 - p^2, \quad \mathfrak{Q}_2 = -(a+f+p)(\mu + a^2 + f^2 + p^2),$$

and $\mu = -a^2 - f^2 - p^2$ provides an algebraic Ricci soliton with associated left-invariant metric given by

$$[e_1, e_4] = ae_1, \quad [e_2, e_4] = fe_2, \quad [e_3, e_4] = pe_3.$$
 (5.15)

Since b = c = h = 0, necessarily some of a, f and p must be non-zero. Working in the same homothetic class, we can assume that a = 1 (see Remark 5.17). Now, the isometry

$$e_2 \longmapsto e_3$$

interchanges (f, p) with (p, f), while the isometry

$$(e_1, e_2, e_3, e_4) \longmapsto (e_2, e_1, e_3, -e_4)$$

interchanges (-1, p) with (-1, -p). Besides, if $f \neq 0$, the homothety

$$(e_1, e_2, e_3, e_4) \longmapsto \frac{1}{f}(e_2, e_1, e_3, e_4)$$

interchanges (f, p) with $(\frac{1}{f}, \frac{p}{f})$ and, if $p \neq 0$, the homothety

$$(e_1, e_2, e_3, e_4) \longmapsto \frac{1}{p}(e_3, e_2, e_1, e_4)$$

interchanges (f, p) with $(\frac{f}{p}, \frac{1}{p})$. The relations above allow us to show that any metric in this case is homothetic to a metric in Theorem 5.4-(*iii*). Since $(f, p) \sim (p, f)$, we restrict the domain of (f, p) to $f \leq p$. Now, using the identification $(f, p) \sim (\frac{1}{f}, \frac{p}{f})$, we remove the set

$$\{(f, p) \colon f \le p, |f| > 1\}$$

since every homothetic class has a representative with $|f| \leq 1$. Now, considering $(f, p) \sim (\frac{f}{p}, \frac{1}{p})$ we can remove the set

$$\{(f, p) \colon f < p, |f| \le 1, p > 1\}$$

so that the domain of (f, p) reduces to

$$\{(f, p): -1 \le f \le p \le 1\}$$

Finally, since $(-1, p) \sim (-1, -p)$ we can also eliminate the segment

$$\{(-1,p): p < 0\}.$$

The particular points (0,0), (0,1) and (1,1) are also excluded because the space is locally symmetric in those cases. Hence, we conclude that this case corresponds to Assertion *(iii)* in Theorem 5.4.

Case (*iii.a*).

In this case, the structure constants satisfy a = 0, $p = -f \neq 0$, c = b and

$$2b^2 - f^2 + h^2 = 0.$$

Algebraic Ricci solitons are characterized by the system of polynomial equations $\{\mathfrak{Q}_i = 0\}$, where

$$\begin{aligned} \mathfrak{Q}_1 &= \rho_{22} - \rho_{33}, \quad \mathfrak{Q}_2 &= \rho_{12} + \frac{b}{f}(\rho_{44} - \mu), \quad \mathfrak{Q}_3 &= \rho_{12} + \rho_{13}, \\ \mathfrak{Q}_4 &= \rho_{23} - \frac{h}{2f}(\rho_{44} - \mu), \quad \mathfrak{Q}_5 &= b\left(2f(\rho_{11} - \rho_{22}) - 3h(\rho_{44} - \mu)\right), \end{aligned}$$

with $\rho_{44} \neq \mu$ and $2b^2 - f^2 + h^2 = 0$. A direct calculation leads to

$$\begin{split} \mathfrak{Q}_1 &= 0, \quad \mathfrak{Q}_2 = -\frac{b}{f}(\mu + 3f^2), \quad \mathfrak{Q}_3 = 0, \\ \mathfrak{Q}_4 &= \frac{h}{2f}(\mu + 6f^2), \quad \mathfrak{Q}_5 = 3bh(\mu + 2f^2), \end{split}$$

where $\mu \neq -2f^2$ and $2b^2 - f^2 + h^2 = 0$. Since $f \neq 0$, the cases b = h = 0 and $bh \neq 0$ are not possible. Next we analyse the cases b = 0, $h \neq 0$ and h = 0, $b \neq 0$ separately.

If b = 0 and $h \neq 0$, then $h = \varepsilon f$, with $\varepsilon^2 = 1$, and the system $\{\mathfrak{Q}_i = 0\}$ reduces to $\mu = -6f^2$. We take f = 1 working in the same homothetic class and the left-invariant metric associated to the algebraic Ricci soliton is given by

$$[e_2, e_4] = e_2 + \varepsilon e_3, \quad [e_3, e_4] = -\varepsilon e_2 - e_3.$$
 (5.16)

The isometry $e_3 \mapsto -e_3$ interchanges ε and $-\varepsilon$. Hence, we can set $\varepsilon = 1$ and Assertion (i) in Theorem 5.4 follows.

If h = 0 and $b \neq 0$ then we have $b = \frac{\varepsilon}{\sqrt{2}}f$, with $\varepsilon^2 = 1$, and the algebraic Ricci solitons are determined by $\mu = -3f^2$. As in the previous case, we can take f = 1 so that the associated left-invariant metric is given by

$$[e_1, e_4] = \frac{\varepsilon}{\sqrt{2}}(e_2 + e_3), \quad [e_2, e_4] = -\frac{\varepsilon}{\sqrt{2}}e_1 + e_2, \quad [e_3, e_4] = -\frac{\varepsilon}{\sqrt{2}}e_1 - e_3.$$
(5.17)

The isometry $(e_1, e_2, e_3, e_4) \mapsto (e_1, -e_2, -e_3, e_4)$ interchanges ε with $-\varepsilon$, so we can set $\varepsilon = 1$ and this case corresponds to Assertion (*ii*) in Theorem 5.4.

5.3.3 Algebraic Bach solitons on $\mathbb{R}^3 \rtimes \mathbb{R}$

Einstein metrics are Bach-flat in dimension four and so they are steady Bach solitons. Therefore, we focus on the non-Einstein situation and use Theorem 5.19 to determine all the non-trivial algebraic Bach solitons. A long but straightforward calculation shows that the non-zero components \mathfrak{B}_{ij} of the Bach tensor of $\mathbb{R}^3 \rtimes \mathbb{R}$ are given by

$$\begin{split} & 6\mathfrak{B}_{11} = a^4 - f^4 - p^4 - a^3f + 3af^3 - a^3p + 3ap^3 + f^3p + fp^3 - 2a^2f^2 - 2a^2p^2 \\ & -7a^2fp + 3af^2p + 3afp^2 + 9(b^2 + c^2)a^2 - 3(5b^2 + h^2)f^2 - 3(5c^2 + h^2)p^2 \\ & + 6(b^2 + 2c^2)af + 6(2b^2 + c^2)ap - 6(2b^2 + 2c^2 - h^2)fp + 18bch(f - p), \end{split}$$

$$\begin{split} 6\mathfrak{B}_{12} &= -b\left(a-f\right)\left(2a^2+2f^2-p^2-16af-10ap-10fp\right)\\ &+ 6ch\left(a+f-2p\right)\left(a+f+p\right)-3b\left((4b^2+4c^2+h^2)a-(4b^2+c^2+4h^2)f\right)\\ &\quad -3(c^2-h^2)p\right), \end{split}$$

$$\begin{split} 6\mathfrak{B}_{13} &= -c\left(a-p\right)\left(2a^2-f^2+2p^2-10af-16ap-10fp\right)\\ &\quad -6bh\left(a-2f+p\right)\left(a+f+p\right)-3c\left((4b^2+4c^2+h^2)a-3(b^2-h^2)f\right)\\ &\quad -(b^2+4c^2+4h^2)p\right), \end{split}$$

$$\begin{split} & 6\mathfrak{B}_{22} = -a^4 + f^4 - p^4 + 3a^3f - af^3 + a^3p + ap^3 - f^3p + 3fp^3 - 2a^2f^2 - 2f^2p^2 \\ & + 3a^2fp - 7af^2p + 3afp^2 - 3(5b^2 + c^2)a^2 + 9(b^2 + h^2)f^2 - 3(c^2 + 5h^2)p^2 \\ & + 6(b^2 + 2h^2)af - 6(2b^2 - c^2 + 2h^2)ap + 6(2b^2 + h^2)fp - 18bch(a - p), \end{split}$$

$$\begin{split} 6\mathfrak{B}_{23} &= h\left(f-p\right)\left(a^2-2f^2-2p^2+10af+10ap+16fp\right)\\ &\quad - 6bc\left(2a-f-p\right)\left(a+f+p\right)+3h\left(3(b^2-c^2)a-(4b^2+c^2+4h^2)f\right)\\ &\quad + (b^2+4c^2+4h^2)p\right), \end{split}$$

$$\begin{split} & 6\mathfrak{B}_{33} = -a^4 - f^4 + p^4 + a^3f + af^3 + 3a^3p - ap^3 + 3f^3p - fp^3 - 2a^2p^2 - 2f^2p^2 \\ & + 3a^2fp + 3af^2p - 7afp^2 - 3(b^2 + 5c^2)a^2 - 3(b^2 + 5h^2)f^2 + 9(c^2 + h^2)p^2 \\ & + 6(b^2 - 2c^2 - 2h^2)af + 6(c^2 + 2h^2)ap + 6(2c^2 + h^2)fp + 18bch(a - f). \end{split}$$

Recall that the Bach tensor is trace-free, so $\mathfrak{B}_{44} = -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33}$. In the locally symmetric case, analysing the cases in Remark 5.16, a direct calculation shows that the only case which is not Bach-flat corresponds to

$$a = f = c = h = 0, \quad p \neq 0$$

and

$$\widehat{\mathfrak{B}} = \frac{p^4}{6} \operatorname{diag}[-1, -1, 1, 1].$$

This case gives a non-trivial algebraic Bach soliton with soliton constant $\mu = \frac{p^4}{6}$ (see Theorem 5.19-(*i*)) which is also an algebraic Ricci soliton (see Equation (5.13)).

If the space is not locally symmetric, we need to analyse the four cases obtained in Theorem 5.19 separately. Note that, with no further assumption,

$$\mathfrak{B}_{14}=\mathfrak{B}_{24}=\mathfrak{B}_{34}=0,$$

so Case (iii.b) in Theorem 5.19 is not possible.

Case (i).

Take $a = f \neq p$. The conditions that determine an algebraic Bach soliton can be expressed as the vanishing of the polynomials

$$\begin{aligned} \mathfrak{Q}_{1} &= \mathfrak{B}_{13}, \quad \mathfrak{Q}_{2} &= \mathfrak{B}_{23}, \quad \mathfrak{Q}_{3} &= b\mathfrak{B}_{12}, \quad \mathfrak{Q}_{4} &= b(\mathfrak{B}_{11} - \mathfrak{B}_{22}), \\ \mathfrak{Q}_{5} &= c(\mathfrak{B}_{11} - \mathfrak{B}_{33}) + h\mathfrak{B}_{12}, \quad \mathfrak{Q}_{6} &= h(\mathfrak{B}_{22} - \mathfrak{B}_{33}) + c\mathfrak{B}_{12}, \\ \mathfrak{Q}_{7} &= -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33} - \mu, \\ \mathfrak{Q}_{8} &= a(\mathfrak{B}_{11} + \mathfrak{B}_{22}) + p\mathfrak{B}_{33} - (2a + p)\mu, \end{aligned}$$

and a straightforward calculation shows that

$$\begin{split} \mathfrak{Q}_{1} &= \frac{1}{6}(a-p)\left(c(9a^{2}-2p^{2}+26ap)+6bh(2a+p)-3c(b^{2}+4c^{2}+4h^{2})\right),\\ \mathfrak{Q}_{2} &= \frac{1}{6}(a-p)\left(h(9a^{2}-2p^{2}+26ap)-6bc(2a+p)-3h(b^{2}+4c^{2}+4h^{2})\right),\\ \mathfrak{Q}_{3} &= \frac{1}{2}b(a-p)\left(4ch(2a+p)-3b(c^{2}-h^{2})\right),\\ \mathfrak{Q}_{4} &= 2b(a-p)\left((c^{2}-h^{2})(2a+p)+3bch\right),\\ \mathfrak{Q}_{5} &= \frac{1}{6}(a-p)\left(2c(2a+p)(p^{2}-4ap+12c^{2}+12h^{2})+9bh(c^{2}+h^{2})\right),\\ \mathfrak{Q}_{6} &= \frac{1}{6}(a-p)\left(2h(2a+p)(p^{2}-4ap+12c^{2}+12h^{2})-9bc(c^{2}+h^{2})\right),\\ \mathfrak{Q}_{7} &= \frac{1}{6}(a-p)^{2}\left(p^{2}-4ap+9c^{2}+9h^{2}\right)-\mu,\\ \mathfrak{Q}_{8} &= \left(2a+p\right)\left(\frac{1}{6}(a-p)^{2}(p^{2}-4ap+9c^{2}+9h^{2})-\mu\right). \end{split}$$

Now, it is easy to see that

$$h\mathfrak{Q}_1 - c\mathfrak{Q}_2 = b(2a+p)(c^2+h^2)(a-p).$$

Next we show that c = h = 0 by analysing the cases p = -2a and b = 0. If p = -2a, which implies $a \neq 0$ since $a \neq p$, we obtain

$$\mathfrak{Q}_1 = -\frac{3}{2}ac(17a^2 + b^2 + 4c^2 + 4h^2), \quad \mathfrak{Q}_2 = -\frac{3}{2}ah(17a^2 + b^2 + 4c^2 + 4h^2),$$

so c = h = 0. If $p \neq -2a$ and b = 0 a straightforward calculation shows that

$$\mathfrak{Q}_1 = \frac{1}{6}c(a-p)(9a^2 - 2p^2 + 26ap - 12c^2 - 12h^2),$$

$$\mathfrak{Q}_2 = \frac{1}{6}h(a-p)(9a^2 - 2p^2 + 26ap - 12c^2 - 12h^2),$$

and

$$\mathfrak{Q}_5 = \frac{1}{3}c(a-p)(2a+p)(p^2 - 4ap + 12c^2 + 12h^2),$$

$$\mathfrak{Q}_6 = \frac{1}{3}h(a-p)(2a+p)(p^2 - 4ap + 12c^2 + 12h^2),$$

from where it follows that

$$4(2a+p)\mathfrak{Q}_1 + 13\mathfrak{Q}_5 = c(a-p)(2a+p)(6a^2+3p^2+44c^2+44h^2),$$

$$4(2a+p)\mathfrak{Q}_2 + 13\mathfrak{Q}_6 = h(a-p)(2a+p)(6a^2+3p^2+44c^2+44h^2).$$

Hence, again c = h = 0. Assuming the vanishing of c and h the system $\{Q_i = 0\}$ reduces to

$$\mathfrak{Q}_7 = \frac{1}{6}(a-p)^2(p-4a)p - \mu,$$

$$\mathfrak{Q}_8 = (2a+p)(\frac{1}{6}(a-p)^2(p-4a)p - \mu),$$

or equivalently,

$$\mu = \frac{1}{6}(a-p)^2(p-4a)p.$$

This leads to an algebraic Bach soliton with associated left-invariant metric given by

$$[e_1, e_4] = ae_1 + be_2, \quad [e_2, e_4] = -be_1 + ae_2, \quad [e_3, e_4] = pe_3,$$

which is locally symmetric if and only if a = 0 or p = 0, and it is Bach-flat if and only if p = 0 or p = 4a. Note that, in any case, it is also an algebraic Ricci soliton (see Equation (5.14)).

Case (ii).

If $a \neq f \neq p$, $a \neq p$ and the Bach operator is diagonal, the algebraic Bach solitons are determined by the vanishing of the polynomials

$$\begin{aligned} \mathfrak{Q}_1 &= \mathfrak{B}_{12}, \quad \mathfrak{Q}_2 &= \mathfrak{B}_{13}, \quad \mathfrak{Q}_3 &= \mathfrak{B}_{23}, \\ \mathfrak{Q}_4 &= b(\mathfrak{B}_{11} - \mathfrak{B}_{22}), \quad \mathfrak{Q}_5 &= c(\mathfrak{B}_{11} - \mathfrak{B}_{33}), \\ \mathfrak{Q}_6 &= h(\mathfrak{B}_{22} - \mathfrak{B}_{33}), \quad \mathfrak{Q}_7 &= -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33} - \mu, \\ \mathfrak{Q}_8 &= a\mathfrak{B}_{11} + f\mathfrak{B}_{22} + p\mathfrak{B}_{33} - (a + f + p)\mu. \end{aligned}$$

$$(5.18)$$

Next we split the analysis in several cases, depending on whether bch vanishes or not. According to Remark 5.17, it is enough to consider the case b = c = h = 0, the case b = c = 0, $h \neq 0$ and the case $ch \neq 0$. We will show that there exist algebraic Bach solitons only in the first case.

<u>Case 1. b = c = h = 0.</u>

If b = c = h = 0, a direct calculation shows that the Bach operator is diagonal and that Equation (5.18) takes the form

$$\mathfrak{Q}_7 = \frac{1}{6} \left(a^2 + (f-p)^2 - 2a(f+p) \right) \left(a^2 + f^2 + p^2 - af - ap - fp \right) - \mu,$$

$$\mathfrak{Q}_8 = (a+f+p)\mathfrak{Q}_7.$$

Therefore, algebraic Bach solitons are determined by

$$\mu = \frac{1}{6} \left(a^2 + (f - p)^2 - 2a(f + p) \right) \left(a^2 + f^2 + p^2 - af - ap - fp \right).$$

The associated left-invariant metric is given by

$$[e_1, e_4] = ae_1, \quad [e_2, e_4] = fe_2, \quad [e_3, e_4] = pe_3,$$

so these algebraic Bach solitons are also algebraic Ricci solitons (see Equation (5.15)).

Case 2. $b = c = 0, h \neq 0$.

In this case we work with the polynomials \mathfrak{Q}_3 and \mathfrak{Q}_6 in Equation (5.18) which are given by

$$\mathfrak{Q}_3 = \frac{1}{6}h(f-p)\left(a^2 - 2(f-p)^2 + 10a(f+p) + 12fp - 12h^2\right),$$

$$\mathfrak{Q}_6 = \frac{1}{3}h(f-p)(a+f+p)\left(a^2 + (f-p)^2 - 2a(f+p) + 12h^2\right)$$

If a + f + p = 0 we take p = -a - f and

$$\mathfrak{Q}_3 = -\frac{1}{6}h(a+2f)\left(10(a+f)^2 + a^2 + 10f^2 + 12h^2\right) \neq 0,$$

while if $a + f + p \neq 0$ we calculate

$$2(a+f+p)\mathfrak{Q}_3 + 5\mathfrak{Q}_6 = h(f-p)(a+f+p)\left(2a^2 + (f+p)^2 + 16h^2\right) \neq 0.$$

Since none of these two polynomials above vanishes, we conclude that no algebraic Bach soliton may exist in this case.

Case 3. $ch \neq 0$.

Next we show that non-trivial algebraic Bach solitons cannot exist when $ch \neq 0$. To do so, we consider the polynomials

$$\begin{split} \mathfrak{Q}_5 &= c(\mathfrak{B}_{11} - \mathfrak{B}_{33}), \quad \mathfrak{Q}_6 &= h(\mathfrak{B}_{22} - \mathfrak{B}_{33}), \\ \mathfrak{Q}_7 &= -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33} - \mu, \\ \mathfrak{Q}_8 &= a\mathfrak{B}_{11} + f\mathfrak{B}_{22} + p\mathfrak{B}_{33} - (a + f + p)\mu, \end{split}$$

given in Equation (5.18). It is easy to see that

$$\mathfrak{B}_{11} = \mathfrak{B}_{22} = \mathfrak{B}_{33}, \quad \mu = -3\mathfrak{B}_{11}, \quad (a+f+p)(\mathfrak{B}_{11}-\mu) = 0.$$

Hence, $(a + f + p)\mathfrak{B}_{11} = 0$. Since the Bach operator must be diagonal, if $\mathfrak{B}_{11} = 0$ then any possible algebraic Bach soliton would be trivial. Thus, p = -a - f and a direct calculation shows that the polynomials \mathfrak{Q}_2 and \mathfrak{Q}_3 in Equation (5.18) are given by

$$\begin{aligned} \mathfrak{Q}_2 &= -\frac{1}{6}c\left((2a+f)(20a^2+11f^2+20af)\right) \\ &\quad +3(5b^2+8c^2+5h^2)a-3(2b^2-4c^2-7h^2)f\right), \\ \mathfrak{Q}_3 &= -\frac{1}{6}h\left((a+2f)(11a^2+20f^2+20af)\right) \\ &\quad -3(2b^2-7c^2-4h^2)a+3(5b^2+5c^2+8h^2)f\right), \end{aligned}$$

from where it immediately follows that

$$h\mathfrak{Q}_2 - c\mathfrak{Q}_3 = -\frac{1}{12}ch(a-f)\left(47(a+f)^2 + 11(a^2+f^2) + 6(7b^2+c^2+h^2)\right) \neq 0.$$

This implies that non-trivial algebraic Bach solitons cannot exist if $ch \neq 0$.

Case (iii.a).

In this case, the structure constants satisfy a = 0, $p = -f \neq 0$, c = b and

$$2b^2 - f^2 + h^2 = 0$$

In this situation, the algebraic Bach solitons are determined by the system of polynomial equations $\{\mathfrak{Q}_i = 0\}$, where

$$\begin{split} \mathfrak{Q}_1 &= \mathfrak{B}_{22} - \mathfrak{B}_{33}, \quad \mathfrak{Q}_2 &= \mathfrak{B}_{12} - \frac{b}{f}(\mathfrak{B}_{11} + \mathfrak{B}_{22} + \mathfrak{B}_{33} + \mu), \\ \mathfrak{Q}_3 &= \mathfrak{B}_{12} + \mathfrak{B}_{13}, \quad \mathfrak{Q}_4 &= \mathfrak{B}_{23} + \frac{h}{2f}(\mathfrak{B}_{11} + \mathfrak{B}_{22} + \mathfrak{B}_{33} + \mu), \\ \mathfrak{Q}_5 &= b\left((2f + 3h)\mathfrak{B}_{11} - (2f - 3h)\mathfrak{B}_{22} + 3h\mathfrak{B}_{33} + 3h\mu\right), \end{split}$$

with $\mu \neq -(\mathfrak{B}_{11} + \mathfrak{B}_{22} + \mathfrak{B}_{33})$ and $h^2 = f^2 - 2b^2$. A long but direct calculation leads to $\mathfrak{Q}_1 = \mathfrak{Q}_3 = 0$ and

$$\begin{split} \mathfrak{Q}_2 &= \quad \frac{b}{3f} \left(40f^4 - 45b^2f^2 - 3\mu \right), \\ \mathfrak{Q}_4 &= -\frac{h}{6f} \left(88f^4 - 45b^2f^2 - 3\mu \right), \\ \mathfrak{Q}_5 &= -3bh \left(8f^4 - 15b^2f^2 - \mu \right), \end{split}$$

where $\mu \neq 8f^4 - 9b^2f^2$ and $h^2 = f^2 - 2b^2$. Note that

$$3fh\mathfrak{Q}_2 + \mathfrak{Q}_5 = 16bhf^4$$

Hence, since $f \neq 0$, it follows that the cases b = h = 0 and $bh \neq 0$ are not possible. In what follows we analyse the cases b = 0, $h \neq 0$ and h = 0, $b \neq 0$ separately.

If b = 0 and $h \neq 0$, then $h = \varepsilon f$, with $\varepsilon^2 = 1$, and the system of polynomial equations $\{\mathfrak{Q}_i = 0\}$ reduces to

$$\mathfrak{Q}_4 = \frac{\varepsilon}{6}(3\mu - 88f^4) = 0.$$

Thus, $\mu = \frac{88}{3}f^4$ and the left-invariant metric associated to the algebraic Bach soliton is given by

$$[e_2, e_4] = e_2 + \varepsilon e_3, \quad [e_3, e_4] = -\varepsilon e_2 - e_3,$$

where we have taken f = 1 – working in the same homothetic class. Note that these spaces are algebraic Ricci solitons (see Equation (5.16)).

Finally, assume that h = 0 and $b \neq 0$. In this case, $b = \frac{\varepsilon}{\sqrt{2}}f$, with $\varepsilon^2 = 1$, and the system $\{\mathfrak{Q}_i = 0\}$ leads to the soliton constant $\mu = \frac{35}{6}f^4$. As we did in the case just above, we can take f = 1 and the left-invariant metric associated to the algebraic Bach soliton is given by

$$[e_1, e_4] = \frac{\varepsilon}{\sqrt{2}}(e_2 + e_3), \quad [e_2, e_4] = -\frac{\varepsilon}{\sqrt{2}}e_1 + e_2, \quad [e_3, e_4] = -\frac{\varepsilon}{\sqrt{2}}e_1 - e_3,$$

which also determines an algebraic Ricci soliton (see Equation (5.17)).

5.4 Algebraic solitons on $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$

We follow the description of the left-invariant metrics on semi-direct extensions of the Poincaré and the Euclidean groups given in Section 1.4.2. Recall that there exists an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ such that

$$[e_1, e_3] = -\lambda_2 e_2, \qquad [e_2, e_3] = \lambda_1 e_1,$$

$$[e_1, e_4] = b e_1 - A \lambda_2 e_2, \qquad [e_2, e_4] = A \lambda_1 e_1 + b e_2,$$

$$[e_3, e_4] = C e_1 + D e_2,$$

(5.19)

where $\lambda_1 \lambda_2 \neq 0$. The associated Lie group to $\widetilde{E}(2) \times \mathbb{R}$ if λ_1 , λ_2 do not change sign, and corresponds to $E(1,1) \times \mathbb{R}$ otherwise.

Remark 5.21. $\widetilde{E}(2) \rtimes \mathbb{R}$ is locally symmetric if and only if C = D = 0 and $\lambda_2 = \lambda_1 - in$ which case the metric is flat if b = 0, and homothetic to the product $\mathbb{H}^3 \times \mathbb{R}$ otherwise. Analogously, $E(1,1) \rtimes \mathbb{R}$ is locally symmetric if and only if C = D = 0, $\lambda_2 = -\lambda_1$ and $b^2 = (A^2 + 1)\lambda_1^2 - in$ which case it is isometric to the product of two surfaces of constant distinct negative curvatures if $A \neq 0$, and homothetic to $\mathbb{H}^2 \times \mathbb{H}^2$ otherwise. We show in Remark 5.24 that these metrics are also realized as left-invariant metrics on $\mathbb{R}^3 \rtimes \mathbb{R}$.

Remark 5.22. The parameters $(\lambda_1, \lambda_2, A, b, C, D)$ in Equation (5.19) can be transformed into $(\lambda_2, \lambda_1, A, b, -D, C)$ through the isometry

$$(e_1, e_2, e_3, e_4) \mapsto (-e_2, e_1, e_3, e_4).$$

Therefore, any left-invariant metric (5.19) with D = 0 is isomorphically isometric to a left-invariant metric with C = 0.

5.4.1 Algebraic *T*-solitons on $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$

Theorem 5.23. $E(1,1) \rtimes \mathbb{R}$ or $\widetilde{E}(2) \rtimes \mathbb{R}$ are non-trivial algebraic *T*-solitons with soliton constant μ if and only if

$$T = \begin{pmatrix} T_{11} & 0 & 0 & 0 \\ * & T_{11} & 0 & 0 \\ * & * & A^2(T_{44} - \mu) + \mu & -A(T_{44} - \mu) \\ * & * & * & T_{44} \end{pmatrix} \neq \mu \langle \cdot, \cdot \rangle$$

and b = 0, with C = D = 0 or $T_{11} = A^2(T_{44} - \mu) + T_{44}$.

Remark 5.24. In the case where b = 0 and C = D = 0, the left-invariant metric associated to an algebraic *T*-soliton takes the form

$$[e_1, e_3] = -\lambda_2 e_2, \quad [e_1, e_4] = -A\lambda_2 e_2, \quad [e_2, e_3] = \lambda_1 e_1, \quad [e_2, e_4] = A\lambda_1 e_1.$$

Setting $\zeta = \sqrt{A^2 + 1}$ and considering the orthonormal basis given by

$$\tilde{e}_1 = \frac{1}{\sqrt{2}}(e_1 + e_2), \qquad \tilde{e}_2 = -\frac{1}{\sqrt{2}}(e_1 - e_2),$$

 $\tilde{e}_3 = -\frac{1}{\zeta}(Ae_3 - e_4), \quad \tilde{e}_4 = \frac{1}{\zeta}(e_3 + Ae_4),$

the corresponding non-zero Lie brackets are now

$$[\tilde{e}_1, \tilde{e}_4] = \frac{\zeta}{2} (\lambda_1 - \lambda_2) \tilde{e}_1 - \frac{\zeta}{2} (\lambda_1 + \lambda_2) \tilde{e}_2, \quad [\tilde{e}_2, \tilde{e}_4] = \frac{\zeta}{2} (\lambda_1 + \lambda_2) \tilde{e}_1 - \frac{\zeta}{2} (\lambda_1 - \lambda_2) \tilde{e}_2.$$

Thus, these metrics are isomorphically isometric to metrics on $\mathbb{R}^3 \rtimes \mathbb{R}$ and therefore the corresponding algebraic *T*-solitons have already been described in Theorems 5.19 and 5.20.

Proof. The vanishing of the divergence of T is given by the system of polynomial equations

$$3bT_{14} - \lambda_2 T_{23} - A\lambda_2 T_{24} = 0,$$

$$\lambda_1 T_{13} + A\lambda_1 T_{14} + 3bT_{24} = 0,$$

$$(\lambda_1 - \lambda_2)T_{12} - CT_{14} - DT_{24} - 2bT_{34} = 0,$$

$$bT_{11} + bT_{22} - 2bT_{44} + A(\lambda_1 - \lambda_2)T_{12} + CT_{13} + DT_{23} = 0,$$
(5.20)

while the conditions for $\mathfrak{D} = \widehat{T} - \mu \operatorname{Id}$ to be a derivation will be determined by a system of twenty two (up to duplicities) polynomial equations on the soliton constant μ , the structure constants (5.19) and the components T_{ij} , given by $\{\mathfrak{P}_{ijk} = 0\}$, where

$$\begin{split} \mathfrak{P}_{211} &= -\lambda_1(T_{13} + AT_{14}) + bT_{24}, \\ \mathfrak{P}_{212} &= -bT_{14} - \lambda_2(T_{23} + AT_{24}), \\ \mathfrak{P}_{311} &= (\lambda_1 + \lambda_2)T_{12} - CT_{14} + bT_{34}, \\ \mathfrak{P}_{312} &= -\lambda_2(T_{11} - T_{22} + T_{33} - \mu) - DT_{14} - A\lambda_2T_{34}, \\ \mathfrak{P}_{313} &= \lambda_2T_{23}, \\ \mathfrak{P}_{314} &= \lambda_2T_{24}, \\ \mathfrak{P}_{321} &= -\lambda_1(T_{11} - T_{22} - T_{33} + \mu) - CT_{24} + A\lambda_1T_{34}, \\ \mathfrak{P}_{322} &= -(\lambda_1 + \lambda_2)T_{12} - DT_{24} + bT_{34}, \\ \mathfrak{P}_{323} &= -\lambda_1T_{13}, \\ \mathfrak{P}_{324} &= -\lambda_1T_{14}, \\ \mathfrak{P}_{411} &= b(T_{44} - \mu) + A(\lambda_1 + \lambda_2)T_{12} + CT_{13}, \\ \mathfrak{P}_{412} &= -A\lambda_2(T_{11} - T_{22} + T_{44} - \mu) + DT_{13} - \lambda_2T_{34}, \\ \mathfrak{P}_{413} &= -bT_{13} + A\lambda_2T_{23}, \\ \mathfrak{P}_{414} &= -bT_{14} + A\lambda_2T_{24}, \\ \\ \mathfrak{P}_{422} &= b(T_{44} - \mu) - A(\lambda_1 + \lambda_2)T_{12} + DT_{23}, \\ \mathfrak{P}_{423} &= -A\lambda_1T_{13} - bT_{23}, \\ \mathfrak{P}_{424} &= -A\lambda_1T_{14} - bT_{24}, \\ \\ \mathfrak{P}_{431} &= -C(T_{11} - T_{33} - T_{44} + \mu) - DT_{12} + bT_{13} + \lambda_1(AT_{23} - T_{24}), \\ \mathfrak{P}_{432} &= -D(T_{22} - T_{33} - T_{44} + \mu) - CT_{12} - \lambda_2(AT_{13} - T_{14}) + bT_{23}, \\ \\ \mathfrak{P}_{433} &= -CT_{13} - DT_{23}, \\ \\ \mathfrak{P}_{434} &= -CT_{14} - DT_{24}. \end{split}$$

Since $\lambda_1 \lambda_2 \neq 0$ and

$$\mathfrak{P}_{323} = -\lambda_1 T_{13}, \quad \mathfrak{P}_{324} = -\lambda_1 T_{14}, \quad \mathfrak{P}_{313} = \lambda_2 T_{23}, \quad \mathfrak{P}_{314} = \lambda_2 T_{24},$$
$$\mathfrak{P}_{312} = -\lambda_2 (T_{11} - T_{22} + T_{33} - \mu) - DT_{14} - A\lambda_2 T_{34},$$
$$\mathfrak{P}_{321} = -\lambda_1 (T_{11} - T_{22} - T_{33} + \mu) - CT_{24} + A\lambda_1 T_{34},$$

it immediately follows that

$$T_{13} = T_{14} = T_{23} = T_{24} = 0, \quad T_{22} = T_{11}, \quad T_{33} = \mu - AT_{34}.$$
 (5.21)

Using these conditions, we have

$$\mathfrak{P}_{412} = -\lambda_2 \left(T_{34} + A(T_{44} - \mu) \right),$$

which leads to

$$T_{34} = -A(T_{44} - \mu), \quad T_{33} = A^2(T_{44} - \mu) + \mu.$$
 (5.22)

Now, a direct computation leads to

$$\mathfrak{P}_{311} = (\lambda_1 + \lambda_2)T_{12} - Ab(T_{44} - \mu),$$

$$\mathfrak{P}_{322} = -(\lambda_1 + \lambda_2)T_{12} - Ab(T_{44} - \mu),$$

while the third equation in (5.20) now is

$$(\lambda_1 - \lambda_2)T_{12} + 2Ab(T_{44} - \mu) = 0,$$

which implies that

$$T_{12} = 0. (5.23)$$

At this point, Equations (5.21), (5.22) and (5.23) show that

$$T = \begin{pmatrix} T_{11} & 0 & 0 & 0 \\ * & T_{11} & 0 & 0 \\ * & * & A^2(T_{44} - \mu) + \mu & -A(T_{44} - \mu) \\ * & * & * & T_{44} \end{pmatrix}$$

The system of polynomial equations $\{\mathfrak{P}_{ijk} = 0\}$ reduces to

$$\begin{aligned} \mathfrak{P}_{311} &= \mathfrak{P}_{322} = -Ab(T_{44} - \mu) = 0, \\ \mathfrak{P}_{411} &= \mathfrak{P}_{422} = b(T_{44} - \mu) = 0, \\ \mathfrak{P}_{431} &= -C\left(T_{11} - A^2(T_{44} - \mu) - T_{44}\right) = 0, \\ \mathfrak{P}_{432} &= -D\left(T_{11} - A^2(T_{44} - \mu) - T_{44}\right) = 0, \end{aligned}$$

and the vanishing conditions for the divergence of T given in Equation (5.20) are

$$Ab(T_{44} - \mu) = 0, \quad b(T_{11} - T_{44}) = 0.$$

Note that if $b \neq 0$ the solution is determined by $T_{11} = T_{44} = \mu$ and the algebraic *T*-soliton is trivial. Finally, for b = 0 the solution is given by $\mathfrak{P}_{431} = \mathfrak{P}_{432} = 0$ and therefore C = D = 0 or, otherwise, $T_{11} = A^2(T_{44} - \mu) + T_{44}$, which completes the proof.

5.4.2 Algebraic Ricci solitons on $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$

In this case, the components ρ_{ij} of the Ricci tensor are given by

 $\begin{aligned} &2\rho_{11} = (A^2 + 1)(\lambda_1^2 - \lambda_2^2) - 4b^2 + C^2, & 2\rho_{13} = AD\lambda_2 - 3bC, \\ &2\rho_{12} = -2Ab(\lambda_1 - \lambda_2) + CD, & 2\rho_{14} = -D\lambda_2, \\ &2\rho_{22} = -(A^2 + 1)(\lambda_1^2 - \lambda_2^2) - 4b^2 + D^2, & 2\rho_{23} = -AC\lambda_1 - 3bD, \\ &2\rho_{33} = -(\lambda_1 - \lambda_2)^2 - C^2 - D^2, & 2\rho_{24} = C\lambda_1, \\ &2\rho_{44} = -A^2(\lambda_1 - \lambda_2)^2 - 4b^2 - C^2 - D^2, & 2\rho_{34} = -A(\lambda_1 - \lambda_2)^2. \end{aligned}$

We study the locally symmetric case in the first place. The possibilities we have in this situation are (see Remark 5.21) the Lie group $\tilde{E}(2) \rtimes \mathbb{R}$ with $\lambda_2 = \lambda_1$ and C = D = 0, or $E(1,1) \rtimes \mathbb{R}$, with $\lambda_2 = -\lambda_1$, C = D = 0 and $b^2 = (A^2 + 1)\lambda_1^2$. Besides, Theorem 5.23 guarantees that if there exists a non-trivial algebraic Ricci soliton, then necessarily b = 0, so the latter case is not possible, while in the former case a direct calculation shows that the space is flat due to b = 0.

In the non-symmetric case, according to Theorem 5.23, non-trivial algebraic Ricci solitons must satisfy $\rho_{14} = \rho_{24} = 0$, from where it follows that C = D = 0. Moreover, b must vanish. Now, Remark 5.24 shows that this case reduces to the previously considered situation of $\mathbb{R}^3 \rtimes \mathbb{R}$.

Remark 5.25. Even though we have mentioned that the non-symmetric case reduces to $\mathbb{R}^3 \rtimes \mathbb{R}$, we will solve it here for the sake of completeness. We showed that in this situation b = 0 and C = D = 0. Therefore, $\rho_{14} = \rho_{24} = 0$ and also $\rho_{13} = \rho_{23} = 0$, $\rho_{12} = 0$. Note that we can assume that $\lambda_1 \neq \lambda_2$ (see Remark 5.21) and so the algebraic Ricci solitons are characterized by

$$\rho_{11} = \rho_{22}, \quad \rho_{33} = A^2(\rho_{44} - \mu) + \mu, \quad \rho_{34} = -A(\rho_{44} - \mu).$$

If we compute the components of the Ricci tensor, we obtain

$$\begin{split} \rho_{11} &= -\rho_{22} = \frac{1}{2} (A^2 + 1) (\lambda_1^2 - \lambda_2^2), \\ \rho_{33} &= -\frac{1}{2} (\lambda_1 - \lambda_2)^2, \\ \rho_{34} &= A \rho_{33}, \quad \rho_{44} = A^2 \rho_{33}, \end{split}$$

which combined with the previous relations give

$$(A^{2} + 1)(\lambda_{1}^{2} - \lambda_{2}^{2}) = 0,$$

$$(A^{2} - 1)(2\mu + (A^{2} + 1)(\lambda_{1} - \lambda_{2})^{2}) = 0,$$

$$A(2\mu + (A^{2} + 1)(\lambda_{1} - \lambda_{2})^{2}) = 0.$$

Therefore $\lambda_2 = -\lambda_1$, $\mu = -2(A^2 + 1)\lambda_1^2$ and the algebraic Ricci solitons are given by the left-invariant metrics

$$[e_1, e_3] = \lambda_1 e_2, \quad [e_1, e_4] = A\lambda_1 e_2, \quad [e_2, e_3] = \lambda_1 e_1, \quad [e_2, e_4] = A\lambda_1 e_1.$$
(5.24)

Finally, setting $\zeta = A^2 + 1$ and considering the orthogonal basis

$$\tilde{e}_1 = \frac{1}{\lambda_1 \sqrt{2\zeta}} (e_1 + e_2), \quad \tilde{e}_2 = -\frac{1}{\lambda_1 \sqrt{2\zeta}} (e_1 - e_2),$$
$$\tilde{e}_3 = -\frac{1}{\lambda_1 \zeta} (Ae_3 - e_4), \quad \tilde{e}_4 = \frac{1}{\lambda_1 \zeta} (e_3 + Ae_4),$$

the non-zero Lie brackets reduce to

$$[\tilde{e}_1, \tilde{e}_4] = \tilde{e}_1, \quad \text{and} \quad [\tilde{e}_2, \tilde{e}_4] = -\tilde{e}_2.$$

Besides, $\langle \tilde{e}_i, \tilde{e}_j \rangle = \frac{1}{\lambda_{1\zeta}^2} \langle e_i, e_j \rangle$, so these metrics are isomorphically homothetic to the metric in Theorem 5.4-(*iii*) for f = -1 and p = 0.

5.4.3 Algebraic Bach solitons on $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$

First we compute the components \mathfrak{B}_{ij} of the Bach tensor of $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$. A long but straightforward calculation shows that they are determined by

$$\begin{split} 24\mathfrak{B}_{11} &= -4(A^2+1)^2 \left(5\lambda_1^4 - 3\lambda_2^4 - 3\lambda_1^3\lambda_2 + \lambda_1\lambda_2^3\right) \\ &\quad + (A^2+1) \left((28b^2 - 40C^2 + 3D^2)\lambda_1^2 - (20b^2 - C^2 - 8D^2)\lambda_2^2 \\ &\quad - 4(2b^2 - 3C^2 - D^2)\lambda_1\lambda_2\right) \\ &\quad - 42AbCD(\lambda_1 + \lambda_2) + b^2(43C^2 + D^2) - 4(C^2 + D^2)(5C^2 + D^2), \\ 12\mathfrak{B}_{12} &= 16A(A^2 + 1)b \left(\lambda_1^3 - \lambda_2^3\right) - (A^2 + 1)CD \left(8\lambda_1^2 + 8\lambda_2^2 + 5\lambda_1\lambda_2\right) \\ &\quad + Ab \left((16C^2 - 5D^2)\lambda_1 + (5C^2 - 16D^2)\lambda_2\right) + CD(21b^2 - 8C^2 - 8D^2), \\ 12\mathfrak{B}_{13} &= -A(A^2 + 1)D \left(8\lambda_2^3 - \lambda_1^2\lambda_2 - 4\lambda_1\lambda_2^2\right) \\ &\quad + 3bC \left(8(A^2 + 1)\lambda_1^2 - \lambda_2^2 - (3A^2 + 4)\lambda_1\lambda_2\right) \\ &\quad + AD \left(9b^2\lambda_1 + 4(3b^2 - 2C^2 - 2D^2)\lambda_2\right) - 3bC(3b^2 - 8C^2 - 8D^2), \\ 12\mathfrak{B}_{14} &= -\lambda_2 \left((A^2 + 1)D \left(\lambda_1^2 - 8\lambda_2^2 + 4\lambda_1\lambda_2\right) + 3AbC(\lambda_1 + \lambda_2) + D(3b^2 - 8C^2 - 8D^2)\right), \\ 24\mathfrak{B}_{22} &= 4(A^2 + 1)^2 \left(3\lambda_1^4 - 5\lambda_2^4 - \lambda_1^3\lambda_2 + 3\lambda_1\lambda_2^3\right) \\ &\quad - (A^2 + 1) \left((20b^2 - 8C^2 - D^2)\lambda_1^2 - (28b^2 + 3C^2 - 40D^2)\lambda_2^2 \\ &\quad + 4(2b^2 - C^2 - 3D^2)\lambda_1\lambda_2\right) \\ &\quad + 42AbCD(\lambda_1 + \lambda_2) + b^2(C^2 + 43D^2) - 4(C^2 + D^2)(C^2 + 5D^2), \end{split}$$

$$\begin{split} &12\mathfrak{B}_{23} = A(A^2+1)C\left(8\lambda_1^3 - 4\lambda_1^2\lambda_2 - \lambda_1\lambda_2^2\right) - 3bD\left(\lambda_1^2 - 8(A^2+1)\lambda_2^2 + (3A^2+4)\lambda_1\lambda_2\right) \\ &\quad -AC\left(4(3b^2 - 2C^2 - 2D^2)\lambda_1 + 9b^2\lambda_2\right) - 3bD(3b^2 - 8C^2 - 8D^2), \\ &12\mathfrak{B}_{24} = -\lambda_1\left((A^2+1)C\left(8\lambda_1^2 - \lambda_2^2 - 4\lambda_1\lambda_2\right) + 3AbD(\lambda_1+\lambda_2) - C(3b^2 - 8C^2 - 8D^2)\right), \\ &24\mathfrak{B}_{33} = -4(A^2 - 3)(A^2 + 1)\left(\lambda_1^4 + \lambda_2^4 - \lambda_1^3\lambda_2 - \lambda_1\lambda_2^3\right) \\ &\quad + \left((A^2+3)(8C^2 - D^2) - 12(A^2+1)b^2\right)\lambda_1^2 - \left((A^2+3)(C^2 - 8D^2) + 12(A^2+1)b^2\right)\lambda_2^2 \\ &\quad - 4((A^2+3)(C^2 + D^2) - 6(A^2+1)b^2)\lambda_1\lambda_2 - 18AbCD(\lambda_1 - \lambda_2) \\ &\quad - 3(C^2 + D^2)(19b^2 - 4C^2 - 4D^2), \\ &12\mathfrak{B}_{34} = 8A(A^2+1)\left(\lambda_1^4 + \lambda_2^4 - \lambda_1^3\lambda_2 - \lambda_1\lambda_2^3\right) \\ &\quad + A\left((8C^2 - D^2)\lambda_1^2 - (C^2 - 8D^2)\lambda_2^2 - 4(C^2 + D^2)\lambda_1\lambda_2\right) + 9bCD(\lambda_1 - \lambda_2). \end{split}$$

Recall that the Bach tensor is trace-free, so $\mathfrak{B}_{44} = -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33}$. Next we analyse non-trivial algebraic Bach solitons using Theorem 5.23. Since *b* has to be zero, in the locally symmetric case, which has already been detailed in Remark 5.21, we only have to deal with the case $\lambda_2 = \lambda_1$ and C = D = 0. A direct calculation shows that the space is flat in this situation, and so it is Bach-flat.

In the non-symmetric case we set b = 0 and assume that the corresponding Bach tensor is of the form given in Theorem 5.23. Since

$$\mathfrak{B}_{12} = -\frac{1}{24}CD\left((A^2+1)(5(\lambda_1+\lambda_2)^2+11(\lambda_1^2+\lambda_2^2))+16C^2+16D^2\right)$$

must vanish, it follows that CD = 0. According to Remark 5.22, we can assume C = 0 and work in the isometric class of the initial metric. Moreover, Remark 5.24 shows that the case where D = 0 reduces to the previously considered situation of $\mathbb{R}^3 \rtimes \mathbb{R}$. Hence, we assume that $D \neq 0$. Since

$$\mathfrak{B}_{14} = \frac{1}{12}D\lambda_2 \left(8D^2 - (A^2 + 1)(\lambda_1^2 - 8\lambda_2^2 + 4\lambda_1\lambda_2)\right)$$

must vanish, then

$$D^{2} = \frac{1}{8}(A^{2} + 1)(\lambda_{1}^{2} - 8\lambda_{2}^{2} + 4\lambda_{1}\lambda_{2}).$$
(5.25)

Considering $\mathfrak{B}_{11} - \mathfrak{B}_{22}$, which must vanish, we obtain

 $\frac{3}{16}(A^2+1)^2\lambda_1^3(7\lambda_1-4\lambda_2)=0,$

from where it immediately follows that $\lambda_2 = \frac{7}{4}\lambda_1$. This leads to an incompatibility with Equation (5.25), since $\lambda_1^2 - 8\lambda_2^2 + 4\lambda_1\lambda_2 = -\frac{33}{2}\lambda_1^2 < 0$, and therefore no algebraic Bach solitons exist if $D \neq 0$.

Remark 5.26. For the sake of completeness, we will solve the non-symmetric case when D = 0, even though we already know that it reduces to $\mathbb{R}^3 \rtimes \mathbb{R}$. If b = 0 and C = D = 0, then

$$\mathfrak{B}_{12}=\mathfrak{B}_{13}=\mathfrak{B}_{14}=\mathfrak{B}_{23}=\mathfrak{B}_{24}=0.$$

Moreover, since

$$\mathfrak{B}_{11} = -\frac{1}{6}(A^2 + 1)^2 \left(5\lambda_1^4 - 3\lambda_2^4 - 3\lambda_1^3\lambda_2 + \lambda_1\lambda_2^3\right),$$

$$\mathfrak{B}_{22} = -\frac{1}{6}(A^2 + 1)^2 \left(3\lambda_1^4 - 5\lambda_2^4 - \lambda_1^3\lambda_2 + 3\lambda_1\lambda_2^3\right),$$

must be equal, then

$$\mathfrak{B}_{11} - \mathfrak{B}_{22} = -\frac{2}{3}(A^2 + 1)^2(\lambda_1^2 - \lambda_2^2)\left((\lambda_1 - \frac{1}{2}\lambda_2)^2 + \lambda_1^2 + \frac{7}{4}\lambda_2^2\right) = 0.$$

Hence, either $\lambda_2 = \lambda_1$ or $\lambda_2 = -\lambda_1$. In the former case $(b = C = D = 0, \lambda_2 = \lambda_1)$ the space is flat (see Remark 5.21). In the latter case $(b = C = D = 0, \lambda_2 = -\lambda_1)$ the algebraic Bach solitons are determined by

$$\mathfrak{B}_{33} = A^2(\mathfrak{B}_{44} - \mu) + \mu, \quad \mathfrak{B}_{34} = -A(\mathfrak{B}_{44} - \mu),$$

and a direct calculation shows that

$$\mathfrak{B}_{33} - A^2(\mathfrak{B}_{44} - \mu) - \mu = (A^2 - 1) (\mu - 2(A^2 + 1)^2 \lambda_1^4),$$

$$\mathfrak{B}_{34} + A(\mathfrak{B}_{44} - \mu) = -A (\mu - 2(A^2 + 1)^2 \lambda_1^4).$$

We obtain $\mu = 2(A^2 + 1)^2 \lambda_1^4$ and the associated left-invariant metrics are given by

$$[e_1, e_3] = \lambda_1 e_2, \quad [e_1, e_4] = A\lambda_1 e_2, \quad [e_2, e_3] = \lambda_1 e_1, \quad [e_2, e_4] = A\lambda_1 e_1,$$

so the algebraic Bach solitons are also the algebraic Ricci soliton (see Equation (5.24)) that have already been covered by the analysis of $\mathbb{R}^3 \rtimes \mathbb{R}$.

5.5 Algebraic solitons on $H^3 \rtimes \mathbb{R}$

We follow the description of the left-invariant metrics on semi-direct extensions of the Heisenberg group given in Section 1.4.2 and consider an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ such that

$$[e_1, e_2] = \gamma e_3, \qquad [e_1, e_4] = ae_1 - ce_2 + He_3, [e_3, e_4] = (a+d)e_3, \qquad [e_2, e_4] = ce_1 + de_2 + Fe_3,$$
(5.26)

where $\gamma \neq 0$.

Remark 5.27. $H^3 \rtimes \mathbb{R}$ is locally symmetric if and only if the structure constants satisfy

$$a = d = \frac{\varepsilon}{2}\gamma$$
 and $F = H = 0$,

where $\varepsilon^2 = 1$. The isomorphism determined by $e_4 \mapsto -e_4$ is an orientation reversing isometry so we can set $\varepsilon = 1$ up to a change of orientation. A straightforward calculation shows that the anti-self-dual Weyl curvature operator W^- vanishes an so the underlying manifold is homothetic to the complex hyperbolic plane \mathbb{CH}^2 (see [55]). This is the only Einstein metric on $H^3 \rtimes \mathbb{R}$.

Remark 5.28. The isometry $(e_1, e_2, e_3, e_4) \mapsto (-e_2, e_1, e_3, e_4)$ transforms (γ, a, c, d, H, F) in Equation (5.26) into $(\gamma, d, c, a, -F, H)$. Thus, any left-invariant metric (5.26) with H = 0 is isomorphically isometric to a left-invariant metric with F = 0.
5.5.1 Algebraic *T*-solitons on $H^3 \rtimes \mathbb{R}$

Theorem 5.29. $H^3 \rtimes \mathbb{R}$ is a non-trivial algebraic *T*-soliton with soliton constant μ if and only if one of the following conditions holds.

(i) $a + d \neq 0$. In this case, the space is isomorphically isometric to a Riemannian Lie group given by one of the following possibilities.

(i.a) $a \neq \pm d$, c = 0, F = 0, and the tensor field \widehat{T} is diagonal,

$$\widehat{T} = \operatorname{diag}[T_{11}, T_{22}, T_{11} + T_{22} - \mu, \mu] \neq \mu \operatorname{Id},$$

with

$$H(T_{22} - \mu) = 0, \quad (2a+d)T_{11} + (a+2d)T_{22} - 3(a+d)\mu = 0.$$

(*i.b*) $a = d \neq 0, c = 0, F = H = 0$ and

$$T = \begin{pmatrix} T_{11} & T_{12} & 0 & 0 \\ * & 2\mu - T_{11} & 0 & 0 \\ * & * & \mu & 0 \\ * & * & * & \mu \end{pmatrix} \neq \mu \langle \cdot, \cdot \rangle.$$

- (ii) a + d = 0. In this case, the space is isomorphically isometric to a Riemannian Lie group given by one of the following possibilities.
 - (ii.a) a = -d, $a^2 \neq c^2$, F = H = 0 and the tensor field \widehat{T} is diagonal,

 $\widehat{T} = \text{diag}[T_{11}, T_{11}, 2T_{11} - \mu, \mu], \quad T_{11} \neq \mu.$

(*ii.b*) a = d = c = 0 and

$$T = \begin{pmatrix} T_{11} & T_{12} & 0 & \frac{1}{\gamma} \left(F(T_{22} - T_{33} + T_{44} - \mu) + HT_{12} \right) \\ * & T_{22} & 0 & \frac{-1}{\gamma} \left(H(T_{11} - T_{33} + T_{44} - \mu) + FT_{12} \right) \\ * & * & T_{33} & 0 \\ * & * & * & T_{44} \end{pmatrix} \neq \mu \langle \cdot, \cdot \rangle,$$

with

$$(H^2 - \gamma^2)T_{11} + (F^2 - \gamma^2)T_{22} - (F^2 + H^2 - \gamma^2)T_{33} + (F^2 + H^2)T_{44} + 2FHT_{12} - (F^2 + H^2 - \gamma^2)\mu = 0.$$

(ii.c) $a = -d, c = a \neq 0$ and

$$T = \begin{pmatrix} T_{11} & T_{12} & 0 & T_{14} \\ * & T_{11} & 0 & T_{14} \\ * & * & 2T_{11} - \frac{1}{\gamma}(F - H)T_{14} - \mu & 0 \\ * & * & * & \mu - 2T_{12} \end{pmatrix} \neq \mu \langle \cdot, \cdot \rangle$$

where

$$HT_{11} - (F - 2H)T_{12} - \frac{1}{\gamma} \left(H(F - H) + \gamma^2\right) T_{14} - H\mu = 0,$$

$$FT_{11} + (2F - H)T_{12} - \frac{1}{\gamma} \left(F(F - H) - \gamma^2\right) T_{14} - F\mu = 0.$$

Remark 5.30. If a = c = d = 0 as in Assertion (*ii.b*) in the previous theorem, then the left-invariant metrics associated to the corresponding algebraic T-solitons are given by

$$[e_1, e_2] = \gamma e_3, \quad [e_1, e_4] = He_3, \quad [e_2, e_4] = Fe_3$$

Taking $\zeta=\sqrt{F^2+H^2+\gamma^2}$ and considering the orthonormal basis

$$\begin{split} \tilde{e}_1 &= \frac{1}{\zeta} \left(Fe_1 - He_2 + \gamma e_4 \right), \\ \tilde{e}_2 &= \frac{1}{\sqrt{2(F^2 + H^2)}} \left(He_1 + Fe_2 - \sqrt{F^2 + H^2} e_3 \right), \\ \tilde{e}_3 &= \frac{-1}{\sqrt{2(F^2 + H^2)}} \left(He_1 + Fe_2 + \sqrt{F^2 + H^2} e_3 \right), \\ \tilde{e}_4 &= \frac{1}{\zeta \sqrt{F^2 + H^2}} \left(\gamma Fe_1 - \gamma He_2 - (F^2 + H^2) e_4 \right), \end{split}$$

the only non-zero brackets now correspond to

$$[\tilde{e}_2, \tilde{e}_4] = \frac{\zeta}{2}(\tilde{e}_2 + \tilde{e}_3), \quad [\tilde{e}_3, \tilde{e}_4] = -\frac{\zeta}{2}(\tilde{e}_2 + \tilde{e}_3).$$

These metrics are isomorphically isometric to metrics on $\mathbb{R}^3 \rtimes \mathbb{R}$ and therefore the corresponding algebraic *T*-solitons have already been covered by Theorem 5.19.

We will show that the same conclusion holds true when the structure constants a = -dand $a = c \neq 0$ as in Theorem 5.29-(*ii.c*) if, in addition, H = -F and $4a^2 - 2F^2 - \gamma^2 = 0$ (equivalently, the Ricci operator is of rank three). The left-invariant metrics associated to the algebraic T-solitons are given by

$$[e_1, e_2] = \gamma e_3, \quad [e_1, e_4] = ae_1 - ae_2 - Fe_3, \quad [e_2, e_4] = ae_1 - ae_2 + Fe_3, \tag{5.27}$$

and if we consider the orthonormal basis determined by

$$\begin{split} \tilde{e}_1 &= \frac{1}{2a\sqrt{2}} \left(Fe_1 + Fe_2 + 2ae_3 + \gamma e_4 \right), \\ \tilde{e}_2 &= \frac{-1}{4a} \left((2a - F)e_1 - (2a + F)e_2 + 2ae_3 - \gamma e_4 \right), \\ \tilde{e}_3 &= \frac{-1}{4a} \left((2a + F)e_1 - (2a - F)e_2 - 2ae_3 + \gamma e_4 \right), \\ \tilde{e}_4 &= \frac{1}{2a\sqrt{2}} \left(\gamma e_1 + \gamma e_2 - 2Fe_4 \right), \end{split}$$

the non-zero brackets correspond to

$$[\tilde{e}_1, \tilde{e}_4] = a(\tilde{e}_2 + \tilde{e}_3), \quad [\tilde{e}_2, \tilde{e}_4] = -a(\tilde{e}_1 - \sqrt{2}\tilde{e}_2), \quad [\tilde{e}_3, \tilde{e}_4] = -a(\tilde{e}_1 + \sqrt{2}\tilde{e}_3).$$

Thus, the above metrics are isomorphically isometric to metrics on $\mathbb{R}^3 \rtimes \mathbb{R}$ and so the corresponding algebraic *T*-solitons have already been covered by Theorem 5.19.

Proof. Since we are working with a divergence-free tensor field T of type (0, 2), a direct calculation shows that this condition is equivalent to

$$(3a+2d)T_{14} + \gamma T_{23} - cT_{24} + HT_{34} = 0,$$

$$\gamma T_{13} - cT_{14} - (2a+3d)T_{24} - FT_{34} = 0,$$

$$3(a+d)T_{34} = 0,$$

$$aT_{11} + dT_{22} + (a+d)(T_{33} - 2T_{44}) + HT_{13} + FT_{23} = 0.$$

(5.28)

The conditions for $\mathfrak{D} = \widehat{T} - \mu \operatorname{Id}$ to be a derivation can be expressed in terms of a system of twenty two (up to duplicities) polynomial equations on the soliton constant μ , the structure constants (5.26) and the components T_{ij} , given by $\{\mathfrak{P}_{ijk} = 0\}$, where

$$\begin{split} \mathfrak{P}_{211} &= -\gamma T_{13} - cT_{14} + aT_{24}, \\ \mathfrak{P}_{212} &= -dT_{14} - \gamma T_{23} - cT_{24}, \\ \mathfrak{P}_{213} &= \gamma (T_{11} + T_{22} - T_{33} - \mu) - FT_{14} + HT_{24}, \\ \mathfrak{P}_{214} &= -\gamma T_{34}, \\ \mathfrak{P}_{311} &= aT_{34}, \\ \mathfrak{P}_{312} &= -cT_{34}, \\ \mathfrak{P}_{313} &= -(a+d)T_{14} + \gamma T_{23} + HT_{34}, \\ \mathfrak{P}_{322} &= dT_{34}, \\ \mathfrak{P}_{322} &= dT_{34}, \\ \mathfrak{P}_{323} &= -\gamma T_{13} - (a+d)T_{24} + FT_{34}, \\ \mathfrak{P}_{411} &= a(T_{44} - \mu) + 2cT_{12} - HT_{13}, \\ \mathfrak{P}_{412} &= -c(T_{11} - T_{22} + T_{44} - \mu) - (a-d)T_{12} - HT_{23}, \\ \mathfrak{P}_{413} &= H(T_{11} - T_{33} + T_{44} - \mu) + FT_{12} + dT_{13} + cT_{23} + \gamma T_{24}, \\ \mathfrak{P}_{414} &= -aT_{14} + cT_{24} - HT_{34}, \\ \mathfrak{P}_{421} &= -c(T_{11} - T_{22} - T_{44} + \mu) + (a-d)T_{12} - FT_{13}, \\ \mathfrak{P}_{422} &= d(T_{44} - \mu) - 2cT_{12} - FT_{23}, \\ \mathfrak{P}_{423} &= F(T_{22} - T_{33} + T_{44} - \mu) + HT_{12} - cT_{13} - \gamma T_{14} + aT_{23}, \\ \mathfrak{P}_{431} &= -dT_{13} + cT_{23}, \\ \mathfrak{P}_{431} &= -dT_{13} + cT_{23}, \\ \mathfrak{P}_{432} &= -cT_{13} - aT_{23}, \\ \end{split}$$

$$\mathfrak{P}_{433} = (a+d)(T_{44}-\mu) + HT_{13} + FT_{23},$$

$$\mathfrak{P}_{434} = -(a+d)T_{34}.$$

We start by considering

$$\begin{aligned} \mathfrak{P}_{214} &= -\gamma T_{34}, \\ \mathfrak{P}_{211} + \mathfrak{P}_{323} - \mathfrak{P}_{424} &= -2(\gamma T_{13} - FT_{34}), \\ \mathfrak{P}_{212} - \mathfrak{P}_{313} + \mathfrak{P}_{414} &= -2(\gamma T_{23} + HT_{34}), \end{aligned}$$

which imply that

$$T_{13} = T_{23} = T_{34} = 0, (5.29)$$

and split our analysis differentiating the cases $a + d \neq 0$ and a + d = 0.

Case 1: $a + d \neq 0$.

Using Equation (5.29), since

$$\begin{aligned} \mathfrak{P}_{313} &= -(a+d)T_{14}, \\ \mathfrak{P}_{323} &= -(a+d)T_{24}, \\ \mathfrak{P}_{433} &= (a+d)(T_{44}-\mu), \\ \mathfrak{P}_{213} &= \gamma(T_{11}+T_{22}-T_{33}-\mu) - FT_{14} + HT_{24}, \end{aligned}$$

and $a + d \neq 0$ we obtain

$$T_{14} = T_{24} = 0, \quad T_{44} = \mu, \quad T_{33} = T_{11} + T_{22} - \mu.$$
 (5.30)

Next we show that the situation is different depending on whether c vanishes or not.

Case 1.1:
$$c \neq 0$$
.

In this case Equations (5.29) and (5.30) imply that

$$\mathfrak{P}_{411} = 2cT_{12}, \quad \mathfrak{P}_{421} = -c(T_{11} - T_{22}) + (a - d)T_{12},$$

so the condition $c \neq 0$ leads to

$$T_{12} = 0, \quad T_{22} = T_{11}. \tag{5.31}$$

At this point, Equations (5.29), (5.30) and (5.31) make \hat{T} diagonal,

$$\widehat{T} = \operatorname{diag}[T_{11}, T_{11}, 2T_{11} - \mu, \mu]$$

and, moreover, the vanishing conditions of the divergence of T given in Equation (5.28) reduce to

$$(a+d)(T_{11}-\mu) = 0.$$

Hence, we conclude that there are no non-trivial algebraic T-soliton in this case.

<u>Case 1.2: c = 0.</u>

Using Equation (5.30) together with Equation (5.29) and c = 0, it is straightforward to see that

$$T = \begin{pmatrix} T_{11} & T_{12} & 0 & 0 \\ * & T_{22} & 0 & 0 \\ * & * & T_{11} + T_{22} - \mu & 0 \\ * & * & * & \mu \end{pmatrix}$$

The system $\{\mathfrak{P}_{ijk} = 0\}$ now corresponds to

$$\mathfrak{P}_{412} = -\mathfrak{P}_{421} = -(a-d)T_{12} = 0,$$

$$\mathfrak{P}_{413} = FT_{12} - H(T_{22} - \mu) = 0,$$

$$\mathfrak{P}_{423} = -F(T_{11} - \mu) + HT_{12} = 0,$$

while the vanishing of the divergence of T given by Equation (5.28) reduces to

$$(2a+d)T_{11} + (a+2d)T_{22} - 3(a+d)\mu = 0.$$

If $a - d \neq 0$, then $T_{12} = 0$. Besides, if $FH \neq 0$, then $T_{11} = T_{22} = \mu$ and the tensor field T is trivial. Thus, according to Remark 5.28, we can take F = 0 (working with an isomorphically isometric metric if necessary) and Assertion (*i.a*) immediately follows.

If a = d, they are both different from zero, since $a + d \neq 0$. Therefore, the last equation above implies that

$$T_{22} = 2\mu - T_{11}$$

while the other equations give

$$FT_{12} + H(T_{11} - \mu) = 0, \quad HT_{12} - F(T_{11} - \mu) = 0.$$

Note that if either F or H is non-zero, then the corresponding algebraic T-soliton is trivial. Therefore, F = H = 0 and Assertion (*i.b*) follows.

Case 2: a + d = 0.

In this situation, we distinguish two cases depending on whether a^2 and c^2 are equal or not.

Case 2.1: $a^2 \neq c^2$.

A direct calculation involving Equation (5.29) and the condition d = -a shows that

$$\begin{aligned} \mathfrak{P}_{211} &= -cT_{14} + aT_{24}, \\ \mathfrak{P}_{212} &= aT_{14} - cT_{24}, \\ 2a\mathfrak{P}_{411} + c(\mathfrak{P}_{412} - \mathfrak{P}_{421}) &= 2(a^2 - c^2)(T_{44} - \mu), \\ \mathfrak{P}_{213} &= \gamma(T_{11} + T_{22} - T_{33} - \mu) - FT_{14} + HT_{24}, \end{aligned}$$

so $a^2 \neq c^2$ implies that

$$T_{14} = T_{24} = 0, \quad T_{44} = \mu, \quad T_{33} = T_{11} + T_{22} - \mu.$$
 (5.32)

This last equation together with (5.29) lead to

$$\mathfrak{P}_{411} = 2cT_{12}, \quad \mathfrak{P}_{421} = 2aT_{12} - c(T_{11} - T_{22}),$$

while the vanishing of the divergence of T given in Equation (5.28) reduces to

$$a(T_{11} - T_{22}) = 0.$$

Hence, since $a^2 \neq c^2$, it follows that

$$T_{12} = 0, \quad T_{11} = T_{22}. \tag{5.33}$$

Putting together Equations (5.29), (5.32) and (5.33) we obtain that \hat{T} is diagonal,

$$\widehat{T} = \text{diag}[T_{11}, T_{11}, 2T_{11} - \mu, \mu],$$

and the system of polynomial equations $\{\mathfrak{P}_{ijk}=0\}$ reduces to

$$\mathfrak{P}_{413} = H(\mu - T_{11}) = 0, \quad \mathfrak{P}_{423} = F(\mu - T_{11}) = 0.$$

Thus, there are non-trivial algebraic T-solitons when $T_{11} \neq \mu$ and F = H = 0, so Assertion (*ii.a*) is obtained.

Case 2.2: $a^2 = c^2$.

We set $c = \varepsilon a$, with $\varepsilon^2 = 1$, and assume that the conditions in Equation (5.29) hold. In this case, Equation (5.28), which gives the conditions for T to be divergence-free, transforms into

$$a(T_{11} - T_{22}) = 0, \quad a(\varepsilon T_{14} - T_{24}) = 0$$

If a = 0, then c = d = 0 and the tensor T is divergence-free, while the system $\{\mathfrak{P}_{ijk} = 0\}$ reduces to

$$\mathfrak{P}_{423} = F(T_{22} - T_{33} + T_{44} - \mu) + HT_{12} - \gamma T_{14} = 0,$$

$$\mathfrak{P}_{413} = H(T_{11} - T_{33} + T_{44} - \mu) + FT_{12} + \gamma T_{24} = 0,$$

$$\mathfrak{P}_{213} = \gamma (T_{11} + T_{22} - T_{33} - \mu) - FT_{14} + HT_{24} = 0.$$

Clearing T_{14} and T_{24} in the first and second equations above, respectively, Assertion (*ii.b*) is immediately obtained.

If $a \neq 0$, then we compute

$$\mathfrak{P}_{411} = a(2\varepsilon T_{12} + T_{44} - \mu),$$

which together with the conditions for T to be a divergence-free tensor give

$$T_{22} = T_{11}, \quad T_{24} = \varepsilon T_{14}, \quad T_{44} = \mu - 2\varepsilon T_{12}.$$

Thus, the system of polynomial equations $\{\mathfrak{P}_{ijk} = 0\}$ reduces to

$$\mathfrak{P}_{213} = \gamma (2T_{11} - T_{33} - \mu) - (F - \varepsilon H)T_{14} = 0,$$

$$\mathfrak{P}_{413} = H(T_{11} - T_{33}) + (F - 2\varepsilon H)T_{12} + \varepsilon \gamma T_{14} = 0,$$

$$\mathfrak{P}_{423} = F(T_{11} - T_{33}) + (H - 2\varepsilon F)T_{12} - \gamma T_{14} = 0,$$

and from the first equation above we obtain an expression for T_{33} . Note that the corresponding left-invariant metrics are given by

$$[e_1, e_2] = \gamma e_3, \quad [e_1, e_4] = ae_1 - \varepsilon ae_2 + He_3, \quad [e_2, e_4] = \varepsilon ae_1 - ae_2 + Fe_3,$$

and the isometry

$$(e_1, e_2, e_3, e_4) \mapsto (e_1, -e_2, -e_3, e_4)$$

interchanges (ε, a, F, H) and $(-\varepsilon, a, F, -H)$. Thus we can take $\varepsilon = 1$ working, if necessary, with an isomorphically isometric metric and Assertion *(ii.c)* is obtained.

5.5.2 Algebraic Ricci solitons on $H^3 \rtimes \mathbb{R}$

A straightforward calculation shows that the Ricci tensor is determined by

$$\begin{aligned} &2\rho_{11} = -4a(a+d) - H^2 - \gamma^2, & 2\rho_{12} = -2c(a-d) - FH, \\ &2\rho_{13} = -H(2a+3d) + Fc, & 2\rho_{14} = F\gamma, \\ &2\rho_{22} = -4d(a+d) - F^2 - \gamma^2, & 2\rho_{23} = -F(3a+2d) - Hc, \\ &2\rho_{33} = -4(a+d)^2 + F^2 + H^2 + \gamma^2, & 2\rho_{24} = -H\gamma, \\ &2\rho_{44} = -4(a^2 + d^2 + ad) - F^2 - H^2. \end{aligned}$$

We use Theorem 5.29 to analyse all the possibilities for non-trivial algebraic Ricci solitons. In the locally symmetric case, Remark 5.27 guarantees that $a = d = \pm \frac{1}{2}\gamma$, F = H = 0, and a direct calculation shows that

$$\operatorname{Ric} = -\frac{3}{2}\gamma_1^2 \operatorname{Id},$$

so the corresponding algebraic Ricci solitons are trivial.

In the non-symmetric case, we study each of the five cases in Theorem 5.29 separately. Note that we can take $\gamma = 1$ and work in the same homothetic class.

Case (*i.a*).

We take c = 0, F = 0 and $d \neq \pm a$. Algebraic Ricci solitons must have diagonal Ricci operator, so the expression $\rho_{24} = -\frac{1}{2}H$ implies that H = 0. With this last condition, the Ricci operator is diagonal. Now, algebraic Ricci solitons are characterized by the vanishing of the polynomials

$$\begin{aligned} \mathfrak{Q}_1 &= \rho_{33} - \rho_{11} - \rho_{22} + \mu, \quad \mathfrak{Q}_2 &= \rho_{44} - \mu, \\ \mathfrak{Q}_3 &= (2a+d)\rho_{11} + (a+2d)\rho_{22} - 3(a+d)\mu, \end{aligned}$$

and a direct calculation shows that

$$\begin{split} \mathfrak{Q}_1 &= \mu + \frac{3}{2}, \\ \mathfrak{Q}_2 &= -\mu - 2(a^2 + d^2 + ad), \\ \mathfrak{Q}_3 &= -\frac{1}{2}(a+d) \left(6\mu + 8(a^2 + d^2 + ad) + 3\right). \end{split}$$

Therefore the system $\{\mathfrak{Q}_i = 0\}$ is equivalent to

$$4(a^2 + d^2 + ad) - 3 = 0, \quad \mu = -\frac{3}{2},$$

which determine algebraic Ricci solitons with associated left-invariant metrics given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = ae_1, \quad [e_2, e_4] = de_2, \quad [e_3, e_4] = (a+d)e_3.$$
 (5.34)

Note that the isometry $e_4 \mapsto -e_4$ transforms (a, d) into (-a, -d), while the isometry

$$(e_1, e_2, e_3, e_4) \mapsto (e_2, -e_1, e_3, e_4)$$

transforms (a, d) into (d, a). As a consequence of this together with $d \neq \pm a$, we can assume that |a| < d. Besides, the condition $4(a^2 + d^2 + ad) - 3 = 0$ implies that $a \in (-\frac{\sqrt{3}}{2}, \frac{1}{2})$. This corresponds to Theorem 5.4-(*iv*) for $a \neq -\frac{\sqrt{3}}{2}$.

Case (*i.b*).

Since $a = d \neq 0$ and c = F = H = 0, a direct calculation shows that the Ricci tensor satisfies

$$\rho_{33} = -8a^2 + \frac{1}{2} \quad \text{and} \quad \rho_{44} = -6a^2.$$

Since an algebraic Ricci soliton must satisfy $\rho_{33} = \rho_{44} = \mu$, we get $a = \pm \frac{1}{2}$, which implies that the space is locally symmetric (see Remark 5.27).

Case (*ii.a*).

In this case, d = -a, $a^2 \neq c^2$ and F = H = 0. A direct calculation shows that $\rho_{11} = \rho_{22} = -\frac{1}{2}$ and, moreover, the Ricci operator is diagonal if and only if $\rho_{12} = 0$, so the algebraic Ricci solitons are determined by the vanishing of the polynomials

$$\mathfrak{Q}_1 = \rho_{12}, \quad \mathfrak{Q}_2 = \rho_{33} - 2\rho_{11} + \mu, \quad \mathfrak{Q}_3 = \rho_{44} - \mu.$$

Computing these expressions we obtain

$$\mathfrak{Q}_1 = -2ac, \quad \mathfrak{Q}_2 = \mu + \frac{3}{2}, \quad \mathfrak{Q}_3 = -\mu - 2a^2,$$

which imply that c = 0 and $a = \pm \frac{\sqrt{3}}{2}$. With these conditions we obtain an algebraic Ricci soliton with associated left-invariant metric given by Equation (5.34) with

$$a = -d = \pm \frac{\sqrt{3}}{2}.$$

As we did in the previous case, we can set $a = -\frac{\sqrt{3}}{2}$, and this metric corresponds to Theorem 5.4-(*iv*) for this value of a.

Case (ii.b).

This case has already been covered by the analysis on $\mathbb{R}^3 \rtimes \mathbb{R}$ (see Remark 5.30).

Case (*ii.c*).

In this last case d = -a and $c = a \neq 0$. Since

$$\rho_{34} = 0 \quad \text{and} \quad \rho_{13} = -\rho_{23} = \frac{1}{2}a(F+H),$$

then we must take H = -F in order for the Ricci tensor to have the matrix form given in Theorem 5.29-(*ii.c*). Now, an algebraic Ricci soliton is characterized by the vanishing of the polynomials

$$\begin{split} \mathfrak{Q}_1 &= \rho_{11} - \rho_{22}, & \mathfrak{Q}_2 &= \rho_{33} - 2\rho_{11} + 2F\rho_{14} + \mu, \\ \mathfrak{Q}_3 &= \rho_{44} + 2\rho_{12} - \mu, & \mathfrak{Q}_4 &= \rho_{24} - \rho_{14}, \\ \mathfrak{Q}_5 &= -F\rho_{11} - 3F\rho_{12} + (2F^2 - 1)\rho_{14} + F\mu, \end{split}$$

and a direct calculation leads to $\mathfrak{Q}_1 = \mathfrak{Q}_4 = 0$ and

$$\mathfrak{Q}_2 = \mu + 3F^2 + \frac{3}{2}, \quad \mathfrak{Q}_3 = -\mu - 6a^2, \quad \mathfrak{Q}_5 = F(\mu + 6a^2).$$
 (5.35)

Consequently, $4a^2 - 2F^2 - 1 = 0$ and it follows from Remark 5.30 that this case has already been covered by the analysis on $\mathbb{R}^3 \rtimes \mathbb{R}$.

Remark 5.31. For the sake of completeness, we will include here the solution of the two cases covered by the analysis on $\mathbb{R}^3 \rtimes \mathbb{R}$.

Case (*ii.b*) in Theorem 5.29 is given by a = c = d = 0. In this case, it is easy to check that $\rho_{13} = \rho_{23} = \rho_{34} = 0$, so the matrix form in Theorem 5.29-(*ii.b*) is satisfied and an algebraic Ricci soliton is determined by the vanishing of the polynomials

$$\begin{split} \mathfrak{Q}_1 &= \rho_{14} - F(\rho_{22} - \rho_{33} + \rho_{44} - \mu) - H\rho_{12}, \\ \mathfrak{Q}_2 &= \rho_{24} + H(\rho_{11} - \rho_{33} + \rho_{44} - \mu) + F\rho_{12}, \\ \mathfrak{Q}_3 &= (H^2 - 1)\rho_{11} + (F^2 - 1)\rho_{22} - (F^2 + H^2 - 1)\rho_{33} \\ &+ (F^2 + H^2)\rho_{44} + 2FH\rho_{12} - (F^2 + H^2 - 1)\mu. \end{split}$$

A direct calculation gives

$$\begin{split} \mathfrak{Q}_1 &= \quad \tfrac{1}{2}F(3F^2+3H^2+2\mu+3), \\ \mathfrak{Q}_2 &= -\tfrac{1}{2}H(3F^2+3H^2+2\mu+3), \\ \mathfrak{Q}_3 &= -\tfrac{1}{2}(F^2+H^2-1)(3F^2+3H^2+2\mu+3) \end{split}$$

so the system $\{\mathfrak{Q}_i = 0\}$ is equivalent to $\mu = -\frac{3}{2}(F^2 + H^2 + 1)$. The associated left-invariant metrics are given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = He_1, \quad [e_2, e_4] = Fe_3,$$
 (5.36)

and considering the orthogonal basis

$$\begin{split} \tilde{e}_1 &= \frac{-2}{F^2 + H^2 + 1} \left(Fe_1 - He_2 + e_4 \right), \\ \tilde{e}_2 &= \frac{\sqrt{2}}{\sqrt{(F^2 + H^2 + 1)(F^2 + H^2)}} \left(He_1 + Fe_2 - \sqrt{F^2 + H^2}e_3 \right), \\ \tilde{e}_3 &= \frac{-\sqrt{2}}{\sqrt{(F^2 + H^2 + 1)(F^2 + H^2)}} \left(He_1 + Fe_2 + \sqrt{F^2 + H^2}e_3 \right), \\ \tilde{e}_4 &= \frac{2}{(F^2 + H^2 + 1)\sqrt{F^2 + H^2}} \left(Fe_1 - He_2 - (F^2 + H^2)e_4 \right), \end{split}$$

the only non-zero brackets are

$$[\tilde{e}_2, \tilde{e}_4] = \tilde{e}_2 + \tilde{e}_3$$
 and $[\tilde{e}_3, \tilde{e}_4] = -\tilde{e}_2 - \tilde{e}_3$.

Moreover, $\langle \tilde{e}_i, \tilde{e}_j \rangle = \frac{4}{F^2 + H^2 + 1} \langle e_i, e_j \rangle$, which shows that the above metric Lie groups are isomorphically homothetic to the metric in Theorem 5.4-(*i*).

Finally, in Case (*ii.c*) of Theorem 5.29, using Equation (5.35) we obtain that $4a^2-2F^2-1=0$ and $\mu = -6a^2$ are the conditions that determine the algebraic Ricci solitons in this case. These have associated left-invariant metrics given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = ae_1 - ae_2 - Fe_3, \quad [e_2, e_4] = ae_1 - ae_2 + Fe_3.$$
 (5.37)

Considering the orthogonal basis

$$\begin{split} \tilde{e}_1 &= \frac{1}{4a^2} \left(Fe_1 + Fe_2 + 2ae_3 + e_4 \right), \\ \tilde{e}_2 &= \frac{-1}{4a^2\sqrt{2}} \left((2a - F)e_1 - (2a + F)e_2 + 2ae_3 - e_4 \right), \\ \tilde{e}_3 &= \frac{-1}{4a^2\sqrt{2}} \left((2a + F)e_1 - (2a - F)e_2 - 2ae_3 + e_4 \right), \\ \tilde{e}_4 &= \frac{1}{4a^2} \left(e_1 + e_2 - 2Fe_4 \right), \end{split}$$

the non-zero brackets correspond to

$$[\tilde{e}_1, \tilde{e}_4] = \frac{1}{\sqrt{2}}(\tilde{e}_2 + \tilde{e}_3), \quad [\tilde{e}_2, \tilde{e}_4] = -\frac{1}{\sqrt{2}}\tilde{e}_1 + \tilde{e}_2, \quad [\tilde{e}_3, \tilde{e}_4] = -\frac{1}{\sqrt{2}}\tilde{e}_1 - \tilde{e}_3,$$

and $\langle \tilde{e}_i, \tilde{e}_j \rangle = \frac{1}{2a^2} \langle e_i, e_j \rangle$. Hence, the metric Lie groups above are isomorphically homothetic to the metric given in Assertion (*ii*) of Theorem 5.4.

5.5.3 Algebraic Bach solitons on $H^3 \rtimes \mathbb{R}$

A long but straightforward calculation shows that the components \mathfrak{B}_{ij} of the Bach tensor of $H^3 \rtimes \mathbb{R}$ are determined by

$$\begin{aligned} 24\mathfrak{B}_{11} &= -16a^3d + 48ad^3 + 84a^2c^2 + 16a^2d^2 - 108c^2d^2 + 24ac^2d \\ &\quad + (F^2 - 20H^2 - 20\gamma^2)a^2 - 21(F^2 - H^2)c^2 - 3(4F^2 + 19H^2 + 4\gamma^2)d^2 \\ &\quad + 78FHac - 4(22H^2 + 7\gamma^2)ad + 78FHcd \\ &\quad - 4(F^2 + H^2 + \gamma^2)(F^2 - 3H^2 - 3\gamma^2), \end{aligned}$$

$$\begin{split} &12\mathfrak{B}_{12} = -18a^3c + 24ac^3 - 24c^3d + 18cd^3 - 58a^2cd + 58acd^2 \\ &\quad - 3FH\left(4a^2 - 7c^2 + 4d^2\right) + (31F^2 - 8H^2 - 2\gamma^2)ac - 53FHad \\ &\quad + (8F^2 - 31H^2 + 2\gamma^2)cd + 8FH(F^2 + H^2 + \gamma^2), \\ &12\mathfrak{B}_{13} = -3Fc^3 - 9Hd^3 + 33Fa^2c + 3Hac^2 - 28Ha^2d - 48Had^2 + 24Hc^2d \\ &\quad - 9Fcd^2 + 53Facd + 8(F^2 + H^2 + \gamma^2)(2Ha - Fc + 3Hd), \\ &12\mathfrak{B}_{14} = \gamma \left(3F\left(a^2 - c^2\right) - 3Hac + 14Fad + 15Hcd - 8F(F^2 + H^2 + \gamma^2)\right), \\ &24\mathfrak{B}_{22} = 48a^3d - 16ad^3 - 108a^2c^2 + 16a^2d^2 + 84c^2d^2 + 24ac^2d \\ &\quad - 3(19F^2 + 4H^2 + 4\gamma^2)a^2 + 21(F^2 - H^2)c^2 - (20F^2 - H^2 + 20\gamma^2)d^2 \\ &\quad - 78FHac - 4(22F^2 + 7\gamma^2)ad - 78FHcd \\ &\quad + 4(F^2 + H^2 + \gamma^2)(3F^2 - H^2 + 3\gamma^2), \\ &12\mathfrak{B}_{23} = -9Fa^3 + 3Hc^3 + 9Ha^2c + 24Fac^2 - 48Fa^2d - 28Fad^2 + 3Fc^2d \\ &\quad - 33Hcd^2 - 53Hacd + 8(F^2 + H^2 + \gamma^2)(3Fa + Hc + 2Fd), \\ &12\mathfrak{B}_{24} = \gamma \left(3H\left(c^2 - d^2\right) + 15Fac - 14Had - 3Fcd + 8H(F^2 + H^2 + \gamma^2)\right), \\ &24\mathfrak{B}_{33} = -16a^3d - 16ad^3 - 12a^2c^2 - 48a^2d^2 - 12c^2d^2 + 24ac^2d \\ &\quad + (43F^2 + 28H^2 + 28\gamma^2)a^2 - 9(F^2 + H^2)c^2 + (28F^2 + 43H^2 + 28\gamma^2)d^2 \\ &\quad - 54FHac + (104F^2 + 104H^2 + 44\gamma^2)ad + 54FHcd - 20(F^2 + H^2 + \gamma^2)^2. \end{split}$$

Recall that the Bach tensor is trace-free, so $\mathfrak{B}_{44} = -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33}$. We study the existence of non-trivial algebraic Bach solitons using Theorem 5.29. In the locally symmetric case, the underlying manifold is homothetic to the complex hyperbolic plane (see Remark 5.27), and so it is self-dual, and thus Bach-flat.

In the non-symmetric case, we proceed as in the previous section by checking the five cases given in Theorem 5.29 separately. Without loss of generality we can set $\gamma = 1$ remaining in the same homothetic class.

Case (*i.a*).

In this first case, c = 0, F = 0 and $d \neq \pm a$. Note that

$$\mathfrak{B}_{12} = \mathfrak{B}_{14} = \mathfrak{B}_{23} = \mathfrak{B}_{34} = 0$$

and recall that $\mathfrak{B}_{44} = -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33}$. If we now impose the diagonal form and the conditions in Theorem 5.29-(*i.a*) to the Bach operator we have that an algebraic Bach soliton is determined by the vanishing of the polynomials

$$\begin{aligned} \mathfrak{Q}_{1} &= \mathfrak{B}_{13}, \quad \mathfrak{Q}_{2} &= \mathfrak{B}_{24}, \quad \mathfrak{Q}_{3} &= -2(\mathfrak{B}_{11} + \mathfrak{B}_{22}), \\ \mathfrak{Q}_{4} &= H(\mathfrak{B}_{11} + 2\mathfrak{B}_{22} + \mathfrak{B}_{33}), \\ \mathfrak{Q}_{5} &= (5a + 4d)\mathfrak{B}_{11} + (4a + 5d)\mathfrak{B}_{22} + 3(a + d)\mathfrak{B}_{33}, \\ \mathfrak{Q}_{6} &= \mu + \mathfrak{B}_{11} + \mathfrak{B}_{22} + \mathfrak{B}_{33}. \end{aligned}$$

First of all we use \mathfrak{Q}_2 and \mathfrak{Q}_3 to show that H must be zero. Since

$$\mathfrak{Q}_2 = -\frac{1}{12}H(3d^2 + 14ad - 8H^2 - 8)$$

must vanish, if $H \neq 0$ then d cannot be null and $a = \frac{-1}{14d}(3d^2 - 8H^2 - 8)$. With this value for a a long but direct calculation shows that

$$\mathfrak{Q}_3 = \frac{1}{223636d^2} \left(d^2 + 2H^2 + 2 \right) \left(652H^2 (407d^2 + 64H^2 + 128) + (326d^2 - 83)^2 + 34839 \right)$$

which leads to a contradiction since $\mathfrak{Q}_3 = 0$ and none of the factors above vanishes. Consequently H = 0 and the system $\{\mathfrak{Q}_i = 0\}$ becomes

$$\begin{aligned} \mathfrak{Q}_3 &= -\frac{2}{3}(ad-1)(4(a^2+d^2+ad)-3), \\ \mathfrak{Q}_5 &= -\frac{2}{3}(ad-1)(4(a^2+d^2+ad)-3)(a+d) \\ \mathfrak{Q}_6 &= \mu + \frac{1}{6}(4ad-1)(a^2+d^2-ad-1), \end{aligned}$$

or equivalently,

$$\mu = -\frac{1}{6}(4ad - 1)(a^2 + d^2 - ad - 1), \quad (ad - 1)(4(a^2 + d^2 + ad) - 3) = 0.$$

Thus, we obtain algebraic Bach solitons with associated left-invariant metrics given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = ae_1, \quad [e_2, e_4] = de_2, \quad [e_3, e_4] = (a+d)e_3.$$

If $4(a^2+d^2+ad)-3 = 0$, the space is an algebraic Ricci soliton (see Equation (5.34)). Otherwise, ad = 1, so these structure constants must be non-zero and $d = \frac{1}{a}$. In this case, $\mu = -\frac{(a^2-1)^2}{2a^2}$, the associated left-invariant metrics transform into

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = ae_1, \quad [e_2, e_4] = \frac{1}{a}e_2, \quad [e_3, e_4] = \frac{a^2+1}{a}e_3,$$

and their Bach operators are diagonal, given by

$$\widehat{B} = \frac{1}{2a^2} \left[-3(a^4 - 1), 3(a^4 - 1), (a^2 - 1)^2, -(a^2 - 1)^2 \right].$$

The assumption $d \neq \pm a$ implies that $a \neq \pm 1$. Thus, these spaces are neither Bach-flat nor algebraic Ricci solitons. Furthermore, the isometry $e_4 \mapsto -e_4$ interchanges a with -a, while $(e_1, e_2, e_3, e_4) \mapsto (e_2, -e_1, e_3, e_4)$ interchanges a with $\frac{1}{a}$, so we can restrict the parameter a to (0, 1). These metrics correspond to Assertion *(ii)* in Theorem 5.8.

Case (*i.b*).

Since $a = d \neq 0$ and c = F = H = 0, a direct calculation shows that the Bach operator is diagonal,

$$\widehat{\mathfrak{B}} = \frac{4a^4 - 5a^2 + 1}{6} \operatorname{diag}[3, 3, -5, -1].$$

Since \mathfrak{B}_{33} and \mathfrak{B}_{44} must be equal, any algebraic Bach soliton is clearly Bach-flat in this case.

Case (*ii.a*).

The structure constants must satisfy d = -a, $a^2 \neq c^2$ and F = H = 0. A direct calculation shows that the diagonalizability of the Bach operator depends only on \mathfrak{B}_{12} and the condition $\mathfrak{B}_{11} = \mathfrak{B}_{22}$. Recall that the Bach tensor is trace-free and so $\mathfrak{B}_{44} = -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33}$. Hence, the existence of algebraic Bach solitons is determined by the vanishing of the polynomials

$$\mathfrak{Q}_1 = \mathfrak{B}_{12}, \quad \mathfrak{Q}_2 = -3\mathfrak{B}_{11} - \mathfrak{B}_{22}, \quad \mathfrak{Q}_3 = \mu + \mathfrak{B}_{11} + \mathfrak{B}_{22} + \mathfrak{B}_{33}$$

and it is not difficult to see that

$$\begin{aligned} \mathfrak{Q}_1 &= \frac{1}{3}ac(20a^2 + 12c^2 - 1), \\ \mathfrak{Q}_2 &= \frac{2}{3}(4a^4 + (12c^2 + 1)a^2 - 3), \\ \mathfrak{Q}_3 &= \mu - \frac{1}{6}(12a^4 + 36a^2c^2 - a^2 - 1). \end{aligned}$$

Thus, the system of polynomial equations $\{\mathfrak{Q}_i = 0\}$ is equivalent to c = 0, $a = \pm \frac{\sqrt{3}}{2}$ and $\mu = \frac{5}{6}$, which characterize the algebraic Bach solitons whose left-invariant metrics are given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = \pm \frac{\sqrt{3}}{2}e_1, \quad [e_2, e_4] = \mp \frac{\sqrt{3}}{2}e_2,$$

so they are also algebraic Ricci solitons (see Equation (5.34) with $a = -d = \pm \frac{\sqrt{3}}{2}$).

Case (ii.b).

This case reduces to the analysis of $\mathbb{R}^3 \rtimes \mathbb{R}$, according to Remark 5.30.

Case (ii.c).

We take d = -a and $c = a \neq 0$. The Bach tensor must have the matrix form given in Theorem 5.29-(*ii.c*), and a direct calculation shows that $\mathfrak{B}_{34} = 0$ and

$$\mathfrak{B}_{13} = -\mathfrak{B}_{23} = -\frac{2}{3}a(4a^2 + F^2 + H^2 + 1)(H + F)$$

Consequently, H = -F. Moreover, since $\mathfrak{B}_{44} = -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33}$, an algebraic Bach soliton is characterized by the vanishing of the polynomials

$$\begin{split} \mathfrak{Q}_1 &= \mathfrak{B}_{11} - \mathfrak{B}_{22}, \qquad \mathfrak{Q}_2 &= \mathfrak{B}_{33} - 2\mathfrak{B}_{11} + 2F\mathfrak{B}_{14} + \mu, \\ \mathfrak{Q}_3 &= \mathfrak{B}_{44} + 2\mathfrak{B}_{12} - \mu, \qquad \mathfrak{Q}_4 &= \mathfrak{B}_{24} - \mathfrak{B}_{14}, \\ \mathfrak{Q}_5 &= -F\mathfrak{B}_{11} - 3F\mathfrak{B}_{12} + (2F^2 - 1)\mathfrak{B}_{14} + F\mu. \end{split}$$

A direct calculation leads to $\mathfrak{Q}_1 = \mathfrak{Q}_4 = 0$ and

$$\mathfrak{Q}_{2} = \frac{1}{6} (16a^{2} - 22F^{2} - 11)(a^{2} + 2F^{2} + 1) + \mu,$$

$$\mathfrak{Q}_{3} = \frac{1}{6} (11a^{2} - 2F^{2} - 1)(16a^{2} + 2F^{2} + 1) - \mu,$$

$$\mathfrak{Q}_{5} = -F \left(\frac{1}{6} (11a^{2} - 2F^{2} - 1)(16a^{2} + 2F^{2} + 1) - \mu\right),$$
(5.38)

and thus

$$\mathfrak{Q}_2 + \mathfrak{Q}_3 = 2(4a^2 + 2F^2 + 1)(4a^2 - 2F^2 - 1)$$

We conclude that $4a^2 - 2F^2 - 1 = 0$ and Remark 5.30 shows that the analysis reduces to that in $\mathbb{R}^3 \rtimes \mathbb{R}$.

Remark 5.32. For the sake of completeness, we include here the study of the two cases covered by the analysis carried out on $\mathbb{R}^3 \rtimes \mathbb{R}$.

Considering the situation of Case (*ii.b*) in Theorem 5.29, we take a = c = d = 0. For these values, $\mathfrak{B}_{13} = \mathfrak{B}_{23} = \mathfrak{B}_{34} = 0$ and the Bach tensor has the matrix form given in Theorem 5.29-(*ii.b*). Since $\mathfrak{B}_{44} = -\mathfrak{B}_{11} - \mathfrak{B}_{22} - \mathfrak{B}_{33}$, the vanishing of the polynomials

$$\begin{split} \mathfrak{Q}_{1} &= \mathfrak{B}_{14} + F(\mathfrak{B}_{11} + 2\mathfrak{B}_{33} + \mu) - H\mathfrak{B}_{12}, \\ \mathfrak{Q}_{2} &= \mathfrak{B}_{24} - H(\mathfrak{B}_{22} + 2\mathfrak{B}_{33} + \mu) + F\mathfrak{B}_{12}, \\ \mathfrak{Q}_{3} &= -(F^{2} + 1)\mathfrak{B}_{11} - (H^{2} + 1)\mathfrak{B}_{22} - (2F^{2} + 2H^{2} - 1)\mathfrak{B}_{33} \\ &+ 2FH\mathfrak{B}_{12} - (F^{2} + H^{2} - 1)\mu, \end{split}$$

determines the algebraic Bach solitons in this case. A direct calculation shows that

$$\begin{aligned} \mathfrak{Q}_1 &= -F\left(\frac{11}{6}(F^2 + H^2 + 1)^2 - \mu\right), \\ \mathfrak{Q}_2 &= H\left(\frac{11}{6}(F^2 + H^2 + 1)^2 - \mu\right), \\ \mathfrak{Q}_3 &= (F^2 + H^2 - 1)(\frac{11}{6}(F^2 + H^2 + 1)^2 - \mu), \end{aligned}$$

and therefore the system $\{\mathfrak{Q}_i = 0\}$ is equivalent to $\mu = \frac{11}{6}(F^2 + H^2 + 1)^2$. The associated left-invariant metrics are given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = He_1, \quad [e_2, e_4] = Fe_3,$$

so they are algebraic Ricci solitons as well (see Equation (5.36)).

Finally, in Case (*ii.c*) of Theorem 5.29 we use Equation (5.38) to obtain that $4a^2-2F^2-1=0$ and $\mu = \frac{70}{3}a^4$ are the conditions which characterize an algebraic Bach soliton. Note that these algebraic Bach solitons, whose associated left-invariant metrics are given by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = ae_1 - ae_2 - Fe_3, \quad [e_2, e_4] = ae_1 - ae_2 + Fe_3,$$

are algebraic Ricci solitons as well (see Equation (5.37)).

Part III

Homogeneous manifolds and homogeneous structures

Jensen proved in [88] a well-known result that states that, in the Riemannian homogeneous setting, Einstein metrics are symmetric, which constitutes a very rigid situation. The Weyl curvature tensor of any Einstein metric is divergence-free – in other words, it is harmonic –, a condition that is equivalent to the fact that the corresponding Schouten tensor is a Codazzi tensor. If the scalar curvature is constant, then the Ricci tensor is also a Codazzi tensor. Podestà and Spiro showed in [123] that the condition div W = 0 implies local symmetry in the locally homogeneous four-dimensional case.

On four-dimensional oriented four-manifolds, the decomposition of the Weyl curvature tensor into its self-dual and anti-self-dual components $W = W^+ + W^-$, which is conformal invariant, makes the harmonicity condition div W = 0 become div $W^+ + \text{div } W^- = 0$. A fourdimensional manifold is said to have half-harmonic Weyl curvature tensor if either div $W^+ = 0$ or div $W^- = 0$. These conditions are natural generalizations of the Einstein condition and have received special attention during the last decade (see [46,48,107] and [134]).

In Theorem 6.1 we give the complete description – up to homotheties – of four-dimensional locally homogeneous Riemannian manifolds with half-harmonic Weyl curvature tensor. The manifolds in this result are naturally equipped with a self-dual homogeneous structure which is not necessarily the canonical homogeneous structure of the underlying Lie group. Given the fact that homogeneous manifolds may admit different presentations as homogeneous spaces, they may as well admit more than one homogeneous structure. Therefore we decided to study which Lie groups admit at least two inequivalent homogeneous structures. In Theorem 7.1 we show that a three-dimensional non-symmetric homogeneous space admits at least two inequivalent homogeneous structures if and only if its isometry group is four-dimensional.

Motivated by the results obtained in Theorem 6.1, we started to investigate four-dimensional self-dual homogeneous structures, obtaining some partial classification results. Nevertheless, the whole classification of four-dimensional Riemannian self-dual homogeneous structures remains an open problem.

Chapter 6 Homogeneous four-manifolds with half-harmonic Weyl curvature

In this chapter we will work in the Riemannian homogeneous setting to study what the manifolds that admit half-harmonic Weyl curvature are like. The contents of this chapter are partially contained in the work [37].

6.1 Summary of results

A four-dimensional Riemannian manifold (M, g) has harmonic Weyl curvature if its Weyl tensor is divergence free, i.e., $\delta W = \operatorname{div}_1 W = 0$ or, equivalently, if its Cotton tensor vanishes. Even though the Weyl curvature tensor is conformally invariant, the condition $\delta W = 0$ is not. In fact, for any conformal metric $\tilde{g} = e^{2\sigma}g$, one has that $\operatorname{div}_1 \widetilde{W} = \operatorname{div}_1 W - \iota_{\nabla_\sigma} W$. Einstein metrics have harmonic Weyl curvature, which is the reason why the condition $\delta W = 0$ has been investigated in order to extend the geometric properties of Einstein metrics to more general structures. On the other hand, any locally symmetric manifold trivially has harmonic Weyl curvature and it follows from [15, 54] that the converse is also true in the four-dimensional homogeneous setting (see also [123]).

We have already mentioned that the space of two-forms on an oriented four-dimensional Riemannian manifold splits as the direct sum of the spaces of self-dual and anti-self-dual two-forms, $\Lambda^2_{\pm}(TM)$, under the action of SO(4) and that such decomposition corresponds to the one given by the ± 1 eigenspaces of the Hodge-star operator

$$\star \colon \Lambda^2(TM) \to \Lambda^2(TM),$$

so that the self-dual and anti-self-dual components of any two-form Ω are given by $\Omega_{\pm} = \frac{1}{2} (\Omega \pm \star \Omega)$. Since the Weyl conformal curvature operator $\mathcal{W} \in \text{End } \Lambda^2(TM)$ and the Hodgestar operator commute, the bundles $\Lambda^2_{\pm}(TM)$ remain invariant under its action. In what follows, we will denote by \mathcal{W}^{\pm} the self-dual and anti-self-dual components of the Weyl curvature operator, which are its restrictions to the self-dual and anti-self-dual subspaces of the space of two-forms, respectively, and an oriented Riemannian four-manifold is said to be self-dual (resp. anti-self-dual) if $\mathcal{W}^- = 0$ (resp. $\mathcal{W}^+ = 0$). Given the decomposition of the Weyl curvature operator into its self-dual and anti-self-dual components, its divergence decomposes accordingly and so (M, g) is said to have *half-harmonic Weyl curvature* if either $\delta \mathcal{W}^+ = 0$ or $\delta \mathcal{W}^- = 0$. The conditions $\delta \mathcal{W}^{\pm} = 0$ have been extensively investigated during the past decade in order to describe different classes of Riemannian manifolds such as Ricci solitons [46, 48, 134] and quasi-Einstein metrics [107].

A first example of metrics that have half-harmonic Weyl curvature can be found in the Kähler setting. The geometry of four-dimensional Kähler manifolds is, to a great extent, codified by their self-dual and anti-self-dual Weyl curvatures. In fact, the anti-self-dual Weyl curvature codifies the Bochner tensor and so the condition $\delta W^- = 0$ is equivalent to weak Bochner-flatness ($\delta B = 0$) [8, 87]. It was shown in [89] that weakly Bochner-flat Kähler surfaces of constant scalar curvature are locally symmetric (thus having harmonic Weyl curvature). On the other hand, the self-dual component of the Weyl operator is $W^+ = \frac{\tau}{12} \operatorname{diag}[2, -1, -1]$ and

$$\tau \operatorname{div}_1 W^+ + \iota_{\nabla \tau} W^+ = 0.$$

Therefore, $\delta W^+ = 0$ if and only if the scalar curvature τ is constant. Moreover, any fourdimensional Kähler surface with $W^+ \neq 0$ is conformal to a metric with $\delta W^+ = 0$.

We will show that four-dimensional homogeneous manifolds with half-harmonic Weyl curvature tensor are either symmetric or locally homothetic to the only non-symmetric anti-selfdual homogeneous manifold or to the 3-symmetric space, so non-symmetric homogeneous fourmanifolds with half-harmonic Weyl curvature have an underlying complex structure that is either Kähler or locally conformally Kähler. This description is summarised in the main result of this chapter – whose proof is detailed in Section 6.2 – as follows.

Theorem 6.1. Let (M, g) be a four-dimensional locally homogeneous Riemannian manifold with half-harmonic Weyl curvature tensor. Then it is symmetric or locally homothetic to one of the following semi-direct extensions of the Heisenberg group:

(i) The left-invariant metric on $H^3 \rtimes \mathbb{R}$ determined by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = -e_1, \quad [e_2, e_4] = -e_2, \quad [e_3, e_4] = -2e_3,$$

(ii) or the left-invariant metric on $H^3 \rtimes \mathbb{R}$ determined by

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = \frac{1}{2}e_1, \quad [e_2, e_4] = -e_2, \quad [e_3, e_4] = -\frac{1}{2}e_3,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis.

Remark 6.2. The metrics in Theorem 6.1-(*i*) are anti-self-dual and correspond to those in [55]. The anti-self-dual Weyl curvature operator has eigenvalues $\{-2, 1, 1\}$ and $\Omega_{-} = e^1 \wedge e^2 - e^3 \wedge e^4$ is an eigenvector associated to the distinguished eigenvalue -2. A straightforward calculation shows that the associated almost complex structure $J_{-}e_1 = e_2$, $J_{-}e_3 = -e_4$ is integrable and $d\Omega_{-} = -e^4 \wedge \Omega_{-}$. Therefore, the structure is locally conformally Kähler with respect to the opposite orientation. Furthermore, it was shown in [39] that these metrics are conformally Ricci-flat and, therefore, quasi-Einstein.

Remark 6.3. The self-dual and anti-self-dual Weyl curvature operators of the metrics in Theorem 6.1-(*ii*) have opposite eigenvalues $\{\pm \frac{1}{4}, \pm \frac{1}{2}, \pm \frac{1}{4}\}$. The eigenvectors corresponding to the distinguished eigenvalues of multiplicity one,

$$\Omega_{\pm} = e^1 \wedge e^3 \mp e^2 \wedge e^4,$$

define a symplectic pair so that the underlying manifold is Kähler and opposite almost Kähler. While Ω_+ determines a Kähler structure $J_+e_1 = e_3$, $J_+e_2 = -e_4$, the opposite almost complex structure $J_-e_1 = e_3$, $J_-e_2 = e_4$ determined by Ω_- is not integrable and the Ricci operator is J_- invariant. Now it follows from [7, Theorem 1] that it corresponds to the unique four-dimensional 3-symmetric space, which is the only homogeneous non-symmetric Kähler surface. Moreover, it is an algebraic Ricci soliton [42], thus being an expanding Ricci soliton.

Remark 6.4. The Bach tensor, $\mathfrak{B} = \operatorname{div}_2 \operatorname{div}_4 W + \frac{1}{2}W[\rho]$, of the metrics given by Theorem 6.1-(i) vanishes, since they are anti-self-dual. Therefore, these metrics are trivially steady Bach solitons (see [81]). On the contrary, the Bach tensor of the Kähler metrics in Theorem 6.1-(*ii*),

$$\mathfrak{B} = \operatorname{div}_2 \operatorname{div}_4 W + \frac{1}{2} W[\rho] = \operatorname{div}_2 \operatorname{div}_4 W^+ + \frac{1}{2} W^+[\rho] = \frac{1}{2} W^+[\rho]$$

is determined by the non-zero components

$$\mathfrak{B}(e_1, e_1) = \mathfrak{B}(e_3, e_3) = -\mathfrak{B}(e_2, e_2) = -\mathfrak{B}(e_4, e_4) = -\frac{3}{8}.$$

A straightforward calculation shows that $\mathcal{D} = \hat{\mathfrak{B}} - \frac{3}{8}$ Id is a derivation of the Lie algebra in Theorem 6.1-(ii), where $\hat{\mathfrak{B}}$ is the (1, 1)-tensor field associated to the (0, 2)-Bach tensor \mathfrak{B} . Let φ_t be the one-parameter family of automorphisms of the Lie algebra determined by $d\varphi_{t_{|e}} = e^{-\frac{t}{2}\mathcal{D}}$. Proceeding as in Lauret's work [97], the vector field X on $H^3 \rtimes \mathbb{R}$ given by $X = \frac{d}{dt}_{|t=0}\varphi_t$ defines a Bach soliton, i.e., $\mathcal{L}_X \langle \cdot, \cdot \rangle + \mathfrak{B} = \frac{3}{8} \langle \cdot, \cdot \rangle$. Consequently, the 3-symmetric space is both an algebraic Ricci soliton and an algebraic Bach soliton. Moreover, since the metric does not split as a product $\mathbb{R}^k \times N^{4-k}$, neither the Ricci nor the Bach solitons are gradient (see [81, 121]). The soliton above does not correspond to those studied in [81] since the underlying structure is not the product $H^3 \times \mathbb{R}$.

Remark 6.5. The norm of the self-dual Weyl curvature operator of any Kähler surface satisfies $\|\mathcal{W}^+\|^2 = \frac{1}{24}\tau^2$. If the scalar curvature is constant, then it follows from the Weitzenböck formula (see [18])

$$\Delta \|\mathcal{W}^+\|^2 = -\tau \|\mathcal{W}^+\|^2 + 36 \det_{\Lambda^2_+} \mathcal{W}^+ - 2\|\nabla \mathcal{W}^+\|^2$$

that the self-dual Weyl curvature operator is parallel. Therefore, homogeneous four-manifolds with half-harmonic Weyl curvature are such that $\nabla W^+ = 0$.

Remark 6.6. The Bach tensor of the anti-self-dual metric in Theorem 6.1-(*i*) vanishes and so it is critical for the functional \mathcal{F}_t with $t = -\frac{1}{3}$. The energy of this functional is strictly negative and this metric is the only one which is Bach-flat with non-zero energy on a semi-direct extension of the Heisenberg group.

The 3-symmetric space in Theorem 6.1-(*ii*) is a Ricci soliton and thus it is critical for the functional \mathcal{F}_t with $t = -\frac{1}{2}$. This is the only metric which is critical with zero energy on a semi-direct extension of the Heisenberg group.

6.2 Half-harmonic Weyl curvature on homogeneous manifolds

Simply connected four-dimensional homogeneous Riemannian manifolds are either symmetric or isometric to a Lie group with a left-invariant metric (see [15]). Any non-symmetric homogeneous metric is thus realized either on the product Lie groups $SU(2) \times \mathbb{R}$ and $\widetilde{SL}(2, \mathbb{R}) \times \mathbb{R}$, or on the semi-direct extensions $E(1, 1) \rtimes \mathbb{R}$, $\widetilde{E}(2) \rtimes \mathbb{R}$, $H^3 \rtimes \mathbb{R}$ and $\mathbb{R}^3 \rtimes \mathbb{R}$, where E(1, 1), $\widetilde{E}(2)$, H^3 and \mathbb{R}^3 denote the Poincaré group, the the universal covering of the Euclidean group, the Heisenberg group and the Abelian group, respectively.

As we have already mentioned, any symmetric four-dimensional homogeneous Riemannian manifold has harmonic Weyl curvature, therefore this situation will not provide examples of strictly half-harmonic Weyl curvature. De Smedt and Salamon proved that if (G, g) is a simply connected Lie group with a left-invariant metric and $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis of its Lie algebra, there are two cases in which there exist orientation-reversing isometric automorphisms (see [55, Lemma 2.2]).

1. The Lie algebra \mathfrak{g} has non-zero centre \mathfrak{z} and the isometric automorphism that reverses the orientation of (G, g) is given by

$$(e_1, e_2, e_3, e_4) \longmapsto (-e_1, e_2, e_3, e_4)$$

for an orthonormal basis such that $e_1 \in \mathfrak{z}$. This situation corresponds to the products $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$.

2. G corresponds to the semi-direct extension $\mathbb{R}^3 \rtimes \mathbb{R}$. In this situation, if we fix an orthonormal basis such that $e_2, e_3, e_4 \in \mathfrak{r}^3$, then

$$(e_1, e_2, e_3, e_4) \longmapsto (e_1, -e_2, -e_3, -e_4)$$

determines an orientation-reversing isometric automorphism of G.

As a consequence, the Weyl curvature of these three Lie groups cannot be strictly half-harmonic. Nevertheless, we will address these cases quickly in the following remarks for the sake of completeness.

Remark 6.7. We will use a direct approach to show that the Weyl curvature of $\mathbb{R}^3 \rtimes \mathbb{R}$ cannot be strictly half-harmonic. Let us consider left-invariant metrics on the semi-direct extensions of the Abelian Lie algebra as described in Section 1.4.2. In this case there exists an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ such that

$$[e_1, e_4] = ae_1 + be_2 + ce_3, \qquad [e_2, e_4] = -be_1 + fe_2 + he_3,$$

$$[e_3, e_4] = -ce_1 - he_2 + pe_3.$$
 (6.1)

Now, a straightforward calculation shows that the divergences $\operatorname{div}_1 W^{\pm}(e_i, e_j, e_k)$ are determined – up to the corresponding symmetries – by

$$\operatorname{div}_1 W^{\pm}(e_i, e_j, e_k) = \frac{1}{4} \mathfrak{P}_{ijk}^{\pm}$$

where \mathfrak{P}_{ijk}^{\pm} are the polynomials on the structure constants in (6.1) given by

$$\begin{split} \mathfrak{P}_{112}^{\pm} &= \mp \left(c(a^2 - 2p^2 + af + ap - fp) - bh(a - 2f + p) \right), \\ \mathfrak{P}_{113}^{\pm} &= \pm \left(b(a^2 - 2f^2 + af + ap - fp) + ch(a + f - 2p) \right), \\ \mathfrak{P}_{114}^{\pm} &= a^2(f + p) - af^2 - ap^2 - 2(b^2 + c^2)a + 2b^2f + 2c^2p, \\ \mathfrak{P}_{212}^{\pm} &= \mp \left(h(f^2 - 2p^2 + af - ap + fp) - bc(2a - f - p) \right), \\ \mathfrak{P}_{213}^{\pm} &= \pm \left(a^2f - (a + p)f^2 + fp^2 - 2b^2a + 2(b^2 + h^2)f - 2h^2p \right), \\ \mathfrak{P}_{214}^{\pm} &= -b(2a^2 - f^2 - af + ap - fp) - ch(a + f - 2p), \\ \mathfrak{P}_{312}^{\pm} &= \mp \left(a^2p + f^2p - (a + f)p^2 - 2c^2a - 2h^2f + 2(c^2 + h^2)p \right), \\ \mathfrak{P}_{313}^{\pm} &= \pm \left(h(2f^2 - p^2 + af - ap - fp) - bc(2a - f - p) \right), \\ \mathfrak{P}_{314}^{\pm} &= -c(2a^2 - p^2 + af - ap - fp) - bc(2a - f - p) \right), \\ \mathfrak{P}_{412}^{\pm} &= -b(a - f)^2, \\ \mathfrak{P}_{413}^{\pm} &= -c(a - p)^2, \\ \mathfrak{P}_{414}^{\pm} &= \mp h(f - p)^2. \end{split}$$

Consequently, the conditions $\delta W^+ = 0$ and $\delta W^- = 0$ are equivalent.

Remark 6.8. As in the previous remark, we will make a direct approach to show that the Weyl curvatures of $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$ cannot be strictly half-harmonic.

Let us consider left-invariant metrics on the product Lie groups $\widetilde{SL}(2,\mathbb{R}) \times \mathbb{R}$ and $SU(2) \times \mathbb{R}$ as described in Section 1.4.2. We recall that in this case there exists an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ of the Lie algebra $\mathfrak{sl}(2,\mathbb{R}) \times \mathbb{R}$ or $\mathfrak{su}(2) \times \mathbb{R}$ such that

$$[e_1, e_2] = \lambda_3 e_3, \qquad [e_1, e_3] = -\lambda_2 e_2, \qquad [e_2, e_3] = \lambda_1 e_1,$$

$$[e_1, e_4] = k_3 \lambda_2 e_2 - k_2 \lambda_3 e_3, \qquad [e_2, e_4] = k_1 \lambda_3 e_3 - k_3 \lambda_1 e_1,$$

$$[e_3, e_4] = k_2 \lambda_1 e_1 - k_1 \lambda_2 e_2,$$

(6.2)

where $\lambda_1 \lambda_2 \lambda_3 \neq 0$. The associated Lie group corresponds to $SU(2) \times \mathbb{R}$ if $\lambda_1, \lambda_2, \lambda_3$ do not change sign, and to $\widetilde{SL}(2, \mathbb{R}) \times \mathbb{R}$ otherwise.

Now, a long but straightforward calculation shows that the divergence $\operatorname{div}_1 W^+(e_i, e_j, e_k)$ is determined – up to the corresponding symmetries – by

$$\operatorname{div}_1 W^+(e_i, e_j, e_k) = \frac{1}{16} \mathfrak{P}^+_{ijk},$$

where \mathfrak{P}_{ijk}^+ are the polynomials on the structure constants in (6.2) given by

$$\mathfrak{P}_{112}^{+} = -2 \left(2\lambda_1^3 - \lambda_3^3 - \lambda_1^2\lambda_3 \right) k_2^3 + 2 \left(\lambda_3^3 - 2\lambda_1\lambda_2^2 + \lambda_1\lambda_2\lambda_3 \right) k_1^2 k_2 - \left(4\lambda_1^3 - \lambda_1^2(\lambda_2 + \lambda_3) - \lambda_2^2(\lambda_1 + 2\lambda_3) + \lambda_1\lambda_2\lambda_3 \right) k_2 k_3^2 + \lambda_3 \left(\lambda_1^2 - \lambda_2^2 - 3\lambda_1\lambda_2 + 4\lambda_1\lambda_3 - \lambda_2\lambda_3 \right) k_1 k_3$$

$$\begin{split} &- \left(4\lambda_1^3 - 2\lambda_3^3 - \lambda_1^2(\lambda_2 + 2\lambda_3) - \lambda_2^2(\lambda_1 - \lambda_3) + \lambda_2\lambda_3^2 \right) k_2, \\ \mathfrak{P}_{113}^+ &= -2 \left(2\lambda_1^3 - \lambda_2^3 - \lambda_1^2\lambda_2 \right) k_3^3 + 2 \left(\lambda_3^2 - 2\lambda_1\lambda_3^2 + \lambda_1\lambda_2\lambda_3 \right) k_1^2 k_3 \\ &- \left(4\lambda_1^3 - \lambda_1^2(\lambda_2 + \lambda_3) - \lambda_3^2(\lambda_1 + 2\lambda_2) + \lambda_1\lambda_2\lambda_3 \right) k_2^2 k_3 \\ &- \lambda_2 \left(\lambda_1^2 - \lambda_3^2 + 4\lambda_1\lambda_2 - 3\lambda_1\lambda_3 - \lambda_2\lambda_3 \right) k_1 k_2 \\ &- \left(4\lambda_1^3 - 2\lambda_3^2 - \lambda_1^2(2\lambda_2 + \lambda_3) + \lambda_2^2(\lambda_1 + 3\lambda_2) \right) k_1 k_2 k_3 \\ &+ 2 \left(\lambda_2^3 + \lambda_3^3 - \lambda_1^2(\lambda_2 + 2\lambda_3) + \lambda_2^2\lambda_3 \right) k_1^2 \\ &- \left(4\lambda_1^3 - 2\lambda_3^3 - \lambda_1^2(\lambda_2 + 2\lambda_3) + \lambda_2\lambda_3 \right) k_2^2 \\ &- \left(4\lambda_1^3 - 2\lambda_3^3 - \lambda_1^2(\lambda_2 + 2\lambda_3) + \lambda_2\lambda_3 \right) k_2^2 \\ &- \left(4\lambda_1^3 - 2\lambda_3^3 - \lambda_1^2(\lambda_2 + \lambda_3) + \lambda_2\lambda_3 \right) k_2^2 \\ &- \left(2\lambda_1^3 - \lambda_3^3 - \lambda_2\lambda_3 \right) k_1^3 - 2 \left(\lambda_3^3 - 2\lambda_1^2(\lambda_2 + \lambda_3) + \lambda_2\lambda_3 \right) k_1 k_2^2 \\ &+ \left(4\lambda_2^3 - \lambda_1^2(\lambda_2 + 2\lambda_3) - \lambda_2^2(\lambda_1 + \lambda_3) + \lambda_1\lambda_2\lambda_3 \right) k_1 k_2^2 \\ &+ \left(4\lambda_2^3 - \lambda_1^2(\lambda_2 + 2\lambda_3) - \lambda_2^2(\lambda_1 + \lambda_3) + \lambda_1\lambda_2\lambda_3 \right) k_1 k_2^2 \\ &- \lambda_3 \left(\lambda_1^2 - \lambda_2^2 + 3\lambda_1\lambda_2 + \lambda_1\lambda_3 - 4\lambda_2\lambda_3 \right) k_2 k_3 \\ &+ \left(4\lambda_2^3 - 2\lambda_3^3 - \lambda_1^2(\lambda_2 - \lambda_3) - \lambda_2^2(\lambda_1 + 2\lambda_3) + \lambda_1\lambda_3^2 \right) k_1, \\ \mathfrak{P}_{213}^+ = \left(\lambda_1^2 (4\lambda_2 + 3\lambda_3) + \lambda_2^2(\lambda_1 - \lambda_3) - \lambda_3^2(3\lambda_1 + 4\lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(4\lambda_2^3 - 2\lambda_3^3 - \lambda_1^2(\lambda_3 + \lambda_2^2(2\lambda_1 + \lambda_3) - \lambda_3^2(3\lambda_1 + 4\lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(2\lambda_1^3 - 2\lambda_2^3 + \lambda_1\lambda_2^2 \right) k_3^3 \\ &- 2 \left(\lambda_1^3 - 2\lambda_2^3 + \lambda_1\lambda_2^2 \right) k_3^3 \\ &- \left(\lambda_1^3 - 2\lambda_2^3 + \lambda_1\lambda_2^2 \right) k_2^3 \\ &+ \left(2\lambda_1^3 - 2\lambda_2^3 + \lambda_1\lambda_2^2 \right) k_3^2 \\ &+ 2 \left(\lambda_1^3 - 2\lambda_2^3 + \lambda_1\lambda_2 - \lambda_1\lambda_2^2 + \lambda_3^2(\lambda_1 + \lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(2\lambda_1^3 - 4\lambda_2^3 - \lambda_1^2(\lambda_2 + \lambda_3) - \lambda_3^2(\lambda_1 + \lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(2\lambda_1^3 - 4\lambda_3^3 - \lambda_1^2(\lambda_2 + \lambda_3) + \lambda_3^2(\lambda_1 + \lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(2\lambda_1^3 - 4\lambda_3^3 - \lambda_1^2(\lambda_2 + \lambda_3) + \lambda_3^2(\lambda_1 + \lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(2\lambda_1^3 - 4\lambda_3^3 - \lambda_1^2(\lambda_2 - \lambda_1) \lambda_2^2 + \lambda_3^2(\lambda_1 + \lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(2\lambda_1^3 - 4\lambda_3^3 - \lambda_1^2(\lambda_2 - \lambda_1) \lambda_2^2 + \lambda_3^2(\lambda_1 + \lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(2\lambda_1^3 - 2\lambda_3^3 + \lambda_1\lambda_2^2 + \lambda_3^2(\lambda_1 + \lambda_2) \right) k_1 k_2 k_3 \\ &+ \left(2\lambda_1^3 - 2\lambda_2^3 + \lambda_1\lambda_2 + \lambda_3^2 + \lambda_3^2 + \lambda_3^2(\lambda_1 + \lambda_2) \right)$$

$$\begin{split} &-\lambda_{1} \left(\lambda_{2}^{2}-\lambda_{3}^{2}+\lambda_{1}\lambda_{2}-4\lambda_{1}\lambda_{3}+3\lambda_{2}\lambda_{3}\right)k_{1}k_{3}\\ &-\left(2\lambda_{1}^{3}-4\lambda_{3}^{3}-\lambda_{2}\lambda_{1}^{2}-\lambda_{2}^{2}(\lambda_{1}-\lambda_{3})+\lambda_{3}^{2}(2\lambda_{1}+\lambda_{2})\right)k_{2},\\ &\mathfrak{P}_{412}^{+}=-2\left(\lambda_{1}^{3}+\lambda_{2}^{3}-\lambda_{1}^{2}\lambda_{2}-\lambda_{1}\lambda_{2}^{2}\right)k_{3}^{3}\\ &-\left(2\lambda_{2}^{3}-\lambda_{2}^{2}(\lambda_{1}+\lambda_{3})+\lambda_{3}^{2}(2\lambda_{1}-\lambda_{2})-\lambda_{1}\lambda_{2}\lambda_{3}\right)k_{1}^{2}k_{3}\\ &-\left(2\lambda_{1}^{3}-\lambda_{1}^{2}(\lambda_{2}+\lambda_{3})-\lambda_{3}^{2}(\lambda_{1}-2\lambda_{2})-\lambda_{1}\lambda_{2}\lambda_{3}\right)k_{2}^{2}k_{3}\\ &+\left(\lambda_{1}^{2}(3\lambda_{2}-\lambda_{3})-\lambda_{2}^{2}(3\lambda_{1}-\lambda_{3})-\lambda_{3}^{2}(\lambda_{1}-\lambda_{2})\right)k_{1}k_{2}\\ &-2\left(\lambda_{1}^{3}+\lambda_{3}^{3}-\lambda_{1}^{2}\lambda_{2}-\lambda_{1}\lambda_{2}^{2}\right)k_{3},\\ &\mathfrak{P}_{413}^{+}=2\left(\lambda_{1}^{3}+\lambda_{3}^{3}-\lambda_{1}^{2}\lambda_{3}-\lambda_{1}\lambda_{3}^{2}\right)k_{2}^{3}\\ &+\left(2\lambda_{3}^{3}+\lambda_{2}^{2}(2\lambda_{1}-\lambda_{3})-\lambda_{3}^{2}(\lambda_{1}+\lambda_{2})-\lambda_{1}\lambda_{2}\lambda_{3}\right)k_{1}^{2}k_{2}\\ &+\left(2\lambda_{1}^{3}-\lambda_{1}^{2}(\lambda_{2}+\lambda_{3})-\lambda_{2}^{2}(\lambda_{1}-2\lambda_{3})-\lambda_{1}\lambda_{2}\lambda_{3}\right)k_{2}k_{3}^{2}\\ &-\left(\lambda_{1}^{2}(\lambda_{2}-3\lambda_{3})+\lambda_{2}^{2}(\lambda_{1}-\lambda_{3})+\lambda_{3}^{2}(3\lambda_{1}-\lambda_{2})\right)k_{1}k_{3}\\ &+2\left(\lambda_{1}^{3}+\lambda_{3}^{3}-\lambda_{1}^{2}\lambda_{3}-\lambda_{1}\lambda_{3}^{2}\right)k_{2},\\ &\mathfrak{P}_{414}^{+}=-2\left(\lambda_{2}^{3}+\lambda_{3}^{3}-\lambda_{2}^{2}\lambda_{3}-\lambda_{2}\lambda_{3}^{2}\right)k_{1}^{3}\\ &-\left(2\lambda_{3}^{3}+\lambda_{1}^{2}(2\lambda_{2}-\lambda_{3})-\lambda_{3}^{2}(\lambda_{1}+\lambda_{2})-\lambda_{1}\lambda_{2}\lambda_{3}\right)k_{1}k_{2}^{2}\\ &-\left(2\lambda_{2}^{3}-\lambda_{1}^{2}(\lambda_{2}-2\lambda_{3})-\lambda_{3}^{2}(\lambda_{1}+\lambda_{2})-\lambda_{1}\lambda_{2}\lambda_{3}\right)k_{1}k_{3}^{2}\\ &-\left(2\lambda_{3}^{3}+\lambda_{1}^{2}(2\lambda_{2}-\lambda_{3})-\lambda_{3}^{2}(\lambda_{1}+\lambda_{2})-\lambda_{1}\lambda_{2}\lambda_{3}\right)k_{1}k_{3}^{2}\\ &-\left(2\lambda_{2}^{3}+\lambda_{3}^{3}-\lambda_{2}^{2}\lambda_{3}-\lambda_{2}\lambda_{3}^{2}\right)k_{1}. \end{split}$$

For its part, $\operatorname{div}_1 W^-(e_i,e_j,e_k)$ is given – up to the corresponding symmetries – by

$$\operatorname{div}_1 W^-(e_i, e_j, e_k) = \frac{1}{16} \mathfrak{P}^-_{ijk},$$

where \mathfrak{P}_{ijk}^- are polynomials on the structure constants in (6.2) that can be expressed in terms of the polynomials \mathfrak{P}_{ijk}^+ as follows.

$$\begin{split} \mathfrak{P}_{112}^{-} &= -\mathfrak{P}_{112}^{+} + 2k_1k_3\lambda_3 \left(\lambda_1^2 - \lambda_2^2 - 3\lambda_1\lambda_2 + 4\lambda_1\lambda_3 - \lambda_2\lambda_3\right), \\ \mathfrak{P}_{113}^{-} &= -\mathfrak{P}_{113}^{+} - 2k_1k_2\lambda_2 \left(\lambda_1^2 - \lambda_3^2 + 4\lambda_1\lambda_2 - 3\lambda_1\lambda_3 - \lambda_2\lambda_3\right), \\ \mathfrak{P}_{114}^{-} &= -\mathfrak{P}_{114}^{+} + 2k_1k_2k_3(\lambda_2 - \lambda_3) \left(\lambda_1^2 + 4\lambda_1\lambda_2 + 4\lambda_1\lambda_3 + 3\lambda_2\lambda_3\right), \\ \mathfrak{P}_{212}^{-} &= -\mathfrak{P}_{212}^{+} - 2k_2k_3\lambda_3 \left(\lambda_1^2 - \lambda_2^2 + 3\lambda_1\lambda_2 + \lambda_1\lambda_3 - 4\lambda_2\lambda_3\right), \\ \mathfrak{P}_{213}^{-} &= \mathfrak{P}_{213}^{+} - 2k_1k_2k_3(\lambda_1 - \lambda_3) \left(\lambda_2^2 + 4\lambda_1\lambda_2 - \lambda_1\lambda_3 - 4\lambda_2\lambda_3\right), \\ \mathfrak{P}_{214}^{-} &= \mathfrak{P}_{214}^{+} - 2k_1k_2\lambda_1 \left(\lambda_2^2 - \lambda_3^2 + 4\lambda_1\lambda_2 - \lambda_1\lambda_3 - 3\lambda_2\lambda_3\right), \\ \mathfrak{P}_{312}^{-} &= \mathfrak{P}_{312}^{+} - 2k_1k_2k_3(\lambda_1 - \lambda_2) \left(\lambda_3^2 + 3\lambda_1\lambda_2 + 4\lambda_1\lambda_3 + 4\lambda_2\lambda_3\right), \\ \mathfrak{P}_{313}^{-} &= -\mathfrak{P}_{313}^{+} + 2k_2k_3\lambda_2 \left(\lambda_1^2 - \lambda_3^2 + \lambda_1\lambda_2 - 4\lambda_1\lambda_3 + 3\lambda_2\lambda_3\right), \\ \mathfrak{P}_{314}^{-} &= \mathfrak{P}_{314}^{+} + 2k_1k_3\lambda_1 \left(\lambda_2^2 - \lambda_3^2 + \lambda_1\lambda_2 - 4\lambda_1\lambda_3 + 3\lambda_2\lambda_3\right), \\ \mathfrak{P}_{412}^{-} &= \mathfrak{P}_{413}^{+} + 2k_1k_3(\lambda_1 - \lambda_2) \left(\lambda_3^2 - 3\lambda_1\lambda_2 + \lambda_1\lambda_3 + \lambda_2\lambda_3\right), \\ \mathfrak{P}_{413}^{-} &= \mathfrak{P}_{414}^{+} - 2k_2k_3(\lambda_2 - \lambda_3) \left(\lambda_2^2 + \lambda_1\lambda_2 - 3\lambda_1\lambda_3 + \lambda_2\lambda_3\right), \\ \mathfrak{P}_{414}^{-} &= -\mathfrak{P}_{414}^{+} - 2k_2k_3(\lambda_2 - \lambda_3) \left(\lambda_1^2 + \lambda_1\lambda_2 - 3\lambda_1\lambda_3 - 3\lambda_2\lambda_3\right). \end{split}$$

Since $\lambda_1 \neq 0$, we can consider the orthogonal basis $\hat{e}_i = \frac{1}{\lambda_1} e_i$ and assume that $\lambda_1 = 1$ for the rest of our calculations, just working in the homothetic class of the initial metric. In what follows, we will make use of Gröbner bases (see Section 1.6 and the use we make of them in the following sections for a better understanding). In particular, we consider the polynomial ring $\mathbb{R}[\lambda_2, \lambda_3, k_1, k_2, k_3]$ and fix the graded reverse lexicographical order to compute a Gröbner basis of the ideal generated by the polynomials \mathfrak{P}_{ijk}^+ . In this way, we obtain that the polynomials

$$\begin{aligned} \mathbf{g}_1 &= k_1 (k_1^2 + k_2^2 + k_3^2 + 1)^2 (\lambda_2 - \lambda_3), \\ \mathbf{g}_2 &= (k_2^2 + k_3^2 + 1) (k_1^2 + k_2^2 + k_3^2 + 1) (k_1 (4\lambda_2 - 5\lambda_3 + 1) + 3k_2 k_3 (\lambda_3 - 1)), \\ \mathbf{g}_3 &= (k_2^2 + k_3^2 + 1) (k_1^2 + k_2^2 + k_3^2 + 1) (3k_1 (\lambda_2 - \lambda_3) + k_2 k_3 (\lambda_2 + \lambda_3 - 2)), \end{aligned}$$

belong to the ideal. Note that the expressions for \mathfrak{P}_{ijk}^- given above imply that the space cannot be strictly half-harmonic if any two of the constants k_1 , k_2 and k_3 vanish. Hence, excluding that case, the vanishing of the three polynomials \mathbf{g}_1 , \mathbf{g}_2 and \mathbf{g}_3 clearly leads to $\lambda_1 = \lambda_2 = \lambda_3 = 1$, so again \mathfrak{P}_{ijk}^- coincides with \mathfrak{P}_{ijk}^+ – up to a sign in some cases –, so the conditions $\delta \mathcal{W}^+ = 0$ and $\delta \mathcal{W}^- = 0$ are equivalent.

6.2.1 Half-harmonic Weyl curvature on $H^3 \rtimes \mathbb{R}$

We follow the description of the left-invariant metrics on semi-direct extensions of the Heisenberg group given in Section 1.4.2. We fix an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ such that

$$[e_1, e_2] = \gamma e_3, \qquad [e_1, e_4] = ae_1 - ce_2 + He_3, [e_3, e_4] = (a+d)e_3, \quad [e_2, e_4] = ce_1 + de_2 + Fe_3,$$
(6.3)

where $\gamma \neq 0$.

For a given orthonormal basis $\{e_1, e_2, e_3, e_4\}$ on the Lie algebra \mathfrak{g} , we consider the orientation induced by the volume form $\omega = e^1 \wedge e^2 \wedge e^3 \wedge e^4$ and denote by $\{E_i^{\pm}\}$ the corresponding orthonormal basis of the spaces of self-dual and anti-self-dual two-forms $\Lambda^2_{\pm}(\mathfrak{g})$ given by

$$E_1^{\pm} = \frac{1}{\sqrt{2}} \left(e^{12} \pm e^{34} \right), \quad E_2^{\pm} = \frac{1}{\sqrt{2}} \left(e^{13} \mp e^{24} \right), \quad E_3^{\pm} = \frac{1}{\sqrt{2}} \left(e^{14} \pm e^{23} \right),$$

where we are using the notation $e^{ij} = e^i \wedge e^j$ and $\{e^i\}$ is the dual basis of $\{e_i\}$.

The Levi-Civita connection ∇ is determined by (6.3) and the expressions

$$\nabla_{e_1}e_1 = -a e_4, \qquad \nabla_{e_1}e_2 = \frac{1}{2}\gamma e_3,
\nabla_{e_1}e_3 = -d e_4, \qquad \nabla_{e_1}e_4 = a e_1 + \frac{1}{2}H e_3,
\nabla_{e_2}e_2 = -d e_4, \qquad \nabla_{e_2}e_3 = \frac{1}{2}(\gamma e_1 - F e_4), \qquad (6.4)
\nabla_{e_2}e_4 = d e_2 + \frac{1}{2}F e_3, \qquad \nabla_{e_3}e_3 = -(a + d) e_4,
\nabla_{e_3}e_4 = \frac{1}{2}(H e_1 + F e_2 + 2(a + d) e_3), \quad \nabla_{e_4}e_4 = 0.$$

The self-dual and anti-self-dual Weyl curvature operators have associated matrices (\mathcal{W}_{ij}^{\pm}) whose components $\mathcal{W}_{ij}^{\pm} = \langle \mathcal{W}^{\pm}(E_j^{\pm}), E_i^{\pm} \rangle = \langle \mathcal{W}(E_j^{\pm}), E_i^{\pm} \rangle$ are determined by

$$\begin{aligned} \mathcal{W}_{11}^{\pm} &= -\frac{1}{6} \left(4ad \pm 3\gamma(a+d) + 2\gamma^2 - F^2 - H^2 \right), \\ \mathcal{W}_{12}^{\pm} &= \frac{1}{4} \left(Fa + Hc + 2Fd \pm 2\gamma F \right), \qquad \mathcal{W}_{13}^{\pm} = \mp \frac{1}{4} \left(2Ha - Fc + Hd \pm 2\gamma H \right), \\ \mathcal{W}_{22}^{\pm} &= \frac{1}{6} \left(2ad \pm 3\gamma d + \gamma^2 - 2F^2 + H^2 \right), \qquad \mathcal{W}_{23}^{\pm} = \pm \frac{1}{2} \left(ac - cd + FH \right). \end{aligned}$$

The components of the self-dual and anti-self-dual Weyl curvature tensors are given by

$$W^{\pm}(e_i, e_j, e_k, e_\ell) = \frac{1}{2} \{ W(e_i, e_j, e_k, e_\ell) \pm W(e_i, e_j, e_{\bar{k}}, e_{\bar{\ell}}) \}$$

where $e_{\bar{k}}, e_{\bar{\ell}}$ are such that $e^k \wedge e^{\ell} \wedge e^{\bar{k}} \wedge e^{\bar{\ell}} = e^1 \wedge e^2 \wedge e^3 \wedge e^4$ and $\{e_1, e_2, e_3, e_4\}$ is a positively oriented orthonormal basis. The non-zero components of the self-dual Weyl curvature tensor $W^+_{ijk\ell} = W^+(e_i, e_j, e_k, e_\ell)$ are determined by

$$\begin{split} W_{1212}^+ &= W_{1234}^+ = W_{3434}^+ = \frac{1}{2}\mathcal{W}_{11}^+, \\ W_{1313}^+ &= -W_{1324}^+ = W_{2424}^+ = \frac{1}{2}\mathcal{W}_{22}^+, \\ W_{1213}^+ &= -W_{1224}^+ = W_{1334}^+ = -W_{2434}^+ = \frac{1}{2}\mathcal{W}_{12}^+, \\ W_{1214}^+ &= W_{1223}^+ = W_{1434}^+ = W_{2334}^+ = \frac{1}{2}\mathcal{W}_{13}^+, \\ W_{1314}^+ &= W_{1323}^+ = -W_{1424}^+ = -W_{2324}^+ = \frac{1}{2}\mathcal{W}_{23}^+, \\ W_{1414}^+ &= W_{1423}^+ = W_{2323}^+ = -\frac{1}{2}\left(\mathcal{W}_{11}^+ + \mathcal{W}_{22}^+\right), \end{split}$$

up to the corresponding symmetries. The divergence

$$\operatorname{div}_{1} W^{+}(e_{i}, e_{j}, e_{k}) = \sum_{\alpha=1}^{4} \left(\nabla_{e_{\alpha}} W^{+} \right) \left(e_{\alpha}, e_{i}, e_{j}, e_{k} \right)$$

is determined – up to the corresponding symmetries – by

$$\operatorname{div}_1 W^+(e_i, e_j, e_k) = \frac{1}{16} \mathfrak{P}_{ijk},$$

where \mathfrak{P}_{ijk} are polynomials on the structure constants in equation (6.3). For instance, using the Levi-Civita connection (6.4) and the expressions for the components $W^+_{ijk\ell}$ above, one obtains

$$\begin{split} \mathfrak{P}_{112} &= 16 \sum_{\alpha=1}^{4} \left(\nabla_{e_{\alpha}} W^{+} \right) \left(e_{\alpha}, e_{1}, e_{1}, e_{2} \right) \\ &= \left(4Hd^{2} + 8Had - 4Fac - 4\gamma Ha + 2\gamma Fc + 6\gamma Hd - 4\gamma^{2} H \right) \\ &+ \left(4Ha^{2} + 2Hd^{2} - 4Fac + 2Had + 4\gamma Ha - \gamma Fc + \gamma Hd \right) \\ &- \left(2Hc^{2} + 4Fac - 4Had + 2Fcd - 4\gamma Ha + 4\gamma Fc - 2\gamma Hd \right) \\ &+ 2H^{3} + 2F^{2}H - 2\gamma^{2}H \right) \\ &= 2H \left(2a^{2} - c^{2} + 3d^{2} \right) - 8Fac + 14Had - 6FcD + 4\gamma Ha \\ &- 3\gamma Fc + 9\gamma Hd - 2H \left(F^{2} + H^{2} + \gamma^{2} \right). \end{split}$$

It is straightforward to see that the remaining components are given by

$$\begin{split} \mathfrak{P}_{113} &= -4(2a^2c - 3cd^2 + acd) + 2\gamma(ac - dc) - 7FHa + (4F^2 - 3H^2)c - 10FHd, \\ \mathfrak{P}_{114} &= -8(ac^2 + ad^2 - c^2d) - 4\gamma(a^2 + ad) - 2(F^2 - 3H^2 - \gamma^2)a - 7FHc + 9H^2d \\ &+ 2\gamma(F^2 + H^2 + \gamma^2), \\ \mathfrak{P}_{212} &= 2F(3a^2 - c^2 + 2d^2) + 6Hac + 14Fad + 8Hcd + 9\gamma Fa + 3\gamma Hc + 4\gamma Fd \\ &- 2F(F^2 + H^2 + \gamma^2), \\ \mathfrak{P}_{213} &= 8(a^2d - ac^2 + c^2d) + 4\gamma(d^2 + ad) - 9F^2a - 7FHc - 2(3F^2 - H^2 + \gamma^2)d \\ &- 2\gamma(F^2 + H^2 + \gamma^2), \\ \mathfrak{P}_{214} &= 4(3a^2c - 2cd^2 - acd) - 2\gamma(ac - cd) + 10FHa + (4H^2 - 3F^2)c + 7FHd, \\ \mathfrak{P}_{312} &= 8(a^2d + ad^2) + 4\gamma(a^2 + d^2 + 2ad) - (7F^2 + 6H^2 + 2\gamma^2)a \\ &- (6F^2 + 7H^2 + 2\gamma^2)d - 4\gamma(F^2 + H^2 + \gamma^2), \\ \mathfrak{P}_{313} &= -2F(2a^2 - c^2 + 2d^2) - 2Hac - 14Fad - 10Hcd - 5\gamma Fa - \gamma Hc - 4\gamma Fd \\ &+ 4F(F^2 + H^2 + \gamma^2), \\ \mathfrak{P}_{412} &= 4(a^2c + cd^2 - 2acd) + 3FHa + (F^2 + H^2)c - 3FHd, \\ \mathfrak{P}_{413} &= -2Hd^2 - 2Fac + 4Fcd + 2\gamma Fc - 4\gamma Hd - 2H(F^2 + H^2 + \gamma^2), \\ \mathfrak{P}_{414} &= -2Fa^2 - 2H(2ac - cd) - 4\gamma Fa - 2\gamma Hc - 2F(F^2 + H^2 + \gamma^2). \\ \end{split}$$

Since $\gamma \neq 0$, we can consider the orthogonal basis $\hat{e}_i = \frac{1}{\gamma} e_i$ and work in the homothetic class of the initial metric assuming $\gamma = 1$. The divergence δW^+ vanishes if and only if the

structure constants in Equation (6.3) satisfy the system of polynomial equations $\{\mathfrak{P}_{ijk} = 0\}$, where $\mathfrak{P}_{ijk} \in \mathbb{R}[a, c, d, F, H]$. Let $\mathcal{I} \subset \mathbb{R}[a, c, d, F, H]$ be the ideal generated by the polynomials \mathfrak{P}_{ijk} . Fixing a monomial order on the polynomial ring $\mathbb{R}[a, c, d, F, H]$, a Gröbner basis of \mathcal{I} is a finite subset $\mathcal{G} = \{\mathbf{g}_1, \dots, \mathbf{g}_r\}$ such that the leading term of any element of \mathcal{I} is divisible by the leading term of one of the \mathbf{g}_i with respect to the given ordering. The Hilbert Basis Theorem guarantees that any non-zero ideal admits a Gröbner basis and Buchberguer's algorithm, among others, provides a constructive strategy to find one such basis (see Section 1.6). The zero-set of $\{\mathfrak{P}_{ijk} = 0\}$ and the roots of $\mathcal{I} = \langle \mathfrak{P}_{ijk} \rangle = \langle \mathcal{G} \rangle$ coincide, and \mathcal{G} can be considered a *good* basis for our purposes if it contains one or more polynomials for which we can determine all the real roots in terms of simple conditions on the structure constants in Equation (6.3).

Computing a Gröbner basis \mathcal{G} of $\mathcal{I} = \langle \mathfrak{P}_{ijk} \rangle$ with respect to the lexicographical order, we get a basis of 29 polynomials, one of which is

$$\mathbf{g}_{1} = H \left(F^{2} + H^{2} + 1 \right) \left(4H^{2} + 1 \right) \left(4H^{2} + 81 \right) \left(64H^{2} + 81 \right) \\ \times \left(\left(F^{2} - H^{2} \right)^{2} + F^{2} + H^{2} \right) \left(233317175H^{4} + 255147165H^{2} + 69861528 \right).$$

Consequently, H = 0 and the polynomials \mathfrak{P}_{414} and \mathfrak{P}_{412} are reduced to

$$\mathfrak{P}_{414} = -2F\left(\left(a+1\right)^2 + F^2\right),\\ \mathfrak{P}_{412} = c\left(4\left(a-d\right)^2 + F^2\right).$$

Therefore, F = 0 and there are two different possibilities depending on whether or not d = a.

If d = a, then $\mathfrak{P}_{114} = -2(a+1)(2a-1)(2a+1)$, from where we are led to the cases a = -1 or $a = \pm \frac{1}{2}$.

If $a = \pm \frac{1}{2}$, the metric is Einstein and therefore locally symmetric [88] and so this case does not provide strictly half-harmonic examples.

On the other hand, if a = -1, then the associated left-invariant metric

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = -e_1 - \kappa e_2, \quad [e_2, e_4] = \kappa e_1 - e_2, \quad [e_3, e_4] = -2e_3$$

is anti-self-dual and not locally symmetric, thus corresponding to that previously obtained by de Smedt and Salamon [55]. A straightforward calculation shows that the sectional curvature of the left-invariant metrics described above does not depend on the real parameter κ . Consequently, the corresponding metric Lie groups are homothetic (see [96]), although not isomorphically homothetic. Assertion (i) in Theorem 6.1 now follows.

If $d \neq a$, then c = 0 and

$$\mathfrak{P}_{114} = -2 \left(4ad^2 + 2a(a+d) - a - 1 \right),$$

$$\mathfrak{P}_{312} = 2 \left(4ad(a+d) + 2(a+d)^2 - (a+d) - 2 \right)$$

We consider $\mathfrak{P}_{114} + \frac{1}{2}\mathfrak{P}_{312} = (a-d)(2a-1)(2d-1)$, so either $a = \frac{1}{2}$ (and then \mathfrak{P}_{114} shows that d = -1) or $d = \frac{1}{2}$ (in which case \mathfrak{P}_{114} shows that a = -1). Now the two cases above are equivalent through the transformation

$$(e_1, e_2, e_3, e_4) \mapsto (e_2, -e_1, e_3, e_4).$$

A straightforward calculation now shows that the resulting left-invariant metric

$$[e_1, e_2] = e_3, \quad [e_1, e_4] = \frac{1}{2}e_1, \quad [e_2, e_4] = -e_2, \quad [e_3, e_4] = -\frac{1}{2}e_3,$$

satisfies $\delta W^+ = 0$ and $\delta W \neq 0$. This corresponds to Assertion (ii) in Theorem 6.1.

Remark 6.9. In a completely analogous way, left-invariant metrics on $H^3 \rtimes \mathbb{R}$ with $\delta W^- = 0$ are in correspondence with the ones described above.

6.2.2 Half-harmonic Weyl curvature on $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$

In this section we will proceed as in the previous one and we will see that there are no leftinvariant metrics with strictly half-harmonic Weyl curvature on $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$.

Next we show that any left-invariant metric with half-harmonic Weyl tensor on $E(1,1) \rtimes \mathbb{R}$ or $\widetilde{E}(2) \rtimes \mathbb{R}$ satisfies $\delta W = 0$, thus being necessarily symmetric. We proceed as in the previous section.

We follow the description of the left-invariant metrics on semi-direct extensions of the Poincaré and the Euclidean groups as in Section 1.4.2. We consider an orthonormal basis $\{e_1, e_2, e_3, e_4\}$ such that

$$[e_1, e_3] = -\lambda_2 e_2, \qquad [e_2, e_3] = \lambda_1 e_1,$$

$$[e_1, e_4] = b e_1 - A \lambda_2 e_2, \qquad [e_2, e_4] = A \lambda_1 e_1 + b e_2,$$

$$[e_3, e_4] = C e_1 + D e_2,$$

(6.5)

where $\lambda_1 \lambda_2 \neq 0$. The associated Lie group corresponds to $\tilde{E}(2) \times \mathbb{R}$ if λ_1 , λ_2 do not change sign, and corresponds to $E(1, 1) \times \mathbb{R}$ otherwise.

The self-dual and anti-self-dual Weyl curvature operators have associated matrices (\mathcal{W}^{\pm}) whose components $\mathcal{W}_{ij}^{\pm} = \langle \mathcal{W}^{\pm}(E_j^{\pm}), E_i^{\pm} \rangle = \langle \mathcal{W}(E_j^{\pm}), E_i^{\pm} \rangle$ are determined by

$$\begin{split} \mathcal{W}_{11}^{\pm} &= \frac{1}{6} \left\{ \left(A^2 + 1 \right) \left(\lambda_1 - \lambda_2 \right)^2 + 2 \left(C^2 + D^2 \right) \right\}, \\ \mathcal{W}_{12}^{\pm} &= \frac{1}{4} \left\{ \left(2AC \mp D \right) \lambda_1 + \left(AC \mp 2D \right) \lambda_2 + bD \right\}, \\ \mathcal{W}_{13}^{\pm} &= \mp \frac{1}{4} \left\{ \left(AD \pm 2C \right) \lambda_1 - \left(2AD \pm C \right) \lambda_2 + bC \right\}, \\ \mathcal{W}_{22}^{\pm} &= \frac{1}{6} \left\{ \left(2A^2 - 1 \right) \lambda_1^2 - \left(A^2 - 2 \right) \lambda_2^2 - \left(A^2 + 1 \right) \lambda_1 \lambda_2 \mp 3b \left(\lambda_1 - \lambda_2 \right) - C^2 - D^2 \right\}, \\ \mathcal{W}_{23}^{\pm} &= \frac{1}{2} \left\{ A \left(\lambda_1^2 - \lambda_2^2 \right) \pm Ab \left(\lambda_1 - \lambda_2 \right) \right\}. \end{split}$$

Proceeding as in the previous section, a long but standard calculation shows that the nonzero components of the divergence $\operatorname{div}_1 W^+$ (up the corresponding symmetries) are given by $\operatorname{div}_1 W^+(e_i, e_j, e_k) = \frac{1}{16} \mathfrak{P}_{ijk}$, where \mathfrak{P}_{ijk} are the following polynomials on the structure constants in Equation (6.5):

$$\begin{split} \mathfrak{P}_{112} &= -(A^2+1)C(4\lambda_1^2-\lambda_2^2-\lambda_1\lambda_2) + b(AD+5C)\lambda_1 - b(7AD+3C)\lambda_2 \\ &+ 4C(b^2-C^2-D^2), \\ \mathfrak{P}_{113} &= 2A(A^2+1)(2\lambda_1^3-\lambda_2^3-\lambda_1^2\lambda_2) - 4Ab(\lambda_1^2-\lambda_2^2) - (4Ab^2-4AC^2-CD)\lambda_1 \\ &+ (4Ab^2-AC^2-2AD^2+4CD)\lambda_2 + 5bCD, \\ \mathfrak{P}_{114} &= -2(A^2+1)(2\lambda_1^3-\lambda_2^3-\lambda_1^2\lambda_2) - 2b(3A^2+1)(\lambda_1^2-\lambda_2^2) + (ACD+4b^2-4C^2)\lambda_1 \\ &+ (4ACD-4b^2+C^2+2D^2)\lambda_2 - b(7C^2+2D^2), \\ \mathfrak{P}_{212} &= (A^2+1)D(\lambda_1^2-4\lambda_2^2+\lambda_1\lambda_2) + b(7AC-3D)\lambda_1 - b(AC-5D)\lambda_2 \\ &+ 4D(b^2-C^2-D^2), \\ \mathfrak{P}_{213} &= -2(A^2+1)(\lambda_1^3-2\lambda_2^3+\lambda_1\lambda_2^2) - 2b(3A^2+1)(\lambda_1^2-\lambda_2^2) \\ &+ (4ACD+4b^2-2C^2-D^2)\lambda_1 + (ACD-4b^2+4D^2)\lambda_2 + b(2C^2+7D^2), \\ \mathfrak{P}_{214} &= -2A(A^2+1)(\lambda_1^3-2\lambda_2^3+\lambda_1\lambda_2^2) + 2b(3A^2+1)(\lambda_1^2-\lambda_2^2) \\ &+ (4Ab^2-2AC^2-AD^2-4CD)\lambda_1 - (4Ab^2-4AD^2+CD)\lambda_2 - 5bCD, \\ \mathfrak{P}_{312} &= 2(A^2+1)(\lambda_1^3+\lambda_2^3-\lambda_1^2\lambda_2-\lambda_1\lambda_2^2) + (3ACD+2C^2-D^2)\lambda_1 \\ &- (3ACD+C^2-2D^2)\lambda_2 + 9b(C^2+D^2), \\ \mathfrak{P}_{313} &= -D\lambda_1^2 + (2A^2D-AC+2D)\lambda_2^2 - (AC+D)\lambda_1\lambda_2 - b(10AC-3D)\lambda_1 \\ &+ b(2AC-9D)\lambda_2 - 2D(3b^2-C^2-D^2), \\ \mathfrak{P}_{314} &= -(2A^2C+AD+2C)\lambda_1^2 + C\lambda_2^2 - (AD-C)\lambda_1\lambda_2 + b(2AD+9C)\lambda_1 \\ &- b(10AD+3C)\lambda_2 + 2C(3b^2-C^2-D^2), \\ \mathfrak{P}_{413} &= (2A^2C-AD+2C)\lambda_1^2 - A^2C\lambda_2^2 - A(AC+D)\lambda_1\lambda_2 + b(AD+4C)\lambda_1 \\ &- 3AbD\lambda_2 + 2C(b^2+C^2+D^2), \\ \mathfrak{P}_{414} &= -A^2D\lambda_1^2 + (2A^2D+AC+2D)\lambda_2^2 - A(AC+D)\lambda_1\lambda_2 + b(AD+4C)\lambda_1 \\ &- 3AbD\lambda_2 + 2C(b^2+C^2+D^2), \\ \mathfrak{P}_{414} &= -A^2D\lambda_1^2 + (2A^2D+AC+2D)\lambda_2^2 - A(AC+D)\lambda_1\lambda_2 + b(AD+4C)\lambda_1 \\ &- b(AC-4D)\lambda_2 + 2D(b^2+C^2+D^2). \\ \end{array}$$

Since $\lambda_1 \lambda_2 \neq 0$, we consider the orthonormal basis $\hat{e}_i = \frac{1}{\lambda_1} e_i$ and work in the homothetic class of the initial metric. This way, we can assume that $\lambda_1 = 1$ in what follows. The divergence δW^+ vanishes if and only if the structure constants in equation (6.5) satisfy the system of polynomial equations $\{\mathfrak{P}_{ijk} = 0\}$, where $\mathfrak{P}_{ijk} \in \mathbb{R} [\lambda_2, A, b, C, D]$. Let $\mathcal{I}_1 \subset \mathbb{R} [\lambda_2, A, b, C, D]$ be the ideal generated by the polynomials \mathfrak{P}_{ijk} . We compute a Gröbner basis \mathcal{G}_1 of \mathcal{I}_1 with respect to the lexicographical order and we see that one of the 39 polynomials in it is

$$\mathbf{g}_{1} = D^{5} \left(C^{2} + D^{2} \right) \left(49D^{2} + 16 \right) \left(49D^{2} + 144 \right) \left(193600D^{4} + 16560D^{2} + 2187 \right).$$

It follows immediately that D = 0. Now we compute a Gröbner basis \mathcal{G}_2 of the ideal \mathcal{I}_2 generated by the polynomials $\mathcal{G}_1 \cup \{D\} \subset \mathbb{R}[\lambda_2, A, b, C, D]$ with respect to the lexicographical order, obtaining that the polynomial

$$\mathbf{g}_2 = C \left(A^2 + (b+1)^2 + C^2 \right)$$

belongs to \mathcal{G}_2 . Therefore, C = 0, and the polynomial \mathfrak{P}_{312} is reduced to

$$\mathfrak{P}_{312} = 2 \left(A^2 + 1 \right) \left(\lambda_2 - 1 \right)^2 \left(\lambda_2 + 1 \right).$$

This leads to two different possibilities depending on whether $\lambda_2 = 1$ or $\lambda_2 = -1$.

If $\lambda_2 = 1$, then the left-invariant metric is locally conformally flat. Moreover, it is flat or locally isometric to a product $\mathbb{R} \times N(c)$, where N(c) is a three-dimensional manifold of constant sectional curvature.

If $\lambda_2 = -1$, the polynomial \mathfrak{P}_{114} is reduced to

$$\mathfrak{P}_{114} = 8 \left(b^2 - A^2 - 1 \right),$$

so $b = \pm \sqrt{A^2 + 1}$. In this situation, the corresponding left-invariant metric is Einstein (thus locally symmetric [88]) if A = 0 and locally isometric to a product $M_1(c_1) \times M_2(c_2)$ of two surfaces of constant curvature $c_1^2 \neq c_2^2$ otherwise.

The Weyl tensor is divergence-free in all the cases above, which shows that there are no non-trivial examples in this case.

Remark 6.10. Proceeding in a completely analogous way, one gets that the metrics with half-harmonic Weyl conformal tensor corresponding to the condition $\delta W^- = 0$ are again the ones described just above.

Chapter 7 Three-dimensional Riemannian homogeneous structures

In this chapter we will devote ourselves to the study of homogeneous structures and give a complete classification of the homogeneous structures on non-symmetric three-dimensional Riemannian Lie groups. We will see that one such group admits a non-canonical homogenous structure if and only if its isometry group is four-dimensional. The results in this chapter are contained in the work [38].

Before we start, we will make a short introduction to the world of homogeneous structures.

7.1 Homogeneous structures

The characterization of Riemannian locally symmetric spaces as those whose curvature is parallel with respect to the Levi-Civita connection was originally given by Cartan. Ambrose and Singer extended this characterization to homogeneous Riemannian manifolds showing that a connected, complete and simply connected *n*-dimensional Riemannian manifold (M, g) is homogeneous if and only if there exists a (1, 2)-tensor field T on M such that

$$\widetilde{\nabla}g = 0, \qquad \widetilde{\nabla}R = 0, \qquad \widetilde{\nabla}T = 0,$$
(7.1)

where $\widetilde{\nabla}$ is the *Ambrose-Singer connection* given by $\widetilde{\nabla} = \nabla - T$, ∇ is the Levi-Civita connection of the metric g, and R denotes the Riemannian curvature tensor for which we adopt the sign convention $R(X, Y) = \nabla_{[X,Y]} - [\nabla_X, \nabla_Y]$ (see [3]).

The tensor field T is said to be a *homogeneous structure* on M. We will also denote by T the associated (0,3)-tensor field given by T(X,Y,Z) = g(T(X,Y),Z). Conditions (7.1) were further investigated by Tricerri and Vanhecke in [130], where they considered the space $\mathcal{T}(\mathcal{V})$ of such tensor fields on a vector space $(\mathcal{V}, \langle \cdot, \cdot \rangle)$ and decomposed it into three irreducible components under the action of the orthogonal group as $\mathcal{T}(\mathcal{V}) = \mathcal{T}_1(\mathcal{V}) \oplus \mathcal{T}_2(\mathcal{V}) \oplus \mathcal{T}_3(\mathcal{V})$. The subspaces of such decomposition are

$$\begin{aligned} \mathcal{T}_1(\mathcal{V}) &= \{ T \in \mathcal{T}(\mathcal{V}) \colon T(x, y, z) = \langle x, y \rangle \varphi(z) - \langle x, z \rangle \varphi(y) \} \\ \mathcal{T}_2(\mathcal{V}) &= \{ T \in \mathcal{T}(\mathcal{V}) \colon c_{12}(T) = 0, \ \sigma_{x, y, z} T(x, y, z) = 0 \}, \\ \mathcal{T}_3(\mathcal{V}) &= \{ T \in \mathcal{T}(\mathcal{V}) \colon T(x, y, z) + T(y, x, z) = 0 \}, \end{aligned}$$

where $\varphi \in \mathcal{V}^*$, $\sigma_{x,y,z}$ is the cyclic sum with respect to x, y, z and $c_{12}(T)$ denotes the contraction

$$c_{12}(T)(z) = \sum_{i} T(e_i, e_i, z)$$

for an arbitrary orthonormal basis $\{e_i\}$ of \mathcal{V} . The projections of a homogeneous structure T on each of these subspaces are given by

$$p_{1}(T)(x, y, z) = \frac{1}{n-1} \langle x, y \rangle c_{12}(T)(z) - \frac{1}{n-1} \langle x, z \rangle c_{12}(T)(y),$$

$$p_{3}(T)(x, y, z) = \frac{1}{3} \sigma_{x,y,z} T(x, y, z),$$

$$p_{2}(T)(x, y, z) = (T - p_{1}(T) - p_{3}(T))(x, y, z).$$
(7.2)

Homogeneous manifolds admitting a homogeneous structure in one of the eight different classes induced by the decomposition above have been extensively studied in the literature. It was shown in [130] that naturally reductive spaces correspond to non-vanishing homogeneous structures of type \mathcal{T}_3 and that a Riemannian manifold admits a non-vanishing structure of type \mathcal{T}_1 if and only if it is locally isometric to the real hyperbolic space. The latter also holds true for homogeneous structures of type $\mathcal{T}_1 \oplus \mathcal{T}_3$, $T \notin \mathcal{T}_1$ and $T \notin \mathcal{T}_3$, in dimension greater than three, as shown in [117]. Riemannian manifolds of dimension less than or equal to four that admit a homogeneous structure of type \mathcal{T}_2 were described in [93] (see also [34]). Homogeneous structures in the class $\mathcal{T}_1 \oplus \mathcal{T}_2$ in dimension less than or equal to four were described in [73], and those in this class whose fundamental one-form is closed were investigated in [118]. It was shown in [93] that a three-dimensional non-symmetric space admitting a homogeneous structure of type \mathcal{T}_3 also admits a \mathcal{T}_2 -structure.

In dimension two, the decomposition above reduces to $\mathcal{T}(\mathcal{V}) = \mathcal{T}_1(\mathcal{V})$. As a consequence, a surface admits a non-zero homogeneous structure if and only if it is isometric to the hyperbolic plane. Dimension three is particularly relevant to the study of homogeneous spaces. First, it is the lowest possible dimension admitting locally homogeneous metrics which are not locally symmetric and, secondly, any three-dimensional homogeneous manifold is either symmetric or locally isometric to a Lie group endowed with a left-invariant metric [126].

The special case in which (M, g) is a Lie group G equipped with a left-invariant metric $\langle \cdot, \cdot \rangle$ is of special interest for our purposes. Let T^{∇} be the *canonical homogeneous structure* defined by

$$2\langle T^{\nabla}(X,Y),Z\rangle = \langle [X,Y],Z\rangle - \langle [Y,Z],X\rangle + \langle [Z,X],Y\rangle,$$

for left-invariant vector fields X, Y and Z. Then the corresponding Ambrose-Singer connection $\widetilde{\nabla} = \nabla - T^{\nabla}$ satisfies $\widetilde{\nabla}_X Y = 0$ for any two left-invariant vector fields. This structure is equivalent to the description $G = G/\{e\}$, which corresponds to the action $G \times G \to G$.

7.2 Riemannian homogeneous structures in dimension three

On the basis of the above outlined, in this section we will clarify the classification of the Riemannian homogeneous structures in dimension three, describing all the possible ones in the non-symmetric case. The following result characterizes the non-symmetric Lie groups admitting more than one homogeneous structure.

Theorem 7.1. A non-symmetric simply connected three-dimensional Riemannian Lie group admits a homogeneous structure different from the canonical one if and only if it admits a naturally reductive homogeneous structure. Moreover, in such a case, it admits exactly a one-parameter family of homogeneous structures.

The explicit description of all the homogeneous structures on non-symmetric Lie groups is given in Theorem 7.2 and Theorem 7.3.

Recall that a three-dimensional complete and simply connected manifold is naturally reductive if and only if it admits a non-vanishing homogeneous structure of type \mathcal{T}_3 . In this case (M, g)is a real space form \mathbb{R}^3 , \mathbb{S}^3 or \mathbb{H}^3 , or it is isometric either to the special unitary group SU(2), or to the universal covering of $\widetilde{SL}(2,\mathbb{R})$ or to the 3-dimensional Heisenberg group H^3 , endowed with a suitable left-invariant metric described in terms of the Lie algebras (up to rotations) as

$$H^{3}: [e_{1}, e_{2}] = \lambda e_{3}, \qquad \lambda \neq 0,$$

$$SU(2): [e_{1}, e_{2}] = \mu e_{3}, \quad [e_{2}, e_{3}] = \lambda e_{1}, \quad [e_{3}, e_{1}] = \lambda e_{2}, \quad \lambda \mu > 0,$$

$$\widetilde{SL}(2, \mathbb{R}): [e_{1}, e_{2}] = \mu e_{3}, \quad [e_{2}, e_{3}] = \lambda e_{1}, \quad [e_{3}, e_{1}] = \lambda e_{2}, \quad \lambda \mu < 0,$$

(7.3)

where $\{e_1, e_2, e_3\}$ is an orthonormal basis (see [130]).

In this way, (M, g) is naturally reductive if and only if it is isometric to a Lie group endowed with a left-invariant metric whose isometry group is at least four-dimensional.

Theorem 7.1 is thus connected to the following theorem by Meeks and Perez (see [103]): a simply connected, three-dimensional Lie group with a left-invariant metric $(G_1, \langle \cdot, \cdot \rangle_1)$ is isometric to a second Lie group $(G_2, \langle \cdot, \cdot \rangle_2)$ which is not isomorphic to G_1 if and only if its isometry group has dimension at least four.

7.2.1 Summary of results

We study the unimodular and non-unimodular cases separately. The unimodular case is dealt with in Section 7.2.2, and the non-unimodular case is considered in Section 7.2.3. We will see that the Riemannian homogeneous structures on a non-symmetric three-dimensional Lie group G equipped with a left-invariant metric are given as follows, from where the proof of Theorem 7.1 is obtained at once.

Unimodular Lie groups

The left-invariant Riemannian metrics $\langle \cdot, \cdot \rangle$ on three-dimensional unimodular Lie groups G were described by Milnor (see [104]) in terms of three structure constants $(\lambda_1, \lambda_2, \lambda_3)$, so that the Lie algebra becomes

$$[e_1, e_2] = \lambda_3 e_3,$$
 $[e_1, e_3] = -\lambda_2 e_2,$ $[e_2, e_3] = \lambda_1 e_1,$
for an orthonormal basis $\{e_1, e_2, e_3\}$. It now follows that $(G, \langle \cdot, \cdot \rangle)$ is symmetric if and only if $\lambda_1 = 0$ and $\lambda_2 = \lambda_3$ (up to rotations), in which case it is flat, or $\lambda_1 = \lambda_2 = \lambda_3 \neq 0$, and the sectional curvature is constant and positive. We focus on the non-symmetric situation and we have the following result.

Theorem 7.2. Let $(G, \langle \cdot, \cdot \rangle)$ be a unimodular Lie group equipped with a non-symmetric leftinvariant Riemannian metric. Then there are two mutually excluding cases.

(i) The three structure constants $\lambda_1, \lambda_2, \lambda_3$ are different and the only homogeneous structure is the canonical one, which is given by

$$T^{\nabla} = -(\lambda_1 - \lambda_2 - \lambda_3)e^1 \otimes (e^2 \wedge e^3) - (\lambda_1 - \lambda_2 + \lambda_3)e^2 \otimes (e^1 \wedge e^3) + (\lambda_1 + \lambda_2 - \lambda_3)e^3 \otimes (e^1 \wedge e^2).$$

The canonical homogeneous structure is of type \mathcal{T}_2 if $\lambda_1 + \lambda_2 + \lambda_3 = 0$ (see also [73]) and it is of type $\mathcal{T}_2 \oplus \mathcal{T}_3$ otherwise.

(ii) Up to a rotation, the structure constants $\lambda_1 = \lambda_2 \neq \lambda_3$, $\lambda_3 \neq 0$ and there exists a oneparameter family of homogeneous structures

$$T = \lambda_3 e^1 \otimes (e^2 \wedge e^3) - \lambda_3 e^2 \otimes (e^1 \wedge e^3) + 2\kappa e^3 \otimes (e^1 \wedge e^2), \quad \kappa \in \mathbb{R},$$

which corresponds to the canonical structure for $\kappa = \frac{1}{2}(2\lambda_1 - \lambda_3)$. Moreover, it is of type \mathcal{T}_2 if $\kappa = -\lambda_3$, of type \mathcal{T}_3 if $\kappa = \frac{1}{2}\lambda_3$, and of type $\mathcal{T}_2 \oplus \mathcal{T}_3$ otherwise.

The unimodular Lie groups in Theorem 7.2-(*ii*) correspond to SU(2), $\widetilde{SL}(2,\mathbb{R})$ and H^3 with left-invariant metric as in (7.3), and these include the homogeneous structures on Berger spheres previously considered in [75].

Non-unimodular Lie groups

Non-unimodular Riemannian Lie groups $(G, \langle \cdot, \cdot \rangle)$ are semi-direct extensions $\mathbb{R}^2 \rtimes \mathbb{R}$ of the Abelian group. It was shown in [104] that there exists an orthonormal basis $\{e_1, e_2, e_3\}$ so that

$$[e_1, e_2] = \alpha e_2 + \beta e_3, \qquad [e_1, e_3] = \gamma e_2 + \delta e_3, \qquad [e_2, e_3] = 0,$$

where the trace of the endomorphism determining the semi-direct extension satisfies $\alpha + \delta \neq 0$. Moreover, one may rotate the orthonormal basis $\{e_2, e_3\}$ of the unimodular kernel to assume that their images by the endomorphism are orthogonal, i.e., $\alpha\gamma + \beta\delta = 0$. The Riemannian Lie group $(G, \langle \cdot, \cdot \rangle)$ is symmetric if and only if $\beta = \delta = \gamma = 0$ (up to the isometry $e_2 \mapsto e_3$), and so it is isometric to $\mathbb{R} \times \mathbb{H}^2(-\alpha^2)$, or if $\alpha = \delta \neq 0$ and $\gamma = -\beta$, in which case it is a space of constant sectional curvature $\mathbb{H}^3(-\delta^2)$. In this setting, we have the following result.

Theorem 7.3. Let $(G, \langle \cdot, \cdot \rangle)$ be a non-unimodular Lie group equipped with a left-invariant nonsymmetric Riemannian metric. Then there are two mutually excluding cases.

- (i) If $\delta = \gamma = 0$, $\alpha\beta \neq 0$, then the homogeneous structures are given by one of the following possibilities.
 - (*i.a*) The one-parameter family

$$T = \beta e^1 \otimes (e^2 \wedge e^3) - \beta e^2 \otimes (e^1 \wedge e^3) + 2\kappa e^3 \otimes (e^1 \wedge e^2), \quad \kappa \in \mathbb{R}.$$

In this case the homogeneous structure is of type \mathcal{T}_2 if $\kappa = -\beta$, of type \mathcal{T}_3 if $\kappa = \frac{1}{2}\beta$, and of type $\mathcal{T}_2 \oplus \mathcal{T}_3$ otherwise.

(i.b) The canonical homogeneous structure

$$T^{\nabla} = \beta e^1 \otimes (e^2 \wedge e^3) - 2\alpha e^2 \otimes (e^1 \wedge e^2) - \beta e^2 \otimes (e^1 \wedge e^3) - \beta e^3 \otimes (e^1 \wedge e^2),$$

which is of the generic type $\mathcal{T}_1 \oplus \mathcal{T}_2 \oplus \mathcal{T}_3$.

(ii) If $\delta \alpha \neq 0$, $\beta = -\frac{\alpha \gamma}{\delta}$ and $\alpha \neq \delta$, then the only homogeneous structure is the canonical one, which is given by

$$T^{\nabla} = -\frac{(\alpha+\delta)\gamma}{\delta}e^1 \otimes (e^2 \wedge e^3) - 2\alpha e^2 \otimes (e^1 \wedge e^2) + \frac{(\alpha-\delta)\gamma}{\delta}e^2 \otimes (e^1 \wedge e^3) + \frac{(\alpha-\delta)\gamma}{\delta}e^3 \otimes (e^1 \wedge e^2) - 2\delta e^3 \otimes (e^1 \wedge e^3),$$

and is of type $\mathcal{T}_1 \oplus \mathcal{T}_2$ if $\gamma = 0$, and $\mathcal{T}_1 \oplus \mathcal{T}_2 \oplus \mathcal{T}_3$ otherwise.

The non-unimodular Lie groups given in Theorem 7.3 are semi-direct extensions $\mathbb{R}^2 \rtimes \mathbb{R}$ of the Abelian Lie group determined by an endomorphism $-\operatorname{ad}(e_1)$. Assertion (*i*) in Theorem 7.3 corresponds to the special situation where det $\operatorname{ad}(e_1) = 0$, and they are isometric (although not isomorphically isometric) to a left-invariant metric on $\widetilde{SL}(2, \mathbb{R})$ as in (7.3), which corresponds to Theorem 7.2-(*ii*) (cf. [103, 130]). We would like to emphasize that isometries between Riemannian Lie groups need not preserve the Lie group structure, since they are not necessarily realized by group isomorphisms, as evidenced in the above-mentioned situation. On the other hand, the Lie groups in Theorem 7.3-(*ii*) correspond to the generic situation, where one can always specify the orthonormal basis $\{e_2, e_3\}$ so that it is given by eigenvectors of the self-adjoint part of $\operatorname{ad}(e_1)$ (cf. [104]).

Non-symmetric simply connected homogeneous three-dimensional Riemannian manifolds with four-dimensional isometry group are isometric to the unitary group SU(2), the universal cover of $\widetilde{SL}(2,\mathbb{R})$, or the Heisenberg group with the special metrics (7.3). It follows from the description of homogeneous structures in Theorem 7.2 and Theorem 7.3 that (see also [130]) a non-symmetric three-dimensional Riemannian Lie group admits a homogeneous structure different from the canonical one if and only if its isometry group is four-dimensional

Remark 7.4. A more conceptual proof of this last statement can be summarized as follows. For any three-dimensional Lie groups $(G_1, \langle \cdot, \cdot \rangle_1)$ and $(G_2, \langle \cdot, \cdot \rangle_2)$ equipped with a left-invariant Riemannian metric, it follows from Theorems 7.2 and 7.3 that the infinitesimal models associated to their canonical homogeneous structures are isomorphic if and only if the Lie groups $(G_1, \langle \cdot, \cdot \rangle_1)$ and $(G_2, \langle \cdot, \cdot \rangle_2)$ are isomorphically isometric (see [34]). Besides, any non-symmetric homogeneous three-manifold with four-dimensional isometry group admits more than one homogeneous structure. It follows from the works in [103, 126] that a homogeneous three-manifold with threedimensional isometry group is isometric to a unique Riemannian Lie group, in which case any homogeneous structure is isomorphic to the canonical one.

7.2.2 Homogeneous structures on non-symmetric unimodular Lie groups

Following [104], if \mathfrak{g} is unimodular then there exists an orthonormal basis $\{e_1, e_2, e_3\}$ of \mathfrak{g} such that

$$[e_1, e_2] = \lambda_3 e_3, \qquad [e_1, e_3] = -\lambda_2 e_2, \qquad [e_2, e_3] = \lambda_1 e_1.$$

Let T be a (0,3)-tensor field so that the connection $\widetilde{\nabla} = \nabla - T$ makes the metric tensor parallel, i.e., $T_{xyz} + T_{xzy} = 0$ for $x, y, z \in \mathfrak{g}$. Denoting by $\{e^1, e^2, e^3\}$ the dual basis of $\{e_1, e_2, e_3\}$, then the tensor field T can be written as

$$T = 2\sum_{i} \sum_{j < k} T_{ijk} e^i \otimes (e^j \wedge e^k).$$

Therefore, the non-zero components of the connection $\widetilde{\nabla} = \nabla - T$ are given by

$$\widetilde{\nabla}_{112} = -T_{112}, \quad \widetilde{\nabla}_{223} = -T_{223}, \quad \widetilde{\nabla}_{123} = -\frac{1}{2}(\lambda_1 - \lambda_2 - \lambda_3) - T_{123}, \\
\widetilde{\nabla}_{113} = -T_{113}, \quad \widetilde{\nabla}_{313} = -T_{313}, \quad \widetilde{\nabla}_{213} = -\frac{1}{2}(\lambda_1 - \lambda_2 + \lambda_3) - T_{213}, \\
\widetilde{\nabla}_{212} = -T_{212}, \quad \widetilde{\nabla}_{323} = -T_{323}, \quad \widetilde{\nabla}_{312} = -\frac{1}{2}(\lambda_1 + \lambda_2 - \lambda_3) - T_{312},$$
(7.4)

while the (0, 4)-curvature tensor field is determined by

$$R_{1212} = \frac{1}{4} \left((\lambda_1 - \lambda_2)^2 - 3\lambda_3^2 + 2(\lambda_1 + \lambda_2)\lambda_3 \right),$$

$$R_{1313} = \frac{1}{4} \left((\lambda_1 - \lambda_3)^2 - 3\lambda_2^2 + 2(\lambda_1 + \lambda_3)\lambda_2 \right),$$

$$R_{2323} = \frac{1}{4} \left((\lambda_2 - \lambda_3)^2 - 3\lambda_1^2 + 2(\lambda_2 + \lambda_3)\lambda_1 \right).$$

(7.5)

Let $\Re_{ikj\ell;r} = (\widetilde{\nabla}_{e_r}R)(e_i, e_j, e_k, e_\ell)$. A straightforward calculation involving

Equations (7.4) and (7.5) shows that the condition $\widetilde{\nabla}R = 0$ in Equation (7.1) is given by

$$2 \mathfrak{R}_{1213;1} = (\lambda_1 - \lambda_2 - \lambda_3)(\lambda_2 - \lambda_3)(\lambda_1 - \lambda_2 - \lambda_3 + 2T_{123}) = 0,$$

$$\mathfrak{R}_{1213;2} = (\lambda_1 - \lambda_2 - \lambda_3)(\lambda_2 - \lambda_3)T_{223} = 0,$$

$$\mathfrak{R}_{1213;3} = (\lambda_1 - \lambda_2 - \lambda_3)(\lambda_2 - \lambda_3)T_{323} = 0,$$

$$\mathfrak{R}_{1223;1} = (\lambda_1 - \lambda_2 + \lambda_3)(\lambda_1 - \lambda_3)T_{113} = 0,$$

$$2 \mathfrak{R}_{1223;2} = (\lambda_1 - \lambda_2 + \lambda_3)(\lambda_1 - \lambda_3)(\lambda_1 - \lambda_2 + \lambda_3 + 2T_{213}) = 0,$$

$$\mathfrak{R}_{1223;3} = (\lambda_1 - \lambda_2 + \lambda_3)(\lambda_1 - \lambda_3)T_{313} = 0,$$

$$\mathfrak{R}_{1323;1} = (\lambda_1 + \lambda_2 - \lambda_3)(\lambda_1 - \lambda_2)T_{112} = 0,$$

$$-\mathfrak{R}_{1323;2} = (\lambda_1 + \lambda_2 - \lambda_3)(\lambda_1 - \lambda_2)T_{212} = 0,$$

$$2 \mathfrak{R}_{1323;3} = (\lambda_1 + \lambda_2 - \lambda_3)(\lambda_1 - \lambda_2)(\lambda_1 + \lambda_2 - \lambda_3 - 2T_{312}) = 0.$$

(7.6)

From here, depending on the eigenvalues λ_i , we are led to the following two possibilities.

Case of three different eigenvalues

In this case, if $\lambda_1 - \lambda_2 - \lambda_3$, $\lambda_1 - \lambda_2 + \lambda_3$ and $\lambda_1 + \lambda_2 - \lambda_3$ do not vanish, then Equations (7.4) and (7.6) clearly imply that $\widetilde{\nabla}_X Y = 0$ for all left-invariant vector fields. Therefore, the only homogeneous structure is the canonical one given by $T_x y = \nabla_x y$, for $x, y \in \mathfrak{g}$, i.e.,

$$T = -(\lambda_1 - \lambda_2 - \lambda_3)e^1 \otimes (e^2 \wedge e^3) - (\lambda_1 - \lambda_2 + \lambda_3)e^2 \otimes (e^1 \wedge e^3) + (\lambda_1 + \lambda_2 - \lambda_3)e^3 \otimes (e^1 \wedge e^2).$$
(7.7)

Next we show that the same holds if any of $\lambda_1 - \lambda_2 - \lambda_3$, $\lambda_1 - \lambda_2 + \lambda_3$ and $\lambda_1 + \lambda_2 - \lambda_3$ vanishes. Suppose that $\lambda_3 = \lambda_1 - \lambda_2$ (the other two cases are obtained in a completely analogous way). Then, Equations (7.6) implies

$$T_{112} = T_{113} = T_{212} = T_{313} = 0, \quad T_{213} = -\lambda_1 + \lambda_2, \quad T_{312} = \lambda_2.$$
 (7.8)

Let $\mathfrak{T}_{ijk;r} = (\widetilde{\nabla}_{e_r}T)(e_i, e_j, e_k)$. A straightforward calculation using Equations (7.4) and (7.8) shows that the condition $\widetilde{\nabla}T = 0$ in Equation (7.1) reduces to

$$\begin{aligned} \mathfrak{T}_{313;r} &= -\mathfrak{T}_{212;r} = (\lambda_1 - 2\lambda_2)T_{r23} = 0, \\ \mathfrak{T}_{223;r} &= T_{r23}T_{323} = 0, \\ \mathfrak{T}_{323;r} &= -T_{r23}T_{223} = 0, \end{aligned}$$

or, equivalently, $T_{123} = T_{223} = T_{323} = 0$. Thus, the only homogeneous structure is given by

$$T = -2(\lambda_1 - \lambda_2)e^2 \otimes (e^1 \wedge e^3) + 2\lambda_2 e^3 \otimes (e^1 \wedge e^2),$$

which corresponds to the structure given by Equation (7.7) for $\lambda_3 = \lambda_1 - \lambda_2$.

Finally, the projections of the homogeneous structure given in Equation (7.7) are obtained by a direct calculation. In particular, $p_1(T) = 0$ and

$$p_{2}(T) = -\frac{2}{3}(2\lambda_{1} - \lambda_{2} - \lambda_{3})e^{1} \otimes (e^{2} \wedge e^{3}) - \frac{2}{3}(\lambda_{1} - 2\lambda_{2} + \lambda_{3})e^{2} \otimes (e^{1} \wedge e^{3}) + \frac{2}{3}(\lambda_{1} + \lambda_{2} - 2\lambda_{3})e^{3} \otimes (e^{1} \wedge e^{2}),$$
$$p_{3}(T) = \frac{1}{3}(\lambda_{1} + \lambda_{2} + \lambda_{3})(e^{1} \otimes (e^{2} \wedge e^{3}) - e^{2} \otimes (e^{1} \wedge e^{3}) + e^{3} \otimes (e^{1} \wedge e^{2})).$$

Case of two different eigenvalues

In this case, without loss of generality, we can assume $\lambda_1 = \lambda_2 \neq \lambda_3$. Moreover, $\lambda_3 \neq 0$ since the space would be locally symmetric otherwise. Thus, Equation (7.6) implies

$$T_{113} = T_{223} = T_{313} = T_{323} = 0, \quad T_{123} = -T_{213} = \frac{\lambda_3}{2}.$$
 (7.9)

Let $\mathfrak{T}_{ijk;r} = (\widetilde{\nabla}_{e_r}T)(e_i, e_j, e_k)$. A straightforward calculation using Equations (7.4) and (7.9) shows that the condition $\widetilde{\nabla}T = 0$ in Equation (7.1) reduces to

$$\begin{aligned} \mathfrak{T}_{112;r} &= T_{r12}T_{212} = 0, \ (r = 1, 2), \quad \mathfrak{T}_{112;3} = -\frac{1}{2}(2\lambda_1 - \lambda_3 - 2T_{312})T_{212} = 0, \\ \mathfrak{T}_{212;r} &= -T_{r12}T_{112} = 0, \ (r = 1, 2), \quad \mathfrak{T}_{212;3} = -\frac{1}{2}(2\lambda_1 - \lambda_3 - 2T_{312})T_{112} = 0, \end{aligned}$$

or, equivalently, $T_{112} = T_{212} = 0$. Thus, we obtain a one-parameter family of homogeneous structures given by

$$T = \lambda_3 e^1 \otimes (e^2 \wedge e^3) - \lambda_3 e^2 \otimes (e^1 \wedge e^3) + 2\kappa e^3 \otimes (e^1 \wedge e^2), \quad \kappa \in \mathbb{R}.$$

In the particular case where $\kappa = \frac{1}{2}(2\lambda_1 - \lambda_3)$, it corresponds to the canonical structure. Finally, a direct calculation shows that the projections of these structures are such that $p_1(T) = 0$ and

$$p_2(T) = \frac{1}{3}(\lambda_3 - 2\kappa) \left(e^1 \otimes (e^2 \wedge e^3) - e^2 \otimes (e^1 \wedge e^3) - 2e^3 \otimes (e^1 \wedge e^2) \right),$$

$$p_3(T) = \frac{2}{3}(\lambda_3 + \kappa) \left(e^1 \otimes (e^2 \wedge e^3) - e^2 \otimes (e^1 \wedge e^3) + e^3 \otimes (e^1 \wedge e^2) \right).$$

7.2.3 Homogeneous structures on non-symmetric and non-unimodular Lie groups

If \mathfrak{g} is non-unimodular then there exists an orthonormal basis $\{e_1, e_2, e_3\}$ of \mathfrak{g} such that (see [104])

 $[e_1, e_2] = \alpha e_2 + \beta e_3,$ $[e_1, e_3] = \gamma e_2 + \delta e_3,$ $[e_2, e_3] = 0,$

where $\alpha + \delta \neq 0$ and $\alpha \gamma + \beta \delta = 0$.

A straightforward calculation shows that a non-unimodular Lie group corresponding to a Lie algebra above is locally symmetric if and only if it is of constant negative sectional curvature (which corresponds to the cases when $ad(e_1)$ is a multiple of the identity or it has complex eigenvalues), or it is locally isometric to a product $\mathbb{R} \times N(c)$, where N(c) is a surface of constant negative sectional curvature (if $ad(e_1)$ is of rank-one and $\{e_2, e_3\}$ is an orthonormal basis of eigenvectors).

Using the same notation as in the previous section, the non-zero components of the connection $\widetilde{\nabla} = \nabla - T$ are given by

$$\widetilde{\nabla}_{112} = -T_{112}, \qquad \widetilde{\nabla}_{223} = -T_{223}, \qquad \widetilde{\nabla}_{123} = \frac{1}{2}(\beta - \gamma - 2T_{123}), \\
\widetilde{\nabla}_{113} = -T_{113}, \qquad \widetilde{\nabla}_{313} = -\delta - T_{313}, \qquad \widetilde{\nabla}_{213} = -\frac{1}{2}(\beta + \gamma + 2T_{213}), \\
\widetilde{\nabla}_{212} = -\alpha - T_{212}, \qquad \widetilde{\nabla}_{323} = -T_{323}, \qquad \widetilde{\nabla}_{312} = -\frac{1}{2}(\beta + \gamma + 2T_{312}),$$
(7.10)

while the (0, 4)-curvature tensor field is determined by

$$R_{1212} = -\frac{1}{4} \left(4\alpha^2 + 3\beta^2 - \gamma^2 + 2\beta\gamma \right),$$

$$R_{1313} = \frac{1}{4} \left(\beta^2 - 3\gamma^2 - 4\delta^2 - 2\beta\gamma \right),$$

$$R_{2323} = \frac{1}{4} \left(\beta^2 + \gamma^2 - 4\alpha\delta + 2\beta\gamma \right).$$

(7.11)

As in the previous section, set $\Re_{ikj\ell;r} = (\widetilde{\nabla}_{e_r} R)(e_i, e_j, e_k, e_\ell)$. Equations (7.10) and (7.11) imply that the condition $\widetilde{\nabla} R = 0$ in Equation (7.1) is given by

$$-2 \mathfrak{R}_{1213;1} = (\alpha^{2} + \beta^{2} - \gamma^{2} - \delta^{2})(\beta - \gamma - 2T_{123}) = 0,$$

$$\mathfrak{R}_{1213;2} = (\alpha^{2} + \beta^{2} - \gamma^{2} - \delta^{2})T_{223} = 0,$$

$$\mathfrak{R}_{1213;3} = (\alpha^{2} + \beta^{2} - \gamma^{2} - \delta^{2})T_{323} = 0,$$

$$-\mathfrak{R}_{1223;1} = (\alpha^{2} + \beta^{2} - \alpha\delta + \beta\gamma)T_{113} = 0,$$

$$-2 \mathfrak{R}_{1223;2} = (\alpha^{2} + \beta^{2} - \alpha\delta + \beta\gamma)(\beta + \gamma + 2T_{213}) = 0,$$

$$-\mathfrak{R}_{1223;3} = (\alpha^{2} + \beta^{2} - \alpha\delta + \beta\gamma)(\delta + T_{313}) = 0,$$

$$\mathfrak{R}_{1323;1} = (\gamma^{2} + \delta^{2} - \alpha\delta + \beta\gamma)T_{112} = 0,$$

$$\mathfrak{R}_{1323;2} = (\gamma^{2} + \delta^{2} - \alpha\delta + \beta\gamma)(\alpha + T_{212}) = 0,$$

$$2 \mathfrak{R}_{1323;3} = (\gamma^{2} + \delta^{2} - \alpha\delta + \beta\gamma)(\beta + \gamma + 2T_{312}) = 0.$$

(7.12)

Next we analyse the cases $\delta = 0$ and $\delta \neq 0$ separately.

Case $\delta = 0$

Since $\alpha + \delta \neq 0$ and $\alpha \gamma + \beta \delta = 0$, in this case $\gamma = 0$ and $\alpha \neq 0$. Moreover, $\beta \neq 0$ because the space is assumed not to be locally symmetric. Thus, Equation (7.12) reduces to

$$T_{113} = T_{223} = T_{313} = T_{323} = 0, \quad T_{123} = -T_{213} = \frac{\beta}{2}.$$
 (7.13)

Following the notation in the previous section, we set $\mathfrak{T}_{ijk;r} = (\widetilde{\nabla}_{e_r}T)(e_i, e_j, e_k)$. A straightforward calculation involving Equations (7.10) and (7.13) shows that the condition $\widetilde{\nabla}T = 0$ in Equation (7.1) reduces to

$$\begin{aligned} \mathfrak{T}_{112;1} &= T_{112}T_{212} = 0, & -\mathfrak{T}_{212;1} = (T_{112})^2 = 0, \\ \mathfrak{T}_{112;2} &= (\alpha + T_{212})T_{212} = 0, & -\mathfrak{T}_{212;2} = (\alpha + T_{212})T_{112} = 0, \\ 2\mathfrak{T}_{112;3} &= (\beta + 2T_{312})T_{212} = 0, & -2\mathfrak{T}_{212;3} = (\beta + 2T_{312})T_{112} = 0. \end{aligned}$$

Hence, $T_{112} = 0$ and either $T_{212} = 0$ or $T_{212} = -\alpha$, $T_{312} = -\frac{\beta}{2}$.

If $T_{112} = T_{212} = 0$, we obtain a one-parameter family of homogeneous structures given by

$$T = \beta e^1 \otimes (e^2 \wedge e^3) - \beta e^2 \otimes (e^1 \wedge e^3) + 2\kappa e^3 \otimes (e^1 \wedge e^2), \quad \kappa \in \mathbb{R}.$$

The projections of these homogeneous structures are obtained by a direct calculation. In particular, $p_1(T) = 0$ and

$$p_2(T) = \frac{1}{3}(\beta - 2\kappa) \left(e^1 \otimes (e^2 \wedge e^3) - e^2 \otimes (e^1 \wedge e^3) - 2e^3 \otimes (e^1 \wedge e^2)\right),$$

$$p_3(T) = \frac{2}{3}(\beta + \kappa) \left(e^1 \otimes (e^2 \wedge e^3) - e^2 \otimes (e^1 \wedge e^3) + e^3 \otimes (e^1 \wedge e^2)\right).$$

If $T_{112} = 0$ and $T_{212} = -\alpha$, $T_{312} = -\frac{\beta}{2}$, then Equations (7.10) and (7.13) clearly imply that $\tilde{\nabla}_X Y = 0$ for left-invariant vector fields. Therefore, the only homogeneous structure is the canonical one, given by

$$T = \beta e^1 \otimes (e^2 \wedge e^3) - 2\alpha e^2 \otimes (e^1 \wedge e^2) - \beta e^2 \otimes (e^1 \wedge e^3) - \beta e^3 \otimes (e^1 \wedge e^2).$$

In this case, the projections of this homogeneous structure are given by

$$p_1(T) = -\alpha \left(e^2 \otimes \left(e^1 \wedge e^2 \right) + e^3 \otimes \left(e^1 \wedge e^3 \right) \right),$$

$$p_2(T) = \frac{2}{3}\beta e^1 \otimes \left(e^2 \wedge e^3 \right) - \alpha e^2 \otimes \left(e^1 \wedge e^2 \right) - \frac{2}{3}\beta e^2 \otimes \left(e^1 \wedge e^3 \right)$$

$$- \frac{4}{3}\beta e^3 \otimes \left(e^1 \wedge e^2 \right) + \alpha e^3 \otimes \left(e^1 \wedge e^3 \right),$$

$$p_3(T) = \frac{1}{3}\beta \left(e^1 \otimes \left(e^2 \wedge e^3 \right) - e^2 \otimes \left(e^1 \wedge e^3 \right) + e^3 \otimes \left(e^1 \wedge e^2 \right) \right).$$

Case $\delta \neq 0$

In this case, since $\alpha \gamma + \beta \delta = 0$, we have $\beta = -\frac{\alpha \gamma}{\delta}$. Moreover, the space is locally symmetric if either $\alpha = \delta$ or $\alpha = \gamma = 0$. Now Equation (7.12) becomes

$$\begin{split} & 2\delta^{3}\,\mathfrak{R}_{1213;1} = \,(\alpha^{2} - \delta^{2})(\gamma^{2} + \delta^{2})((\alpha + \delta)\gamma + 2\delta T_{123}) = 0, \\ & \delta^{2}\,\mathfrak{R}_{1213;2} = \,(\alpha^{2} - \delta^{2})(\gamma^{2} + \delta^{2})T_{223} = 0, \\ & \delta^{2}\,\mathfrak{R}_{1213;3} = \,(\alpha^{2} - \delta^{2})(\gamma^{2} + \delta^{2})T_{323} = 0, \\ & -\delta^{2}\,\mathfrak{R}_{1223;1} = \,\alpha(\alpha - \delta)(\gamma^{2} + \delta^{2})T_{113} = 0, \\ & 2\delta^{3}\,\mathfrak{R}_{1223;2} = \,\alpha(\alpha - \delta)(\gamma^{2} + \delta^{2})((\alpha - \delta)\gamma - 2\delta T_{213}) = 0, \\ & -\delta^{2}\,\mathfrak{R}_{1223;3} = \,\alpha(\alpha - \delta)(\gamma^{2} + \delta^{2})(\delta + T_{313}) = 0, \\ & -\delta\,\mathfrak{R}_{1323;1} = \,(\alpha - \delta)(\gamma^{2} + \delta^{2})T_{112} = 0, \\ & -\delta\,\mathfrak{R}_{1323;2} = \,(\alpha - \delta)(\gamma^{2} + \delta^{2})(\alpha + T_{212}) = 0, \\ & 2\delta^{2}\,\mathfrak{R}_{1323;3} = \,(\alpha - \delta)(\gamma^{2} + \delta^{2})((\alpha - \delta)\gamma - 2\delta T_{312}) = 0. \end{split}$$

Note that $\alpha^2 - \delta^2 \neq 0$, since $\alpha + \delta \neq 0$ and if $\alpha - \delta = 0$ then the space is locally symmetric. Besides, if $\alpha = 0$ then $\beta = 0$ and the metric is isometric to the one in the case corresponding to $\delta = 0$. Hence, the previous equations together with Equation (7.10) imply that $\widetilde{\nabla}_X Y = 0$ for left-invariant vector fields and the only homogeneous structure is the canonical one, which corresponds to

$$\begin{split} T &= -\frac{(\alpha+\delta)\gamma}{\delta}e^1 \otimes (e^2 \wedge e^3) - 2\alpha e^2 \otimes (e^1 \wedge e^2) + \frac{(\alpha-\delta)\gamma}{\delta}e^2 \otimes (e^1 \wedge e^3) \\ &+ \frac{(\alpha-\delta)\gamma}{\delta}e^3 \otimes (e^1 \wedge e^2) - 2\delta e^3 \otimes (e^1 \wedge e^3), \end{split}$$

and its projections are given by

$$p_{1}(T) = -(\alpha + \delta) \left(e^{2} \otimes (e^{1} \wedge e^{2}) + e^{3} \otimes (e^{1} \wedge e^{3})\right),$$

$$p_{2}(T) = -\frac{2(\alpha + \delta)\gamma}{3\delta}e^{1} \otimes (e^{2} \wedge e^{3}) - (\alpha - \delta)e^{2} \otimes (e^{1} \wedge e^{2})$$

$$+ \frac{2(\alpha - 2\delta)\gamma}{3\delta}e^{2} \otimes (e^{1} \wedge e^{3})$$

$$+ \frac{2(2\alpha - \delta)\gamma}{3\delta}e^{3} \otimes (e^{1} \wedge e^{2}) + (\alpha - \delta)e^{3} \otimes (e^{1} \wedge e^{3}),$$

$$p_{3}(T) = -\frac{(\alpha + \delta)\gamma}{3\delta} \left(e^{1} \otimes (e^{2} \wedge e^{3}) - e^{2} \otimes (e^{1} \wedge e^{3}) + e^{3} \otimes (e^{1} \wedge e^{2})\right).$$

7.3 Self-dual and anti-self-dual homogeneous structures

Dimension four is of special interest for the study of Riemannian manifolds. In this case, the fact that the rotation group SO(4) is not simple gives rise to the concepts of self-duality and anti-self-duality in this context.

Even though \mathcal{T}_1 and \mathcal{T}_3 are irreducible under the action of SO(4), \mathcal{T}_2 splits into two irreducible components and the decomposition of the space of homogeneous structures on a fourdimensional vector space $(\mathcal{V}, \langle \cdot, \cdot \rangle)$ is now given by

$$\mathcal{T}(\mathcal{V}) = \mathcal{T}_1(\mathcal{V}) \oplus \mathcal{T}_2^+(\mathcal{V}) \oplus \mathcal{T}_2^-(\mathcal{V}) \oplus \mathcal{T}_3(\mathcal{V}),$$

where

$$\mathcal{T}_2^{\pm} = \{ T \in \mathcal{T}_2 \colon T(x, \bar{y}, \bar{z}) = \pm T(x, y, z) \}$$

for all $x, y, z \in \mathcal{V}$, where (\bar{y}, \bar{z}) stands for the dual of (y, z) according to the relation $e^k \wedge e^{\ell} \wedge e^{\bar{k}} \wedge e^{\bar{\ell}} = e^1 \wedge e^2 \wedge e^3 \wedge e^4$ for any positively oriented orthonormal basis $\{e_1, e_2, e_3, e_4\}$. The projections of a homogeneous structure $T \in \mathcal{T}$ on these two subspaces are given by

$$p_2^{\pm}(T)(x,y,z) = \frac{1}{2} \{ p_2(T)(x,y,z) \pm p_2(T)(x,\bar{y},\bar{z}) \},\$$

for any $x, y, z \in \mathcal{V}$.

A homogenous structure T on a four-dimensional Riemannian manifold is said to be *self-dual* (resp. *anti-self-dual*) if $p_2^-(T) = 0$ (resp. $p_2^+(T) = 0$).

Self-dual and anti-self-dual homogeneous structures appear naturally in relation to the study of homogeneous manifolds with half-harmonic Weyl curvature.

Proposition 7.5. Let (M, g) be a simply connected homogeneous four-manifold with half-harmonic Weyl tensor. Then it is symmetric or it admits a self-dual or anti-self-dual homogeneous structure.

Proof. Let $(G, \langle \cdot, \cdot \rangle)$ be a half-harmonic non-symmetric homogenous four-manifold as in Theorem 6.1-(*i*), determined by the Lie algebra structure

$$[e_1, e_2] = e_3, \ [e_1, e_4] = -e_1, \ [e_2, e_4] = -e_2, \ [e_3, e_4] = -2e_3.$$

Then, the Riemannian Lie group $(G, \langle \cdot, \cdot \rangle)$ is anti-self-dual and its canonical homogeneous structure is given by

$$T = 2e^1 \otimes (e^1 \wedge e^4) + e^1 \otimes (e^2 \wedge e^3) - e^2 \otimes (e^1 \wedge e^3)$$
$$+2e^2 \otimes (e^2 \wedge e^4) - e^3 \otimes (e^1 \wedge e^2) + 4e^3 \otimes (e^3 \wedge e^4)$$

and a straightforward calculation shows that $p_2^+(T) = 0$, which shows that it is an anti-self-dual homogeneous structure. Moreover, one has that $p_1(T)$ and $p_3(T)$ are non-zero, and thus it is a homogeneous anti-self-dual structure which is not \mathcal{T}_2 .

The homogeneous manifold in Theorem 6.1-(*ii*) corresponds to the unique four-dimensional 3-symmetric space, and it is isometric to the Lie group determined by the Lie algebra

$$[e_1, e_2] = e_3, \ [e_1, e_4] = \frac{1}{2}e_1, \ [e_2, e_4] = -e_2, \ [e_3, e_4] = -\frac{1}{2}e_3.$$

Considering the Kähler and opposite almost Kähler structures (J_+, J_-) determined by the Kähler forms

$$\Omega_+ = e^1 \wedge e^2 - e^3 \wedge e^4$$
, and $\Omega_- = e^1 \wedge e^2 + e^3 \wedge e^4$,

the covariant derivative of the opposite almost Kähler structure J_{-} determines a homogeneous structure

$$T_{XY} = \frac{1}{2}J_{-}(\nabla_X J_{-})Y$$

which is self-dual [130]. Furthermore, this structure is \mathcal{T}_2 , and it was shown by Nicolodi in [109] that it is the only self-dual \mathcal{T}_2 structure. Furthermore this structure is not the canonical one. \Box

Remark 7.6. It follows after a long calculation that there are no strictly self-dual or anti-selfdual homogeneous structures on the semi-direct extensions $E(1,1) \rtimes \mathbb{R}$ and $\widetilde{E}(2) \rtimes \mathbb{R}$. On the contrary, semi-direct extensions of the Heisenberg group support such homogeneous structures. Indeed $H_3 \rtimes \mathbb{R}$ admits a self-dual homogeneous structure if and only if it is homothetic to a semi-direct extension with left-invariant metric determined by the following Lie algebras given in terms of an orthonormal basis $\{e_1, e_2, e_3, e_4\}$:

(i)
$$[e_1, e_2] = e_3$$
, $[e_1, e_4] = \alpha e_1$, $[e_2, e_4] = \alpha e_2$, $[e_3, e_4] = 2\alpha e_3$, $\alpha \neq 0, \pm \frac{1}{2}$.

In this case,

$$T = -2\alpha e^1 \otimes (e^1 \wedge e^4) + e^1 \otimes (e^2 \wedge e^3) - e^2 \otimes (e^1 \wedge e^3)$$
$$-2\alpha e^2 \otimes (e^2 \wedge e^4) - (2\alpha - 1)e^3 \otimes (e^1 \wedge e^2) - 4\alpha e^3 \otimes (e^3 \wedge e^4)$$

is of type $\mathcal{T}_1 \oplus \mathcal{T}_2^+$ if $\alpha = \frac{3}{2}$, or of type $\mathcal{T}_1 \oplus \mathcal{T}_2^+ \oplus \mathcal{T}_3$ otherwise.

(*ii*)
$$[e_1, e_2] = e_3$$
, $[e_1, e_4] = e_1$, $[e_2, e_4] = -\frac{1}{2}e_2$, $[e_3, e_4] = \frac{1}{2}e_3$.

In this case,

$$T = -e^2 \otimes (e^1 \wedge e^3) + e^2 \otimes (e^2 \wedge e^4) - e^3 \otimes (e^1 \wedge e^2) - e^3 \otimes (e^3 \wedge e^4)$$

is of type \mathcal{T}_2^+ .

There are semi-direct extensions of the Abelian group \mathbb{R}^3 which admit self-dual homogeneous structures. Considering the Lie algebra

$$[e_1, e_4] = e_1, \quad [e_2, e_4] = fe_2 + he_3, \quad [e_3, e_4] = -he_2 + fe_3, \quad f \neq 1,$$

where $\{e_1, e_2, e_3, e_4\}$ is an orthonormal basis, one can check that the homogeneous structures

$$T = -2e^1 \otimes (e^1 \wedge e^4) + 2(f-1)e^1 \otimes (e^2 \wedge e^3)$$
$$-2fe^2 \otimes (e^2 \wedge e^4) - 2fe^3 \otimes (e^3 \wedge e^4)$$

are self-dual and never correspond to the canonical homogeneous structure.

Conclusions and open problems

The main achievements of this work can be outlined as follows.

- **C.1** We completed the classification of four-dimensional locally conformally flat Kähler, para-Kähler and null-Kähler structures.
- **C.2** We determined all the left-invariant four-dimensional para-Kähler Lie groups up to automorphisms preserving the symplectic structure, showing that all the locally conformally flat structures are realizable as left-invariant Kähler or para-Kähler structures on Lie groups.
- **C.3** We gave a complete description of all the left-invariant Ricci solitons on four-dimensional Lorentzian Lie groups.
- **C.4** We determined all the Riemannian algebraic Bach solitons, showing that they are algebraic Ricci solitons or belong to one of two exceptional families.
- **C.5** We classified the Riemannian homogeneous four-dimensional manifolds with half-harmonic Weyl curvature.
- **C.6** We showed that a non-symmetric three-dimensional homogeneous manifold admits more than one homogeneous structure if and only if its isometry group has dimension four.

A number of open problems naturally arise as a consequence of our work.

P.1 The complete classification of four-dimensional Bochner-flat para-Kähler structures of non-constant scalar curvature.

It was shown in [67] that any Bochner flat para-Kähler surface of constant scalar curvature is locally a para-complex space form or a locally conformally flat para-Kähler surface as described in Theorem 2.1. The problem of non-constant scalar curvature remains open. Such Bochner flat para-Kähler surfaces are locally isometric to a cotangent bundle with a modified Riemannian extension but a precise parametrization of such structures is still under consideration.

P.2 Algebraic Lorentzian Ricci solitons.

Algebraic Ricci solitons on Lorentzian Lie groups are well-understood in dimension three [14]. One expects to be able to solve the four-dimensional case by following the strategy developed in Chapter 5, although the calculations seem to be much more involved. P.3 The existence of non-trivial left-invariant Bach solitons.

The existence of homogeneous gradient Bach solitons and algebraic Bach solitons on four-dimensional Riemannian Lie groups was discussed in Chapter 5. While there are no non-trivial Ricci solitons on four-dimensional Lie groups with a left-invariant soliton vector field, it is an open problem whether the same statement holds true in the case of Bach solitons.

P.4 Algebraic T-solitons.

It was shown by Arroyo and Lafuente [9] that any Riemannian expanding homogeneous Ricci soliton is homothetic to an algebraic Ricci soliton in dimension four. Hence, four-dimensional homogeneous Ricci solitons are either symmetric or algebraic.

The situation seems to be much more complicated for other geometric flows where it is not clear whether any non-symmetric soliton is necessarily algebraic. This is the case in the completely solvable case, where isometries are isomorphisms of the group, but the general situation is still an open question.

P.5 The complete classification of four-dimensional Riemannian self-dual homogeneous structures.

Homogeneous four-manifolds with half-harmonic Weyl curvature are equipped with a (not necessarily canonical) self-dual homogeneous structure. Non-symmetric semi-direct extensions of the Heisenberg group H^3 , the Euclidean group $\tilde{E}(2)$, or the Poincaré group E(1,1) do not admit any other self-dual homogeneous structure. As pointed out in Section 7.3, there are however other self-dual homogeneous structures on semi-direct extensions $\mathbb{R}^3 \rtimes \mathbb{R}$. It is an open problem to classify such homogeneous structures and to understand their underlying geometries.

Bibliography

- Z. Afifi, Riemannian extensions of affine connected spaces, *Quart. J. Math., Oxford Ser.* (2) 5 (1954), 312–320.
- [2] D. Alekseevsky, B. Guilfoyle, and W. Klingenberg, On the geometry of spaces of oriented geodesics, Ann. Global Anal. Geom. 40 (2011), 389–409. Erratum: Ann. Global Anal. Geom. 50 (2016), 97–99.
- [3] W. Ambrose and I. M. Singer, On homogeneous Riemannian manifolds, *Duke Math. J.* **25** (1958), 647–669.
- [4] H. Anciaux, Spaces of geodesics of pseudo-Riemannian space forms and normal congruences of hypersurfaces, *Trans. Amer. Math. Soc.* **366** (2014), 2699–2718.
- [5] A. Andrada, Hypersymplectic Lie algebras, J. Geom. Phys. 56 (2006), 2039–2067.
- [6] C. Andrada, L. Barberis, I. Dotti, and G. Ovando, Product structures on four-dimensional solvable Lie algebras, *Homology Homotopy Appl.* **7**(1) (2005), 9–37.
- [7] V. Apostolov, J. Armstrong, and T. Drăghici, Local rigidity of certain classes of almost Kähler 4-manifolds, Ann. Global Anal. Geom. 21 (2002), 151–176.
- [8] V. Apostolov, D. M. J. Calderbank, and P. Gauduchon, The geometry of weakly self-dual Kähler surfaces, *Compositio Math.* 135 (2003), 279–322.
- [9] R. Arroyo and R. Lafuente, Homogeneous Ricci solitons in low dimensions, *Int. Math. Res. Not. IMRN* **13** (2015), 4901–4932.
- [10] R. Bach, Zur Weylschen Relativitätstheorie und der Weylschen Erweiterung des Krümmungstensorbegriffs, Math. Z. 9 (1921), 110–135.
- [11] E. Bahuaud and D. Helliwell, Short-time existence for some higher-order geometric flows, *Comm. Partial Differential Equations* **36** (2011), 2189–2207.
- [12] E. Bahuaud and D. Helliwell, Uniqueness for some higher-order geometric flows, *Bull. London Math. Soc.* **47** (2015), 980–995.
- [13] M. Barros and A. Romero, Indefinite Kähler manifolds, Math. Ann. 261 (1982), 55–62.

- [14] W. Batat and K. Onda, Algebraic Ricci solitons of three-dimensional Lorentzian Lie groups, J. Geom. Phys. 114 (2017), 138–152.
- [15] L. Bérard-Bergery, Les spaces homogènes Riemanniens de dimension 4, Riemannian geometry in dimension 4 (Paris 1978/1979) 3 (1981), 40–60.
- [16] M. Berger, Quelques formules de variation pour une structure riemannienne, Ann. Sci. École Norm. Sup. 3 (1970), 285–294.
- [17] M. Berger, P. Gauduchon, and E. Mazet, *Le spectre d'une variété riemannienne*, Lecture Notes in Mathematics, **194**, Springer-verlag, Berlin - New York, 1971.
- [18] A. L. Besse, *Einstein manifolds*, Ergebnisse der Mathematik **10**, Springer, Berlin, 1987.
- [19] M. Blau and M. O'Loughlin, Homogeneous plane waves, *Nuclear Phys. B* 654 (1-2) (2003), 135–176.
- [20] S. Bochner, Curvature and Betti numbers, II, Ann. Math. 50 (1949), 77–93.
- [21] A. Borowiec, M. Francaviglia, and I. Volovich, Anti-Kählerian manifolds, *Differential Geom. Appl.* 12 (2000), 281–289.
- [22] Ch. Boubel and L. Bérard-Bergery, On pseudo-Riemannian manifolds whose Ricci tensor is parallel, *Geom. Dedicata* 86 (2001), 1–18.
- [23] M. Brozos-Vázquez, S. Caeiro-Oliveira, E. García-Río, and R. Vázquez Lorenzo, Fourdimensional homogeneous critical metrics for quadratic curvature functionals, to appear.
- [24] M. Brozos-Vázquez, G. Calvaruso, E. García-Río, and S. Gavino-Fernández, Threedimensional Lorentzian homogeneous Ricci solitons, *Israel J. Math.* 188 (2012), 385–403.
- [25] M. Brozos-Vázquez, E. García-Río, and S. Gavino-Fernández, Locally Conformally Flat Lorentzian Gradient Ricci Solitons, J. Geom. Anal. 23 (2013), 1196–1212.
- [26] M. Brozos-Vázquez, E. García-Río, and X. Valle-Regueiro, Isotropic quasi-Einstein manifolds, *Classical Quantum Gravity* 36 (2019), 245005, 13 pp.
- [27] R. Bryant, Bochner-Kähler metrics, J. Amer. Math. Soc. 14 (2001), 623–715.
- [28] M. Cahen, J. Leroy, M. Parker, F. Tricerri, and L. Vanhecke, Lorentz manifolds modelled on a Lorentz symmetric space, J. Geom. Phys. 7 (4) (1990), 571–581.
- [29] E. Calabi, Extremal Kähler metrics, *Seminar on Differential Geometry* (S. T Yau, ed.), Princeton, 1982.
- [30] G. Calvaruso, Homogeneous structures on three-dimensional Lorentzian manifolds, J. *Geom. Phys.* 57 (2007), 1279–1291.

- [31] G. Calvaruso, Four-dimensional paraKähler Lie algebras: classification and geometry, *Houston J. Math.* **41** (2015), 733–748.
- [32] G. Calvaruso, A complete classification of four-dimensional paraKähler Lie algebras, *Complex Manifolds* **2** (2015), 1–10.
- [33] G. Calvaruso and M. Castrillón-López, Cyclic Lorentzian Lie groups, Geom. Dedicata 181 (2016), 119–136.
- [34] G. Calvaruso and M. Castrillón-López, *Pseudo-Riemannian homogeneous structures*, Developments in Mathematics 59, Springer, Cham, 2019.
- [35] G. Calvaruso, R. Storm, and J. van der Veken, Parallel and totally geodesic hypersurfaces of non-reductive homogeneous four-manifolds, *Math. Nachr.* **293** (2020), 1707–1729.
- [36] G. Calvaruso and A. Zaeim, Four-dimensional Lorentzian Lie groups, *Differential Geom. Appl.* **31** (2013), 496–509.
- [37] E. Calviño-Louzao, M. Ferreiro-Subrido, E. García-Río, and R. Vázquez-Lorenzo, Homogeneous four-manifolds with half-harmonic Weyl curvature, *Forum Math.* 34 (2022), 1497–1505.
- [38] E. Calviño-Louzao, M. Ferreiro-Subrido, E. García-Río, and R. Vázquez-Lorenzo, Homogenous Riemannian structures in dimension three, *Rev. Real Acad. Cienc. Exactas Fis. Nat. Ser. A-Mat.* **117** (2023), Paper no. 70, 11 pp.
- [39] E. Calviño-Louzao, X. García-Martínez, E. García-Río, I. Gutiérrez-Rodríguez, and R. Vázquez-Lorenzo, Conformally Einstein and Bach-flat four-dimensional homogeneous manifolds, J. Math. Pures Appl. (9) 130 (2019), 347–374.
- [40] E. Calviño-Louzao, E. García-Río, P. Gilkey, I. Gutiérrez-Rodríguez, and R. Vázquez-Lorenzo, Affine surfaces which are Kähler, para-Kähler, or nilpotent Kähler, *Results Math.* 73 (2018), Art. 135, 24 pp.
- [41] E. Calviño-Louzao, E. García-Río, P. Gilkey, and R. Vázquez-Lorenzo, The geometry of modified Riemannian extensions, *Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci.* 465 (2009), 2023–2040.
- [42] E. Calviño-Louzao, E. García-Río, I. Gutiérrez-Rodríguez, and R. Vázquez-Lorenzo, Fourdimensional homogeneous Kähler Ricci solitons, *Contemp. Math.* 775 (2021), 31–39.
- [43] E. Calviño-Louzao, E. García-Río, M. Sixto-Neira, and E. Vázquez-Abal, Biharmonic maps on tangent and cotangent bundles, J. Geom. Phys 101 (2016), 1–10.
- [44] E. Calviño-Louzao, E. García-Río, M. E. Vázquez-Abal, and R. Vázquez-Lorenzo, Local rigidity and nonrigidity of symplectic pairs, Ann. Global Anal. Geom. 41 (2012), 241–252.

- [45] E. Calviño-Louzao, E. García-Río, M. E. Vázquez-Abal, and R. Vázquez-Lorenzo, Geometric properties of generalized symmetric spaces, *Proc. Roy. Soc. Edinburgh Sect. A* 145 (2015), 47–71.
- [46] X. Cao and H. Tran, The Weyl tensor of gradient Ricci solitons, *Geom. Topol.* **20** (2016), 389–436.
- [47] G. Catino, P. Mastrolia, D. D. Monticelli, and M. Rigoli, Analytic and geometric properties of generic Ricci solitons, *Trans. Amer. Math. Soc.* 368 (2016), 7533–7549.
- [48] Q. Chen and M. Zhu, On rigidity of gradient Kähler-Ricci solitons with harmonic Bochner tensor, Proc. Amer. Math. Soc. 140 (2012), 4017–4025.
- [49] B. Chow, S.-C. Chu, D. Glickenstein, Ch. Guenther, J. Isenberg, T. Ivey, D. Knopf, P. Lu, F. Luo, and L. Ni, The Ricci flow: techniques and applications. Part I. Geometric aspects. *Mathematical Surveys and Monographs*, 135. American Mathematical Society, Providence, RI, 2007.
- [50] B. Y. Chu, Symplectic homogeneous spaces, Trans. Amer. Math. Soc. 197 (1974), 145–159.
- [51] R. S. Clark and M. R. Bruckheimer, Sur les structures presque tangentes, *C. R. Acad. Sci. Paris* **251** (1960), 627–629.
- [52] R. S. Clark and M. R. Bruckheimer, Tensor structures on a differentiable manifold, Ann. Mat. Pura Appl., (IV), 65 (1961), 123–141.
- [53] D. A. Cox, J. Little, and D. O'Shea, *Ideals, Varieties, and Algorithms. An Introduction to Computational Algebraic Geometry and Commutative Algebra*, 4th ed., Undergrad. Texts Math., Springer, Cham, 2015.
- [54] J. E. D'Atri, Codazzi tensors and harmonic curvature for left invariant metrics, *Geom. Dedicata* **19** (1985), 229–236.
- [55] V. de Smedt and S. Salamon, Anti-self-dual metrics on Lie groups, *Contemp. Math.*, **308** (2002), 63–75.
- [56] A. Derdziński, Self-dual Kähler manifolds and Einstein manifolds of dimension four, *Compositio Math.* 49 (1983), 405–433.
- [57] A. Derdziński, Connections with skew-symmetric Ricci tensor on surfaces, *Results Math.* 52 (2008), 223–245.
- [58] L. di Cerbo, Generic properties of homogeneous Ricci solitons, Adv. Geom. 14 (2014), 225–237.
- [59] S. Dragomir and L. Ornea, Locally conformally Kähler geometry, *Progress in Mathematics*, vol. 155, Birkhäuser Boston, In., Boston, MA, 1998.

- [60] M. Dunajski, Null Kähler geometry and isomonodromic deformations, *Commun. Math. Phys.* **391**, 77-105 (2022).
- [61] M. Dunajski and M. Przanowski, *Null-Kähler structures, symmetries and integrability*, Topics in mathematical physics, general relativity and cosmology in honor of Jerzy Plebański. Proceedings of the 2002 international conference, Cinvestay, Mexico City, Mexico, September 17–20, 2002. Hackensack, NJ: World Scientific. 147-155 (2006).
- [62] L. P. Eisenhart, Non-Riemannian Geometry, American Math. Soc. Colloquium publications, 8 (1927).
- [63] H. A. Eliopoulos, On the general theory of differentiable manifolds with almost tangent structure, *Canadian Math. Bull.*, **8** (1965) 721–748.
- [64] J.-C. Faugére, A new efficient algorithm for computing Gröbner bases (F4), *J. Pure Appl. Algebra* **139** (1999), 61–88.
- [65] J.-C. Faugére, A new efficient algorithm for computing Gröbner bases (F5), *Proceedings of the 2002 International Symposium on Symbolic and Algebraic Computation* (2002), 75–83.
- [66] M. E. Fels and A. G. Renner, Non-reductive Homogeneous Pseudo-Riemannian Manifolds of Dimension Four, *Canad. J. Math.* **58** (2) (2006), 282–311.
- [67] M. Ferreiro-Subrido, *Bochner-flat para-Kähler surfaces*, Developments in Lorentzian Geometry. GELOMA 2021. Springer Proceedings in Mathematics and Statistics 389 (2022), 81–92.
- [68] M. Ferreiro-Subrido, E. García-Río, and R. Vázquez-Lorenzo, Locally conformally flat Kähler and para-Kähler manifolds, *Ann. Global Anal. Geom.* 59 (2021), 483–500.
- [69] M. Ferreiro-Subrido, E. García-Río, and R. Vázquez-Lorenzo, Ricci solitons on fourdimensional Lorentzian Lie groups, *Anal. Math. Phys.* 12, (2022), 35 pp.
- [70] M. Ferreiro-Subrido, E. García-Río, and R. Vázquez-Lorenzo, Geometry of fourdimensional Kähler and para-Kähler Lie groups, to appear.
- [71] M. Ferreiro-Subrido, E. García-Río, and R. Vázquez-Lorenzo, Four-dimensional Riemannian algebraic solitons associated to geometric flows, to appear.
- [72] A. Fino, Almost K\u00e4hler 4-dimensional Lie groups with J-invariant Ricci tensor, Differential Geom. Appl. 23 (2005), 26–37.
- [73] P. M. Gadea, J. C. González-Dávila, and J. A. Oubiña, Cyclic homogeneous Riemannian manifolds, Ann. Mat. Pura Appl. (4) 195 (2016), 1619–1637.
- [74] P. M. Gadea and A. Montesinos Amilibia, Spaces of constant para-holomorphic sectional curvature, *Pacific J. Math.* 136 (1989), 85–101.

- [75] P. M. Gadea and J. A. Oubiña, Homogeneous Riemannian structures on Berger 3-spheres, *Proc. Edinburgh Math. Soc.* **48** (2005), 375–387.
- [76] E. García-Río, P. Gilkey, and S. Nikcevic, Homogeneity of Lorentzian three-manifolds with recurrent curvature, *Math. Nachr.* 287 (2014), 32–47.
- [77] E. García-Río and R. Vázquez-Lorenzo, Four-dimensional Osserman symmetric spaces, *Geom. Dedicata* **88** (2001), 147–151.
- [78] W. Globke and Th. Leistner, Locally homogeneous pp-waves, J. Geom. Phys. 108 (2016), 83–101.
- [79] A. R. Gover and P. A. Nagy, Four-dimensional conformal C-spaces, Q. J. Math. 58 (2007), 443–462.
- [80] A. Gray and L. Vanhecke, Riemannian geometry as determined by the volumes of small geodesic balls, *Acta Math.* 142 (1979), 157–198.
- [81] E. Griffin, Gradient ambient obstruction solitons on homogeneous manifolds, *Ann. Global Anal. Geom.* **60** (2021), 469–499.
- [82] B. Guilfoyle and W. Klingenberg, An indefinite Kähler metric on the space of oriented lines, J. London Math. Soc. (2) 75 (2005), 497–509.
- [83] R. S. Hamilton, Three-manifolds with positive Ricci curvature, J. Differential Geom. 17 (1982), 255–306.
- [84] D. Helliwell, Bach flow on homogeneous products, *Symmetry, Integrability and Geometry: Methods and Applications (SIGMA)*, **16** (2020), Paper no. 027, 35 pp.
- [85] P. T. Ho, Bach Flow, J. Geom. Phys. 133 (2018), 1-9.
- [86] C. S. Houh, On a Riemannian manifod M_{2n} with an almost tangent structure, *Canadian Math. Bull.*, **12** (1969) 759–769.
- [87] W. Jelonek, Compact Kähler surfaces with harmonic anti-self-dual Weyl tensor, *Differential Geom. Appl.* 16 (2002), 267–276.
- [88] G. R. Jensen, Homogeneous Einstein spaces of dimension four, J. Differential Geom. 3 (1969), 309–249.
- [89] J. Kim, On Kähler manifolds with harmonic Bochner curvature tensor, *Ann. Global Anal. Geom.* **35** (2009), 339–343.
- [90] A. W. Knapp, *Lie groups beyond an introduction*, 2Ed, Progress Math. 140, Birkhäuser, Boston, MA, 2002.
- [91] O. Kowalski, On regular curvature structures, Math. Z. 125 (1972), 129–138.

- [92] O. Kowalski, Z. Dusek, and B. Opozda, On locally nonhomogeneous pseudo-Riemannian manifolds with locally homogeneous Levi-Civita connections, *Internat. J. Math.* 14 (2003), 559–572.
- [93] O. Kowalski and F. Tricerri, Riemannian manifolds of dimension ≤ 4 admitting a homogeneous structure of class \mathcal{T}_2 , *Confer. Sem. Mat. Univ. Bari* **222** (1987), 24 pp.
- [94] C. N. Kozameh, E. T. Newman, and K. P. Tod, Conformal Einstein spaces, *Gen. Relativ. Gravit.* **17** (1985) 343–352.
- [95] W. Kühnel, *Differential geometry. Curves-surfaces-manifolds*, Student Mathematical Library, vol. 77, American Mathematical Society, Providence, RI, 2015.
- [96] R. S. Kulkarni, Curvature and metric, Ann. of Math. (2) 91 (1970), 311–331.
- [97] J. Lauret, Ricci soliton homogeneous nilmanifolds, Math. Ann. 319 (2001), 715–733.
- [98] J. Lauret, Ricci soliton solvmanifolds, J. Reine Angew. Math. 650 (2011), 1–21.
- [99] C. LeBrun, Bach-flat Kähler surfaces, J. Geom. Anal. 30 (2020), 2491–2514.
- [100] Th. Leistner, Conformal holonomy of C-spaces, Ricci-flat, and Lorentzian manifolds, *Differential Geom. Appl.* **24** (2006), 458–478.
- [101] M. W. Mansouri and A. Oufkou, The classification of left-invariant para-Kähler structures on simply connected four-dimensional Lie groups, *Complex Manifolds* 9 (2022), 1–17.
- [102] Y. Matsushita, Walker 4-manifolds with proper almost complex structures, *J. Geom. Phys.* 55 (2005), 385–398.
- [103] W. H. Meeks III and J. Pérez, Constant mean curvature surfaces in metric Lie groups, *Geometric analysis: partial differential equations and surfaces*, 25–110, Contemp. Math., 570, Amer. Math. Soc., Providence, RI, 2012.
- [104] J. Milnor, Curvatures of left invariant metrics on Lie groups, Adv. Math. 21 (1976), 293–329.
- [105] A. Naber, Noncompact shrinking four solitons with nonnegative curvature, J. Reine Angew. Math. 645 (2010), 125–153.
- [106] R. Narasimhan and Y. Nievergelt, *Complex analysis in one variable*, Second edition. Birkhäuser Boston, Inc., Boston, MA, 2001.
- [107] B. L. Neto, Generalized quasi-Einstein manifolds with harmonic anti-self dual Weyl tensor, Arch. Math. (Basel) 106 (2016), 489–499.
- [108] A. Newlander and L. Nirenberg, Complex analytic coordinates in almost complex manifolds, Ann. of Math. (2) 65 (1957), 391–404.

- [109] L. Nicolodi, Strutture omogenee autodualie antiautoduali, *Bolletino U.M.I.* (7) **3**-B (1989), 789-800.
- [110] K. Nomizu and T. Sasaki, *Affine differential geometry*, Cambridge Tracts in Mathematics, vol. 111, Cambridge University Press, Cambridge, 1994, Geometry of affine immersions.
- [111] Z. Olzak, On conformally flat parakählerian manifolds, *Math. Balkanica* **5** (1991), 302–307.
- [112] B. O'Neil, Semi-Riemannian geometry with applications to relativity, Pure and Applied Mathematics, vol. 103, Academic Press, Inc. [Harcourt Brace Jovanovich, Publishers], New York, 1983.
- [113] B. Opozda, Locally symmetric connections on surfaces, *Results Math.* 20 (1991), 725–743.
- [114] T. Otero-Casal, *Métricas de Einstein en grupos de Lie lorentzianos*, Publicaciones del Departamento de Geometría y Topología, Universidad de Santiago de Compostela.
- [115] G. Ovando, Four dimensional symplectic Lie algebras, *Beiträge Algebra Geom.* **47** (2006), 419–434.
- [116] G. Ovando, Invariant pseudo-Kähler metrics in dimension four, *J. Lie Theory* **16** (2006), 371–391.
- [117] A. M. Pastore, On the homogeneous Riemannian structures of type $\mathcal{T}_1 \oplus \mathcal{T}_3$, Geom. Dedicata **30** (1989), 235–246.
- [118] A. M. Pastore and F. Verroca, Some results on the homogeneous Riemannian structures of class $\mathcal{T}_1 \oplus \mathcal{T}_2$, *Rend. Mat. Appl.* (7) **11** (1991), 235–246.
- [119] E. M. Patterson, Symmetric Kähler spaces, J. Lond. Math. Soc. 30 (1955), 286–291.
- [120] E. M. Patterson and A. G. Walker, Riemannian extensions, *Quart. J. Math., Oxford Ser.* (2) 3 (1952), 19–28.
- [121] P. Petersen and W. Wylie, On gradient Ricci solitons with symmetry, *Proc. Amer. Math. Soc.* **137** (2009), 2085–2092.
- [122] P. Petersen and W. Wylie, Rigidity of homogeneous gradient soliton metrics and related equations, *Differential Geom. Appl.* **84** (2022), Paper No. 101929, 29 pp.
- [123] F. Podestà and A. Spiro, Four-dimensional Einstein-like manifolds and curvature homogeneity, *Geom. Dedicata* 54 (1995), 225–243.
- [124] S. Rahmani, Métriques de Lorentz sur les groupes de Lie unimodulaires, de dimension trois, *J. Geom. Phys.* **9** (1992), 295–302.

- [125] H. S. Ruse, A. G. Walker, and T. J. Willmore, *Harmonic spaces*, Consiglio Nazionale delle Ricerche Monografie Matematiche, 8 Edizioni Cremonese, Rome 1961.
- [126] K. Sekigawa, On some 3-dimensional curvature homogeneous spaces, *Tensor (N.S.)* 31 (1977), 87–97.
- [127] G. Székelyhidi, An Introduction to Extremal Kähler Metrics, Graduate Studies in Mathematics 152, American Mathematicas Society, Province, Rhode Island, 1981.
- [128] H. Takagi, Conformally flat Riemannian manifolds admitting a transitive group of isometries, *Tohoku Math. J.* (2) 27 (1975), 103–110.
- [129] S. Tanno, 4-dimensional conformally flat Kähler manifolds, *Tohoku Math. J.* **24** (1972), 501–504.
- [130] F. Tricerri and L. Vanhecke, *Homogeneous structures on Riemannian manifolds*, London Math. Soc. Lecture Notes Series 83, Cambridge University Press, London, 1983.
- [131] I. Vaisman, Locally conformally symplectic manifolds, *Internat. J. Math. Math. Sci.* 8 (1985), 521–536.
- [132] A. G. Walker, Canonical form for a Riemannian space with a parallel field of null planes, *Quart. J. Math.* Oxford Ser. (2) 1 (1950), 69–79.
- [133] Th. H. Wears, On algebraic solitons for geometric evolution equations on threedimensional Lie groups, *Tbilisi Math. J.* 9 (2016), 33–58.
- [134] J.-Y. Wu, P. Wu, and W. Wylie, Gradient shrinking Ricci solitons of half harmonic Weyl curvature, *Calc. Var. Partial Differential Equations* **57** (2018), Paper No. 141, 15 pp.
- [135] Z. Yan and S. Deng, Double extensions on Riemannian Ricci nilsolitons, *J. Geom. Anal.* 31 (2021), 9996–10023.